Representations and invariant equations of E(3)

Mayer Humi

Department of Mathematical Sciences, Worcester Polytechnic Institute, Worcester, Massachusetts 01609

(Received 5 May 1987; accepted for publication 5 August 1987)

Using methods analogous to those introduced by Gel'fand *et al.* [Representations of the Rotation and Lorentz Groups and Their Applications (Pergamon, New York, 1963)] for the Lorentz group the matrix elements for the representations of the Lie algebra of the Euclidean group in three dimensions E(3) are explicitly derived. These results are then used to construct invariant equations with respect to this group and to show, in particular, that the nonrelativistic analog to the Dirac equation is not unique.

I. INTRODUCTION

The Euclidean group in three dimensions E(3) plays in classical mechanics the same role that the Lorentz group plays in relativistic mechanics.¹⁻³ Moreover, from a mathematical point of view, the Lie algebras of these two groups contain six generators J_i , K_i , 1 = 1,2,3, whose commutation relations (CR) for the Lorentz algebra are

$$[J_i, J_j] = \epsilon_{ijk} J_k, \tag{1.1}$$

$$[J_i, K_j] = \epsilon_{ijk} K_k, \tag{1.2}$$

$$[K_i, K_j] = -\epsilon_{ijk} J_k. \tag{1.3}$$

The CR of the Lie algebra of E(3) differ from those above only in the third set [Eq. (1.3)], which is replaced by

$$[K_i, K_j] = 0. (1.4)$$

[For more details regarding the generators of E(3) and the notations used for the rest of this paper see the Appendix.]

In view of these similarities it is surprising to find that the representations of these algebras and their respective invariant equations were treated rather differently in the literature.

Thus while complete and explicit expressions for the matrix elements of the irreducible representations of the Lorentz algebra and its invariant equations exist in the literature⁴ the same is not true for E(3). As for this latter algebra the "general form" of the matrix elements for the finite (indecomposable) irreducible representations were given (somewhat) indirectly in Ref. 5. What has been left open, however, was the computation of the interlocking constants between the various irreducible representations of O(3) that appear in the decomposition of an E(3) representation (especially when can these interlocking constants be nonzero). These constants can be computed by *ad hoc* methods (as suggested in Ref. 5) only in some very simple cases.

In view of these circumstances it is our first objective in this paper (Sec. II) to derive explicitly, the expressions for the matrix elements for some of the irreducible representations of E(3) (both finite and infinite dimensional) by methods that are completely analogous to those used by Gel'fand *et al.* for the Lorentz group.⁴

In Sec. III we present a systematic approach to the construction of first-order invariant equations with respect to E(3) subject to the constraint that each component of the wave function satisfies the Galilean energy momentum relation $2mE = m^2 + p^2$ (Refs. 6-8). As a result we show that the nonrelativistic analog to the Dirac equation is not unique and that there is no nonrelativistic analog to the Majorana equation.⁹

II. REPRESENTATIONS OF E(3)

To construct the representations of E(3) we first observe that Gel'fand *et al.*⁴ already found the most general solution of the CR(1.1), (1.2). Adopting the notations used by Gel'fand *et al.* we can write this solution as follows:

$$H_{3}\xi_{l,m}^{\tau} = m\xi_{l,m}^{\tau}, \qquad (2.1)$$

$$H_{+}\xi_{l,m}^{\tau} = \alpha_{m+1}^{l}\xi_{l,m+1}^{\tau}, \qquad (2.2)$$

$$H_{-\xi_{l,m}}^{\tau} = \alpha_{m}^{l} \xi_{l,m-1}^{\tau}, \qquad (2.3)$$

$$F_{3}\xi_{l,m}^{\tau} = \sum_{\tau'} \left\{ C_{l}^{\tau\tau'} d(l,m) \xi_{l-1,m}^{\tau} - A_{l}^{\tau\tau'} \times m \xi_{l,m}^{\tau} - C_{l+1}^{\tau\tau'} d(l+1,m) \xi_{l+1,m}^{\tau'} \right\},$$
(2.4)

$$F_{+}\xi_{l,m}^{\tau} = \sum_{\tau'} \left\{ C_{l}^{\tau\tau'}g_{1}(l,m)\xi_{l-1,m+1}^{\tau'} - A_{l}^{\tau\tau'}g_{2}(l,m)\xi_{l,m+1}^{\tau'} + C_{l+1}^{\tau\tau'}g_{3}(l,m)\xi_{l+1,m+1}^{\tau} \right\}, \qquad (2.5)$$

$$F_{-}\xi_{l,m}^{\tau} = \sum_{\tau'} \left\{ -C_{l}^{\tau\tau'}g_{1}(l,-m)\xi_{l-1,m-1}^{\tau'} -A_{l}^{\tau\tau'}g_{2}(l,-m)\xi_{l,m-1}^{\tau'} -C_{l+1}^{\tau\tau'}g_{3}(l,-m)\xi_{l+1,m-1}^{\tau'} \right\}, \qquad (2.6)$$

where

$$\alpha_m^l = [(l+m)(l-m+1)]^{1/2},$$

$$d(l,m) = [l^2 - m^2]^{1/2},$$

$$g_1(l,m) = [(l-m)(l-m-1)]^{1/2},$$

$$g_2(l,m) = [(l-m)(l+m+1)]^{1/2},$$

$$g_3(l,m) = [(l+m+1)(l+m+2)]^{1/2}.$$

It is therefore clear that the only step one must take in order to construct the representations of E(3), rather than O(3,1)is to subject this general solution to the constraints given by Eqs. (1.4) rather than those of Eqs. (1.3). To carry this program out we begin by constructing two special types of representations that are important from a physical point of view. There are the following. (1) Representations of E(3) whose decomposition with respect to O(3) contains each irreducible representation of O(3) at most once.

(2) Representations of E(3) in whose decomposition with respect to O(3) only one irreducible representation of this group appears (several times).

Case 1: Since each representation of O(3) appears only once in the decomposition we can drop the degeneracy index τ .

Furthermore, we observe that if the representation is irreducible and l_0 , l_1 are, respectively, the lowest and highest l's that appear in the decomposition then each $l_0 + n$, $n = 1,2,...,l_1 - l_0$ must also appear in the decomposition. In fact, if some representation $l_0 + n$ is missing then we can infer [since the generators of the algebra, according to (2.1)-(2.5), can connect states of $l_0 + n - 1$ only with those of $l_0 + n$ and $(l_0 + n - 2)$] that the representations $l_0,...,l_0 + n - 1$ form an invariant subspace in contradiction to the assumption that the representation is irreducible.

Thus an irreducible representation of E(3) that belongs to this class is a "ladder representation" which is either finite or infinite (presently we show that the ladder must be infinite).

To construct these irreducible representations we apply any of the constraints $[F_+, F_3] = [F_+, F_-] = [F_-, F_3] = 0$ to Eqs. (2.4)-(2.6) under the present assumptions and obtain the following equations:

$$[A_{l}(l+1) - A_{l-1}(l-1)]C_{l} = 0, \qquad (2.7)$$

$$[A_{l+1}(l+2) - A_l l]C_{l+1} = 0, (2.8)$$

$$[2l-1]C_l^2 - [2l+3]C_{l+1}^2 - A_l^2 = 0.$$
 (2.9)

To solve these equations we first observe [using Eqs. (2.4)–(2.6)] that for an irreducible representation $C_{l+1} = 0$, only if $l = l_1$ (if such a finite l_1 exists) and similarly $C_l = 0$ only if $l = l_0$. Hence for other *l*'s that appear in the decomposition of the representation with respect to O(3), both C_l, C_{l+1} are nonzero and we infer from Eqs. (2.7) and (2.8) that

$$A_l = A_{l-1}(l-1)/(l+1),$$

which leads to

$$A_{l} = A_{l_{0}} [l_{0}(l_{0}+1)/l(l+1)], \qquad (2.10)$$

where A_{l_0} is an arbitrary constant. Furthermore, to complete the analogy with the Lorentz group we rewrite (2.10) in the form

$$A_l = i l_0 L / l(l+1). \tag{2.11}$$

To evaluate the C_l 's we now multiply Eq. (2.9) by (2l + 1)and sum the resulting equations for $l = l_0, ..., l$. This yields, after some algebra,

$$C_{l}^{2} = L^{2} \left[\left(l^{2} - l_{0}^{2} \right) / (4l^{2} - 1)l^{2} \right].$$
 (2.12)

We infer from this relation that contrary to the Lorentz group a ladder representation of E(3) must be an infinitedimensional representation. In fact (2.12) implies that either $C_l = 0$ for all l (the representation is then reducible) or $C_l \neq 0$ for all $l_0 + n$, n = 1, 2, ... We thus proved the following.

Proposition 1: All irreducible ladder representations of E(3) are of infinite dimension (except the trivial one-dimensional representation). The matrix elements of these repre-

sentations are given by Eqs. (2.1)-(2.6), where A_i , C_i are determined by Eqs. (2.11), (2.12), respectively.

It appears to us that the explicit computation of these matrix elements is new.

Case 2: When the decomposition of the representation under consideration with respect to O(3) contains only one irreducible representation l of this group with multiplicity n, then obviously the matrix elements of J_i are given by the Kronecker product

$$J_i = I \times J_i(l), \tag{2.13}$$

where $J_i(l)$ are the matrix elements of the irreducible representation l of O(3) and I is the unit matrix of dimension n. The matrix elements of the other generators of the algebra are given by the following proposition.

Proposition 2: Under the present assumption the matrix elements of K_i are given by

$$K_i = N \times J_i(l), \tag{2.14}$$

where N is an $n \times n$ matrix so that $N^2 = 0$. Moreover, if N admits an invariant (proper) subspace then the representation is reducible otherwise the representation is indecomposable.¹⁰

Proof: It is easy to show that if J_i , K_i are given by Eqs. (2.13) and (2.14) then all the CR of the algebra are satisfied. In fact,

 $[J_i, K_j] = N \times [J_i(l), J_j(l)] = N \times \epsilon_{ijk} J_k(l) = \epsilon_{iik} K_k$

and

$$[K_i,K_j] = N^2 \times \epsilon_{ijk} J_k = 0.$$

To show that this is the only possible solution of the CR under present assumptions it is enough to observe that the matrix elements of K_3 are [using (2.4)]

$$(K_3)_{l,l,m,m'}^{\tau\tau'} = A^{\tau\tau'} m \delta_{m,m'}, \qquad (2.15)$$

i.e., $K_3 = N \times J_3(l)$.

An appropriate form of the matrix N which is important in the construction of E(3) invariant equations is given by

$$\mathbf{N} = \begin{bmatrix} 0 & & & \\ \alpha_2 & & 0 & \\ \vdots & 0 & \ddots & \\ \alpha_{n-1} & & & \\ \epsilon, \beta_2 & & \cdots & \beta_{n-1} & 0 \end{bmatrix}, \quad (2.16)$$

where $\alpha_2 \beta_2 + \cdots + \alpha_{n-1} \beta_{n-1} = 0$ and ϵ is an arbitrary parameter.

We now proceed to discuss the irreducible representations of E(3) in the general case, i.e., when the decomposition of the representation with respect to O(3) contains the representations $l_0l_0 + 1...,l_1$, each with multiplicity n_i . (Obviously such a representation is irreducible only if all the representations $l_0 + n$, $n = 0,...,l_1 - l_0$ appear at least once in the decomposition.) To construct the matrix elements of F_+ , F_- , and F_3 in this case we rewrite Eqs. (2.4)-(2.6) in block matrix form:

$$F_{3} = \begin{bmatrix} N_{l_{0},l_{0}} \times H_{3}(l_{0}), & N_{l_{0},l_{0}+1} \times D(l_{0}), & 0, & \cdots \\ N_{l_{0}+1,l} \times D(l_{0}), & N_{l_{0}+1,l_{0}+1} \times H_{3}(l_{0}+1), & N_{l_{0}+1,l_{0}+2} \times D(l_{0}+1), & 0 \cdots \\ 0, & N_{l_{0}+2,l_{0}+1} \times D(l_{0}+1), & N_{l_{0}+2,l_{0}+2} \times H_{3}(l_{0}+2), & 0 \cdots \\ & & \cdots \end{bmatrix},$$
(2.17)

where the $N_{l_0 l_i}$ are matrices of dimension $n_i \times n_j$ and

$$[D(l)]_{mm'} = \delta_{mm'} \sqrt{(l+1)^2 - m^2}.$$
 (2.18)

Similar expressions can be written for F_+ , F_- . The CR (1.4) now reduce to constraints on the matrices $N_{I'_{i},I_i}$ in the form

$$N_{l_0,l_0} = N_{l_0,l_0+1}N_{l_0+1,l_0} = N_{l_0,l_0}N_{l_0,l_0+1} = \dots = 0,$$
(2.19)

which is, in general, a redundant system of equation for the matrix elements of these matrices.

A particularly interesting case is obtained when the multiplicity of each l in the representation is the same, i.e., $n_i = n$, $i = 0,...,l_1 - l_0$. In this "degenerate" case all the equations of (2.19) can be satisfied if we choose

 $N_{l,l} = N_{l,l \pm 1} = M$, where $M^2 = 0$.

III. INVARIANT EQUATIONS

The study of E(3) invariant equations was initiated by Levi-Leblond⁶ who constructed by *ad hoc* methods a nonrelativistic analog of the Dirac equation and studied its properties. In this section, however, we study the construction of such equations from a covariant point of view⁷ and especially consider the uniqueness of the nonrelativistic analog of the Dirac equation.

Definition 1: Let an operator realization of p_{λ} be given (see the Appendix). We say that the equation

$$(L_{\lambda}p^{\lambda} + \widetilde{m})\psi = 0, \quad \psi \in \mathbb{R}^{n}$$

$$(3.1)$$

is invariant with respect to E(3) if $L_{\lambda}p^{\lambda}$ is a scalar of the E(3) algebra, viz.,

$$\left[L_{\lambda}p^{\lambda},J_{\mu\nu}\right] = 0. \tag{3.2}$$

We observe that this definition is equivalent to the one introduced in Ref. 4 from a group theoretical point of view for the Lorentz group.

At this point on objection might be raised to the effect that *m* appears in (3.1) twice, once as a scalar and once as an operator (albeit as a scalar multiplication operator). The answer to this is that this dichotomy is inherent to the definition of mass in nonrelativistic mechanics. In fact *m* appears as a scalar of the (pure) Galileo group and then is added to the group as an operator through a central extension.⁶ In any event since *m* is constant both as a scalar and as an operator the equations under consideration describe entities with constant mass. Furthermore, by imposing proper constraints on L^{μ} one can insure (see Proposition 3) that each component of ψ satisfies the Galilean energy-momentum relation. Thus it is appropriate to refer to equations of the form (3.1) as nonrelativistic invariant wave equations.

Corollary 7: Equation (3.1) is invariant with respect to E(3) if and only if

$$\begin{bmatrix} L_{\lambda}, J_{\mu\nu} \end{bmatrix} = i [g_{\lambda\mu} L_{\lambda} - g_{\lambda\nu} L_{\mu}] .$$
(3.3)

Proof: From (3.2) we infer

$$0 = [L^{\lambda} p_{\lambda} J_{\mu\nu}] = L^{\lambda} [p_{\lambda} J_{\mu\nu}] + [L_{\lambda} J_{\mu\nu}] p^{\lambda}$$
$$= iL^{\lambda} [g_{\lambda\mu} p_{\nu} - g_{\lambda\nu} p_{\mu}] + [L_{\lambda} J_{\lambda\nu}] p^{\lambda}$$
$$= i[g_{\nu\lambda} L_{\mu} - g_{\mu\lambda} L_{\nu}] p^{\lambda} + [L_{\lambda} J_{\mu\nu}] p^{\lambda},$$

from which Eq. (3.3) follows.

Furthermore, to insure the physical meaning of such an invariant equation we require that each component of ψ satisfies the Galilean energy-momentum relation

$$E = m + p^2/2m,$$
 (3.4)

which leads to the following result.

Proposition 3: Each component of ψ satisfies Eq. (3.4) if

$$\{L_{\mu}, L_{\nu}\} = L_{\mu}L_{\nu} + L_{\nu}L_{\mu} = 2g_{\mu\nu}I.$$
(3.5)

Proof: This result is obvious if we multiply (3.1) by $(L^{\mu}p_{\mu} - \tilde{m})$ and require that (3.4) is satisfied. Observe, however, that since $g_{\mu\nu}$ is not diagonal Eq. (3.5) implies

$$L_4^2 = 0, \quad \{L_4, L_5\} = 2I.$$
 (3.6)

We now turn our attention to finite-dimensional equations which describe nonrelativistic particles with definite spin, viz., equations based on finite-dimensional indecomposable representations of the form given by Eqs. (2.13), (2.14), and (2.16).

To begin with we infer from (3.3) that

$$[[L_3, H_+], H_-] = 2L_3. \tag{3.7}$$

Hence using the results in Ref. 4 regarding O(3) invariant equations we obtain

$$L_3 = i \cdot D \times \bar{J}_3(l) \tag{3.8}$$

and consequently

$$L_1 = i \cdot D \times \tilde{J}_1(l), \quad L_2 = i \cdot D \times \tilde{J}_2(l). \tag{3.9}$$

Furthermore, if we write L_4 , L_5 in block form then the CR's

$$[L_4, \tilde{J}_i] = [L_5, \tilde{J}_i] = 0 \tag{3.10}$$

imply, using Schur's lemma, that

$$L_4 = \lambda_4 \times I, \quad L_5 = \lambda_5 \times I,$$
 (3.11)

where D, λ_4 , λ_5 are $n \times n$ matrices. From Eq. (3.5) it then follows that

$$\{\lambda_4, \lambda_5\} = I, \quad \lambda_5^2 = -I, \quad D^2 \times \tilde{J}_i^2(l) = I \times I,$$

(3.12)

$$\lambda_{4}^{2} = \{\lambda_{4}, D\} = \{\lambda_{5}, D\} = 0.$$

Thus to complete the construction of the invariant equations under consideration we must solve for D, λ_4 , λ_5 using the remaining CR. However, from

$$[L_4, J_{34}] = 0, \quad [L_5, J_{34}] = -iL_3, \quad (3.13)$$

$$[L_3, J_{34}] = -iL_4, \tag{3.14}$$

$$[[L_5, F_+], F_-] = -2L_4, \qquad (3.15)$$

it follows that

$$\lambda_4 N - N \lambda_4 = 0, \qquad (3.16)$$

$$\lambda_5 N - N \lambda_5 = D, \qquad (3.17)$$

$$(ND - DN) \times \tilde{J}_3^2(l) = \lambda_4, \qquad (3.18)$$

and

$$N\lambda_5 N \times \{H_+(l), H_-(l)\} = 2\lambda_4 \times I.$$
 (3.19)

(The remaining CR lead to the same constraints on λ_4 , λ_5 , D.)

Proposition 4: Invariant equations of the form (3.1) based on the indecomposable representations (2.13) and (2.14) can be constructed only for $l = \frac{1}{2}$.

Proof: From Eqs. (3.18) and (3.19) it follows that to construct E(3) invariant equations we must have

$$\{H_+(l), H_-(l)\} = \alpha I, \quad \alpha, \beta \in \mathbb{R},$$

and

$$\tilde{J}_3^2 = \beta I. \tag{3.20}$$

However, this can be satisified only for $l = \frac{1}{2}$, where

$$\left\{H_{+}\left[\frac{1}{2}\right], H_{-}\left[\frac{1}{2}\right]\right\} = I, \quad \tilde{J}_{3}^{2} = \frac{1}{4}I.$$
(3.21)

We conclude from this proposition that invariant equations describing particles with spin $\neq \frac{1}{2}$ must be based on "mixed" indecomposable representations ($l \neq \text{const}$) and will contain, therefore, some extraneous components that must be eliminated by some subsidiary conditions (this is similar to the situation in relativistic mechanics).

Corollary: For $l = \frac{1}{2}$,

$$N\lambda_5 N = 2\lambda_4$$
, $[N,D] = 4\lambda_4$, $D^2 = 4I$. (3.22)

This is a direct consequence of (3.12), (3.18), (3.19), (3.21).

The nonrelativistic analog to the Dirac equation derived in Refs. 1 and 6 is based on a representation of E(3) in the form (2.13)-(2.15) with dim N=2. These equations are⁶

$$(\boldsymbol{\sigma} \cdot \boldsymbol{P})\boldsymbol{\varphi} + 2m\chi = 0, \quad (\boldsymbol{\sigma} \cdot \boldsymbol{P})\chi + \boldsymbol{E}\boldsymbol{\varphi} = 0,$$
 (3.23)

where φ , χ are two component functions and σ are Pauli matrices. We observe however that in these equations φ , χ satisfy the energy-momentum relation in the form $E = P^2/2m$ rather than Eq. (3.4). Hence E should be replaced by E - m to conform to our notation. Using this observation we can rewrite Eq. (3.23) in matrix form as

$$\begin{cases} (I \times \boldsymbol{\sigma}) \cdot \mathbf{P} + \begin{bmatrix} \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix} \times I \end{bmatrix} E \\ + m \begin{bmatrix} 0 & 2 \\ -1 & 0 \end{bmatrix} \times I \end{bmatrix} \begin{bmatrix} \varphi \\ \chi \end{bmatrix} = 0. \tag{3.24}$$

On the other hand, if we use Eqs. (3.12), (3.16)–(3.19), and (3.22) to determine a solution for D, λ_4 , λ_5 based on the same representation of E(3), we find

$$\lambda_4 = \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix}, \quad \lambda_5 = \begin{bmatrix} a & 2 \\ c & -a \end{bmatrix}, \quad D = \begin{bmatrix} 2 & 0 \\ \gamma & -2 \end{bmatrix}$$
(3.25)

subject to the constraints

$$a^2 + 2c = -1, \quad 2a + \gamma = 0.$$
 (3.26)

In matrix form the corresponding equations can be written as

$$\left\{ \left(iD \times \frac{\sigma_i}{2} \right) p^i + [\lambda_4 \times I] \widetilde{E} + [(\lambda_5 + I) \times I] \widetilde{m} \right\} \begin{bmatrix} \varphi \\ \chi \end{bmatrix} = 0, \qquad (3.27)$$

and it is easily verified that (3.27) is not equivalent to (3.24).

Furthermore, one can find solutions to Eqs. (3.12) and (3.16)-(3.19) based on higher-dimensional representations of E(3) as illustrated by the following proposition.

Proposition 5: Let N be a 4×4 matrix in the form (2.16) with $\epsilon = \alpha_2 = \alpha_3 = \beta_2 = -\beta_3 = 2$, then the matrices $\lambda_4 = N$ and

$$\lambda_{5} = \begin{bmatrix} 0 & 1 & 0 & 0 \\ -1 & 0 & 0 & 0 \\ \alpha & 1 & 0 & -1 \\ -1 & \alpha & 1 & 0 \end{bmatrix},$$

$$D = 2 \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ 0 & -2 & 1 & 0 \\ 2 + 2\alpha & 0 & 0 & -1 \end{bmatrix},$$
(3.28)

where α is an arbitrary parameter, are solutions of equations (3.12) and (3.16)–(3.19).

Thus we showed the existence of a new nonrelativistic analog to the Dirac equation and demonstrated that the number of components in such an equation is indeterminate from a purely nonrelativistic point of view.

We now turn to an attempt to construct a nonrelativistic analog to the Majorana equation based on the self-coupling of a ladder representation l_0 , $l_0 + 1$,... (case 1 of Sec. II).

To begin with we introduce an orthogonal basis $\xi_{l,m}$ on the representation space and set

$$L_{4}\xi_{l,m} = \sum D_{l,m;l',m'}\xi_{l',m'}.$$
(3.29)

However, since $[L_4, J_{\mu\nu}] = 0$ and the representation is irreducible it follows from Schur's lemma that

$$D_{l,m;l',m'} = \rho \delta_{ll'} \delta_{mm'}. \tag{3.30}$$

To calculate the other matrices L_{μ} we observe that it is sufficient to find L_5 . Setting

$$L_{5}\xi_{l,m} = \sum c_{l,m;l',m'}\xi_{l',m'}$$
(3.31)

we infer from $[L_5, H_3] = 0$ that

$$c_{l,m;l'm'} = c_{l,l',m} \delta_{mm'}.$$
 (3.32)

Furthermore, from $[L_5, H_{\pm}] = 0$ we deduce that

$$\alpha_{l,m+1}c_{l,l',m+1} - \alpha_{l',m+1}c_{l,l',m} = 0, \qquad (3.33)$$

$$\alpha_{l,m}c_{l,l',m-1} - \alpha_{l',m}c_{l,l',m} = 0.$$
(3.34)

By simple algebraic manipulations it then follows that

$$c_{l,m;l',m'} = c_l \delta_{ll'} \delta_{mm'}.$$
 (3.35)

To compute the c_l 's and ρ we now invoke the relation (3.15). By applying this relation to $\xi_{l,m}$ we obtain after a long algebra that $\rho = 0$ and $c_1 = c = \text{const}$, viz., $L_4 = 0$, $L_5 = cI$. Using the CR

$$[L_{5}, J_{i4}] = -iL_{i} \tag{3.36}$$

this implies that $L_i = 0$. We conclude then that the relativistic Majorana equation has no nonrelativistic analog.

APPENDIX: COMMUTATION RELATIONS FOR E(3)

In this appendix we describe the relationship between the different notations for the generators of the Lie algebra of the Galilean group G and its E(3) subgroup. We also give an explicit differential realization for the commutation relations (CR) of the extended (physical) Galilean group.⁶

1. Real basis^{1,6}

The Lie algebra of the (pure) Galilean group has ten generators which we denote by J_i , K_i , P_i , i = 1, 2, 3, and E. The J_i 's are the generators of the Lie algebras of O(3) the K_i 's are the boost generators, and P_i , E are the generators for the translations in space and time, respectively [E(3) is generated by J_i , k_i , i = 1, 2, 3].

The nonzero CR of G are

$$\begin{bmatrix} J_i, J_j \end{bmatrix} = \epsilon_{ijk} J_k, \quad \begin{bmatrix} J_i, K_j \end{bmatrix} = \epsilon_{ijk} K_k,$$

$$\begin{bmatrix} J_i, P_j \end{bmatrix} = \epsilon_{ijk} P_k, \quad \begin{bmatrix} K_i E \end{bmatrix} = P_i.$$
 (A1)

The CR for the extended ("physical") Lie algebras of G (which is obtained by a central extension) are the same as in (A1) except that

 $[K_i, P_i] = 0$

is replaced by

$$[K_i, P_i] = m\delta_{ij}. \tag{A2}$$

Thus the extended group has an additional generator m. However, m is a scalar of the algebra as it commutes with all other generators of G.

2. Complex basis

If we consider G over the complex numbers and define

$$\begin{split} \tilde{J}_j &= iJ_j, \quad \tilde{K}_j = iK_j, \quad iP_j, \quad \tilde{E} = iE, \\ j &= 1, 2, 3, \quad \tilde{m} = im, \end{split}$$
 (A3)

then the CR of G (extended) take the form

$$\begin{bmatrix} \tilde{J}_i, \tilde{J}_j \end{bmatrix} = i\epsilon_{ijk}\tilde{J}_k, \quad \begin{bmatrix} \tilde{J}_i, \tilde{K}_j \end{bmatrix} = i\epsilon_{ijk}\tilde{K}_k, \\ \begin{bmatrix} \tilde{J}_i, \tilde{P}_j \end{bmatrix} = i\epsilon_{ijk}P_k, \quad \begin{bmatrix} \tilde{K}_i, \tilde{E} \end{bmatrix} = iP_i, \quad \begin{bmatrix} \tilde{K}_i, \tilde{P}_i \end{bmatrix} = i\tilde{m}\delta_{ij}.$$
 (A4)

3. Covariant notation7.11

By introducing the nonsingular "metric"

$$g_{\alpha\beta} = \begin{bmatrix} -I & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 0 \\ 0 & 1 & -1 \end{bmatrix},$$
 (A5)

where I is the unit matrix in three dimensions and

$$J_{ij} = \epsilon_{ijk} \tilde{J}_k, \quad J_{i4} = \tilde{K}_i, p_i = \tilde{P}_i, \quad p_4 = \tilde{m}, \quad p_5 = \tilde{E} - \tilde{m},$$
(A6)

we can write the CR of the extended G in covariant notation

$$\begin{bmatrix} J_{\mu\nu}, J_{\rho\sigma} \end{bmatrix} = i \begin{bmatrix} g_{\nu\rho} J_{\mu\sigma} - g_{\mu\rho} J_{\nu\sigma} + g_{\mu\sigma} J_{\nu\rho} - g_{\nu\sigma} J_{\mu\rho} \end{bmatrix},$$
(A7)
(A7)

$$[p_{\mu},J_{\rho\sigma}] = i[g_{\mu\rho}p_{\sigma} - g_{\mu\sigma}p_{\rho}]$$
(A8)

(note that $g_{44} = 0$). We observe that the metric tensor $g_{\alpha\beta}$ can be used only to raise and lower indices of vectors over E(3) (Ref. 11), e.g., if $p_{\alpha'} = [p_{1,}p_{2,}p_{3,}\widetilde{m},\widetilde{E} - \widetilde{m}]$ then

$$p^{\alpha} = (-p_1, -p_2, -p_3, \widetilde{E}, \widetilde{m}).$$
 (A9)

Furthermore, $p_{\alpha}p^{\alpha} = \tilde{m}^2$ is the Galilean energy-momentum-mass relation.

A differential realization of the generators (A6) and their CR is given by

$$J_{ij} = i \left[x_j \frac{\partial}{\partial x_i} - x_i \frac{\partial}{\partial x_j} \right], \quad J_{i4} = i \left[t \frac{\partial}{\partial x_i} + m x_i \right],$$

$$p_i = -i \frac{\partial}{\partial x_i}, \quad p_4 = im, \quad p_5 = i \left[\frac{\partial}{\partial t} - m \right].$$
(A10)

4. A basis of raising and lowering operators

Since E(3) and the Lorentz group O(3,1) have the same number of generators (and six of the nine CR are the same) it is convenient to introduce a basis of E(3) similar to the one used by Gel'fand *et al.*⁴ to construct the representations and invariant equations of O(3,1). Thus in Gel'fand's notation

$$A_{12} = J_1, \quad A_{13} = J_2, \quad A_{23} = J_1, \quad B_i = K_i, \quad i = 1, 2, 3.$$
(A11)

Hence

$$\begin{split} H_{+} &= \tilde{J}_{1} + i \tilde{J}_{2}, \quad H_{-} = \tilde{J}_{1} - i \tilde{J}_{2}, \quad H_{3} = \tilde{J}_{3}, \\ F_{+} &= \tilde{K}_{1} + i \tilde{K}_{2}, \quad F_{-} = \tilde{K}_{1} - i \tilde{K}_{2}, \quad F_{3} = \tilde{K}_{3}. \end{split}$$
 (A12)

¹J.-M Levy-Leblond, *Galilei Group and Galilean Invariance in Group The*ory and its Applications, edited by E. M. Loebl (Academic, New York, 1971), Vol. II; this review article contains an extensive list of references; A. S. Wightman, Rev. Mod. Phys. **34**, 845 (1965).

- ²J. Voisin, J. Math. Phys. 6, 1519 (1965).
- ³W. I. Fushchich and R. M. Chemiha, J. Phys. A 18, 3491 (1985).
- ⁴I. M. Gel'fand, R. A. Minlos, and Z. Ya. Shapiro, *Representations of the Rotation and Lorentz Groups and Their Applications* (Pergamon, New York, 1963).
- ⁵C. George and M. Levi-Nahas, J. Math. Phys. 7, 980 (1966).
- ⁶J.-M Levy-Leblond, Commun. Math. Phys. **6**, 286 (1967); **12**, 64 (1969); J. Math. Phys. **4**, 776 (1963).
- ⁷M. Humi, J. Math. Phys. 10, 1897 (1969).
- ⁸H. P. Kunzle and C. Duval, Ann. Inst. H. Poincaré 41, 363 (1984); H. P.
- Kunzle and J. M. Nester, J. Math. Phys. 25, 1009 (1984).
- ⁹E. Majorana, Nuovo Cimento 9, 335 (1932).
- ¹⁰P. Furlan, V. B. Petkova, G. M. Sotkov, and I. T. Todorov, Rev. Nuovo Cimento 8, 3 (1985); I. M. Gel'fand and V. A. Ponomarev, Russ. Mat. Surv. 23, 1 (1968); C. Rendere and D. J. Griffiths, Phy. Rev. D 2, 317 (1970).

¹¹G. Pinski, J. Math. Phys. 9, 1927 (1968).

Weight-2 zeros of 3/ coefficients and the Pell equation

J. D. Louck and P. R. Stein

Theoretical Division, Los Alamos National Laboratory, Los Alamos, New Mexico 87545

(Received 16 June 1987; accepted for publication 30 July 1987)

All weight-2 zeros of the Wigner 3j coefficients may be obtained from the quadratic Diophantine equation known as Pell's equation. These zeros may then be classified by the orbits of a discrete, infinite-order subgroup of the Lorentz group SO(1,1). This is carried out by transforming the "polynomial part" of a weight-2 3j coefficient to Pellian form and obtaining the fundamental zeros numerically. The relation of this polynomial to a family of binary quadratic forms is also given, together with a discussion of the invariance group.

I. INTRODUCTION AND RÉSUMÉ

A number of papers¹⁻¹² have appeared recently on the subject of (nontrivial) zeros of the angular momentum coefficients known as 3j and 6j symbols. All such zeros are integer solutions of known polynomial expressions,^{12,13} that is, of Diophantine equations. The approach to the "zeros problem" via the study of Diophantine equations,^{3,5,7-9,11,12} using number theoretic methods, is to be distinguished from the mathematical structure that Racah¹⁴ used to "explain" the first known such zero, which was a Lie algebra–subalgebra relation. Racah's method was extended by Judd,¹⁵ and more recently has been studied systematically (though not exhaustively) by Vanden Berghe *et al.*¹⁶ Other important approaches have been either purely computational,^{2,17,18} or through the use of physical models,^{13,15,19} or simply suggestive of an underlying structure, not fully implemented.^{1,4,6}

The Diophantine approach shows that the number of zeros of the angular momentum coefficients is (denumerably) infinite, but does not suggest physical applications. Although several consequences of the existence of these zeros are known, ^{13,15,19,20} there is at present no systematic study of the subject. The mathematical relationship—if one exists—between solutions of Diophantine equations and Lie algebra-subalgebra structures has not been established.

The zeros of the 3j and 6j coefficients may all be classified by their weight, which is defined to be the smallest integer in the associated Regge²¹ and Bargmann²² arrays, respectively. The weight may assume values 1,2,... and is related to the total degree of the polynomial (in several variables) whose zeros are sought. The simplest case is weight 1. All weight-1 zeros of both the 3j and 6j coefficients have been given explicitly.^{2,5,7-10} Certain general classes of zeros of weight-2 6j coefficients have also been found, ¹² but this problem is not completely solved. The success in finding these special zeros may be attributed to the fact that they can be related to the solutions of classic (well-known) Diophantine equations. It appears likely that other Diophantine equations will need to be studied in classifying zeros of higher weight.¹¹

In this paper, we classify all zeros of weight-2 3j coefficients in terms of the classic Diophantine equation known as the (generalized) Pell equation. We use a technique developed earlier¹² for weight-2 zeros of 6j coefficients, as described in Secs. II-V. [It is also of interest to study the polynomial in question from a more general viewpoint. This is

done in Sec. VI, where it is shown that either the Lorentz group SO(1,1) or the rotation group SO(2) is an underlying symmetry of this polynomial.]

Before giving a more detailed description of the problem, we define some notations used throughout. The real line is denoted by \mathbb{R} , the plane by \mathbb{R}^2 , etc. The set of all integers is denoted by Z, the subset of natural numbers (including zero) by N, and the subset of positive and negative integers by \mathbb{Z}_+ and \mathbb{Z}_- , respectively. The obvious extensions \mathbb{Z}^2 , \mathbb{N}^2 ,... are also used. For typographical convenience, we employ We one nonstandard notation. denote the $(n+1) \times (n+1)$ matrix transformation A of a column vector $\mathbf{x} = \operatorname{col}(x_1, x_2, \dots, x_n, 1)$, to a column vector $x' = col(x'_1, x'_2, ..., x'_n, 1)$, where A has the (n + 1)st row (0,0,...,0,1) by

$$(x'_1,...,x'_n) = A \cdot (x_1,...,x_n).$$
(1.1)

The general polynomial, the set of whose zeros contains all nontrivial zeros of the 3j coefficients, has been given earlier.¹² It was also pointed out that the weight-2 zeros originate from a family of Pell equations. Since the background of this method has been presented in detail in Ref. 12, we proceed immediately to the description of the polynomial whose zeros are the weight-2 zeros of the 3j coefficients. It is convenient to use the variables (u_1, u_2, x_1, x_2) given by the following Regge array:

$$\begin{bmatrix} 2 & x_1 & x_2 + u_1 - 2 \\ u_1 + u_2 - 2 & x_2 & x_1 - u_2 + 2 \\ x_1 + x_2 - u_2 & u_1 & u_2 \end{bmatrix}.$$
 (1.2)

The domain \mathbb{D}^4 of the variables (u_1, u_2, x_1, x_2) is, by definition, the set of all points $(u_1, u_2, x_1, x_2) \in \mathbb{N}^4$ such that the remaining four entries in the Regge array also belong to N. The polynomial of interest associated with the Regge array (1.2) is given by

$$Q_{u_1,u_2}(x_1,x_2) = 2u_1(u_1-1)(x_1-u_2+1)(x_1-u_2+2) -4u_1u_2(x_1-u_2+2)x_2 +2u_2(u_2-1)x_2(x_2-1).$$
(1.3)

The zeros of the Diophantine equation

$$Q_{u_1,u_2}(x_1,x_2) = 0$$

with

$$(u_1, u_2, x_1, x_2) \in \mathbb{D}^4,$$
 (1.4b)

(1.4a)

give all nontrivial zeros for which 2 occurs in the Regge array—a zero of a weight-2 3j coefficient, or, briefly, a weight-2 zero. If any of the entries in the Regge array (1.2) are equal to 1, the polynomial (1.3) reduces to that for special weight-1 zeros. Since *all* weight-1 zeros have already been determined, we will generally restrict the determination of the zeros of $Q_{u_1,u_2}(x_1,x_2)$ to those that satisfy not only

 $(u_1, u_2, x_1, x_2) \in \mathbb{N}^4$,

but also

$$u_1 \ge 2, \quad u_2 \ge 2, \quad x_1 \ge u_2, \quad x_2 \ge 2.$$
 (1.5b)

For completeness, we write down the complete expression for the 3j coefficient corresponding to the Regge array (1.2) in the form containing the polynomial (1.3) as a factor:

$$\begin{bmatrix} 2 & x_1 & x_2 + u_1 - 2 \\ u_1 + u_2 - 2 & x_2 & x_1 - u_2 + 2 \\ x_1 + x_2 - u_2 & u_1 & u_2 \end{bmatrix} = \begin{pmatrix} (u_1 + u_2)/2 & (x_1 + x_2)/2 & (x_1 + x_2 + u_1 - u_2) \\ (u_1 + u_2 - 4)/2 & (-x_1 + x_2)/2 & (x_1 - x_2 - u_1 - u_2 + 4)/2 \end{pmatrix}$$
$$= (-1)^{x_1 + u_1 + u_2} \frac{1}{4} \left[\frac{2(u_1 + u_2 - 2)!}{u_1! u_2!} \right]^{1/2}$$
$$\times \left[\frac{(x_1 - u_2 + 3)_{u_2 - 2}(x_2 + 1)_{u_1 + u_2}}{(x_1 + x_2 - u_2 + 1)_{u_1 + u_2 + 1}} \right]^{1/2} Q_{u_1, u_2}(x_1, x_2), \quad (1.6)$$

(1.5a)

where $u_1 \ge 2$ and $u_2 \ge 2$ and $(x)_a = x(x+1)\cdots(x+a-1)$ for $a \in \mathbb{N}$ with $(x)_0 = 1$. The 3*j* symbol in this result is related to the standard angular momentum notation by

 $\begin{pmatrix} j_1 & j_2 & j_3 \\ m_1 & m_2 & m_3 \end{pmatrix}.$

We regard u_1 and u_2 as parameters, and (x_1,x_2) as variables in the expression (1.3); that is, we select values u_1 and u_2 in the domain (1.5) and determine the values of (x_1,x_2) in this domain that are zeros of $Q_{u_1,u_2}(x_1,x_2)$. Observe that each Q_{u_1,u_2} is an inhomogeneous quadratic polynomial in (x_1,x_2) . This viewpoint corresponds to interpreting a 3j coefficient in terms of a *Wigner operator* specified by the parameters (u_1,u_2) acting in a separable Hilbert space with a standard orthonormal basis given by

$$\{|\frac{1}{2}(x_1+x_2), \frac{1}{2}(-x_1+x_2)\rangle | x_1, x_2 \in \mathbb{N}\}.$$
 (1.7)

This operator structure of the 3*j* coefficients has been developed in detail in Ref. 13. We will not pursue it here except to note that the zeros (x_1,x_2) of $Q_{u_1,u_2}(x_1,x_2)$ correspond to vectors in the null space of the Wigner operator specified by (u_1,u_2) . These null space vectors are distinct from those belonging to the *characteristic* null space (see Ref. 13). As will be shown in the sequel, for each pair (u_1,u_2) (i.e., a specified Wigner operator), there is a denumerably infinite set of zeros $(x_1,x_2) \in \mathbb{N}^2$ of the quadratic polynomial Q_{u_1,u_2} (hence an infinite set of vectors in the null space of the corresponding Wigner operator). This corresponds to the *hyperbolic structure* of Q_{u_1,u_2} for each pair of parameters (u_1,u_2) , as shown in Sec. II.

The "hyperbolic" classification (to be demonstrated) of 3j coefficients using the solutions of the Diophantine equation (1.4a) for each pair $(u_1, u_2) \in \mathbb{N}^2$ is not the only one. It must also be possible to classify these zeros by dividing them into finite sets according to the value of the "magic square" parameter of a Regge array. This parameter is the common sum of the entries in the separate rows and columns. It is given by $x_1 + x_2 + u_1$ for the array (1.2). This suggests a change of coordinates.

Consider the transformation of coordinates

$$(u_1, u_2, x_1, x_2) \mapsto (v_1, v_2, y_1, y_2),$$
 (1.8)

$$y_1 = u_1, \quad y_2 = x_1 - u_2 + 2,$$
 (1.9a)

$$v_1 = u_1 + u_2 - 2, \quad v_2 = x_1 + x_2 - u_2$$

with the inverse

$$x_1 = -y_1 + y_2 + v_1, \quad x_2 = -y_2 + v_2 + 2,$$

 $u_1 = y_1, \quad u_2 = v_1 - y_1 + 2.$ (1.9b)

In the new coordinates, the Regge array (1.2) becomes

$$\begin{bmatrix} 2 & y_2 - y_1 + v_1 & y_1 - y_2 + v_2 \\ v_1 & v_2 - y_2 + 2 & y_2 \\ v_2 & y_1 & v_1 - y_1 + 2 \end{bmatrix}$$
(1.10a)

which has magic square parameter

$$v_1 + v_2 + 2.$$
 (1.10b)

The domain $\overline{\mathbb{D}}^4$ of the variables (v_1, v_2, y_1, y_2) is defined with the same phrases (*mutatis mutandis*) used in defining the domains \mathbb{D}^4 of (u_1, u_2, x_1, x_2) , namely, it is the set of all $(v_1, v_2, y_1, y_2) \in \mathbb{N}^4$ with the additional property that the remaining four entries in the Regge array (1.10a) also belong to N. The polynomial (1.3) is expressed in terms of the new variables by

$$Q_{u_1,u_2}(x_1,x_2) = P_{v_1,v_2}(y_1,y_2)$$

= $2y_1(y_1-1)y_2(y_2-1)$
 $- 4y_1(v_1-y_1+2)y_2(v_2-y_2+2)$
 $+ 2(v_1-y_1+1)(v_1-y_1+2)$
 $\times (v_2-y_2+1)(v_2-y_2+2).$ (1.11)

The coefficients of $y_1^2 y_2^2, y_1^2 y_2$, and $y_1 y_2^2$ in $P_{v_1,v_2}(y_1,y_2)$ vanish identically in (v_1,v_2) . Thus $P_{v_1,v_2}(y_1,y_2)$ is an inhomogeneous quadratic polynomial in (y_1,y_2) . This polynomial can have only the finite number of zeros that correspond to zeros of the 3*j* coefficient

$$\begin{pmatrix} (v_1+2)/2 & (-y_1+v_1+v_2+2)/2 & (y_1+y_2)/2 \\ (v_1-2)/2 & (y_1-2y_2-v_1+v_2+2)/2 & (-y_1+2y_2-v_2)/2 \end{pmatrix}$$

The reason is that for $(v_1, v_2) \in \mathbb{N}^2$, we have $(v_1, v_2, y_1, y_2) \in \overline{\mathbb{D}}^4$ if an only if $(y_1, y_2) \in \mathbb{N}^2$ and

 $0 \leq y_1 \leq v_1 + 2, \quad 0 \leq y_2 \leq v_2 + 2.$ (1.13)

Indeed, as we show in Sec. II, the number of zeros $(y_1, y_2) \in \mathbb{N}^2$ of the Diophantine equation

$$P_{v_1,v_2}(y_1,y_2) = 0 \tag{1.14}$$

for given $(v_1, v_2) \in \mathbb{N}^2$ is finite in consequence of its *elliptical* structure.

The set of all 3*j* coefficients is mapped into itself, up to phase, under the group of transformations of the angular momentum quantum numbers $(j_1, m_1, j_2, m_2, j_3, m_3)$ corresponding to permutations of rows, permutations of columns, and transposition of the Regge array. This is a 72-element group which is isomorphic to the direct product group $S_3 \times S_3 \times S_2$, where S_n denotes the symmetric group on n symbols. Thus each zero of the Diophantine equation (1.4) gives rise to 72 zeros (or less, if the Regge array is symmetric or has some equal columns or rows) corresponding to the permutations and the transposition mentioned above.

We have already used the Regge symmetry to write the entry 2 in the upper left-hand corner of the arrays (1.2) and (1.10a). The 2 remains in this position under an eight element group of transformations K of the variables (u_1, u_2, x_1, x_2) generated by the interchange of columns 2 and 3, of rows 2 and 3, and by transposition (isomorphic to $S_2 \times S_2 \times S_2$). The transformation group is given explicitly by

$$K = \{I, r, t, rt, tr, trt, rtr, (rt)^2 = (tr)^2\}.$$
 (1.15a)

0 \

The matrices r and t, which correspond to column 2-column 3 interchange and transposition, are defined by the matrices 0 0

$$r = \begin{pmatrix} 0 & 1 & 0 & 0 & 0 \\ 1 & 0 & 0 & 0 & 0 \\ 1 & 0 & 0 & 1 & -2 \\ 0 & -1 & 1 & 0 & 2 \\ 0 & 0 & 0 & 0 & 1 \end{pmatrix},$$

$$t = \begin{pmatrix} 0 & -1 & 1 & 0 & 2 \\ 0 & 1 & 0 & 0 & 0 \\ 1 & 1 & 0 & 0 & -2 \\ 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 \end{pmatrix}.$$
(1.15b)

The transformations of (u_1, u_2, x_1, x_2) are then

 $(u_1, u_2, x_1, x_2) \mapsto (u_1', u_2', x_1', x_2')$

/∩

1

$$= k \cdot (u_1, u_2, x_1, x_2), \quad k \in K.$$
 (1.15c)

The Regge symmetries of the 3*j* coefficients imply that the polynomial $Q_{u,u}$, (x_1, x_2) is invariant under the group K; that is,

$$Q_{u_1',u_2'}(x_1',x_2') = Q_{u_1,u_2}(x_1,x_2), \qquad (1.16)$$

for each transformation (1.15c). We do not take the symmetry (1.16) into account in the classification of zeros of

 $Q_{u_1,u_2}(x_1,x_2)$ given in Sec. V; that is, we have not attempted to partition these zeros into the equivalence classes resulting from the definition $(u'_1, u'_2, x'_1, x'_2) \sim (u_1, u_2, x_1, x_2)$, if Eq. (1.16) is satisfied. Thus, with each zero of Eqs. (1.4), we associate a set of 72 zeros, obtained from the given one by the Regge group. Some of these sets will be identical because of (1.16); some elements in a given set may also be equal if the Regge array is symmetric or has equal columns or rows.

Results similar to the above also apply to the polynomial $P_{v_1,v_2}(y_1,y_2)$. [We will not deal directly with this expression, but let us note that its zeros can be obtained from those of $Q_{u_1,u_2}(x_1,x_2)$ by use of the transformation (1.9).]

II. TRANSFORMATION TO DIAGONAL FORM

In studying the properties of the polynomials $Q_{u_1,u_2}(x_1,x_2)$ and $P_{v_1,v_2}(y_1,y_2)$ defined by Eqs. (1.3) and (1.11), respectively, we can regard (u_1, u_2, x_1, x_2) and (v_1, v_2, y_1, y_2) as arbitrary points in \mathbb{R}^4 . We do this in much of this section, except when we want to relate the results to zeros of 3*j* coefficients.

The inhomogeneous polynomial (1.3) is written in standard matrix form as

$$Q_{u_1,u_2}(x_1,x_2) = (x_1 \ x_2 \ 1)A\begin{pmatrix}x_1\\x_2\\1\end{pmatrix}.$$
 (2.1)

where A is the real 3×3 symmetric matrix

$$A = (a_{ij}), \tag{2.2a}$$
 with

$$a_{11} = 2u_1(u_1 - 1), \quad a_{12} = -2u_1u_2,$$

$$a_{22} = 2u_2(u_2 - 1), \quad a_{13} = -u_1(u_1 - 1)(2u_2 - 3),$$

$$a_{23} = 2u_1u_2(u_2 - 2) - u_2(u_2 - 1),$$

$$a_{33} = 2u_1(u_1 - 1)(u_2 - 1)(u_2 - 2).$$

(2.2b)

The polynomial (2.1), that is, the matrix A, is brought to diagonal form using the upper triangular transformation (see Ref. 12 for the details):

$$\Delta_{\mathcal{A}} = \begin{pmatrix} a_{11} & a_{12} & a_{33} \\ 0 & \alpha_{22} & \alpha_{23} \\ 0 & 0 & 1 \end{pmatrix},$$
(2.3a)

where

$$\alpha_{22} = \det \begin{bmatrix} a_{11} & a_{12} \\ a_{12} & a_{22} \end{bmatrix}, \quad \alpha_{23} = \det \begin{bmatrix} a_{11} & a_{13} \\ a_{12} & a_{23} \end{bmatrix}.$$
 (2.3b)

Defining new variables (ξ_1, ξ_2) by

$$\begin{pmatrix} \xi_1 \\ \xi_2 \\ 1 \end{pmatrix} = \Delta_A \begin{pmatrix} x_1 \\ x_2 \\ 1 \end{pmatrix}, \tag{2.4a}$$

we obtain

 $(\det \Delta_A)Q_{\mu_1,\mu_2}(x_1,x_2) = p_2\xi_1^2 + \xi_2^2 + p_1p_3,$ (2.4b)

where

$$p_1 = a_{11}, \quad p_2 = \alpha_{22}, \quad p_3 = \det A, \quad \det \Delta_A = p_1 p_2.$$
(2.4c)

[Note that (2.4b) is a true "quadratic form" with the variable $\xi_3 = 1$.] Explicit evaluation of the p_i and α_{23} , using the coefficients (2.2), gives

$$p_1 = 2u_1(u_1 - 1), \quad p_2 = -4u_1u_2(u_1 + u_2 - 1),$$

$$p_3 = \frac{1}{2}(u_1 - 1)(u_2 - 1)(u_1 + u_2)p_2, \quad (2.5)$$

$$\alpha_{23} = \frac{1}{2}(u_1 - 1)p_2.$$

Let us digress to consider the application of this diagonalization procedure to the problem of 3j coefficients; in this case all the entries in the matrix Δ_A are integral. Accordingly, the map $(x_1,x_2)\mapsto(\xi_1,\xi_2)$ given by Eq. (2.4a) is, for $(x_1,x_2)\in\mathbb{N}^2$, from \mathbb{N}^2 to \mathbb{Z}^2 . The converse of this result is, however, not true in general, since the inverse transformation is

$$x_2 = (\xi_2 - \alpha_{23})/p_2, \tag{2.6a}$$

$$x_1 = (\xi_1 - a_{12}x_2 - a_{13})/p_1.$$
 (2.6b)

Nonetheless, if solutions of

$$Q_{u_1,u_2}(x_1,x_2) = 0 \tag{2.7a}$$

exist in the domain (1.5), they must correspond to solutions $(\xi_1,\xi_2)\in \mathbb{Z}^2$ of the Diophantine equation

$$p_2\xi_1^2 + \xi_2^2 + p_1p_3 = 0. (2.7b)$$

This equation can be simplified further because the coefficient α_{23} contains p_2 as a factor. Hence we can write

$$\xi_2/p_2 = x_2 + \frac{1}{2}(u_1 - 1),$$
 (2.8a)

so that ξ_2/p_2 is always half-integral for $x_2 \in \mathbb{N}$ and $u_1 \in \mathbb{N}$. Consequently, it is necessary that ξ_2 have the form

$$\xi_2 = -2u_1u_2(u_1 + u_2 - 1)y, \quad y \in \mathbb{Z}, \tag{2.8b}$$

in order that $x_2 \in \mathbb{N}$.

Let us return to the general diagonalization (2.4b) for which the only restriction is det $\Delta_A \neq 0$, that is, $p_1 \neq 0$, $p_2 \neq 0$. Under the further change of variable

$$\xi_1 = x, \quad x \in \mathbb{R}, \tag{2.9a}$$

$$\xi_2 = -2u_1u_2(u_1 + u_2 - 1)y, \quad y \in \mathbb{R}, \tag{2.9b}$$

(2.4b) reduces to

$$2u_{1}(u_{1}-1)Q_{u_{1},u_{2}}(x_{1},x_{2})$$

$$=x^{2}-u_{1}u_{2}(u_{1}+u_{2}-1)y^{2}$$

$$+u_{1}(u_{1}-1)^{2}(u_{2}-1)(u_{1}+u_{2}).$$
(2.10)

The relation between the variables (x_1, x_2) and (x, y) is

$$x_1 = \frac{x + u_1 u_2 y}{2u_1(u_1 - 1)} + \frac{1}{2} (u_2 - 3), \qquad (2.11a)$$

$$x_2 = \frac{1}{2}(y - u_1 + 1).$$
 (2.11b)

Relation (2.10) is valid for all points $(u_1, u_2, x_1, x_2) \in \mathbb{R}^4$, excepting the points that have $u_1 = 0$ or $u_1 = 1$, with the transformation between coordinates given by Eqs. (2.11).

The problem of finding all nontrivial weight-2 zeros of 3*j* coefficients has now been reduced to the following: For each pair $(u_1, u_2) \in \mathbb{N}^2$ with $u_1 > 2$ and $u_2 > 2$, find all solutions $(x, y) \in \mathbb{Z}^2$ of the Pell equation

$$x^{2} - u_{1}u_{2}(u_{1} + u_{2} - 1)y^{2}$$

= $-u_{1}(u_{1} - 1)^{2}(u_{2} - 1)(u_{1} + u_{2})$ (2.12)

such that the coordinates (x_1, x_2) given by Eqs. (2.11) satisfy $(x_1, x_2) \in \mathbb{N}^2$ and $x_1 \ge u_2, x_2 \ge 2$. The corresponding weight-2 3*j* coefficients that are zero are then given by Eq. (1.6).

That such zeros exist is shown by the example $u_1 = u_2 = 2$, x = 38, y = 11, $x_1 = 20$, $x_2 = 5$, which corresponds to

$$\begin{pmatrix} 2 & \frac{25}{2} & \frac{25}{2} \\ 0 & -\frac{15}{2} & \frac{15}{2} \end{pmatrix} = 0.$$
 (2.13)

As will be shown in the next section, this zero is one of a denumerably infinite family corresponding to certain points $(x,y) \in \mathbb{N}^2$ on the hyperbola

$$x^2 - 12y^2 = -8. (2.14)$$

We refer to the classification of the weight-2 zeros of 3*j* coefficients using the Pell equation (2.12) as the *hyperbolic classification*. This method is developed in some detail in Sec. III.

The diagonalization procedure described by Eqs. (2.3)–(2.5) also applies to the polynomial

$$P_{v_1,v_2}(y_1,y_2) = (y_1 \ y_2 \ 1) B \begin{pmatrix} y_1 \\ y_2 \\ 1 \end{pmatrix}, \qquad (2.15)$$

where B is the real 3×3 symmetric matrix

$$\boldsymbol{B} = (\boldsymbol{b}_{ij}), \tag{2.16a}$$

with

$$b_{11} = 2(v_2 + 1)(v_2 + 2), \quad b_{12} = 2(v_1 + 1)(v_2 + 1),$$

$$b_{22} = 2(v_1 + 1)(v_1 + 2),$$

$$b_{13} = -(v_2 + 1)(v_2 + 2)(2v_1 + 3), \quad (2.16b)$$

$$b_{23} = -(v_1 + 1)(v_1 + 2)(2v_2 + 3),$$

$$b_{33} = 2(v_1 + 1)(v_1 + 2)(v_2 + 1)(v_2 + 2).$$

The final result, which is analogous to Eqs. (2.10) and (2.11), is

$$2(v_{2} + 1)(v_{2} + 2)P_{v_{1},v_{2}}(y_{1},y_{2})$$

$$= X^{2} + (v_{1} + 1)(v_{2} + 1)(v_{1} + v_{2} + 3)Y^{2}$$

$$- (v_{1} + 2)(v_{2} + 1)(v_{2} + 2)^{2}(v_{1} + v_{2} + 2),$$
(2.17)

where the transformation between the coordinates (y_1,y_2) and (X,Y) is

$$y_1 = \frac{X - (v_1 + 1)(v_2 + 1)Y}{2(v_2 + 1)(v_2 + 2)} + \frac{1}{2}(v_1 + 2), \quad (2.18a)$$

$$y_2 = \frac{1}{2}(Y + v_2 + 2).$$
 (2.18b)

Relation (2.17) is valid for all points $(v_1, v_2, y_1, y_2) \in \mathbb{R}^4$, excepting the points having $v_2 = -2$ or $v_2 = -1$, with the transformation between coordinates given by Eqs. (2.18).

The problem of finding all nontrivial weight-2 zeros of 3*j* coefficients has now been reduced to the following: For each pair $(v_1, v_2) \in \mathbb{N}^2$, with $v_1 > 2$ and $v_2 > 2$, find all solutions $(X, Y) \in \mathbb{Z}^2$ that belong to the ellipse

$$X^{2} + (v_{1} + 1)(v_{2} + 1)(v_{1} + v_{2} + 3)Y^{2}$$

 $= (v_1 + 2)(v_2 + 1)(v_2 + 2)^2(v_1 + v_2 + 2)$ (2.19) such that the coordinates (y_1, y_2) given by Eqs. (2.18) satisfy $(y_1, y_2) \in \mathbb{N}^2$ and $2 < y_1 < v_1$, $2 < y_2 < v_2$. The corresponding weight-2 3j coefficients that are zero are then given by Eq. (1.12).

That such zeros exist is shown by the example $v_1 = 2$, $v_2 = 23$, X = 1080, Y = 15, $y_1 = 2$, $y_2 = 20$, which corresponds to the zero given by Eq. (2.13) [see Eqs. (1.9a) and (1.12)]. This zero comes from the ellipse

$$X^2 + 2016Y^2 = 1620\ 000. \tag{2.20}$$

We refer to the classification of the weight-2 zeros of 3j coefficients using integral points on the ellipse (2.19) as the *elliptical classification*. All weight-2 zeros of 3j coefficients can be classified by either the hyperbolic or the elliptic scheme, *or both*.

III. GENERAL PROCEDURE FOR SOLVING PELL'S EQUATION

Our procedure for solving Pell's equation,

$$x^2 - Dy^2 = N, \tag{3.1}$$

where D is a positive integer and N integral but not necessarily positive, has been described in detail in Ref. 12. All proofs of the results stated below are given there. For the reader's convenience, we summarize briefly this *method of orbits*.

We order all points $(x,y) \in \mathbb{R}^2$ by the rule (x,y) < (x',y')if x < x' and (x,y) < (x,y') if y < y'. The point (x,y) is called positive if x > 0 and $y \ge 0$. We also use the following notations.

(i)
$$(u_0,v_0)$$
 denotes the least positive solution of
 $u^2 - Dv^2 = 1$ (3.2)

(such a solution always exists; see, for example, $LeVeque^{23}$).

(ii) (x_0, y_0) denotes the least positive solution of the Pell equation (3.1) for given D and N, whenever solutions of this equation exist.

With the solution (u_0, v_0) of Eq. (3.2) we associate the matrix g_0 of unit determinant given by

$$g_0 = \begin{pmatrix} u_0 & v_0 D \\ v_0 & u_0 \end{pmatrix}. \tag{3.3}$$

The basic result for solutions of the general Pell equation (3.1) is the following: For each $n \in \mathbb{Z}$ and each solution (x,y) of (3.1), the point $(x^{(n)}, y^{(n)})$ defined by

$$\binom{x^{(n)}}{y^{(n)}} = g_0^n \binom{x}{y}$$
(3.4)

is a solution of (3.1).

Relation (3.4) is the principal result needed for classifying all solutions of Pell's equation (3.1) by means of the orbits of the group G_0 defined by

$$G_0 = \{ g_0^n | n \in \mathbb{Z} \}.$$
(3.5)

Each $g \in G_0$ maps a point $(x,y) \in \mathbb{R}^2$ of the hyperbola $x^2 - Dy^2 = N$ to a new point

$$(x,y) \mapsto (x',y') = g(x,y)$$
(3.6a)

of the hyperbola. The action of g on the point (x,y) is given by matrix multiplication

$$\begin{pmatrix} x'\\ y' \end{pmatrix} = g \begin{pmatrix} x\\ y \end{pmatrix}.$$
(3.6b)

In particular, $(x,y) \in \mathbb{Z}^2$ implies $(x',y') \in \mathbb{Z}^2$, and conversely. The orbit of the point $(x,y) \in \mathbb{Z}^2$ is the set of points $G_0(x,y)$ defined by

$$G_0(x,y) = \{ g(x,y) | g \in G_0 \}.$$
(3.7)

A set of fundamental solutions with respect to G_0 of Pell's equation (3.1) is given by

$$\mathbb{D} = \{ (x,y) | x^2 - Dy^2 = N; \ (x,y) \in \mathbb{N}^2; \\ 1 \leq y \leq u_0 x_0 + v_0 y_0 - 1 \}.$$
(3.8a)

Here (u_0,v_0) and (x_0,y_0) denote the least positive solutions of Eqs. (3.2) and (3.1), respectively. The set D depends, of course, on D and N in Eq. (3.1), that is, we really have

$$\mathbf{D} = \mathbf{D}(D, N). \tag{3.8b}$$

The set of all integer solutions of the Pell equation (3.1)belonging to the *positive branch* of the hyperbola (x > 0, if N > 0; y > 0, if N < 0) is given by the sets

$$g_0^n \mathbb{D}, \quad n = 0, \pm 1, \pm 2, \dots$$
 (3.9)

Here multiplication of \mathbb{D} from the left by g_0^n means that each element of \mathbb{D} is multiplied by g_0^n . Alternatively, the set of all integer solutions (3.9) of Pell's equation belonging to the positive branch may be classified by orbits under the action of the group G_0 ,

$$G_0(x,y) = \{g_0^n(x,y) | n = 0, \pm 1, \pm 2, ...\},$$
 (3.10a)
where

(*x*

$$x,y)\in\mathbb{D}.$$
 (3.10b)

One obtains, of course, *all* integer solutions of Eq. (3.1) from those given by the sets (3.9) [or (3.10)] by adjoining those with reversed sign of the x coordinate if N > 0 and of the y coordinate if N < 0.

The number of orbits, that is, the number $|\mathbb{D}|$ of elements in the set D, is not known in general. Bounds on this number have been given by Stolt.²⁴ For the special Pell equation of interest here, we determine the number of orbits numerically, as described in the next section.

IV. PROPERTIES OF THE PELL EQUATION FOR WEIGHT-2 ZEROS OF 3/ COEFFICIENTS

The Pell equation of interest here for determining all weight-2 zeros of 3j coefficients is given by Eq. (2.12). Because both D and N are polynomials in the variables $(u_1, u_2) \in \mathbb{N}^2$, this Pell equation has many nice properties that allow us to give a reasonably comprehensive treatment. In terms of the notation (3.1) for the general Pell equation, Eq. (2.12) is

$$x^2 - Dy^2 = N, \tag{4.1a}$$

where the integers D and N are given by

$$D = u_1 u_2 (u_1 + u_2 - 1), \tag{4.1b}$$

$$\mathbf{V} = -u_1(u_1 - 1)^2(u_2 - 1)(u_1 + u_2), \qquad (4.1c)$$

where

$$(u_1, u_2) \in \mathbb{N}^2, \quad u_1 \ge 2, \quad u_2 \ge 2.$$
 (4.1d)

Throughout this section, D and N denote the polynomials (4.1b) and (4.1c), respectively, subject to conditions (4.1d) unless otherwise specified.

One of the special properties of Eq. (4.1a) is that it possesses two *parametric solutions*. Namely, for all

$$(u_1, u_2, x, y) \in \mathbb{R}^4,$$
 (4.2)

the following values of x and y solve Eq. (4.1a):

$$x = u_1(u_1 - 1), \quad y = u_1 - 1;$$
 (4.3a)

$$x = u_1(u_1 + 2u_2 - 1), \quad y = u_1 + 1.$$
 (4.3b)

For all $(u_1, u_2) \in \mathbb{N}^2$ with $u_1 \ge 2$ and $u_2 \ge 2$, one can prove by elementary methods that these two solutions are the only parametric solutions of Eq. (4.1a). In fact, each parametric solution must have $y = u_1 + a$, $a \in \mathbb{Z}$, since this is the only way that the term $-u_1^4 u_2$ on the right-hand side can be canceled identically by a term from the left-hand side. The values $a = \pm 1$ are then determined by substituting $y = u_1 + a$ in Eq. (4.1a).

Numerical calculations show that the parametric solution (4.3a) is sometimes, but not always, the smallest positive solution (x_0,y_0) of the equation. More generally, it can be shown that

$$(u_1 - 1)\sqrt{(u_2 - 1)/u_2} < y_0 \le u_1 - 1,$$
 (4.4)

but we cannot give the exact result for (x_0, y_0) .

Let us summarize, for the special Pell equation (4.1), the method of solution given in Sec. III [see Eqs. (3.8)]. The fundamental solutions are

$$\mathbb{D}(u_1, u_2) = \mathbb{D}(D, N), \tag{4.5}$$

where D and N are defined by Eqs. (4.1b) and (4.1c). As noted earlier, (u_0,v_0) and (x_0,y_0) are the least positive solutions, respectively, of $u^2 - Dv^2 = 1$ and $x^2 - Dy^2 = N$. The set $\mathbb{D}(u_1,u_2)$ contains at least two points, namely, the two parametric solutions (4.3). For a specific (u_1,u_2) , the set of all integral solutions of Eq. (4.1a) belonging to the positive branch of the hyperbola is given by

$$g_0^n \mathbb{D}(u_1, u_2), \quad n = 0, \pm 1, \pm 2, \dots$$
 (4.6)

The orbit $G_0(x,y)$ of the fundamental solution $(x,y) \in \mathbb{D}(u_1,u_2)$ is

$$G_0(x,y) = \{ g_0^n(x,y) | n \in \mathbb{Z} \}.$$
(4.7)

An integral solution of the Pell equation does not, in general, correspond to a zero of the 3j coefficient. This is because the map $(x,y)\mapsto(x_1,x_2)$ given by Eqs. (2.11) is not from integers into integers. For that property to hold, it is necessary (but not sufficient) that $u_1(u_1 - 1)$ divide $x + u_1u_2 y$; that is,

$$k = (x + u_1 u_2 y) / u_1 (u_1 - 1) \in \mathbb{Z}.$$
(4.8)

When this condition is fulfilled, the map (2.11) $(x,y)\mapsto(x_1,x_2)$ is

$$x_1 = (k + u_2 - 3)/2,$$
 (4.9a)

$$x_2 = (y - u_1 + 1)/2.$$
 (4.9b)

From the above results we conclude the following: For each $(u_1, u_2) \in \mathbb{N}$ with $u_1 \ge 2$, $u_2 \ge 2$, the necessary and sufficient conditions that a solution (x, y) of the Pell equation correspond to a weight-2 zero of a 3j coefficient are

(i)
$$x + u_1 u_2 y = 0 [mod u_1(u_1 - 1)];$$

(ii)
$$(u_1, u_2)$$
 and (k, y) satisfy $k \ge u_2 + 3$ with k and u_2 of

opposite parity, and $y \ge u_1 + 3$ with y and u_1 of opposite parity.

We formulate two theorems that are basic for obtaining the zeros of the 3j coefficients from the zeros of the Pell equation. First, we define the subset $A(u_1,u_2)$ of the set $\mathbb{D}(u_1,u_2)$ of fundamental solutions of the Pell equation by

$$\mathbb{A}(u_1, u_2) = \{(x, y) \in \mathbb{D}(u_1, u_2) | \text{condition (i) holds} \}.$$
(4.10)

Theorem 4.1: A necessary and sufficient condition that a solution (x',y') of the Pell equation be divisible by $u_1(u_1-1)$ is that $(x',y')\in G_0(x,y)$, where $(x,y)\in A(u_1,u_2)$.

Proof: Let (x,y) and (x',y') be any two integer solutions of the Pell equation that are related by $(x',y') = g_0(x,y)$, where g_0 is the generator (3.3) of G_0 . We prove the theorem by showing that, if $x + u_1u_2y$ is divisible by $u_1(u_1 - 1)$, then so is $x' + u_1u_2y'$, and conversely. Define k and k' [see (4.8)] by

$$k = (x + u_1 u_2 y)/u_1(u_1 - 1),$$
 (4.11a)

$$k' = (x' + u_1 u_2 y^1) / u_1 (u_1 - 1).$$
 (4.11b)

The transformation $(x,y)\mapsto(x',y')=g_0(x,y)$ then gives $(k,y)\mapsto(k',y')=h_0(k,y)$; that is,

$$\binom{k'}{y'} = h_0 \binom{k}{y}, \tag{4.12a}$$

where h_0 is the matrix of unit determinant defined by

$$h_0 = \begin{pmatrix} u_0 + u_1 u_2 v_0 & -u_2 (u_2 - 1) v_0 \\ u_1 (u_1 - 1) v_0 & u_0 - u_1 u_2 v_0 \end{pmatrix}.$$
 (4.12b)

Clearly, k and y integral imply k' and y' integral, and conversely, since h_0^{-1} is obtained from h_0 by replacing v_0 by $-v_0$.

The significance of Theorem 4.1 is in showing that for obtaining solutions of the Pell equation (4.1a) that map back to integer solutions (x_1,x_2) of $Q_{u_1,u_2}(x_1,x_2) = 0$ it is sufficient to consider those fundamental solutions in the set $A(u_1,u_2)$. We must still select from the orbit $G_0(x,y)$, $(x,y) \in A(u_1,u_2)$, those points having parity such that (x_1,x_2) defined by Eqs. (4.9) are integral. This selection is also reduced to the set $A(u_1,u_2)$ by the next result.

Theorem 4.2: Each point in the orbit $G_0(x,y)$, $(x,y) \in \mathbb{A}(u_1,u_2)$ has the same parity as the point (x,y).

Proof: It is sufficient to show that the maps (x,y) $\mapsto (x',y') = g_0(x,y)$ and $(x,y) \mapsto (x'',y'') = g_0^{-1}(x,y)$ conserve the parity of (x,y) for any integer solution of the Pell equation. We do this by considering all cases. To this end we write (a,b) = (e,o) if the integer pair (a,b) has a even and b odd, etc. Then for $(u_1,u_2) = (o,o)$ we have that D is odd, N is even, and (x,y) = (e,e) or (o,o), while $(u_0,v_0) = (o,e)$ or (e,o). For $(u_1,u_2) \neq (o,o)$, we have that D is even, N is even, and (x,y) = (e,e) or (e,o), while $(u_0,v_0) = (o,e)$ or (o,o). Considering all eight cases in turn, we verify parity conservation for the map $(x,y) \mapsto (x',y') = g_0(x,y)$. This result also applies to $(x,y) \mapsto (x'',y'') = g_0^{-1}(x,y)$, since g_0^{-1} is obtained from g_0 by the map $v_0 \mapsto -v_0$, which conserves parity.

It remains to explain briefly our numerical method for solving the general Pell equation (3.1), where D is a positive, nonsquare integer. We do not treat the elliptical case here, although, in fact, it is much simpler than the hyperbolic case. If D is a perfect square, there are, as in the elliptical case, only a finite number of solutions for given N, and these are easily dealt with.

Let us choose D and N randomly (in some appropriate fashion) from a very large integer lattice of (D,N) points. Then the Pell equation has, with high probability, no solutions at all. LeVeque²³ gives bounds on x and y [depending on (u_0,v_0) , D, and N] such that if the equation has solutions it must have at least one satisfying these bounds. In our case, Eq. (4.1a), there are parametric solutions, so this problem does not arise.

If we take the least positive solution (x_0,y_0) and operate on it as in Eq. (3.4) (with n = 1) we get the value $(x_0^{(1)}, y_0^{(1)})$, the second point on the first orbit. We then solve for all $(x',y') < (x_0^{(1)},y_0^{(1)})$ thus generating the set D of (3.8a), and hence all fundamental solutions of our special Pell equation. The procedure is to transpose the equation to $x^2 = Dy^2 + N$ and to let y vary from y_0 to $y_0^{(1)}$. For each y, we test the righthand side of this expression for "squareness"; whenever it is a square, we have a solution. We used various Cray computing machines and worked in double precision. Since Crays have not been designed with such number theoretic problems in mind, it was necessary to do some preprocessing in order to speed up the calculations. Briefly, this consisted in examining a fixed number of trailing bits in $Dy^2 + N$ to eliminate those values which could not possibly be squares. This is quite fast compared to determining whether the expression in question is a square.

There was one further problem of a technical nature. The process described above requires knowledge of the matrix g_0 [Eq. (3.3)], hence of (u_0, v_0) , the least positive solution of Eq. (3.2). A look at Kortum and McNiel²⁵ shows that these integers can be very large indeed; manipulating them would require multiprecision arithmetic, where "multi" means greater than "double." The very large solution appearing in the set D would probably be of no interest in the physics of the present problem; consequently, we made no effort to calculate all orbits in these cases.

Finally, the code was modified to exclude solutions not satisfying the divisibility conditions of Eq. (4.8); parity violations were dealt with by eye.

V. THE WEIGHT-2 ZEROS OF 3/ COEFFICIENTS

We have tabulated in Table I the elements (x,y) in the set $D(u_1,u_2)$ for $2 \le u_1 \le 6$, $2 \le u_2 \le 6$, together with the integer k defined by Eq. (4.8), and the solution (u_0,v_0) of $u^2 - Dv^2 = 1$. These integers are presented as (x,y)(k)when k exists, and as (x,y) otherwise. Recall that $A(u_1,u_2)$ is the subset of all fundamental solutions of the Pell equation (4.1a) such that $(x,y) \in D(u_1,u_2)$ and $x + u_1u_2 y$ is divisible by $u_1(u_1 - 1)$, this quotient defining the integer k. Thus the coordinates (k,y) and the elements of the matrix h_0 in Eqs. (4.12) are completely determined for each point $(x,y) \in A(u_1,u_2)$.

We shall now show how to obtain all weight-2 zeros of the 3j coefficients from the fundamental solutions $(x,y) \in \mathbb{A}(u_1,u_2)$, Eqs. (4.12), and Eqs. (4.9). We must take into account the transformations (4.8) and (4.9), as well as the parity requirements and conditions $k \ge u_2 + 3$ and $y \ge u_1 + 3$ given just below Eqs. (4.9).

Let L denote the linear transformation

$$\binom{k}{y} = L\binom{x}{y}, \tag{5.1a}$$

$$L = \begin{pmatrix} 1/u_1(u_1 - 1) & u_1u_2/u_1(u_1 - 1) \\ 0 & 1 \end{pmatrix}.$$
 (5.1b)

The group H_0 defined by

$$H_0 = LG_0 L^{-1} = \{h_0^n | n \in \mathbb{Z}\}$$
(5.2)

is isomorphic to G_0 . The matrix h_0 [see Eq. (4.12b)] is completely determined by (u_0, v_0) and (u_1, u_2) .

Next, we define the set $\mathbb{B}(u_1,u_2)$ to be the image of $\mathbb{A}(u_1,u_2)$ under the transformation L,

$$\mathbb{A}(u_1, u_2) \mapsto \mathbb{B}(u_1, u_2). \tag{5.3}$$

Thus the elements of $\mathbb{B}(u_1, u_2)$ are the pairs of integers $(k, y) \in \mathbb{N}^2$ obtained by applying the transformation L to each element $(x, y) \in \mathbb{A}(u_1, u_2)$.

We can now prove the following result.

Theorem 5.1: For each $h \in H_0$, the transformation

 $(k,y)\mapsto h(k,y), (k,y)\in \mathbb{B}(u_1,u_2)$

conserves parity; that is, the integer pair (k',y') = h(k,y) has the same parity as (k,y).

Proof: It is sufficient to prove that the transformation $(n_1,n_2) \mapsto h_0(n_1,n_2)$ conserves the parity of (n_1,n_2) , where (n_1,n_2) is any pair of integers. This is done by considering all cases for (u_1,u_2) , (u_0,v_0) , and (n_1,n_2) . We find that the diagonal elements of the matrix h_0 [see Eq. (4.12b)] are always odd and the off-diagonal ones even. This implies that $(n'_1,n'_2) = h_0(n_1,n_2)$ and (n_1,n_2) have the same parity.

The significance of Theorem 5.1 for weight-2 zeros of 3j coefficients is that we can restrict our considerations to those elements $(k,y) \in \mathbb{B}(u_1,u_2)$ that satisfy the required parity properties. This motivates us to define the subset $S(u_1,u_2)$ of $\mathbb{B}(u_1,u_2)$ by

$$S(u_1, u_2) = \{(k, y) \in \mathbb{B}(u_1, u_2) | u_2 + k \text{ odd}; u_1 + y \text{ odd}\}.$$
(5.4)

Not all points (k,y) in the set $S(u_1,u_2)$ are mapped by the transformation (4.9) to coordinates (x_1,x_2) that give weight-2 zeros of a 3*j* coefficient. This is because the "threshold conditions" $k \ge u_2 + 3$, $y \ge u_1 + 3$ must still be satisfied. This situation suggests defining the pair of integers $(k^*,y^*) \in \mathbb{N}^2$ to be the least pair in the sets

...,
$$h_0^{-2}S(u_1,u_2), h_0^{-1}S(u_1,u_2), S(u_1,u_2),$$

 $h_0S(u_1,u_2), h_0^2S(u_1,u_2),...,$ (5.5a)

such that

$$k^* \ge u_2 + 3, \quad y^* \ge u_1 + 3.$$
 (5.5b)

[Note that h_0 is defined in (4.12b).] We find, in fact, that

$$(k^*, y^*) \in h_0^{-1} S(u_1, u_2), S(u_1, u_2), \text{ or } h_0 S(u_1, u_2).$$
 (5.6)

This result may be proved by using the parametric solution (4.3a). This parametric point gives

TABLE I. Fundamenta	l solutions of	f Pell's equation.
---------------------	----------------	--------------------

(<i>u</i> ₁ , <i>u</i> ₂)	(u_0, v_0)	(x,y)(k) and (x,y)
(2,2)	(two orbits)	(7,2)	(2,1)(3) (10,3)(11)
(2,3)	(two orbits)	(5,1)	(2,1)(4) (14,3)(16)
(2,4)	(three orbits)	(19,3)	(2,1)(5) $(18,3)(21)$ $(82,13)(93)$
(2,5)	(four orbits)	(31,4)	(2,1)(6) $(22,3)(26)$ $(38,5)(44)$ $(178,23)(204)$
(2,6)	(six orbits)	(55,6)	(2,1)(7) (16,2)(20) (26,3)(31) (82,9)(95) (128,14)(148) (394,43)(455)
(3,2)	(two orbits)	(5,1)	(6,2)(3) (18,4)(7)
(3,3)	(six orbits)	(161,24)	(6,2)(4) (24,4)(10) (66,10)(26) (174,26)(68) (456,68)(178) (1194,178)(466)
(3,4)	(four orbits)	(17,2)	(6,2)(5) (30,4)(13) (66,8)(27) (186,22)(75)
(3,5)	(six orbits)	(41,4)	(6,2)(6) (36,4)(16) (69,7)(29) (111,11)(46) (204,20)(84) (594,58)(244)
(3,6)	(square)	(1,0)	(6,2)(7) (42,4)(19)
(4,2)	(four orbits)	(19,3)	(12,3)(3) $(28,5)$ $(68,11)(13)$ $(132,21)(25)$
(4,3)	(four orbits)	(17,2)	(12,3)(4) $(36,5)(8)$ $(108,13)(22)$ $(228,27)(46)$
(4,4)	(eight orbits)	(127,12)	(12,3)(5) (44,5) (68,7)(15) (156,15)(33) (348,33)(73) (772,73) (1132,107)(237) (2508,237)(525)
(4,5)	(ten orbits)	(721,57)	(12,3)(6) (52,5) (108,9)(24) (212,17)(46) (492,39)(106) (948,75)(204) (2188,173) (4212,333)(906) (8108,641)(1744) (18708,1479)(4024)
(4,6)	(nine orbits)	(485,33)	(12,3)(7) (60,5)(15) (156,11)(35) (276,19)(61) (660,45)(145) (1572,107)(345) (2748,187)(603) (6540,445)(1435) (15564,1059)(3415)
(5,2)	(four orbits)	(31,4)	(20,4)(3)(40,6)(5)(200,26)(23)(340,44)(39)
(5,3)	(seven orbits)	(41,4)	(20,4)(4) (50,6)(7) (85,9)(11) (160,16)(20) (295,29) (470,46)(58) (860,84)(106)
(5,4)	(eight orbits)	(721,57)	(20,4)(5) (60,6)(9) (300,24)(39) (580,46)(75) (1340,106)(173) (2580,204)(333) (11460,906)(1479) (22060,1744)(2847)
(5,5)	(square)	(1,0)	(20,4)(6) (70,6)(11) (155,11)
(5,6)	(12 orbits)	(1351,78)	(20,4)(7) (80,6)(13) (160,10)(23) (340,20)(47) (620,36)(85) (1280,74)(175) (2320,134)(317) (4780,276)(653) (8660,500)(1183) (17840,1030)(2437) (32320,1866)(4415) (66580,3844)(9095)
(6,2)	(nine orbits)	(55,6)	(12,4)(2) (30,5)(3) (54,7) (114,13)(9) (180,20)(14) (282,31) (558,61)(43) (870,95)(67) (1356,148)
(6,3)	(square)	(1,0)	(30,5)(4) (66,7) (222,19)
(6,4)	(12 orbits)	(485,33)	(30,5)(5)(78,7)(114,9)(11)(210,15)(19)(510,35)(45)(894,61) (1218,83)(107)(2130,145)(187)(5070,345)(445)(8862,603) (12066,821)(1059)(21090,1435)(1851)
(6,5)	(12 orbits)	(1351,78)	(30,5)(6) $(90,7)(10)$ $(210,13)(20)$ $(390,23)(36)$ $(810,47)(74)(1470,85)(134)$ $(3030,175)(276)$ $(5490,317)(500)(11310,653)(1030)$ $(20490,1183)(1866)$ $(42210,2437)(3844)(76470,4415)(6964)$
(6,6)	(eight orbits)	(199,10)	(30,5)(7) (102,7) (366,19)(35) (690,35)(65) (1290,65)(121) (2406,121) (7422,373)(695) (13830,695)(1295)

$$(k,y) = (u_2 + 1, u_1 - 1) \in \mathbb{S}(u_1, u_2).$$
(5.7)

showing that the set $S(u_1, u_2)$ is never empty; on the other hand, the point (5.7) does not satisfy the threshold condition (5.5b). It is, however, true that

$$h_0^{-1}(u_2+1,u_1-1) < h_0(u_2+1,u_1-1),$$
 (5.8a)

$$h_0(u_2 + 1, u_1 - 1) \ge (u_2 + 3, u_1 + 3)$$
 (5.8b)

for $u_1 \ge 2$, $u_2 \ge 2$, which proves (5.6). Relation (5.8a) is trivial to prove; (5.8b) may be proved directly from the definition (4.12b) of h_0 , using the property $v_0 \ge 1$, $u_0 > u_1 + 1$ of the least positive solution of $u^2 - Dv^2 = 1$ for $u_1 \ge 2$, $u_2 \ge 2$ and D nonsquare.

The orbit $H_0(k,y)$ of the point $(k,y) \in S(u_1,u_2)$ under the action of the group H_0 is, by definition, the set of points

$$H_0(k,y) = \{h_0^n(k,y) | n \in \mathbb{Z}\}.$$
(5.9)

The results above furnish the complete answer to the problem of obtaining all weight-2 zeros of the 3*j* coefficient

in terms of Pell's equation. We state this result as a theorem, using the notations above.

Theorem 5.2: For each pair $(u_1, u_2) \in \mathbb{N}^2$ with $u_1 \ge 2$, $u_2 \ge 2$, we have

$$\begin{bmatrix} 2 & x_1 & x_2 + u_1 - 2 \\ u_1 + u_2 - 2 & x_2 & x_1 - u_2 + 2 \\ x_1 + x_2 - u_2 & u_1 & u_2 \end{bmatrix} = 0, \quad (5.10)$$

where the integer pair $(x_1, x_2) \in \mathbb{N}^2$ with $x_1 \ge u_2, x_2 \ge 2$ is any of the pairs given by

$$x_1 = (k' + u_2 - 3)/2,$$
 (5.11a)

$$x_2 = (y' - u_1 + 1)/2,$$
 (5.11b)

for each (k', y') such that

 $(k',y') \in H_0(k,y),$ (5.12a)

$$(k',y') \ge (k^*,y^*),$$
 (5.12b)

in which the point (k,y) itself runs over all elements in the set $S(u_1,u_2)$.

Remark: the Regge symmetries and Theorem 5.2 give *all* nontrivial weight-2 zeros of the 3*j* coefficients in terms of positive solutions of the Pell equation. Because of parity, modular, and threshold restrictions, not all solutions of the Pell equation map back to weight-2 zeros. In the final result, given by Theorem 5.2, the zeros (5.10) are classified according to the orbits $H_0(k,y)$, where (k,y) is a point in $S(u_1,u_2)$, there being an orbit for each such point. By construction, each point (x',y') in the orbit $H_0(k,y)$ has parity such that the pair (x_1,x_2) is integral, but not necessarily positive nor greater than the threshold required to make all entries in the

$$\mathbb{D}(6,2) = \begin{cases} (12,4), & (30,5), & (54,7), & (114,13), \\ (282,31), & (558,61), & (870,95), & (1356,148) \end{cases}$$

The set of all solutions belonging to the positive branch y > 0 is

 $\{g_0^n \mathbb{D}(6,2) | n = 0, \pm 1, ...\},$ (5.14b)

 $\{G_0(x,y)|(x,y)\in\mathbb{D}(6,2)\}.$ (5.14c)

The subset $A(6,2) \subset D(6,2)$ of fundamental solutions whose points satisfy $x + 12y = 0 \mod 30$ is

$$= \left\{ \begin{array}{ccc} (12,4), & (30,5), & (114,13), & (180,20) \\ (558,61), & (870,95) & \\ \end{array} \right\}$$
(5.15)

with the corresponding set B(6,2) of values (k,y) given by B(6,2)

$$= \left\{ \begin{array}{ccc} (2,4), & (3,5), & (9,13), & (14,20) \\ (43,61), & (67,95) \end{array} \right\}.$$
(5.16)

Finally, the subset S(6,2) of points in B(6,2) having the proper parity is

$$S(6,2) = \{(3,5), (9,13), (43,61), (67,95)\}.$$
 (5.17)

The group H_0 is given by

$$H_0 = \{h_0^n | n = 0, \pm 1, \pm 2, ...\},$$
 (5.18a)

where

$$h_0 = \begin{pmatrix} 127 & -12\\ 180 & -17 \end{pmatrix}.$$
 (5.18b)

The orbits of the group H_0 on the set S(6,2) are

$$H_0(3,5), H_0(9,13), H_0(43,61), H_0(67,95).$$
 (5.19)

We see that there are four disjoint denumerably infinite families of weight-2 zeros of the 3j coefficients corresponding to $(u_1, u_2) = (6, 2)$. These zeros are given by the Regge array

$$\begin{bmatrix} 2 & (k'-1)/2 & (y'+3)/2 \\ 6 & (y'-5)/2 & (k'-1)/2 \\ (k'+y'-10)/2 & 6 & 2 \end{bmatrix} = 0,$$
(5.20)

where (k',y') is a positive point in any of the four orbits

Regge array greater than 2. The condition $(k',y') \ge (k^*,y^*)$ assures this property for the corresponding points (x_1,x_2) .

It is useful to illustrate the results given here and in Sec. IV by an example from Table I. Consider $(u_1, u_2) = (6,2)$, which has D = 84 and N = -1200. The least positive solution (u_0, v_0) of $u^2 - 84v^2 = 1$ is $(u_0, v_0) = (55,6)$, which gives

$$g_0 = \begin{pmatrix} 55 & 504 \\ 6 & 55 \end{pmatrix}. \tag{5.13}$$

The set D(6,2) of fundamental solutions of $x^2 - 84y^2$ = -1200 is

$$\left. \begin{array}{c} (180,20) \\ (5.14a) \end{array} \right\}.$$

(5.19) satisfying $(k',y') \ge (5,9) = (k^*,y^*)$. This includes, in fact, all points in the orbits (5.19) with the exception of the four points:

$$(1,5) = h_0^{-1}(67,95) \in H_0(67,95),$$

$$(1,7) = h_0^{-1}(43,61) \in H_0(43,61),$$

$$(3,31) = h_0^{-1}(9,13) \in H_0(9,13),$$

$$(3,5) = h_0^0(3,5) \in H_0(3,5).$$

The result for weight-2 zeros of the 3j coefficients as given by Theorem 5.2 is, we believe, a comprehensive and elegant statement of the origin of these zeros in terms of a classical Diophantine equation, the Pell equation.

VI. RELATION TO BINARY QUADRATIC FORMS

We have chosen to solve the Pell equation (3.1) in determining the weight-2 zeros of the 3j coefficients. This procedure required rejecting from the set of fundamental solutions those that failed to satisfy the divisibility condition (4.8). This step can be avoided by using a different formulation of the problem. This alternative, but closely related, method is of interest because it relates the properties of the weight-2 3jcoefficients directly to those of binary quadratic forms, which is a well-studied subject in number theory.²⁶⁻²⁸

A (real) binary quadratic form $q_{a,b,c}$ is a homogeneous polynomial in $(x,y) \in \mathbb{R}^2$ with coefficients $(a,b,c) \in \mathbb{R}^3$,

$$q_{a,b,c}(x,y) = ax^{2} + bxy + cy^{2}.$$
 (6.1)

The discriminant d of $q_{a,b,c}$ is

$$d = b^2 - 4ac. \tag{6.2}$$

The form is said to be *definite* if d < 0; *indefinite* if d > 0.

To transform $Q_{u_1,u_2}(x_1,x_2)$ [see Eq. (1.3)] to binary quadratic form, we eliminate only the linear terms. Thus carrying out the transformation

$$(u_1, u_2, x_1, x_2) = U \cdot (u_1, u_2, x, y), \tag{6.3a}$$

$$U = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 \\ 0 & \frac{1}{2} & \frac{1}{2} & 0 & -\frac{3}{2} \\ -\frac{1}{2} & 0 & 0 & \frac{1}{2} & \frac{1}{2} \\ 0 & 0 & 0 & 0 & 1 \end{pmatrix},$$
(6.3b)

we obtain the new polynomial F,

 $F(u_1, u_2, x, y) = Q_{u_1, u_2}(x_1, x_2) = q_{a,b,c}(x, y) - m.$ (6.4) The coefficients *a,b,c* are the following functions of u_1, u_2 :

$$a = u_1(u_1 - 1)/2, \quad c = u_2(u_2 - 1)/2,$$

$$b = -u_1u_2, \quad m = -(u_1 - 1)(u_2 - 1)(u_1 + u_2)/2.$$
(6.5)

The quadratic polynomial $P_{v_1,v_2}(y_1,y_2)$ may be similarly transformed:

$$(v_1, v_2, y_1, y_2) = V (v_1, v_2, x, y),$$

$$V = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 \\ \frac{1}{2} & 0 & \frac{1}{2} & 0 & 1 \\ 0 & \frac{1}{2} & 0 & \frac{1}{2} & 1 \\ 0 & 0 & 0 & 0 & 1 \end{pmatrix},$$
(6.6b)

$$G(v_1, v_2, x, y) = P_{v_1, v_2}(y_1, y_2) = q_{a,b,c}(x, y) - m, \quad (6.7)$$

with coefficients

$$a = (v_2 + 1)(v_2 + 2)/2, \quad c = (v_1 + 1)(v_1 + 2)/2,$$

$$b = (v_1 + 1)(v_2 + 1), \quad (6.8)$$

$$m = (v_1 + 2)(v_2 + 2)(v_1 + v_2 + 2)/2.$$

The discriminants of these two forms $q_{a,b,c}$ with coefficients (6.5) and (6.8), respectively, are

$$d_1(u_1, u_2) = u_1 u_2(u_1 + u_2 - 1),$$
 (6.9a)

$$d_2(v_1, v_2) = -(v_1 + 1)(v_2 + 1)(v_1 + v_2 + 3).$$
 (6.9b)

Consequently, for $u_1 \ge 1$, $u_2 \ge 2$ and $v_1 \ge 0$, $v_2 \ge 0$, the corresponding forms $q_{a,b,c}$ are indefinite and definite, respectively.

In studying the relations between the families of quadratic polynomials described above, we take the various parameters and coordinates to be defined individually on the real line; an *n*-tuple composed of these is taken to lie in \mathbb{R}^n ; for example,

$$(u_1, u_2, x, y) \in \mathbb{R}^4.$$
 (6.10)

With this extension, the relation

$$G(-u_2-1,-u_1-1,-x,y)=F(u_1,u_2,x,y)$$
(6.11)

is valid for the full domain (6.10). Consequently, there is only one family of binary quadratic forms occurring here; namely, the form (6.4) with coefficients given by Eqs. (6.5). The discriminants then also agree, $d_2(-u_2 - 1, -u_1 - 1) = d_1(u_1, u_2)$.

Relation (1.11) can also be expressed in terms of the variables (6.10). In terms of the transformation W defined by $(v_1, v_2, y_1, y_2) = W \cdot (u_1, u_2, x_1, x_2)$ [see Eqs. (1.9)], U defined by Eqs. (6.3), and V by Eqs. (6.6), the relation between G and F is

$$G(T \cdot (u_1, u_2, x, y)) = F(u_1, u_2, x, y), \qquad (6.12a)$$

where

$$T = V^{-1}WU = \begin{pmatrix} 1 & 1 & 0 & 0 & -2 \\ -\frac{1}{2} & -\frac{1}{2} & \frac{1}{2} & \frac{1}{2} & -1 \\ 1 & -1 & 0 & 0 & 0 \\ \frac{1}{2} & -\frac{1}{2} & \frac{1}{2} & -\frac{1}{2} & 0 \\ 0 & 0 & 0 & 0 & 1 \end{pmatrix}.$$
(6.12b)

Combining relations (6.11) and (6.12) we obtain the following theorem.

Theorem 6.1: The function F is invariant under the transformation S defined by

$$S = \begin{pmatrix} \frac{1}{2} & \frac{1}{2} & -\frac{1}{2} & -\frac{1}{2} & 0\\ -1 & -1 & 0 & 0 & 1\\ -1 & 1 & 0 & 0 & 0\\ \frac{1}{2} & -\frac{1}{2} & \frac{1}{2} & -\frac{1}{2} & 0\\ 0 & 0 & 0 & 0 & 1 \end{pmatrix};$$
(6.13a)

that is,

$$F(u_1, u_2, x, y) = F(S \cdot (u_1, u_2, x, y)).$$
(6.13b)

This symmetry of F is in addition to the eight Regge array symmetries of the group K defined by Eq. (1.15). The latter symmetry is expressed by $F(u_1, u_2, x, y) = F(m \cdot (u_1, u_2, x, y))$, each $m \in M = U^{-1}KU$. In particular, the interchange symmetry $u_1 \leftrightarrow u_2$, $x \leftrightarrow y$ of F is the transformation $U^{-1}rU$.

The group M' generated by the group M and the transformation S is a finite group of symmetries of the function F. This group will be described in detail in another paper dealing with the general symmetry group of the "polynomial part" of a 3j coefficient.

The group M' is the same for all $(u_1, u_2, x, y) \in \mathbb{R}^4$; that is, the group elements do not depend on the point in \mathbb{R}^4 . The function F has, in addition, a continuous group of symmetries at each point $(u_1, u_2) \in \mathbb{R}^2$. The nature of this group depends on the point (u_1, u_2) . Since this group structure belongs to an arbitrary binary quadratic form $q_{a,b,c}(x,y)$ defined by Eq. (6.1), we give its general formulation.

We define the graph A_d of the function $u^2 - dv^2 = 1$ by

$$\mathbb{A}_{d} = \{(u,v) | u^{2} - dv^{2} = 1\}, \tag{6.14}$$

where d is the discriminant of $q_{a,b,c}$. To each point $(u,v) \in \mathbf{A}_d$, we let correspond the 2×2 unimodular matrix

$$h(u,v) = \begin{pmatrix} u - bv & -2cv \\ 2av & u + bv \end{pmatrix},$$
(6.15)

where a,b,c are the coefficients in $q_{a,b,c}$, which we regard as specified. Then the set H defined by

$$H = \{h(u,v) | (u,v) \in \mathbb{A}_d\}$$
(6.16)

is a group under matrix multiplication, which may also be expressed as h(u,v)h(u',v') = h(uu' + vv'd,uv' + vu'); the identity is h(1,0), and $h(u,v)^{-1} = h(u, -v)$. [Here the coefficients have been suppressed in writing H and h(u,v).]

The principal result for the group H, which may be proved by direct substitution, is the following.

Theorem 6.2: The group H is an invariance group of the binary quadratic form $q_{a,b,c}$; that is,

$$q_{a,b,c}(x',y') = q_{a,b,c}(x,y), \qquad (6.17a)$$

for

$$\binom{x'}{y'} = h\binom{x}{y}, \quad \text{each } h \in H.$$
 (6.17b)

It is not difficult to prove that H is isomorphic to the Lorentz group SO(1,1) for d>0, to the rotation group SO(2) for d<0, and to a group $\Delta(2)$ of 2×2 triangular matrices for d=0.

The invariance group H is important for the study of the Diophantine equation

$$q_{a,b,c}(x,y) = m,$$
 (6.18)

where now $(a,b,c,x,y) \in \mathbb{Z}^5$ and $m \in \mathbb{Z}$ is in the range of $q_{a,b,c}$. The group *H* is significant because of its integer subgroups: integer points $(u,v) \in A_d$ correspond to elements $h(u,v) \in H$ with integer entries, and such elements of *H* map integer solutions of (6.18) to integer solutions. Not all integer elements of *H*, however, lead to distinct solutions. This is because all integer solutions of $u^2 - dv^2 = 1$ belong to a single orbit of the group $\{G_{0}, -G_0\}$, where G_0 is defined by Eq. (3.5) with $D = d = b^2 - 4ac$. The implication of this result is that only the subgroup $H_0 = \{h_0^n | n \in \mathbb{Z}\} \subset H$, where

$$h_0 = \begin{pmatrix} u_0 - bv_0 & -2cv_0 \\ 2av_0 & v_0 + bv_0 \end{pmatrix}, \tag{6.19}$$

is relevant in the classification of the set of solutions of Eq. (6.18) by the orbits of $\{H_0, -H_0\}$.

Let $(x,y) \in \mathbb{Z}^2$ be a solution of Eq. (6.18), if such exists. We can define a fundamental solution with respect to H_0 to be the least positive solution in the orbit $\pm H_0(x,y)$. (Any set of orbit representatives could serve as a set of fundamental solutions.) In this way, the problem of finding all integer solutions of Eq. (6.18) is reduced to that of finding the fundamental ones, which are always finite in number (including zero). For d > 0 and nonsquare (hyperbolic case), the matrix h_0 generates an infinite-order discrete group $H_0 \subset$ SO(1,1), and the infinitely many solutions of Eq. (6.18) are classified by a finite number of orbits $\pm H_0(x,y)$, one for each fundamental solution (x,y). For d > 0 and square (hyperbolic case) or for d < 0 (elliptical case), we have $(u_0,v_0) = (1,0)$ so that $h_0 = H_0 = I$, the identity matrix. The invariance groups $\{I, -I\} \subset SO(1,1)$ and $\{I, -I\} \subset SO(2)$ now yield only the points (x,y) and (-x, -y) in the same orbit.

In the application of the preceding results on binary quadratic forms to the special forms with coefficients (6.5) and (6.8), an extra richness of structure appears. This is brought out nicely by considering the surface \mathbb{F} in \mathbb{R}^4 defined by

$$\mathbb{F} = \{ (u_1, u_2, x, y) \in \mathbb{R}^4 | F(u_1, u_2, x, y) = 0 \}.$$
 (6.20)

The set of *lattice points* \mathbb{L} of this surface is the set of points of \mathbf{F} with integer coordinates,

$$\mathbb{L} = \{ (u_1, u_2, x, y) \in \mathbb{Z}^4 | F(u_1, u_2, x, y) = 0 \}.$$
 (6.21)

In particular, the set L contains all points that correspond to weight-2 zeros of the 3j coefficients, both in the hyperbolic and elliptical classifications [see Eqs. (6.3), (6.4), (6.6), (6.7), and (6.11)]. Unlike the general case (6.18), the set L is never empty, since it always contains the points

$$(u_1, u_2, u_2 + 1, u_1 - 1)$$
, all $(u_1, u_2) \in \mathbb{Z}^2$,

and

$$(u_1, u_2, u_2 + 1 + (4u_2/(u_1 - 1)), u_2 + 1)$$
, all $(u_1, u_2) \in \mathbb{Z}^2$
such that $u_1 - 1$ divides $4u_2$.

Each (u_1, u_2) plane in F has a symmetry group that is either isomorphic to SO(1,1) (d>0), to SO(2) (d<0), or to $\Delta(2)$ (d=0). If we restrict the surface F to its subset of lattice points L, the relevant group for the (u_1, u_2) plane, with $(u_1, u_2) \in \mathbb{Z}^2$, is restricted to the discrete subgroup $\{H_0, -H_0\}$, so that all points in L can be classified by orbits relative to this subgroup. (This is, to be sure, a rather trivial classification in some cases.)

In addition to the group structure of points in a (u_1, u_2) plane described above, there is also the symmetry of the surface F corresponding to the transformations belonging to the finite group M' (see Theorem 6.1 *et seq.*) These transformations map points from one (u_1, u_2) plane to another. The following example illustrates this property.

Take $(u_1, u_2) = (6, 2)$. This is our earlier example [see Eqs. (5.13)-(5.18)] rephrased in terms of binary quadratic forms. We have

$$F(6,2,x,y) = 15x^2 - 12xy + y^2 + 20 = 0.$$
 (6.22)

The (6,2) plane in \mathbb{L} contains an infinite number of lattice points given by the orbits $\pm H_0(x_1,y_1)$, where $(x_1,y_1)\in\mathbb{B}(6,2)$. These orbits, of course, constitute all integer solutions of Eq. (6.22).

Under the transformation S given by (6.13a), an integer solution (x,y) of (6.22) is mapped to an integer solution of

$$F(u'_1, u'_2, x', y') = 0, (6.23a)$$

where

$$(u'_1, u'_2) = (4 - \frac{1}{2}(x + y), -7),$$
 (6.23b)

$$(x',y') = (-4,2+\frac{1}{2}(x-y)).$$
 (6.23c)

The point (u'_1, u'_2, x', y') belongs to L. The original (6,2) plane is, however, hyperbolic, while for $|x + y| \ge 10$, the (u'_1, u'_2) plane is elliptical. The integer solutions (infinite in number) of

$$q_{15,-12,1}(x,y) = -20 \tag{6.24a}$$

are distributed by the map S into integer solutions, one each, of the infinite number of Diophantine equations.

$$q_{a',b',c'}(-4,2+\frac{1}{2}(x-y))=m',$$
 (6.24b)

where a', b', c', and m' are expressed in terms of (u'_1, u'_2) by Eqs. (6.5).

The properties of the function $F(u_1, u_2, x, y)$ discussed in this section are intended only to indicate the deep structure associated with the polynomial part of a 3*j* coefficient. We have not implemented methods discussed in Refs. 26–28 of solving the Diophantine equation $F(u_1, u_2, x, y) = 0$, since from a calculational viewpoint this would repeat much of Secs. III-V, which have been based directly on Pell's equation.

ACKNOWLEDGMENTS

We were assisted by our colleague W. A. Beyer in formulating the method of solving Pell's equation in Secs. II and III (see Ref. 12).

This work was performed under the auspices of the U.S. Department of Energy under Contract No. W-7405-ENG-36 and the Office of Basic Energy Sciences, Department of Applied Mathematics.

¹A. Lindner, *Drehimpulse in der Quantenmechanik* (Teubner, Stuttgart, 1984).

²S. Brudno, "Nontrivial zeros of the Wigner (3-j) and Racah (6-j) coefficients. I. Linear solutions," J. Math. Phys. 26, 434 (1985).

³S. Brudno and J. D. Louck, "Nontrivial zeros of the Racah quadrupole invariant," J. Math. Phys. 26, 1125 (1985).

- ⁴K. S. Rao, "A note on the classification of the zeros of angular momentum coefficients," J. Math. Phys. 26, 2260 (1985).
- ⁵S. Brudno and J. D. Louck, "Nontrivial zeros of weight 1 3*j* and 6*j* coefficients: Relation to Diophantine equations of equal sums of like powers," J. Math. Phys. **26**, 2092 (1985).
- ⁶A. Lindner, "Non-trivial zeros of the Wigner (3*j*) and Racah (6*j*) coefficients," J. Phys. A: Math. Gen. 18, 3071 (1985).
- ⁷A. Brenner, "On Diophantine equations and nontrivial Racah coefficients," J. Math. Phys. 27, 1181 (1986).
- ⁸A. Bremner and S. Brudno, "A complete determination of the zeros of weight-1 6*j* coefficients," J. Math. Phys. 27, 2613 (1986).
- ⁹J. J. Labarthe, "Parametrization of the linear zeros of 6*j* coefficients," J. Math. Phys. 27, 2964 (1986).
- ¹⁰K. Srinivasa Rao and V. Rajeswari, "An algorithm to generate the polynomial zeros of degree one of the Racah coefficients," J. Phys. A: Math Gen. 20, 507 (1987).
- ¹¹S. Brudno, "Nontrivial zeros of the Wigner (3*j*) and Racah (6*j*) coefficients. II. Some nonlinear solutions," J. Math. Phys. 28, 124 (1987).
- ¹²W. A. Beyer, J. D. Louck, and P. R. Stein, "Zeros of the Racah coefficients and the Pell equation," Acta Appl. Math. 7, 257 (1986).
- ¹³L. C. Biedenharn and J. D. Louck, Angular Momentum in Quantum Physics, Vol. 8; The Racah-Wigner Algebra in Quantum Theory, Vol. 9, Encyclopedia of Mathematics and Its Applications, edited by G. C. Rota (Addison-Wesley, Reading, MA, 1981).
- ¹⁴G. Racah, "Theory of complex spectra. I," Phys. Rev. **61**, 186 (1942); II, **62**, 438 (1942); III, **63**, 368 (1943); IV, **76**, 1352 (1949).
- ¹⁵B. R. Judd, "Topics in atomic theory," in *Topics in Atomic and Nuclear Theory* (Caxton, Christchurch, New Zealand, 1970), pp. 1–60.
- ¹⁶G. Vanden Berghe, H. DeMeyer, and J. Van der Jeugt, "Tensor operator

realizations of E_6 and structural zeros of the 6*j*-symbol," J. Math. Phys. 25, 2585 (1984).

- ¹⁷S. H. Koozekanani and L. C. Biedenharn: "Non-trivial zeros of the Racah (6-j) coefficient," Rev. Mex. Fiz. 23, 327 (1974).
- ¹⁸M. J. Bowick, "Regge symmetries and null 3-j and 6-j symbols," thesis, University of Canterbury, Christchurch, New Zealand, 1976.
- ¹⁹A. de-Shalit and I. Talmi, *Nuclear Shell Theory* (Academic, New York, 1963).
- ²⁰J. N. Ginocchio, "A schematic model for monopole and quadrupole pairing in nuclei," Ann. Phys. **126**, 214 (1980).
- ²¹T. Regge, "Symmetry properties of Clebsch-Gordan coefficients," Nuovo Cimento 10, 544 (1958).
- ²²V. Bargmann, "On the representations of the rotation group," Rev. Mod. Phys. 34, 829 (1962).
- ²³W. J. LeVeque, *Topics in Number Theory* (Addison-Wesley, Reading, MA, 1956), Vol. 1.
- ²⁴B. Stolt, "On the Diophantine equation $u^2 Dv^2 = \pm 4N$. I," Ark. Mat. 2, 1 (1951); II, 2, 251 (1952); III, 3 (1954). See also P. Heichelheim, "Some remarks on Stolt's theorem for Pellian equations," Ark. Mat. 12, 167 (1974).
- ²⁵R. Kortum and G. McNiel, "A table of periodic continued fractions," Lockheed Missiles and Space Division, 1961, Technical Report AD 640524, Parts I and II (distributed by DTIC, Cameron Station, Alexandria, VA, 22314).
- ²⁶E. Landau, Elementary Number Theory (Chelsea, New York, 1958).
- ²⁷E. D. Dickson, *Introduction to the Theory of Numbers* (University of Chicago Press, Chicago, 1929).
- ²⁸T. Nagell, Introduction to Number Theory (Wiley, New York, 1951).

Construction of space-time by gauge translations

P. K. Smrz

Department of Mathematics, University of Newcastle, Newcastle, N. S. W. 2308, Australia

(Received 4 February 1987; accepted for publication 13 August 1987)

The following procedure is described: Starting with a connection in a principal fiber bundle P(M,G), where G is either the Poincaré group or one of the de Sitter groups, a connection in the bundle of linear frames of a submanifold N of M is constructed by using the translational components of the original connection for frame identification. The dimension of N is gauge dependent, and the flat four-dimensional Minkowski space-time may appear of dimension less than 4 when certain gauges are used. It is shown that in the case of a de Sitter group, the minimum dimension to which the flat four-dimensional space-time can be reduced is 1, while the number is 0 for the Poincaré group. The gauge transformation that achieves the maximum dimension reduction in the de Sitter case is constant and leads to infinite strings as a result. Variable continuous gauge transformations that can reduce the dimension over a finite region of the base manifold are also considered.

I. INTRODUCTION

Ever since Kibble¹ proposed a theory with translations included in the gauge group of general relativity, the subject has been extremely popular. The emphasis has been, however, on the introduction of torsion into general relativity (Riemann-Cartan geometry) and the construction of Lagrangians. Some examples of the literature are given in Ref. 2.

In this paper it is the geometry of the construction that is primarily studied, expanding a short note listed in Ref. 3. The main geometric role played by the translations in the gauge group is that it can provide the necessary "soldering,"4 i.e., it changes a principal fiber bundle with no special relationship between the points on the fibers and the base manifold into the bundle of linear frames of the base manifold. The soldering is based on an identification of the translational component of the original connection form with the canonical form of the bundle of linear frames. The identification is made on a partially fixed cross section subject only to Lorentz gauge transformations. Since the Poincaré group and de Sitter groups are indistinguishable as far as the adjoint action of the Lorentz subgroup is concerned, the soldering can be done in the same way for all three cases. However, the procedure depends on the selection of the reference cross section, and that is where the difference between the Poincaré and the de Sitter groups shows up. When the rank of the translational component of the connection form is less than 4, full soldering is not possible, but in some cases it can still be carried out on a submanifold of the base manifold. In particular, a connection may yield the flat four-dimensional space-time in one gauge, and a space of a lower dimension in another gauge. One can ask about the minimum dimension to which the flat four-dimensional space-time can be reduced by a transformation. In the de Sitter case, which is, in general, geometrically more interesting than the Poincaré case, the minimum dimension is 1. This could be somehow connected with the fact that the most elementary constituents of matter seem to have the appearance of one-dimensional strings.

Section II introduces the construction of a bundle of linear frames from the principal fiber bundle with Poincaré

or de Sitter structure using the formalism of the modern differential geometry.⁵ Section III discusses a physical interpretation of the construction. Section IV describes explicitly the construction of Sec. II in terms of coordinates, while Sec. V is concerned with the construction of a flat Minkowski space and its gauge dependence. Section VI gives an example of a partial dimension reduction in spherical coordinates.

II. SOLDERING BY GAUGE TRANSLATIONS

Chapter III of Ref. 5 describes the reduction of the bundle of affine frames to the bundle of linear frames and the affine connection. Since it is closely related to the construction described later, it will be briefly reviewed. A linear frame of an *n*-dimensional manifold M is a basis $(X_1,...,X_n)$ of the tangent vector space $T_x(M)$ at $x \in M$. An affine frame is a linear frame together with a point in $T_x(M)$ regarded as an affine space. There is a natural map from the bundle of linear frames L(M) into the bundle of affine frames A(M)defined by

$$\gamma(X_1,...,X_n) = (0_x;X_1,...,X_n), \qquad (2.1)$$

where 0_x is the origin of $T_x(M)$. If $\tilde{\omega}$ is a connection form on A(M) then its image on L(M) splits naturally into Gl (n, \mathbb{R}) components,

$$\gamma^* \widetilde{\omega} = \omega + \phi \,. \tag{2.2}$$

Every linear frame $(X_1,...,X_n)$ at $x \in M$ can be considered as a linear map u of \mathbb{R}^n onto $T_x(M)$ by

$$u(a^{1},...,a^{n}) = \sum_{i=1}^{n} a^{i} X_{i} .$$
(2.3)

The canonical form θ on L(M) is the \mathbb{R}^n -valued form defined by

$$\theta(X) = u^{-1}(\pi(X)), \qquad (2.4)$$

where X is a tangent vector in the bundle manifold of L(M), and π is the canonical projection in L(M) from the bundle manifold to the base manifold. It is easy to see that

$$(R_{a}^{*}\theta)(X) = a^{-1}(\theta(X)), \qquad (2.5)$$

where R_a is the right action of $a \in Gl(n, \mathbb{R})$ on L(M). Since the form defined by (2.2) has the same transformation law

(2.5) as θ , it is possible to define an affine connection in A(M) as a connection for which $\phi = \theta$. The curvature form $\tilde{\Omega}$ of an affine connection in A(M) decomposes under the injection γ defined by (2.1) according to

$$\gamma^* \hat{\Omega} = \Omega + \Theta \,, \tag{2.6}$$

where Ω and Θ are, respectively, the curvature and torsion forms on L(M).

If an affine connection in A(M) is reducible to a Poincaré type subgroup of the affine group, then the linear connection in L(M) is reducible to the corresponding Lorentz type subgroup of $Gl(n, \mathbb{R})$.

The aim of this section is to use the above for defining a construction of a bundle of linear frames from a principal fiber bundle P(M,G), where M is a four-dimensional manifold, and G is the Poincaré group. Only local aspects of the construction will be considered in this paper. Thus the bundle manifold P can be considered as $M \times G$, and a subbundle Q(M,H) of P can be identified. Here $Q = M \times H$ and H is the Lorentz subgroup of G. If γ is the natural injection of Q into P and $\tilde{\omega}$ is a connection form on P,

$$\gamma^* \omega = \omega + \theta \,, \tag{2.7}$$

where ω and θ are forms on Q with values in the Lorentz algebra and \mathbb{R}^n , respectively. Under the right action of $a \in H$, θ behaves exactly as in (2.5). It can thus be identified with a canonical form of a bundle of linear frames, which is already reduced to a Lorentz structure. By the identification it is meant that (2.4) is used to determine u^{-1} , which in its turn defines the frames by (2.3). To ensure the existence of u we must have a condition

$$\theta(X) = 0 \Longrightarrow \pi(X) = 0.$$
(2.8)

Let us now assume that

$$\theta(X) = 0 \quad \text{if } \pi(X) \in U_x , \qquad (2.9)$$

where U_x is a subspace of $T_x(M)$. Selecting another subspace V_x of $T_x(M)$ such that

$$T_x(M) = U_x \oplus V_x$$
 and $\theta(X) \neq 0$
if $\pi(X) \neq 0$ and $\pi(X) \in V_x$,

an invertible map v^{-1} : $V_x \to \mathbb{R}^p$ can be defined by

$$\rho(\theta(X)) = v^{-1}(\pi(X)), \quad \pi(X) \in V_x , \quad (2.10)$$

where ρ is a canonical projection $\mathbb{R}^4 \to \mathbb{R}^p$, $p = \text{Dim } V_x$. If V_x defines an involutive distribution on M then there exists (locally) a submanifold N of M such that $T_x(N) = V_x$. In that case map v can be used to define linear frames on N. A connection in L(N) is defined by ω with values restricted to the subgroup of the Lorentz group determined by projection ρ .

Several comments about the above described construction are in order.

(i) Depending on the original connection in P(M,G) it may or may not be possible to construct submanifold N. When the construction is possible, it is in general not unique. The dimension of N, however, is unique since it is determined by the rank of θ .

(ii) The construction does not work only for the Poincaré group in the role of G. If G is ten-dimensional and its Lie algebra can be written as

$$\mathsf{G} = \mathsf{H} \oplus \mathsf{T}, \tag{2.11}$$

where H is the Lie algebra of the Lorentz group and T is an *h*dimensional subspace of G that transforms under the adjoint map ad H in the same way as the translations in the Poincaré group, then G can be used for the construction. Indeed, injection γ will still exist, $\gamma^* \tilde{\omega}$ will still decompose as in (2.7), and it will still have the correct behavior under the right action of H. Since the construction is on the subbundle Q(M,H), the Lie aglebraic properties of T are not important, except for its transformation under ad H. In this way G can be also a de Sitter group of type either (4,1) or (3,2).

(iii) The construction may be possible also when M has dimension higher than 4. Since θ can have rank at most 4, some dimension reduction must occur, and the success of the construction will depend on the existence of the involutive distribution V_x .

(iv) Most importantly, the construction depends on the particular way in which P is represented as $M \times G$. Such a representation involves a selection of a reference cross section corresponding to the identity element of G. In physical applications this is called gauge, and a change from one reference cross section to another is called a gauge transformation. Gauge invariance of the construction is retained only as far as a Lorentz gauge transformation is concerned (or the appropriate subgroup of the Lorentz group in case of a lower dimension), but a gauge transformation involving translations can change the result, including the final dimension of the constructed submanifold. Gauge dependent results are usually not studied in differential geometry. After all, a gauge transformation is just a special kind of a coordinate transformation, and the true modern differential geometry is concerned only with the properties that can be described in a coordinate-free fashion. Alas, in physical applications the selection of a gauge and gauge transformations often play an important role and, as discussed in the next section, the role could be crucial in case of translations.

III. PHYSICAL INTERPRETATION

Before the construction described in Sec. II is demonstrated on particular examples, let us discuss the physical motivation and interpretation of the results. For a physical understanding of concepts like connection and reference cross section, it is better to introduce the connection via the horizontal lift, rather than the connection form.

If P(M,G) is a principal fiber bundle with a connection form ω , the horizontal lift $X^{(h)}$ of $X \in T_x(M)$ is the unique tangent vector at $p \in P$, $x = \pi(p)$, satisfying

$$\pi(X^{(h)}) = X$$
 and $\omega(X^{(h)}) = 0$. (3.1)

Physically, transport in the horizontal direction implies no measurable change of the geometrical properties described by G. For example, if P(M,G) were a bundle of linear frames of a three-dimensional manifold and G the group of rotations, a frame transported in the horizontal direction would be perceived as parallel to the original frame, i.e., without any observable rotation.

Thus $\theta(X) = 0$ of Sec. II can be interpreted as "no observable translation" in the direction of $\pi(X)$. Dimension of

M is then just the number of parameters needed to describe physical fields, while the dimension of the submanifold N is the observable physical dimension. Our macroscopical experience with space-time tells us that the observable dimension is 4. Since the dimension depends on the selected reference cross section (gauge), it must be assumed that the gauge is to some extent fixed. It is not too surprising, since even the Lorentz gauge, though theoretically free to choose, would be difficult to set up practically if its Lorentz frames involved high velocities. Hence it is possible that our macroscopical methods of measuring space-time intervals restrict us to a fixed gauge as far as translations are concerned. Such restrictions may not apply in the microworld of elementary particles, where the same space-time may appear to have a different dimension due to the fact that it is viewed from a different gauge.

IV. EXPRESSION IN LOCAL COORDINATES

To investigate particular cases one has to formulate everything explicitly, using local coordinate systems.

Let *M* be a four-dimensional manifold with a local coordinate system $(x^{\mu}, \mu = 1,...,4)$. Consider a principal fiber bundle P(M,G), where *G* is either the Poincaré group or one of the de Sitter groups. In a particular gauge, a connection in *P* can be defined by the horizontal lift of $\partial / \partial x^{\mu} \in T_x(M)$,

$$X_{\mu}^{(h)} = \frac{\partial}{\partial x^{\mu}} - \frac{1}{2} A_{\mu}^{ij}(x) W_{ji} - A_{\mu}^{i}(x) T_{i}. \qquad (4.1)$$

Here W_{ji} and T_i are the right invariant vector fields in G, with W_{ji} spanning the Lorentz algebra. Functions $A_{\mu}^{ij}(x)$ are the Lorentz components of the connection, while $A_{\mu}^{i}(x)$ correspond to the form θ of (2.7) by

$$\theta^{i}(X_{\mu}^{(h)}) = A_{\mu}^{i}(x) . \tag{4.2}$$

The coordinates in the Poincaré group can be chosen as the elements of 5×5 matrices

$$\begin{bmatrix} X_{j+}^{i+} \\ Y_{j+} \\ Y_{i+} \end{bmatrix},$$
(4.3)

where

$$X_{j}^{i}g_{il}X_{k}^{l} = g_{jk},$$

$$g_{ik} = \text{diag}(1,1,1,-1).$$
(4.4)

The right-invariant vector fields are then

$$W_{ij} = g_{ik} W_j^k - g_{jk} W_i^k, \quad W_j^k = X_m^k \frac{\partial}{\partial X_m^j}, \quad (4.5)$$

and

$$T_i = g_{ij} X^j_k \frac{\partial}{\partial Y_k}.$$
(4.6)

The coordinates in a de Sitter group can be chosen as the elements of 5×5 matrices $[X_{q}^{p}]$ satisfying

$$X_{r}^{p}g_{pg}X_{s}^{q} = g_{rs} ,$$

$$g_{rs} = \text{diag}(1,1,1,-1,\pm 1) .$$
(4.7)

The right-invariant vector fields are then

$$W_{pq} = g_{pr}W_q^r - q_{qr}W_p^r, \quad W_p^r = X_s^r \frac{\partial}{\partial X_s^p}, \quad (4.8)$$

where the restriction of indices to the range 1, 2, 3, 4 defines W_{ij} , while

$$T_i = W_{5i} . \tag{4.9}$$

A gauge transformation is characterized by the right action of group G on the reference cross section with the group element being a function of position in M. This causes a change of the coordinate system in $P = M \times G$ with the new coordinates in G measuring the position on the fibers from the new cross section. If the variable group elements are characterized by $a_j^i(x)$ and $a_j(x)$ in case of the Poincaré case and by $a_q^p(x)$ in the de Sitter cases, the respective transformations are

$$\overline{x}^{\mu} = x^{\mu}, \quad \overline{X}_{j}^{i} = b_{k}^{i}(x)X_{j}^{k},$$

$$\overline{Y}_{j} = Y_{j} - a_{i}(x)b_{k}^{i}(x)X_{j}^{k},$$
(4.10)

and

$$X_{q}^{p} = b_{r}^{p}(x)X_{q}^{r}.$$
(4.11)

Here matrices $\begin{bmatrix} b_j^i \end{bmatrix}$ and $\begin{bmatrix} b_q^p \end{bmatrix}$ are inverses of $\begin{bmatrix} a_j^i \end{bmatrix}$ and $\begin{bmatrix} a_q^p \end{bmatrix}$.

Applying the transformations (4.10) and (4.11) to the horizontal lift (4.1) yields the gauge transformation of the connection components.

For the Poincaré case

$$\overline{A}_{\mu}^{ij} = b_{k}^{i} A_{\mu}^{kl} b_{l}^{j} + (\partial_{\mu} b_{k}^{j}) b_{l}^{i} g^{kl},
\overline{A}_{\mu}^{i} = b_{k}^{i} A_{\mu}^{k} + b_{k}^{i} A_{\mu}^{kl} b_{l}^{j} a_{j} + g^{il} (\partial_{\mu} a_{l})
+ g^{il} b_{k}^{i} (\partial_{\mu} b_{k}^{j}) a_{i}.$$
(4.12)

For the de Sitter case

$$\overline{A}_{\mu}^{pq} = b_{r}^{p} A_{\mu}^{rs} b_{s}^{q} + (\partial_{\mu} b_{r}^{q}) b_{s}^{p} g^{rs}, \qquad (4.13)$$

where

 $A_{\mu}^{i5} \equiv A_{\mu}^{i}$ of (4.1).

Restricting the gauge transformation to the Lorentz subgroup means $a_j(x) = 0$ for the Poincaré group, and $a_5^i(x) = a_j^5(x) = 0$, $a_5^5(x) = 1$ for the de Sitter groups. Of course, the transformation of A_{μ}^i is then identical in all cases and given by

$$\overline{A}_{\mu}^{i}(x) = b_{k}^{i}(x)A_{\mu}^{k}(x), \qquad (4.14)$$

which is just an explicit expression for the transformation (2.5) of the canonical form.

Curvature of the connection in P can be calculated by taking the commutator

$$[X_{\mu}^{(h)}, X_{\nu}^{(h)}] = \frac{1}{2} R_{\mu\nu}^{ij} W_{ji} + S_{\mu\nu}^{i} T_{i} . \qquad (4.15)$$

The two terms on the right-hand side of (4.15) correspond to the decomposition (2.6).

V. FLAT SPACE CONSTRUCTION AND ITS GAUGE DEPENDENCE

Let us now assume that in a particular gauge the connection describes a flat Minkowski space. Explicitly it means that the connection is described by a horizontal lift of the form

$$X_{\mu}^{(h)} = \frac{\partial}{\partial x^{\mu}} - A_{\mu}^{i}(x)T_{i}, \qquad (5.1)$$

P. K. Smrz 2826

where the matrix $[a^i_{\mu}]$ is invertible for all x, and

$$\partial_{\mu}A^{i}_{\nu} - \partial_{\nu}A^{i}_{\mu} = 0.$$
 (5.2)

Here $A^{i}_{\mu}(x)$ are proportional to the partial derivatives of Minkowski coordinates x^{i} , i = 1,...,4, with respect to general coordinates x^{μ} , $\mu = 1,...,4$. If the coordinates are considered as measured in units of length, we can write

$$A^{i}_{\mu}(x) = \frac{1}{l} \frac{\partial x^{i}}{\partial x^{\mu}}, \qquad (5.3)$$

where *l* is a constant length.

Calculating the curvature of the connection defined by (5.1), according to (4.15) one obtains

$$R^{ij}_{\mu\nu} = S^{i}_{\mu\nu} = 0$$
(5.4)
for the Poincaré group and

$$R^{ij}_{\mu\nu} = \pm (A^{i}_{\nu}A^{j}_{\mu} - A^{i}_{\mu}A^{j}_{\nu}), \quad S^{i}_{\mu\nu} = 0$$
 (5.5)

for the de Sitter groups. The curvature of the connection in L(M) generated according to the construction described in Sec. II is, of course, 0 in both cases.

Under a gauge transformation the components of the curvature in (4.15) transform according to the action of ad $a^{-1}(x)$, $a(x)\in G$. In the Poincaré case, $R^{ij}_{\mu\nu}$ and $S^{i}_{\mu\nu}$ remain 0 in all gauges. For de Sitter groups

$$\overline{S}_{\mu\nu}^{i} = \pm b_{k}^{i} (A_{\nu}^{k} A_{\mu}^{l} - A_{\mu}^{k} A_{\nu}^{l}) b_{l}^{5}, \qquad (5.6)$$

and requirement $\overline{S}_{\mu\nu}^{i} = 0$ leads to $b_{1}^{5} = 0$, $b_{5}^{i} = 0$, and $b_{5}^{5} = 1$. Thus only a Lorentz gauge transformation is permitted if $S_{\mu\nu}^{i}$ is to remain 0. At the same time, the induced connection in L(M) is not changed by Lorentz gauge transformations. Thus in the de Sitter case, the flat connection in L(M) with $S_{\mu\nu}^{i} = 0$ is unique. The geometric meaning of $S_{\mu\nu}^{i}$ in L(M) is that of torsion. Hence we can state the following.

If a flat connection is generated from a connection in P(M,G) with G being a de Sitter group, the gauge is determined up to a Lorentz gauge transformation by the requirement of zero torsion.

This is quite pleasing from the physical point of view, since it hints on a possible relationship between space-time measurements and the choice of the gauge. Macroscopical measurements of space and time are based on the use of geodesics (light rays) and a connection constructed from geodesics (using the Schild's ladder) comes automatically with zero torsion.⁶

Let us now investigate how the dimension of submanifold N constructed from the connection described by (5.1) depends on the choice of the gauge. In particular, we should be interested in the minimum dimension to which the flat four-dimensional space-time can be reduced, and in the gauge that can achieve the maximum reduction.

In the Poincaré case the answer is rather trivial. Using Minkowski coordinates $(A_{\mu}^{i} = (1/l)\delta_{\mu}^{i}), \overline{A}_{\mu}^{i} = 0$ in (4.12) leads to

$$b_k^i = \delta_k^i, \quad a_l = -(1/l)g_{lk}x^k.$$
 (5.7)

Thus the rank of θ is 0, and so is the minimum dimension. This is not surprising, since the original connection in P(M,G) is flat. In the de Sitter case it is easy to see that the minimum dimension must be greater than 0. Namely, assume that it is 0. Then the horizontal lift (4.1) has only the Lorentz component in the appropriate gauge. But then the curvature has also only the Lorentz component, and that is impossible since $S^{i}_{\mu\nu}$ can be 0 only in the original gauge. Thus the minimum dimension must be at least 1. The direct way to show that it is in fact equal to 1 is to solve the equations $\overline{A}^{i5}_{\mu} = 0$, i = 1,...,4, for as many values of index μ as possible. Separating index 5 from the remaining values Eq. (4.13) reads

$$A^{i}_{\mu} = b^{j}_{k}A^{k}_{\mu}b^{5}_{5} - b^{i}_{5}A^{k}_{\mu}b^{5}_{k} + (\partial_{\mu}b^{5}_{5})b^{i}_{5}g^{55} + (\partial_{\mu}b^{5}_{i})b^{i}_{k}g^{jk} = 0.$$
(5.8)

Working in Minkowski coordinates we have for, say $\mu = 1$, $(1/l)b_1^i b_5^5 - (1/l)b_5^i b_5^5 + b_5^i (\partial_1 b_5^5)g^{55}$

$$+ b_{k}^{i} (\partial_{1} b_{j}^{5}) g^{jk} = 0.$$
 (5.9)

After multiplication by matric $[a_a^p]$ one obtains

$$(1/l)\delta_1^p - (1/l)\delta_5^p b_1^5 + \delta_5^p (\partial_1 b_5^5) g^{55} + g^{pj} (\partial_1 b_j^5) = 0,$$

for p = 1,...5, which can be written as

$$(1/l)b_{5}^{5} + g^{11}(\partial_{1}b_{1}^{5}) = 0,$$

- $(1/l)b_{1}^{5} + g^{55}(\partial_{1}b_{5}^{5}) = 0,$ (5.10)
 $\partial_{1}b_{2}^{5} = \partial_{1}b_{3}^{5} = \partial_{1}b_{4}^{5} = 0.$

Similar sets of equations must be valid for every dimension μ for which $\overline{A}_{\mu}^{i} = 0$. If only one more dimension is added, say $\mu = 2$, the resulting set of equations leads to b_{1}^{5} $= b_{2}^{5} = b_{5}^{5} = 0$. The process can be extended by one more dimension. The third dimension cannot be chosen arbitrarily, since $\begin{bmatrix} b_{q}^{p} \end{bmatrix}$ must belong to the particular de Sitter group and due to $b_{5}^{5} = 0$ the equation

$$b_{i}^{5}g^{ij}b_{j}^{5} = g^{55}$$
(5.11)

must be satisfied. Thus if the de Sitter group is the type (4,1) (e.g., $g^{55} = +1$), the maximum dimension reduction can be achieved by

$$b_1^5 = b_2^5 = b_4^5 = b_5^5 = 0, \quad b_3^5 = 1,$$

while if the type is (3,2) $(g^{55} = -1)$ then we have

 $b_1^5 = b_2^5 = b_3^5 = b_5^5 = 0, \quad b_4^5 = 1.$

In both cases the minimum dimension is 1, but its character (spatial or temporal) depends on the type of the de Sitter group.

VI. DIMENSION REDUCTION IN SPHERICAL COORDINATES

The maximum reduction described in the previous section is allowed by the ultimate symmetry of the flat spacetime expressed in Cartesian coordinates. The gauge transformation is also constant and cannot be continuously changed into the identity. It is of interest to study variable transformations that can reduce the dimension of a finite region of space-time while remaining everywhere continuous. In particular, eliminating the radial distance could lead to a model of a pointlike particle that would at the same time occupy a finite region of the base manifold and thus preserve an internal structure. This could be used also for dealing with classical singularities caused by pointlike character of particles. The Schwarzschild metric should be investigated in this manner, but here only the reduction of a flat space-time metric in spherical coordinates will be considered as a preliminary study.

In the initial gauge we can choose

$$A_r^1 = (1/l)\cos\theta, \quad A_r^2 = (1/l)\sin\theta\sin\phi, \quad (6.1)$$
$$A_r^3 = (1/l)\sin\theta\cos\phi, \quad (6.1)$$

and

$$A_r^{ij} = 0$$
 for all $i, j = 1,...,4$. (6.2)

The same reasoning that lead to Eqs. (5.10) now results in

$$A_{r}^{i}b_{5}^{5} + g^{ij}(\partial_{r}b_{j}^{5}) = 0, \quad i = 1,2,3,$$

$$g^{4j}(\partial_{r}b_{j}^{5}) = 0, \quad -A_{r}^{k}b_{k}^{5} + (\partial_{r}b_{5}^{5})g^{55} = 0.$$
(6.3)

Eliminating
$$b_i^{5}$$
 yields

$$g^{55}(\partial_r^2 b_5^5) + (1/l^2)b_5^5 = 0, \qquad (6.4)$$

with the general solution

$$b_5^5 = a\cos(r/l) + b\sin(r/l)$$
, (6.5)

if $g^{55} = +1$ [the case of (4,1) de Sitter group], and

$$b_5^5 = ae^{r/l} + be^{-r/l}, (6.6)$$

if $g^{55} = -1$ [the case of (3,2) de Sitter group].

For example, a complete solution for the (4,1) case can be written as

$$\begin{bmatrix} b_{q}^{p} \end{bmatrix} = \begin{bmatrix} 0 & \cos\phi & -\sin\phi & 0 & 0\\ \sin\theta & -\cos\theta\sin\phi & -\cos\theta\cos\phi & 0 & 0\\ \cos(r/l)\cos\theta & \cos(r/l)\sin\theta\sin\phi & \cos(r/l)\sin\theta\cos\phi & 0 & \sin(r/l)\\ 0 & 0 & 0 & 1 & 0\\ -\sin(r/l)\cos\theta & -\sin(r/l)\sin\theta\sin\phi & -\sin(r/l)\sin\theta\cos\phi & 0 & \cos(r/l) \end{bmatrix}.$$
(6.7)

At $r = 2\pi nl$, *n* integer, the gauge transformation (6.7) reduces to a 3×3 transformation of spatial frames. The fact that it does not reach the identity is due to the nonexistence of a global frame selection on a spherical surface, and it could be fixed by using an appropriate patching. The important feature is that the translational components of the gauge transformation disappear on the surface of radius $2\pi nl$. Thus it is possible to have a continuous gauge transformation that leaves the outside of the sphere as the four-dimensional space-time, while the inside region loses one observable dimension, namely the radial distance.

- ²F. W. Hehl, P. von der Heyde, G. D. Kerlick, and J. M. Nester, Rev. Mod. Phys. **48**, 393 (1976); K. Hayashi and T. Shirafuji, Prog. Theor. Phys. **64**,
- 866 (1980); R. P. Wallner, Acta Phys. Austriaca 54, 165 (1982).
- ³P. K. Smrz, Lett. Nuovo Cimento 38, 141 (1983).
- ⁴A. Trautman, Rep. Math. Phys. 1, 29 (1970).
- ⁵S. Kobayashi and K. Nomizu, *Foundations of Differential Geometry* (Interscience, New York, 1963), Vol. 1.
- ⁶C. W. Misner, K. S. Thorne, and J. A. Wheeler, *Gravitation* (Freeman, San Francisco, 1973), p. 249.

¹T. W. B. Kibble, J. Math. Phys. 2, 212 (1961).

Quantization: Towards a comparison between methods

G. M. Tuynman

Mathematisch Instituut, Roetersstraat 15, NL 1018 WB, Amsterdam, The Netherlands

(Received 16 April 1987; accepted for publication 5 August 1987)

In this paper it is shown that the procedure of geometric quantiztion applied to Kähler manifolds gives the following result: the Hilbert space H consists, roughly speaking, of holomorphic functions on the phase space M and to each classical observable f (i.e., a real function on M) is associated an operator f on \mathcal{H} as follows: first multiply by $f + \frac{1}{4}\hbar\Delta_{dR}f(\Delta_{dR})$ being the Laplace-de Rham operator on the Kähler manifold M) and then take the holomorphic part [see G. M. Tuynman, J. Math. Phys. 27, 573 (1987)]. This result is correct on compact Kähler manifolds and correct modulo a boundary term $\int_M d\alpha$ on noncompact Kähler manifolds. In this way these results can be compared with the quantization procedure of Berezin [Math. USSR Izv. 8, 1109 (1974); 9, 341 (1975); Commun. Math. Phys. 40, 153 (1975)], which is strongly related to quantization by *-products [e.g., see C. Moreno and P. Ortega-Navarro; Ann. Inst. H. Poincaré Sec. A: 38, 215 (1983); Lett. Math. Phys. 7, 181 (1983); C. Moreno, Lett. Math. Phys. 11, 361 (1986); 12, 217 (1986)]. It is shown that on irreducible Hermitian spaces [see S. Helgason, Differential Geometry, Lie Groups and Symmetric Spaces (Academic, Orlando, FL, 1978)] the contravariant symbols (in the sense of Berezin) of the operators f as above are given by the functions $f + \frac{1}{2}\hbar\Delta_{dR}f$. The difference with the quantization result of Berezin is discussed and a change in the geometric quantization scheme is proposed.

I. PRELIMINARIES

Let (M,ω) be a symplectic manifold describing the phase space of some physical system in classical mechanics. By *n* we always denote half the (real) dimension of *M* and by ϵ_k we denote the 2*k*-form on *M*,

$$\boldsymbol{\epsilon}_{k} = (-1)^{k(k-1)/2}/k! \, \omega^{k}$$

in particular, $\epsilon_0 = 1$ and ϵ_n is the Liouville volume element on M. On M we choose the orientation such that ϵ_n is positive and hence we can (and will) identify densities and volume forms on M.

Geometric quantization with the metalinear correction incorporated then constructs a Hilbert space \mathscr{H} out of sections of a complex line bundle QB over M and assigns (tries to assign) to an observable f (i.e., $f:M \to \mathbb{R}$) an operator f on \mathscr{H} . In Ref. 1 a general formula is derived for the local expression of the inner product on \mathscr{H} and moreover, for those observables satisfying $[X_f, P + P^{\dagger}] \subset P + P^{\dagger}$ a general (local) expression for f is given that is derived by means of an infinitesimal pairing (Ref. 2) $[X_f$ denotes the Hamiltonian vector field associated to the function $f: i_{Xf}\omega + df = 0$, P denotes the polarization and \dagger means complex conjugation (and in case of matrices it means transposition as well)].

We now briefly recall those formulas, in case the polarization P is a positive Kähler polarization. The prequantum bundle L (see, for instance, Refs. 2 or 3) is constructed by means of a set of local potentials ϑ_j on U_j for ω (i.e., $d\vartheta_j = \omega$), where $\{U_j\}$ covers the symplectic manifold M. On $U_j \cap U_k$ one supposes $\vartheta_j - \vartheta_k = du_{jk}$ for some function u_{jk} and then the transition functions of the bundle L are given by $\exp(iu_{jk}/\hbar)$. In the sequel we will identify local sections of L with functions on M, always with respect to such a trivialization, i.e., depending on the choices of local symplectic potentials ϑ_i . The quantum bundle QB is defined as the tensor product of L with a line bundle associated to the bundle of metalinear P frames. Since the bundle of metalinear P frames has (locally) a canonical trivialization in the case of a Kähler polarization (see Ref. 1), it follows that our trivialization of L defines a trivilization of QB. Let ψ_0 be a local nowhere-zero section of L (i.e., identified with a local nowhere-zero function) that is covariant constant along the polarization P (with respect to the connection ∇ on L, see below) and let $z^1, ..., z^n$ be local complex coordinates on M. These coordinates exist because (M,ω) together with a positive Kähler polarization determine a complex structure on M for which ω is a (positive) Kähler form (e.g., see Ref. 4). With ψ_0 and z' as above each global section ψ of QB which is covariant constant along P determines (uniquely) a local holomorphic function h(z) such that locally $\psi = h\psi_0$. Moreover, \mathcal{H} consists of these covariant constant sections of OB; to two such sections ψ and ψ' is associated a density \equiv measure \equiv volume form (ψ', ψ) on M and the inner product $\langle \psi', \psi \rangle$ in \mathcal{H} is calculated by integration of (ψ', ψ) over M.

Because of the one-to-one relation between sections of QB that are covariant constant in the direction of P and (local) holomorphic functions on M we will use the two notations ψ and h, with $\psi = h\psi_0$ interchangeably. With these conventions the local expression for (ψ', ψ) as given in Ref. 1 becomes

$$(\psi',\psi)(z,z^{\dagger}) = (h',h)(z,z^{\dagger}) = H_{h'h}(z,z^{\dagger})\epsilon_{n}, H_{h'h}(z,z^{\dagger}) = h'(z)^{\dagger}h(z)|\psi_{0}(z,z^{\dagger})|^{2}\cdot\sqrt{\det((2\pi i\hbar)^{-1} \times \omega(X_{z^{t}},X_{z^{k}}))_{j,k=1,\dots,n}}.$$
(1.1)

Since (ψ', ψ) and ϵ_n are globally defined densities it follows that $H_{h'h}$ is a globally defined function on M. Then \mathcal{H} con-

sists of those ψ that are covariant constant in the direction of P and for which $\int_{M} (\psi, \psi')$ is finite. If we omit the condition that h (and h') should be holomorphic (i.e., if we omit the condition that ψ should be covariant constant in the direction of P) then (1.1) defines a density for each pair of global sections of QB and one can consider the Hilbert space $L^{2}(M,P)$ of square integrable sections. Then \mathcal{H} is the closed subspace of holomorphic sections of $L^{2}(M,P)$,

$$\mathscr{H} \equiv L^{2}(M,P)_{\text{hol}} \subset L^{2}(M,P), \qquad (1.2)$$

and the orthogonal projection $\pi: L^2(M,P) \to \mathcal{H}$ admits a (local) integral representation (local in the sense that it depends upon the chosen trivialization of QB). Now let f be an observable, then for $h \in \mathcal{H}$ we can define a section $L_f h$ of QB (which need not be covariant constant in the direction of P) by

$$(L_f h)\psi_0 = -i\hbar X_f(h\psi_0) + \left(f - \vartheta(X_f) - \frac{1}{2}i\hbar \sum_{j=1}^n a_{jj}\right)h\psi_0 \qquad (1.3)$$

or equivalently with our identification of a section $h\psi_0$ with the function h,

$$L_{f}h = -i\hbar X_{f}h + \left(f - \vartheta(X_{f}) - i\hbar X_{f}\log\psi_{0}\right)$$
$$-\frac{1}{2}i\hbar\sum_{j}a_{jj}h,$$

where a_{jk} is defined by the equation

$$\begin{bmatrix} X_f, X_{Z^j} \end{bmatrix} = \sum_k (a_{jk} X_{Z^k} + b_{jk} X_{Z^{k\dagger}})$$

and where ϑ is a local potential for $\omega(d\vartheta = \omega)$, used to define the trivialization of the prequantum bundle L; the combination

$$\nabla_X(h\psi_0) = X(h\psi_0) - (i/\hbar)\vartheta(X)(h\psi_0)$$

is the local expression for the connection ∇ on L acting on the section $h\psi_0$ in the direction X. Using the general Blattner-Kostant-Sternberg (BKS) kernel one can show that the operator **f** acting on h as defined by the method of infinitesimal pairing is given by the orthogonal projection of L_ch on \mathcal{H} ; more precisely,

domain
$$\mathbf{f} = \{h \in \mathcal{H} | L_f h \in L^2(M, P)\},\$$

and

$$\mathbf{f}h = \pi L_f h \quad \Leftrightarrow \\ \forall h' \in \mathcal{H} \quad \forall h \in \text{domain } \mathbf{f}: \ \langle h', \mathbf{f}h \rangle = \int_M (h', L_f h).$$

$$(1.4)$$

II. ON KÄHLER MANIFOLDS

In this section we briefly recall some elementary facts about Kähler manifolds, and then apply them to formulas (1.1) and (1.3). Our basic reference will be the book of Weil⁵ and we will adopt the conventions used therein. A Kähler manifold M is a complex manifold with a (complex) Hermitian inner product G on each tangent space: if $z^1,...,z^n$ are local complex coordinates then

$$G(m) = g_{ik}(m) dz^{j} \otimes dz^{k\dagger} \quad \text{with} \quad g_{ik}^{\dagger} = g_{ki};$$

 (g_{ik}) should be positive definite and we denote by g the de-

terminant of (g_{ik}) ,

 $g = \det(g_{jk}).$

The imaginary part ω of G, $\omega = \text{Im } G$, is an antisymmetric two-tensor, hence a two-form. In the local coordinates z^{j} we have

$$\omega = \frac{1}{2} i g_{jk} dz^k \wedge dz^{j\dagger},$$

and the condition M is Kähler equivalent to the condition ω symplectic. On the other hand, the real part of G defines a Riemannian structure on M seen as a real manifold (of dimension 2n) and hence there exists a Laplace-de Rham operator Δ_{dR} on forms. In the local coordinates this operator is expressed on functions $f: M \to \mathbb{R}$ as

$$\Delta_{\mathrm{dR}} f = -4g^{jk} \cdot \frac{\partial^2 f}{\partial z^j \partial z^{k\dagger}} \quad \text{with} \quad g^{jk} \cdot g_{km} = \delta_m^j. \quad (2.1)$$

The invariant volume form associated to this Riemannian structure is the form ϵ_n ; if we denote $z^j = x^j + iy^j$ then

$$\epsilon_n = (\frac{1}{2}i)^n \cdot g \cdot dz^1 \wedge \cdots \wedge dz^n \wedge dz^{1\dagger} \wedge \cdots \wedge dz^{n\dagger}$$

 $= g \cdot dx^1 \wedge \cdots \wedge dx^n \wedge dy^1 \wedge \cdots \wedge dy^n = g \cdot \operatorname{Leb}^{2n}$

with Lebⁿ the Lebesgue measure on \mathbb{R}^{n} .

The complex structure on M defines a natural grading of the k-forms and the exterior derivative splits accordingly, d = d' + d'' (d' the del operator and d'' the del-bar operator). Using the Grothendieck-Dolbeault lemma on the closed two-form ω of type (1,1) one can show the local existence of a real function F on M such that

$$\omega = \frac{1}{2} i d' d'' F \Leftrightarrow g_{jk} = \frac{\partial^2 F}{\partial z^k \partial z^{j\dagger}}$$

From this it follows that the local one-form ϑ defined by

$$\vartheta = \frac{1}{4}i(d"F - d'F) = \frac{1}{4}i\left(\frac{\partial F}{\partial z^{k\dagger}}dz^{k\dagger} - \frac{\partial F}{\partial z^{k}}dz^{k}\right)$$

is a local symplectic potential. Computing $X_{r^{j}}$ we find

$$\begin{split} X_{z^{j}} &= -2ig^{jk}\frac{\partial}{\partial z^{k+}}, \quad \vartheta(X_{z^{j}}) = \frac{1}{4}iX_{z^{j}}F, \\ \omega(X_{z^{j+}},X_{z^{k}}) &= 2ig^{jk}, \end{split}$$

from which we deduce that the "function" ψ_0 defined by

$$\psi_0 = \exp(-\frac{1}{4}F/\hbar)$$

is nowhere vanishing and covariantly constant in the direction of the polarization P spanned by the $X_{z^{j}}$ [see the definition of ∇ after formula (1.3)]. This polarization is indeed a positive Kähler polarization on M seen as symplectic manifold, and conversely, if P is a positive Kähler polarization on a symplectic manifold then there exist local complex coordinates z^{j} (turning M into a complex Kähler manifold) such that $X_{z^{j}}$ span P, a fact that should explain our interest in the local Hamiltonian vector fields $X_{z^{j}}$. Finally the condition $d\omega = 0$ implies that

$$\sum_{j} \left(\frac{\partial g^{jk}}{\partial z^{j}} + g^{jk} \frac{\partial \log g}{\partial z^{j}} \right) = 0 \quad (\forall k),$$

and we find after some calculations that the expression $-\frac{1}{2i\hbar\Sigma_{a\,ij}}$ in (1.3) is given by

$$-\frac{1}{2}\,\hbar\sum a_{jj}=-\frac{1}{4}\,\hbar\Delta_{\mathrm{dR}}f-\hbar\frac{\partial\log g}{\partial z^{j}}\cdot g^{jk}\cdot\frac{\partial f}{\partial z^{k\dagger}}\,.$$

Combining all these facts and substituting them in (1.1) and (1.3) we find

$$H_{h'h}(z,z^{\dagger}) = (\pi\hbar)^{-n/2}h'(z)^{\dagger}h(z)\exp(-\frac{1}{2}F/\hbar)g^{-1/2}$$
(2.2)

and

$$L_{f} = -i\hbar X_{f}h + h \cdot \left[f - \frac{1}{4} \hbar \Delta_{dR} f - \frac{\partial (F + \hbar \log g)}{\partial z^{j}} \cdot g^{jk} \cdot \frac{\partial f}{\partial z^{k\dagger}} \right].$$

Since $(h', L_f h)$ figures in the expression for f we can integrate by parts with respect to X_f inside $(h', L_f h)$ and we find

$$(h', L_{f}h) = (h', [(1 + \frac{1}{4}\hbar\Delta_{dR})f]h) + R_{f}(h', h), \qquad (2.3)$$

$$R_{f}(h', h) = \sum_{j,k=1}^{n} \frac{\partial}{\partial z^{j}} [(\pi\hbar)^{-n/2}h'(z)^{\dagger}h(z) \\ \times \exp\left(\frac{-F}{2\hbar}\right)g^{1/2}2\hbar g^{jk}\frac{\partial f}{\partial z^{k\dagger}}] \text{ Leb}^{2n}$$

$$= (-1)^{n-1}i\hbar d'[H_{h'h}d''(f\epsilon_{n-1})]$$

$$\equiv (-1)^{n-1}i\hbar d[H_{h'h}d''(f\epsilon_{n-1})] = d\alpha. (2.4)$$

From this expression we see that $\alpha = (-1)^{n-1}$ $\times i\hbar H_{h'h}d''$ ($f\epsilon_{n-1}$) is a well-defined global (2n-1)-form on M (because $H_{h'h}$, f and ϵ_{n-1} are global), so $R_f(h',h)$ is an exact 2n-form and the following theorem immediately follows.

Theorem 2.1: If M is a compact Kähler manifold without boundary, if $f: M \to \mathbb{R}$ is any observable, and if $\psi, \psi \in \mathcal{H}$ are global sections of QB, then

 $f\psi = \pi([(1 + \frac{1}{4}\hbar\Delta_{dR})f]\psi)$ (orthogonal projection), or equivalently

$$\langle \psi', \mathbf{f}\psi \rangle = \int_{\mathcal{M}} \left(\psi', \left[\left(1 + \frac{1}{4} \, \hbar \Delta_{\mathrm{dR}} \right) f \right] \psi \right).$$

Remark 2.2: If M is not compact then there might be an extra term resulting from $\int_M d\alpha$, which depends on the behavior of α "at infinity" (or at the "boundary" of M). In Secs. IV and V we will consider the cases $M = \mathbb{C}$ and M is an irreducible Hermitian symmetric space of noncompact type and in these cases we will redefine f as if Theorem 2.1 were true for these noncompact spaces. We then will give some examples to make plausible that the boundary term $\int d\alpha$ is 0 for physically interesting observables, which should justify the redefinition of f.

Remark 2.3: There is a remarkable resemblance between $R_{f}(h',h)$ and the inner product (h',h):

$$(h',h) = H_{h',h}\epsilon_n = (-1)^{n-1}i(2n)^{-1}H_{h',h}dd''(F\epsilon_{n-1}),$$

$$R_f(h',h) = d\left[(-1)^{n-1}i\hbar H_{h',h}d''(f\epsilon_{n-1})\right].$$

III. INTERMEZZO

In this (brief) section we recall some definitions of Berezin^{6,7} in order to be able to compare our results with his. Let \mathcal{H} be a Hilbert space and M^0 a manifold with measure μ and let $E = \{e_m | m \in M^0\}$ be a set of vectors in \mathcal{H} .

Definition 3.1: The system E is called (super) complete iff

$$\forall f,g \in \mathscr{H}: \quad \langle f,g \rangle = \int_{\mathcal{M}^{\circ}} \langle f,e_m \rangle \langle e_m,g \rangle d\mu(m).$$

It follows that \mathscr{H} can be embedded in $L^2(M^0,\mu)$ by $f \in \mathscr{H} \to f(m) = \langle e_m, f \rangle.$

Definition 3.2: Suppose E is a super complete system and A is an operator such that E is contained in the domain of A. then the function A_0 defined by

$$A_0(m) = \langle e_m, \mathbf{A} e_m \rangle / \langle e_m, e_m \rangle$$

(2.2)

is called the covariant symbol of the operator A.

Definition 3.3: Let E be a super complete system and A an operator that can be calculated by

$$\langle f, Ag \rangle = \int_{M^0} \langle f, e_m \rangle A^0(m) \langle e_m, g \rangle d\mu(m)$$

for some function A^0 on M^0 , then A^0 is called the *contravar*iant symbol of the operator A.

Proposition 3.4: Suppose A is an operator on \mathcal{H} for which both the covariant and contravariant symbol exist, then we have

$$A_0(m') = \int_{\mathcal{M}_0} \frac{\langle e_{m'}, e_m \rangle \langle e_m, e_{m'} \rangle}{\langle e_{m'}, e_{m'} \rangle} A^0(m) d\mu(m).$$

IV. QUANTIZATION ON C

On $M = \mathbb{C}$ with complex coordinate z = x + iy we use the symplectic form $\omega = \frac{1}{\lambda} dz \wedge dz^{\dagger} = \lambda dx \wedge dy$. For $\lambda > 0$ the polarization $P = \mathbb{C}X_z$ is a positive Kähler polarization, so we can apply the theory of Secs. I and II. The global function $F = \lambda z z^{\dagger}$ satisfies $\omega = \frac{1}{2} i d' d'' F$ so the general (local) formulas (1.1), (1.2), (2.1), (2.2), and (2.4) become global and reduce to

$$L^{2}(\mathbb{C},P) = \left\{h: \mathbb{C} \to \mathbb{C} \left| \int_{\mathbb{C}} |h(z)| \right|^{2} \\ \times \exp\left(\frac{-\lambda z z^{\dagger}}{2\hbar}\right) d \operatorname{Leb}^{2} < \infty \right\},$$

$$\mathscr{H} = \left\{h \in L^{2}(\mathbb{C},P) \mid h \text{ is holomorphic}\right\},$$

$$\langle h',h \rangle = \pi^{-1/2} \left(\frac{\lambda}{\hbar}\right)^{1/2} \int_{\mathbb{C}} h'(z)^{\dagger} h(z) \\ \times \exp\left(\frac{-\lambda z z^{\dagger}}{2\hbar}\right) d \operatorname{Leb}^{2},$$

$$\Delta_{d\mathbb{R}} = -4\lambda^{-1} \frac{\partial^{2}}{\partial z \partial z^{\dagger}} = -\lambda^{-1} (\partial_{x}^{2} + \partial_{y}^{2}),$$

$$R_{f}(h',h) = d \left[i\pi^{-1/2}(\hbar/\lambda)^{1/2} h'(z)^{\dagger} h(z) \\ \times \exp\left(-\frac{1}{2}\lambda z z^{\dagger}/\hbar\right) d'' f \right].$$

Remark 4.1: In physics a symplectic form has the same units as an angular momentum $[kg m^2 sec^{-1}] \equiv [Joule]$ sec]; in mathematics coordinates usually do not have any units, so one expects the symplectic form to have no units. In order to separate these two aspects of a symplectic form we have introduced a parameter λ which has the same units as

Planck's constant \hbar (i.e., Joule sec) and we have introduced dimensionless coordinates z = x + iy; all units have been absorbed in the parameter λ . As is clear from the above formulas, in relevant formulas this parameter always occurs in the combination λ / \hbar , which is dimensionless (the combination $\hbar\Delta_{dR}$ is dimensionless and it is this combination that occurs in the expression for the operators), so one might be tempted to omit one of them (as does Berezin, see Ref. 6). We will not do so, because λ and \hbar have different physical interpretations: λ represents a physical quantity and \hbar represents the quantum mechanical yard stick with which to measure λ . In Secs. V and VI a parameter λ will be introduced in the same way as in this section and with the same motivations; in this section one can imagine that λ represents the mass of the (one-dimensional) system, in case $M = S^2 (= M_{1,1}^I)$, see Sec. VI) λ represents the intrinsic angular momentum (spin) (see for instance Refs. 8 and 9).

For an observable f we now define (redefine) (the domain of) f as follows:

domain
$$\mathbf{f} = \{h \in \mathcal{H} | [(1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}})f]h \in L^2(\mathbb{C},P)\}$$

 $fh = \pi([1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}})f]h)$
 $= \pi([f - \frac{1}{4}(\hbar/\lambda)(\partial_x^2 + \partial_y^2)f]h).$

This definition is different from the definition given in Sec. I, but we will make plausible by means of examples that in interesting cases if $h \in \text{domain } f$ and $h' \in \mathscr{H}$ then $\int R_f(h',h) = 0$, which implies that then the two definitions "coincide," the only possible difference being the case that in (2.3) the left-hand side $(h', L_f h)$ is finite whereas both terms on the right-hand side are infinite.

Proposition 4.2: If $\partial f / \partial z^{\dagger}$ is bounded on C then $\int_{C} R_{f}(h',h) = 0$ for all $h',h \in \mathcal{H}$.

Proof: For $h',h\in\mathcal{H}$ the function $G = \pi^{-1/2} \times (\lambda/\hbar)^{1/2}h'(z)^{\dagger}h(z)\exp(-\frac{1}{2}\lambda zz^{\dagger}\hbar)$ is absolutely integrable over \mathbb{C} (it is the integrand of $\langle h',h \rangle$) so

$$\int_{\mathcal{C}} |G| d \operatorname{Leb}^2 = \int_0^\infty dR \int_0^{2\pi} d\varphi R |G| < \infty,$$

hence there exists a sequence R_k tending to ∞ such that

$$\lim_{k\to\infty}\int_{|z|=R_k}R_k|G|d\varphi=0.$$

We now have

$$|R_{f}(h',h)| = \left|\lim_{k \to \infty} \int_{|z| < R_{k}} d\left[G,d''f\right]\right|$$
$$= \lim_{k \to \infty} \left|\int_{|z| = R_{k}} Gd''f\right|$$
$$\leq \lim_{k \to \infty} \int_{|z| = R_{k}} |G|R_{k} \left|\frac{\partial f}{\partial z^{\dagger}}\right| d\varphi$$
$$\leq \operatorname{const} \lim_{k \to \infty} \int_{0}^{2\pi} |G|R_{k} d\varphi = 0,$$

where the last inequality is a consequence of our assumption. Q.E.D.

Proposition 4.3: If $|\partial f / \partial z^{\dagger}| \leq c_1 |(1 + \frac{1}{4}\hbar\Delta_{dR})f| + c_2$ for some positive constants c_1 and c_2 then $h \in \text{domain } f$ and $h' \in \mathscr{H} \Rightarrow \int_{\mathbb{C}} R_f(h',h) = 0.$ **Proof:** If we define G as in Proposition 4.2 then the conditions show that the function $c_1 | [(1 + \frac{1}{4}\hbar\Delta_{dR})f]G| + c_2 |G|$ has a finite integral over C and then we can apply the same reasoning as in Proposition 4.2. Q.E.D.

Remark 4.4: Proposition 4.2 is clearly a special case of Proposition 4.3, although the statement is slightly stronger. The condition mentioned in Proposition 4.3 is true for the observables x,y (linear momentum and position), x^2, y^2 , and $x^2 + y^2$ (kinetic energy, quadratic potential, and the harmonic oscillator) so for these observables both definitions coincide, i.e., for these observables the procedure of geometric quantization gives the result of Theorem 2.1 for the (noncompact) manifold C.

Remark 4.5: The condition of Proposition 4.3 is not satisfied by $f = xy = z^2 - z^{\dagger 2}$. However, if we are interested in the angular momentum, then we have to consider the symplectic manifold $\mathbb{C}^n(n > 1)$ and there one can show with the same arguments that for $f = x_1 y_2 - x_2 y_1 = z_1 z_2^{\dagger} - z_2 z_1^{\dagger}$ then again the two definitions coincide.

We now turn our attention to the orthogonal projection $\pi: L^2(\mathbb{C}, P) \to \mathcal{H}$; it is well known that π admits an integral representation with a Bergman kernel $K_{\lambda}(w, z^{\dagger})$ defined by

$$K_{\lambda}(w,z^{\dagger}) = \frac{1}{2} \pi^{-1/2} \left(\frac{\lambda}{\hbar}\right)^{1/2} \exp\left(\frac{\lambda w z^{\dagger}}{2\hbar}\right)$$

and

$$(\pi g)(w) = \pi^{-1/2} \left(\frac{\lambda}{\hbar}\right)^{1/2} \int_{\mathbb{C}} K_{\lambda}(w, z^{\dagger}) g(z, z^{\dagger})$$
$$\times \exp\left(\frac{-\lambda z z^{\dagger}}{2\hbar}\right) d \operatorname{Leb}^{2}(z).$$

If we define the elements $e_{w} \in \mathcal{H}$ by $e_{w}(z) = K_{\lambda}(z, w^{\dagger})$ then this becomes

$$(\pi g)(w) = \langle e_w, g \rangle$$
 [inner product in $L^2(\mathbb{C}, P)$].

Since π is a projection we have the reproducing property formula for $g \in \mathcal{H}: g(w) = \langle e_w, g \rangle$, whence the set $\{e_w | w \in \mathbb{C}\}$ is a supercomplete system in the sense of Berezin if we denote by μ the measure defined by

$$d\mu(z) = \pi^{-1/2} (\lambda / \hbar)^{1/2} \exp(-\frac{1}{2}\lambda z z^{\dagger} / \hbar) d \operatorname{Leb}^{2}(z).$$

With these conventions the expression for f becomes

$$\langle h', \mathbf{f}h \rangle = \int h'(z)^{\dagger} \left[\left\{ 1 + \frac{1}{4} \hbar \Delta_{\mathrm{dR}} \right\} f \right] h(z) d\mu(z)$$

= $\int \langle h', e_z \rangle \left[\left(1 + \frac{1}{4} \hbar \Delta_{\mathrm{dR}} \right) f \right] \langle e_z, h \rangle d\mu(z),$

so the contravariant symbol of f is the function

contravar(f) = $(1 + \frac{1}{4}\hbar\Delta_{dR})f$.

For $M = \mathbb{C}$ the relation between covariant and contravariant symbols is known (see Ref. 6),

$$A_0 = \exp(-\frac{1}{2}\hbar\Delta_{\rm dR})A^0,$$

from which we deduce

$$\operatorname{covar}(\mathbf{f}) = (1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}})\exp(-\frac{1}{2}\hbar\Delta_{\mathrm{dR}})f. \quad (4.1)$$

G. M. Tuynman 2832

V. QUANTIZATION ON IRREDUCIBLE HERMITIAN SYMMETRIC SPACES OF NONCOMPACT TYPE

According to the general theory¹⁰ the four series of irreducible noncompact Hermitian symmetric spaces are just the classical bounded domains in \mathbb{C}^n (Ref. 11). These domains are characterized by matrices as described in Table I (matrix A > 0 means all eigenvalues positive). In the sequel we will often identify a set of coordinates $(z^1,...,z^n)$ with the corresponding matrix, so depending on the context z means either the matrix or the set of coordinates.

These four series can also be described as certain homogeneous spaces; this alternative description is listed in Table II. When seen as bounded domains, these spaces possess a (classical) Bergman kernel $K(z,w^{\dagger})$ (see Ref. 11) that is also listed in Table II (up to a multiplicative constant). We now introduce a Kähler structure on Ω (for all four series simultaneously) by

$$\omega = \frac{1}{2}id'd''F \quad \text{with } F(z,z^{\dagger}) = \lambda \log K(z,z^{\dagger})$$
for λ positive and real.

The associated (positive definite) Hermitian metric is (apart from a multiplicative constant) the usual Bergman metric which is invariant under the holomorphic diffeomorphisms of Ω . Since each Ω is homogeneous (the group G can be realized as holomorphic isometries of Ω) there exist positive constants $c(\Omega)$ such that

$$g = \det\left(\frac{\partial^2 F}{\partial z \,\partial z^{\dagger}}\right) = c(\Omega)\lambda^n K(z,z^{\dagger}).$$

Finally, the polarization P spanned by $(X_{z^{\prime}})$ is a positive Kähler polarization with respect to the symplectic form ω defined above and, moreover, F is global so the (local) formula (1.1), (1.2), (2.2), and (2.4) are global:

$$L^{2}(\Omega, P) = \left\{h: \Omega \to \mathbb{C} \left| \int_{\Omega} |h(z)|^{2} K(z, z^{\dagger})^{1-\lambda/\hbar/2} d \operatorname{Leb}^{2n} < \infty \right\}, \quad \mathscr{H} = \left\{h\in L^{2}(\Omega, P) \left| h \text{ is holomorphic} \right\}, \\ \langle h', h \rangle = c(\Omega)^{-1/2} (\pi\lambda\hbar)^{-n/2} \int_{\Omega} h'(z)^{\dagger} h(z) K(z, z^{\dagger})^{-(1+\lambda/\hbar)/2} \epsilon_{n} \\ = c(\Omega)^{1/2} \pi^{-n/2} (\lambda/\hbar)^{n/2} \int_{\Omega} h'(z)^{\dagger} h(z) K(z, z^{\dagger})^{(1-\lambda/\hbar)/2} d \operatorname{Leb}^{2n}, \\ R_{f}(h', h) = d \left[(-1)^{n-1} i\hbar c(\Omega)^{-1/2} (\pi\lambda\hbar)^{-n/2} h'(z)^{\dagger} h(z) K(z, z^{\dagger})^{-(1+\lambda/\hbar)/2} d''(f\epsilon_{n-1}) \right].$$

Remark 5.1: For $\lambda / \hbar = 1$ we see that \mathcal{H} is the ordinary Hilbert space of holomorphic, square integrable functions on the domain Ω (with respect to the Lebesgue measure). For more comments on the interpretation of the parameter λ , see Remark 4.1.

Remark 5.2: In Ref. 12 Berezin introduces a parameter h^{-1} when quantizing irreducible Hermitian symmetric spaces of noncompact type, a parameter which in our treatment is given by $\frac{1}{2}(1 + \lambda / \hbar)$ (see also Remark 4.1). After the introduction of h^{-1} he gives a definition of "allowed value" for h^{-1} that is, roughly speaking, the following condition: the Hilbert space \mathcal{H} depends upon the parameter h^{-1} (i.e., on λ) and h_0^{-1} is allowed if $\mathcal{H}(h^{-1})$ is "analytic" in a neighborhood of h_0^{-1} . He then shows that the set of allowed values consists of a continuous part and a discrete part. In our case with $\lambda > 0$ the corresponding value of h^{-1} always lies in the continuous part, i.e., in the set of allowed values.

For irreducible Hermitian symmetric spaces of compact type there is in Ref. 12 also a parameter h^{-1} and a definition of allowed value, but in the compact case the definition of allowed value is different from the noncompact case (see for more details Sec. VI after Theorem 6.5).

As on C we now define (redefine) for an observable $f: \Omega \to \mathbb{R}$ (the domain of) the operator f as follows:

domain
$$\mathbf{f} = \left\{h \in \mathscr{H} \mid h\left(1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}}\right) f \in L^2(\Omega, P) \land \forall h' \in \mathscr{H}: \int_{\Omega} R_f(h', h) = 0\right\}, \quad fh = \pi(\left[(1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}}) f\right]h),$$

and again we wish to make plausible that this definition does not differ very much from the definition given in Sec. I. We do this by considering the case $\Omega_{1,1}^{I}$, which can be interpreted as the Lobatschevsky plane $\cong \Omega_{1,1}^{I} = D^{1} = \{z \in \mathbb{C} | |z| < 1\}$.

Proposition 5.3: If $f: D^1 \to \mathbb{R}$ is such that $\partial f / \partial z^{\dagger}$ is bounded by $(1 - zz^{\dagger})^{-1}$ on D^1 then $\int_{D^1} R_f(h', h) = 0$ for all $h', h \in \mathcal{H}$. Proof: On D^1 we have $K(z, z^{\dagger}) = (1 - zz^{\dagger})^{-2}$ so assuming $h', h \in \mathcal{H}$ we find

TABLE I. Irreducible	Hermitian	symmetric	spaces of	noncompact 1	ype.
----------------------	-----------	-----------	-----------	--------------	------

Name	Description	Coordinates	Condition	$n \equiv C$ dimension
$\Omega^{\mathrm{I}}_{p,q}$ Ω^{II}_{p} $\Omega^{\mathrm{III}}_{p}$ $\Omega^{\mathrm{III}}_{p}$	matrices z of order $p \times q$ symmetric matrices z of order $p \times p$ antisymmetric matrices z of order $p \times p$ row vectors z in \mathbb{C}^p	all pq entries of z the upper triangle including the diagonal the upper triangle without the diagonal all p entries	$I - zz^{\dagger} > 0$ $\left\{1 + zz^{T} ^{2} - 2zz^{\dagger} > 0$ $\left zz^{T} < 1 (^{T} = \text{transpose})\right $	$p \cdot q$ $\frac{1}{2}p(p+1)$ $\frac{1}{2}p(p-1)$ p

$$\int_{D^{-1}} |h'(z)^{\dagger}h(z)K(z,z^{\dagger})^{-(1+\lambda/\hbar)/2}K(z,z^{\dagger})|d\operatorname{Leb}^{2} < \infty$$

$$\Leftrightarrow \int_{0}^{1} (1-R^{2})^{-1} dR \int_{0}^{2\pi} |h'(z)^{\dagger}h(z)|(1-R^{2})^{\lambda/\hbar}R d\varphi < \infty$$

$$\Rightarrow \exists R_{k} \uparrow 1: \lim_{k \to \infty} \int_{0}^{2\pi} |h'(z)^{\dagger}h(z)|(1-R_{k}^{2})^{\lambda/\hbar}R_{k} d\varphi = 0$$

$$\Rightarrow \left| \int_{D^{-1}} R_{f}(h',h) \right| \leq \lim_{k \to \infty} \int_{|z|=R_{k}} |h'^{\dagger}h| \operatorname{const}(1-R_{k}^{2})^{1+\lambda/\hbar} \left| \frac{\partial f}{\partial z^{\dagger}} dz^{\dagger} \right|$$

$$\leq \lim_{k \to \infty} \operatorname{const} \int_{0}^{2\pi} |h'(z)^{\dagger}h(z)|(1-R_{k}^{2})^{\lambda/\hbar}R_{k} d\varphi = 0.$$
Q.E.D.

Proposition 5.4: Let $f: D^1 \to \mathbb{R}$ and suppose there exist positive constants c_1 and c_2 such that $(1 - zz^{\dagger})|\partial f \partial z^{\dagger}| \leq c_1|(1 + \frac{1}{4}\hbar\Delta_{dR})f| + c_2$ on D^1 then $h \in \text{domain } \mathbf{f}$ and $h' \in \mathcal{H} \Longrightarrow \int R_f(h',h) = 0.$

Proof: If one pastes together the arguments of the proofs of Proposition 4.3 and 5.3 then one obtains easily a proof for this proposition. Q.E.D.

Example 5.5: The group SU(1,1) acts on D^{1} as the group of holomorphic symplectic isometries by

$$g = \begin{pmatrix} a & b \\ b^{\dagger} & a^{\dagger} \end{pmatrix}; \quad z \to \frac{az+b}{b^{\dagger}z+a^{\dagger}},$$

$$g \in SU(1,1) \Leftrightarrow |a|^2 - |b|^2 = 1.$$

We can exhibit three one-parameter subgroups $g_j(t)$ of SU(1,1),

$$g_1(t) = \begin{pmatrix} \cosh t & \sinh t \\ \sinh t & \cosh t \end{pmatrix},$$

$$g_2(t) = \begin{pmatrix} \cosh t & i \sinh t \\ -i \sinh t & \cosh t \end{pmatrix}, \quad g_3(t) = \begin{pmatrix} e^{it} & 0 \\ 0 & e^{-it} \end{pmatrix},$$

which become after differentiation the generators of the Lie algebra su(1,1):

$$g'_{1}(0) = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad g'_{2}(0) = \begin{pmatrix} 0 & i \\ -i & 0 \end{pmatrix}$$
$$g'_{3}(0) = \begin{pmatrix} i & 0 \\ 0 & -i \end{pmatrix}.$$

The momentum map^{8,13,14} associated to this action maps these three generators of the Lie algebra to the functions:

$$f_1 = -i\lambda \frac{z^{\dagger} - z}{1 - zz^{\dagger}}, \quad f_2 = \lambda \frac{z^{\dagger} + z}{1 - zz^{\dagger}}, \quad f_3 = \lambda \frac{1 + zz^{\dagger}}{1 - zz^{\dagger}}.$$

TABLE II. Bergman kernels for the bounded domains.

Name	$K(z,w^{\dagger})$	Alternative description as G/K (see Ref. 10)
$\Omega^{I}_{p,q}$	$\det(1_p - zw^{\dagger})^{-p-q}$	$SU(p,q)/S(U(p) \times U(q))$
$\hat{\Omega}_{p}^{\hat{\Pi}}$	$\det(1_p - zw^{\dagger})^{-p-1}$	$Sp(p,\mathbf{R})/U(p)$
Ω_{P}^{in}	$\det(1_p - zw^{\dagger})^{-p+1}$	$SO^{*}(2p)/U(p)$
$\Omega_p^{\rm iv}$	$(1-2zw^{\dagger}+zz^{T}(ww^{T})^{\dagger})^{-p}$	$SO_0(p,2)/(SO(p)\times SO(2))$

2834 J. Math. Phys., Vol. 28, No. 12, December 1987

One can show that these functions satisfy the condition of Proposition 5.4, so in computing f_j we can forget about the term $R_f(h',h)$. Since SU(1,1) acts holomorphically, the associated Hamiltonian vector fields X_{f_j} leave the polarization P invariant, which shows that $L_f h$ is already in \mathcal{H} and we find

$$\begin{split} (L_{f_1}h)(z) &= (\mathbf{f}_1)h(z) \\ &= i\hbar \Big((1-z^2)\frac{dh}{dz} - \Big(1+\frac{\lambda}{\hbar}\Big)zh(z) \Big) \\ &= \pi (h \ (1+\frac{1}{4}\hbar\Delta_{\mathrm{dR}})f) \,, \end{split}$$

$$\begin{split} (L_{f_2}h)(z) &= (\mathbf{f}_2h)(z) \\ &= -\hbar \Big((1+z^2) \frac{dh}{dz} + \Big(1+\frac{\lambda}{\hbar}\Big) zh(z) \Big) \\ &= \pi (h \ (1+\frac{1}{4}\hbar \Delta_{\mathrm{dR}})f) \ , \\ (L_{f_3}h)(z) &= (\mathbf{f}_3h)(z) \\ &= -\hbar \Big(2z \frac{dh}{dz} + \Big(1+\frac{\lambda}{\hbar}\Big) zh(z) \Big) \\ &= \pi (h \ (1+\frac{1}{4}\hbar \Delta_{\mathrm{dR}})f) \ . \end{split}$$

On the other hand, one verifies easily that the action $\rho(g)$ of SU(1,1) on \mathcal{H} given by

$$(\rho(g)h)(z) = (a - b^{\dagger}z)^{-(1 + \lambda/\hbar)}h(g^{-1}z),$$

geSU(1,1) as above,

is a projective representation of SU(1,1) on \mathscr{H} . For fixed $g \in SU(1,1)$ and |z| < 1 we have $|a - b^{\dagger}z| > 0$, so $(a - b^{\dagger}z)^{-(1+\lambda/\pi)}$ is holomorphic on D^1 ; however, such a branch cannot be chosen consistently for all $g \in SU(1,1)$ simultaneously (see also Ref. 15). For the one-parameter groups $\rho(g_j(t))$ a consistent choice can be made, resulting in a unitary representation of \mathbb{R} on \mathscr{H} , showing that their generators f_j are (essentially) self-adjoint operators:

$$\rho(g_i(t)) = \exp(-it\mathbf{f}_i/\hbar).$$

Back to the general case, the (generalized) Bergman kernel $K_{\lambda}(z,w^{\dagger})$ associated with the orthogonal projection $\pi: L^{2}(\Omega, P) \to \mathcal{H}$ is calculated by Berezin,¹²

$$K_{\lambda}(z,w^{\dagger}) = \tilde{n}_{\lambda}(\Omega)K(z,w^{\dagger})^{(1+\lambda/\hbar)/2},$$

i.e.,

$$(\pi g)(w) = c(\Omega)^{-1/2} (\pi \lambda \hbar)^{-n/2}$$

$$\times \int_{\Omega} K_{\lambda}(w, z^{\dagger}) g(z, z^{\dagger}) K(z, z^{\dagger})^{-(1+\lambda/\hbar)/2} \epsilon_{n}$$

$$= c(\Omega)^{-1/2} (\pi \lambda \hbar)^{-n/2} \tilde{n}_{\lambda}(\Omega)$$

$$\times \int_{\Omega} g(z, z^{\dagger}) \left[\frac{K(w, z^{\dagger})}{K(z, z^{\dagger})} \right]^{-(1+\lambda/\hbar)/2} \epsilon_{n},$$

where the branch of the mapping $\zeta \to \zeta^{(1+\lambda/\hbar)/2}$ is chosen such that $K_{\lambda}(z,w^{\dagger})$ is real for z = w, and the constant $\tilde{n}_{\lambda}(\Omega)$ is determined by $\tilde{n}_{\lambda}(\Omega) = ||1||^{-2}$, where ||1|| denotes the norm in \mathcal{H} of the function which is constant 1 on Ω (it should be noted that indeed $1 \in \mathcal{H}$ for each $\lambda > 0$). If we now define for each $w \in \Omega$ the function $e_w \in \mathcal{H}$ by

$$e_w(z) = K_\lambda(z, w^{\dagger}),$$

then because of the reproducing property of the orthogonal projection we know that the set $\{e_w | w \in \Omega\}$ is a supercomplete system of elements of \mathcal{H} and, moreover, by definition of domain f, we can show with the same reasoning as in Sec. IV,

$$contravar(\mathbf{f}) = (1 + \frac{1}{4}\hbar\Delta_{dR})f.$$

The relation between covariant and contravariant symbols is in these cases more complicated then for $M = \mathbb{C}$; explicit expressions for the connecting operator in terms of powers of Δ_{dR} is given by Moreno.^{16–18}

VI. QUANTIZATION ON IRREDUCIBLE HERMITIAN SYMMETRIC SPACES OF COMPACT TYPE

In this section the main problem is not to "show" that $R_f(h',h)$ does not contribute to f, because that is guaranteed by compactness, but to exhibit a supercomplete system of vectors in \mathcal{H} . In particular, we will show that the allowed

values of the parameter h^{-1} of Berezin in Ref. 12 (see Remark 5.2) correspond exactly to the (pre)quantizable values of our parameter λ (quantizable in the sense of Kostant and Souriau, a condition on the cohomology class of the symplectic form).

The four series of irreducible Hermitian symmetric spaces of compact type (which are dual to the noncompact ones) are listed in Table III. As in the noncompact case z denotes either the local coordinates $(z^1,...,z^n)$ or the corresponding matrix. To obtain a full set of local charts we need a (partial) action of G on the local coordinates; therefore we represent each $g \in G$ as block matrices g_{ij} (i, j = 1, 2) of appropriate sizes as follows:

$$\begin{pmatrix} g_{11} & g_{12} \\ g_{21} & g_{22} \end{pmatrix} = A(M)gA(M)^{-1},$$

with

$$A(M_{p,q}^{I}) = \mathbf{1}_{p+q}, \quad A(M_{p}^{II}) = \mathbf{1}_{2p},$$
$$A(M_{p}^{II}) = \begin{pmatrix} \mathbf{1}_{p} & i\mathbf{1}_{p} \\ i\mathbf{1}_{p} & \mathbf{1}_{p} \end{pmatrix}, \quad A(M_{p}^{IV}) = \begin{pmatrix} \mathbf{1}_{p} & 0 \\ 0 & i\mathbf{1}_{2} \end{pmatrix}$$

(note that as a result each matrix g_{ij} is a special unitary matrix). Using these block matrices one defines the action of G on the local coordinates by

 $gz = (g_{11}z + g_{12})(g_{21}z + g_{22})^{-1}$, for types I, II, and III and

$$gz = \frac{g_{11}z + g_{12}v_z}{(1,i) [g_{21}z + g_{22}v_z]} \quad \text{for type IV,}$$

where v_z is given by $v_z = (\frac{1}{2}(z^Tz + 1), \frac{1}{2}i(z^Tz - 1))^T$. Since the stabilizer of $0 \in \mathbb{C}^n$ is just the subgroup K in the definition of the homogeneous space G/K (Table III) we can use this action of G on \mathbb{C}^n to define a set of local charts $\{U_g | g \in G\}$ for G/K; the chart $\varphi_g : U_g \to \mathbb{C}^n$ is defined by

$$U_{g} = \{ [h] \in G/K | \det((gh)_{22}) \neq 0 \}$$

$$\varphi_{g}: U_{g} \to \mathbb{C}^{n}, [h] \to gh(0) = (gh)_{12}(gh)_{22}^{-1} \}$$
for type I, II, and III,

$$U_{g} = \{ [h] \in G/K | (1,i)(gh)_{22}(1,-i)^{T} \neq 0 \}$$

$$\varphi_{g}: U_{g} \to \mathbb{C}^{n}, [h] \to gh(0) = (gh)_{12}(1,-i)^{T}/(1,i)(gh)_{22}(1,-i)^{T} \}$$
for type IV,

a chart which covers G/K except for a submanifold of lower dimension. The local diffeomorphisms between two charts U_g and U_g is given by $z \rightarrow kz$ for some $k \in G$, more precisely,

$$\varphi_{g'} \circ (\varphi_g)^{-1} : z \to (g'g^{-1})(z).$$
 (6.1)

On each M = G/K there exists a Kähler structure ω (invariant with respect to the action of G) which is given in

local coordinates z by $\omega = \frac{1}{2}id'd''F(z,z^{\dagger})$, where F is defined by

$$F(z,z^{\dagger}) = -\lambda \log K(z,z^{\dagger}), \quad \lambda > 0,$$

$$K(z,w^{\dagger}) = N(z,w^{\dagger})^{-\nu(M)}.$$
(6.2)

The defining functions $N(z,w^{\dagger})$ and v(M) are given by

FABLE III. Irreducible Hermitian	n symmetric spaces o	f compact type.
---	----------------------	-----------------

Name	Description as G/K	Local coordinates	$n \equiv C$ dimension	
$M_{p,q}^{I}$ M_{p}^{II} M_{p}^{III} M_{p}^{IIV} M_{p}^{IV}	$SU(p+q)/S(U(p) \times U(q))$ Sp(p)/U(p) SO(2p)/U(p) $SO(p+2)/(SO(p) \times SO(2))$	matrices z of order $p \times q$ symmetric matrices z of order $p \times p$ antisymmetric matrices z of order $p \times p$ vectors z of length p	$p \cdot q$ $\frac{1}{2}p(p+1)$ $\frac{1}{2}p(p-1)$ p	<u> </u>

$$N(z,w^{\dagger}) = \begin{cases} \det(1+zw^{\dagger}), & \text{for types I, II, and III,} \\ 1+2z^{T}z+(z^{T}z)(z^{T}z)^{\dagger}, & \text{for type IV,} \end{cases}$$
(6.3)

$$\nu(M_{p,q}^{I}) = p + q, \quad \nu(M_{p}^{II}) = p - 1, \\
\nu(M_{p}^{II}) = p + 1, \quad \nu(M_{p}^{IV}) = p.$$
(6.4)

Since $\lambda > 0$ the associated Hermitian metric is positive definite and the polarization P spanned by $(X_{z^{i}})$ is a positive Kähler polarization, so again we can apply the general formalism of Sec. II. Since M is homogeneous there exist constants c(M) such that

$$\det\left(\frac{\partial^2 F}{\partial z \,\partial z^{\dagger}}\right) = c(M)\lambda^n K(z, z^{\dagger}) \tag{6.5}$$

(here we omit the letter g for this determinant to avoid confusion with the elements of the group G); furthermore, since each local chart U_g covers M except for a set of measure 0 (with respect to $\epsilon_n =$ the Liouville measure) it follows that instead of integrating over M we may restrict the integration to the local chart $U_{id} \cong \mathbb{C}^n$. Hence formulas (1.1), (1.2), and (2.2) become

$$L^{2}(M,P) = \left\{ h: \mathbb{C}^{n} \to \mathbb{C} | h \text{ represents a global section of QB} \right.$$

and $\int_{\mathbb{C}^{n}} |h(z)|^{2} K(z, z^{\dagger})^{(\lambda/\hbar + 1)/2}$
 $\times d \operatorname{Leb}^{2n} < \infty \right\},$

 $\mathscr{H} = \{h \in L^2(M, \mathcal{P}) | h \text{ holomorphic} \},\$ $\langle h', h \rangle = c(M)^{-1/2} (\pi \lambda \tilde{\pi})^{-n/2}$

$$\times \int_{\mathbb{C}^{n}} h'(z)^{\dagger} h(z) K(z, z^{\dagger})^{(\lambda/\bar{n}-1)/2} \epsilon_{n}$$

$$= c(M)^{1/2} \pi^{-n/2} \left(\frac{\lambda}{\bar{n}}\right)^{n/2}$$

$$\times \int_{\mathbb{C}^{n}} h'(z)^{\dagger} h(z) K(z, z^{\dagger})^{\lambda/\bar{n}+1)/2} d \operatorname{Leb}^{2n},$$

$$(6.6)$$

and for any observable $f: M \to \mathbb{R}$ we have (Theorem 2.1)

 $\langle h', \mathbf{f}h \rangle = \langle h', [(1 + \frac{1}{4}\hbar\Delta_{\mathrm{dR}})f]h \rangle [\operatorname{in} L^2(M, P)!].$

The only remaining problem now is the determination of the structure of the (nontrivial) bundle QB. According to the general theory the bundle QB is the tensor product of the prequantum bundle L and a line bundle ML associated to the bundle of metalinear P frames. It is not true that both bundles do always exist: ML exists iff the first Chern class $c_1(\omega)^{\mathbb{R}}$ of the symplectic manifold is even, and L exists iff the symplectic form ω/\hbar determines an integer cohomology class, and QB exists iff $[\omega/\hbar]$ and $\frac{1}{2}c_1(\omega)^{\mathbb{R}} \in \mathbb{Z}$. However, if we are only interested in the bundle QB and not in the two constituents separately, then one can reduce the condition on existence to

$$[\omega/\hbar] + \frac{1}{2}c_1(\omega)^{\mathbf{R}} \in \mathbb{Z}, \tag{6.7}$$

a result which can be obtained by using Mp^c structures.¹⁹ What we will do is to construct explicitly the transition functions of the bundle QB and then finding those values of λ for which this system of transition functions is indeed well defined, which in turn are those values of λ for which (6.7) is satisfied. Since $\epsilon_n = \text{const } K(z,z^{\dagger})d \text{ Leb}^{2n}$ (6.5) is invariant under the action of G it follows that

$$K(gz,(gz)^{\dagger}) = \left|\frac{\partial gz}{\partial z}\right|^{-2} K(z,z^{\dagger}), \qquad (6.8)$$
$$\frac{\partial gz}{\partial z} = \det(\operatorname{Jacobian}(z \to gz)).$$

Now suppose that the holomorphic functions $h_i^{(\prime)}$ on U_i (i = 1,2) are local representations of global sections $\psi^{(\prime)}$ of QB, then from Sec. I we know that the function $H_{h'h}$ (1.1) is a global function on M, hence $h_1^{(\prime)}(z) = \operatorname{tr}_{12}(z) \cdot h_2^{(\prime)}(gz)$ [where g is the diffeomorphism (6.1) between the two charts $U_1 \rightarrow U_2$ and where $\operatorname{tr}_{12}(z)$ is the transition function between the two trivializations above U_1 and U_2] implies

$$\begin{split} h_{1}'(z)^{\dagger}h_{1}(z)K(z,z^{\dagger})^{(\lambda/\hbar-1)/2} \\ &= |\mathrm{tr}_{12}(z)|^{2}h_{2}'(gz)^{\dagger}h_{2}(gz)K(z,z^{\dagger})^{(\lambda/\hbar-1)/2} \\ &= h_{2}'(gz)^{\dagger}h_{2}(gz)K(gz,(gz)^{\dagger})^{(\lambda/\hbar-1)/2} \\ &= h_{2}'(gz)^{\dagger}h_{2}(gz) \left|\frac{\partial gz}{\partial z}\right|^{(1-\lambda/\hbar)}K(z,z^{\dagger})^{(\lambda/\hbar-1)/2} \\ &\Rightarrow |\mathrm{tr}_{12}(z)|^{2} = \left|\frac{\partial gz}{\partial z}\right|^{(1-\lambda/\hbar)}. \end{split}$$

Because of the identification of local sections ψ which are covariant constant along P with holomorphic functions h, the transition functions are holomorphic (QB is in cases under investigation a holomorphic bundle). Since $\partial gz/\partial z$ is holomorphic, too, we deduce that

$$\operatorname{tr}_{12}(z) = \left(\frac{\partial gz}{\partial z}\right)^{(1-\lambda/\hbar)/2}.$$
(6.9)

Remark 6.1: If one computes the transition functions of the bundles L and ML separately, one finds that for ML it is $(\partial gz/\partial z)^{1/2}$ and for $L(\partial gz/\partial z)^{-\lambda/\hbar/2}$, which also shows that for QB = $L \otimes ML$ the transition function is $(\partial gz/\partial z)^{1/2}$ $\times (\partial gz/\partial z)^{-\lambda/\hbar/2} = (\partial gz/\partial z)^{(1-\lambda/\hbar)/2}$.

Remark 6.2: In reality one has a degree of freedom, one can modify (6.9) by a phase factor $\exp(i\varphi)$ (which vanishes when taking the absolute values). This freedom determines a Cech 1-cocycle with values in U(1) (the phase factor depends on the intersection $U_1 \cap U_2$) and hence an element of $H^1(M,U(1))$, which classifies the space of the inequivalent (pre)quantizations. However, each irreducible Hermitian symmetric space of compact type is simply connected, hence $H^1(M,U(1)) = \{0\}$, so modulo a natural equivalence of bundles L and QB our choice in (6.9) is unique.

We now want to "compute" the values of λ for which the transition functions (6.9) are well defined (perhaps by using subcharts on which the function $\zeta \rightarrow \zeta^{(1-\lambda/\hbar)/2}$ is defined) and for which they satisfy (at the same time) the cocycle condition $tr_{12}tr_{23} = tr_{13}$.

Proposition 6.3: If $w = g^{-1}(0) \in \mathbb{C}^n$ exists then

$$\frac{\partial gz}{\partial z} = \operatorname{const}(g)K(z,w^{\dagger}) = \operatorname{const} N(z,w^{\dagger})^{-\nu(M)}.$$

The proof of this proposition can be found in Ref. 12, Theorem 2.1. It can be verified easily for the spaces of type I, II, and III by using relation (6.8) to calculate $\partial gz/\partial z$ explicitly $(\Rightarrow \partial gz/\partial z = [\det(g_{21}z + g_{22})]^{-\nu(M)}$, a formula that is valid for all $g \in G$).

Corollary 6.4: $\operatorname{tr}_{12}(z) = \operatorname{const} N(z, w^{\dagger})^{(\lambda/\hbar - 1) \cdot v(M)/2}$.

In Ref. 12 it is shown that for the manifolds of type I, II, and IV $N(z,w^{\dagger})$ is an irreducible polynomial in the unknowns (z,w^{\dagger}) and for type III $N(z,w^{\dagger})$ is the square of an irreducible polynomial. This fact is the main ingredient of the proof of the following theorem, a proof which will be delegated to the Appendix.

Theorem 6.5: (a) The values of λ for which the transition functions (6.9) define a bundle are given below $[\nu(M)$ is defined by (6.4)]:

$$\frac{1}{2}(\lambda/\hbar - 1)\nu(M) \in \mathbb{Z}, \text{ for types I, II, and IV,} \\ (\lambda/\hbar - 1)\nu(M) \in \mathbb{Z}, \text{ for type III,} (6.10)$$

or equivalently,

$$\frac{\text{name}}{M_{p,q}^{\text{I}}} \quad \frac{\text{allowed } \lambda \text{'s } (k \in \mathbb{Z})}{p+q} \quad \frac{\text{name}}{p+q} \quad \frac{\text{allowed } \lambda \text{'s } (k \in \mathbb{Z})}{p-1}$$

$$M_{p}^{\text{II}} \quad \frac{p-1+k}{p-1} \, \tilde{n} \quad M_{p}^{\text{III}} \quad \frac{p-1+k}{p-1} \, \tilde{n}$$

(b) For $0 < \lambda < \hbar$ and λ allowed (i.e., $\lambda > 0$ and k < 0) dim $\mathcal{H} = 0$, for $\lambda = \hbar$ (i.e, k = 0) dim $\mathcal{H} = 1$ and for k > 0: $0 < \dim \mathcal{H} < \infty$.

When we compare our allowed values with the allowed values of Berezin, we see that we can get a complete agreement if we identify our "parameter" $\frac{1}{2}(\lambda/\hbar - 1)$ [which appears in the transition function (6.9) and in the description of the inner product (6.6)] with the parameter h^{-1} of Berezin,¹² which appears in exactly the same way in the inner product. However, Berezin's definition of allowed values does not involve any condition on the bundle existence, it involves a certain condition on the Bergman kernel on the local chart \mathbb{C}^n (see below). Since the allowed values agree, we may deduce that on the local chart U_{id} the (generalized) Bergman kernel $K_{\lambda}(z,w^{\dagger})$ that is associated to the orthogonal projection $\pi: L^2(M,P) \to \mathcal{H}$ is given by

 $K_{\lambda}(z,w^{\dagger}) = \tilde{n}_{\lambda}(M)K(z,w^{\dagger})^{(1-\lambda/\hbar)/2}$

[in fact it is the validity of this relation between the function $K(z,w^{\dagger})$ (6.2), (6.3), and the Bergman kernel K_{λ} that is the condition of Berezin], i.e., for $k \in L^{2}(M,P)$,

$$(\pi k)(w) = c(M)^{-1/2} (\pi \lambda \hbar)^{-n/2} \int_{\mathbb{C}^n} K_\lambda(w, z^{\dagger}) k(z, z^{\dagger})$$
$$\times K(z, z^{\dagger})^{(\lambda/\hbar - 1)/2} \epsilon_n$$
$$= c(M)^{-1/2} (\pi \lambda \hbar)^{-n/2} \tilde{n}_\lambda(M) \int_{\mathbb{C}^n} k(z, z^{\dagger})$$
$$\times \left[\frac{K(z, z^{\dagger})}{K(w, z^{\dagger})} \right]^{(\lambda/\hbar - 1)/2} \epsilon_n.$$

The constant $\tilde{n}_{\lambda}(M)$ is determined by $\tilde{n}_{\lambda}(M) = ||1||^{-2}$, where ||1|| is the norm of the global section ψ which is represented by the constant (holomorphic) function 1 on the local chart U_{id} ; the branch of the function $\zeta \rightarrow \zeta^{(1-\lambda/\hbar)/2}$ is determined in the same way as for the tansition functions (6.9), $K_{\lambda}(z,w^{\dagger})$ is a polynomial in (z,w^{\dagger}) that is constant 1 for z = 0. As in Sec. V it follows that the set $\{e_w(z) = K_\lambda(z,w^{\dagger}) | w \in \mathbb{C}^n\}$ is a supercomplete system [with M° of Definition (3.1) given by the (local) chart $\mathbb{C}^n \cong U_{id}$] and that for any observable $f: M \to \mathbb{R}$,

contravar(f) = $(1 + \frac{1}{4}\hbar\Delta_{dR})f$.

As in the noncompact case we refer to Refs. 18 and 20 for the relation between covariant and contravariant symbols.

VII. SUMMARY AND DISCUSSION

In Sec. II we showed that on Kähler manifolds the geometric quantization scheme gives the following results: modulo a boundary term $\int d\alpha$ the operator f associated to a classical observable f is given by the process (1) multiply by $(1 + \frac{1}{\hbar}\Delta_{dR})f$ and then (2) take the holomorphic part (\mathcal{H} is the subspace of holomorphic sections of QB with regard to the space of all "square integrable" sections of QB). For compact Kähler manifolds the boundary term $\int d\alpha$ is 0 and the above described result is exact. It was made plausible that for the physically interesting observables the result is also true on irreducible Hermitian symmetric spaces of noncompact type, from which one can deduce that it is true for all simply connected Hermitian symmetric spaces (see Ref. 10 Chap. VIII, §6). On the other hand, the above description of f can NOT be true in general: for $M = \{|z| < 1\} \subset \mathbb{C}$, $\omega = \frac{1}{2} dz \wedge dz^{\dagger}$ and $f = z + z^{\dagger}$ (a bounded observable) the contribution $\int d\alpha$ is NOT zero for all $h, h' \in \mathcal{H}$.

In Secs. IV-VI we showed (forgetting for the moment about the boundary term) that on the *quantizable* irreducible Hermitian symmetric spaces (which gave a condition in the compact case) a supercomplete system (in the sense of Berezin) exists and that the contravariant symbol of any operator **f** associated to a classical observable *f* is given by contrav(**f**) = $(1 + \frac{1}{4}\hbar\Delta_{dR})f$. Moreover, the quantization condition in the compact case coincides exactly with Berezin's definition of allowed values.

In order to relate our results to results obtained by *-product quantization we need a few comments on the close relationship between Berezin's quantization method and quantization by *-products. Both methods have as their basic assumption that quantum mechanics is a deformation of classical mechanics; the algebra of operators (especially observables) on the quantum mechanical Hilbert space depends upon a parameter \hbar and in the limit $\hbar \downarrow 0$ this algebra reduces to the Poisson algebra of functions on the classical phase space (in quantum mechanics products of operators and commutators; in classical mechanics pointwise products of functions and Poisson brackets). More abstractly they study deformations (depending on a parameter) of the Poisson algebra. On the one hand Berezin realizes such a deformation as the set of covariant symbols of operators on a Hilbert space, deducing the product structure on these symbols from the product structure of the corresponding operators. On the other hand, *-products study such deformations abstractly. To compute physical relevant data (i.e., spectra of observables) they use a beautiful theorem²¹ stating that to compute the spectrum of an "operator" [i.e., an element of the (deformed) algebra] one only needs the algebra structure, i.e., the *-product structure.

In Ref. 12 Berezin has computed the quantization on irreducible Hermitian symmetric spaces; for these cases the connection between covariant symbols and contravariant symbols and the associated *-products are calculated by Moreno in Refs. 16–18, 20. This gives us a way to compare the geometric quantization scheme with the quantization by *-products (deformations of the Poisson algebra), although the comparison works only in the case of Kähler manifolds (more precisely, for Hermitian symmetric spaces).

The results which have been discussed up till now are obtained by (conventional) geometric quantization including the metalinear correction. I now propose to "change" this quantization scheme in the case of positive Kähler polarizations: "hypothesis": the contravariant symbol of the operator f associated to the classical observable f is given by

$$f_{\rm contravar} = \exp(\frac{1}{4}\hbar\Delta_{\rm dR})f,$$

instead of $(1 + \frac{1}{4}\hbar\Delta_{dR})f$. At first this might seem a fancy change, but let us discuss the merits of this hypothesis. One of the first merits is that when we compare this quantization scheme in \mathbb{C}^n [resulting in operators on the Bargmann representation) with the usual geometric quantization scheme on \mathbb{C}^n with the vertical polarization [resulting in operators on the Schrödinger representation $\mathcal{H}_s = L^2(\mathbb{R}^n)$] then the results are the same for operators of the form $p^2 + pa(q) + V(q)$ (i.e., at most quadratic in the momentum variables). If $f_{\text{contravar}}$ had been given by $(1 + \frac{1}{4}\hbar\Delta_{dR})f$ then the results would have been different (see Ref. 1 for explicit calculations).

A second merit of our hypothesis can be found in the comparison with the Weyl-Wigner quantization scheme on \mathbb{C}^n . In Ref. 6 it is shown that the Weyl symbol f_w of an operator f is related to the contravariant symbol $f_{\text{contravar}}$ of the same operator by

$$f_{\rm W} = \exp(-\frac{1}{4}\hbar\Delta_{\rm dR})f_{\rm contravar}$$

With our hypothesis we find that $f_w = f$, i.e., the Weyl symbol is the classical observable we started with. It follows from the above observation that our hypothesis gives the same quantization prescription as the Weyl-Wigner quantization scheme, from which one deduces that our hypothesis gives on \mathbb{C}^n the same quantization results as Berezin's and as the *-product approach (since these also coincide with the Weyl-Wigner procedure).

Apart from the above-mentioned merits of our hypothesis (i.e., giving the same results as various other quantization schemes) there is another merit: in Ref. 9 a problem was encountered in the (geometric) quantization of the classical hydrogen atom including spin: the hyperfine interaction term contains an incorrect factor 2 (for 1-densities the incorrect factor is §) that cannot be absorbed by rescaling. Assuming our hypothesis, this factor disappears (after reinterpreting the classical intrinsic angular momentum). More generally the situation is as follows: suppose we have two (isolated) physical systems without interaction, described by the symplectic (Kähler) manifolds (M_1, ω_1) and (M_2,ω_2) . Then $(M_1 \times M_2, \omega_1 + \omega_2)$ describes the two systems simultaneously and one can quantize it. Now suppose we consider observables f_1 on M_1 and f_2 on M_2 , then $f(m_1,m_2) = f_1(m_1)f_2(m_2)$ is an observable on $M_1 \times M_2$. Quantization of (M_i, ω_i) yields a Hilbert space \mathcal{H}_i (of holomorphic sections of QB_i) and operators \mathbf{f}_i on \mathcal{H}_i ; quantization of $M_1 \times M_2$ yields a Hilbert space \mathcal{H} (which is in a certain sense the tensor product $\mathcal{H}_1 \otimes \mathcal{H}_2$) and one gets an operator f associated to $f = f_1 f_2$. When we now assume that after quantization the two systems should remain isolated (a reasonable assumption in physics), then one expects that

$$\begin{bmatrix} \mathbf{f}_1, \mathbf{f}_2 \end{bmatrix} = 0 \\ \mathbf{f} = \mathbf{f}_1 \mathbf{f}_2.$$
 (noninterference axiom).

Although the relation $[\mathbf{f}_1, \mathbf{f}_2] = 0$ is satisfied, the second relation $\mathbf{f} = \mathbf{f}_1 \mathbf{f}_2$ is not satisfied. Investigating their contravariant symbols we find

contravar(f)

$$= f_1 f_2 + \frac{1}{4} \hbar (\Delta_1 + \Delta_2) f_1 f_2$$

= $f_1 f_2 + \frac{1}{4} \hbar f_1 \Delta_2 f_2 + \frac{1}{4} \hbar f_2 \Delta_1 f_1,$

 $contravar(\mathbf{f}_1\mathbf{f}_2)$

$$= (f_1 + \frac{1}{4}\hbar\Delta_1 f_1)(f_2 + \frac{1}{4}\hbar\Delta_2 f_2)$$

= contravar(f) + 16⁻¹\hbar^2(\Delta_1 f_1)(\Delta_2 f_2),

so geometric quantization does not comply with our noninterference axiom [it is precisely the absence of the term $16^{-1} \#^2(\Delta_1 f_1)(\Delta_2 f_2)$ that explains the incorrect factor $\frac{2}{5}$ in the quantization of the hydrogen atom in Ref. 9]. On the other hand, our hypothesis, which changes the quantization results, does comply with the noninterference axiom:

$$\exp(\frac{1}{\hbar}(\Delta_1 + \Delta_2))f_1f_2 = [\exp(\frac{1}{\hbar}\Delta_1)f_1][\exp(\frac{1}{\hbar}\Delta_2)f_2].$$

At this point we must remember that our hypothesis is valid in the context of Kähler manifolds with their associated Kähler polarizations [so that \mathcal{H}_i consists (locally) of holomorphic functions on M_i]. If M_1 and M_2 are two cotangent bundles $M_i = T^*Q_i$ and if we use the vertical polarizations $(\Rightarrow \mathcal{H}_i \cong \text{functions on } Q_i)$ then the axiom of noninterference is satisfied by the geometric quantization procedure (at least for the directly quantizable observables, i.e., the ones which are linear in p).

To summarize, the merits of our hypothesis are threefold: (1) the quantization results in \mathbb{C}^n are the same as for Weyl-Wigner, Berezin, and *-products, (2) the quantization results in \mathbb{C}^n are the same as for the vertical polarization, and (3) the quantization result satisfies the noninterference axiom.

Considering the above-mentioned arguments in favor of the hypothesis, I propose to change the geometric quantization scheme in such a way that the factor $(1 + \frac{1}{4}\hbar\Delta_{dR})$ in (2.3) is replaced by $\exp(\frac{1}{4}\hbar\Delta_{dR})$, i.e., considering geometric quantization as a first order approximation in \hbar to "real" quantum mechanics. Unfortunately however, I have no conceptual idea how to change geometric quantization in order to obtain this result.

Remark 7.1: The arguments in favor of our hypothesis are arguments with physics in mind; from the point of view of representation theory of Lie groups this hypothesis is not so good. It should also be mentioned that the change $(1 + \frac{1}{4}\hbar\Delta_{dR}) \rightarrow \exp(\frac{1}{4}\hbar\Delta_{dR})$ does not explain the substitution $4\lambda^3 - 3\lambda^4 \rightarrow \lambda^2 J_0(-2i \log \lambda)$ made by Rawnsley and Sternberg in Ref. 22 to obtain (by means of geometric quantization) representations of SL(3,R). Remark 7.2: All preceding results concerning geometric quantization were derived by using the metalinear correction. When one uses $\frac{1}{2}$ -densities instead of $\frac{1}{2}$ -forms, there are some (minor) differences, some of which will be listed below.

(1) $H_{h'h}(z,z^{\dagger}) = (\pi\hbar)^{-n/2}h'(z)^{\dagger}h(z)\exp(-\frac{1}{2}f/\hbar)$

[formula (2.2)].

(2)
$$L_f h = -i\hbar X_f h + h \cdot \left[f - \frac{\partial F}{\partial z^j} \cdot g^{jk} \cdot \frac{\partial f}{\partial z^{k^\dagger}} \right]$$

(3) Starting at formula (2.3) the term $(1 + \frac{1}{4}\hbar\Delta_{dR})$ should be replaced everywhere by $(1 + \frac{1}{4}\hbar\Delta_{dR})$,

(4) The formulas $R_f(h'h) = d\alpha$, $\alpha = (-1)^{n-1} \times i\hbar H_{h'h} d'' (f_{\epsilon_{n-1}})$ remain valid (with the changed function $H_{h'h}$).

(5) In Sec. V the exponent $-\frac{1}{2}(1 + \lambda / \hbar)$ should be replaced by $-\frac{1}{2}\lambda / \hbar$ and in Sec. VI the exponent $\frac{1}{2}(\lambda / \hbar - 1)$ should be replaced by $\frac{1}{2}\lambda / \hbar$ (undoing of the metalinear correction).

The results with $\frac{1}{2}$ -densities are exactly equal to results obtained by using prequantization only and restricting oneself to holomorphic sections.

APPENDIX: PROOF OF THEOREM 6.5

We start with some notations: (1) $\mu(g,z) = \partial gz/\partial z$ and (2) $\mathbb{C}(z)$ denotes the field of quotients of polynomials in the variables $z_1,...,z^n$.

Proof of (a): sufficiency: We suppose that λ satisfies (6.10) and we have to prove that the transition functions (6.9) really define a holomorphic line bundle. Berezin proved in Ref. 12 that $N(z,w^{\dagger})$ is (the square of) an irreducible polynomial in the unknowns (z,w^{\dagger}) , so it follows from Proposition 6.3, Corollary 6.4, and the condition (6.10) on λ that for certain $g \in G$ [i.e., $g^{-1}(0)$ should exist] that

$$\operatorname{tr}_{12}(z) = \operatorname{const}^{\circ} N(z, w^{\dagger})^{(\lambda/\hbar - 1) \cdot v(M)/2} \in \mathbb{C}(z)$$

Since there exists a neighborhood of $id \in G$ for which this condition on g is satisfied and since $\mu(g,z)$ (see above) satisfies the relation

$$\mu(g \cdot g', z) = \mu(g, g'z) \cdot \mu(g', z),$$

it follows that $\operatorname{tr}_{12}(z) \in \mathbb{C}(z)$ for all $g \in G$ [use that for nonzero constants there always exists a $[(\lambda/\hbar - 1) \cdot \nu(M)/2]$ power and use that the components of gz are in $\mathbb{C}(z)$]. Hence for each pair of charts U_1 and U_2 connected by the element $g \in G$ the function $\operatorname{tr}_{12}(z)$ exists as a holomorphic function (in particular, as a quotient of polynomials).

The next step is to choose the functions $\operatorname{tr}_{12}(z)$ in such a way that they satisfy the cocycle condition, using the cover $\{U_g | g \in G\}$ (the different choices are different branches of the complex logarithm). For two charts U_g and $U_{g'}$ the connecting group element is $g'g^{-1}$ [see (6.1)] so $\operatorname{tr}_{(g,g')}(z)$ should be a choice for $\mu(g'g^{-1},z)^{(1-\lambda/\hbar)/2}$. We are going to define $\operatorname{tr}_{(g,g')}$ in two steps. We start with

$$\operatorname{tr}_{(e,g)}(z) = \mu(g,z)^{(1-\lambda/2)}$$

with an arbitrary choice of the branch of the log [except for

 $\operatorname{tr}_{(e,e)}(z) = 1$]. We know that $\operatorname{tr}_{(e,g)}(z) \in \mathbb{C}(z)$. Then we define

$$\mathrm{tr}_{(g,g')}(z) = \mathrm{tr}_{(e,g')}(g^{-1}z)/\mathrm{tr}_{(e,g)}(g^{-1}z) \in \mathbb{C}(z).$$

This $\operatorname{tr}_{(g,g')}(z)$ is for each fixed $z \in \mathbb{C}$ a $[(1 - \lambda /\hbar)/2]$ power of $\partial (g'g^{-1})z/\partial z$, hence it is a holomorphic solution of (6.9) and moreover the system of transition functions $\operatorname{tr}_{(g,g')}(z)$ defined in this way obviously satisfies the cocycle relation

$$\operatorname{tr}_{(g,g')}(z) \cdot \operatorname{tr}_{(g',g'')}((g'g^{-1})z) = \operatorname{tr}_{(g,g')}(z),$$

which proves that for λ 's satisfying (6.10) there exists a line bundle QB with transition functions defined by (6.9). N.B. In general, one cannot expect that for any choice of branches $\mu(g,z)^{(1-\lambda/\hbar)/2}$ the relation $\mu(gh,z)^{(1-\lambda/\hbar)/2}$ $= \mu(g,hz)^{(1-\lambda/\hbar)/2} \cdot \mu(h,z)^{(1-\lambda/\hbar)/2}$ is satisfied for all $g,h\in G$ simultaneously (there exist easy counterexamples); this fact shows the necessity of our detour by $\operatorname{tr}_{(e,g)}(z)$.

Proof of (a): necessity: We have seen that there exist λ 's for which the line bundle QB is defined and in the same way one can show that there exist λ 's for which the prequantum bundle L is defined. This shows that the group of periods of the symplectic form ω on M is discrete (see Ref. 15) and since M is compact it must be isomorphic to \mathbb{Z} , implying that the set of allowed values of λ is discrete (countably infinite). Hence there exists a smallest positive λ_0 for which L exists, and this λ_0 with the corresponding L_0 we will call the elementary prequantization. The general theory now shows that, given one quantum bundle QB(λ) for a certain value of λ , all other QB's (for other values of λ) can be obtained by tensoring with the elementary prequantization: $QB(\lambda) \otimes L_0(\lambda_0) = QB(\lambda + \lambda_0).$

We know two obvious solutions λ_1 and λ_2 : $\lambda_1 = \hbar$ [where QB is trivial: tr(z) = 1, see (6.9)] and λ_2 defined by $(\lambda_2/\hbar - 1) \cdot \nu(M)/2 = 1$ (for type I, II, and IV; $\frac{1}{2}$ for type III) with tr(z) = const $S(z,w^{\dagger})$, where $S(z,w^{\dagger})$ is the irreducible polynomial from Berezin [$S(z,w^{\dagger}) = N(z,w^{\dagger})$ for type I, II, and IV and $S(z,w^{\dagger})^2 = N(z,w^{\dagger})$ for type III]. From this we deduce that the difference $\lambda_2 - \lambda_1$ is an integer multiple k of λ_0 .

Now choose any $g \in G$ such that $S(z, w^{\dagger})$ has a simple zero in $z [w^{\dagger}$ depends on g; such a g exists, although it is not true, in general, that for all $w S(z, w^{\dagger})$ should have a simple zero in z]. From $\lambda_2 - \lambda_1 = k\lambda_0$ it follows that $QB(\lambda_2)$ $\cong QB(\lambda_1) \otimes \otimes^k L_0 \cong \otimes^k L_0$ [remember that $QB(\lambda_1)$ is trivial]. Denote by T(z) the transition function of L_0 between the two local charts U_e and U_g (our special g of above!) then by the bundle equivalence there exist nonzero holomorphic functions κ_e on U_e and κ_g on U_g such that $T = \kappa_e^{-1} S \kappa_g$. Since $U_e \cong U_g \cong \mathbb{C}^n$ there exist global k th roots of κ so there exists on $U_e \cap U_g$ a k th root of $S(z, w^{\dagger})$. Because $U_e \cap U_g$ $= \{z \in U_e | S(z, w^{\dagger}) \neq 0\}$ and because $S(z, w^{\dagger})$ has a simple zero in z this leads to a contradiction except if k = 1, proving the necessity of condition (6.10).

Proof of (b): If k = 0 (i.e., $\lambda = \lambda_1$ and QB is trivial) then \mathscr{H} consists of global holomorphic sections of the trivial bundle over a compact complex manifold, hence these sections must be constant, implying $\mathscr{H} \cong \mathbb{C}$. Another way to derive this result is by using the transition functions, if the global section ψ is represented on two local charts by $h_1(z)$ and $h_2(z)$ then $h_1(z) = h_2(gz)$. Now both $h_1(z)$ and $h_2(z)$ are

global holomorphic functions on \mathbb{C}^n and gz is a quotient of polynomials with nonconstant denominator (in general) so $h_1(z)$ and $h_2(z)$ must be constant.

Now suppose k < 0 (and $\lambda > 0$) then we know from the proof of part (a) that $tr_{12}(z) = polynomial(z)^k$ so the relation $h_1(z) = tr_{12}(z)h_2(z)$ combined with $h_1(z)$ globally holomorphic shows that this is possible only if $h_1(z) = 0 = h_2(z)$, hence $\mathscr{H} = \{0\}$.

For k > 0 the relation $h_1(z) = \operatorname{tr}_{12}(z)h_2(z)$ poses a condition on $h_1(z)$ and $h_2(z)$ which restricts these holomorphic functions to polynomials of fixed degree, so \mathcal{H} is finite dimensional. The same conclusion can be obtained by inspecting the condition on the sections imposed by \mathcal{H} [see (6.6)], $h: \mathbb{C}^n \to \mathbb{C}$ holomorphic is an element of \mathcal{H} iff

$$\int |h(z)|^2 N(z,z^{\dagger})^{-(\lambda/\hbar+1)\cdot\nu(M)/2} d\operatorname{Leb}^{2n} < \infty,$$

which also shows that h is restricted to polynomials of fixed degree $[N(z,z^{\dagger})$ is a polynomial]. Q.E.D.

¹G. M. Tuynman, "Generalized Bergman kernels and geometric quantization," J. Math. Phys. 28, 593 (1987).

²D. J. Simms and N. Woodhouse, *Lectures on Geometric Quantization* (Springer, Berlin, 1977).

- ³B. Kostant, "Quantization and unitary representations" in *Lectures in Modern Analysis and Applications III*, edited by C. T. Taam, LNM 170 (Springer, Berlin, 1970).
- ⁴J. Sniatycki, Geometric Quantization and Quantum Mechanics (Springer, Berlin, 1980).

- ⁵A. Weil, Variétés Kähleriennes (Hermann, Paris, 1958).
- ⁶F. A. Berezin, "Quantization," Math. USSR Izv. 8, 1109 (1974).
- ⁷F. A. Berezin, "General concept of quantization," Commun. Math. Phys. **40**, 153 (1975).
- ⁸J.-M. Souriau, Structures des systèmes dynamiques (Dunod, Paris, 1969).
- ⁹C. Duval, J. Elhadad, and G. M. Tuynman, "Hyperfine interaction in a classical hydrogen atom and geometric quantization," J. Geom. Phys. 3, 401 (1986).
- ¹⁰S. Helgason, Differential Geometry, Lie Groups and Symmetric Spaces (Academic, Orlando, FL, 1978).
- ¹¹L. K. Hua, "Harmonic analysis of functions of several complex variables in the classical domains," Trans. Math. Monogr. 6 (Am. Math. Soc., Providence, RI, 1963).
- ¹²F. A. Berezin, "Quantization in complex symmetric spaces," Math. USSR Izv. 9, 341 (1975).
- ¹³R. Abraham and J. E. Marsden, Foundations of Mechanics (Benjamin/ Cummings, Reading, MA, 1978), 2nd ed.
- ¹⁴V. Guillemin and S. Sternberg, Symplectic Techniques in Physics (Cambridge U. P., Cambridge, 1984).
- ¹⁵G. M. Tuynman and W. A. J. J. Wiegerinck, "Central extensions and physics," preprints of the Mathematisch Instituut, Report 87-10, Amsterdam, 1987.
- ¹⁶C. Moreno and P. Ortega-Navarro, "Deformations of the algebra of functions on hermitian symmetric spaces resulting from quantization," Ann. Isnt. H. Poincaré Sec. A 38, 215 (1983).
- ¹⁷C. Moreno, "*-products on some Kähler manifolds," Lett. Math. Phys. 11, 361 (1986).
- ¹⁸C. Moreno, "Invariant star products and representations of compact semisimple Lie groups," Lett. Math. Phys. 12, 217 (1986).
- ¹⁹J. Rawnsley and P. Robinson, "*Mp^c* structures and geometric quantization," University of Warwick, preprint 1984.
- ²⁰C. Moreno and P. Ortega-Navarro "*-products on D¹(C), S² and related spectral analysis," Lett. Math. Phys. 7, 181 (1983).
- ²¹C. Moreno (private communication).
- ²²J. Rawnsley and S. Sternberg, "On representations associated to the minimal nilpotent coadjoint orbit of SL(3,R)," Am. J. Math. 104, 1153 (1982).

A class of integrable potentials

Tanaji Sen^{a)}

Institute For Theoretical Physics, State University of New York at Stony Brook, Stony Brook, New York 11794

(Received 16 June 1986; accepted for publication 30 July 1987)

A class of time independent two-dimensional integrable potentials, all possessing an invariant of the same general form, is constructed. One of these potentials is superintegrable, its invariants realize the symmetry algebra sO(3) for negative energies, e(2) for zero energy, and sO(2,1) for positive energies. A transformation of coupling constants reveals that in parabolic coordinates this potential is the harmonic oscillator acted on by constant forces. This and another potential in the class may be considered as successive extensions of the Kepler potential. The analytic properties of these integrable systems in the complex time plane are also discussed.

I. INTRODUCTION

The study of integrable systems has witnessed a rapid growth in the past few years. Mostly study has focused on infinite-dimensional systems governed by nonlinear evolution equations. Of late, the identification and study of finitedimensional integrable Hamiltonian systems has also proved to be of considerable interest. This has followed the realization that an arbitrary Hamiltonian system is likely to be nonintegrable and perhaps chaotic, the canonical example of such a system being the Hénon–Heiles system.¹ Interest in integrable Hamiltonian systems has been partly motivated by the practical need of realizing them in plasma physics and accelerators and partly by the desire to understand what characterizes integrability.

We will consider only two-dimensional, time-independent, classical Hamiltonian systems here. Such a system is integrable if there exists an additional analytic, single-valued, globally defined constant of the motion besides the energy. Darboux² may have been the first to give a systematic method of constructing a constant (or invariant) for a Hamiltonian system. In this direct method one postulates the form of the invariant, usually polynomial in the momenta. Demanding that the Poisson bracket of the invariant and the Hamiltonian vanish leads to a set of equations whose solution gives both the integrable potential and the complete invariant. This method has been used by several authors³ to construct a variety of integrable potentials. Most of the results obtained up to now have been reviewed by Hietarinta.⁴ All integrable potentials admitting invariants which are linear or quadratic in the momenta are known. At higher orders our knowledge is much less complete. The majority of results at higher order have been found by restricting the form of the invariant or making a particular choice for the form of the potential. In this paper we continue that tradition and choose our polynomial invariant to be of a certain allowable form and construct a class of integrable potentials which admit this type of invariant. There do exist other methods of identifying integrable systems, in particular, the Lax pair method,⁵ and Painlevé analysis.⁶ Indeed the latter

has proved to be a useful complement to the direct method in integrability studies.

This paper is organized as follows. In Sec. II we introduce the method due to Darboux and write the particular choice of invariant. Assuming that the invariant is of *n*th order in the momenta, we find the necessary general form of a potential for it to admit an invariant of this type. In Sec. III and IV we carry out an explicit construction of the integrable potentials at n = 3 and n = 4, respectively. The potential we find in Sec. III,

$$V = \alpha z^{-1/2} + \beta \bar{z}^{-1/2} + \gamma (z\bar{z})^{-1/2}, \qquad (1.1)$$

is superintegrable and turns out to be closely related to the two-dimensional Kepler potential. One of the four potentials we construct in Sec. IV can be viewed as the next extension of (1.1). In Sec. III we also look briefly at the quantum mechanics of (1.1) and obtain its energy spectrum by the use of its dynamical symmetry group. In Sec. V we perform a Painlevé analysis on the equations of motion of the integrable systems to test the nature of the singularities. Finally we conclude in Sec. VI with some remarks.

II. AN ANSATZ FOR THE INVARIANT

Consider the Hamiltonian to be of the standard form

$$H = \frac{1}{2}(p_x^2 + p_y^2) + V(x,y)$$
(2.1)

admitting a polynomial invariant (of nth order in the momenta) of the general form

$$I = \sum_{\substack{i=0\\i+j\leq n}}^{n} \sum_{\substack{j=0\\i+j\leq n}}^{n} a_{ij}(x,y) p_{x}^{i} p_{y}^{j}, \qquad (2.2)$$

where i + j is either even or odd, according as *n* is even or odd.⁷ The coefficients $a_{ij}(x,y)$ with total order i + j = n can be determined to be⁸

$$a_{pn-p} = \sum_{r=0}^{n-p} \sum_{s=0}^{p} (-1)^{s} {r+s \choose r} A_{p-sn-p+s}^{r+s} x^{r} y^{s},$$

$$p = 0, 1, ..., n,$$
(2.3)

where the A_{ij}^k are constants whose values depend on the potential. So far most of the invariants known⁴ with n > 2 are those where a_{pn-p} are constants for all p. Thompson⁹ and more recently Leach¹⁰ and Sen¹¹ have considered invariants

^{a)} Present address: Department of Applied Physics, Columbia University, New York, New York 10027.

with

$$a_{p\,n-p} = (-1)^p \binom{n}{n-p} x^{n-p} y^p, \qquad (2.4)$$

the invariant being

$$I = (xp_y - yp_x)^n + \text{lower-order terms.}$$
(2.5)

Here we concentrate on finding those integrable potentials for which

$$a_{pn-p} = (-1)^{p} \left[\binom{n-2}{n-p} x^{n-p} y^{p-2} + \binom{n-2}{n-p-2} x^{n-p-2} y^{p} \right], \qquad (2.6)$$

i.e., the invariant is of the form

$$I = (xp_y - yp_x)^{n-2} (p_x^2 + p_y^2) + \text{lower-order terms.}$$
(2.7)

For I to be an invariant, its time derivative or equivalently its Poisson bracket with H (since I is explicitly time independent) must vanish. This condition results in a system of equations, the complete solution of which proceeds in two steps. First, we find that the potential V must obey an nthorder linear partial differential equation (PDE). The solution of this PDE determines V as a sum of n arbitrary functions of known arguments. Next, these arbitrary functions are subject to certain nonlinear equations. Solving these nonlinear equations in the second stage of the analysis is considerably simpler if we choose an appropriate coordinate system. This choice of coordinate system is indicated by the arguments of the functions appearing in the first step. For our choice of invariant (2.7), these natural coordinates are the complex coordinates. Complex coordinates were also used by Kaushal et al.¹² but no new integrable systems were found.

We now make the point transformation to complex coordinates and perform the analysis from the beginning in this system. The Hamiltonian (2.1) is now of the form

$$H = 2p_z p_{\bar{z}} + V(z,\bar{z}) \tag{2.8}$$

and the invariant (2.7) is

$$I(z,\bar{z},p_{z},p_{\bar{z}}) = (zp_{z} - \bar{z}p_{\bar{z}})^{n-2} p_{z} p_{\bar{z}} + \sum_{\substack{i=0\\i+j \le n-2}}^{n-2} \sum_{\substack{j=0\\i+j \le n-2}}^{n-2} c_{ij}(z,\bar{z}) p_{z}^{i} p_{\bar{z}}^{j}.$$
(2.9)

Here the coefficients $c_{ij}(z,\overline{z})$ of total order i+j=n are (p=1,...,n-1),

$$c_{pn-p} = (-1)^{n-p-1} {\binom{n-2}{n-p-1}} z^{p-1} \overline{z}^{n-p-1}$$
(2.10)

and

$$c_{n0} = 0 = c_{0n}. (2.11)$$

The condition that the Poisson bracket of I and H vanish results in a system of equations (i, j = 0, 1, ..., n - 1),

$$2(\partial_z c_{ij-1} + \partial_{\bar{z}} c_{i-1j}) = (i+1)c_{i+1i} \partial_z V + (j+1)c_{ij+1} \partial_{\bar{z}} V, \qquad (2.12)$$

obtained by equating the different powers of $p_z p_{\overline{z}}$ to 0. One must recall here the stipulation $c_{ij} = 0$ if i, j < 0 or i + j > n.

The solution to this set of equations at a particular n determines the integrable potentials and the complete invariants.

Consider the following set of n equations (j=0,1...n-1):

$$2(\partial_z c_{n-j-1j-1} + \partial_{\overline{z}} c_{n-j-2j})$$

= $(n-j)c_{n-jj} \partial_z V + (j+1)c_{n-j-1j+1} \partial_{\overline{z}} V$
(2.13)

obtained from (2.12) by setting i + j = n - 1. The *n*th-order linear PDE for V is obtained by eliminating the coefficients of order (n - 2) on the left-hand side from this system of equations. Some of these PDE's and their solutions (obtained by the method of characteristics) are

$$n = 3 \quad (z \partial_z^2 \partial_{\bar{z}} - \bar{z} \partial_z \partial_{\bar{z}}^2) V = 0$$

$$\Rightarrow V(z,\bar{z}) = A(z) + B(\bar{z}) + C(z\bar{z}), \qquad (2.14)$$

$$n = 4 \quad [z^2 \partial_z^3 \partial_{\bar{z}} - \bar{z}^2 \partial_z \partial_{\bar{z}}^3 + 5(z \partial_z^2 \partial_{\bar{z}} - \bar{z} \partial_z \partial_{\bar{z}}^2)] V = 0$$

$$\Rightarrow V = A(z) + B(\bar{z}) + C(z\bar{z}) + (1/z\bar{z})D(\bar{z}/z), \qquad (2.15)$$

$$n = 5 \left[(z^{3}\partial_{z}^{4}\partial_{\bar{z}} + z^{2}\bar{z}\partial_{z}^{3}\partial_{\bar{z}}^{2} - z\bar{z}^{2}\partial_{z}^{2}\partial_{\bar{z}}^{3} - \bar{z}^{3}\partial_{z}\partial_{\bar{z}}^{4} \right]$$

+ $12(z^{2}\partial_{z}^{3}\partial_{\bar{z}} - \bar{z}^{2}\partial_{z}\partial_{\bar{z}}^{3})$
+ $30(z\partial_{z}^{2}\partial_{\bar{z}} - \bar{z}\partial_{z}\partial_{\bar{z}}^{2}) V = 0$
 $\Rightarrow V = A(z) + B(\bar{z}) + C(z\bar{z})$
+ $\frac{1}{z\bar{z}}D_{1}\left(\frac{\bar{z}}{z}\right) + \frac{1}{(z\bar{z})^{3/2}}D_{2}\left(\frac{\bar{z}}{z}\right),$ (2.16)

where A, B, C, ... are arbitrary functions of their arguments. It can be shown that at *n*th order the solution of the linear PDE is

$$V = A(z) + B(\bar{z}) + C(z\bar{z}) + \sum_{k=2}^{n-2} \frac{1}{(z\bar{z})^{k/2}} D_k\left(\frac{\bar{z}}{z}\right).$$
(2.17)

This only completes the first stage of the analysis. At the second stage we find that these *n* arbitrary functions in (2.17) are also subject to $\lfloor n/2 \rfloor - 1$ additional nonlinear equations ($\lfloor n/2 \rfloor$ is the smallest integer> n/2). The solutions to these nonlinear equations (which must be found at each value of *n* separately) determine the exact forms of the integrable potentials. In the following two sections we carry out this process at n = 3,4.

III. CUBIC INVARIANT

The invariant in complex coordinates is

$$I = (zp_z - \bar{z}p_{\bar{z}})p_z p_{\bar{z}} + c_{10} p_z + c_{01} p_{\bar{z}}.$$
(3.1)

The equations determining c_{10} and c_{01} are [from (2.12)],

$$2 \partial_{\overline{z}} c_{10} = z \partial_{\overline{z}} V,$$

$$2 \partial_{z} c_{10} + 2 \partial_{\overline{z}} c_{01} = 2(z \partial_{z} V - \overline{z} \partial_{\overline{z}} V),$$

$$2 \partial_{z} c_{01} = -\overline{z} \partial_{z} V$$
(3.2)

and

$$c_{10}\,\partial_z\,V + c_{01}\,\partial_{\bar{z}}\,V = 0. \tag{3.3}$$

Eliminating c_{10} and c_{01} from (3.2) we get the PDE in (2.14). We now choose to write the solution as

$$V = A'(z) + B'(\bar{z}) + C(z\bar{z}).$$
(3.4)
The derivatives (denoted by primes) are taken for later convenience. The equations in (3.2) can now be solved for c_{10} and c_{01} . The general solution is

$$2c_{10} = 2zA'(z) - A(z) + zB'(\bar{z}) + zC(z\bar{z}),$$
(3.5)
$$2c_{01} = -[\bar{z}A'(z) + 2\bar{z}B'(\bar{z}) - B(\bar{z}) + \bar{z}C(z\bar{z})],$$

where we have absorbed three arbitrary constants, one each in A, B, and C. The potential must also satisfy (3.3). This is a nonlinear compatibility condition that will determine the functional forms of A, B, and C. Substituting from (3.5), this equation can be written as

$$f(z) - g(\bar{z}) + h(z,\bar{z}) = 0,$$
 (3.6)

where

$$f(z) = (2zA' - A)A'', \quad g(\overline{z}) = (2\overline{z}B' - B)B'',$$

$$h(z,\overline{z}) = [\overline{z}(zA' - A) - z(\overline{z}B' - B)]C' \quad (3.7)$$

$$+ [zA'' - \overline{z}B'']C + zB'A'' - \overline{z}A'B''.$$

To solve (3.6), set¹³

$$f(z) = a, \quad g(\overline{z}) = b, \quad h(z,\overline{z}) = b - a,$$
 (3.8)

where a and b are constants. The first two equations in (3.8) are used to determine A and B, respectively, and then the last can be used to find C. The general solution of

f(z) = (2zA' - A)A'' = a,

where $a \neq 0$ can only be given in the parametric form

$$A(z) = z(t - a/t) + k_1/t,$$

$$z = t^2 \exp\left(\frac{t^2}{2a}\right) \left\{\frac{k_1}{a} \frac{d}{da^{-1}} \left[\sqrt{2a} \operatorname{erf}\left(\frac{t}{\sqrt{2a}}\right)\right] + k_2\right\},$$
(3.9)

where t is the parameter, k_1 and k_2 are arbitrary constants. A similar solution holds for $B(\bar{z})$ with $b \neq 0$. The last of (3.8) can be written in the form

$$\partial_{z} [(zA' - A)(B' + C)] - \partial_{\overline{z}} [(\overline{z}B' - B)(A' + C)] = b - a.$$
(3.10)

This equation together with the above solutions for A(z) and $B(\bar{z})$ has not proved tractable when either of a, b or both are nonzero, nor even in the case b - a = 0. The question whether there do exist solutions to (3.8) under the above conditions is left open. However, it can be shown that there is no solution with A, B, and C analytic in their arguments, which is compatible with (3.10) and the solution for A(z) and $B(\bar{z})$ obtained above.

With a = 0 = b, A(z) and $B(\overline{z})$ are given by

$$A(z) = \alpha z^{1/2}, \quad B(\bar{z}) = \beta \bar{z}^{1/2},$$
 (3.11)

where α , β are arbitrary constants, complex in general. [We can ignore the cases $A''(z) = 0 = B''(\overline{z})$ since they lead to a potential with spherical symmetry.] The equation for C now reduces to

$$2z\bar{z}C' + C = 0, (3.12)$$

$$C(z\overline{z}) = \gamma(z\overline{z})^{-1/2}, \qquad (3.13)$$

where γ is an arbitrary constant, complex in general. The

integrable Hamiltonian is

$$H = 2p_z p_{\bar{z}} + \alpha z^{-1/2} + \beta \bar{z}^{-1/2} + \gamma (z\bar{z})^{-1/2} \qquad (3.14)$$

(after rescaling $\alpha_{,\beta}$) and is complex in general. As we will see, this system turns out to be closely related to the Kepler potential. Most real integrable systems remain integrable under complex extension but complex integrable potentials by themselves have also been considered before (see, e.g., Hietarinta¹⁴). Physically we will be interested only in the real versions but for the purpose of studying integrability we work with the general complex potential. To complete the determination of the invariant, the coefficients c_{10} and c_{01} may be found using (3.5). The complete invariant is

$$I = (zp_{z} - \bar{z}p_{\bar{z}})p_{z}p_{\bar{z}} + \frac{1}{2}[\beta\bar{z}^{-1/2} + \gamma(z\bar{z})^{-1/2}]zp_{z} - \frac{1}{2}[\alpha z^{-1/2} + \gamma(z\bar{z})^{-1/2}]\bar{z}p_{\bar{z}}.$$
 (3.15)

Before we discuss the system (3.14) in detail, let us consider the possibility of other solutions to (3.6), with a = 0 = b. If A and B both vanish then C can be arbitrary. The potential is spherically symmetric and the Hamiltonian admits a linear invariant, the angular momentum. In all other cases, e.g., if only one of A, B, C vanishes, then it is straightforward to show that the only possible solutions are special cases of (3.14).

The Hamiltonian (3.14) reduces to the two-dimensional Kepler system when $\alpha = 0 = \beta$. That system has the angular momentum and the two components of the Runge-Lenz vector as invariants. Checking for lower-order invariants we find that the above system also admits two quadratic invariants

$$I_{1} = (zp_{z} - \overline{z}p_{\overline{z}})p_{z} + \frac{1}{2}[\beta \overline{z}^{1/2} - \alpha z^{-1/2}\overline{z} - \gamma z^{-1/2}\overline{z}^{1/2}], \qquad (3.16)$$

$$I_{2} = (zp_{z} - \bar{z}p_{\bar{z}})p_{\bar{z}} + \frac{1}{2}[\beta z\bar{z}^{-1/2} - \alpha z^{1/2} - \gamma z^{1/2}\bar{z}^{-1/2}]. \quad (3.17)$$

Thus this system has four invariants including the energy. For an autonomous system with s degrees of freedom there may exist only 2s - 1 independent invariants for the system to be able to evolve in time. The invariants constructed here are related by the algebraic equation

$$(I_1I_2 + \frac{1}{4}\gamma^2)H + \frac{1}{2}(\alpha^2 I_1 - \beta^2 I_2) + \frac{1}{2}\alpha\beta\gamma = 2I_3^2, \quad (3.18)$$

where the cubic invariant has been relabeled I_3 . With $\alpha = 0 = \beta$, I_1 and I_2 reduce to the two components of the Runge-Lenz vector and I_3 to the product of the angular momentum and the energy. The existence of the quadratic invariants implies that the Hamilton-Jacobi equation separates in one of the orthogonal coordinate systems. The system above separates in parabolic coordinates ζ_{1}, ζ_{2} : $z = (\zeta_1 + i\zeta_2)^2$; in which the Hamiltonian is

$$H = \frac{1}{\zeta_1^2 + \zeta_2^2} \left[\frac{1}{2} \left(p_{\zeta_1}^2 + p_{\zeta_2}^2 \right) + (\alpha + \beta) \zeta_1 + i(\beta - \alpha) \zeta_2 + \gamma \right].$$
(3.19)

This is real valued if we take $\beta = \alpha^*$ and it turns out to be one of the four potentials found by Winternitz *et al.*¹⁵ for which the Hamilton-Jacobi equation separates in two coordinate systems (here two parabolic coordinate systems mutually perpendicular to each other) and thus admits two independent quadratic invariants. Consider the physically more interesting case of the real potential. Define the following real-valued invariants:

$$J_{1} = i(\epsilon/2H)^{1/2} [I_{1} + I_{2} + (1/2H)(\alpha^{2} - \alpha^{*2})],$$

$$J_{2} = (\epsilon/2H)^{1/2} [I_{1} - I_{2} - (1/2H)(\alpha^{2} + \alpha^{*2})],$$
 (3.20)

$$J_{3} = -(2i/H)I_{3},$$

where $\epsilon = \operatorname{sgn} H$. The Poisson bracket relations between them are

$$\{J_1, J_2\} = -\epsilon J_3, \ \{J_2, J_3\} = J_1, \ \{J_3, J_1\} = J_2.$$
 (3.21)

Thus the Poisson algebra is isomorphic to the Lie algebra of SO(3) for negative energies, to E(2) for zero energy, and to SO(2,1) for positive energies, respectively. This of course is also true for the two-dimensional Kepler potential. The realization of a dynamical symmetry group (energy dependent) by the invariants singles out this potential from the other three potentials found by Winternitz *et al.*¹⁵ These potentials are

(i)
$$V = \alpha (x^2 + y^2) + \beta_1 x^{-2} + \beta_2 y^{-2}$$

 $= \alpha r^2 + (1/r^2) [\beta_1 \sec^2 \theta + \beta_2 \csc^2 \theta],$
(ii) $V = \alpha (4x^2 + y^2) + \beta x + \gamma y^{-2}$
 $= [1/(\xi_1^2 + \xi_2^2)] [4\alpha (\xi_1^6 + \xi_2^6) + \beta (\xi_1^4) - \xi_2^4) + (\gamma/4) (\xi_1^{-2} + \xi_2^{-2})],$
(iii) $V = \frac{\alpha}{2r} + \frac{1}{4r^2} [\beta_1 \sec^2 \frac{\theta}{2} + \beta_2 \csc^2 \frac{\theta}{2}]$
 $= \frac{1}{\xi_1^2 + \xi_2^2} [\alpha + \beta_1 \xi_1^{-2} + \beta_2 \xi_2^{-2}],$ (3.22)

where (r,θ) are the polar coordinates and (ζ_1,ζ_2) are parabolic coordinates. The configuration paths in each of the four potentials are closed periodic trajectories. Furthermore, potential (iii) also reduces to the Kepler potential when $\beta_1 = 0 = \beta_2$. However, for these three potentials, the invariants do not close under the Poisson bracket operation to give a finite-dimensional algebra. Classically at least, these potentials do not admit a symmetry group even though their motions are completely degenerate.

Besides the harmonic oscillator, the Kepler potential, and the four-potentials mentioned above, at least three other potentials are known to be superintegrable (i.e., admit three algebraically independent invariants) in two dimensions. In an arbitrary number of dimensions N the known superintegrable potentials, admitting 2N - 1 independent invariants, are again the harmonic oscillator and the Kepler potential. It would be interesting to check if the higher-dimensional analogs of the potentials listed in (3.22) and in particular of (3.19), are superintegrable or not.

The close connection of the system (3.19) to the Kepler system can also be seen in the following way. It is well known that under the time coordinate transformation

$$t \to s; \frac{ds}{dt} = (\zeta_1^2 + \zeta_2^2)^{-1},$$

the Kepler potential is transformed into the harmonic oscil-

lator when expressed in parabolic coordinates.¹⁶ We have

$$\frac{d\zeta_1}{ds}=p_{\zeta_1},\quad \frac{d\zeta_2}{ds}=p_{\zeta_2},$$

and the energy equation (3.19) for the real potential can be written as

$$-\gamma = (\frac{1}{2}p_{\xi_1}^2 + \frac{1}{2}p_{\xi_2}^2) - E(\zeta_1^2 + \zeta_2^2) + \kappa_1\zeta_1 + \kappa_2\zeta_2,$$
(3.23)

where $\kappa_1 = \alpha + \alpha^*$, $\kappa_2 = i(\alpha^* - \alpha)$ are real constants. Here $-\gamma$ can be interpreted as the Hamiltonian for a twodimensional isotropic harmonic oscillator (with coupling constant -E) acted on by a constant force. This noncanonical time transformation also serves as an example of a coupling constant metamorphosis between two integrable systems.¹⁷

One may try to look for a geometrical reason behind the degeneracy of the system (3.19), given that it is so closely related to the Kepler sytem. The Kepler potential is of course spherically symmetric but in addition, the equations of motion resulting from it are invariant under the transformation

$$\mathbf{r} \rightarrow \mathbf{r}(1 + \mathbf{c} \cdot \mathbf{r})^{-1}, \quad dt \rightarrow dt(1 + \mathbf{c} \cdot \mathbf{r})^{-2},$$

where c is an arbitrary constant vector. This invariance has been shown to lead to the conservation of the Runge-Lenz vector.¹⁸ No such invariance transformation (i.e., a point transformation in space) exists here. Again, unlike the Kepler potential, the equations of motion of (3.19) do not admit any finite, nontrivial (i.e., besides time translation) space-time symmetry transformation. See the Appendix for the proof. Alternatively one might try transforming to momentum space, as was done for the hydrogen atom by Fock who reduced the motion by a stereographic projection to free flow on a hypersphere (for bound states) or on a hyperboloid (for scattering states). The reason why this works is that the invariants for the Kepler potential involve the space coordinates only linearly or have 1/r whose Fourier transform is simple. In our case, the nonlinear terms occurring with α , β in (3.15)–(3.17) would appear to rule out such a simple picture in momentum space. Thus the realization of the space on which the motion of the system (3.19) would be a free flow is still left open.

Let us look briefly at the quantum mechanics of this system. The presence of the quadratic invariants implies that the quantum version of this potential is integrable too.¹⁹ The quantum invariants can be calculated from (3.15)-(3.17) by use of either of the symmetrization, Born–Jordan, or Wigner–Weyl correspondence rules²⁰ to get

$$\begin{split} \bar{I}_{1} &= (zp_{z} - \bar{z}p_{\bar{z}})p_{z} - (i/2)p_{z} \\ &+ \frac{1}{2}(\beta \bar{z}^{1/2} - \alpha z^{-1/2} \bar{z} - \gamma z^{-1/2} \bar{z}^{1/2}), \\ \bar{I}_{2} &= (zp_{z} - \bar{z}p_{\bar{z}})p_{\bar{z}} + (i/2)p_{\bar{z}} \\ &+ \frac{1}{2}(\beta z \bar{z}^{-1/2} - \alpha z^{1/2} + \gamma z^{1/2} \bar{z}^{-1/2}), \quad (3.24) \\ \bar{I}_{3} &= (zp_{z} - \bar{z}p_{\bar{z}})p_{z}p_{\bar{z}} + \frac{1}{2}(\beta \bar{z}^{-1/2} + \gamma (z \bar{z})^{-1/2})zp_{z} \\ &- \frac{1}{2}(\alpha z^{-1/2} + \gamma (z \bar{z})^{-1/2})\bar{z}p_{\bar{z}} \\ &+ (i/8)(\alpha z^{-1/2} - \beta \bar{z}^{-1/2}). \end{split}$$

The relation (3.18) between the invariants is altered to

$$(\hat{I}_1\hat{I}_2 + i\hat{I}_3 + \frac{1}{4}\gamma^2)\hat{H} + \frac{1}{2}(\alpha^2\hat{I}_1 - \beta^2\hat{I}_2) + \frac{1}{2}\alpha\beta\gamma$$

 $= 2(\hat{I}_3^2 - \frac{1}{16}\hat{H}^2).$ (3.25)

For the real potential the same expressions as in (3.20) can be used to define the quantities \hat{J}_1 , \hat{J}_2 , \hat{J}_3 . Now the commutator algebras of these invariants yields the same Lie algebras as in the classical case. Following Pauli's derivation of the hydrogen spectrum, the bound state energy spectrum of this model can be obtained by using the Casimir operator for SO(3),

$$C = \hat{J}_{1}^{2} + \hat{J}_{2}^{2} + \hat{J}_{3}^{2}$$

= $-\left(\frac{1}{4} + \frac{1}{2\hat{H}}\left(\gamma^{2} + \frac{2|\alpha|^{2}\gamma}{\hat{H}} + \frac{|\alpha|^{4}}{\hat{H}^{2}}\right)\right)$ (3.26)

[using (3.25)]. Equating this to the eigenvalues of C, viz, j(j+1) where $j = 0, \frac{1}{2}, 1,...$ we get the energy eigenvalue equation (n = 2j + 1 = 1,2,3...),

$$n^{2}E^{3} + 2\gamma^{2}E^{2} + |\alpha|^{2}\gamma E + 2|\alpha|^{4} = 0, \qquad (3.27)$$

a result also obtained by Winternitz *et al.*¹⁵ by solving the Schrödinger equation.

Analogous to the classical result, the commutator algebra of the quantum invariants for the potentials in (3.22) does not close. However, the symmetry group SU(2) [or O(3), locally isomorphic to it] can be realized at the quantum mechanical level by defining appropriate ladder operators.¹⁵

IV. QUARTIC INVARIANT

The invariant is

$$I = (zp_z - \bar{z}p_{\bar{z}})^2 p_z p_{\bar{z}} + c_{20} p_z^2 + c_{11} p_z p_{\bar{z}} + c_{02} p_{\bar{z}}^2 + c_{00}.$$
(4.1)

The system of equations (2.12) at n = 4 reduces to

$$2 \partial_{\overline{z}} c_{20} = z^2 \partial_{\overline{z}} V,$$

$$2 \partial_{z} c_{20} + 2 \partial_{\overline{z}} c_{11} = 3z^2 \partial_{z} V - 4z\overline{z} \partial_{\overline{z}} V,$$

$$2 \partial_{z} c_{11} + 2 \partial_{\overline{z}} c_{02} = -4z\overline{z} \partial_{z} V + 3\overline{z}^2 \partial_{\overline{z}} V,$$

$$2 \partial_{z} c_{02} = \overline{z}^2 \partial_{z} V$$

(4.2)

and

$$2 \,\partial_z c_{00} = c_{11} \,\partial_z \,V + 2c_{02} \,\partial_{\bar{z}} \,V,$$

$$2 \,\partial_{\bar{z}} c_{00} = 2c_{20} \,\partial_z \,V + c_{11} \,\partial_{\bar{z}} \,V.$$
(4.3)

Differentiating thrice to eliminate the coefficients on the lefthand side of (4.2) we find that the potential obeys the linear PDE (2.15). As before, we write the general solution for V as

$$V(z,\bar{z}) = A'(z) + B'(\bar{z}) + C(z\bar{z}) + (1/z\bar{z})D(\bar{z}/z).$$
(4.4)

The coefficients at order 2 may be obtained by integrating (4.2) to get the general solution

$$c_{20} = \frac{1}{2} [3z^{2}A' - 2zA + z^{2}B' + z^{2}C + zD/\overline{z} + k_{2}],$$

$$c_{11} = -[\overline{z}(2zA' - A) + z(2\overline{z}B' - B) + z\overline{z}C + 3D + k_{1}],$$

$$c_{02} = \frac{1}{2} [\overline{z}^{2}A' + 3\overline{z}^{2}B' - 2\overline{z}B + \overline{z}^{2}C + (\overline{z}/z)D + k_{0}],$$

(4.5)

where k_0 , k_1 , k_2 are arbitrary constants. Two other constants have been absorbed in A and B. The nonlinear compatibility condition on V can be obtained by eliminating c_{00} from (4.3). It is

$$[2c_{20}\partial_{z}^{2} - 2c_{02}\partial_{\bar{z}}^{2} + (2\partial_{z}c_{20} - \partial_{\bar{z}}c_{11})\partial_{z} + (\partial_{z}c_{11} - 2\partial_{\bar{z}}c_{02})\partial_{\bar{z}}]V = 0.$$
(4.6)

As was the case for the cubic invariant, this can be written in the form

$$f(z) - g(\bar{z}) + h(z,\bar{z}) = 0,$$
 (4.7)

where

$$f(z) = (3z^{2}A' - 2zA + k_{2})A''' + 3(z^{2}A'' + 2zA' - A)A'', \quad g(\bar{z}) = (3\bar{z}^{2}B' - 2\bar{z}B + k_{0})B''' + 3(\bar{z}^{2}B'' + 2\bar{z}B' - B)B'',$$

$$h(z,\bar{z}) = [2\bar{z}^{2}(z^{2}A' - zA) - 2z^{2}(\bar{z}^{2}B' - \bar{z}B) + k_{2}\bar{z}^{2} - k_{0}z^{2}]C'' + 3[\bar{z}(z^{2}A'' + zA' - A) - z(\bar{z}^{2}B'' + \bar{z}B' - B)]C'$$

$$+ [z^{2}A''' + 3zA'' - \overline{z}^{2}B''' - 3\overline{z}B'']C + \frac{1}{z^{4}} \left[2\overline{z}(zA' - A) - 2z(\overline{z}B' - B) + k_{2}\frac{\overline{z}}{z} - k_{0}\frac{z}{\overline{z}} \right]D'''$$

$$+ \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(3z^{2}A'' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A''' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A''' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A''' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A''' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A''' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A'' - 5zA' + 5A) - \frac{1}{z^{2}}(3\overline{z}^{2}B'' - \overline{z}B' + B) + \frac{4k_{2}}{z^{2}} - \frac{2k_{0}}{z^{2}} \right]D'' + \frac{1}{z^{2}} \left[-\frac{1}{z^{2}}(z^{2}A'' - A) + \frac{1}{z^{2}}(z^{2}A'' - A) + \frac{1}{z^{2}}(z^{2}A'' - A) + \frac{1}{z^{2}}(z^{2}A'' - A) \right]$$

$$z^{2} [z^{2} - \frac{1}{\overline{z}} (\overline{z}^{3}B''' + \overline{z}B' - B) + \frac{k_{2}}{z^{2}} - \frac{k_{0}}{\overline{z}^{2}}]D + (z^{2}A''' + 3zA'')B' - A'(\overline{z}^{2}B''' + 3\overline{z}B'').$$

On the grounds of tractability, we solve (4.7) by setting each of f, g, and h to 0 and look for power law solutions for A(z) and $B(\overline{z})$. This also requires $k_2 = 0 = k_0$. The power law solutions obtained from f(z) = 0 are

$$A(z) = z, z^{1/2}, z^{1/3},$$

and similarly for $B(\bar{z})$. Writing $A(z) = \alpha z^r$, $B(\bar{z}) = \beta \bar{z}^s$, where α, β are arbitrary complex constants, the equation $h(z,\bar{z}) = 0$ reduces to

$$\begin{aligned} & [2\alpha(r-1)z^{r+1}z^{s}-2\beta(s-1)z^{2}\overline{z}^{s+1}]C''+3[\alpha(r-1)(r+1)z^{r}\overline{z}-\beta(s-1)(s+1)z\overline{z}^{s}]C'+[\alpha r(r-1)(r+1)z^{r-1}\\ & -\beta s(s-1)(s+1)\overline{z}^{s-1}]C+z^{-4}[2\alpha(r-1)z^{r}\overline{z}-2\beta(s-1)z\overline{z}^{s}]D''+z^{-2}[-\alpha(r-1)(3r-5)z^{r-1}\\ & -\beta(s-1)(3s-1)\overline{z}^{s-1}]D'+z^{-1}\overline{z}^{-1}[\alpha(r-1)^{3}z^{r-1}-\beta(s-1)^{3}\overline{z}^{s-1}]D-\alpha\beta rs[(r-1)(r+1)\\ & -(s-1)(s+1)]z^{r-1}\overline{z}^{s-1}=0. \end{aligned}$$

$$(4.9)$$

2845 J. Math. Phys., Vol. 28, No. 12, December 1987

This is a linear equation for the two functions $C(z\overline{z})$ and $D(\bar{z}/z)$. It will be solved by the method of separating variables. There are six distinct values for the pair (r,s): (1,1), $(1,\frac{1}{2}), (1,\frac{1}{3}), (\frac{1}{2},\frac{1}{2}), (\frac{1}{2},\frac{1}{3}), (\frac{1}{3},\frac{1}{3}).$ With r = 1, A'(z) reduces to a constant and can be dropped from the potential, i.e., we can put $\alpha = 0$ for r = 1. Similarly for s = 1, set $\beta = 0$.

Consider first (r,s) = (1,1). In this case (4.9) is satisfied identically so C and D can be completely arbitrary. The Hamiltonian is

$$H = 2p_z p_{\bar{z}} + C(z\bar{z}) + (1/z\bar{z})D(\bar{z}/z)$$

$$\equiv \frac{1}{2} \left(p_r^2 + \frac{1}{r^2} p_{\theta}^2 \right) + F(r) + \frac{1}{r^2} G(\theta), \qquad (4.10)$$

where r, θ are the polar coordinates. For this system the Hamilton-Jacobi equation separates in polar coordinates,²¹ thus there exists an invariant quadratic in the momenta. The quartic invariant found here is just the product of that invariant and the Hamiltonian.

We give as an example, the method of solution for one of the other cases, $(r,s) = (\frac{1}{2},\frac{1}{2})$. On separating variables, (4.9) reduces to

$$\begin{split} &8\xi^{2}C'' + 18\xi C' + 3C = \lambda\xi^{-1}, \\ &8(\alpha\eta^{1/2} - \beta)\eta^{2}D'' + 2(7\alpha\eta^{1/2} - \beta)\eta D' \\ &+ (\alpha\eta^{1/2} - \beta)D = -\lambda(\alpha\eta^{1/2} - \beta), \\ &\text{where } \xi = z\overline{z}, \ \eta = \overline{z}/z, \ C = C(\xi), \ D = D(\eta), \text{ and } \lambda \text{ is the} \end{split}$$

separation constant. These equations have the solutions

$$C = \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} + \lambda \xi^{-1},$$

$$D = (\eta^{1/2} - \delta)^{-2} [\nu_1 \eta^{1/4} (\eta^{1/2} + \delta) + \nu_2 \eta^{1/2}] - \lambda,$$

where $\delta = \beta / \alpha$ and $\mu_1, \mu_2, \nu_1, \nu_2$ are arbitrary complex constants. The potential is

$$V(z,\overline{z})$$

1

$$= \alpha z^{-1/2} + \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} + \xi^{-1} (\eta^{1/2} - \delta)^{-2} [\nu_1 \eta^{1/4} (\eta^{1/2} + \delta) + \nu_2 \eta^{1/2}].$$
(4.11)

The λ terms cancel out and we have rescaled α , β .

For the case $(\frac{1}{2},\frac{1}{3})$ it is easy to show that there exists no solution to (4.9). In the other three cases the analysis proceeds similarly.

Thus we have four new and distinct cases of integrable potentials with quartic invariants

(I)
$$V(z,\overline{z}) = \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} + \xi^{-1}$$

 $\times [\nu_1 \eta^{1/4} + \nu_2 \eta^{1/2}],$
(II) $V(z,\overline{z}) = \beta \overline{z}^{-2/3} + \mu_1 \xi^{-1/3} + \mu_2 \xi^{-2/3} + \xi^{-1}$
 $\times (\nu_1 \eta^{1/3} + \nu_2 \eta^{2/3}),$
(III) $V(z,\overline{z}) = \alpha z^{-1/2} + \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4}$
 $+ \xi^{-1} (\eta^{1/2} - \delta)^{-2}$
 $\times [\nu_1 \eta^{1/4} (\eta^{1/2} + \delta) + \nu_2 \eta^{1/2}],$

TABLE I. Integrable potentials with quartic invariants.

Potential	Invariant
$V = A'(z) + B'(\bar{z}) + C(\xi) + \xi^{-1}D(\eta)$	$I = I_4 + I_2$ $I_4 = (zp_z - \overline{z}p_2)^2 p_z p_{\overline{z}}$
$\xi = z\overline{z}, \eta = \overline{z}/z$	I_2
(I) $\beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} + \xi^{-1} (\nu_1 \eta^{1/4} + \nu_2 \eta^{1/2})$	$\frac{1}{2}(B'+C+\xi^{-1}D)z^{2}p_{z}^{2}-(C+3\xi^{-1}D)\xi p_{z}p_{\bar{z}}+\frac{1}{2}(-B'+C+\xi^{-1}D)\overline{z}^{2}p_{z}^{2}$ - {($v_{1}+v_{2}\eta^{1/4}$)[$\mu_{1}z^{-3/4}\overline{z}^{-1/4}+v_{2}z^{-1}\overline{z}^{-1/2}+z^{-3/2}\overline{z}^{-1/2}(v_{1}+v_{2}\eta^{1/4})$] + ($\beta/4$)($\mu_{2}\eta^{-1/4}-v_{1}\xi^{-1/4}+\beta z$)}
(II) $\beta \overline{z}^{-2/3} + \mu_1 \xi^{-1/3} + \mu_2 \xi^{-2/3} + \xi^{-1} (\nu_1 \eta^{1/3} + \nu_2 \eta^{2/3})$	$\frac{1}{2}(B'+C+\xi^{-1}D)z^{2}p_{z}^{2}+(B'-C-3\xi^{-1}D)\xi p_{z}p_{\bar{z}}+\frac{1}{2}(-3B'+C+\xi^{-1}D)\bar{z}^{2}p_{\bar{z}}^{2}$ - {($v_{1}+v_{2}\eta^{1/3}$)[$\mu_{1}z^{-2/3}+\mu_{2}z^{-1}\bar{z}^{-1/3}+z^{-5/3}\bar{z}^{1/3}(v_{1}+v_{2}\eta^{1/3})$] + $\beta z^{2/3}(\mu_{2}\xi^{-1/3}+v_{2}z^{-4/3}+\beta\eta^{-1/3})$ }
(III) $\alpha z^{-1/2} + \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} $ + $\xi^{-1} (\eta^{1/2} - \delta)^{-2} $ × $[\nu_1 \eta^{1/4} (\eta^{1/2} + \delta) + \nu_2 \eta^{1/2}]$	$\frac{1}{2}(-A'+B'+C+\xi^{-1}D)z^{2}p_{z}^{2}-(C+3\xi^{-1}D)\xi p_{z}p_{\overline{z}}+\frac{1}{2}(A'-B'+C+\xi^{-1}D)\overline{z}^{2}p_{\overline{z}}^{2}$ $+\frac{1}{2}\mu_{2}(\alpha\eta^{1/4}+\beta\eta^{-1/4})-\frac{1}{2}\nu_{1}\xi^{-1/4}+\frac{1}{4}(\alpha\overline{z}^{1/2}-\beta z^{1/2})^{2}-(\eta^{1/2}-\delta)^{-2}z^{-1}$ $\times [\nu_{1}(\mu_{1}\eta^{-1/4}+\mu_{2}\overline{z}^{-1/2})(\eta^{1/2}+\delta)+\nu_{2}(\mu_{1}+\mu_{2}\xi^{-1/4})]$ $-(\eta^{1/2}-\delta)^{-4}z^{-2}[\nu_{1}\eta^{-1/4}(\eta^{1/2}+\delta)+\nu_{2}]^{2}$
(IV) $\alpha^2 z^{-2/3} + \beta^2 \overline{z}^{-2/3} + \mu_1 \xi^{-1/3} + \mu_2 \xi^{-2/3} + \xi^{-1} \eta^{1/3} [\nu_1(\eta^{1/3} - \delta)^{-2} + \nu_2(\eta^{1/3} + \delta)^{-2}]$	$\frac{1}{2}(-3A'+B'+C+\xi^{-1}D)z^{2}p_{z}^{2}+(A'+B'-C-3\xi^{-1}D)\xi p_{z}p_{\overline{z}}+\frac{1}{2}(A'-3B'+C)$ $+\xi^{-1}D)\overline{z}^{2}p_{\overline{z}}^{2}+\mu_{2}(\alpha^{2}\eta^{1/3}+\beta^{2}\eta^{-1/3})+z\overline{z}^{-1/3}(\alpha^{2}\eta^{2/3}-\beta^{2})^{2}-[\mu_{1}(\nu_{1}+\nu_{2})$ $-2\alpha\beta(\nu_{1}-\nu_{2})]\overline{z}^{-2/3}+\nu_{1}(\eta^{1/3}-\delta)^{-2}[(2\beta^{2}-\mu_{1}\delta)(2\xi^{-1/3}-\delta\overline{z}^{-2/3})$ $-\mu_{2}z^{-1}\overline{z}^{-1/3}]+\mu_{2}(\eta^{1/3}+\delta)^{-2}[(2\beta^{2}+\mu_{1}\delta)(2\xi^{-1/3}+\delta\overline{z}^{-2/3})-\mu_{2}z^{-1}\overline{z}^{-1/3}]$
$\delta = \beta / \alpha$	$-z^{-5/3}\overline{z}^{-1/3}[\nu_1(\eta^{1/3}-\delta)^{-2}+\nu_2(\eta^{1/3}+\delta)^{-2}]^2$

where
$$\delta = \beta / \alpha$$
,
(IV) $V(z,\bar{z}) = \alpha^2 z^{-2/3} + \beta^2 \bar{z}^{-2/3} + \mu_1 \xi^{-1/3} + \mu_2 \xi^{-2/3} + \xi^{-1} \eta^{1/3} [\nu_1(\eta^{1/3} - \delta)^{-2} + \nu_2(\eta^{1/3} + \delta)^{-2}],$
(4.12)

where $\delta = \beta / \alpha$, along with those obtained by $z \leftrightarrow \overline{z}$.

To complete the determination of the invariant, the equations in (4.3) must be integrated to find c_{00} for each of the potentials. The potentials together with their invariants are listed in Table I. With α and β equal to 0, the quartic invariants reduce to products of a quadratic invariant and the Hamiltonian.

Now some comments may be made on the nature of the potentials. All the integrable potentials constructed here (including the one in the previous section) are discontinuous along the positive real axis because of terms like $\bar{z}^{-1/2}$, etc. Potentials I and II in (4.12) are complex valued for all choices of β . Potential III is real valued under either of the following choices: (i) $\alpha = \beta$ and all coefficients are real; (ii) $\beta = \alpha^*$ and $v_1^2/v_1^{*2} = v_2/v_2^* = \alpha^{*2}/\alpha^2$; μ_1 , μ_2 are real. Potential IV is real valued under either of the following choices: (i) $\alpha = \beta$ and all coefficients are real; (ii) $\beta = \alpha^*$ and $v_1/v_1^* = v_2/v_2^* = \alpha^{*2}/\alpha^2$; μ_1 , μ_2 are real.

Potential III can be considered as an extension of the potential in (3.14) which itself was an extension of the Kepler potential. Do there exist extensions of these potentials with invariants at higher order? At fifth and sixth order, with power law solutions for A(z) and $B(\overline{z})$, no integrable potentials exist. Such is likely to be the case at higher orders as well. Thus there would appear to be no extensions to (3.14) and (4.11) along the lines considered here. With A and B set to 0, the solution (2.17) of the linear PDE for the potential reduces to

$$V = C(z\bar{z}) + \sum_{k=2}^{n-2} \frac{1}{(z\bar{z})^{k/2}} D_k\left(\frac{\bar{z}}{z}\right)$$
$$\equiv F(r) + \sum_{k=0}^{n-4} \frac{1}{r^{k+2}} G_k(\theta).$$
(4.13)

It has been found (Sen^{11}) that this is precisely the necessary general form of a potential for it to admit an invariant of the form given in (2.5) but of order (n-2). Thus if the potential is not to admit a lower order invariant both A and B must not vanish. The possibility of solutions other than power law for A(z) [or $B(\bar{z})$] at fifth and sixth order has not been investigated. Even at fourth order our results are incomplete and there might well exist other solutions to (4.7) besides those found here. The complexity of the forms of $f(z), g(\bar{z}), \text{ and } h(z,\bar{z})$ in (4.8) would appear to indicate that construction of other solutions may not be straightforward. It should also be noted that the quantum integrability of the potentials in (4.12) is not assured but must be checked for each case by computing the Moyal bracket of I and H. However, that question is not considered here.

At this stage one might well question the utility of constructing invariants at high orders. Certainly from a practical point of view, the behavior (in say, configuration space) of an integrable potential with a high-order invariant might appear to be little different from that of a nonintegrable potential. However, the existence of the invariant permits one to make global statements about the system. For instance, if the motion is bounded and a periodic solution is known to exist, then there must in fact be a one-parameter (or d - 1parameter for *d*-dimensional integrable systems) family of periodic solutions. Moreover, many systems can be approximated by integrable ones and it is useful to understand the kinds of behavior exhibited by a variety of integrable systems. If one's interest is more in the mathematical question of what determines integrability, then explicit examples of integrable potentials can be used to test any proposed integrability criteria. We do precisely this in the next section.

V. PAINLEVÉ ANALYSIS

The original conjecture that integrable systems should possess the Painlevé property, i.e., their only movable singularities in the complex time plane must be poles, was made by Ablowitz, Ramani, and Segur²² for nonlinear partial differential equations. This has been extended to discrete dynamical systems and has proved to be useful in identifying various integrable cases.⁶ There is, however, no general proof as of yet that integrable systems must satisfy this property and the relation between this analytic property and the existence of an invariant is still unclear. In this section the integrable potentials constructed in the previous two sections are tested for this property,

$$V = \alpha z^{-1/2} + \beta \overline{z}^{-1/2} + \gamma (z\overline{z})^{-1/2}.$$
 (5.1)

This is the potential in (3.14). The second-order equations of motion are

$$\ddot{z} = \beta \overline{z}^{-3/2} + \gamma z^{-1/2} \overline{z}^{-3/2}, \quad \ddot{z} = \alpha z^{-3/2} + \gamma z^{-3/2} \overline{z}^{-1/2}.$$
(5.2)

Assume that the dominant behavior of z and \overline{z} as $\tau = (t - t_0) \rightarrow 0$ is $z \sim a\tau^p$, $\overline{z} \sim b\tau^q$. Balancing the most singular terms, one finds the two possibilities: (i) $p = q = \frac{4}{3}$. The resonances are $r = -1, \frac{2}{3}, -\frac{1}{3}, -\frac{2}{5}$; (ii) $p = q = \frac{2}{3}$. The resonances are $r = 0, -1, \frac{2}{3}, -\frac{1}{3}$.

Besides -1,0 the other values of r should be positive in order to get the complete four-parameter solution. Since that does not hold in either case, the leading order behaviors obtained above cannot describe the general solution. However, we know that for this potential, the Hamilton-Jacobi equation separates in parabolic coordinates leading to two onedimensional problems. The trajectories can be obtained exactly. This was done by Winternitz *et al.*¹⁵ and we quote their result (for the real potential $\beta = \alpha^*$),

$$\zeta_{i} = \frac{\kappa_{i}}{2E} + \left(\frac{\kappa_{i}^{2}}{4E^{2}} - \frac{a_{i}}{E}\right)^{1/2} \sin\left(\sqrt{-2E}\,\tau + b_{i}\right), \quad i = 1, 2,$$
(5.3)

where ζ_i are the parabolic coordinates, *E* is the energy, τ is a time parameter, and $\kappa_1 = \alpha + \alpha^*$, $\kappa_2 = i(\alpha^* - \alpha)$. Here a_i , b_i are the four arbitrary parameters. Clearly this does not show any singular behavior.

Now consider the potentials with quartic invariants giv-

en in (4.12), in order,

$$V = \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/2} + \xi^{-1} (\nu_1 \eta^{1/4} + \nu_2 \eta^{1/2}).$$
 (5.4)

The equations of motion are

$$\ddot{z} = \beta \overline{z}^{-3/2} + \mu_1 z^{-1/2} \overline{z}^{-3/2} + \frac{3}{2} \mu_2 z^{-3/4} \overline{z}^{-7/4} + \frac{3}{2} \nu_1 z^{-5/4} \overline{z}^{-7/4} + \nu_2 z^{-3/2} \overline{z}^{-3/2},
$$\ddot{z} = \mu_1 z^{-3/2} \overline{z}^{-1/2} + \frac{3}{2} \mu_2 z^{-7/4} \overline{z}^{-3/4} + \frac{5}{2} \nu_1 z^{-9/4} \overline{z}^{-3/4} + 3 \nu_2 z^{-5/2} \overline{z}^{-1/2}.$$
(5.5)$$

Assuming $z \sim a\tau^p$, $\overline{z} \sim b\tau^q$ as $\tau = (t - t_0) \rightarrow 0$, we have the following possibilities: (1) $p = q = \frac{1}{2}$. The resonances are $r = -\frac{1}{4}, -1, 1, \frac{1}{2}$. This gives a three-parameter solution. (ii) p = 2 = -q. The resonances are r = -1, 0, 1, 2. This gives a four-parameter solution. (iii) p = -3, q = 5. The resonances are r = -2, -1, 0, 1. This gives a three-parameter solution. Thus only (ii) can describe the general solution. The expansions for z, \overline{z} read

$$z = a\tau^2 + \cdots, \quad \overline{z} = (\nu_2/2)^{2/3}a^{-5/3}\tau^{-2} + \cdots.$$
 (5.6)

Potential (II) in (4.12) shows similar behavior. Now consider potential (III),

$$V(z,\overline{z}) = \alpha z^{-1/2} + \beta \overline{z}^{-1/2} + \mu_1 \xi^{-1/2} + \mu_2 \xi^{-3/4} + \xi^{-1} (\eta^{1/2} - \delta)^{-2} [\nu_1 \eta^{1/4} (\eta^{1/2} + \delta) + \nu_2 \eta^{1/2}], \qquad (5.7)$$

where $\delta = \beta / \alpha$. It was remarked earlier in Sec. IV that this potential can be considered as an extension of the potential in (3.14) and indeed the singularity analysis shows similar behavior. The equations of motion for this potential are

$$\ddot{z} = \beta \overline{z}^{-3/2} + \mu_1 z^{-1/2} \overline{z}^{-3/2} + \frac{3}{2} \mu_2 z^{-3/4} \overline{z}^{-7/4} + (z^{-1/2} \overline{z}^{1/2} - \delta)^{-3} [\frac{5}{2} \nu_1 z^{-9/4} \overline{z}^{-3/4} + 3 \nu_2 z^{-2} \overline{z}^{-1} + \delta (3 \nu_1 z^{-7/4} \overline{z}^{-5/4} - \nu_2 z^{-3/2} \overline{z}^{-3/2} - \frac{3}{2} \delta \nu_1 z^{-5/4} \overline{z}^{-7/4}],$$

$$\ddot{z} = \alpha z^{-3/2} + \mu_1 z^{-3/2} \overline{z}^{-1/2} + \frac{3}{2} \mu_2 z^{-7/4} \overline{z}^{-3/4} + (z^{-1/2} \overline{z}^{1/2} - \delta)^{-3} [\frac{3}{2} \nu_1 z^{-13/4} \overline{z}^{1/4} + \nu_2 z^{-3} - \delta (3 \nu_1 z^{-11/4} \overline{z}^{-1/4} + 3 \nu_2 z^{-5/2} \overline{z}^{-1/2} + \frac{5}{2} \delta \nu_1 z^{-9/4} \overline{z}^{-3/4}]].$$
(5.8)

With $z \sim a\tau^p$, $\overline{z} \sim b\tau^q$, none of the various possibilities for p and q give a four-parameter solution. The most we get are the three-parameter solutions with either (i) p = 5, q = -3. Resonances are r = -2, -1,0,1; (ii) p = -3, q = 5. The same resonances as above.

We argue that the leading order behavior is not expected to be singular because of the following reason. For p = 5, q = -3 the most singular terms on the right-hand sides of (5.8) arise from the term $v_1 z^{-3/4} \overline{z}^{-5/4}$ in the potential (5.7). Now $z^{-3/4} \overline{z}^{-5/4} = r^{-2} e^{i\theta/2}$, where (r,θ) are the polar coordinates. With a potential of the form $V = r^{-2} f(\theta)$ the Hamilton-Jacobi equation separates and the coordinates are expressible in terms of periodic functions of time. A similar argument holds for case (ii). Thus the potential (called here the second extension of the Kepler potential) shows similar nonsingular behavior as the potential in (5.1) (the first extension of the Kepler potential).

Potential (IV) in (4.12) shows similar nonsingular behavior as the previous one. Thus it is seen that the equations of motion for the integrable potentials constructed in Sec. III and IV either do not exhibit nonsingular behavior or else have the Painlevé property.

VI. CONCLUDING REMARKS

In this work we have constructed examples of two-dimensional integrable Hamiltonian systems which admit an invariant of the chosen form (2.7). The potential in (3.14)reveals itself to be a close relative of the Kepler potential. Under the same time transformation which turns the Kepler potential into the harmonic oscillator in parabolic coordinates, the system (3.14) is transformed into the harmonic oscillator acted on by a constant force. The system (3.14) is also superintegrable and its invariants provide a representation of the Lie algebra sO(3) for negative energies and sO(2,1) for positive energies. Unlike the Kepler potential however, its equations of motion do not admit any finite symmetries. The potentials in (4.12) admit a quartic invariant of the form (2.7) and of them, potential (III) can be considered as an extension of (3.14). Recently it was shown by Yoshida²³ that potentials of the form

$$V = a/r + bx^n + cy^n, \tag{6.1}$$

where a, b, c, n are constants are integrable only if (i) n = 1, b, c arbitrary, (ii) n = 2, b = c, and (iii) n = 2, b = 4c (or c = 4b). Other perturbations to the Kepler potential of this form do not lead to an integrable system. In this context one may view (3.14) and potential III in (4.12) as perturbations of the Kepler potential which do preserve integrability. One should bear in mind that all potentials of the form

$$V = a/r + (1/r^2) f(\theta),$$
 (6.2)

where f is an arbitrary function, are integrable and admit a quadratic invariant. In other words, the Kepler system may also remain integrable under a whole class of perturbations unlike the cases mentioned above where integrability is preserved with only specific perturbations.

The analytic structure of the integrable systems constructed here has been studied and it was found that they either do not show singular behavior or else their movable singularities are indeed poles, thus verifying the Painlevé conjecture.

Let us indicate possible extensions to this work. Using (2.3) we can rewrite the general form (2.2) of an invariant as

$$I = C_{00}^{n}L^{n} + (C_{10}^{n-1}p_{x} + C_{01}^{n-1}p_{y})L^{n-1} + (C_{20}^{n-2}p_{x}^{2} + C_{11}^{n-2}p_{x}p_{y} + C_{02}^{n-2}p_{y}^{2})L^{n-2} + \cdots + C_{n0}^{0}p_{x}^{n} + C_{n-11}^{0}p_{x}^{n-1}p_{y} + \cdots + C_{0n}^{0}p_{y}^{n} + lower-order terms in the momenta, (6.3)$$

where C_{20}^{k} are constants and $L = xp_y - yp_x$. In this work we took $C_{20}^{n-2} = 1 = C_{02}^{n-2}$ and all others 0 while in Sen,¹¹ invariants with $C_{00}^{n} = 1$ and all other constants 0 have been studied. Clearly one must try to establish similar results or show the nonexistence of integrable potentials for other

choices of coefficients. In addition to polynomial invariants, Darboux's method can also be used⁴ to treat invariants which are rational or transcendental in the momenta. This extension needs to be studied in depth.

ACKNOWLEDGMENTS

I thank Professor Max Dresden for his encouragement and for reading the manuscript and Professor Jacques Perk for his comments and especially for his help in analyzing (3.6).

This work was supported in part by NSF Grant No. 473-2377A.

APPENDIX: LIE SYMMETRIES OF (3.14)

Prince and Eliezer²⁴ have shown that the Kepler potential admits three point Lie symmetry generators. They also claim that as a consequence the Runge-Lenz vector is conserved but this has been disputed.²⁵ We follow the notation of Prince and Eliezer and show that the system (3.14) (the first extension of the Kepler system) admits only the trivial time translation generator.

A system of second-order differential equations,

$$\mathbf{F}(t,\mathbf{x},\dot{\mathbf{x}},\ddot{\mathbf{x}}) = 0,\tag{A1}$$

admits a point Lie symmetry if it is invariant under the infinitesimal point transformations

$$t' = t + \epsilon \xi(\mathbf{x}, t), \quad x'_i = x_i + \epsilon \eta_i(\mathbf{x}, t)$$
 (A2)

(ϵ is an infinitesimal parameter) generated by the operator

$$U = \xi(\mathbf{x}, t)\partial_t + \eta_i(\mathbf{x}, t)\partial_{\mathbf{x}_i}.$$
 (A3)

The condition of invariance is

$$U''\mathbf{F} = 0, \tag{A4}$$

where U'' is the twice extended operator

$$U'' = \xi \partial_t + \eta_i \partial_{x_i} + \eta'_i \partial_{\dot{x}_i} + \eta''_i \partial_{\dot{x}_i}, \qquad (A5)$$

with

$$\eta_i^k(\mathbf{x}, \dot{\mathbf{x}}, ..., \mathbf{x}^{(k)}, t) \equiv \frac{d}{dt} \eta_i^{k-1} - x_i^{(k)} \frac{d}{dt} \xi,$$
 (A6)

where $\mathbf{x}^{(k)} \equiv (d^k/dt^k)\mathbf{x}$. The equations of motion for the system (3.14) are

$$\ddot{z} = \beta \overline{z}^{-3/2} + \gamma z^{-1/2} \overline{z}^{-3/2}, \quad \ddot{z} = \alpha z^{-3/2} + \gamma z^{-3/2} \overline{z}^{-1/2}.$$
(A7)

The infinitesimal symmetry generator is

$$U = \xi(z,\overline{z},t)\partial_t + \eta_1(z,\overline{z},t)\partial_z + \eta_2(z,\overline{z},t)\partial_{\overline{z}}, \qquad (A8)$$

and its twice extended form is

$$U'' = \xi \partial_t + \eta_1 \partial_z + \eta_2 \partial_{\bar{z}} + (\dot{\eta}_1 - \dot{z}\dot{\xi})\partial_{\dot{z}} + (\dot{\eta}_2 - \dot{\bar{z}}\dot{\xi})\partial_{\dot{z}} + (\ddot{\eta}_1 - 2\ddot{z}\dot{\xi} - \dot{z}\ddot{\xi})\partial_{\ddot{z}} + (\ddot{\eta}_2 - 2\ddot{\bar{z}}\dot{\xi} - \ddot{z}\ddot{\xi})\partial_{\ddot{z}}.$$
 (A9)

Applying this to (A7) and equating the coefficients of different powers of \dot{z} , $\dot{\bar{z}}$ to 0, we get ($\xi_{zz} \equiv \partial_z^2 \xi$, etc.),

(A10)

(A13)

$$\eta_{1,zz} = 2\xi_{,zi}; \quad \eta_{1,z\bar{z}} = \xi_{,\bar{z}i}; \quad \eta_{1,\bar{z}\bar{z}} = 0, \tag{A11}$$

-1/2--3/2 6

-3/2

$$\begin{aligned} \eta_{2,zz} &= 0; \quad \eta_{2,z\bar{z}} = \xi_{zt}; \quad \eta_{2,\bar{z}\bar{z}} = 2\xi_{,\bar{z}t}, \quad 2\eta_{1,zt} = 3(\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2})\xi_{,z} + (\alpha z^{-3/2} + \gamma z^{-3/2}\bar{z}^{-1/2})\xi_{,\bar{z}} + \xi_{,tt}, \\ \eta_{1,\bar{z}t} &= (\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2})\xi_{,\bar{z}}, \quad \eta_{2,zt} = (\alpha z^{-3/2} + \gamma z^{-3/2}\bar{z}^{-1/2})\xi_{,z}, \\ 2\eta_{2,\bar{z}t} &= (\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2})\xi_{,z} + 3(\alpha z^{-3/2} + \gamma z^{-3/2}\bar{z}^{-1/2})\xi_{,\bar{z}} + \xi_{,tt}, \\ \eta_{1,z}(\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2}) + \eta_{1,\bar{z}}(\alpha z^{-3/2} + \gamma z^{-3/2}\bar{z}^{-1/2}) + \eta_{1,tt} + \frac{1}{2}\eta_{1}\gamma(z\bar{z})^{-3/2} \\ &+ \frac{3}{2}\eta_{2}(\beta\bar{z}^{-5/2} + \gamma z^{-1/2}\bar{z}^{-5/2}) - 2(\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2})\xi_{,t} = 0, \\ \eta_{2,z}(\beta\bar{z}^{-3/2} + \gamma z^{-1/2}\bar{z}^{-3/2}) + \eta_{2,\bar{z}}(\alpha z^{-3/2} + \gamma z^{-3/2}\bar{z}^{-1/2}) + \eta_{2,tt} + \frac{1}{2}\eta_{2}\gamma(z\bar{z})^{-3/2} \end{aligned}$$

$$(z^{2} + \gamma z^{-5/2} \overline{z}^{-1/2}) - 2(\alpha z^{-3/2} + \gamma z^{-3/2} \overline{z}^{-1/2})\xi_{t} = 0.$$

It is straightforward to check that the only solution of these equations when $\alpha \neq 0 \neq \beta$ is

$$\xi = \text{const}, \, \eta_1 = 0 = \eta_2, \tag{A14}$$

giving the Lie symmetry generator

 $\xi_{,zz} = 0; \quad \xi_{,z\overline{z}} = 0; \quad \xi_{,\overline{zz}} = 0,$

 $+\frac{3}{2}\eta_1(\alpha z^{-5/2})$

$$U = \partial_t. \tag{A15}$$

This time translation symmetry that exists for all time-independent potentials is associated only with the conservation of energy. The importance of Lie symmetries arises by virtue of Lie's theorem that the infinitesimal point transformations can be integrated to give a finite continuous group of transformations. This is usually not the case for other kinds of local symmetries. The lack of any finite space-time symmetry transformations for this completely degenerate potential shows that there is no direct connection between such symmetries and degeneracy.

¹M. Hénon and C. Heiles, Astron. J. 69, 73 (1964).

-3/2

- ²G. Darboux, Arch. Neerl. (ii) 6, 371 (1901).
- ³E. T. Whittaker, Analytical Dynamics of Particles (Cambridge U.P., Cambridge, 1937), 4th ed.; C. R. Holt, J. Math. Phys. 23, 1037 (1982); L. S. Hall, Physica D 8, 90 (1983); A. Ankiewicz and C. Pask, J. Phys. A: Math. Gen. 16, 4203 (1983); G. Thompson, ibid. 17, 985 (1984); M. Wojciechowska and S. Wojciechowski, Phys. Lett. A 105, 11 (1984).
- ⁴J. Hietarinta, Phys. Rep. 147, 87 (1987).

⁵M. A. Olshanetsky and A. M. Perelomov, Invent. Math. 37, 93 (1976); M. Adler, Commun. Math. Phys. 55, 195 (1977).

- ⁶B. Dorizzi, B. Grammaticos, and A. Ramani, J. Math. Phys. 24, 2282, 2289 (1983).
- ⁷Compare Holt in Ref. 3.
- ⁸T. Sen, Phys. Lett. A 111, 97 (1985).
- ⁹G. Thompson, J. Math. Phys. 25, 3474 (1984).
- ¹⁰P. G. L. Leach J. Math. Phys. 27, 153, 2445 (1986).
- ¹¹T. Sen, Phys. Lett. A 122, 100 (1987).
- ¹²R. S. Kaushal, S. C. Mishra, and K. C. Tripathy, J. Math. Phys. 26, 420 (1985).
- ¹³This is not the most general case. In particular, there exists the solution $A(z) = z(a + m \log z), \quad B(\overline{z}) = \overline{z}(b + m \log \overline{z}), \quad C(z\overline{z})$ —arbitrary,

where a, b, m are constants, for which (3.8) is not satisfied. However, this solution leads to a spherically symmetric potential. I consider only the ansatz (3.8) here since this is the only one to have led to a nonspherically symmetric potential. I thank Jacques Perk for suggesting the transformation which converts (3.6) to an autonomous equation and for pointing out the above solution.

- ¹⁴J. Hietarinta, Phys. Lett. A 96, 273 (1983).
- ¹⁵P. Winternitz, J. A. Smorodinsky, M. Uhlir, and I. Fris, Sov. J. Nucl. Phys. 4, 444 (1967).
- ¹⁶E. Stiefel and G. Scheifele, *Linear and Regular Celestial Mechanics* (Springer, Berlin, 1971).
- ¹⁷J. Hietarinta, B. Grammaticos, B. Dorizzi, and A. Ramani, Phys. Rev.

Lett. 53, 1707 (1984).

- ¹⁸K. Mariwalla, Phys. Rep. 20, 287 (1975).
- ¹⁹J. Hietarinta, J. Math. Phys. 25, 1833 (1984).
- ²⁰L. Cohen, J. Math. Phys. 7, 781 (1966).
- ²¹L. D. Landau and E. M. Lifshitz, *Mechanics* (Pergamon, New York, 1976), 3rd ed.
- ²²M. J. Ablowitz, A. Ramani, and H. Segur, J. Math. Phys. 21, 715, 1006 (1980).
- ²³H. Yoshida, Phys. Lett. A 120, 388 (1987).
- ²⁴G. E. Prince and C. J. Eliezer, J. Phys. A: Math. Gen. 14, 587 (1981).
- ²⁵R. L. Schafir, J. Phys. A: Math. Gen. 14, L269 (1981); K. H. Mariwalla, *ibid.* 15, L467 (1982).

Second-order equation fields and the inverse problem of Lagrangian dynamics

G. Thompson

Department of Mathematics, University of Edinburgh, Edinburgh EH9 3JZ, United Kingdom

(Received 11 February 1987; accepted for publication 1 July 1987)

The transformation properties of determined, autonomous systems of second-order ordinary differential equations, identified as vector fields on the tangent bundle of the space of dependent variables, are derived and studied. The inverse problem of Lagrangian dynamics is studied from this transformation viewpoint as well as the problem of alternative Lagrangians. In particular, regular Lagrangians which are analytic as functions of the first derivatives are considered. Finally, the inverse problem for second-order systems corresponding to the geodesic flow of a symmetric linear connection is investigated.

I. INTRODUCTION

The inverse problem of Lagrangian dynamics asks for necessary and sufficient conditions that a given system of second-order ordinary differential equations are the Euler– Lagrange equations corresponding to a regular Lagrangian function. Allied to the inverse problem is the question of when two Lagrangian functions, which are not trivially equivalent (a term which will be explained in Sec. III), determine the same second-order system, the so-called problem of alternative Lagrangians. There has been a certain amount of progress made recently in answering these important questions, which have all sorts of theoretical and practical ramifications, but a complete solution is, for the moment at least, not in sight.

The main aim of this paper is to make a thorough study of the transformation properties of systems of autonomous second-order ordinary differential equations and apply the results obtained to the inverse problem. For the sake of simplicity, throughout this paper I shall only consider autonomous systems, though for the most part this is not an essential restriction. In Sec. II I consider the correspondence between such systems and certain kinds of vector fields known as second-order equation fields. This correspondence leads in a natural way to a "transformation law" for secondorder systems, several simple consequences of which are presented. The material of Sec. II (and Sec. III) is not intended to be simply an exercise in mathematical erudition: a modern approach to a system of differential equations (even in the context of Euclidean space) is to consider two such systems as being equivalent, if one can be transformed into the other using a change of coordinates belonging to some specified pseudogroup of coordinate transformations. This pseudogroup is chosen at the outset and is dictated by the significance attached to the independent and dependent variables. A set of differential equations then is more properly regarded as an equivalence class of systems, any two representatives of the class being related by a coordinate transformation of the given pseudogroup.

The other theme of the paper, in Secs. III–V, is a study of a restricted class of second-order equation fields, namely, those corresponding to second-order systems analytic in the first derivatives. In particular, the inverse problem for such systems is investigated. In Sec. III the inverse problem is introduced and some basic points concerning it are made. In Sec. IV some simple consequences of a direct approach to the inverse problem are derived. This approach is an alternative to the Helmholtz equations which provide necessary and sufficient conditions for the existence of a regular Lagrangian, whose Euler–Lagrange field is the given second-order equation field.^{1,2} In Sec. V I consider the inverse problem for second-order equation fields which are the geodesic sprays of symmetric linear connections. It is shown that the conditions for the existence of a Lagrangian have a simple geometric interpretation. These conditions are used to exhibit several examples of sprays which admit alternative Lagrangians.

Sections II and III employ some geometric machinery of the tangent bundle, a comprehensive treatment of which can be found in Crampin,³ whereas Secs. IV and V are analytical in character. As regards notation, M consistently denotes a "smooth," that is class C^{∞} , manifold. The tangent bundle submersions from TM to M and TTM to TM will be denoted by π and Π , respectively, and the tangent mapping of π from TTM to TM is denoted by $T\pi$. The second-order tangent bundle to M, that is, equivalence classes of (germs of) curves on M which agree to second order, is denoted by $T^{2}M$. If (x') is a system of local coordinates on M, there is a naturally induced system (x^i, u^i) on TM which I refer to as an adapted coordinate system. Similarly $(x^{i}, u^{i}, y^{i}, v^{i})$ denotes a coordinate system on TTM induced from (x^i, u^i) on TM. The notation of classical tensor calculus is used with a repeated index denoting summation. For a (covariant) tensor field of valence *n* whose components are $A_{i_1\cdots i_n}$, its symmetric part is denoted by $A_{(i_1 \cdots i_n)}$.

Finally, I mention two lifting constructions on the tangent bundle. First, a one-form α on M defines in a natural way a real-valued function on TM, denoted by $\tilde{\alpha}$. If $u \in TM$ and $\pi(u) = x, \tilde{\alpha}(u)$ is defined as the result of pairing u with $\alpha(x)$ (as elements of $T_x M$ and T_x^*M , respectively). In an adapted coordinate system (x^i, u^i) on TM if locally α is represented by $\alpha_i dx^i$, $\tilde{\alpha}$ is given locally by $(x^i, u^i) \mapsto \alpha_i u^i$. Second, if A is a tensor field on M of type (1, n) which is symmetric in its covariant arguments, it determines naturally a vertical vector field on TM written A^v . If (x^i, u^i) is an adapted coordinate system and A is represented locally by $A_{i_1}^j \cdots_{i_n} (\partial/x^j) \otimes dx^{i_1} dx^{i_2} \cdots dx^{i_n}, A^v$ is given locally by $A_{i_1}^j \cdots_{i_n} u^{i_1} u^{i_2} \cdots u^{i_n} (\partial/\partial u^j)$. This vertical lift construction is the natural generalization of the usual vertical lift construction in which a vector field on M is lifted to a vertical field on TM which is "constant on fibers."^{3,4}

II. FOUNDATIONAL REMARKS ON SECOND-ORDER EQUATIONS

A second-order equation field (SOEF) Γ on an *m*-dimensional manifold *M* is a section of *TTM* over *TM* with respect to both the submersion maps $T\pi$ and Π . In other words, Γ is a vector field on *TM* such that for all *u* in *TM*,

$$(\pi_*\Gamma)_u = u. \tag{2.1}$$

A SOEF provides a convenient geometric description of a regular, determined, autonomous system of second-order ordinary differential equations. To say the system is "determined" is to say that there are as many independent equations as there are dependent variables and by "regularity" I mean that the second-order derivatives can always be solved for as functions of the first-order derivatives and dependent variables. If (x^i, u^i) is a local coordinate system on TM adapted to the tangent bundle fibration, Γ assumes the form

$$\Gamma = u^{i} \frac{\partial}{\partial x^{i}} + f^{i} \frac{\partial}{\partial u^{i}}, \qquad (2.2)$$

where the f^{i} 's are functions of x^{i} and u^{i} . The second-order system corresponding to Γ is given by

$$\ddot{\mathbf{x}}^i = f^i(\mathbf{x}^j, \dot{\mathbf{x}}^j), \tag{2.3}$$

where the dot denotes differentiation with respect to the independent variable t. Setting $u^i = \dot{x}^i$ is the well-known device which effectively converts (2.3) into (2.2).

The second-order system (2.3) is amenable to a somewhat different geometric interpretation which at once explains the significance of the regularity condition and enables one to interpret (2.3) in terms of the second-order tangent bundle T^2M . In fact consider T^2M as a bundle over TM. (One should note carefully that T^2M naturally has the structure of an affine rather than a vector bundle over TM; for more about affine bundles see Crampin and Thompson.⁵) If (x^i, u^i, v^i) is an adapted coordinate system on T^2M , then one may consider (2.3) as defining (locally) a codimension m submanifold, Σ say, of T^2M by the conditions

$$v^{i} = f^{i}(x^{j}, u^{j})$$
 (2.4)

In fact Σ actually defines a section σ of T^2M over TM and thus may be identified with TM. Notice that it is precisely the regularity condition which guarantees that Σ defines a section of T^2M over TM.

I shall explain next how the considerations of the previous paragraph lead naturally to the definition of a SOEF as given above starting from the second-order system (2.3). Note first of all that T^2M is naturally an embedded submanifold of TTM. If (x^i, u^i, y^i, v^i) is a coordinate system on TTMconsonant with the adapted coordinate system (x^i, u^i, y^i, v^i) on T^2M , T^2M is locally the submanifold given by the conditions $u^i = y^i$. Let the submanifold map from T^2M to TTM be denoted by j. Notice that $j(T^2M)$ is invariant under the action of the canonical involution I on TTM. It follows that the map $j \circ \sigma$ defines a section of *TTM* over *TM* with respect to both the submersion maps $T\pi$ and Π , in other words is a SOEF on *M*.

I next introduce the notion of isomorphism SOEF's, which makes the collection of SOEF's on manifolds diffeomorphic to a fixed manifold into a groupoid. Suppose that Γ_1 and Γ_2 are SOEF's on M_1 and M_2 , respectively. Then Γ_1 and Γ_2 [or more accurately (Γ_1, M_1) and (Γ_2, M_2)] are isomorphic or equivalent, if there exists a diffeomorphism ϕ of M_1 with M_2 such that at each point of TM_1 ,

$$T\phi)_*\Gamma_1 = \Gamma_2. \tag{2.5}$$

Equation (2.5) enables us to define what it means for two second-order systems to be equivalent (at least secondorder systems corresponding to SOEF's). It is important to appreciate that this is the appropriate categorical meaning of equivalence of second-order systems; one could envisage other notions of equivalence, but these will either destroy some property which is characteristic of such systems or preserve some uncharacteristic property.

We can ask what (2.5) means locally. To this end, let $(\bar{x}^i, \bar{u}^i), (\bar{x}^i, \bar{u}^i)$ be two adapted coordinate systems on TM_1 and TM_2 , respectively, and suppose that

$$\Gamma_1 = u^i \frac{\partial}{\partial x^i} + f^i \frac{\partial}{\partial u^i}, \qquad (2.6)$$

$$\Gamma_2 = \bar{u}^i \frac{\partial}{\partial \bar{x}^i} + \bar{f}^i \frac{\partial}{\partial \bar{u}^i}.$$
(2.7)

If now the transformation $T\phi$ is described locally by

$$\bar{\mathbf{x}}^i = \bar{\mathbf{x}}^i(\mathbf{x}^{\ j}) , \qquad (2.8)$$

$$\bar{u}^{i} = \frac{\partial \bar{x}^{i}}{\partial x^{j}} u^{j}, \qquad (2.9)$$

we find (2.5) gives the following transformation law of second-order equations:

$$\bar{f}^{i} = f^{a} \frac{\partial \bar{x}^{i}}{\partial x^{a}} + \bar{u}^{a} \bar{u}^{b} \frac{\partial^{2} \bar{x}^{i}}{\partial x^{c} \partial x^{e}} \frac{\partial x^{c}}{\partial \bar{x}^{a}} \frac{\partial x^{e}}{\partial \bar{x}^{b}}.$$
(2.10)

Equation (2.10) may be interpreted in the spirit of classical tensor analysis; it says, assuming that Γ_1 and Γ_2 are analytic in the fiber u, that every term transforms tensorially, except the quadratic one, which transforms according to the transformation law for Christoffel symbols. Thus, for $n \neq 2$, the term of degree n in f^a , $A^a_{j_1\cdots j_n} u^{j_1}\cdots u^{j_n}$ say, arises from the type (1,n) tensor field $A^a_{j_1\cdots j_n}$ on M and $A^a_{j_1\cdots j_n} u^{j_1}\cdots u^{j_n} (\partial/\partial u^a)$ is just its vertical lift in the sense of Sec. I. We can therefore decompose a given SOEF Γ , which is analytic in u, into a sum of a spray together with various vector fields which are vertical lifts. Thus I shall write

$$\Gamma = \Lambda + A_0^{\nu} + A_1^{\nu} + A_3^{\nu} + A_4^{\nu} + \cdots, \qquad (2.11)$$

where Λ is the spray associated to Γ and A_n^{ν} is the vertical lift field corresponding to the terms of degree *n* in *u* and which is invariantly associated to Γ . It is worth noting too that the notion of a spray, which is by definition a SOEF in which the f^{j*} s are homogeneous quadratic polynomials in u^j , is well defined in virtue of (2.10); this is not obvious *a priori*.

I shall present next some applications of (2.5). I shall make use of the well-known result in connection theory

which asserts that if the curvature of a torsion-free (linear) connection vanishes, one can always introduce local coordinates relative to which the Christoffel symbols of the connection are zero.^{6,7}

Proposition 2.1: Suppose that Γ_1 and Γ_2 are two SOEF's analytic in the fiber variables. Suppose further, that Γ_1 has a nonzero term of degree $n \ (\neq 2)$ in u but that Γ_2 does not. Then Γ_1 and Γ_2 are not equivalent.

Proposition 2.2: Suppose we are given a second-order system of type (2.3) with corresponding SOEF Γ . Then we can find a local representation as

$$\ddot{\bar{x}}^i = \bar{f}^i(\bar{x}^j) \tag{2.12}$$

iff (i) $\Gamma = \Lambda + A_0^{\nu}$, that is, the f^{i} 's are even quadratic in u, and (ii) the curvature of the connection corresponding to Λ vanishes.

Proposition 2.3: Suppose we are given a second-order system of type (2.3) with corresponding SOEF Γ . Then we can find a linear representation

$$\ddot{x}^{i} = A^{i} + B^{i}_{j} \ddot{x}^{j} + C^{i}_{j} \ddot{x}^{j},$$
 (2.13)

where A^{i} is a constant *m* vector and B_{j}^{i} and C_{j}^{i} are constant $m \times m$ matrices iff (i) $\Gamma = \Lambda + A_{0}^{\nu} + A_{1}^{\nu}$, that is, the f^{b} s are quadratic in *u*; (ii) the curvature of the connection ∇ corresponding to Λ vanishes; and (iii) $\nabla A_{1} = 0$ and $\nabla^{2}A_{0} = 0$.

Condition (ii) says that we can remove the term quadratic in u and thereafter, according to (2.10), we must only consider transformations which are affine linear in x. (Any other transformation would reintroduce terms quadratic in u.) Condition (iii) then guarantees that the term of degree 0 in u is affine linear in x.

Thus for autonomous systems of second-order ordinary differential equations there is a complete characterization of those systems that are equivalent to linear systems. Moreover, this characterization yields an effective practical test for determining whether a given second-order system can be linearized. This is in stark contrast to the situation for even scalar nonautonomous second-order equations. (See Ref. 8 for some recent results on this subject.)

III. THE INVERSE PROBLEM OF LAGRANGIAN DYNAMICS

In this section I shall briefly review the geometrical approach to Lagrangian dynamics and describe the inverse problem in the light of the material of Sec. II. This is intended to put the inverse problem into some kind of perspective, prior to the more specific analysis in Secs. IV and V.

To formulate Lagrangian theory invariantly, we shall have to investigate the geometry of the tangent bundle TM in greater detail. In particular we shall need the canonical Liouville vector Δ and the 1-1 tensor field S, the so-called vertical endomorphism. The vector field Δ arises out of the action of the multiplication group of nonzero reals on each fiber of TM; specifically, if F is any real-valued function on TM and u an arbitrary point in TM,

$$(\Delta F)(u) = \frac{d}{dt} \left(F(tu) \right) \Big|_{t=0}.$$
(3.1)

The tensor S is defined as follows. Let $u \in TM$ and $X \in T_u(TM)$; then $S(X)_u$ is $\Delta(\pi_*X)$, which makes sense because π_*X is an element of TM. In an adapted coordinate system (x^i, u^i) on TM, Δ and S assume the form $u^i(\partial/\partial u^i)$ and $(\partial/\partial u^i) \otimes dx^i$, respectively.

Now suppose that L: $TM \rightarrow \mathbf{R}$ is a regular Lagrangian. This means that in the adapted coordinates (x^i, u^i) ,

$$\det\left(\frac{\partial^2 L}{\partial u^i \partial i^j}\right) \neq 0.$$
(3.2)

Now since S is a 1-1 tensor, it can be applied to the one-form dL to produce another one-form $S^{\circ}dL$. It follows from (3.2) that the two-form $\omega_L = d(S^{\circ}dL)$ is a symplectic form. The Euler-Lagrange field Λ determined by L is the unique vector field which satisfies

$$\dot{a}(\Lambda)\omega_L = -d(\Delta L - L) . \qquad (3.3)$$

One may readily check that Λ so defined is indeed a SOEF on M. The inverse problem of Lagrangian dynamics asks for necessary and sufficient conditions that a given SOEF is the Euler-Lagrange field corresponding to some regular Lagrangian function on TM. For more on the invariant formulation of Lagrangian dynamics the reader can refer to Refs. 3, 9, and 10.

Now consider all pairs $(\overline{M}, \overline{L})$ consisting of a manifold \overline{M} diffeomorphic to M and a regular Lagrangian function \overline{L} on $T\overline{M}$. The collection of all such pairs forms a category, indeed groupoid, in the obvious way. We can now describe the assignment of a SOEF corresponding to a regular Lagrangian as a functor \mathcal{F} from this groupoid to the groupoid of SOEF's on manifolds diffeomorphic to M. The inverse problem is then precisely that of characterizing the image of \mathcal{F} and the problem of alternative Lagrangians is essentially that of describing the extent to which \mathcal{F} fails to be injective.

Concerning the problem of alternative Lagrangians, it is well known that there are several rather trivial ways in which a given regular Lagrangian may be modified and yet still yield the same Euler-Lagrange vector field. Specifically, let L be such a regular Lagrangian; then the Lagrangian L, where

$$\overline{L} = \lambda L + \tilde{\alpha} + c, \qquad (3.4)$$

 $\lambda, c \in \mathbb{R}$, and α is a closed one-form on M ($\tilde{\alpha}$ denoting the corresponding real-valued function on TM which is linear in u) does indeed determine the same Euler-Lagrange vector field as L. The term $\tilde{\alpha}$ is known usually as a "gauge term" and is the global, autonomous version of the addition of a total time derivative of a function on M to the Lagrangian. Two Lagrangians L and \overline{L} related as in (3.4) will be called trivially equivalent; the problem of alternative Lagrangians is to be solved modulo this equivalence relation.

An important and basic question relating to the inverse problem is whether the property of a SOEF being a Euler-Lagrange field is invariant under transformations of the form $T\phi$, for ϕ a diffeomorphism of M. This fundamental, conceptual point is usually not addressed in the literature, though the answer to the question just posed can hardly be said to be obvious, given the definitions of a SOEF and the Euler-Lagrange field. In fact, let us apply the diffeomorphism $T\phi$ to either side of (3.3), noting that since Δ and S are natural objects on TM they are certainly preserved by $T\phi$, that is, $\Delta \cdot T\phi = \Delta$ and $S \cdot T\phi = S$; one obtains

$$i((T\phi)_*^{-1}\Lambda)\omega_{L\cdot T\phi} = -d\left(\Delta(L\cdot T\phi) - L\cdot T\phi\right). \quad (3.5)$$

Equation (3.5) says precisely that $(T\phi)_{*}^{-1}\Lambda$ is the Euler-Lagrange vector field corresponding to the regular Lagrangian $L \cdot T\phi$.

In terms of an adapted coordinate system (x^i, u^i) on *TM*, the f^{j*} s in a Euler-Lagrange field [see (2.3)] are determined in terms of the Lagrangian *L* according to the conditions

$$\frac{\partial^2 L}{\partial u^i \partial u^j} f^j = \frac{\partial L}{\partial x^i} - u^j \frac{\partial^2 L}{\partial x^j \partial u^i}.$$
(3.6)

The reader may care to give a proof in local coordinates of the result derived in the previous paragraph using (3.6). [One has to check that the f^{j*} s defined by (3.6) transform according to (2.10).] Perhaps the most obvious and direct approach to the inverse problem is to consider (3.6) as a system of m second-order partial differential equations for the unknown L; one has to give necessary and sufficient conditions on the f^{j} 's for there to exist a solution L which also satisfies the transversality condition (3.2). Of course this is a highly nontrivial problem. Now an answer of sorts to the inverse problem, as is well known, is provided by the Helmholtz conditions.^{1,2} However, this "solution" is no more an answer than that which says that a Lagrangian exists, if and only if there is a regular solution to (3.6). Indeed, the Helmholtz conditions, as given by Crampin¹ for example, are really only a way of recasting (3.6) as a first-order system of differential equations in terms of the intermediate variables $g_{ii} = \partial^2 L / \partial u^i \partial u^j$ (together with some purely algebraic conditions on the g_{ii} 's).

The one case in which (3.6) can be analyzed completely is when m = 1 so that (3.6) reduces to a single equation. In that case the existence of a Lagrangian L follows, at least if one assumes that the single function f is analytic in x and u, from the Cauchy-Kowalewski theorem. Furthermore, Lcan be found such that $\partial L^2 / \partial u^2 \neq 0$, at least on an open set in TM, and so L is a regular Lagrangian. As to the problem of alternative Lagrangians, it is easy to verify the following: if Lis a regular solution of (3.6) (with m = 1), and the energy H_L is defined by $H_L = u(\partial L / \partial u) - L$, then up to trivially equivalent Lagrangians, every other Lagrangian is given by \overline{L} where F is an arbitrary smooth function of H_L and

$$\overline{L} = u \int \frac{F(H_L) du}{u^2} \,. \tag{3.7}$$

[Of course in (3.7) an arbitrary function of x enters because of the integration with respect to u, but this simply results in the addition of a gauge term to the Lagrangian.]

The case of m = 1 is of fairly limited interest in its own right of course; however, it is useful for checking that results conjectured for arbitrary values of m reduce to a correct result for m = 1. Furthermore, if one is given a Lagrangian system with m arbitrary, one may always take the "direct product" (in the obvious informal sense) of it with a onedimensional Lagrangian and the resulting Lagrangian system will be one which admits alternative Lagrangians. Any attempt to solve the problem of alternative Lagrangians then, has to incorporate some way of detecting when this situation occurs; indeed it begs the question of whether alternative Lagrangians always arise in this way.

IV. LAGRANGIANS ANALYTIC IN u

The notion of equivalence of SOEF's introduced in Sec. II, namely, an equivalence that preserves the structure of tangent bundles, suggests that we consider systems of second-order equations like (2.3) in which the f^k 's are analytic functions of u. For a diffeomorphism of two tangent bundles which is lifted (or "prolonged") from a diffeomorphism of the corresponding manifolds, preserves all such notions as (real-valued) functions being "polynomial or analytic in the fiber." (On the other hand, it has to be admitted that this assumption of analyticity is a significant restriction; it would exclude, for example, the interesting families of alternative Lagrangians for a particle moving under a spherically symmetric potential given by Henneaux and Shepley¹¹).

In this section I investigate what the implications are of expanding the f^{j} 's in powers of u and using the decomposition given by (2.11). I shall begin by deriving some simple consequences of (3.6) in the shape of the following proposition.

Proposition 4.1: Let Γ be an SOEF on M. Then if there exists a Lagrange function L on TM such that Γ is the Euler-Lagrangian vector field of L and if L is analytic in u (about the zero section of TM), Γ is analytic in u. Furthermore, if L and the f^{j} 's are polynomial [in u and u^{i} , respectively, where (x^{i}, u^{i}) is an adapted coordinate system], then the f^{j} 's are at most quadratic polynomials. Finally, if it is assumed simply that L is quadratic in u (and regular), the f^{j} 's are necessarily at most quadratic in u^{i} .

Proof: The proof of the first statement is quite straightforward. [One may show that the matrix $(\partial^2 L / \partial u^i \partial u^j)^{-1}$ consists of functions analytic in u, the coefficients of which are determined in terms of the coefficients in the series expansion of L; I shall return to this point below.]

The proof of the second statement is clear from (3.6), because if L is of degree, say n, in u, the right-hand side of (3.6) is of degree at most n. Thus the left-hand side is of degree no more than n, which evidently means that the f^{j} 's can be at most quadratic.

To prove the third assertion, differentiate (3.6) fiberwise three times assuming that L is quadratic in u. One easily obtains

$$\frac{\partial^2 L}{\partial u^i \partial u^j} \frac{\partial^3 f^j}{\partial u^k \partial u^m \partial u^n} = 0.$$
(4.1)

Now (4.1) implies that the f^j 's are quadratic in u^i because the matrix $\partial^2 L / \partial u^i \partial u^j$ is nonsingular.

It is important to appreciate that the converse of each assertion made in Proposition 4.1 is false. Counterexamples can be obtained by considering a free particle (when the f^{j} 's are identically 0) in one dimension and noting some of the remarks made in Sec. III on one-dimensional systems. Similarly it is easy to generalize to appropriate counterexamples with an arbitrary number of dependent variables.

To return to the proof of the first assertion in Proposition 4.1, let us suppose that L is developed as a power series in u. Specifically, let (x^i, u^i) be an adapted coordinate system and suppose

$$+ (1/4!)A_{ijkl}u^{i}u^{j}u^{k}u^{l} + \cdots, \qquad (4.2)$$

$$L = A + A_i u^i + (1/2!) A_{ij} u^i u^j + (1/3!) A_{ijk} u^i u^j u^k$$

where $A, A_i, A_{ij}, A_{ijk}, A_{ijkl}$, etc., are functions of x^i only. Then one may show from (3.6) that the first three terms in f^j are given explicitly by

$$f^{j} = \frac{\partial A}{\partial x^{a}} (A^{-1})^{aj} + \left[\left(\frac{\partial A_{k}}{\partial x^{a}} - \frac{\partial A_{a}}{\partial x^{k}} \right) (A^{-1})^{aj} - \frac{\partial A}{\partial x^{a}} A_{ikl} (A^{-1})^{ij} (A^{-1})^{al} \right] u^{k} \\ + \frac{1}{2} \left[-(A^{-1})^{aj} \left(\frac{\partial A_{la}}{\partial x^{k}} + \frac{\partial A_{ka}}{\partial x^{l}} - \frac{\partial A_{kl}}{\partial x^{a}} \right) + \left(\left(\frac{\partial A_{a}}{\partial x^{l}} - \frac{\partial A_{l}}{\partial x^{a}} \right) A_{ikm} + \left(\frac{\partial A_{a}}{\partial x^{k}} - \frac{\partial A_{k}}{\partial x^{a}} \right) A_{ilm} \right) (A^{-1})^{ij} (A^{-1})^{am} \\ + \frac{\partial A}{\partial x^{a}} ((A_{ikm}A_{lnp} + A_{ilm}A_{knp}) (A^{-1})^{ij} (A^{-1})^{mp} (A^{-1})^{an} - A_{iklm} (A^{-1})^{am} (A^{-1})^{ij}) \right] u^{k} u^{l} + \cdots .$$

$$(4.3)$$

The terms in f_j of degree 3 and higher can be determined, in principle, in a similar manner, the term of degree *n* depending on the coefficients corresponding to terms in *L* of degree up to and including n + 2.

There are some other points worth noting about (4.3). First, if in (4.2) A is constant and the term $A_i u^i$ corresponds to a closed one-form on M, that is, is a pure gauge term, the zeroth- and first-order terms in the expansion (4.3) are 0. Conversely, suppose that one is given a SOEF and that in the expansion of the corresponding f^j 's, the zeroth- and first-order terms are 0; then if there is a regular Lagrangian L such that Γ is its Euler-Lagrange vector field, in the expansion (4.2), A must be constant and $A_i u^i$ must be a gauge term. In other words, in looking for such a Lagrangian, one may assume without loss of generality that its lowest-order terms are quadratic in u.

The second point about (4.3) concerns the spray associated to SOEF Γ which is the Euler-Lagrange vector field of a regular Lagrangian L. Suppose that in an adapted coordinate system (x^i, u^i) the Christoffel symbols of this spray are Γ_{kl}^j so that (4.3) gives

$$\Gamma_{kl}^{j} = -\frac{1}{2} \bigg[(A^{-1})^{aj} (A_{la,k} + A_{ka,l} - A_{kl,a}) + ((A_{l,a} - A_{a,l})A_{ikm} + (A_{k,a} - A_{a,k})A_{ilm}) (A^{-1})^{ij} (A^{-1})^{am} \\ + \frac{\partial A}{\partial x^{a}} ((A_{ikm}A_{lnp} + A_{ilm}A_{knp}) (A^{-1})^{ij} (A^{-1})^{mp} (A^{-1})^{an} - A_{iklm} (A^{-1})^{am} (A^{-1})^{ij}) \bigg].$$

$$(4.4)$$

Thus if we interpret A_{ij} as a Riemannian or pseudo-Riemannian metric on M (A_{ij} must be nonsingular if L is to be regular on the zero section of TM), we see that the spray associated to Γ is the Levi connection of A_{ij} if either L is quadratic in u or is trivially equivalent to a Lagrangian whose lowest-order terms are quadratic (and possibly in other cases, too). By performing maniupulations similar to these in Riemannian geometry,⁶ one can derive from (4.4)

$$A_{bk,l} = A_{bj} \Gamma^{j}_{kl} + A_{kj} \Gamma^{j}_{lb} + B_{bkl},$$
(4.5)

where

$$B_{bkl} = \frac{1}{2} \left[(A_{a,k} - A_{k,a}) A_{blm} + 2(A_{a,l} - A_{l,a}) A_{bkm} + (A_{a,b} - A_{b,a}) A_{klm} \right] (A^{-1})^{am} \\ + \frac{1}{2} A_{,a} (A_{blm} A_{knp} + 2A_{bkm} A_{lnp} + A_{klm} A_{bnp}) (A^{-1})^{mp} (A^{-1})^{an} - A_{,a} A_{bklm} (A^{-1})^{am}.$$

$$(4.6)$$

Equation (4.5) can also be written in the form

$$A_{bk;l} = B_{bkl}, \tag{4.7}$$

where the semicolon denotes the covariant derivative with respect to the connection whose components are Γ_{kl}^{j} . Differentiating (4.7) covariantly again and invoking Ricci's identities which measure the extent to which iterated covariant derivatives fail to commute,⁶ one obtains the following integrability conditions:

$$A_{bs}R_{rlk}^{s} + A_{ks}R_{rlb}^{s} = B_{bkr,l} - B_{bk;lr}, \qquad (4.8)$$

where R_{rlk}^{s} are the components of the curvature tensor determined by the Γ_{kl}^{j} 's.

It follows amongst other things from (4.8) that if the right-hand side is 0 (again if L is quadratic or its lowest-order terms are quadratic, for example), that the curvature of the connection associated to Γ has a Ricci tensor which is symmetric.

V. THE INVERSE PROBLEM FOR SPRAYS

I shall consider in this section what is in effect a special case of the inverse problem, but one which is of considerable interest in its own right, namely, when the given SOEF is a spray. We have already seen in Sec. II, that the f^{j} 's being (not necessarily homogeneous) quadratic is an invariant condition, whereas the f^{j} 's being independent of u^{i} , for example, is not. The special status of quadratic systems is further reinforced in Lagrangian theory by Proposition 4.1. The reason for considering sprays first, as opposed to more general quadratic systems, is that sprays are the most natural kinds of SOEF's from the geometric standpoint. In particular, given a spray on M, one may ask if it is the geodesic flow of the Levi-Civita connection of some Riemannian or pseudo-Riemannian metric on M. This is indeed a special case of the inverse problem because one is asking for a quadratic

Lagrangian for which Λ is the associated Euler-Lagrange vector field.

Suppose then that Λ is a spray corresponding to the second-order system

$$\ddot{\mathbf{x}}^{j} = -\Gamma^{j}_{ab}\dot{\mathbf{x}}^{a}\dot{\mathbf{x}}^{b}.$$
(5.1)

If we look for a Lagrangian analytic in u, by a previous remark we may assume that A = 0 and $A_i = 0$ in (4.2). Inserting the remaining terms in (4.2) into (3.6) with the f^{j} 's given by the right-hand side of (5.1), one finds the following conditions for n = 2,3,...:

$$nA_{k(i_{1}\cdots i_{n-1},i_{n})} - A_{i_{1}\cdots i_{n},k} = n(n-1)\Gamma^{j}_{(i_{1},i_{2}}A_{i_{3}\cdots i_{n})jk}.$$
(5.2)

Thus, for a spray and indeed essentially in that case alone, (3.6) decouples by degree. In other words, (5.2) are precisely the conditions that there should exist a Lagrangian, which is a homogeneous polynomial of degree n in u ($n \ge 2$), and, for different values of n, these conditions are independent of each other.

From (5.2) it follows by symmetrization that

$$(n-1)A_{(i_1\cdots i_n,k)} = n(n-1)\Gamma^j_{(i_1i_2}A_{i_3\cdots i_nk)}j, \qquad (5.3)$$

which in turn, by choosing k as a distinguished index in the symmetrization in the left-hand side of (5.3), implies that

$$[1/(n+1)]A_{i_{1}\cdots i_{n},k} + [n/(n+1)]A_{k(i_{1}\cdots i_{n-1},i_{n})}$$

= $n\Gamma^{j}_{(i_{1}i_{2}}A_{i_{3}\cdots i_{n}k)}j.$ (5.4)

From (5.2) and (5.4) we easily obtain

$$A_{i_{1}\cdots i_{n},k} = \binom{n+1}{2} \Gamma^{j}_{(i_{1}i_{2}}A_{i_{3}\cdots i_{n}k)j} - \binom{n}{2} \Gamma^{j}_{(i_{1}i_{2}}A_{i_{3}\cdots i_{n})kj}.$$
(5.5)

Equation (5.5) can be rewritten as

$$i_{i_{1}\cdots i_{n},k} = \Gamma^{j}_{i_{1}k}A_{ji_{2}\cdots i_{n}} + \Gamma^{j}_{i_{2}k}A_{i_{1}ji_{3}\cdots i_{n}} + \cdots + \Gamma^{j}_{i_{n}k}A_{i_{1}\cdots i_{n-1}j},$$
(5.6)

which in turn is equivalent to

A

$$A_{i_1\cdots i_n;k} = 0, \tag{5.7}$$

where the semicolon denotes the covariant derivative with respect to the connection defined by the Γ_{jk}^{i} 's. Conversely, since (5.7) is equivalent to (5.5) and also implies (5.3), it is clear that (5.7) is equivalent to (5.2). I summarize the preceding discussion in the following theorem.

Theorem 5.1: Given a spray Λ on M, a function \overline{L} on TManalytic in u is a regular Lagrangian for Λ iff \overline{L} satisfies the following conditions: \overline{L} is trivially equivalent to a Lagrangian L which has an expansion of the form (4.2) in which A = 0 and $A_i = 0$, A_{ij} is nonsingular and each nonzero term in (4.2) corresponds to a symmetric, covariant tensor field on M parallel with respect to the parallel transport defined by the connection associated to Λ .

A number of observations about Theorem 5.1 are in order. First of all, given a spray Λ , if there is to be a regular Lagrangian whose Euler-Lagrange field is Λ , (5.7) must be satisfied for some nonsingular A_{ij} in the case n = 2. In other words, Λ must be the geodesic spray of the metric A_{ij} . Suppose that such a metric A_{ij} exists; then $L = \frac{1}{2}A_{ij}u^i u^j$ is a regular Lagrangian for Λ and we may enquire whether an alternative Lagrangian exists. In fact, such a Lagrangian will exist if either one can find another metric compatible with the connection defined by Λ or else there is symmetric tensor $A_{i_1 \cdots i_n}$ of valence *n* parallel with respect to parallel transport defined by that connection. Condition (5.7) is very restrictive; despite this, when a metric A_{ij} exists, all tensor powers of A_{ij} satisfy (5.7). Thus $A_{(ij}A_{kl)}$ satisfies (5.7) with n = 4 because

$$[A_{(ij}A_{kl})]_{;m} = A_{(ij}[A_{kl})_{;m}] + A_{(kl}[A_{ij})_{;m}]$$

= $2A_{(ij}A_{kl})_{;m}$
= 0,

since $A_{kl;m} = 0$. So for a spray Λ compatible with a metric A_{ij} , there are always many (nontrivial) alternative Lagrangians obtained by adding to the Lagrangian L functions on TM corresponding to symmetric tensor powers of the metric.

Another somewhat different way in which solutions to (5.7) arise is in the context of locally symmetric spaces.⁷ A manifold M with a metric A_{ij} is said to be locally symmetric if the curvature tensor R^{i}_{jkl} is parallel with respect to the parallel transport of the Levi-Civita connection defined by A_{ij} , in other words, if

$$R_{jkl;m}^{i} = 0. (5.8)$$

Equation (5.8) implies of course that

$$R_{jkl;m}^{j} = 0,$$
 (5.9)

that is, the Ricci tensor $R_{kl} = R_{jkl}^{j}$ is parallel. Since we are dealing with a manifold with a metric, the Ricci tensor is also symmetric and thus satisfies (5.7) with n = 2. Of course it may be that R_{kl} is a constant multiple of A_{ij} in which case (M, A_{ij}) is said to be an Einstein manifold. Furthermore, there is no reason to suppose in general that R_{ij} is nondegenerate and so may not serve as an alternative metric compatible with the connection ∇ .

Again suppose that Λ is a spray on M which is given in an adapted coordinate system (x^i, u^i) by

$$\Lambda = u^{i} \frac{\partial}{\partial x^{i}} - \Gamma^{i}_{jk} u^{j} u^{k} \frac{\partial}{\partial u^{i}}.$$
 (5.10)

Then one may readily show that the necessary and sufficient condition that a polynomial homogeneous of degree n in u, $B_{i,\dots,i}u^{i_1}\cdots u^{i_n}$ say, is a first integral of Λ is

$$B_{(i_1\cdots i_n;i_{n+1})} = 0. (5.11)$$

A symmetric, covariant tensor field $B_{i_1 \cdots i_n}$ which satisfies (5.11) is said to be a Killing tensor field of the connection ∇ corresponding to Λ .^{12,13} [The notion of a contravariant Killing vector on a manifold with metric is extremely well known. The Killing condition on the corresponding covariant object is precisely (5.11) with n = 1, so that (5.11) is the natural generalization of this condition.] Equation (5.11) is of course a weaker condition than (5.7). In particular, when looking for Lagrangians L corresponding to Λ , each term in the expansion (4.2) of degree 2 or more is a first integral of Λ . This shows how restrictive (5.7) is: to the

knowledge of the author, there are no known examples of manifolds with a metric that have a homogeneous first integral of degree n in u, which is notrivial when n > 2. [A Killing tensor of degree n is said to be trivial if it consists of a (finite) sum of symmetrized products of lower-order Killing tensors.] For n = 2, however, a limited number of examples are known such as the Kerr metric¹² and the Ricci tensor in locally symmetric spaces as described above.

The conditions given in Theorem 5.1 for a spray to admit a Lagrangian are easily extended to systems in which the f^{j} 's are quadratic in u^{i} , though the geometric interpretation becomes somewhat more contrived. Indeed, instead of (5.1) consider the following system:

$$\ddot{x}^{j} = -(\Gamma^{j} + \Gamma^{j}_{a} u^{a} + \Gamma^{j}_{ab} u^{a} u^{b}), \qquad (5.12)$$

where Γ^{j} , Γ^{j}_{a} , and Γ^{j}_{ab} are functions of x^{i} only. Then if one seeks a Lagrangian of the form (4.2), one finds by modifying (5.2)–(5.7) that the following conditions are necessary and sufficient for the existence of such a Lagrangian:

$$A_{,i} = -A_{ij}\Gamma^{j}, \tag{5.13}$$

$$A_{k,i} - A_{i,k} = -A_{ij} \Gamma_k^j,$$
 (5.14)

$$A_{i_{1}\cdots i_{n,k}} = \frac{1}{n-1} \Gamma^{j} A_{jki_{1}\cdots i_{n}} + \frac{n(n+1)}{2(n-1)} \Gamma^{j}_{(i_{1}} A_{i_{2}\cdots i_{n}k)j} - (n/2) \Gamma^{j}_{(i_{1}} A_{i_{2}\cdots i_{n})kj} \quad (n \ge 2).$$
(5.15)

ACKNOWLEDGMENT

The author gratefully acknowledges the support of the Science and Engineering Research Council of the United Kingdom in the form of a Research Fellowship.

- ¹M. Crampin, J. Phys. A: Math. Gen. 14, 2567 (1981).
- ²W. Sarlet, J. Phys. A: Math. Gen. 15, 1503 (1982).
- ³M. Crampin, J. Phys. A: Math. Gen. 16, 3755 (1983).
- ⁴K. Yano and S. Ishihara, *Tangent and Cotangent Bundles* (Dekker, New York, 1973).
- ⁵M. Crampin and G. Thompson, Proc. Cambridge Philos. Soc. 98, 61 (1985).
- ⁶M. Spivak, A Comprehensive Introduction to Differential Geometry (Publish or Perish, Waltham, MA, 1970), Vol. 2.
- ⁷J. A. Wolf, Spaces of Constant Curvature (McGraw-Hill, New York, 1967).
- ⁸W. Sarlet, F. M. Mahomed, and P. G. L. Leach, J. Phys. A: Math. Gen. 20, 277 (1987).
- ⁹J. Klein, Ann. Inst. Fourier 12, 1 (1962).
- ¹⁰C. Godbillon, Geométrie differentielle et méchanique analytique (Herman, Paris, 1969).
- ¹¹M. Henneaux and L. C. Shepley, J. Math. Phys. 23, 2101 (1982).
- ¹²N. M. J. Woodhouse, Commun. Math. Phys. 44, 9 (1975).
- ¹³G. Thompson, J. Math. Phys. 27, 2693 (1986).

The inverse scattering problem for the soft ellipsoid

George Dassios^{a)}

Department of Mathematics, University of Patras, Patras 26110, Greece

(Received 17 March 1987; accepted for publication 24 June 1987)

A soft triaxial ellipsoid, of unknown semiaxes and orientation, is excited into secondary radiation by a plane acoustic wave of a fixed low frequency. It is proved that one measurement of the leading low-frequency coefficient and exactly six measurements of the second low-frequency coefficient of the real part of the forward or the backward scattering amplitude are enough to specify completely both the semiaxes, as well as the orientation of the ellipsoid. Therefore, only the first two low-frequency coefficients of the real part of the scattering amplitude are needed in order to solve the inverse scattering problem for the soft ellipsoid. For the case of spheroids, the number of measurements is restricted to one for the first and three for the second coefficient. Finally, the sphere is specified by a single measurement of the leading coefficient. The special cases where the orientation or the semiaxes are known are also discussed.

I. INTRODUCTION

The problem of scattering of a plane acoustic wave of fixed frequency by a soft body has a history of approximately one century. Its complete mathematical analysis can be found in Ref. 1, where also the low-frequency approximation of the wave problem has been reduced to certain potential problems through Neumann series expansions. Sleeman² has applied Kleinman's method¹ to the case of an ellipsoid and he evaluated the low-frequency coefficients up to the second order in terms of elliptic functions. In Ref. 3, a different technique has been used to evaluate the four leading coefficients of the scattering amplitude, using also Lamé products but in their algebraic form, which simplifies the final results considerably. In fact, the scattering amplitude is given explicitly in Ref. 3, in terms of the three semiaxes of the ellipsoid, the directional cosines of incidence, the directional cosines of observation, and four well known elliptic integrals connected through two algebraic relations. The actual geometry of the scatterer is implicit in the values of these elliptic integrals, while the directions of incidence and orientation, which are at our disposal to control, specify the relative orientation of the ellipsoid. It is in fact the simplicity of the expression of the fourth low-frequency coefficient of the scattering amplitude that allows for an exact solution of the corresponding inverse problem.

In order to solve completely the inverse problem, all we need to specify are the three semiaxes as well as the three Euler angles that fix the position of the principal axes of the ellipsoid. Therefore, we need to excite ("see") the ellipsoid from six different angles and for each one of them to measure the forward or the backward scattering amplitude, which conveys all the information about the scatterer that can be "seen" from the particular direction of incidence. It turns out that these six measurements give rise to a highly nonlinear system involving four elliptic integrals. Therefore, we need one more piece of information to effectively solve the inverse problem. The choice of the six directions of incidence should be such as to make an optimum use of the symmetry inherited in the ellipsoid. We note that our technique depends crucially upon the inversion symmetry, as well as on the particular form that the scattering theorem assumes for scatterers having inversion symmetry.⁴

We remark that as long as we are in the low-frequency realm no measurement of any frequency is necessary.⁵ Also, only the expression of the zeroth- and fourth-order low-frequency coefficients of the normalized scattering amplitude,³ or equivalently, the first two nonvanishing terms of its real part are needed for the complete solution of the inverse problem.

As it is well known,⁵ the inverse scattering problem is intrinsically nonlinear. In the case of the ellipsoid the nonlinearities enter the problem via the elliptic integrals, the second and fourth powers of the values of the semiaxes, and the quadratic expressions of the components of the directions of incidence and observation. To a large extent, the worst part of these nonlinearities has been eliminated by algebraic manipulations and the use of the seventh measurement. Nevertheless we cannot avoid dealing at least with one standard elliptic integral for the determination of the semiaxes. The effect of nonlinearity in finding the orientation is reflected upon the nonlinear expressions providing the three Euler angles.

Angell and Kleinman have reduced the inverse scattering problem for an ellipsoid to a constrained optimization problem.⁶ They have considered all the physically important boundary conditions in acoustics as well as in electromagnetism except the scalar Dirichlet problem explored in the present paper by a completely different method.

After giving all the necessary information from the solution of the direct problem in Sec. II, we proceed to Sec. III where the solution of the inverse scattering problem is given in its generality. The special cases of known orientation of geometries with higher symmetry are discussed in Sec. IV.

II. THE DIRECT PROBLEM

$$x_1^2/a_1^2 + x_2^2/a_2^2 + x_3^2/a_3^2 = 1, \qquad (1)$$

Let S be the ellipsoid

where $0 < a_3 < a_2 < a_1 < +\infty$ are the three semiaxes and $\hat{\mathbf{x}}_1$, $\hat{\mathbf{x}}_2$, $\hat{\mathbf{x}}_3$ are denoting the unit vectors of the orthogonal Cartesian system that coincides with the principal axes of the given ellipsoid. Suppressing the harmonic time dependence $\exp\{-i\omega t\}$, the direct scattering problem for (1) consists in finding a solution u of the Helmholtz equation

$$\Delta u + k^2 u = 0, \qquad (2)$$

in the region V, exterior to S, which assumes the values

$$u=-e^{i\mathbf{k}\cdot\mathbf{r}},\qquad(3)$$

on S and the asymptotic form

$$r\left(\frac{\partial u}{\partial r} - iku\right) = o(1) \tag{4}$$

as $r \to +\infty$. The vector k determines the direction of propagation of the incident wave $\exp{\{i\mathbf{k}\cdot\mathbf{r}\}}$, while its magnitude k is the wave number. Far away from the scatterer, the scattered field u has the asymptotic form

$$u(\mathbf{r}) = g(\hat{\mathbf{r}}, \hat{\mathbf{k}}) h(kr) + O(1/r^2) , \qquad (5)$$

where $h(x) = e^{ix} / ix$ and $g(\hat{\mathbf{r}}, \hat{\mathbf{k}})$ is the normalized (dimensionless) scattering amplitude which describes the response of the scatterer in the direction $\hat{\mathbf{r}}$ to plane wave excitation of direction $\hat{\mathbf{k}}$. Due to the symmetry of the ellipsoid the scattering amplitude satisfies the following relations⁴:

$$g(\hat{\mathbf{r}},\hat{\mathbf{k}}) = g(\hat{\mathbf{k}},\hat{\mathbf{r}}) \tag{6}$$

and

$$-\operatorname{Re} g(\hat{\mathbf{k}}, \hat{\mathbf{k}}) = \frac{1}{4\pi} \int_{|\hat{\mathbf{p}}| = 1} |g(\hat{\mathbf{p}}, \hat{\mathbf{k}})|^2 \, d\Omega(\hat{\mathbf{p}}) \,. \tag{7}$$

As a consequence of (6) and (7) if the amplitude g has the low-frequency expansion³

$$g(\hat{\mathbf{r}},\hat{\mathbf{k}}) = ikA_1(\hat{\mathbf{r}},\hat{\mathbf{k}}) + k^2A_2(\hat{\mathbf{r}},\hat{\mathbf{k}}) + ik^3A_3(\hat{\mathbf{r}},\hat{\mathbf{k}}) + k^4A_4(\hat{\mathbf{r}},\hat{\mathbf{k}}) + O(k^5)$$
(8)

as $k \rightarrow 0 +$, and if the leading two coefficients of the Im $g(\hat{\mathbf{r}}, \hat{\mathbf{k}})$ are known, then the leading two coefficients of Re $g(\hat{\mathbf{r}}, \hat{\mathbf{k}})$ can be found from⁷

$$A_2(\hat{\mathbf{r}},\hat{\mathbf{k}}) = -\frac{1}{4\pi} \int_{|\hat{\mathbf{p}}|=1} A_1(\hat{\mathbf{p}},\hat{\mathbf{r}}) A_1(\hat{\mathbf{p}},\hat{\mathbf{r}}) d\Omega(\hat{\mathbf{p}})$$
(9)

and

$$\begin{aligned} A_4(\hat{\mathbf{r}}, \hat{\mathbf{k}}) \\ &= -\frac{1}{4\pi} \int_{|\hat{\mathbf{p}}| = 1} \left[A_1(\hat{\mathbf{p}}, \hat{\mathbf{r}}) A_3(\hat{\mathbf{p}}, \hat{\mathbf{k}}) \right. \\ &+ A_2(\hat{\mathbf{p}}, \hat{\mathbf{r}}) A_2(\hat{\mathbf{p}}, \hat{\mathbf{k}}) + A_3(\hat{\mathbf{p}}, \hat{\mathbf{r}}) A_1(\hat{\mathbf{p}}, \hat{\mathbf{k}}) \left] d\Omega(\hat{\mathbf{p}}) . \end{aligned}$$

(10)

The actual coefficients of $g(\hat{\mathbf{r}}, \hat{\mathbf{k}})$ for the case of the soft ellipsoid are given by³

$$A_1(\hat{\mathbf{r}}, \hat{\mathbf{k}}) = -1/I, \qquad (11)$$

$$A_2(\hat{\mathbf{r}}, \hat{\mathbf{k}}) = -1/I^2, \qquad (12)$$

$$A_{3}(\hat{\mathbf{r}}, \hat{\mathbf{k}}) = 1/I^{3} - (1/3I)(a_{1}^{2} + a_{2}^{2} + a_{3}^{2}) + (1/3I^{2})(a_{1}^{4} I_{1} + a_{2}^{4}I_{2} + a_{3}^{4}I_{3}) - \frac{1}{3} \left(\frac{i_{1}o_{1}}{I_{1}} + \frac{i_{2}o_{2}}{I_{2}} + \frac{i_{3}o_{3}}{I_{3}} \right) + (1/6I)(i_{1}^{2} a_{1}^{2} + i_{2}^{2} a_{2}^{2} + i_{3}^{2} a_{3}^{2})$$

$$+ (1/6I) (o_1^2 a_1^2 + o_2^2 a_2^2 + o_3^2 a_3^2), \quad (13)$$

$$A_4(\hat{\mathbf{r}}, \hat{\mathbf{k}}) = 1/I^4 - (5/9I^2) (a_1^2 + a_2^2 + a_3^2) + (2/3I^3) (a_1^4I_1 + a_2^4I_2 + a_3^4I_3) + (1/6I^2) (i_1^2 a_1^2 + i_2^2 a_2^2 + i_3^2 a_3^2) + (1/6I^2) (o_1^2 a_1^2 + o_2^2 a_2^2 + o_3^2 a_3^2), \quad (14)$$

where $\hat{\mathbf{k}} = (i_1, i_2, i_3), \, \hat{\mathbf{r}} = (o_1, o_2, o_3),$

$$I = \frac{1}{2} \int_{0}^{+\infty} \frac{dx}{\sqrt{x + a_{1}^{2}} \sqrt{x + a_{2}^{2}} \sqrt{x + a_{3}^{2}}},$$
 (15)
$$I_{n} = \frac{1}{2} \int_{0}^{+\infty} \frac{dx}{(x + a_{n}^{2})\sqrt{x + a_{1}^{2}} \sqrt{x + a_{2}^{2}} \sqrt{x + a_{3}^{2}}},$$

$$n = 1,2,3.$$
 (16)

The four elliptic integrals I, I_1, I_2, I_3 are related via the formulas

$$a_1 a_2 a_3 (I_1 + I_2 + I_3) = 1,$$

$$a_1^2 I_1 + a_2^2 I_2 + a_3^2 I_3 = I.$$
(17)

In particular,

$$= k g(\mathbf{k}, \mathbf{k})$$

= $k^2 / I^2 + k^4 [T - (1/3I^2)(i_1^2 a_1^2 + i_2^2 a_2^2 + i_3^2 a_3^2)]$
+ $O(k^6), \quad k \to 0 + ,$ (18)

where

$$T = -1/I^{4} + (5/9I^{2})(a_{1}^{2} + a_{2}^{2} + a_{3}^{2}) - (2/3I^{3})(a_{1}^{4}I_{1} + a_{2}^{4}I_{2} + a_{3}^{4}I_{3}).$$
(19)

The parameter T depends only on the semiaxes of the ellipsoid and it is independent of its orientation. The reason why we choose to work with the real, instead of the imaginary, part of g is due to the fact that the coefficient A_4 has only one nonisotropic term connecting the directional cosines of $\hat{\mathbf{k}}$ to the corresponding semiaxes, while the coefficient A_3 involves two such terms connecting $\hat{\mathbf{k}}$ to the elliptic integrals I_n , n = 1,2,3. Then the only really "bad" part in A_4 is the expression T which is evaluated by introducing one more measurement.

Furthermore, expression (14) and hence also (18) provide the same value independently on whether one measures the forward or the backward scattering amplitude.

III. THE INVERSE PROBLEM

The question we try to answer in this section is the following. Suppose we have a way to measure the coefficients $A_2(\hat{\mathbf{k}}, \hat{\mathbf{k}})$ and $A_4(\hat{\mathbf{k}}, \hat{\mathbf{k}})$ of $g(\hat{\mathbf{k}}, \hat{\mathbf{k}})$, how many measurements (corresponding to different $\hat{\mathbf{k}}$'s) are needed in order to effectively specify the ellipsoid (1) as well as its orientation?

As is shown in the sequel we need to know A_2 for a fixed \hat{k} and A_4 for six different directions of \hat{k} .

In fact, let

$$m_0 = -A_2(\hat{\mathbf{k}}, \hat{\mathbf{k}}) \tag{20}$$

and

$$m_j = -A_4(\hat{\mathbf{k}}_j, \hat{\mathbf{k}}_j), \quad j = 1, 2, 3, 4, 5, 6,$$
 (21)

be the values obtained by the corresponding measurements.

Let $\hat{\mathbf{k}}'$ be an arbitrary direction,

$$\hat{\mathbf{k}}_{j}' = \hat{\mathbf{x}}_{j}', \quad j = 1, 2, 3,$$
 (22)

and

$$\hat{\mathbf{k}}_{4}' = (1/\sqrt{2})(\hat{\mathbf{x}}_{1}' + \hat{\mathbf{x}}_{2}')
\hat{\mathbf{k}}_{5}' = (1/\sqrt{2})(\hat{\mathbf{x}}_{2}' + \hat{\mathbf{x}}_{3}'),$$

$$\hat{\mathbf{k}}_{6}' = (1/\sqrt{2})(\hat{\mathbf{x}}_{3}' + \hat{\mathbf{x}}_{1}'),$$
(23)

where $\{\hat{\mathbf{x}}'_1, \hat{\mathbf{x}}'_2, \hat{\mathbf{x}}'_3\}$ form an orthonormal base of an arbitrary chosen Cartesian system whose origin coincides with the centroid of the ellipsoid.

Let P be the orthogonal matrix that transforms (by rotation) the arbitrarily chosen primed system x'_1, x'_2, x'_3 to the unprimed system x_1, x_2, x_3 determined by the principal directions of the ellipsoid, i.e.,

$$\mathbf{r} = \mathbf{P}\mathbf{r}' \,. \tag{24}$$

Of course, the matrix **P** is not known, since the orientation of the ellipsoid is not known, but its existence and its uniqueness within the orthogonal group is ensured by the orthogonality of both the chosen unprimed system and the system of principal axes of the ellipsoid. The knowledge of the matrix **P** will provide the exact orientation of the ellipsoid. The elements P_{ij} of **P** are expressed via the three Euler angles φ , θ , y (Ref. 8) as follows:

$$P_{11} = \cos \varphi \cos y - \cos \theta \sin \varphi \sin y ,$$

$$P_{12} = \sin \varphi \cos y + \cos \theta \cos \varphi \sin y ,$$

$$P_{13} = \sin \theta \sin y ;$$
(25)

$$P_{21} = -\cos\varphi \sin y - \cos\theta \sin\varphi \cos y,$$

$$P_{22} = -\sin\varphi \sin y + \cos\theta \cos\varphi \cos y,$$
 (26)

$$P_{23} = \sin\theta \cos y;$$

$$P_{31} = \sin \theta \sin \varphi,$$

$$P_{32} = -\sin \theta \cos \varphi,$$

$$P_{33} = \cos \theta.$$

(27)

Therefore, the orientation of the ellipsoid is known whenever the three Euler angles, φ , θ , y are known.

Since the form (18) for the scattering amplitude is referred to as the principal axes system it follows that (18) holds true after the transformation **P** has been applied to the directions of incidence $\hat{\mathbf{k}}$, i.e.,

$$\hat{\mathbf{k}}_{j} = \mathbf{P}\hat{\mathbf{k}}_{j}', \quad j = 1, 2, 3, 4, 5, 6.$$
 (28)

Consequently,

 $m_0 = 1/I^2$ (29)

and

$$m_j = T - (1/3I^2)(\hat{\mathbf{k}}_j)^t \mathbf{A}\hat{\mathbf{k}}_j$$
(30)

for j = 1, 2, ..., 6, where

$$\mathbf{A} = \begin{bmatrix} a_1^2 & 0 & 0\\ 0 & a_2^2 & 0\\ 0 & 0 & a_3^2 \end{bmatrix}.$$
 (31)

In view of (28) and (29), Eqs. (30) take the form

$$3(T-m_j)/m_0 = (\hat{\mathbf{k}}_j')' \mathbf{P}' \mathbf{A} \mathbf{P}(\hat{\mathbf{k}}_j'), \qquad (32)$$

for j = 1, 2, ..., 6.

Substituting (22) and (23) into (32) we obtain the following nonlinear system of equations for the unknown quantities φ , θ , y, a_1, a_2, a_3 :

$$P_{11}^{2} a_{1}^{2} + P_{21}^{2} a_{2}^{2} + P_{31}^{2} a_{3}^{2} = 3(T - m_{1})/m_{0}, \qquad (33)$$

$$P_{12}^{2} a_{1}^{2} + P_{22}^{2} a_{2}^{2} + P_{32}^{2} a_{3}^{2} = 3(T - m_{2})/m_{0}, \qquad (34)$$

$$P_{13}^{2} a_{1}^{2} + P_{23}^{2} a_{2}^{2} + P_{33}^{2} a_{3}^{2} = 3(T - m_{3})/m_{0}, \qquad (35)$$
$$(P_{11} + P_{12})^{2} a_{1}^{2} + (P_{21} + P_{22})^{2} a_{2}^{2}$$

$$+ (P_{31} + P_{32})^2 a_3^2 = 6(T - m_4)/m_0, \qquad (36)$$

$$(P_{12} + P_{13})^2 a_1^2 + (P_{22} + P_{23})^2 a_2^2 + (P_{32} + P_{33})^2 a_3^2 = 6(T - m_5)/m_0, \qquad (37)$$

$$(P_{11} + P_{13})^2 a_1^2 + (P_{21} + P_{23})^2 a_2^2 + (P_{31} + P_{33})^2 a_3^2 = 6(T - m_6)/m_0.$$
(38)

By virtue of (33)-(35), Eqs. (36)-(38) can be written as $P_{12}P_{12}a_1^2 + P_{22}P_{22}a_2^2 + P_{22}P_{22}a_2^2$

$$= 3(m_1 + m_2 - 2m_4)/2m_0, \qquad (39)$$

$$P_{12}P_{13}a_1^2 + P_{22}P_{23}a_2^2 + P_{32}P_{33}a_3^2$$

= $3(m_2 + m_3 - 2m_5)/2m_0$, (40)

$$P_{11}P_{13}a_1^2 + P_{21}P_{23}a_2^2 + P_{31}P_{33}a_3^2$$

= 3(m_3 + m_1 - 2m_6)/2m_0. (41)

Equations (33), (34), (35), (39), (40), and (41) can be combined in the following matrix form:

$$\mathbf{P}^{t}\mathbf{A}\mathbf{P} = (3T/m_{0})\mathbf{I} + \mathbf{M}, \qquad (42)$$

where M is the real symmetric matrix of measurements of the A_4 coefficient given by

$$M = \frac{3}{2m_0} \begin{bmatrix} -2m_1 & m_1 + m_2 - 2m_4 & m_3 + m_1 - 2m_6 \\ m_1 + m_2 - 2m_4 & -2m_2 & m_2 + m_3 - 2m_5 \\ m_3 + m_1 - 2m_6 & m_2 + m_3 - 2m_5 & -2m_3 \end{bmatrix}.$$
 (43)

The matrix M has the real eigenvalues $\lambda_1, \lambda_2, \lambda_3$ and its eigenvectors form an orthonormal basis in \mathbb{R}^3 .

Equation (42) is also written as

$$\mathbf{A} - (3T/m_0)\mathbf{I} = \mathbf{P}\mathbf{M}\mathbf{P}^t.$$

Since the left-hand side of (44) is a diagonal matrix, M is a real symmetric matrix, and P is orthogonal, it follows that the eigenvalues of M are given by

$$\lambda_k = a_k^2 - 3T/m_0, \quad k = 1, 2, 3, \tag{45}$$

George Dassios 2860

while the columns of \mathbf{P}^t are the corresponding orthonormal eigenvectors. Therefore, the squares of the three semiaxes are given by

$$a_k^2 = \lambda_k + 3T/m_0, \quad k = 1, 2, 3.$$
 (46)

The value of the expression T can be calculated from (29), which in view of (15) is written as

$$\int_{0}^{+\infty} \frac{dx}{\sqrt{x + \lambda_1 + 3T/m_0} \sqrt{x + \lambda_2 + 3T/m_0} \sqrt{x + \lambda_3 + 3T/m_0}} = \frac{2}{\sqrt{m_0}}.$$
(47)

The transformation

$$x = (\lambda_1 - \lambda_3)/t^2 - (\lambda_1 + 3T/m_0)$$
(48)

brings the elliptic integral in (47) to its canonical form

$$F(\varphi_{0},\alpha_{0}) = \int_{0}^{\sin\varphi_{0}} \frac{dt}{\sqrt{1-t^{2}}\sqrt{1-t^{2}\sin^{2}\alpha_{0}}} = \sqrt{\frac{\lambda_{1}-\lambda_{3}}{m_{0}}},$$
(49)

where $F(\varphi_0,\alpha_0)$ denotes the incomplete elliptic integral of the first kind,⁹ with amplitude

$$\varphi_0 = \sin^{-1} \sqrt{(\lambda_1 - \lambda_3)/(\lambda_1 + 3T/m_0)}$$
(50)

and modular angle

$$\alpha_0 = \sin^{-1} \sqrt{(\lambda_1 - \lambda_2)/(\lambda_1 - \lambda_3)} .$$
 (51)

From (50) we obtain the value of T as

$$T = [m_0(\lambda_1 - \lambda_3)/3] \cot^2 \varphi_0 - m_0 \lambda_3/3.$$
 (52)

Putting everything together the steps we need to follow in order to evaluate the semiaxes and the orientation of the ellipsoid are the following: (a) we measure, in the forward direction, A_2 for any \hat{k} and A_4 for the six directions given by (22) and (23); (b) we write down the matrix of measurements **M** as it is given by (43); (c) we evaluate the eigenvalues $\lambda_1, \lambda_2, \lambda_3$ (this involves the solution of a cubic equation) and the eigenvectors (P_{11}, P_{12}, P_{13}) , (P_{21}, P_{22}, P_{23}) , and (P_{31}, P_{32}, P_{33}) of **M**; and (d) using tables, or a numerical procedure, we evaluate T from (49)-(52). Then, the semiaxes are given by

$$a_1^2 = (\lambda_1 - \lambda_3)\cot^2 \varphi_0 + (\lambda_1 - \lambda_3), \qquad (53)$$

$$a_2^2 = (\lambda_1 - \lambda_3)\cot^2 \varphi_0 + (\lambda_2 - \lambda_3), \qquad (54)$$

$$a_3^2 = (\lambda_1 - \lambda_3)\cot^2 \varphi_0, \qquad (55)$$

while the orientation is given by the Euler angles

$$\varphi = \sin^{-1} \left(P_{31} / \sqrt{1 - P_{33}^2} \right) , \qquad (56)$$

$$\theta = \sin^{-1} \sqrt{1 - P_{33}^2} , \qquad (57)$$

$$y = \sin^{-1} \left(P_{13} / \sqrt{1 - P_{33}^2} \right), \qquad (58)$$

which are derived from (25)-(27).

IV. DEGENERATE CASES

The case of a spheroid corresponds to $a_2 = a_3$, while the rotational symmetry reduces the number of independent parameters for the determination of the orientation to 2. Therefore, we only need to evaluate the two semiaxes a_1 , a_2 and the two Euler angles φ , θ that fix the axis of the spheroid. As-

suming y = 0, the orthogonal transformation that brings the arbitrarily chosen primed system to the system of principal axes of the spheroid, takes the form

$$\mathbf{P}_{0} = \begin{bmatrix} \cos\varphi & \sin\varphi & 0\\ -\cos\theta\sin\varphi & \cos\theta\cos\varphi & \sin\theta\\ \sin\theta\sin\varphi & -\sin\theta\cos\varphi & \cos\theta \end{bmatrix}.$$
 (59)

Then, the m_1, m_2, m_3 measurements provide us with the system

$$a_1^2 \cos^2 \varphi + a_2^2 \sin^2 \varphi = 3(T - m_1)/m_0, \qquad (60)$$

$$a_1^2 \sin^2 \varphi + a_2^2 \cos^2 \varphi = 3(T - m_2)/m_0, \qquad (61)$$

$$a_2^2 = 3(T - m_3)/m_0, (62)$$

which has the solution

$$a_1^2 = 3(T - m_1 - m_2 + m_3)/m_0, \qquad (63)$$

$$a_2^2 = 3(T - m_3)/m_0, (64)$$

$$\sin^2 \varphi = (m_2 - m_3)/(m_1 + m_2 - 2m_3). \tag{65}$$

Therefore, the three measurements along the axes are enough to determine both semiaxes as well as the first Euler angle φ . Using (59) into Eqs. (39)–(41) we confirm that the measurements m_4, m_5, m_6 are not independent any more and they can be expressed in terms of m_1, m_2 , and m_3 .

Note that any rotation around the axis of the spheroid, which is fixed by the angles φ, θ , leaves the spheroid invariant. Therefore, the measurements m_1, m_2, m_3 will be invariant under any rotation by the third Euler angle y. In particular, if $y = \pi/2$, then the elements of the first column of $\mathbf{P}_{\pi/2}$ are given by

$$P_{\pi/2,11} = -\cos\theta\sin\varphi,$$

$$P_{\pi/2,21} = -\cos\varphi,$$

$$P_{\pi/2,31} = \sin\theta\sin\varphi,$$
(66)

and since the value of m_1 is still the same, Eq. (33) yields

$$a_1^2 \cos^2 \theta \sin^2 \varphi + a_2^2 (\cos^2 \varphi + \sin^2 \theta \sin^2 \varphi)$$

= 3(T - m₁)/m₀, (67)

$$\cos^2 \theta = (m_1 - m_3)/(m_2 - m_3) .$$
 (68)

The value of T is obtained from (29) where the elliptic integral (15) can be evaluated in this case and its value is given by³

$$I = \begin{cases} (1/\sqrt{a_1^2 - a_2^2}) \cos h^{-1} (a_1/a_2), & a_1 > a_2, \\ (1/\sqrt{a_2^2 - a_1^2}) \cos^{-1} (a_1/a_2), & a_1 < a_2. \end{cases}$$
(69)

Therefore,

$$T = [(\delta + 1)m_3 - m_1 - m_2]/(\delta - 1), \qquad (70)$$

where

$$\delta = \begin{cases} \cos h^2 \left[\sqrt{3(2m_3 - m_1 - m_2)} / m_0 \right], & a_1 > a_2, \\ \cos^2 \left[\sqrt{3(m_1 + m_2 - 2m_3)} / m_0 \right], & a_1 < a_2. \end{cases}$$
(71)

Consequently, rotational symmetry reduces the number of necessary measurements by 3.

The case of radial symmetry reduces the number of measurements by another 3. In fact, for a sphere the only quantity we need to evaluate is its radius a and this comes out from the m_0 measurement, since $I = a^{-1}$ and $a = \sqrt{m_0}$.

Of course, the reduction of the total number of measurements, from seven for the ellipsoid, to four for the spheroids, and to one for the sphere assumes *a priori* information about the actual shape of the scatterer. If there is no such information at our disposal then we have to perform all seven measurements and then decide whether the scatterer is an ellipsoid, a spheroid, or a sphere by looking at the number of equal eigenvalues of the matrix of measurements M.

Finally, we discuss the case where the orientation or the semiaxes are known. In particular, if the orientation is known, then by choosing the coordinate system along the principal axes the three Eqs. (33)-(35) take the form

$$a_k^2 = 3(T - m_k)/m_0, \quad k = 1, 2, 3.$$
 (72)

In this case every measurement determines a semiaxis and of course T is evaluated as before via (29). Hence, as it is expected, if the orientation is known the number of measurements is reduced by three (the three Euler angles).

On the other hand, if the three semiaxes are known then m_0 is readily evaluated from (15) and (29). Also T is known from its expression (19), via (16). Hence, the eigenvalues of **M** are known, which in turn implies that the elements of the matrix **M** are connected through the known values of the three invariants of **M**, i.e., the trace $\lambda_1 + \lambda_2 + \lambda_3$, the determinant $\lambda_1\lambda_2\lambda_3$ and the second invariant $\lambda_1\lambda_2 + \lambda_2\lambda_3 + \lambda_3\lambda_1$. Consequently, only three out of the six measurements m_k , k = 1,2,...,6, convey independent information about the orientation of the scatterer.

²B. D. Sleeman, "The low-frequency scalar Dirichlet scattering by a general ellipsoid," J. Inst. Math. Appl. **3**, 291 (1967).

³G. Dassios, "Second order low-frequency scattering by the soft ellipsoid," SIAM J. Math. Appl. **38**, 373 (1980).

⁴V. Twersky, "Certain transmission and reflection theorems," J. Appl. Phys. **25**, 859 (1954).

⁵D. Colton, and R. Kress, *Integral Equation Methods in Scattering Theory* (Wiley, New York, 1983).

⁷J. Van Bladel, "Low frequency scattering by hard and soft bodies," J. Acoust. Soc. Am. 44, 1067 (1968).

⁹M. Abramowitz, and I. A. Stegun, *Handbook of Mathematical Functions* (National Bureau of Standards, Washington, DC, 1964).

¹R. E. Kleinman, "The Dirichlet problem for the Helmholtz equation," A. R. M. A. 18, 205 (1965).

⁶T. S. Angell, and R. E. Kleinman, "Polarizability tensors in low frequency inverse scattering," to appear.

⁸H. Goldstein, *Classical Mechanics* (Addison-Wesley, Reading, MA, 1981).

Integrability of restricted multiple three-wave interactions

Frank Verheest

Instituut voor Theoretische Mechanika, Rijksuniversiteit Gent, Krijgslaan 281, B-9000 Gent, Belgium

(Received 15 April 1987; accepted for publication 30 July 1987)

Using a Hamiltonian framework with complex canonical variables allows for the determination of irreducible forms, which serve as building blocks for polynomial invariants. All the independent invariants in involution are thus obtained for the restricted multiple three-wave interactions, where all triads are coupled through a common pump (or daughter) wave, in the case of equal coupling strengths in all triads. The mixed, common pump/daughter wave case is not integrable.

I. INTRODUCTION

Nonlinear interactions between coherent waves or coupled oscillators can take place in many different physical systems, such as plasma physics, nonlinear optics, ocean wave dynamics, and many more. In the last decade, attention has been focused on the integrability or nonintegrability of the Hamiltonian equations describing the interactions between several wave triads, where each triad of coupled waves interacts with the other triads via one or possibly two common waves.¹⁻⁷

As shown already elsewhere,⁸ the sets of coupled-amplitude equations usually originate in complex notation and they can be studied with great advantage by using a Hamiltonian formalism, in which the complex conjugates of the wave amplitudes are also their canonically conjugated variables. The Hamiltonian remains a real quantity, in contrast to the recent study of integrable multiwave systems by Wojciechowski *et al.*,⁷ who also use complex canonical variables.

Not only is the algebra greatly simplified, but one can use with great advantage the concept of irreducible forms to find in a rather systematic way the real invariants in involution needed to prove complete integrability, and this without having to resort to Lax operators, as was done by Menyuk *et* $al.^{5,6}$ or Wojciechowski *et al.*⁷ These irreducible forms are the simplest combinations possible of wave quantities which remain constant on the fast oscillation time scale, and hence they can serve as building blocks in the construction of polynomial invariants on the slow modulation time scale.⁹

II. BASIC EQUATIONS FOR MANY COUPLED TRIADS WITH A COMMON PUMP WAVE

The nonlinear interaction between 2N + 1 waves is obtained by using a multiple time scale analysis¹⁰ that separates the fast oscillations of each wave or oscillator from the slow modulation of its amplitude due to the nonlinear coupling between the different waves.

For simplicity, we will start from selection rules for the wave or oscillator frequencies, where the wave common to all triads is a so-called pump wave,

$$\omega_0 = \omega_m + \Omega_m + \delta_m \quad (m = 1, ..., N), \tag{1}$$

with analogous relations for the wave vectors, if need be. Other cases of selection rules will be discussed further on.

In (1), δ_m is a small frequency mismatch or detuning. The complex amplitudes of the waves with frequencies ω_0 , ω_m , and Ω_m (m = 1,...N) are denoted by c, a_m , and b_m

$$(m = 1,...,N)$$
, and their slow-time variations are given by

$$\dot{a}_{m} = i\lambda_{m}cb_{m}, \quad \bar{a}_{m} = -i\lambda_{m}\bar{c}b_{m},$$

$$\dot{b}_{m} = i\lambda_{m}c\bar{a}_{m} + i\delta_{m}b_{m}, \quad \dot{b}_{m} = -i\lambda_{m}\bar{c}a_{m} - i\delta_{m}\bar{b}_{m},$$

$$\dot{c} = i\sum_{m=1}^{N}\lambda_{m}a_{m}b_{m}, \quad \dot{c} = -i\sum_{m=1}^{N}\lambda_{m}\bar{a}_{m}\bar{b}_{m}.$$
 (2)

A different scaling on the wave amplitudes can put the frequency detuning into these equations in various ways, but (2) is about the simplest possible form. The dot refers to a derivative on the slow modulation time scale, and the bar to complex conjugation.

The set (2) is derivable from the Hamiltonian

$$H = \sum_{m=1}^{N} (\lambda_m (a_m b_m \overline{c} + \overline{a}_m \overline{b}_m c) + \delta_m b_m \overline{b}_m), \qquad (3)$$

provided Hamilton's equations are written as⁸

$$\dot{a}_{m} = i \frac{\partial H}{\partial \bar{a}_{m}}, \quad \dot{b}_{m} = i \frac{\partial H}{\partial \bar{b}_{m}}, \quad \dot{c} = i \frac{\partial H}{\partial \bar{c}}, \quad (4)$$
$$\dot{\bar{a}}_{m} = -i \frac{\partial H}{\partial \bar{a}_{m}}, \quad \dot{\bar{b}}_{m} = -i \frac{\partial H}{\partial \bar{b}_{m}}, \quad \dot{\bar{c}} = -i \frac{\partial H}{\partial c}.$$

The analysis of Menyuk *et al.*^{5,6} showed that (2) is generically nonintegrable for arbitrary initial conditions and arbitrary frequency mismatches, except when all coupling constants λ_m are equal, in which case one can rescale all amplitudes so as to get $\lambda_m = 1$.

Menyuk *et al.*,⁵ using the Painlevé criteria, also found a second case in which (2) would be integrable for arbitrary initial conditions, namely when

$$\lambda_1 = \dots = \lambda_M = 2\lambda_{M+1} = \dots = 2\lambda_N,$$

$$\delta_1 = \dots = \delta_M = 2\delta_{M+1} = \dots = 2\delta_N.$$
(5)

However, in this case the direct proof of integrability via a complete set of independent invariants in involution has so far eluded all efforts.

Finally, with perfect frequency tuning, one can select special initial conditions, making the value of H zero, and then the system described by (2) is integrable for arbitrary λ_m (Ref. 3).

Here we will look at the restricted multiple three-wave interactions (2) with all $\lambda_m = 1$. The coupling is also restricted in the sense that the different triads share only a common wave, the same common wave in each triad.

III. IRREDUCIBLE FORMS AND INDEPENDENT INVARIANTS IN INVOLUTION

In the Introduction, the irreducible forms have been described as the simplest real polynomial combinations of $a_m \exp i(\mathbf{k}_m \cdot \mathbf{x} - \omega_m t_{\text{fast}}), \qquad b_m \exp i(\mathbf{K}_m \cdot \mathbf{x} - \Omega_m t_{\text{fast}}),$ $c \exp i(\mathbf{k}_0 \cdot \mathbf{x} - \omega_0 t_{\text{fast}}),$ and their complex conjugates remaining invariant on the fast time scale. For a system such as (2), governed by selection rules (1), the irreducible forms are

$$a_{m}\bar{a}_{m}, \quad b_{m}\bar{b}_{m}, \quad c\bar{c}, \quad a_{m}b_{m}\bar{c}+\bar{a}_{m}\bar{b}_{m}c,$$

$$a_{m}b_{m}\bar{a}_{p}\bar{b}_{p}+\bar{a}_{m}\bar{b}_{m}a_{p}b_{p}.$$
 (6)

All other real polynomial forms remaining constant with respect to the fast time scale (or the space scale, for that matter) are necessarily combinations of those mentioned in (6).

Irreducible forms as in (6) are usually not constant on the slow time scale, but suitable combinations can be invariant. Such invariants can be found by taking a closer look at the slow time derivatives of the irreducible forms and seeing how these can cancel.

The irreducible forms of the first line in (6) yield precisely N + 1 independent Manley-Rowe relations or invariants, say one for each triad,

$$a_m \bar{a}_m = b_m \bar{b}_m + C_m, \tag{7}$$

besides a global one,

$$c\bar{c} + \sum_{m=1}^{N} b_m \bar{b}_m = E.$$
(8)

Through the correspondence between invariants or first integrals and symmetries of Hamiltonian systems,¹¹ these Manley-Rowe relations generate the following (N + 1)parameter group of transformations:

$$a_{m} = \hat{a}_{m} \exp i\epsilon_{m},$$

$$b_{m} = \hat{b}_{m} \exp(i\epsilon - i\epsilon_{m}),$$

$$c = \hat{c} \exp i\epsilon.$$

(9)

which leave the set of equations (2) invariant. Since the N + 1 Manley-Rowe relations are already in involution, one could attempt to extend them to a complete set of 2N + 1 first integrals in involution. If this can be done, the additional first integrals will automatically have to be invariant under the (N + 1)-parameter transformation group (9). Real polynomials invariant under this group are precisely the irreducible forms given in (6) or combinations thereof. So it is entirely natural to look upon the irreducible forms as building blocks for additional polynomial invariants in involution.

Returning now to (6), one sees that the irreducible forms of the first line combine to the Hamiltonian (3).

Thus one starts for the other invariants from the irreducible forms $a_m b_m \bar{a}_p \bar{b}_p + \bar{a}_m \bar{b}_m a_p b_p$. To see how we proceed, let us call

$$H_m = a_m b_m \bar{c} + \bar{a}_m \bar{b}_m c + \delta_m b_m \bar{b}_m, \tag{10}$$

so that

$$H = \sum_{m=1}^{N} H_m, \tag{11}$$

with

$$\dot{H}_m = i \sum_{\substack{p=1\\p\neq m}}^N (a_p b_p \bar{a}_m \bar{b}_m - a_m b_m \bar{a}_p \bar{b}_p).$$
(12)

Similarly, setting

$$U_{mp} = \|a_m \bar{a}_p - \bar{b}_m b_p\|^2$$

= $a_m \bar{a}_m a_p \bar{a}_p + b_m \bar{b}_m b_p \bar{b}_p - a_m b_m \bar{a}_p \bar{b}_p$
 $- a_p b_p \bar{a}_m \bar{b}_m,$ (13)

which contain the last set of irreducible forms, one sees that

$$\dot{U}_{mp} = i(\delta_p - \delta_m)(a_m b_m \bar{a}_p \bar{b}_p - \bar{a}_m \bar{b}_m a_p b_p).$$
(14)

If all δ_m are different from each other, one gets immediately a set of N additional invariants.

$$I_m = H_m + \sum_{\substack{p=1\\p \neq m}}^{N} U_{mp} (\delta_p - \delta_m)^{-1}.$$
 (15)

These correspond essentially to the invariants found earlier^{6,7} by use of the Painlevé analysis and the Lax operator framework. However, as

$$H = \sum_{m=1}^{N} I_m = \sum_{m=1}^{N} H_m,$$
 (16)

the Hamiltonian is functionally dependent upon the I_m . The system (2) is thus completely integrable, as we have 2N + 1 invariants (C_m, E, I_m) in involution, meaning that their Poisson brackets vanish. This last point can easily be checked.

If all δ_m are equal, it is clear from (14) that the U_{mp} themselves are $\frac{1}{2}N(N-1)$ additional independent invariants, besides the Manley-Rowe relations. Notice that

$$U_{mm} = C_m^2. (17)$$

However, as the U_{mp} are not all in involution, it becomes necessary to group them in a proper way. This can be done in various equivalent ways, one of them being

$$K_m = \sum_{p=m+1}^{N} U_{mp}.$$
 (18)

There are N-1 of these combinations, and now the 2N+1 independent invariants in involution are (C_m, E, H, K_m) .

For the intermediate cases, where some but not all frequency mismatches are equal, we proceed as follows.

Suppose first, to fix the ideas, that δ_1 and δ_2 are equal, but all the others mutually different. One cannot then use I_1 or I_2 as given in (15), but can take instead U_{12} and

$$I_{12} = H_1 + H_2 + \sum_{p=3}^{N} (U_{1p} + U_{2p}) (\delta_p - \delta_1)^{-1}$$
$$= \lim_{\delta_2 \to \delta_1} (I_1 + I_2), \tag{19}$$

and similarly for every pair of equal detunings.

Were three frequency mismatches to be equal, say $\delta_1 = \delta_2 = \delta_3$, then one replaces I_2 , I_2 , and I_3 by

$$I_{123} = H_1 + H_2 + H_3$$

+ $\sum_{p=4}^{N} (U_{1p} + U_{2p} + U_{3p}) (\delta_p - \delta_1)^{-1}, \quad (20)$
 $K'_1 = U_{12} + U_{13}, \quad K'_2 = U_{23}.$

All possible choices of equal and unequal detunings can be dealt with in an analogous way. The new invariants in (19) or (20) are independent and in involution, with each other and with the others, with H as a dependent invariant.

IV. OTHER CASES AND FINAL REMARKS

In the above discussion we treated multiple three-wave couplings, where the one common wave was a pump wave in each triad.

The case of a common daughter wave in each triad is governed by the selection rules

$$\Omega_m = \omega_0 + \omega_m + \delta_m, \tag{21}$$

but runs entirely on analogous lines and will not be given here explicitly. Such systems have been proposed as models for internal ocean waves or plasma turbulence and are detailed elsewhere^{1,6,7} via different methods.

Rather surprisingly, the mixed case where the common wave is a pump wave in some triads [with selection rules of type (1)] and a daughter wave in the others [with rules of type (21)] is qualitatively quite different, but seems to have been overlooked in previous studies.^{6,7} For N = 2 it seems already nonintegrable, even when λ_1 and λ_2 are equal, and the addition of more triads will not improve matters.⁹

It is hoped that the above given treatment of restricted multiple three-wave couplings amply demonstrate the power of the combined method of a complex Hamiltonian formalism and irreducible forms. Although it was not detailed further for lack of space, the use of irreducible forms also directly points out that all λ_m have to be equal if integrability is to exist.

The fact that the other case, where the couplings and the detunings obey (5) and which was also surmised by Menyuk *et al.*⁵ to be integrable, so far seems intractable could perhaps be an indication that its missing invariant (or invariants) cannot be a simple polynomial.

ACKNOWLEDGMENTS

It is a pleasure to thank Dr. W. Sarlet, Dr. F. Cantrijn, and Professor W. Hereman (University of Wisconsin at Madison) for many fundamental discussions.

The author also thanks the National Fund for Scientific Research (Belgium) for a research grant.

- ¹K. M. Watson, B. J. West, and B. I. Cohen, J. Fluid Mech. 77, 185 (1976).
- ²J. D. Meiss, Phys. Rev. A 19, 1780 (1979).
- ³L. Falk, Proceedings of the 1982 International Conference on Plasma Phys-
- *ics* (Chalmers, Göteborg, 1982), Vol. 1, p. 96.
- ⁴F. J. Romeiras, Phys. Lett. A **93**, 227 (1983).
- ⁵C. R. Menyuk, H. H. Chen, and Y. C. Lee, Phys. Rev. A 27, 1597 (1983).
- ⁶C. R. Menyuk, H. H. Chen, and Y. C. Lee, J. Math. Phys. 24, 1073 (1983).
 ⁷S. Wojciechowski, Z. Jiang, and R. K. Bullough, Phys. Lett. A 117, 399 (1986).
- ⁸F. Verheest, J. Phys. A: Math. Gen. 20, 103 (1987).
- ⁹F. Verheest, to be published in Proc. Egypt Math. Phys. Soc.
- ¹⁰A. H. Nayfeh, *Perturbation Methods* (Wiley-Interscience, New York, 1973).
- ¹¹E. J. Saletan and A. H. Cromer, *Theoretical Mechanics* (Wiley, New York, 1971).

Quantum observables: Compatibility versus commutativity and maximal information

Paul Busch Institut für Theoretische Physik, Universität zu Köln, D-5000 Köln 41, Federal Republic of Germany

Thomas P. Schonbek and Franklin E. Schroeck, Jr. Department of Mathematics, Florida Atlantic University, Boca Raton, Florida 33431

(Received 24 March 1987; accepted for publication 24 June 1987)

Two different approaches to a characterization of the degree of (in)compatibility of quantum observables are investigated. First, recent examples of the (partial) commutativity of spectral measures of incompatible observables are proved to be generic. The analysis is extended to the case of compatible or incompatible unsharp, or stochastic observables, leading to a general criterion for commutativity of position and momentum effects. Further, a recently proposed information theoretic quantification of the (in)compatibility of noncommuting observables is generalized, and the relation between "maximal information," "minimal uncertainty," partial commutativity, and strict correlation is further clarified. Both approaches are illustrated in a number of examples.

I. INTRODUCTION

This work contributes to a more refined understanding of the operational content of noncommutativity of quantum observables. It is well known that noncommutativity is equivalent to incompatibility in the case of sharp observables represented by projection-valued (PV) measures.¹ On the other hand, the so-called unsharp, or stochastic observables [represented by effect-valued, also called positive-operatorvalued (POV) measures] may be compatible, or coexistent (that is, admit joint observables) without being commutative.² Recent investigations have shown that the degree of noncommutativity can be quantified; for example, the incompatible sharp position and momentum observables possess pairs of spectral projections which are either totally noncommutative, or partially commutative, or even commutative.³ The underlying characterization of the degree of (non)commutativity can be extended to the case of stochastic observables. This is the subject of Sec. II where the most general criterion for the commutativity of position and momentum effects is given. We show that the introduction of unsharpness does not necessarily change the degree of noncommutativity. Thus all POV measures representing (unsharp) position and momentum have different choices of arguments (Borel sets) for which they partially or totally commute, for other choices do not commute, and still are coexistent in the sense of having joint observables. We conclude that commutativity is not an ideal characterization of compatibility (simultaneous observability) of observables. We turn to an alternative quantitative measure concerning simultaneous measurability, namely a quantification of the indeterminacy of the simultaneous measurement of any observables.

Noncommuting observables cannot be simultaneously determined with arbitrary accuracy; there must in general be a certain amount of indeterminacy, no matter what measure of the spread is used. This fact raises the question as to how much *information* can simultaneously be available about those observables. In Ref. 4 several functionals $\mathscr{F}[\phi] = f(\langle E \rangle_{\phi}, \langle F \rangle_{\phi})$ expressible as functions of the expectation values of effects *E*, *F* in state ϕ were studied as potential measures of (maximal) information. Here (Sec. III) the maximization of a general class of functionals will be investigated in some detail, leading to a clarification of the relation between "maximal information" and "minimal uncertainty." In particular, maximal information in whatever sense requires the vanishing of the expectation of the commutator and equality in the uncertainty relation, that is, strict correlation for that maximal information state. Thus commutativity in the expectation value arises as a consequence of a general information analysis.

In Sec. IV the main conclusions will be summarized.

II. DEGREE OF NONCOMMUTATIVITY

For pairs of Hilbert space projections P, Q the degree of noncommutativity can be described in terms of the "commutativity projection"

$$\operatorname{com}(P,Q) = (P \land Q) \lor (P \land Q^{\perp})$$
$$\lor (P^{\perp} \land Q) \lor (P^{\perp} \land Q^{\perp})$$
(1)

 $(P \land Q, P \lor Q$ denoting the projectors onto the intersection, union of the ranges of P and Q, respectively, $P^{\perp} = 1 - P$, etc.). The operator com(P, Q) is the projector onto the subspace of vectors on which P and Q do commute. This formulation can immediately be extended to the class of effects E, $F(0 \le E, F \le 1, \text{ that is, } E \text{ and } F \text{ have spectra within the inter$ val [0, 1]): com<math>(E, F) projects onto the subspace of states on which E and F are commutative, that is,

$$\operatorname{com}(E,F)H = \{\phi \in H \mid (EF - FE)\phi = [E,F]\phi = 0\}.$$

(1')

Then one may call E, F commutative if com(E,F) = 1, partially (non)commutative if $0 \neq \text{com}(E,F) \neq 1$, and (totally) noncommutative if com(E,F) = 0. In Ref. 3 examples of pairs of position and momentum spectral projections $E^Q(X)$, $E^P(Y)$ were given for each category. Here Q and P may be introduced as any (irreducible) representation of the Weyl commutation relation; that is, Q, P may be any Fourier couple. For technical reasons, such as the use of Fourier transformations, one employs the Schrödinger representation. Then the results of Ref. 3 can be extended to unsharp position and momentum observables, defined as POV measures through the following:

$$a_{v}^{Q}(X) = \int_{\mathbb{R}} dv(q) E^{Q}(X+q)$$

= $\int_{\mathbb{R}} dE^{Q}(q) (v * \chi_{X})(q) = (v * \chi_{X})(Q),$
$$a_{v}^{P}(Y) = \int_{\mathbb{R}} dv'(p) E^{P}(Y+p)$$

= $\int_{\mathbb{R}} dE^{P}(p) (v' * \chi_{Y})(p) = (v' * \chi_{Y})(P),$
(2)

where v, v' are normalized, positive confidence (Borel) measures, $X, Y \in B(\mathbb{R})$. Thus

$$C(X,Y) \equiv \left[a_{\nu}^{Q}(X), a_{\nu}^{P}(Y)\right] = \iint_{\mathbb{R}^{2}} d\nu(q) d\nu'(p)$$
$$\times \left[E^{Q}(X+q), E^{P}(Y+p)\right]. \tag{3}$$

Now one may readily translate some of the results obtained in Ref. 3 for the spectral projections.

It has been shown in Ref. 3 that $E^Q(X)$, $E^P(Y)$ may be (i) totally noncommutative, e.g., for X, Y being half-lines; (ii) partially commutative, e.g., for X, Y being bounded Borel sets; (iii) commutative for (X,Y) being α -periodic, that is, $(X,Y) = (X + \alpha, Y + 2\pi/\alpha)$ for some positive constant α .

From the above expression for C(X,Y) it is seen that commutativity holds for arbitrary measures v, v' if (X,Y) is α -periodic: in that case also (X + q, Y + p) is α -periodic for arbitrary q, p. Now let us assume v, v' to have bounded supports [-a,a] and [-b,b], respectively. Let X $= [-x_{0r}x_{0}], Y = [-y_{0r}y_{0}]$ and define $X^{\alpha} = [-x_{0}$ $-a,x_{0} + a], Y^{b} = [-y_{0} - b,y_{0} + b]$. Then $X + q \subseteq X^{\alpha}$, $Y + p \subseteq Y^{b}$ for $q \in \text{supp } v$, $p \in \text{supp } v'$, so

$$E = E^{Q}(\mathbb{R} \setminus X^{a}) \wedge E^{P}(\mathbb{R} \setminus Y^{b})$$

$$\leq E^{Q}(\mathbb{R}\setminus X+q)\wedge E^{P}(\mathbb{R}\setminus Y+p),$$

for all $q \in \text{supp } v$, $p \in \text{supp } v'$; and all eigenvectors of E will be eigenvectors of all $E^Q(X+q)$, $E^P(Y+p)$. Thus $C(X,Y)\phi = 0$ on the subspace $E\mathscr{H}$, that is, $a_v^Q(X)$ and $a_v^P(Y)$ are partially commutative. (We will argue below that these effects are not commutative.) Next, if X, Y are halflines, supp v and supp v' bounded, then $a_v^Q(X)$, $a_v^P(Y)$ are totally noncommutative because $E^Q(X)$ and $E^P(Y)$ are: the supports of $a_v^Q(X)$ and $a_v^P(Y)$ are half-lines again so that the intersection of their ranges is $\{0\}$; this holds for all pairs $(a_v^Q(X^i), a_v^P(Y^j))$ with $X^i \in \{X, \mathbb{R} \setminus X\}$, $Y^j \in \{Y, \mathbb{R} \setminus Y\}$ so that there is no common range at all.

The last two examples rest on the assumption of compact supports for v, v'; the introduction of more general kinds of unsharpness (confidence measures) seems to decrease the extent of partial commutativity. On the other hand, it is the introduction of unsharpness that opens the possibility of compatible position and momentum observables,^{2,5} and it is necessary to avoid compact support of at least one of the measures v,v' for achieving compatibility. Take absolutely continuous measures dv(q) = f(q)dq, dv'(p) = g(p)dp, where $f(q) = |\psi_0(q)|^2$, $g(p) = |\psi_0(p)|^2$, for $\psi_0 \in L^2(\mathbb{R})$. Then a_v^Q , a_v^P are marginals of the phase space observable

$$Z \rightarrow A(Z) = \iint_{Z} dq \, dp |\psi_{qp}\rangle \langle \psi_{qp}|, \quad Z \in B(\mathbb{R}^{2}),$$
$$\psi_{qp} = \exp[i(-qP + pQ)]\psi_{0}, \qquad (4)$$

and therefore are coexistent although they do not commute in general. The opposite will also occur: for compactly supported measures and bounded sets X, Y the effects $a_v^Q(X)$, $a_v^P(Y)$ possess no joint observable, but these effects do partially commute as was shown above. This confirms the conclusion of Ref. 2 that commutativity is not essential to coexistence (compatibility) in case of stochastic observables.

We now generalize the α -periodicity condition slightly to get a necessary and sufficient condition for commutativity. For this we consider the following.

Let v, v' be positive normalized Borel measures and define

$$v_X = v * \chi_X, \quad v'_Y = v' * \chi_Y, \quad \text{for } X, Y \in B(\mathbb{R});$$

that is,

$$v_X(q) = \int dv(y) \chi_X(q-y)$$
 a.e., etc.

Then $\nu_X, \nu_Y \in L^{\infty}$ (R,dq). If we also let \mathscr{F} (\mathscr{F}^{-1}) denote (inverse) Fourier transform in $L^2(\mathbb{R},dq)$ and working in the Schrödinger representation for Q, P, we then have, for unsharp observables,

$$C(X,Y)\phi \equiv \left[a_{v}^{Q}(X),a_{v}^{P}(Y)\right]\phi$$
$$= v_{X}\mathcal{F}^{-1}\left[v_{Y}'\mathcal{F}(\phi)\right]$$
$$-\mathcal{F}^{-1}\left[v_{Y}'\mathcal{F}(v_{X}\phi)\right],$$
(5)

for all $\phi \in L^2(\mathbb{R},dq)$. Since $f \in L^{\infty}$, $g \in L^2$ implies $fg \in L^2$, this defines C(X,Y) as a mapping on L^2 . A complete characterization of C(X,Y) = 0 is given by the following theorem the proof of which can be found in the Appendix.

Theorem 1: Let $f,g \in L^{\infty}(\mathbb{R},dq)$. Then

$$f\mathcal{F}^{-1}[g\mathcal{F}(\phi)] = \mathcal{F}^{-1}[g\mathcal{F}(f\phi)]$$
(6)

holds for all $\phi \in L^2(\mathbb{R}, dq)$ if and only if f and g satisfy one of the following: (a) at least one of f or g is constant a.e.; or (b) f and g are both periodic with minimal periods β , α satisfying $2\pi/\alpha\beta \in \mathbb{Z} \setminus \{0\}$ ("generalized α -periodicity").

We remark that case (a) is of little interest here since we identify $f = v_X$, $g = v'_Y$. If we have either constant, this leads to one of $a_v^Q(X)$, $a_v^P(Y)$ being a multiple of the identity. The case (b) with v, v' taken as point measures (delta measures) generalizes the known results for sharp observables.³ For the general case, consider

$$v_X(q) = v_X(q+\beta)$$
 a.e. q

$$\int d\nu(y)\chi_X(q-y) = \int d\nu(y)\chi_X(q+\beta-y) \text{ a.e. } q$$
f

iff

iff

$$v(-X+q) = v(-X+q+\beta)$$
 a.e. q

$$v((-X/(-X+\beta))+q) = v((-X+\beta)/X+q)$$
 a.e. q,

from which we see that periodicity of X leads to periodicity of v_X . We therefore obtain a second derivation of commutativity of the *unsharp* observables from the (generalized) α periodicity of (X,Y). We also note that there are no solutions $v_X \neq 0$ for X essentially bounded or essentially semi-infinite since this leads to a contradiction to v being a normalized positive measure.

In view of the one-to-one correspondence between L^{∞} functions f,g and bounded operator functions of position and momentum,

$$f(Q) = \int_{\mathbf{R}} f(q) dE^{Q}(q), \quad g(P) = \int_{\mathbf{R}} g(p) dE^{P}(p), \quad (7)$$

Eq. (6) can be rewritten as

$$[f(Q),g(P)] = 0$$
(8)

which holds in any Hilbert space representation of the canonical commutation relations if and only if one of the following is satisfied: (a) at least one of f(Q), g(P) is a multiple of the identity operator, and (b) f and g are both periodic with minimal periods β, α satisfying $2\pi/\alpha\beta \in \mathbb{Z} \setminus \{0\}$.

III. INFORMATION MAXIMIZATION

In Ref. 4 various measures of "joint information" for (expected values of) pairs of effects E,F have been studied with regard to their maximization. The first variation (in the sense of the Fréchet differential⁶) of the functional $\mathscr{F}[\phi] = f(\langle E \rangle_{\phi}, \langle F \rangle_{\phi})$ with respect to $\phi \in \mathscr{H}$ is given by

 $\mathcal{F}[\phi + \delta\phi] - \mathcal{F}[\phi] = \delta\mathcal{F}[\delta\phi;\phi] + \epsilon[\delta\phi],$

where $\delta \mathcal{F}$ must be a continuous linear functional in $\delta \phi$ and $\lim_{||\delta \phi|| \to 0} (\epsilon[\delta \phi]/||\delta \phi||) = 0$. Here \mathcal{F} will be locally extremal on a vector ϕ with $||\phi|| = 1$ only if

$$\delta(\mathscr{F} + \lambda \langle \phi | \phi \rangle) = 0,$$

which implies

$$0 = (f_1 E + f_2 F + \lambda \mathbf{1})\phi, \tag{9}$$

where $f_i = (\partial f / \partial x_i)(\langle E \rangle_{\phi}, \langle F \rangle_{\phi})$ (i = 1,2). The Lagrange parameter is found by taking the expectation

$$\lambda = -f_1 \langle E \rangle_{\phi} - f_2 \langle F \rangle_{\phi};$$

therefore Eq. (1) can be written as

$$0 = (f_1 E' + f_2 F')\phi,$$

$$E' = E - \langle E \rangle_{\phi} \mathbf{1}, \quad F' = F - \langle F \rangle_{\phi} \mathbf{1}.$$
(10)

Multiplying this equation either with E' or with F' and taking the expectation yields the system

$$0 = f_1 \langle E'^2 \rangle_{\phi} + f_2 \langle E'F' \rangle_{\phi},$$

$$0 = f_1 \langle F'E' \rangle_{\phi} + f_2 \langle F'^2 \rangle_{\phi}.$$
(11)

Here f_1 and f_2 are fixed as partial derivatives of f; so in order to have local extrema for \mathcal{F} one must have either $f_1 = f_2 = 0$ or have the vanishing of the determinant,

$$\operatorname{Var}_{\phi}(E) \cdot \operatorname{Var}_{\phi}(F) - |\langle E'F' \rangle_{\phi}|^{2} = 0.$$
 (12)

In that latter case one solves (assuming $f_2 \neq 0$)

$$\frac{f_1}{f_2} = -\frac{\langle E'F' \rangle_{\phi}}{\operatorname{Var}_{\phi}(E)} = -\frac{\operatorname{Var}_{\phi}(F)}{\langle F'E' \rangle_{\phi}} = \pm \left[\frac{\operatorname{Var}_{\phi}(F)}{\operatorname{Var}_{\phi}(E)} \right]^{1/2}.$$
 (13)

Here we have assumed f to be real. If $f_2 = 0$ then either $f_1 = 0$ or $\operatorname{Var}_{\phi}(E) = 0$, i.e., $E'\phi = 0$. Therefore we may separate the "trivial" solutions corresponding to $f_1 = f_2 = 0$ (local extremum of f), or $f_1 = 0$, $F'\phi = 0$, or $f_2 = 0$, $E'\phi = 0$, or $F'\phi = E'\phi = 0$, and the remaining solutions of (10) have to satisfy (12) and (13).

These connections can be generalized in several respects, as we shall see in the following theorems. In particular, we shall consider functionals not only on the Hilbert space of pure states but also try to include mixed states. The most natural way of extending the above derivations to this general class of states turns out to be in terms of the Hilbert Schmidt class $B_2(H)$; the choice of the trace class $B_1(H)$ leads to trivial results only.

We shall first prove the following.

Theorem 2: Let \mathscr{F} be a real-valued functional on $B_2(H)$ such that $\mathscr{F}[r] = f(\langle A_1 \rangle_{\rho}, ..., \langle A_n \rangle_{\rho})$, A_i bounded self-adjoint operators, $\rho = rr^*$, $r \in B_2(H)$ [so ρ is a positive element in $B_1(H)$], $\langle A_i \rangle_{\rho} = \text{Tr}[\rho A_i] = (r, A_i r)$, f differentiable. Then \mathscr{F} is Fréchet differentiable. In particular, if \mathscr{F} has a local extremum on $r = \rho^{1/2}$ under the constraint $\text{Tr}[\rho] = (r, r) = 1$, then the first Fréchet variation vanishes, $0 = \delta(\mathscr{F} + \lambda \operatorname{Tr}[\rho])$, which implies

$$0 = \sum_{i} f_i (A_i - \langle A_i \rangle_{\rho} \mathbf{1}) \cdot \rho^{1/2}, \qquad (14)$$

where

$$f_i = \frac{\partial f}{\partial x_i} (\langle A_1 \rangle_{\rho}, ..., \langle A_n \rangle_{\rho}),$$

or, equivalently,

$$0 = \sum_{i} f_{i} (A_{i} - \langle A_{i} \rangle_{\rho} \mathbf{1}) \phi, \text{ for all } \phi \text{ in the range of } \rho.$$
(15)

Proof: The Fréchet differentiability of \mathcal{F} is shown by a straightforward but gory computation. The Fréchet differential of \mathcal{F} is

$$\delta \mathscr{F}[\delta r; r] = \sum f_i [(r, A_i \delta r) + (\delta r, A_i r)].$$

It is linear, i.e., $\delta \mathscr{F}[h+k;r] = \delta \mathscr{F}[h;r] + \delta \mathscr{F}[k;r]$, and bounded with respect to the $B_2(H)$ norm $(||r||_2 = (r,r)^{1/2} = {\mathrm{Tr}[r^*r]}^{1/2})$:

$$|\delta \mathscr{F}[\delta r;r]| \leq \left\{ 2\sum_{i} |f_i| \|\mathcal{A}_i\| \right\} \|\delta r\|_2.$$

Then for a local extremum of $\mathcal F$ one has

$$0 = \delta(\mathscr{F} + \lambda \operatorname{Tr}[\rho]) = \left(\delta r, \left\{\sum_{i} f_{i} A_{i} + \lambda \mathbf{1}\right\} r\right) + \text{c.c.},$$

which yields, due to the arbitrariness of $r \in B_2(H)$,

$$0 = \left\{\sum_{i} f_i A_i + \lambda \mathbf{1}\right\} \cdot \rho^{1/2}.$$

The Lagrange parameter λ is found by evaluation of the condition $1 = \text{Tr}[\rho] = (r,r)$:

$$\lambda = -\sum_i f_i \langle A_i \rangle_{\rho}.$$

The remaining step is straightforward since for bounded A, $A\rho^{1/2} = 0$ is equivalent to $A\rho = 0$ and also to $A\phi = 0$ for all ϕ in the range of ρ . This completes the proof.

At first sight it appears surprising that all vectors in the range of the maximizing ρ satisfy the same equation (15). However, this simply means that the pure states maximizing the functional \mathscr{F} span a closed subspace which is an eigenspace of the bounded linear operator $\sum_{i=1}^{n} f_i A_i$.

Theorem 3: The coefficients α_i in the (linear in ϕ) equation $0 = (\sum_i \alpha_i A_i + \lambda 1)\phi$, or equivalently (in the nonlinear equation) $\sum_i \alpha_i A_i'\phi = 0$ $(A_i' = A_i - \langle A_i \rangle_{\phi})$, are determined as functions $\alpha_i = g_i (\langle A_k' A_i' \rangle_{\phi}; \gamma_j)$ of the (complex) covariances $\langle A_k' A_i' \rangle_{\phi}$ and *r* parameters γ_j where n - r is the rank of the matrix $(\langle A_k' A_i' \rangle_{\phi})$. A nontrivial set $(\alpha_1,...,\alpha_n) \neq (0,...,0)$ requires det $(\langle A_k' A_i' \rangle_{\phi}) = 0$.

Proof: Multiplying the equation with A'_k , k = 1,...,n, and taking the expectation yields the system

$$\sum_{i=1}^{n} \alpha_i \langle A'_k A'_i \rangle_{\phi} = 0, \quad k = 1, \dots, n.$$

The remainder is a simple application of linear algebra.

The case n = 1 has a nontrivial solution iff $\operatorname{Var}_{\phi}(A_1) = 0$; that is, ϕ is an eigenvector of A_1 . The case n = 2 has been solved completely in Ref. 7 with the following result.

Theorem 4: In the uncertainty relation (Cauchy-Schwarz inequality) for self-adjoint A, B,

$$\operatorname{Var}_{\rho}(A) \cdot \operatorname{Var}_{\rho}(B)$$

$$\equiv \langle A^{\prime 2} \rangle_{\rho} \langle B^{\prime 2} \rangle_{\rho} \geq |\langle A^{\prime}B^{\prime} \rangle_{\rho}|^{2}$$

$$= |\frac{1}{2} \langle [A,B] \rangle_{\rho} |^{2} + (\frac{1}{2} \langle \{A^{\prime},B^{\prime}\}_{+} \rangle_{\rho})^{2} \qquad (16)$$

equality is obtained if and only if $\operatorname{Var}_{\rho}(B) = 0$ or $\operatorname{Var}_{\rho}(B) \neq 0$ and $[A' - (\langle B'A' \rangle_{\rho} / \operatorname{Var}_{\rho}(B) B'] \rho^{1/2} = 0$. This latter condition is equivalent to

$$0 = [A' - (\langle B'A' \rangle_{\rho} / \operatorname{Var}_{\rho}(B) \rangle B'] \rho, \qquad (17)$$

or

$$0 = [A' - (\langle B'A' \rangle_{\rho} / \operatorname{Var}_{\rho} (B))B']\phi,$$

for all ϕ in the range of ρ . (18)

So we arrive at just the "linear" equation obtained earlier from the maximization of $f(\langle E \rangle_{\phi}, \langle F \rangle_{\phi})$ [cf. Eq. (10)].

We summarize and combine the above statements.

Theorem 5: A real functional $\mathscr{F}[r] = f(\langle A_1 \rangle_{\rho}, ..., \langle A_n \rangle_{\rho}), \rho = rr^*, r \in B_2(H)$, with f and self-

adjoint bounded A_i such that \mathcal{F} is Fréchet differentiable will be locally extremal only on states ρ satisfying

$$\sum_{i=1}^{n} f_i (A_i - \langle A_i \rangle_{\rho} \mathbf{1}) \rho = 0$$
⁽¹⁹⁾

and for $(f_1, ..., f_n)$ obeying

$$\sum_{i=1}^{n} f_i \langle A'_k A'_i \rangle_{\rho} = 0, \qquad (20)$$

where either $f_1 = \cdots = f_n = 0$ or det $(\langle A'_k A'_i \rangle_{\rho}) = 0$.

The solutions of these equations yield extrema of \mathcal{F} iff at $(x_1,...,x_n) = (\langle A_1 \rangle_{\rho},...,\langle A_n \rangle_{\rho})$ the function f is extremal on the numerical range of $A_1,...,A_n$,

num ran (A_1, \dots, A_n)

$$=\{(x_1,\ldots,x_n)|x_i=\langle A_i\rangle_{\rho},$$

 ρ any positive trace 1 operator}.

Consequences: (i) For n = 1 the only possible solutions are $(\partial f/\partial x)(\langle A \rangle_{\rho}) = 0$ or $\operatorname{Var}_{\rho}(A) = 0$, that is, ρ is an "eigenstate" of A.

(ii) For n = 2, either ρ is a common eigenstate of A, B, or $f_1 = f_2 = 0$ on ρ , or $\operatorname{Var}_{\rho}(A) = 0$ and $f_2 = 0$, or $\operatorname{Var}_{\rho}(B) = 0$ and $f_1 = 0$, or $\operatorname{Var}_{\rho}(a) \neq 0 \neq \operatorname{Var}_{\rho}(B)$ and

$$\frac{f_1}{f_2} = -\frac{\langle A'B'\rangle_{\rho}}{\operatorname{Var}_{\rho}(A)} = -\frac{\operatorname{Var}_{\rho}(B)}{\langle B'A'\rangle_{\rho}} = \pm \left[\frac{\operatorname{Var}_{\rho}(B)}{\operatorname{Var}_{\rho}(A)}\right]^{1/2}$$
(21)

and

$$\operatorname{Var}_{\rho}(A)\operatorname{Var}_{\rho}(B) = \frac{1}{4}(\langle \{A', B'\}_{+}\rangle_{\rho})^{2} = [\operatorname{cov}_{\rho}(A, B)]^{2}.$$
(22)

In particular, the reality of f_1, f_2 forces $\langle A'B' \rangle_{\rho}$ to be real so that only the covariance term appears in (22), and

$$\langle [A,B]_{-} \rangle_{\rho} = 0$$
 on any extremalizing state ρ . (23)
Equation (22) tells that in such states ρ the observables A, B
are strictly (anti-)correlated, $\operatorname{corr}_{\rho}(A,B) = \pm 1$.

With these results we arrive at a generalization and a complete clarification of the relationships (9)-(13) reviewed in the beginning of this section which had been found earlier⁴ in some examples. Furthermore, we obtain a means for comparing the various possible measures of information insofar as different functions f(x,y) admit different solutions for possible extrema. This will be illustrated in a number of examples. First we consider pairs of projections P, Q, so that num ran $(P,Q) \subseteq [0,1] \times [0,1]$, and the unit square will be the natural domain of our functions f(x,y).

Example 1:

$$f(x,y) = x \ln x + (1-x)\ln(1-x) + y \ln y + (1-y)\ln(1-y).$$

Here

$$I_{\phi}(Q) = \langle Q \rangle_{\phi} \ln \langle Q \rangle_{\phi} + \langle 1 - Q \rangle_{\phi} \ln \langle 1 - Q \rangle_{\phi}$$

represents the (missing) information about Q in state ϕ in the sense of negative entropy, and

$$I_{\phi}(Q,P) = I_{\phi}(Q) + I_{\phi}(P) = f(\langle Q \rangle_{\phi}, \langle P \rangle_{\phi})$$

is a measure of the information simultaneously available on

Q and P. Now $f_1 = \ln[x/(1-x)], f_2 = \ln[y/(1-y)]$, and $f_1 = f_2 = 0$ if and only if $x = y = \frac{1}{2}$ which is known to yield the absolute minimum $-2 \ln 2$ of f and thus of $\mathscr{F}[\phi] = I_{\phi}(Q,P)$. The absolute maximum 0 of f is assumed on $(x,y) \in \{(0,0), (0,1), (1,0), (1,1)\}$, which can be realized by $(\langle P \rangle_{a}, \langle Q \rangle_{a})$ if and only if there exist joint eigenstates. Thus it depends on the degree of commutativity whether or not the information can be absolutely maximal or not.

The variance of a projection is $\operatorname{Var}_{\phi}(P)$ $= \langle P \rangle_{\phi} - \langle P \rangle_{\phi}^2$ so that condition (21) on f_1/f_2 reads $[x(1-x)]^{1/2} \ln[x/(1-x)]$ $= + [y(1-y)]^{1/2} \ln[y/(1-y)]$ or

V

$$\Psi(x) = \Psi(y)^{\pm 1}, \quad \Psi(x) = (x/(1-x))^{[x(1-x)]^{1/2}},$$

by continuity $\Psi(0) = \Psi(1) = 1$. Noting where $\Psi(0) = \Psi(\frac{1}{2}) = \Psi(1) = 1$ and that Ψ has a minimum on $x_0 \in (0, \frac{1}{2})$ and a maximum on $x_1 \in (\frac{1}{2}, 1)$ and satisfies $\Psi(1-x) = \Psi(x)^{-1}$, one obtains exactly four solutions $y = x, y = 1 - x, y = x^*$, and $y = 1 - x^*$, where x and x^* are the solutions of $\Psi(x) = u$ for u in the range of Ψ (for $x = x_0$ or $x = x_1$ it follows that $x = x^*$, so in that case there are only two solutions).

Example 2: f(x,y) = x + y. Here $\mathscr{F}[\phi] = \langle P \rangle_{\phi}$ $+\langle Q \rangle_{\phi}$ is the sum of probabilities. Then $f_1 = f_2 = 1$, so Eq. (21) yields

 $1 = f_1/f_2 = [y(1-y)/x(1-x)]^{1/2},$

which has y = x and y = 1 - x as the only solutions.

Example 3: $f(x,y) = x \cdot y$ corresponds to the product of probabilities. Here $f_1 = y$, $f_2 = x$. The absolute minimum occurs on x = 0 or y = 0. Otherwise [Eq. (21)]

 $f_1/f_2 = y/x = [y(1-y)/x(1-x)]^{1/2}$

or

$$[(1-x)/x]^{1/2} = [(1-y)/y]^{1/2}$$

with y = x as the only solution.

Example 4: $f(x,y) = x(1-x) \cdot y(1-y)$ corresponds $\mathscr{F}[\phi] = \operatorname{Var}_{\phi}(P) \cdot \operatorname{Var}_{\phi}(Q).$ Here $f_1 = (1 - 2x)$ to $(1-y), f_2 = x(1-x)(1-2y), \text{ and } f_1 = 0 \text{ iff } x = \frac{1}{2} \text{ or } f_1 = 0$ $y \in \{0,1\}$. Thus the absolute minimum 0 of f occurs on the eigenstates of P,Q, the absolute maximum on $x = y = \frac{1}{2}$. If neither is the case, (21) yields

$$\Psi(x) \equiv [x(1-x)]^{1/2}/(1-2x) = \pm \Psi(y)$$

with $\Psi(1-y) = -\Psi(y), \Psi'(x) > 0$. So the only solutions are y = x and y = 1 - x.

In all of these examples the solutions y = x (and y = 1 - x lead to (local) extrema if the boundary points (x,y) of num ran(P,Q), satisfying y = x (or y = 1 - x) belong to the set num ran(P,Q). We see that the number of possible extrema depends essentially on the degree of symmetry of the functions f(f(x,y) = f(y,x); f(1-x,y)= f(x,y), etc.) as well as on the shape of the (convex) numerical range of P,Q. We remark that Lenard⁸ gives an exhaustive description of num ran(P,Q) so that in principle all extrema of functionals $\mathscr{F}[\phi] = f(\langle P \rangle_{\phi}, \langle Q \rangle_{\phi})$ can actually be determined.

Example 5: The final example deals with a more complicated functional,

$$\mathcal{F}[\phi] = \operatorname{Var}_{\phi}(A) \cdot \operatorname{Var}_{\phi}(B)$$

= $(\langle A^2 \rangle_{\phi} - \langle A \rangle_{\phi}^2) (\langle B^2 \rangle_{\phi} - \langle B \rangle_{\phi}^2)$
= $f(\langle A \rangle_{\phi}, \langle B \rangle_{\phi}, \langle A^2 \rangle_{\phi}, \langle B^2 \rangle_{\phi}).$

Application of Theorem 2 yields for the extremal states,

$$0 = [\operatorname{Var}_{\phi}(B)(A^{2} - 2\langle A \rangle_{\phi}A) + \operatorname{Var}_{\phi}(A)(B^{2} - 2\langle B \rangle_{\phi}B) + \lambda 1]\phi;$$

since $A'^2 - \langle A \rangle_{\phi}^2 \mathbb{1} = A^2 - 2 \langle A \rangle_{\phi} A$, etc., this equation can be brought into the form

$$0 = \left[\operatorname{Var}_{\phi}(B)A'^{2} + \operatorname{Var}_{\phi}(A)B'^{2} - 2\operatorname{Var}_{\phi}(A)\cdot\operatorname{Var}_{\phi}(B)\mathbf{1}\right]\phi.$$

Possible solutions are eigenstates of A or B; further solutions will be found by observing that this equation is precisely an instance of the case n = 2 in Theorems 4 and 5, leading to Eqs. (21)-(23). In particular, Eq. (23) reads

$$\langle [A'^2,B'^2]_-\rangle_{\phi}=0.$$

If we insert for A, B the position Q and momentum P operators (although they are not bounded), then

$$[Q'^2,P'^2]_- = 2i\hbar\{Q',P'\}_+$$

so $\langle [Q'^2, P'^2]_{-} \rangle_{\phi} = 0$ if and only if $\langle \{Q', P'\}_{+} \rangle_{\phi} = 0$ which singles out the coherent (Gaussian) states of minimal uncertainty.

IV. CONCLUDING REMARKS

In Sec. II we constructed examples of generalized, unsharp position and momentum observables whose effects $a_{u}^{Q}(X), a_{u}^{P}(Y)$ may be either totally noncommutative, or partially commutative, or commutative. We established a characterization of the commutative case: $[a_{\nu}^{Q}(X), a_{\nu}^{P}(Y)] = 0$ whenever (X, Y) is α -periodic, $(X,Y) = (X + \alpha, Y + 2\pi/n\alpha)$ for some fixed $\alpha > 0$. For the investigation of partial commutativity the question of positive lower bounds⁹ $a \leq a_v^Q(X), a_v^P(Y)$ may be of relevance, as it is in the case of projections.

One possibility of introducing approximate measurements of noncommuting observables A, B within conventional quantum mechanics is in terms of joint measurements of commuting functions f(A), g(B) of A, B. Then Theorem 1 tells that the only approximate joint measurements of Q, P in this sense are those by means of (generalized) α -periodic functions f, g (as long as only bounded functions are considered). In contrast, after introducing POV observables commutativity is no longer necessary (though still sufficient) for coexistence so that much more flexiblity is gained.

These results show that "commutativity" and "compatibility" are guite unrelated in general from the mathematical standpoint. This is also true from the viewpoint of measurement theory.² There are compatible observables that do not commute; the "price" to be paid for obtaining coexistence "in spite" of noncommutativity is unsharpness, that is, lack of certainty inherent in the results of joint measurements.

This lack of certainty can be estimated in the way indicated in Sec. III, by searching for maxima of "information" functionals $\mathscr{F}[\phi] = f(\langle A_1 \rangle_{\phi}, ..., \langle A_n \rangle_{\phi})$. It turns out that optimal upper and lower bounds for the information are determined by the extrema of f on the numerical range of $(A_1,...,A_n)$. In particular, for n = 2 extrema occur only on strictly correlated states with vanishing expectation of the commutator. This means that on any state of maximal information, in the sense defined here, the variances have to be finite. Thus at least in this respect the usefulness of the variances as uncertainty measures is supported.¹⁰

According to Theorem 2, the information maximizing states ϕ satisfy an equation of the form [Eq. (15)] $\sum_i f_i A_i \phi = \lambda \mathbf{1} \phi$. In the context of U(n) representations with A_i being self-adjoint generators of infinitesimal unitary transformations, this equation generalizes the statement that ϕ is a highest weight state.⁷

For pairs of projections P, Q we found the locations of the solutions of the variational equations in the case of several functions f. The examples show that functions of the form f(x,y) = g(x) + g(y) (examples 1 and 2) are more suitable for the description of uncertainty than f(x,y) $= h(x) \cdot h(y) > 0$ (examples 3 and 4): in the latter type minimal uncertainty in the case of small h remains quite undetermined since the minimum 0 occurs if only one factor is zero. The symmetry f(x,y) = f(y,x) appears natural since Q and P should be treated on equal footing. But this yields y = x as a natural candidate for maximal information (examples 1– 4). If in addition f(1 - x,y) = f(x,y) then also the complements 1 - Q and 1 - P are treated equivalently, and further solutions may lie on the line y = 1 - x (examples 1 and 4).

To conclude, there is again only a very limited connection between (non)commutativity and maximal information. In Ref. 4 it was pointed out that for the totally noncommutative spectral projections $E^Q(\mathbb{R}^+)$, $E^P(\mathbb{R}^+)$ one may have information arbitrarily close to the supremum without reaching it. For bounded X, Y, maximal information is $(E^{Q}(\mathbb{R} \setminus X))$ achieved on the states satisfying $\wedge E^{P}(\mathbb{R} \setminus Y)\phi = \phi$ or $(E^{Q}(X) + E^{P}(Y))\phi = ||E^{Q}(X)$ $+ E^{P}(Y) \|\phi$; note that these $E^{Q}(X)$, $E^{P}(Y)$ are partially commutative. In general, both commutative and partially commutative projections admit absolutely maximal information in the sense of example 1, but this last example indicates that the measure of example 2 [without the symmetry f(x,y) = f(1-x,y) is more suitable since it points more directly to a characterization of the degree of "reality" of $E^{Q}(X)$ and $E^{P}(Y)$: the larger the intervals X, Y are chosen, the larger the maximal probability sum $||E^Q(X) + E^P(Y)||$ will be.

ACKNOWLEDGMENTS

One of the authors (P. B.) wishes to express his gratitude for the kind hospitality extended to him during his stay at the Department of Mathematics at Florida Atlantic University where this work has been carried out. Thanks are also due to Pekka J. Lahti, Finland, for critical reading of the manuscript. P. B. also gratefully acknowledges partial support by means of a travel grant from the Deutsche Forschungsgemeinschaft, Bonn, Federal Republic of Germany.

APPENDIX: PROOF OF THEOREM 1

Assume

$$f\mathcal{F}^{-1}[g\mathcal{F}(\phi)] = \mathcal{F}^{-1}[g\mathcal{F}(f\phi)] \tag{(*)}$$

holds for all $\phi \in L^2(\mathbb{R}, dq)$ and for some nonconstant $f,g \in L^{\infty}(\mathbb{R}, dq)$. Fix f and let G_f denote the set of all $h \in L^{\infty}(\mathbb{R}, dq)$ such that (*) holds with g replaced by h for all $\phi \in L^2(\mathbb{R}, dq)$. One sees that G_f is a nontrivial translation invariant subspace of $L^{\infty}(\mathbb{R}, dq)$ which is closed in the weak*-topology of $L^{\infty} = (L^1)^*$. Let W be the weak*-closed subspace of $L^{\infty}(\mathbb{R}, dq)$ spanned by g and its translates. Then $W \neq \{0\}$; hence W contains a character $V_\beta(x) = \exp(i\beta x)$ of \mathbb{R} , by Theorem 9 of Dunford and Schwartz.¹¹ By Theorem 16 of the same reference, W must contain a V_β with $\beta \neq 0$; in fact, if $V_0 \equiv 1$ is the only character in W then $g = \text{const} \cdot V_0$; i.e., g is constant. Since $W \subset G_f$, we have $V_\beta \in G_f$ and hence

$$f\mathcal{F}^{-1}[V_{\beta}\mathcal{F}(\phi)] = \mathcal{F}^{-1}[V_{\beta}\mathcal{F}(f\phi)];$$

that is, $f\phi_{\beta} = f_{\beta}\phi_{\beta}$, for all $\phi \in L^2(\mathbb{R}, dq)$, where $h_{\beta}(x) = h(x - \beta)$ for functions *h*. It follows that $f = f_{\beta}$ a.e.; i.e., β is a nonzero period of *f*. Conversely, if β is a period of *f*, then $V_{\beta} \in G_f$. Since *f* is measurable and nonconstant, *f* has a minimum period $\beta_0 > 0$. We can now reverse the roles of *f* and *g*; taking the Fourier transform of both sides of (*) we see (*) is equivalent to

$$g\mathcal{F}[f\mathcal{F}^{-1}(\hat{\phi})] = \mathcal{F}[f\mathcal{F}^{-1}(g\hat{\phi})]$$

where $\hat{\phi} = \mathcal{F}\phi$ is again a general element of $L^2(\mathbb{R}, dq)$. We conclude that g must be periodic with minimum period $\alpha_0 > 0$. We may set $f = \sum c_n V_{n\alpha}$ where $\alpha = 2\pi/\beta_0$. This implies

$$\sum_{n} c_n \mathscr{F}^{-1} [g_{n\alpha} \mathscr{F} (V_{n\alpha} \phi)] = \mathscr{F}^{-1} \bigg[g \mathscr{F} \bigg[\sum_{n} c_n V_{n\alpha} \phi \bigg] \bigg];$$

or, taking \mathcal{F} ,

$$\sum c_n g_{n\alpha} \mathscr{F}(V_{n\alpha} \phi) = g \sum c_n \mathscr{F}(V_{n\alpha} \phi), \quad \forall \phi \in L^2;$$

or

$$\sum c_n (g_{n\alpha} - g) \mathscr{F}(V_{n\alpha} \phi) = 0, \quad \forall \phi \in L^2;$$

or

$$\sum c_n (g_{n\alpha} - g) [\mathscr{F}(\phi)]_{n\alpha} = 0, \quad \forall \phi \in L^2.$$

Pick $m \in \mathbb{Z}$. For each $N \in \mathbb{Z}$, define $\phi_N \in L^2$ by

$$(\mathscr{F}\phi_N)(x) = \begin{cases} 1, & x \in [(N-m)\alpha, (N-m+1)\alpha], \\ 0, & \text{otherwise.} \end{cases}$$

Pick $y \in \mathbb{R}$. Then there is a unique N such that $y \in [N\alpha, (N+1)\alpha)$. Thus $[\mathscr{F}(\phi_N)]_{n\alpha}(y) = 0$ unless m = n. Hence

$$c_m(g_{m\alpha} - g)(y) = 0, y \in [N\alpha, (N+1)\alpha], \text{ a.e.}$$

since the choice N was arbitrary

Since the choice N was arbitrary,

$$c_m (g_{ma} - g)(y) = 0$$
 a.e.

Hence if $c_m \neq 0$, then $g_{m\alpha} = g$ a.e.; i.e., $m\alpha$ is a period for g. Since g has minimal period α_0 we must have the following.

If (g, f) is a solution pair for (*) with minimal periods α_0, β_0 , and for f written in the form

$$f=\sum c_n V_{n2\pi/\beta_0},$$

then

 $c_n \neq 0 \Rightarrow \exists$ integer z(n) such that

$$n2\pi/\beta_0=z(n)\alpha_0$$

Let $K = \{n \in \mathbb{Z} | c_n \neq 0\}.$

Let *M* be the subset of Z generated from the finite linear combinations of elements of *K* using integer coefficients. Here *M* is a closed subgroup of Z, so there is an $n_0 \in \mathbb{Z}/\{0\}$ with $M = n_0 \mathbb{Z}$. Thus every element *n* of *k* can be written $n = n_0 r$, some integer *r*. Then

$$f = \sum_{r} c_{n_0 r} V_{n_0 r 2\pi/\beta_0}$$

which has β_0/n_0 as a period. Since β_0 is the minimal period, we must have $n_0 = 1$. Now the equation

 $n2\pi/\beta_0=z(n)\alpha_0,$

z(n) some integer, holds for all $n \in K$ and therefore for all $n \in M$. Thus there is an integer $z \neq 0$ such that

$$2\pi/\alpha_0\beta_0=z.$$

Now we prove sufficiency. If either f or g is constant, (*) holds trivially. Thus suppose f, g are periodic L^{∞} functions with minimal periods β_{0}, α_{0} , respectively. Then one has for $\alpha = 2\pi/\beta_{0}$

$$V_{n\alpha}\mathcal{F}^{-1}[g\mathcal{F}(\phi)] = \mathcal{F}^{-1}[g_{n\alpha}\mathcal{F}(V_{n\alpha}\phi)]$$
$$= \mathcal{F}^{-1}[g\mathcal{F}(V_{n\alpha}\phi)]$$

whenever $n\alpha$ is a period of g. Multiplying by the c_n such that $f = \sum c_n V_{n\alpha}$ we obtain (*) after invoking generalized α -periodicity.

- ¹P. J. Lahti, in *Recent Developments of Quantum Logic*, edited by P. Mittelstaedt and E. W. Stachow (Bibliographisches Institut, Mannheim, 1985), pp. 61–80.
- ²F. E. Schroeck, Found. Phys. 15, 677 (1985).
- ³P. Busch and P. J. Lahti, Phys. Lett. A 115, 259 (1986).
- ⁴P. Busch and P. J. Lahti, J. Phys. A: Math. Gen. 20, 899 (1987).
- ⁵E. Prugovecki, Stochastic Quantum Mechanics and Quantum Spacetime (Reidel, Dordrecht, 1984).
- ⁶H. Sagan, Introduction to the Calculus of Variations (McGraw-Hill, New York, 1969).
- ⁷F. E. Schroeck, in Mathematical Foundations of Quantum Theory, edited
- by A. Marlow (Academic, New York, 1978), pp. 299-327.
- ⁸A. Lenard, J. Funct. Anal. 10, 410 (1972).
- ⁹P. Busch, J. Math. Phys. 25, 1794 (1984).
- ¹⁰J. Hilgevoord and J. B. M. Uffink, Found. Phys. 15, 925 (1985).
- ¹¹N. Dunford and J. T. Schwartz, *Linear Operators* (Interscience, New York, 1963), Vol. II, p. 987.

The problem of moments in the phase-space formulation of quantum mechanics

Francis J. Narcowich Department of Mathematics, Texas A&M University, College Station, Texas 77843-3368

(Received 13 April 1987; accepted for publication 13 August 1987)

Long ago, Moyal [Proc. Cambridge Philos. Soc. 45, 99 (1949)] formulated a moment problem in the context of the Wigner-Weyl phase-space formulation of quantum mechanics. The problem amounts to giving necessary and sufficient conditions for a sequence of numbers to be moments of a Wigner function. In this paper, that problem is solved, and so is a truncated version of it.

I. INTRODUCTION

My purpose in writing this paper is to give necessary and sufficient conditions for the solution of a quantum mechanical moment problem first mentioned by Moyal in his seminal 1949 paper¹: Given a sequence of numbers $m_{j,k}$, where j,kare the multi-indices

$$j = (j_1, ..., j_n), \quad k = (k_1, ..., k_n),$$
 (1.1)

and *n* is some fixed positive integer, find necessary and sufficient conditions for there to be a Wigner distribution function $\rho(\mathbf{q},\mathbf{p}), \mathbf{q} = (q_1,...,q_n), \mathbf{p} = (p_1,...,p_n)$, such that

$$m_{j,k} = \int \int \mathbf{p}^{j} \mathbf{q}^{k} \rho(\mathbf{q}, \mathbf{p}) d^{n} q d^{n} p . \qquad (1.2)$$

Here \mathbf{p}^{j} and \mathbf{q}^{k} stand for these quantities,

$$\mathbf{p}^{j} = p_{1}^{j_{1}} p_{2}^{j_{2}} \cdots p_{n}^{j_{n}}, \quad \mathbf{q}^{k} = q_{1}^{k_{1}} \cdots q_{n}^{k_{n}}.$$
(1.3)

In a recent paper,² O'Connell and I gave a set of *necessary* conditions for there to be a solution to this problem. While I will postpone any detailed discussion of that set, I do wish to make a few remarks about it.

The conditions in it amount to the non-negativity of a hierarchy of quadratic forms, where a given quadratic form has moments up to some fixed, even order for its coefficients. The simplest of these involves moments of order 2 or less, and its non-negativity is equivalent^{2,3} to the position-momentum uncertainty relation. The conditions guaranteeing the non-negativity of other quadratic forms in the set place restrictions on the higher order moments analogous to those for the simplest case; these restrictions may thus be thought of as generalized uncertainty relations.

The set of necessary and sufficient conditions that I will give here includes the set that O'Connell and I derived in Ref. 2; the relationship between the two sets will be discussed in the last section of the paper. For now, I will simply describe the set that I will later show to be necessary and sufficient. This requires some notation and a word or two about the Wigner-Weyl phase-space formulation of quantum mechanics.

Let $z \equiv (q,p)$ denote a point in phase space Γ , which is $\mathbb{R}^n \times \mathbb{R}^n$ equipped with the symplectic form

$$\sigma(z,z') \equiv (2/\hbar) \left(\mathbf{p} \cdot \mathbf{q}' - \mathbf{q} \cdot \mathbf{p}' \right)$$
(1.4)

and the measure

$$dz = dq_1 \cdots dq_n \, dp_1 \cdots dp_n \,. \tag{1.5}$$

Here A(z), B(z), C(z), etc., will denote functions or tempered (Schwartz) distributions defined on phase space, and $\hat{A}, \hat{B}, \hat{C},...$ will denote the operators associated with A, B, C,...via the Weyl transform.⁴ The twisted product of A with B is

$$A \circ B(z) \equiv (\pi \hbar)^{-2n} \iint A(z+z_1)B(z+z_2) \\ \times e^{-i\sigma(z_1,z_2)} dz_1 dz_2, \qquad (1.6)$$

and is the Wigner transform⁴ of \widehat{AB} . Finally, let

$$\langle A,B \rangle \equiv \int A(z) \ \overline{B(z)} \ dz \ .$$
 (1.7)

I will bring in other notation as needed. For the most part I will use the same notation that I used in Ref. 4. (I also remark that Secs. I and II of Ref. 4 contain a brief review of the phase-space formulation of quantum mechanics.)

One can easily show⁵ that if A and B are Schwartz functions—i.e., A,B are in $\mathscr{S}(\mathbb{R}^{2n})$, then $A \circ B$ is too. This allows us to define a notion of positivity on \mathscr{S}' , the set of tempered distributions. A tempered distribution F will be said to be \hbar *positive* if

$$\langle F, A \circ \overline{A} \rangle \geqslant 0 \tag{1.8}$$

for every $A \in \mathscr{S}(\mathbb{R}^{2n})$. When this happens, I will write

$$F \geqslant 0. \tag{1.9}$$

In terms of operators, F being \hbar positive means that

$$\langle F, A \circ \overline{A} \rangle = (2\pi \hbar)^n \operatorname{Tr}[\widehat{F} \widehat{A} \widehat{A}^*] \ge 0.$$
 (1.10)

From this it follows that \widehat{F} is a positive operator, and conversely.

Let P(z) be any polynomial in q, p, that is, let

$$P(z) = \sum_{j,k} c_{j,k} \mathbf{p}^{j} \mathbf{q}^{k}.$$
(1.11)

Next, define the "moment functional" μ via

$$\mu(P) = \sum_{j,k} c_{j,k} m_{j,k} . \qquad (1.12)$$

If the $m_{j,k}$ are of the form (1.1), then it is easy to see that

$$\mu(P) = \int P(z)\rho(z) \, dz = \langle P, \rho \rangle \,. \tag{1.13}$$

Since ρ is a Wigner function, it is the Wigner transform of a non-negative trace class operator; it thus has the form $\rho = G \circ G$. [Take $\hat{G} = (\hat{\rho})^{1/2}$.] From (1.13), one gets

$$\mu(P) = \langle P, G \circ G \rangle, \quad G = \overline{G}. \tag{1.14}$$

If $G \in \mathscr{S}(\mathbb{R}^{2n})$, all of the manipulations I have made are correct. Examining (1.14), I find that $\mu(P) \ge 0$ when P is \hbar positive.

To anyone with experience in classical moment problems, it should not come as a surprise that μ being positive for \hbar -positive *P* is also a sufficient condition for a solution to the moment problem to exist. There is, however, a surprise in the kind of solution one gets. The main result of this paper is the following theorem.

Theorem 1.1: A necessary and sufficient condition for there to exist an (unnormalized) Wigner function $\rho = G \circ G$, with G real valued and in \mathcal{S} , for which (1.2) holds is that $\mu(P) \ge 0$ whenever P is an \hbar -positive polynomial in q,p.

The surprise is, of course, that the Wigner function one gets to solve the moment problem (1.2) not only decays fast enough to support integrals of all powers of **p** and **q**, but it is also infinitely smooth, a Schwartz function in fact. No such regularity arises in solutions to classical moment problems.

I will prove Theorem 1.1 in two major steps. The first step is to extend the moment functional μ to certain spaces that contain the polynomials and that consist of smooth functions which are themselves sufficiently nice that one may use them as symbols in the Weyl calculus; the extensions that I get will be positive on the \hbar -positive elements of these spaces. The next step, which is the harder of the two, is to come up with a representation for certain elements in the dual of the extension spaces; these elements are just the continuous linear functionals that are positive on the \hbar -positive functions in the extension spaces. Putting the two steps together gives the theorem.

The remainder of the paper is organized this way. In Sec. II, I will carry out the first step of the proof. That is, I will prove the existence of a continuous linear functional that is positive on the \hbar -positive elements of the extension spaces and that agrees with the moment functional μ when restricted to the polynomials. In Sec. III, I will carry out the second step in the proof. In addition, I will also solve a truncated version of the moment problem (1.2). Finally, in Sec. IV, I will discuss the connection between the set of necessary conditions that O'Connell and I gave in Ref. 2, and I will make a few concluding remarks.

II. EXTENSIONS OF THE MOMENT FUNCTIONAL

To carry out the "extension" step in the proof of the main result, I need an appropriate set of spaces to work with. It turns out that for the purposes I have in mind the symbol classes S_r introduced by Voros⁵ will do nicely. These are defined this way: Let *r* be a fixed real number. The space S_r is defined as the set of all C^{∞} functions $A: \Gamma \to C$ that satisfy

$$\|A\|_{\alpha,r} \equiv \sup_{x\in\Gamma} \{ |\partial^{\alpha}A| (1+|z|^2)^{(|\alpha|-r)/2} \} < \infty .$$
 (2.1)

Here, $|z|^2 = \mathbf{q} \cdot \mathbf{q} + \mathbf{p} \cdot \mathbf{p}$, α is a multi-index, and ∂^{α} is the partial derivative corresponding to α . The $|| \quad ||_{\alpha,r}$'s are seminorms on S_r , and they can be used in the usual way to construct a topology on S_r .

There are several reasons for using the S_r 's. First, every polynomial of degree less than or, possibly, equal to r is contained in S_r . Second, the twisted product is defined for functions in these spaces. Indeed, if $A \in S_r$ and $B \in S_{r'}$, then⁵ $A \circ B \in S_{r+r'}$. Third, functions in S_r are tempered distributions, for, when |z| is large, they behave like polynomials. The set of functions in S_r that are also \hbar -positive tempered distributions is clearly a cone, which I will denote by S_r^+ . Finally, the relationship between these spaces and the Weyl calculus is well understood.^{5,6}

The space S_0 plays an important role here. One can show that there is a linear topological isomorphism between S_0 and S_r , and that this isomorphism associates the positive cones S_0^+ and S_r^+ with one another. In a bit, I will discuss this isomorphism. For now, I want to take a close look at the properties of S_0 itself. I will begin with this proposition.

Proposition 2.1: Let $A \in S_0$ and let A be real valued. There exists a continuous seminorm $\|\cdot\|_w$, defined on S_0 , such that

$$\|A^{\prime}\|_{w} \pm A \in S_{0}^{+} .$$

Proof: It is tedious, but quite straightforward, to show that the function

$$g_{z}(t) \equiv |t|^{2} / \hbar (1 + |z|^{2})$$
(2.3)

defines what Hörmander terms a "slowly varying" metric on phase space (Ref. 6, Definition 18.4.1), and that the function

$$m_r(z) \equiv (1+|z|^2)^{r/2} \tag{2.4}$$

is "g continuous" for g defined by (2.3). Thus Hörmander's symbol class (Ref. 6, Definition 18.4.2) $S(m_r,g_z)$ is well defined. Inspection of its definition shows that it is a Fréchet space identical with Voros's S_r . In particular, S_0 coincides with $S(1,g_z)$.

On the other hand, $S(1,g_z)$ satisfies the conditions necessary and sufficient for the L^2 -operator norm of the Weyl transform of a symbol in $S(1,g_z)$ to provide a continuous seminorm on $S(1,g_z)$. [See Theorem 18.6.3, Ref. 6. Checking that this is the case is again tedious. The reader should also be aware of notational differences between Ref. 6 and this paper. To compare the two notational systems, set h = 1here, and note that σ defined in (1.4) here is two times the σ Hörmander uses in Ref. 6.] Specifically, if $u(\mathbf{q}) \in L^2(\mathbb{R}^n)$, and if $A \in S(1,g_z)$, then

$$(\widehat{A}u)(\mathbf{q}) \equiv (2\pi\hbar)^{-n} \int e^{i\mathbf{p}\cdot(\mathbf{q}-\mathbf{q}')/\hbar} A\left(\frac{1}{2}(\mathbf{q}+\mathbf{q}'),\mathbf{p}\right)$$
$$\times u(\mathbf{q}') d^{n}\mathbf{q}' d^{n}\mathbf{p} \qquad (2.5)$$

is the Weyl transform of A. The norm $\|\widehat{A}\|_{op}$ is a continuous seminorm on the symbol space $S(1,g_z)$. I will denote this seminorm by

$$\|A\|_{w} \equiv \|\widehat{A}\|_{\text{op}} . \tag{2.6}$$

Let $F(z) \in \mathscr{S}(\Gamma)$, and recall that^{5,7} the corresponding Weyl transform \hat{F} is a trace-class operator. Moreover,⁷

$$\int A(z) \left(F \circ \widetilde{F}\right)(z) \, dz = (2\pi \hbar)^n \operatorname{tr}(\widehat{A} \widehat{F} \widehat{F}^*) \,. \tag{2.7}$$

If A is real, then \widehat{A} is self-adjoint and satisfies

$$- \|\widehat{A}\|_{op} \cdot \widehat{1} \leqslant \widehat{A} \leqslant \|\widehat{A}\|_{op} \cdot \widehat{1}.$$
(2.8)

This operator inequality is preserved if one premultiplies and postmultiplies by \hat{F}^* and \hat{F} , respectively. Hence, one has that

$$-\|\widehat{A}\|_{\mathrm{op}}\widehat{F}^*\widehat{F}\leqslant\widehat{F}^*\widehat{A}\widehat{F}\leqslant\|\widehat{A}\|_{\mathrm{op}}\widehat{F}^*\widehat{F}.$$
(2.9)

Using the fact that taking the trace in (2.9) will leave the sense of the inequalities unchanged and rewriting (2.9) in an obvious way, one gets

$$\operatorname{tr}(\widehat{F}^{*}[\|\widehat{A}\|_{\operatorname{op}} \cdot \widehat{1} \pm \widehat{A}]\widehat{F}) \ge 0.$$
(2.10)

From (2.6), (2.7), (2.10), and $tr(\widehat{RS}) = tr(\widehat{SR})$, one finally arrives at this:

$$\int (\|A\|_w \pm A) F \circ \overline{F} \, dz \ge 0 \,. \tag{2.11}$$

Since $S(1,g_z)$ and S_0 are the same Fréchet space, (2.11) holds for all real $A \in S_0$. From the definition of S_0^+ , one sees that (2.11) implies (2.2).

For possible future reference, I want to draw off a corollary to the proof of Proposition 2.1.

Corollary 2.2: With g_z defined by (2.3) and $m_r(z)$ given by (2.4), Hörmander's symbol class $S(m_r,g_z)$ is the same Fréchet space as Voros's symbol class S_r .

What needs to be done next is to construct a mapping that makes S_0 and S_r topologically isomorphic and that associates S_0^+ with S_r^+ . To do this, first let

$$H(z) \equiv \frac{1}{2} (\mathbf{q} \cdot \mathbf{q} + \mathbf{p} \cdot \mathbf{p}) ; \qquad (2.12)$$

this is, of course, the classical Hamiltonian for n harmonic oscillators with all of the masses and frequencies taken to be 1. Next, define

$$H_r(z) \equiv$$
 Wigner transform of (\dot{H}^r) . (2.13)

(I will use a subscript and not a superscript because I want to avoid confusing powers relative to the twisted product with powers relative to ordinary multiplication.) These functions have the properties listed in the following.

Lemma 2.4: If r and t are real, and if H_r is given by (2.13), then H_r is real valued, belongs to S_{2r} , and satisfies

$$H_r \circ H_t = H_{r+t} . \tag{2.14}$$

Proof: Let $\beta > 0$; define the operator

$$\widehat{\Omega}_{\beta} \equiv \exp(-\beta \widehat{H}) . \tag{2.15}$$

Using standard functional analytic techniques, one can write

$$\widehat{H}^{-r} = \frac{1}{\Gamma(r)} \int_0^\infty \beta^{r-1} \widehat{\Omega}_\beta \, d\beta \tag{2.16}$$

for all r > 0. Taking the Wigner transform of both sides and using the fact that the Wigner transform commutes with the integral in (2.16), one gets

$$H_{-r}(z) = \frac{1}{\Gamma(r)} \int_0^\infty \beta^{r-1} \Omega_\beta(z) \, d\beta \,, \qquad (2.17)$$

where $\Omega_{\beta}(z)$, the Wigner transform of Ω_{β} , is given by

$$\Omega_{\beta}(z) = \operatorname{sech}^{n}(\hbar\beta/2)$$
$$\times \exp(-\left[2H(z)/h\right] \tanh(\hbar\beta/2)). \quad (2.18)$$

The one-dimensional version of (2.18) is derived in Ref. 8. To get (2.18), one need only take products of the onedimensional Wigner functions. I now want to show that $H_{-r} \in S_{-2r}$. First, observe that

$$\partial^{\alpha} H_{-r}(z) = \sum_{|\gamma| < |\alpha|} z^{\gamma} g \int_{0}^{\infty} \beta^{r-1} \times \left[\tanh\left(\frac{\hbar\beta}{2}\right) \right]^{(|\gamma| + |\alpha|)/2} \Omega_{\beta}(z) \, d\beta \,;$$
(2.19)

here, c_{γ} depends on the multi-indices, α, γ , on \hbar , and nothing else.

The integral in (2.19) is easy to estimate for large values of |z|. Just break the interval of integration into [0,1] and $[1,\infty)$. The part of the integral from $[1,\infty)$ decays like $\exp(-\cosh|z|^2)$. On the other hand, the part from [0,1] satisfies

$$\int_0^1 \beta^{r-1} \left[\tanh\left(\frac{\hbar\beta}{2}\right) \right]^{(|\gamma|+|\alpha|)/2} \Omega_\beta(z) \ d\beta$$
$$\leq C \int_0^\infty \beta^{(2r+|\gamma|+|\alpha|)/2-1} \exp(-\beta H) \ d\beta$$

and so it decays like $C|z|^{-2(2r+|r|+|\alpha|)}$. The whole integral thus decays like the power estimate for the [0,1] interval.

By what I have just found concerning the integral in (2.19), I get that for $|z| \to \infty$

$$\left|\partial^{\alpha}H_{-r}(z)\right| \leq C |z|^{-(2r+|\alpha|)},$$

from which it immediately follows that $H_{-r} \in S_{-2r}$. To see that $H_r \in S_{2r}$ for $r \ge 0$, write r this way

$$r=k-(k-r),$$

where k is any integer strictly larger than r. Because of the operator identity

$$\widehat{H}^{r} = \widehat{H}^{r} \cdot \widehat{H}^{-(k-r)}.$$

one has that

$$H_r = H_k \circ H_{-(k-r)} \; .$$

Since $H = H_1$ is in S_2 , because it is a polynomial of degree 2, $H_k = H \circ H \circ \cdots \circ H$ (k times) is in S_{2k} . The function $H_{-(k-r)}$ is, by the analysis above, in S_{2r-2k} . The twisted product of H_k and $H_{-(k-r)}$ is⁵ then in $S_{2k+2r-2k} = S_{2r}$.

The formula (2.14) is the phase-space version of the corresponding operator identity. That $H_r(z)$ is real is obvious.

Lemma 2.4 is the key to constructing the isomorphism between S_0 and S_r that I mentioned earlier. To construct this, first observe⁵ that the mapping $A \rightarrow A^{\circ}H_{-r/4}$ takes S_t continuously into $S_{t-r/2}$, and that $B \rightarrow H_{-r/4} \circ B$ takes $S_{t-r/2}$ continuously into S_{t-r} . Thus

$$T_{r}(A) = H_{-r/4} \circ A \circ H_{-r/4}$$
(2.20)

defines a continuous, linear map from S_t to S_{t-r} , for every $r,t \in \mathbb{R}$. Moreover, T_{-r} inverts T_r , for

$$T_{-r}(T_{r}(A)) = H_{r/4} \circ H_{-r/4} \circ A \circ H_{-r/4} \circ H_{r/4}$$
$$= H_{0} \circ A \circ H_{0}$$
$$= 1 \circ A \circ 1 = A.$$

Similarly, $T_r(T_r(A)) = A$. Since T_r , is a continuous linear map too, the mapping T_r sets up a topological isomorphism between the two Fréchet spaces S_t and S_{t-r} .

It is obvious that this isomorphism satisfies

$$\overline{T_r(A)} = T_r(\overline{A}) , \qquad (2.21)$$

so that real-valued functions are associated with real-valued functions. What requires a little more work, but is more important here, is that S_i^+ and S_{i-}^+ , are associated via this isomorphism. To see that is so, one must show that $A \in S_i$ is \hbar positive if and only if $T_r(A)$ is, too.

Let $A \ge 0$. For every $B \in \mathscr{S}(\Gamma)$, one has that $H_{-r/4} \circ B$ is also in $\mathscr{S}(\Gamma)$, and that

$$\int A(z)(H_{-r/4}\circ B)\circ \overline{(H_{-r/4}\circ B)} \, dz \ge 0. \qquad (2.22)$$

Using the associativity of the twisted product and the identity

$$\int C \circ D \, dz = \int C(z) D(z) \, dz \,, \qquad (2.23)$$

which applies under fairly general conditions, among which are included one of C,D being in $\mathcal{S}(\Gamma)$ while the other is in some S_r , one may justify these manipulations done on the right-hand side of (2.22),

rhs of (2.22) =
$$\int A \circ H_{-r/4} \circ B \circ \overline{B} \circ H_{-r/4} dz$$

= $\int H_{-r/4} \circ A \circ H_{-r/4} \circ B \circ \overline{B} dz$
= $\int T_r(A) \circ B \circ \overline{B} dz$
= $\langle T_r(A), B \circ \overline{B} \rangle$. (2.24)

Hence, for all $B \in \mathscr{S}(\Gamma)$, (2.22) implies that

$$\langle T_r(A), B \circ \overline{B} \rangle \ge 0$$
, (2.25)

and so $T_r(A) \ge 0$. A similar manipulation justifies the reverse implication, and proves the following theorem.

Theorem 2.5: The mapping $T_r: S_t \to S_{t-r}$ defined in (2.20) is a continuous linear bijection for every $r,t \in \mathbb{R}$. The inverse of T_r is T_{-r} . Under T_r , \hbar -positive functions in S_t correspond to \hbar -positive functions in S_{t-r} , and real functions correspond to real functions.

To complete the "extension step," I need one last result.

Theorem 2.6: Let *E* be a self-adjoint subspace of S_r and suppose that $H_{r/2}(z) \in E$. If ϕ_0 is a linear functional that is defined on *E*, real on the real functions in *E*, and positive on $E \cap S_r^+$, then there exists a continuous linear functional ϕ that is defined on S_r , that is non-negative on S_r^+ , and that agrees with ϕ_0 on *E*.

Proof: I begin by noting that \hbar -positive tempered distributions are real valued. The easiest way to see this is to observe that the symplectic Fourier transform of one is a tempered distribution of \hbar -positive type,⁹ and that such a distribution satisfies a symmetry condition [Ref. 9, Proposition 3.1, (iii)] equivalent to the original distribution being real.

The point is that I may work entirely with the spaces $\operatorname{Re}(E)$ and $\operatorname{Re}(S_r)$, which compromise, respectively, the real-valued functions in E and S_r . Doing so allows me to use

Krein's version¹⁰ of the Hahn-Banach theorem to produce the extension ϕ .

Let $A \in \operatorname{Re}(S_r)$, and let

$$D \equiv H_{r/2} - A$$
. (2.26)

Note that for T_r defined by (2.20)

$$T_r(D) = 1 - T_r(A)$$
, (2.27)

and that

 $T_r(D) = 1 - \|T_r(A)\|_w + \|T_r(A)\|_w - T_r(A); \quad (2.28)$

here, $\|\cdot\|_{w}$ is the continuous seminorm that was mentioned in Proposition 2.1. Composing $\|\cdot\|_{w}$ with T_r produces a continuous seminorm on S_r and, hence, on $\operatorname{Re}(S_r)$. The A's in $\operatorname{Re}(S_r)$ for which $\|\operatorname{Tr}(A)\|_{w} < 1$ clearly form a (real) neighbor of 0 in $\operatorname{Re} S_r$; call this neighborhood U. From Proposition 2.1, (2.28), and from the fact that positive constants belong to S_0^+ , one sees that if $A \in U$, then $T_r(D)$ is in S_0^+ . By Theorem 2.5, $T_r(D) \in S_0^+$ if and only if $D \in S_r^+$.

Since the *D*'s in *S*_r⁺ corresponding to the *A*'s in *U* form a neighborhood of $H_{r/2}$, the conditions of Krein's theorem (Ref. 10, Theorem 2.6.3, p. 136) are fulfilled. Hence there exists a continuous, positive linear functional ϕ that extends ϕ_0 from Re(*E*) to Re(*S*_r). Simply by setting $\phi(f + ig) = \phi(f) + i\phi(g)$, for $f,g \in \text{Re}(S_r)$, one gets the extension whose existence is asserted in the statement of the theorem.

I can now finish the extension step. In Theorem 2.6, let *E* be the space of polynomials in **q**,**p** of degree 2k or less. The functional ϕ_0 is replaced by the moment functional μ defined by (1.12). The conclusion of the theorem then gives the following corollary.

Corollary 2.7: For every integer $k \ge 0$, there exists a functional μ_k that is defined and continuous on S_{2k} , that is non-negative on S_{2k}^+ , and that agrees with μ when restricted to the polynomials of degree 2k or less.

I close by remarking that μ_k will not usually be unique.

III. REPRESENTATIONS OF *†*-POSITIVE LINEAR FUNCTIONALS

The next step in showing that the condition given in Theorem 1.1 is sufficient for there to be a solution of the moment problem that I discussed in the Introduction is to get representations for the functionals arising in Corollary 2.7. These functionals have two properties that enable me to get representations for them. First, they are in the topological dual of S_t , $t \ge 0$, and, second, they are non-negative when applied to \hbar -positive functions in S_t . I will call any functional that satisfies these two conditions an \hbar -positive S functional.

One can always restrict an \hbar -positive S functional ϕ to functions in Schwartz space, for $\mathscr{S}(\Gamma)$ is included in S_t for all real t. Moreover, because this inclusion is continuous,⁵ $\phi|_{\mathscr{S}}$ is a continuous linear functional on \mathscr{S} ; that is, $\phi|_{\mathscr{S}}$ is a tempered distribution. Finally, $\phi|_{\mathscr{S}}$ is an \hbar -positive tempered distribution because it is non-negative on S_t^+ and, for every $F \in \mathscr{S}$, $F \circ \overline{F} \in \mathscr{S} \cap S_t^+$. In summary, I have obtained the following lemma.

Lemma 3.1: Every \hbar -positive S functional ϕ , when restricted to \mathcal{S} , is an \hbar -positive tempered distribution.

The restricted \hbar -positive S functional ϕ being a tempered distribution implies that its symplectic Fourier transform $\tilde{\phi}$ is, too. Here $\tilde{\phi}$ is defined in the usual way,^{11,12} via transpose. The version of the symplectic Fourier transform used here is the same as the one in Ref. 4,

$$\widetilde{A}(a) = (\pi \hbar)^{-n} \int A(z) e^{i\sigma(a,z)} dz, \qquad (3.1)$$

where $a \equiv (\mathbf{u}, \mathbf{v}) = (u_1, ..., u_n, v_1, ..., v_n)$. As is the case with the ordinary Fourier transform, the symplectic Fourier transform takes Schwartz functions to Schwartz functions, and tempered distributions to tempered distributions.

The \hbar positivity of ϕ is reflected in $\overline{\phi}$'s being of \hbar -positive type.⁹ This is easy to show. First, recall that¹³

$$\widetilde{A \circ B} = \widetilde{A} \times \widetilde{B}, \qquad (3.2)$$

where $\widetilde{A} \times \widetilde{B}$ is the twisted convolution of \widetilde{A} with \widetilde{B} and is given by

$$\widetilde{A} \times \widetilde{B}(a) = (\pi \hbar)^{-n} \int \widetilde{A}(b) \widetilde{B}(a-b) e^{i\sigma(b,a)} db .$$
 (3.3)

Second, note that

.

$$\widetilde{F}^{\dagger}(a) \equiv \overline{\widetilde{F}}(-a) = \widetilde{\widetilde{F}}(a)$$
. (3.4)

Finally, observe that

$$\widetilde{\phi}(\widetilde{F} \times \widetilde{F}^{\dagger}) = \widetilde{\phi}(\widetilde{F \circ F}) = \phi(F \circ \overline{F}) \ge 0, \qquad (3.5)$$

the last inequality on the right-hand side being a consequence of ϕ 's \hbar positivity.

So far, everything I have said would be true if ϕ were merely an \hbar -positive distribution rather than an \hbar -positive S functional. The main consequence of ϕ 's being an \hbar -positive S functional is that, for $A \in S_0$,

$$\phi(\|A\|_{w} \pm A) \ge 0, \qquad (3.6)$$

because, by Proposition 2.1, $||A||_{w} \pm A \in S_0^+$. Clearly, (3.6) implies that if also $A \in \mathscr{S}(\Gamma)$,

$$\left|\widetilde{\phi}(\widetilde{A})\right| = \left|\phi(A)\right| \leqslant \phi(1) \|A\|_{w} . \tag{3.7}$$

Using a proof somewhat similar to that of Proposition 2.1, one may show that

$$\|A\|_{w} \leq 2^{n} \|\widetilde{A}\|_{L^{1}}, \qquad (3.8)$$

and so

$$|\widetilde{\phi}(\widetilde{A})| \leq 2^n \phi(1) \|\widetilde{A}\|_{L^1} . \tag{3.9}$$

Finally, one may take limits in (3.9) to get that $\tilde{\phi}$ is a continuous, linear functional on L^{-1} . Because L^{∞} is the dual of L^{-1} , there is some function $\tilde{\Phi} \in L^{\infty}$ for which

$$\widetilde{\phi}(\widetilde{A}) = \int \widetilde{\Phi}(a)\widetilde{A}(a) \, da \,. \tag{3.10}$$

Since $\tilde{\phi}$ is an \hbar -positive type tempered distribution that is given by an L^{∞} function, one has, by Theorem 3.1 and Theorem 5.2 of Ref. 9, that there exists a function \tilde{G} such that $\tilde{G}\in L^2$, $\tilde{G}^{\dagger}=\tilde{G}$, and

$$\widetilde{\Phi}(a) = \widetilde{G} \times \widetilde{G}^{\dagger}(a) . \tag{3.11}$$

Unwinding the symplectic Fourier transforms then proves the following theorem.

Theorem 3.2: If ϕ is an \hbar -positive S functional, then there exists a real $G \in L^2$ such that

$$\phi(A) = \int A(z) G^{\circ} G(z) \, dz \tag{3.12}$$

for all $A \in \mathscr{S}(\Gamma)$.

I remark that examination of the proof of Theorem 3.2 shows that one also has the following result.

Corollary 3.3: If ϕ is an \hbar -positive tempered distribution that satisfies

$$|\phi(A)| \leqslant c \|\widetilde{A}\|_{L^{1}}, \qquad (3.13)$$

then, for all $A \in \mathscr{S}(\Gamma)$, ϕ has the form (3.12).

In a certain sense, Theorem 3.2 gives the representation that I need. The difficulty is that it does not contain enough information about G to be useful in getting sufficiency in Theorem 1.1. To overcome this difficulty, I will take a close look at the properties of G when the ϕ that gives rise to it is in S_t^* , the topological dual of S_t . Doing this requires looking at the H_t 's introduced in Sec. II.

The operator \hat{H} corresponding to H(z), which is given by (2.12), is the Hamiltonian for a system of *n* harmonic oscillators. The spectrum of \hat{H} consists of discrete positive eigenvalues, each being of finite multiplicity; ∞ is the only limit point. The eigenvectors ψ_k , k = 0, 1, 2, ..., may be chosen to be real-valued functions. Of course, they are Schwartz functions that comprise an orthonormal basis for $L^2(\mathbb{R}^n)$. I will label the eigenvalue corresponding to ψ_k by λ_k . (Since some eigenvalues are degenerate, one can have $\lambda_k = \lambda'_k$ for $k \neq k'$. This can happen for only a finite number of indices, however.)

Using the ψ_k 's, one may form Schwartz functions that comprise an orthonormal basis for $L^2(\Gamma)$. To do this, define¹⁴

$$\widehat{B}_{j,k} \equiv \psi_j \otimes \psi_k \; ; \tag{3.14}$$

these form an orthonormal basis for the space of Hilbert-Schmidt operators. By virtue of the unitary equivalence¹⁵ between the space of Hilbert-Schmidt operators and $L^2(\Gamma)$, the set $\{B_{j,k}\}$, where $B_{j,k}$ is the Wigner transform of $\hat{B}_{j,k}$, is an orthogonal basis for $L^2(\Gamma)$. One may make it orthonormal by dividing each $B_{j,k}$ by $(2\pi\hbar)^{n/2}$. That each $B_{j,k}$ is a Schwartz function follows from the ψ_j 's being products of Gaussians and Hermite polynomials, together with a standard argument from harmonic analysis.

My reason for introducing the $B_{j,k}$'s is that they diagonalize three different operators. First, define, for $A \in \mathscr{S}(\Gamma)$,

$$K_{+}(A) = H \circ A,$$

$$K_{-}(A) = A \circ H,$$

$$K_{0}(A) = \frac{1}{2}(H \circ A + A \circ H) = \frac{1}{2}(K_{+} + K_{-})(A).$$
(3.15)

Each of these operators has $B_{j,k}$ as an eigenvector; indeed, this is obvious from the following lemma.

Lemma 3.4: Let K_+ , K_- , and K_0 be defined by (3.15). Each operator is essentially self-adjoint on $\mathcal{S}(\Gamma)$, non-negative, and satisfies the following:

$$K_{+}(B_{j,k}) = \lambda_{j}B_{j,k}, \quad K_{-}(B_{j,k}) = \lambda_{k}B_{j,k}, K_{0}(B_{j,k}) = [(\lambda_{j} + \lambda_{k})/2]B_{j,k}.$$
(3.16)

Proof: All three cases are so similar that only one of them needs doing. I will work with the statements involving K_+ , and I will begin by getting (3.16).

Using the definition of $B_{j,k}$, note that these hold:

$$\widehat{K_{+}(B_{j,k})} = \widehat{H}\widehat{B}_{j,k}$$

$$= \widehat{H}(\psi_{j} \otimes \psi_{k})$$

$$= (\widehat{H}\psi_{j}) \otimes \psi_{k}$$

$$= \lambda_{j}\psi_{j} \otimes \psi_{k} = \lambda_{k}\widehat{B}_{j,k}.$$

Taking Wigner transforms gives (3.16) for K_+ .

Next I will show that K_+ is Hermitian on $\mathcal{S}(\Gamma)$. If A and B are Schwartz functions, then one has the following:

$$\langle K_{+}(A),B \rangle = \int H \circ A \,\overline{B} \, dz$$
$$= \int A \circ \overline{B} \circ H \, dz$$
$$= \int A \circ \overline{(H \circ B)} \, dz$$
$$= \langle A, K_{+}(B) \rangle .$$

I remark that if one sets A = B in the equations above, one also has that

$$\langle K_+(A), A \rangle = \int HA \circ \overline{A} dz$$
,

which is non-negative because H is \hbar positive. Thus K_+ is non-negative.

All that remains is essential self-adjointness. This amounts to showing that $\pm i$ are not eigenvalues of K_{+}^{*} (Ref. 16, pp. 98 and 99). If they were, one would have $A \in L^{2}(\Gamma)$ for which $(K_{+}^{*} \pm i)A_{+} = 0$. Thus

$$0 = \langle B_{i,k}, (K_{+}^{*} \pm i)A_{+} \rangle,$$

and so

$$0 = \langle (K_+ \pm i) B_{i,k}, A_{\pm} \rangle.$$

Since $K_{\pm}(B_{j,k}) = \lambda_j B_{j,k}$, and since λ_j is real, this equation implies that $\langle B_{j,k}, A_{\pm} \rangle = 0$. Thus $\{B_{j,k}\}$ being an orthogonal basis then implies that $A_{\pm} = 0$. Consequently, K_{\pm} is essentially self-adjoint.

The essential self-adjointness of these operators is important, for coupling it with (3.16) and the orthogonality of the basis $\{B_{j,k}\}$ yields the spectral resolution for all three operators. One may then use that spectral resolution to prove the next lemma.

Lemma 3.5: Let K_+ and K_- be defined by (3.15), let H_t be given by (3.13), and let Ω_β be given by (2.18). If A is a Schwartz function, then these hold for any real t and all $\beta > 0$:

$$K_{+}^{t}(A) = H_{t} \circ A, \quad K_{-}^{t}(A) = A \circ H_{t},$$

$$\exp(-\beta K_{+})A = \Omega_{\beta} \circ A, \qquad (3.17)$$

$$\exp(-\beta K_{-})A = A \circ \Omega_{\beta}.$$

Proof: For each equation, simply compute the inner product of both sides with each of the $B_{j,k}$, and observe that the results from the two sides are equal. This is sufficient to establish the equations in an L^2 sense, which is all I really need.

The operators K_+ , K_- , and K_0 all have fairly simple expressions in terms of partial differential operators. For example, if $A \in \mathscr{G}(\Gamma)$, then $K_+(A) = H \circ A$. But $H \circ A$ may be directly computed using Voros's Theorem 2.4.1,⁵

$$H \circ A = \sum_{j=1}^{n} \left\{ \frac{1}{2} (p_j^2 + q_j^2) A + \frac{i\hbar}{2} \left(q_j \frac{\partial A}{\partial p_j} - p_j \frac{\partial A}{\partial p_j} \right) - \frac{\hbar^2}{8} \left(\frac{\partial^2 A}{\partial p_j^2} + \frac{\partial^2 A}{\partial q_j^2} \right) \right\}.$$
(3.18)

Using $A \circ H = \overline{H \circ \overline{A}}$, one also has

$$A \circ H = \sum_{j=1}^{n} \left\{ \frac{1}{2} (p_j^2 + q_j^2) A - \frac{i\hbar}{2} \left(q_j \frac{\partial A}{\partial p_j} - p_j \frac{\partial A}{\partial q_j} \right) - \frac{\hbar^2}{8} \left(\frac{\partial^2 A}{\partial p_j^2} + \frac{\partial^2 A}{\partial q_j^2} \right) \right\}.$$
(3.19)

Putting these in vector notation and using them to compute $\frac{1}{2}(A \circ H + H \circ A)$, one easily gets these expressions for K_+ , K_- , and K_0 ,

$$K_{\pm} = \frac{1}{2} |z|^2 \pm \frac{i\hbar}{2} \left(\mathbf{q} \cdot \frac{\partial}{\partial \mathbf{p}} - \mathbf{p} \cdot \frac{\partial}{\partial \mathbf{q}} \right)$$
$$- (\hbar^2/8) \left(\Delta_p + \Delta_q \right), \qquad (3.20)$$
$$K_0 = \frac{1}{2} |z|^2 - (\hbar^2/8) \left(\Delta_p + \Delta_q \right).$$

Observe that K_0 is the quantum mechanical Hamiltonian for 2n decoupled harmonic oscillators. Here K_{\pm} are both perturbations of K_0 by an angular momentum term. While points in the spectrum of K_{\pm} have infinite multiplicity, points in the spectrum of K_0 have only finite multiplicity. Also, for every t > 0, the domain of K_0^t is a compact subset of $L^2(\Gamma)$.

With these comments about K_0 in mind, the significance of my characterization of the G that appears in (3.12) will be apparent.

Theorem 3.6: Let ϕ be an \hbar -positive S functional in the topological dual of S_{2t} , $t \ge 0$. If G is as in (3.12), then G belongs to the domain of $K_0^{t/2}$.

Proof: Since $H_t \in S_{2t}$, and Ω_{β} , which is given in (2.18), belongs to $\mathscr{S}(\Gamma)$, $H_t \circ \Omega_{\beta}$ also belongs to $\mathscr{S}(\Gamma)$.⁵ By Theorem 3.2, one has

$$\phi(H_t \circ \Omega_\beta) = \int (H_t \circ \Omega_\beta)(z) G \circ G(z) \, dz \,. \tag{3.21}$$

Using the associativity of the twisted product and (2.23), one may put (3.21) in this form,

$$\phi(H_t \circ \Omega_\beta) = \langle H_t \circ \Omega_\beta \circ G, G \rangle . \tag{3.22}$$

I want to get a bound for $\phi(H_t \circ \Omega_\beta)$, a bound that is independent of β . To do this, first note that one has the operator inequality

$$\widehat{H}^{\,\prime}-\widehat{H}^{\,\prime}e^{-\beta\widehat{H}}\geq 0\,.$$

In terms of phase-space functions, this inequality becomes

$$H_t - H_t \circ \Omega_\beta \geqslant 0. \tag{3.23}$$

Hence $H_t - H_t \circ \Omega_\beta \in S_{2t}^+$. The \hbar positivity of ϕ then implies that

$$\phi(H_t \circ \Omega_\beta) \leqslant \phi(H_t) , \qquad (3.24)$$
which is the bound I was looking for.

By approximating G with Schwartz functions, one may use Lemma 3.5 to show that

$$H_{t} \circ \Omega_{\beta} \circ G = K_{+}^{t} e^{-K_{+}\beta}(G) , \qquad (3.25)$$

which holds in an L^2 sense. Expand G in the basis $\{B_{i,k}\}$,

$$G = \sum_{j,k} \gamma_{j,k} B_{j,k} ,$$

$$\gamma_{j,k} = (2\pi\hbar)^{-n} \langle G, B_{j,k} \rangle .$$
(3.26)

Using (3.25), (3.26), and (3.16), one sees that

$$\langle H_{\iota} \circ \Omega_{\beta} \circ G, G \rangle = (2\pi\hbar)^{n} \left(\sum_{j,k} \lambda_{j}^{\iota} e^{-\beta\lambda_{j}} |\gamma_{j,k}|^{2} \right).$$
(3.27)

Putting (3.27) together with (3.22) and (3.24), one arrives at the inequality

$$\sum_{j,k} \lambda_j^t e^{-\beta \lambda_j} |\gamma_{j,k}|^2 \leq (2\pi\hbar)^{-n} \phi(H_t) , \qquad (3.28)$$

from which it immediately follows that

$$\sum_{j,k} \lambda_j^t |\gamma_{j,k}|^2 \leq (2\pi\hbar)^{-n} \phi(H_t) . \qquad (3.29)$$

The inequality in (3.29) is precisely the condition required for G to be in the domain of $K_{+}^{t/2}$; indeed,

$$\|K_{+}^{t/2}G\|^{2} = (2\pi\hbar)^{n} \left(\sum_{j,k} \lambda_{j}^{t} |\gamma_{j,k}|^{2}\right) \leq \phi(H_{t}) . \quad (3.30)$$

So far, I have not used the fact the G is a real-valued function. This reality is reflected in the $\gamma_{j,k}$'s obeying

$$\bar{\gamma}_{j,k} = \gamma_{k,j} . \tag{3.31}$$

To see this, first take the complex conjugate of both sides in the equation for $\gamma_{i,k}$,

$$\overline{\gamma}_{j,k} = (2\pi\hbar)^{-n} \langle \overline{G}, \overline{B}_{j,k} \rangle .$$
(3.32)

Since G is real, $G = \overline{G}$. On the other hand,

 $\overline{B}_{j,k} = \widehat{B}_{j,k}^* = (\psi_j \otimes \psi_k)^*, \qquad (3.33)$ from (3.14). But $(x \otimes y)^* = y \otimes x$,¹⁴ so

$$\widehat{B}_{j,k}^* = \psi_k \otimes \psi_j = \widehat{B}_{k,j} .$$
(3.34)

From (3.33) and (3.34), one gets that

$$\overline{B_{j,k}} = B_{k,j} . \tag{3.35}$$

Hence (3.32) may be written

$$\overline{\gamma}_{j,k} = (2\pi\hbar)^{-n} \langle G, B_{k,j} \rangle , \qquad (3.36)$$

from which (3.31) follows directly.

In the sum in (3.30), use $|\gamma_{j,k}| = |\overline{\gamma}_{j,k}| = |\gamma_{k,j}|$, and then interchange summation indices; this yields

$$\|K_{+}^{t/2}G\|^{2} = (2\pi\hbar)^{n} \left(\sum_{j,k} \lambda_{k}^{t} |\gamma_{j,k}|^{2}\right) \leq \phi(H_{t}) . \quad (3.37)$$

As before, this implies that G is in the domain of $K_{-}^{t/2}$, and that

$$\|K_{+}^{t/2}G\|^{2} = \|K_{-}^{k/2}G\|^{2} \leq \phi(H_{t}).$$
(3.38)

The final step in the proof begins with the observation that if a > 0, b > 0, and t > 0, then

$$2^{-t-1}(a^t+b^t) \leq ((a+b)/2)^t \leq a^t+b^t.$$
 (3.39)

Using (3.30), (3.37), and the right half of (3.39), one gets

$$(2\pi\hbar)^n \left(\sum_{j,k} \left(\frac{\lambda_j + \lambda_k}{2}\right)^t |\gamma_{j,k}|^2\right) \leq 2\phi(H_t) . \tag{3.40}$$

From (3.40), (3.16), and the argument that I just used, one again sees that G belongs to the domain of $K_0^{t/2}$. Indeed, putting together the various inequalities that one can arrive at from (3.30) and (3.38)–(3.40), I find that

$$2^{-\iota} \|K_{+}^{\iota/2} G\|^{2} \leq \|K_{0}^{\iota/2} G\|^{2} \leq 2\|K_{+}^{\iota/2} G\|^{2} \leq 2\phi(H_{\iota}).$$
(3.41)

In the course of proving the theorem, I have actually proved somewhat more than what was stated. I shall summarize these additional results below.

Corollary 3.7: With the notation and assumptions of Theorem 3.6, one has that G belongs to the domains of both $K_{+}^{1/2}$ and $K_{-}^{1/2}$.

This corollary will be a help in answering the following natural and important question: To what extent does G determine the functional ϕ ? Although I did not mention it in Theorem 3.2, G is unique if one also requires that it be \hbar positive, so, in a sense, ϕ uniquely determines G. The converse is almost true. It will turn out that if $\phi \in S_t^*$, then for all $A \in S_r$, r < t, $\phi(A)$ is uniquely determined by G. To prove this, I need this lemma.

Lemma 3.8: If $A \in S_t$, then for every $\beta > 0$, both $A \circ \Omega_\beta$ and $\Omega_\beta \circ A$ belong to $\mathscr{S}(\Gamma)$. Moreover, for every $\epsilon > 0$, both $A \circ \Omega_\beta$ and $\Omega_\beta \circ A$ converge, in the topology of $S_{t+\epsilon}$, to A as $\beta \to 0^+$.

Proof: The first statement follows from Voros's theorem 2.4.1.⁵ To get the rest of the lemma, all I really need to show is that, in S_{ϵ} , $\Omega_{\beta} \rightarrow 1$ as $\beta \rightarrow 0^+$; the statement about the convergence of $A \circ \Omega_{\beta}$ and $\Omega_{\beta} \circ A$ is simply a consequence of $\Omega_{\beta} \rightarrow 1$ and the continuity of the twisted product for Voros's symbols.⁵

To show that $\Omega_{\beta} \rightarrow 1$ in S_{ϵ} , note that

$$\partial^{\alpha} (\Omega_{\beta} - 1)(z) = \sum_{|\gamma| \le |\alpha|} c_{\gamma} z^{\gamma} \left[\tanh\left(\frac{\hbar\beta}{2}\right) \right]^{(|\gamma| + |\alpha|)/2} \Omega_{\beta}(z) , \quad (3.42)$$

where c_{γ} depends on α , γ , \hbar , but not on z or β . (Here, γ is a multi-index; it is not related to the $\gamma_{j,k}$'s used earlier.) For R > 0, one may use (3.42) together with a little calculus to show that

$$\sup_{|z|>R} |\partial^{\alpha} (\Omega_{\beta} - 1)(z)(1 + |z|^{2})^{(|\alpha| - \epsilon)/2}| \leq C(1 + R^{2})^{-\epsilon/2},$$
(3.43)

where C is independent of R, β . Since, on compact subsets of Γ , $\partial^{\alpha}(\Omega_{\beta} - 1)$ converges uniformly to 0 as $\beta \rightarrow 0^+$, one has, from (3.43) and (2.1), that

$$\lim_{\substack{\beta \to 0^+}} \sup \|\Omega_{\beta} - 1\|_{\alpha,\epsilon} \leq C(1+R^2)^{-\epsilon/2}.$$
(3.44)

Letting $R \to \infty$ then yields that $\|\Omega_{\beta} - 1\|_{\alpha,\epsilon} \to 0$ for all multiindices α , and so $\Omega_{\beta} \to 1$ in S_{ϵ} .

I can now prove the assertion that I made just prior to Lemma 3.8.

Proposition 3.9: Let $t \ge 0$, r < 2t, and suppose that the *n*-positive S functional ϕ is in S_{2t}^* . If $A \in S_r$, then

$$\phi(A) = \lim_{\beta \to 0^+} \left(\int A \circ \Omega_{\beta} G \circ G \, dz \right). \tag{3.45}$$

Proof: By Lemma 3.8, $A \circ \Omega_{\beta} \in \mathcal{S}(\Gamma)$, and so (3.12) holds:

$$\phi(A \circ \Omega_{\beta}) = \int (A \circ \Omega_{\beta})(z) G \circ G(z) \, dz \,. \tag{3.46}$$

Again by Lemma 3.8, $A \circ \Omega_{\beta} \to A$ in $S_{r+\epsilon}$ for all $\epsilon > 0$. With $\epsilon = 2t - r$, one sees that $A \circ \Omega_{\beta} \to A$ in S_{2t} :

$$\lim_{g\to 0^+} \phi(A \circ \Omega_\beta) = \phi(A) . \tag{3.47}$$

Combining (3.46) and (3.47) yields (3.45).

Thus G uniquely determines ϕ on all of the S_r's with r < 2t. Sad to say, G does not always determine ϕ on S_{2t}. The reason for this is related to the existence of certain functionals in the dual of the space of bounded operators on a Hilbert space. These functionals have the property that they are nonnegative, and they annihilate every compact operator; that is, they "live" on the Calkin algebra.¹⁷ Although I do not need to deal with them here, I will say a bit more about their effects later.

What I do need is to get an explicit formula for ϕ in terms of G when ϕ is restricted to S_r . Let $A \in \mathscr{S}(\Gamma)$. For the present, also suppose that $G \in \mathscr{S}(\Gamma)$. Under these assumptions, one has that

$$\int A(z)G \circ G(z)dz$$

$$= \int H_{t/2} \circ H_{-t/2} \circ A \circ H_{-t/2} \circ H_{t/2} \circ G \circ G dz$$

$$= \int H_{t/2} \circ T_{-2t}(A) \circ H_{t/2} \circ G \circ G dz$$

$$= \int T_{-2t}(A) \circ H_{t/2} \circ G \circ \overline{(H_{t/2} \circ G)} dz$$

$$= \int T_{-2t}(A) \circ K_{+}^{t/2}G(z) \overline{K_{+}^{t/2}G(z)} dz$$

$$= \langle T_{-2t}(A) \circ K_{+}^{t/2}G, K_{+}^{t/2}G \rangle . \qquad (3.48)$$

Here $T_{-2t}(A)$ is defined by (2.20).

If G is merely in the domain of $K_{+}^{t/2}$, one can show that (3.48) still holds. To do this, approximate G by a truncated version of (3.26),

$$G_N \equiv \sum_{j < N, \ k < N} \gamma_{j,k} B_{j,k} .$$
(3.49)

Here G_N is a finite sum of functions in \mathcal{S} , so it itself is in \mathcal{S} . Note that both G_N and $K_+^{1/2}G_N$ converge in L^2 to G and $K_+^{1/2}G$, respectively. Since the twisted product is L^2 continuous, ¹⁵ taking limits in

$$\int AG_N \circ G_N dz = \langle T_{-2t}(A) \circ K_+^{t/2}(G_N), K_+^{t/2}(G_N) \rangle$$

then yields (3.48) for any real G in the domain of $K_{+}^{t/2}$, and for any A in \mathcal{S} .

To get a formula for any A in S_r , r < 2t, just write (3.48) in terms of a trace; that is, recast (3.48) as

$$= \phi(A) = (2\pi\hbar)^n \operatorname{Tr}\left[\widehat{T}_{-2t}(A) \,\widehat{K}_{+}^{t/2}(G) \,\widehat{K}_{+}^{t/2}(G)^*\right].$$
(3.50)

In (3.50), replace A by $A \circ \Omega_{\beta}$, and use (3.45), Lemma 3.8, Theorem 2.5, and (1.7) to get the following proposition.

Proposition 3.10: Let ϕ , r, t, be as in Proposition 3.9. If $A \in S$, and G is as in (3.12), then

$$\phi(A) = (2\pi\hbar)^n \operatorname{Tr}\left[\widehat{T_{-2t}(A)} \,\widehat{K_{+}^{t/2}(G)} \,\widehat{K_{+}^{t/2}(G)}^*\right].$$
(3.51)

One should note that the right-hand side of (3.51) defines an \hbar -positive S functional on S_{2t} ; I will denote this functional by ϕ_G . Equation (3.51) shows that $\phi = \phi_G$ on all of the S_r 's, r < 2t. On the other hand, it is quite possible for $\phi \neq \phi_G$ on S_{2t} . For example, if ϕ_c is a positive, continuous linear functional on the bounded operators on $L^2(\mathbb{R}^n)$, and if ϕ_c annihilates the compact operators, then

$$\phi(A) = \phi_c(\widehat{T}_{-2\iota}(A)) + \phi_G(A) \tag{3.52}$$

defines an \hbar -positive S functional that differs from ϕ_G only for functions in S_{2t} ; on S_r , r < 2t, they agree because $\overline{T_{2t}(A)}$ turns out to be compact for $A \in S_r$.⁵

Suppose that one has another n-positive S functional ϕ' and that it satisfies the same conditions that I imposed on ϕ in Proposition 3.9. In addition, let G' be the functional that corresponds to ϕ' via (3.12). Using (3.51) and few facts¹⁴ concerning the relationships among $\|\cdot\|_{op}$, $\operatorname{Tr}(\cdot)$, and Hilbert-Schmidt norm $\|\cdot\|_{HS}$, one can show that for all $A \in S_r$, r < 2t, one has this inequality,

$$\phi(A) - \phi'(A) | \leq (2\pi\hbar)^{n} \| \widetilde{T}_{-2t}(A) \|_{op} \left[\| \widetilde{K}_{+}^{t/2} G \|_{HS} + \| \widetilde{K}_{+}^{t/2} G' \|_{HS} \right] \left\{ \| \widetilde{K}_{+}^{t/2} (G - G') \|_{HS} \right\}.$$
(3.53)

Putting this in terms of phase-space quantities, i.e., $\|\cdot\|_{op} = \|\cdot\|_{w}$, $(2\pi\hbar)^{n/2}\|\cdot\|_{HS} = \|\cdot\|_{L^{2}}$, one has the following result.

Corollary 3.11: Let $t \ge 0$, r < 2t, and suppose that both ϕ and ϕ' are \hbar -positive S functionals in S_{2t}^* , and that G and G' are the corresponding functions given in (3.12). If $A \in S_r$, then

$$\begin{aligned} |\phi(A) - \phi'(A)| \leq \|\widehat{T}_{-2t}(A)\|_{W} \left[\|\widehat{K}_{+}^{t/2}\widehat{G}\|_{L^{2}} + \|\widehat{K}_{+}^{t/2}\widehat{G'}\|_{L^{2}} \right] \|\widehat{K}_{+}^{t/2}(\widehat{G} - \widehat{G'})\|_{L^{2}}. \end{aligned}$$

$$(3.54)$$

What about the moment problem itself? I am now ready to finish proving that the conditions given in Sec. I are sufficient.

Proof of Theorem 1.1: In Sec. I, I showed that a necessary condition for the moment problem (1.2) to have a solution is that the moment functional be non-negative on every \hbar -positive polynomial P. If I assume, on the other hand, that $\mu(P) \ge 0$ for each \hbar -positive P, then by Corollary 2.7 there exists an \hbar -positive S functional $\mu_k \in S_{2k}^*$ such that $\mu_k(P) = \mu(P)$, when the degree of P is 2k or less. Theorems 3.2 and 3.6 imply that for each μ_k there exists a function G_k such that G_k belongs to the domain of $K_0^{k/2}$. Since, for $j = 0, \dots, \lfloor k/2 \rfloor, S_{4j}^* \supseteq S_{2k}^*, G_k$ belongs to the domain of K_0^0 (which is obvious anyway), and, from (3.41), satisfies

 $\int A(z)G\circ G\,dz$

 $||K_0^j G_k||_{L^2}^2 \leq 2\mu_k(H_{2j}).$

Because H_j is a polynomial of degree $4j \le 2k$, this inequality may be rewritten as

$$\|K_0^j G_k\|_{L^2}^2 \leq 2\mu(H_{2j}), \qquad (3.55)$$

which shows that the bound on the right-hand side above is independent of k as long as $k \ge 2j$.

Recall that the operator K_0 is the Hamiltonian for a system of 2n harmonic oscillators, and that K_0^{-1} is a compact operator. Since, for j = 0,1, (3.55) may be written as

$$\|K_0^{-1}(K_0G_k)\|_{L^2} \leq 2\mu(1), \quad \|K_0G_k\|_{L^2} \leq 2\mu(H_2),$$
(3.56)

and since an L^2 ball of fixed radius is weakly compact, one may extract from the sequence $\{K_0G_k\}$ a weakly convergent sequence. Finally, since compact operators map weakly convergent sequences to strongly convergent ones, the subsequence selected has the property that $K_0^{-1}K_0G_k$ converges strongly in L^2 to some real-valued function G. The inequalities in (3.56) imply

$$\|G\|_{L^2} \leq 2\mu(1) . \tag{3.57}$$

Repeat the process for the subsequence $\{G_{k_i}\}$, only now use

$$||K_0^{-1}K_0^2G_k|| \leq 2\mu(H_2)$$
, $||K_0^2G_k|| \leq 2\mu(H_4)$. (3.58)

(It may be necessary to discard the first term of the subsequence. This will not affect the argument at all.) One may again extract a subsequence for which $\{K_0^2 G_k.\}$ is weakly convergent, and so $\{G_k.\}$ and $\{K_0 G_k.\}$ are strongly convergent. A moment's reflection shows that again

$$G_{k} \rightarrow G$$
, $K_0 G_k \rightarrow K_0 G$,

and

$$||K_0G||_{L^2} \leq 2\mu(H_2)$$
.

This process may be repeated indefinitely. Doing so shows that G is in the domain of K_0^j for all $j \ge 0$, and that

$$\|K_0^j G\|_{L^2} \leq 2\mu(H_j) . \tag{3.59}$$

It is well known that this implies that G must belong to $\mathcal{S}(\Gamma)$.¹⁸

I now want to show that $\rho = G \circ G$ solves the moment problem. Define the functional

$$\mu_G(A) = \int A(z) G^{\circ}G(z) dz . \qquad (3.60)$$

Since G is a real-valued Schwartz function, μ_G is an \hbar -positive S functional, and it belongs to S_i^* for all t. From the way I constructed G, I can pick a subsequence $\{G_{k'}\}$ that has the property that $\{G_{k'}\}, \{K_0G_{k'}\},...,\{K_0^jG_{k'}\}$ all converge in L^2 to G, $K_0G,...,K_0^{j+1}G_{k'}$, respectively. One may also require that $2k' \ge j + 1$, for doing so merely discards a finite number of terms in the sequence.

Let P be any polynomial of degree less than 2j + 2; obviously, P belongs to S_{2j+1} . In Corollary 3.11, put $\phi' = \mu_k$, and $\phi = \mu_G$, note that $\mu_{k'}(P) = \mu(P)$, and use the bounds (3.55) and (3.59) in (3.54). The inequality (3.54) then becomes

 $|\mu(P) - \mu_G(P)|$

$$\leq 2 \|T_{-4j-4}(P)\| \sqrt{\mu(H_{j+1})} \|K_0^{j+1}(G-G_{k'})\|_{L^2} .$$
(3.61)

As $k' \rightarrow \infty$ the right-hand side vanishes. The left-hand side is, however, independent of k' and so is 0 to begin with. Thus

$$\mu(P) = \mu_G(P) \tag{3.62}$$

for all polynomials of degree 2j + 1 or less. Finally, since the construction may be carried out for any j, one sees that (3.62) holds for all polynomials. If one sets $\rho = G \circ G$ and takes P to be any of the monomials in (1.3), then one see that (1.2) holds. [I remark that j above, unlike the j in (1.3), is just an integer; j in (1.3) is a multi-index.] This completes the proof.

The methods used above also yield sufficient conditions for the solution of a truncated moment problem in which one knows moments up to some fixed order.

Corollary 3.12: Suppose that one is given a set of numbers $m_{j,k}$, with $|j| + |k| \le 2\nu + 2$, ν being a non-negative integer. A sufficient condition for the $m_{j,k}$'s to be of the form (1.2) for $|j| + |k| \le 2\nu + 1$ is that the moment functional $\mu(P)$ be non-negative for every \hbar -positive polynomial P having degree $2\nu + 2$ or less.

Proof: Just apply Corollary 2.7, Theorem 3.2, and Proposition 3.10 to get a G for which (1.2) holds, at least in the sense of (3.51), for all polynomials of degree $2\nu + 1$ or less. One runs into difficulty with reproducing the highest order moments because the functional involved may have a term like ϕ_c in (3.52).

IV. CONCLUDING REMARKS

As I mentioned in Sec. I, O'Connell and I gave a set of necessary conditions for a sequence $\{m_{i,k}\}$ to be of the form (1.2)² If one examines that set, one easily sees that it is precisely the set of conditions obtained by requiring that the moment functional μ be non-negative on all \hbar -positive polynomials of the form $P \circ \overline{P}$. Whether this set is sufficient as well as necessary obviously hinges on whether every h-positive polynomial may be written as a sum of polynomials of the form $P \circ \overline{P}$. In the classical, one-dimensional Hamburger moment problem, the analogous question would be whether every non-negative polynomial in one variable can be written as the sum of squares of polynomials. The answer is, of course, yes, and the set of conditions analogous to those of Ref. 2 are indeed sufficient.¹⁹ For the classical, multidimensional problem, the answer is no,^{20,21} and the corresponding set of conditions is not sufficient.²² On the basis of the answer in the classical, multidimensional case, I conjecture that the conditions given in Ref. 2 are not sufficient. I do so with some hesitation, for there are many differences between the quantum mechanical moment problem treated here and the classical, multidimensional moment problem.

The set of restrictions imposed on the moments by requiring that the moment functional μ be non-negative for *n*positive polynomials has an interesting physical interpretation. If one sets

$$P = c_0 + \mathbf{c}_1 \cdot \mathbf{p} + \mathbf{c}_2 \cdot \mathbf{q} \,, \tag{4.1}$$

where c_0 is a complex number, and c_1 , c_2 are in \mathbb{C}^n , then $\mu(P \circ \overline{P}) \ge 0$ gives rise to a quadratic form. Applying the stan-

dard conditions for this form to be non-negative results in the position-momentum uncertainty relations.² If one replaces P in (4.1) by an arbitrary polynomial, then $\mu(P \circ \overline{P}) \ge 0$ is a non-negative quadratic form in the coefficients of P. The restrictions imposed on the moments in this case form a set of generalized uncertainty relations.² Should the conjecture I made above prove true, then there will be \hbar polynomials that are not sums of polynomials of the form $P \circ \overline{P}$. The moment functional μ applied to such polynomials will give rise to additional restrictions on the moments, and so will augment that set of generalized uncertainty relations. In any case, one will arrive at a "complete set" of uncertainty relations for the moments. [By this I mean a set of restrictions satisfied if and only if the $m_{i,k}$'s are of the form (1.2). There are, of course, other uncertainty relations; see Breitenberger's paper²³ for a discussion.]

There are several interesting, unanswered questions concerning the quantum mechanical moment problem discussed here. The first is related to what polynomials are actually \hbar positive. Voros⁵ points out that every semibounded function in S_m gives rise to a semibounded operator on $L^2(\mathbb{R}^n)$. In particular, every non-negative polynomial in **q** and **p** can be made into an \hbar positive one just by adding a positive constant to it. Is the converse also true? Put another way, does every \hbar -positive polynomial differ from a nonnegative polynomial by a constant? If the answer is yes, then the difference between the quantum mechanical moment problem and the classical one is small indeed. The answer to this question might also shed light on the difference between states in quantum mechanics and in classical mechanics.

In proving sufficiency in Theorem 1.1, I merely showed that an appropriate Wigner function ρ existed. Is this function unique? If it is, the moments would serve as a kind of "minimal set" of expectation values required to determine ρ . Also, whether the solution is or is not unique, is there any method—say, some kind of variational procedure—for constructing ρ ? Finally, is there a way of constructing solutions to the truncated moment problem mentioned in Corollary 3.12?

ACKNOWLEDGMENTS

I wish to thank Professor Peter D. Lax for the many helpful and interesting conversations that I had with him. In addition, I wish to thank Professor Ernst Breitenberger for pointing out Ref. 23, and Professor Stephen A. Fulling and Professor Joseph D. Ward for helpful suggestions. Part of the work involved in this paper was done while I was on sabbatical. I would like to thank the Courant Institute and the Ohio State University Mathematics Department for their hospitality, and Texas A&M's Association of Former Students for its support.

- ¹J. E. Moyal, "Quantum mechanics as a statistical theory," Proc. Cambridge Philos. Soc. **45**, 99 (1949).
- ²F. J. Narcowich and R. F. O'Connell, "Necessary and sufficient conditions for a phase-space function to be a Wigner distribution," Phys. Rev. A **34**, 1 (1986).
- ³F. J. Narcowich, "A quantum mechanical moment problem," in *The Physics of Phase Space, Lecture Notes in Physics*, Vol. 278, edited by Y. S. Kim and W. W. Zachary (Springer, Berlin, 1987).
- ⁴F. J. Narcowich, "A Dyson-like expansion for solutions to the quantum Liouville equation," J. Math. Phys. 27, 2502 (1986).
- ⁵A. Voros, "An algebra of pseudodifferential operators and the asymptotics of quantum mechanics," J. Funct. Anal. 29, 104 (1978).
- ⁶L. Hörmander, *The Analysis of Linear Patial Differential Operators III* (Springer, Berlin, 1985).
- ⁷I. Daubechies, "On the distributions corresponding to bounded operators in the Weyl quantization," Commun. Math. Phys. **75**, 229 (1980).
- ⁸M. Hillary, R. F. O'Connell, M. O. Scully, and E. P. Wigner, "Distribution functions in physics: Fundamentals," Phys. Rep. **106**, 121 (1984).
- ⁹F. J. Narcowich, "Distributions of *ħ*-positive type," preprint.
- ¹⁰R. E. Edwards, *Functional Analysis* (Holt, Rinehart and Winston, New York, 1965).
- ¹¹F. G. Friedlander, *Introduction to the Theory of Distributions* (Cambridge U. P., Cambridge, 1982).
- ¹²F. Treves, *Topological Vector Spaces, Distributions, and Kernels* (Academic, New York, 1967).
- ¹³A. Grossmann, G. Loupias, and E. M. Stein, "An algebra of pseudodifferential operators and quantum mechanics in phase space," Ann. Inst. Fourier (Grenoble) 18, 343 (1968).
- ¹⁴J. R. Ringrose, Compact Non-Self-Adjoint Operators (Van Nostrand, Reinhold, London, 1971).
- ¹⁵J. C. T. Pool, "Mathematical aspects of the Weyl correspondence," J. Math. Phys. 7, 66 (1966).
- ¹⁶E. Nelson, *Topics in Dynamics I: Flows* (Princeton U. P., Princeton, NJ, 1970).
- ¹⁷J. W. Calkin, "Two-sided ideals and congruences in the ring of bounded operators in Hilbert space," Ann. Math. 42, 839 (1941).
- ¹⁸B. Simon, "Distributions and their Hermite expansions," J. Math. Phys. 12, 140 (1971).
- ¹⁹J. A. Shohat and J. D. Tamarkin, *The Problem of Moments* (American Mathematical Society, New York, 1943).
- ²⁰D. Hilbert, "Über die darstellung definiter formen als summen von formenquadraten," Math. Ann. **32**, 342 (1888).
- ²¹A. Lax and P. D. Lax, "On sums of squares," Linear Algebra Appl. 20, 71 (1978).
- ²²C. Berg, J. P. R. Christensen, and P. Ressel, Harmonic Analysis on Semigroups (Springer, Berlin, 1985).
- ²³E. Breitenberger, "Uncertainty measures and uncertainty relations for angle observables," Found. Phys. 15, 353 (1985).

Composite systems in quaternionic quantum mechanics

C. G. Nash and G. C. Joshi

School of Physics, University of Melbourne, Parkville, Victoria, 3052, Australia

(Received 3 November 1986; accepted for publication 30 July 1987)

The natural composition of systems in quaternionic quantum mechanics is examined via their lattices of propositions and it is shown that the criticisms that have been made of such a composition are unconvincing.

I. INTRODUCTION

Complex quantum mechanics is at the center of much of modern physics but in spite of this there remain unanswered fundamental questions about why it is preferred above alternative quantum theories. The need for an understanding of this preference is made more pertinent because of the important place phase invariance has taken in particle physics. Quaternionic quantum mechanics, the most prominent alternative, is considered here.

To quote Finkelstein *et al.* on quaternionic quantum mechanics, "The principal conceptual difficulty concerns the theory of composite systems where the ordinary tensor product fails due to noncommutativity."¹ Many authors working in the field since have agreed.²⁻⁵ However, it is in response to this difficulty that Finkelstein *et al.* and the subsequent literature have diverged. While Finkelstein *et al.* considered that a sensible theory could be arrived at despite the requirement that the composite Hilbert space be quaternionic, the rest of the literature has asserted either that one must give up quaternionic quantum mechanics² or give up a quaternionic composite Hilbert space.^{3,4}

What we hope to do here is to put forward a theory of composition in quaternionic quantum mechanics following Finkelstein *et al.* and then to show that the criticisms of such a program that are found in the literature are unconvincing.

The notions of quaternionic Hilbert space used here will be those of Horwitz and Biedenharn.⁴

II. QUATERNIONIC COMPOSITION

Following Aerts,⁶ Zecca,⁷ and Finkelstein et al.,⁸ the composition of distinguishable quantum systems should be viewed as a lattice theoretic problem. Then we ask: given two quantum systems S_1 and S_2 and their corresponding lattices of propositions and sets of states $(\mathcal{L}_1, \mathcal{S}_1)$ and $(\mathcal{L}_2, \mathcal{S}_2)$, what quantum systems S, $(\mathcal{L}, \mathcal{S})$ may be considered as containing both S_1 and S_2 ? That is, what does it mean for one quantum system to contain two others? In line with the above three authors we are going to say it means that if you could ask a question of system 1 (e.g., the particle, that system 1 is, has a momentum p) as a system by itself then you can ask the same question of the composite system (e.g., that particle 1 has momentum p) and similarly for system 2. Mathematically this may be interpreted as the assertion that there exist mappings $h_i: \mathcal{L}_i \to \mathcal{L}$, for i = 1, 2, sending propositions of the subsystems to the corresponding, "the same," propositions in the composite system. This sameness of the elements in h_i (\mathcal{L}_i) and the elements in \mathcal{L}_i implies also that h_i preserves the structure of \mathcal{L}_i . We will call the existence of h_i and the preservation of the structure of \mathcal{L}_i by h_i Axiom 1.

One might describe what has been done as finding a natural and directly interpretable axiom deductable from physically plausable assumptions, which is something that is always useful to do.⁹

It may be that this axiom is too strong to include all systems that we might like to call composite. Composition could possibly be better, or more generally, described via states or via maps from \mathcal{L} to \mathcal{L}_i . However, what we are trying to do here is develop a definitive notion of composition. Axiom 1 will suffice.

Having asserted Axiom 1 we need to consider the relationship between h_1 and h_2 that is the relationship between the two components inside the composite. If we were to describe two completely independent subsystems then we would be able to give very strong requirements governing the relationship between h_1 and h_2 as do Aerts⁶ and Aerts and Daubechies.¹⁰ However, we have traditionally considered a wide range of systems as composite. Systems such as two balls on a string, the hydrogen atom, and three quarks bound by gluons are all different from the sum of the components acting independently.

If we follow wave mechanics where the position observables Q_1 and Q_2 are singled out as multiplication by x_1 and x_2 , respectively, then there is a natural composite Hilbert space and a natural relationship between h_1 and h_2 . For the case of one-dimensional particles the component Hilbert spaces are $L^2_Q(\mathbb{R})_1$ and $L^2_Q(\mathbb{R})_2$, the Lebesgue square integrable functions from R to Q, and the natural Hilbert space for the composite system, a system with two degrees of freedom, is $L^2_Q(\mathbb{R}^2)$. The lattice mappings h_1 and h_2 that Axiom 1 requires become mappings of closed subspaces of $L^2_Q(\mathbb{R})_i$ to closed subspaces of $L^2_Q(\mathbb{R}^2)$ for i = 1, 2. Though it is not necessary it is both convenient and natural for these to take the form

$$h_{1}: S_{1} \subseteq L_{Q}^{2}(\mathbb{R})_{1} \rightarrow \{\psi_{1} \cdot \chi_{2}: (x_{1}, x_{2}) \\ \rightarrow \psi_{1}(x_{1})\chi_{2}(x_{2}) | \psi_{1} \in S_{1}, \ \chi_{2} \in L_{Q}^{2}(\mathbb{R})_{2} \}, \\ h_{2}: S_{2} \subseteq L_{Q}^{2}(\mathbb{R})_{2} \rightarrow \{\psi_{2} \cdot \chi_{1}: (x_{1}, x_{2}) \\ \rightarrow \psi_{2}(x_{2})\chi_{1}(x_{1}) | \psi_{2} \in S_{2}, \ \chi_{1} \in L_{Q}^{2}(\mathbb{R})_{1} \}.$$

$$(1)$$

Observables, as lattice-valued measures, of the component systems may be associated with observables of the composite system via h_1 and h_2 . If O an observable of system *i* maps E, a

Borel subset of R, to O(E) an element of \mathcal{L}_i , then the corresponding composite observable maps E to $h_i(O(E))$. Now as the spectral theorem applies unchanged to quaternionic Hilbert spaces we may express this association in terms of Hermitian operators. To do this we must first note that for all Φ an element of $L^2_Q(\mathbb{R}^2)$, $\Phi(x_1, x_2)$, may be written as a sum,

$$\sum_i \psi_i^1(x_1)\psi_i^2(x_2),$$

where $\psi_i^j \in L_Q^2(\mathbb{R})_j$ just as with the complex case. With this expression of elements of $L_Q^2(R)^2$ an operator A_1 of subsystem 1 can be associated with an operator A of the composite system via

$$(A\Phi)(x_1,x_2) = \sum_i (A_1\psi_i^1)(x_1)\psi_i^2(x_2).$$
(2)

This operator association sends projections onto S_1 , a subspace of $L_Q^2(\mathbb{R})_1$ to projections onto $h_1(S_1)$ so that it corresponds to the association of lattice-valued measures given in Eq. (1). A similar association may be made between an operator A_2 of $L_Q^2(\mathbb{R})_2$ and a composite operator A,

$$(A\Phi)(x_1,x_2) = \sum_i (A_2\phi_i^2)(x_2)\phi_i^1(x_1), \qquad (3)$$

which agrees with h_2 , where

$$\sum_i \phi_i^2(x_2) \phi_i^1(x_1)$$

is an expansion of $\Phi(x_1, x_2)$ with elements of $L_Q^2(\mathbb{R})_2$ leading. In the composition given in Eq. (1) Q_1 commutes with all the observables of system 2 and similarly Q_2 with those of system 1. This is because both Q_1 and Q_2 send real functions to real functions: For i = 1, 2 ($Q_i \Psi$) $(x_1, x_2) = x_i \psi(x_1, x_2)$. On the other hand, the momentum observables P_1 and P_2 do not in general commute with the observables of system 2 or system 1, respectively, in the way that Q_1 and Q_2 do. This is because P_1 is of the form $J_1(\partial/\partial x_1)$ with J_1 an anti-Hermitian unitary operator which commutes with $(\partial/\partial x_1)$ and J_1 defined from the time translation operator in system 1. And J_2 is similarly related to P_2 (Refs. 11 and 12). Here J_1 does not in general send real functions to real functions. This quaternionic nature of J_1 will neither allow it nor P_1 to commute with all the observables of system 2.

This whole discussion could have been carried out just as well with the component systems displayed in momentum space.¹¹ The Hilbert spaces would be the same function spaces but the association between Hermitian operators and physical observables would be different and h_1 and h_2 would not be given by Eq. (1). Here momentum observables P_1 and P_2 would have the form $(P_i\Psi)(p_1, p_2) = p_i\Psi(p_1, p_2)$ in the composite system. So P_1 would commute with all the observables of system 2 and similarly for P_2 . The two compositions, one based on position and the other on momentum, are not equivalent. Finkelstein et al.3,8 have indicated that each such representation of the component systems in terms of function spaces is a definition of what observables are real. So the choice of representation determines which observables of system 1 commute with which observables of system 2. So the choice determines the composite system. Which observables commute has direct physical consequences

when observables are measured simultaneously. So in situations without superselection rules the choice of composition is determined by experiment.

We have shown in Ref. 11 that for quanternionic quantum mechanics (QQM) to be consistent with the remarkably successful description of the world provided by complex quantum mechanics (CQM) it must be that CQM is embedded in QQM. This has the effect that in the wave mechanics representation of system 1 $(J_1\phi_1)(x_1) = q_1\phi_1(x_1)$ for all $\phi_1 \in L^2_O(\mathbb{R})_1$ for some fixed q_1 a pure imaginary quaternion of unit length. Similarly $J_2\phi_2(x_2) = q_2\phi_2(x_2)$ for all $\phi_2 \in L^2_{Q}(\mathbb{R})_2$ for some fixed q_2 a pure imaginary quaternion of unit length. With this restriction of the J's a complex quantum mechanics can be seen to be contained in each of the component systems.¹¹ Complex is defined for observables, states, and scalars via commutation with q_1 and q_2 in system 1 and system 2, respectively. Similarly we expect that "complex" may be defined for the composite system or may be induced from the components. The success of complex quantum mechanics, especially in its use of isolated systems, is strong evidence for the complex observables of system 1 to commute with the complex observables of system 2 at least when operating on complex composite states, which is sufficient to ensure that they do commute. It might be, however, that the commutator of complex observables from different systems is not zero but just small when acting on complex states of the composite system. We believe it would need to be vanishingly small and will not consider the possibility here.

The above requirement that the complex observables of different systems commute restricts the possible representations of the component systems. The wave mechanic representations of systems 1 and 2 must be such that $q_1 = q_2 = q$. Otherwise, in particular, $[P_1, P_2] \neq 0$. Similarly for the momentum representations where $Q_1 = q_1(\partial/\partial p_1)$ and $Q_2 = q_2(\partial/\partial p_2)$ it must be that $q_1 = q_2 = q$.

What is still left ill defined is the relationship between observables of the two systems when one of the observables is not complex. In particular, which if any complex observable of system 1 commutes with all observables of system 2. That this ambiguity remains is not so much the result of an ill defined composition procedure but the result of the hiddenness of quaternion quantum mechanics, since as we have said the choice of composition should be able to be determined by experiment. No measurement has ever been identified as being of a noncomplex observable nor of a noncomplex state. It is with such measurements that the precise nature of composition in quaternionic quantum mechanics must be determined. It also behooves quaternionic quantum mechanics to explain the lack of such experimental evidence, but we will avoid this issue here.

III. ARGUMENTS AGAINST QUATERNIONIC COMPOSITION

To our knowledge there have been two arguments given in the literature purporting to show that the composition of two subsystems described by quaternionic Hilbert spaces cannot in turn be described by a quaternionic Hilbert space. One based upon lattice theoretic considerations is given by Aerts^{2,6} and the other, based on Hilbert space ideas which mirror Aerts's work somewhat, is put forward by Rembielinski.³

Aerts describes a system S as composed of systems S_1 and S_2 if the three following conditions are satisfied.

(1) The structure of S_1 and S_2 is preserved when considered as parts of S.

(2) A measurement on one of the systems S_1 or S_2 does not disturb the other one.

(3) When we know the state of S_1 and of S_2 then we know also the state of S.

He then interprets these as the following lattice theoretic conditions.

(1') There must exist maps $h_1: \mathcal{L}_1 \to \mathcal{L}$ and $h_2: \mathcal{L}_2 \to \mathcal{L}$, where h_1 and h_2 conserve the structure of \mathcal{L}_1 and \mathcal{L}_2 .

(2') If $a_1 \in \mathcal{L}_1$ and $a_2 \in \mathcal{L}_2$, then $h_1(a_1)$ is compatible with $h_2(a_2)$.

(3') If a_1 is an atom of \mathcal{L}_1 and a_2 is an atom of \mathcal{L}_2 then $h_1(a_1) \wedge h_2(a_2)$ is an atom of \mathcal{L} .

Then in the language of generalized Hilbert spaces which represent \mathcal{L}_1 , \mathcal{L}_2 , and \mathcal{L} Aerts proves some very general theorems asserting the existence of a bilinear (σ_1, σ_2) isometry from the component vector spaces to the composite vector space which preserves the composition provided by h_1 and h_2 . These theorems have as their corollary that the Hilbert spaces representing each of the lattices may not all be over the quaternions.

That Aerts should come to such a conclusion is clear from the assumptions he makes. To show this we follow Finkelstein *et al.*⁸ and note for $\psi_1 \in L_Q^2(\mathbb{R})_1$ and $\psi_2 \in L_Q^2(\mathbb{R})_2$ that

$$h_{1}(\{\psi_{1}\alpha: \alpha \in Q\}) \land h_{2}(\{\psi_{2}\beta: \beta \in Q\}) \\ = \{\psi_{1}\cdot\chi_{2}: (x_{1},x_{2}) \rightarrow \psi_{1}(x_{1})\chi_{2}(x_{2}) | \chi_{2} \in L_{Q}^{2}(\mathbb{R})_{2} \} \\ \land \{\psi_{2}\cdot\chi_{1}: (x_{1},x_{2}) \rightarrow \psi_{2}(x_{2})\chi_{1}(x_{1}) | \chi_{1} \in L_{Q}^{2}(\mathbb{R})_{1} \}$$

$$(4)$$

is nonempty if and only if there exist quaternions a and b such that $[\psi_1(x_1)a, \psi_2(x_2)b] = 0$ for all x_1 and x_2 . As $\{\psi_1 \alpha: \alpha \in Q\}$ and $\{\psi_2 \beta: \beta \in Q\}$ are atoms of their respective lattices then the possibility of the above meet being null is in contradiction with 3'.

The assumptions 1,2,3 and their lattice theoretic interpretations 1', 2', 3' are not only inconsistent with quaternionic composites but are similarly inconsistent with some simple complex compositions, in particular the complex quantum mechanics of two hard spheres and the quantum mechanics of two positively charged particles. Just as in quaternionic composition in the complex composition there are pairs of atomic propositions, namely, one from the quantum mechanics of one hard sphere (considered as a separate system) and the other from the quantum mechanics of the other hard sphere, for which the meet of the images of these propositions in the composite system is null. These are those pairs in which the two spheres are required to occupy some volume in common. The quaternionic and the hard sphere compositions are similar in that they both satisfy 1, 1', and 3 for pure states but do not satisfy 2, 2', and 3'. That they satisfy 1 and 1' is clear. For 2 we note that most composite systems do not satisfy 2 in an unqualified way because there are always states of the composite system in which the component systems are correlated. The more specific and much less stringent requirement 2' falls foul of both compositions because in each case there exist these propositions of each of the component systems whose meet in the composite system is null and such pairs can be found whose images in the composite system are not orthogonal. This is sufficient to prevent compatibility. For 3 and 3' we see that when the meet of the images, in the composite system, of two atoms, one from each of the component systems, is not null then the meet is a uniquely determined atom of the composite system. So whenever we know what pure states S_1 and S_2 are in, and this is consistent with the two systems being composed together, then we know the state of S. In this way both compositions satisfy 3 but not 3'.

Rembielinski requires that composition be described by a vector product $f^1 \times f^2 \times \cdots \times f^r$ with f^k an element of the k th Hilbert space. This vector product is required to obey the distributivity condition $f^1 \times f^2 \times \cdots \times (g^k + h^k) \times \cdots \times f^r$ $= f^1 \times f^2 \times \cdots \times g^k \times \cdots \times f^r + f^1 \times f^2 \times \cdots \times h^k \times \cdots$ $\times f^r$ for every k = 1 to r. He notes that states are determined by rays and so requires $f^1 \times f^2 \times \cdots f^k \alpha \times \cdots \times f^r = (f^1 \wedge f^2)$ $\times f^2 \times \cdots \times f^k \times \cdots \times f^r \alpha^{\tau_k}$ for all k, where α is any element of the field of scalars of the component space and α^{C_k} is some element of the field of scalars of the product space. He is then able to show that the scalars of the component spaces commute, thus discounting Q. His argument depends crucially upon this relationship between rays of the component Hilbert spaces and those of the composite Hilbert space. In particular, the fact that the mapping τ_k is independent of f^1 to f'. What we have considered in Sec. II is an alternative relationship, one not essentially based on Hilbert spaces but on lattices, so we have sidestepped his objection by never requiring the composition to be put in terms of individual vectors. It is interesting to note that the implementation of composition by a vector product is precisely what Aerts's theorems assure, so Rembielinski and Aerts have come to the same conclusions somewhat from the same direction.

²D. Aerts, in *Current Issues in Quantum Logic*, edited by E. C. Beltrametti and B. C. van Fraassen (Plenum, New York, 1981).

- ⁴L. P. Horwitz and L. C. Biedenharn, Ann. Phys. (NY) 157, 432 (1984).
- ⁵S. L. Adler, Proceedings of the Tenth Hawaii Topical Conference on High
- Energy Physics (University of Hawaii Press, Honolulu, 1985).

- ⁷A. Zecca, J. Math. Phys. 19, 1482 (1978).
- ⁸D. Finkelstein, J. M. Jauch, and D. Speiser, CERN Report 59-7, 1959.
- ⁹G. W. Mackey, *Mathematical Foundations of Quantum Mechanics* (Benjamin, Reading, MA, 1968).

¹D. Finkelstein, J. M. Jauch, S. Schiminovich, and D. Speiser, J. Math. Phys. **3**, 207 (1962).

³J. Rembielinski, J. Phys. A 14, 2609 (1981).

⁶D. Aerts, J. Math. Phys. 21, 778 (1980).

¹⁰D. Aerts and I. Daubechies, Helv. Phys. Acta 51, 637 (1978).

¹¹C. G. Nash and G. C. Joshi, University of Melbourne preprint UM-P-86/ 73 (unpublished).

¹²P. Truini, L. C. Biedenharn, and G. Cassinelli, Hadronic J. 4, 981 (1981).

Component states of a composite quaternionic system

C. G. Nash and G. C. Joshi

School of Physics, University of Melbourne, Parkville, Victoria, 3052, Australia

(Received 3 November 1986; accepted for publication 30 July 1987)

The problem of finding component states given a composite state is examined for quaternionic quantum mechanics. It is shown that under very loose conditions the component state is forced to be complex.

I. INTRODUCTION

The problem of composing systems in quaternionic quantum mechanics (QQM) is one with many suggested solutions. Elsewhere¹ we have shown that the criticisms leveled against the natural solution that was first mooted by Finkelstein *et al.*² are insufficient to dismiss it and prompt a search for alternative and more elaborate solutions. It has become apparent $^{2-4}$ that the way to approach the problem is via the propositional lattices of the component systems and their embedding within that of the composite system. However, previous work^{1,2} has only gone so far as to establish this embedding; what has not been discussed is the relationship between the states of the composite system and those of the component subsystems. This is what we propose to do here. In order to make the embedding and subsequent analysis more concrete we will discuss the composition of two onedimensional quantum wave mechanical systems. That is, $L_{O}^{2}(\mathbb{R})_{1}$ and $L_{O}^{2}(\mathbb{R})_{2}$ will describe the component systems and $L_{O}^{2}(\mathbb{R}^{2})$ will describe the composite system. We will refer to these Hilbert spaces as H_1 , H_2 , and H, respectively, throughout this paper. These are of course sets of Lebegue square integrable functions from variously \mathbb{R} and \mathbb{R}^2 to Qwith the usual integral inner product. There is no preference for system 1 over system 2 as a component system of the composite system, so to avoid tiresome rewriting of results which focus upon one component system or upon the component systems differently, we note that in all that follows the subscripts or superscripts 1 and 2 may be interchanged.

Fortunately a number of useful structures carry over from complex quantum mechanics (CQM). We know that the states of these systems correspond to density operators acting on the corresponding Hilbert spaces just as in CQM because Gleason's theorem applies unchanged to quaternionic Hilbert spaces.^{5,6} Also the spectral theorem remains unchanged⁶ so observables are self-adjoint operators and may be expanded in terms of projection operators. Similarly the relationship between pure and mixed states is preserved, the latter as convex combinations of the former. Finally as an upshot of all this the probability that a proposition P_M , a projection onto the subspace M, is true when the system is in the state D, a density operator, is $tr(DP_M)$.

The superscripts and subscripts 1,2, or nothing will denote things pertaining to system 1, system 2, or the composite system, respectively, in the following way: functions will be superscripted and inner products will be subscripted.

II. TWO QUESTIONS

There are two questions one can ask of the relationship between composite and component states. First, if the composite system is in a specified state D, what states if any can the component systems be in? Second, if we specify the states of the component systems what states if any can the composite system be in? To answer these questions we must form a criterion with which to say it is consistent for a component system (call it 1) to be in a state D_1 while the composite system is in a state D. Such a criterion is as follows: for any proposition of subsystem 1 the probability of that proposition being true, when subsystem 1 is in state D_1 , is equal to the probability that the "same" proposition, as a proposition of the composite system, is true when the composite system is in state D.

Now the association between a proposition of the component system and the same proposition of the composite system is given by the embedding of the propositional lattice of the component system into that of the composite system as we discussed earlier. In Ref. 1 we argue that this embedding depends upon the choice of representation of the lattice of propositions of the two component systems within their Hilbert spaces. That is, it depends upon which subspaces represent which propositions or equivalently which Hermitian operators represent which physical observables. However, whichever representations are chosen the Hilbert spaces are still the same function spaces and, considered on these function spaces independent of interpretation, the embedding is fixed. So by putting off the discussion of observables for a moment we are free to discuss the embedding and its consequences for the relationship between the states, but without physical interpretation of the states. The embedding is as follows: if P_{M_1} is a proposition of the composite system 1, M_1 a subspace of H_1 , then P_M is the same proposition in the composite system when $M = \{\phi^1 \cdot \psi^2 : \phi^1 \in M_1, \psi^2 \in H_2\},\$ where $\phi^1 \cdot \psi^2$: $(x_1, x_2) \to \phi^1(x_1) \psi^2(x_2)$.

So for D_1 and D states of the component and composite systems, respectively, to be consistent with each other we require that

$$\operatorname{tr}(D_1 P_{\mathcal{M}_1}) = \operatorname{tr}(D P_{\mathcal{M}})$$

for all M_1 a subspace of H_1 [call this requirement (A)].

For further work we need the explicit form of P_M for a given M_1 . Let $\{\chi_k^1\}$ be an orthonormal basis of M_1 and let $\{\xi_j^2\}$ be a real orthonormal basis of H_2 then $\{\chi_k^1 \cdot \xi_j^2\}$ is an orthonormal basis of M. So for $\Psi \in H$,

$$P_{\mathcal{M}}\Psi = \sum_{jk} \chi_k^1 \cdot \xi_j^2 (\chi_k^1 \cdot \xi_j^2, \Psi) .$$

We always can write Ψ as

$$\sum_i \phi_i^1 \cdot \psi_i^2 a_i$$

where $\phi_i^1 \in H_1$, $\psi_i^2 \in H_2$, and $a_i \in Q$ for all *i*. Thus using the properties of the inner products,

$$P_{M}\Psi = \sum_{i} (P_{M_{i}}\phi_{i}^{1}) \cdot \psi_{i}^{2}a_{i} .$$
 (1)

III. FIRST QUESTION

To answer the first question mentioned above we need to start with a state of the composite system D. Then given D, requirement (A) amounts to specifying a probability measure on the propositions of H_1 , namely, $P_{M_1} \rightarrow \text{tr}(DP_M)$. So Gleason's theorem asserts that some D_1 exists such that $\text{tr}(D_1P_{M_1}) = \text{tr}(DP_M)$ for all M_1 and its corresponding M. We may also show that such a D_1 is unique. Suppose D'_1 also satisfies $\text{tr}(D'_1P_{M_1}) = \text{tr}(DP_M)$ for all M_1 , then $\text{tr}(D'_1M_1) = \text{tr}(D_1M_1)$ for all M_1 . Now for M_1 one-dimensional, $M_1 = \langle v \rangle$, this becomes $(v, D_1v) = (v, D'_1v)$ for all $v \in H_1$. So writing $A = D_1 - D'_1$ we note that A is Hermitian and (v, Av) = 0 for all v. Take $u \in H_1$ then

$$0 = (Au + u, A(Au + u))$$

= $(Au, A^{2}u) + (u, A^{2}u) + (Au, Au) + (u, Au)$
= $2||Au||^{2}$. (2)

Thus, Au = 0 for any $u \in H_1$ implying $D'_1 = D_1$. Note that $tr(D_1 P_{M_1})$ for one-dimensional propositions was sufficient to determine D_1 uniquely.

Having now shown the existence and uniqueness of D_1 one may reduce the problem. Any D a state of the composite system can be written as $\sum_i \alpha_i D^i$, where the D^i are pure states of the composite system. For each of these D^{i} we have D_1^i the unique solution to $tr(D_1^i P_{M_1}) = tr(D^i P_M)$ for all M_1 . It is easily shown that $D_1 = \sum_i \alpha_i D_1^i$ satisfies condition (A) uniquely. Thus the problem is reduced to finding the component states corresponding to a pure composite state. We can solve this reduced problem. For D pure then $D = P_{(\Phi)}$ the projection onto the one-dimensional subspace of *H*. Here $\langle \Phi \rangle = \{ \Phi \alpha : \alpha \in Q \}$, where Φ is a unit element of *H*. Now Φ can always be written as $\sum_i \phi_i^1 \cdot \psi_i^2 \lambda_i$, where $\{\psi_i^2\}$ is a real orthonormal set in H_2 , $\phi_i^1 \in H_1$, and $\lambda_i \in Q$ for all *i*. As we have shown above to uniquely determine D_1 it is sufficient to consider one-dimensional subspaces $M_1 = \{\chi^1 \alpha :$ $\alpha \in Q$ = $\langle \chi^1 \rangle$ for $\chi^1 \in H_1$ of unit length. Then for the corresponding M using Eq. (1) and the fact that the ψ_i^2 are real,

$$tr(DP_{M}) = \sum_{ij} \lambda_{i}^{*}(\psi_{i}^{2},(\phi_{i}^{1},\chi^{1})_{1}(\chi^{1},\phi_{j}^{1})_{1}\psi_{j}^{2})_{2}\lambda_{j},$$

$$= \sum_{i} \lambda_{i}^{*}\lambda_{i}|(\phi_{i}^{1},\chi^{1})_{1}|^{2}.$$
 (3)

Then it is easily shown that

$$D_1 = \sum_i \lambda \, {}^*_i \lambda_i P_{\langle \phi_i^1 \rangle}$$

satisfies condition (A) uniquely.

IV. SECOND QUESTION

Given states of the component systems D_1 and D_2 then D, a consistent state of the composite system, is neither guaranteed to be unique nor guaranteed to exist. The lack of

uniqueness comes about just as it does in complex theories.^{6,7} The possible nonexistence of a consistent D is new. This can be traced directly to the correspondence between propositions in the component systems and those in the composite system. The pertinent property is that via this correspondence the meet of some propositions of system 1 with some propositions of system 2 is null.

We can give an account of what happens in the restricted case where D_1 and D_2 are pure. If D exists and is mixed then from what has been said above D_1 and D_2 would have to be mixed. So we know that when it exists D is pure. Suppose D exists and let $D_1 = P_{\langle \phi^1 \rangle}$, $D_2 = P_{\langle \psi^2 \rangle}$, and $D = P_{\langle \Phi \rangle}$, where ϕ^1 , ψ^1 , and Φ are elements of H_1 , H_2 , and H, respectively, and are all of unit norm. Here Φ can always be written as $\sum_i \phi_i^1 \cdot \psi_i^2 \lambda_i$, where ϕ_i^1 is a unit element of H_1 and $\lambda_i \in Q$ for all *i* and where $\{\psi_i^2\}$ is a real orthonormal set in H_2 . Thus from Sec. III

$$P_{\langle \phi^1 \rangle} = D_1 = \sum_i \lambda_i^* \lambda_i P_{\langle \phi_i^1 \rangle} .$$
⁽⁴⁾

Further if $v \in \langle \{\phi_i^1\} \rangle$ and $(v, \phi^1) = 0$ then

$$(v, P_{\langle \phi^{\dagger} \rangle} v) = 0 \neq \left(v, \sum_{i} \lambda_{i}^{*} \lambda_{i} P_{\langle \phi_{i}^{\dagger} \rangle} v \right).$$

This is a contradiction, so $\langle \{\phi_i^1\} \rangle = \langle \phi^1 \rangle$ and therefore $\phi_i^1 = \phi^1 \mu_i$ for some $\mu_i \in Q$ and for all *i*. Therefore

$$\Phi = \sum_{i} \phi^{1} \mu_{i} \cdot \psi_{i}^{2} \lambda_{i} = \phi^{1} \cdot \sum_{i} \psi_{i}^{2} \mu_{i} \lambda_{i} .$$
 (5)

Similarly $\Phi = \psi^2 \cdot \chi^1$ for some $\chi^1 \in H_1$.

Finkelstein *et al.* have shown that Φ can be expressed in these two ways only if for some $\alpha_*\beta \in Q$ $[\phi^1(x_1)\alpha, \psi^2(x_2)\beta] = 0$ for all x_1 and x_2 (Ref. 2). And in that case $\Phi = \phi^1 \gamma \cdot \psi^2 \delta$ for some $\gamma, \delta \in Q$. So these conditions are necessary conditions for *D* to exist. When these conditions are satisfied $D = P_{(\phi^1\gamma \cdot \psi^2)}$ for some $\gamma \in Q$, so from the discussion in Sec. III they are sufficient conditions for *D* to exist. The above necessary and sufficient conditions are precisely the conditions for the propositions corresponding to $P_{(\phi^1)}$ and $P_{(\psi^2)}$ in the composite system to have a non-null meet.² So *D* exists when the atomic propositions that correspond to the pure states of the components have a non-null meet.

V. HIDDEN QQM

The thrust of this work is to show that QQM is hidden. To do this we must take a specific orientation of the physical observables within the component Hilbert spaces. We require in particular that the position observables Q_i have the form

$$(Q_i\phi^i)(x_i) = x_i\phi^i(x_i), \quad i = 1, 2,$$

and that the momentum observables P_i have the form

$$(P_i\phi^i)(x_i) = e_1 \frac{\partial}{\partial x_i} \phi^i(x_i), \quad i = 1,2.$$

That these observables can be written like this is shown in Ref. 8. The complex numbers are then defined as $\langle \{1,e_1\}\rangle = \{\alpha 1 + \beta e_1; \alpha, \beta \in \mathbb{R}\}$ and the complex vectors in the function spaces H_1 and H_2 are those functions that map onto the complex subset of Q. We further show in Ref. 8 that

we can associate those states of QQM that leave the subset containing the complex vectors of the Hilbert space invariant with states of a CQM.

A. An illustration

We will now illustrate how it comes about that QQM might be hidden. Consider $D = P_{\langle \Phi \rangle}$ a state of the composite system $\Phi \in H$ with $\Phi(x_1, x_2) = \phi^1(x_1)\psi^2(x_2)$, ψ^2 a unit element of H_2 and $\phi^1 = (1/\sqrt{2})(\phi_1^1 + e_1\phi_2^1)$ for ϕ_1^1 and ϕ_2^1 real and unit elements of H_1 , $(\phi_1^1, \phi_2^1)_1 = 0$. So we may write Φ as $\psi^2 \cdot \phi_1^1(1/\sqrt{2}) + e_1\psi^2 \cdot \phi_2^1(1/\sqrt{2})$ and then from our previous analysis

$$D_2 = \frac{1}{2} P_{\langle \psi^2 \rangle} + \frac{1}{2} P_{\langle e_1 \psi^2 \rangle} .$$
 (6)

We examine the action of D_2 on a complex element of H_2 ,

$$D_{2}\chi^{2} = \frac{1}{2}\psi^{2}(\psi^{2},\chi^{2})_{2} + \frac{1}{2}e_{1}\psi^{2}(e_{1}\psi^{2},\chi^{2})_{2},$$

$$= \frac{1}{2}\psi^{2}(\psi^{2},\chi^{2})_{2} + \frac{1}{2}e_{1}\psi^{2}(\psi^{2},\chi^{2})_{2}e_{1}^{*}$$

as $\chi^{2}(x_{2}) = \langle \{1,e_{1}\} \rangle$ for all x_{2} ,

$$= \psi_{1}^{2} + e_{1}\psi_{1}^{2}e_{1}^{*}, \text{ where } \psi_{1}^{2} = \frac{1}{2}\psi^{2}(\psi^{2},\chi^{2})_{2}.$$
 (7)

Therefore $D_2\chi^2$ is a complex element of H_2 . So D_2 leaves the subset $L^2_{\{1,e_1\}}$ (\mathbb{R})₂ of H_2 invariant. Thus D_2 may be written as $\sum_i \alpha_i P_{(\xi_i^2(x_2))}$, where the $\xi_i^2(x_2)$ are complex functions in H_2 (Ref. 8). Thus for composite states of the form of D the component states of particle 2 are always complex, independent of the form of $\psi^2(x_2)$. So for such states their quaternionic nature is somewhat hidden.

Pursuing the illustration further we note that the power of the result we have obtained depends very much upon how many and which composite states are of the appropriate form. Apart from the fact that it is not normalizable the momentum state⁸ exp(e_1px_1) would have the appropriate form for $\phi^1(x_1)$. However, in the case where we consider states normalized in a box of finite volume, $\phi^1(x_1) = (1/N)\exp(px_1)$ is a normalized vector for some $N \in \mathbb{R}$. Hence

$$\Phi(x_1, x_2) = \phi^1(x_1)\psi^2(x_2)$$

= $(1/\sqrt{2}N)\psi^2(x_2)\sqrt{2}\cos(px_1)$
+ $(1/\sqrt{2}N)e_1\psi^2(x_2)\sqrt{2}\sin(px_1)$ (8)

is of the appropriate form for any $\psi^2 = L_Q^2 (\text{Box})_2$. It is clear that as well as D_2 being complex D_1 is the momentum state $P_{((1/N)\exp(e,px))}$. This shows, for such a normalization, that a large class of composite states are of the appropriate form, including those in which D_1 is a momentum state. It is instructive for generalization to note that

 $\phi^{1}(x_{1}) = \frac{1}{2}(\exp(e_{1}px_{1}) + \exp(-e_{1}px_{1})) = \cos(px_{1})$ does not leave $\Phi(x_{1},x_{2})$ in the appropriate form, as

$$\Phi(x_1, x_2) = \phi^1(x_1)\psi^2(x_2) = \psi^2(x_2)\cos(px_1)$$

d so $D_2 = P_{\langle \psi^2(x_2) \rangle}$.

B. A generalization

an

With the above in mind we examine a composite state $D = D_{\langle \Phi \rangle}$, where $\Phi(x_1, x_2) = \phi^1(x_1)\psi^2(x_2)$, ϕ^1 a unit ele-

ment of H_1 and ψ^2 a unit element of H_2 . We also require

$$\phi^{1}(x_{1}) = \left(\frac{1}{2\pi}\right)^{1/2} \int_{-\infty}^{\infty} \exp(e_{1}p_{1}x_{1})\tilde{\phi}^{1}(p_{1})dp_{1}$$

with $\phi(p_1) \neq 0$, implying $\phi^1(-p_1) = 0$ for which we will say $\tilde{\phi}^1$ satisfies condition (B). We will say D satisfies (B'). Then setting

$$A^{1} = \left(\frac{1}{2\pi}\right)^{1/2} \int_{-\infty}^{\infty} \cos(p_{1}x_{1}) \tilde{\phi}^{1}(p_{1}) dp_{1}$$

and

$$B^{1} = \left(\frac{1}{2\pi}\right)^{1/2} \int_{-\infty}^{\infty} \sin(p_{1}x_{1})\tilde{\phi}^{1}(p_{1})dp_{1}$$

we may write

$$\Phi(x_1, x_2) = A^{-1}(x_1)\psi^2(x_2) + e_1B^{-1}(x_1)\psi^2(x_2) .$$
 (9)

Note that A^{1} and B^{1} are elements of H_{1} , A^{1} is even, and B^{1} is odd. If we can write

$$B^{1}(x_{1}) = \sum_{i} \chi^{1}_{Bi}(x_{1})b_{i}$$

with $\{\chi_{Bi}^{1}\}$ an orthonormal set of real odd functions in H_1 and $b_i \in Q$ for all *i* and if we can also write

$$A^{1}(x_{1}) = \sum_{i} \chi^{1}_{Ai}(x_{1})a_{i}$$

with $\{\chi_{Ai}^{1}\}$ an orthonormal set of real even functions in H_{1} with $a_{i} = b_{i}$ for all *i* then

$$\Phi(x_{1,x_{2}}) = \sum_{i} \frac{b_{i}}{|b_{i}|} \psi^{2}(x_{2}) \chi^{1}_{Ai}(x_{1}) |b_{i}|$$

+
$$\sum_{i} e_{1} \frac{b_{i}}{|b_{i}|} \chi^{1}_{Bi}(x_{1}) |b_{i}|.$$
(10)

Thus as $(\chi_{Ai}^{i}, \chi_{Bj}^{i})_{1} = 0$ for all *i* and *j* we have, from our previous discussion,

$$D_2 = \sum_i |b_i|^2 (P_{\langle (b_i|b_j|)\psi^2 \rangle} + P_{\langle e_i(b_j|b_j|)\psi^2 \rangle}), \qquad (11)$$

which as before means that D_2 is complex.

We now need to show that we can find such sets $\{\chi_{Bi}^{1}\}$ and $\{\chi_{Ai}^{1}\}$. Take any orthonormal set of real odd functions $\{\chi_{Bi}^{1}\}$ that span the subspace of odd functions of H_{1} ; then as B^{1} is odd

$$B^{1}(x_{1}) = \sum_{i} \chi^{1}_{Bi}(x_{1})b_{i}$$
 with $b_{i} = (\chi^{1}_{Bi}, B^{1})_{1}$.

We wish to find an orthonormal set of real even functions $\{\chi_{Ai}^{1}\}$ such that

$$A^{1}(x_{1}) = \sum_{i} \chi^{1}_{Ai}(x_{1})a_{i}$$

with $a_i = (\chi^1_{A_i}, A^1)$ and $a_i = b_i$. We may write

$$\chi_{Bi}^1(x_1) = \left(\frac{2}{\pi}\right)^{1/2} \int_{-\infty}^{\infty} \widetilde{\chi}_{Bi}^1(p) \sin(px_1) dp$$

with $\tilde{\chi}_{Bi}^{1}(p) = 0$ for p < 0 as χ_{Bi}^{1} is odd. Evaluating b_i we find that

$$b_{1} = \int_{0}^{\infty} \widetilde{\chi}_{Bi}^{1}(p) \widetilde{\phi}^{1}(p) dp - \int_{0}^{\infty} \widetilde{\chi}_{Bi}^{1}(p) \widetilde{\phi}^{1}(-p) dp, \qquad (12)$$

by expanding the integrals. Now let $m_1 = \{p | \bar{\phi}^1(p) \neq 0, p \ge 0\}$ and $m_2 = \{p | \bar{\phi}^1(-p) \neq 0, p \ge 0\}$ then by condition (B) $m_1 \cap m_2 = \phi$. Further, take any $M_1 \supseteq m_2$ and $M_2 \supseteq m_2$ such that $M_1 \cup M_2 = R^+ \cup \{0\}$ and $M_1 \cap M_2 = \phi$, then

$$b_i = \int_{M_1} \widetilde{\chi}_{B_i}^1(p) \widetilde{\phi}^1(p) dp - \int_{M_2} \widetilde{\chi}_{B_i}^1(p) \widetilde{\phi}^1(-p) dp.$$

Similarly we may write

$$\chi^{1}_{Ai}(x_{1}) = \left(\frac{2}{\pi}\right)^{1/2} \int_{-\infty}^{\infty} \widetilde{\chi}^{1}_{Ai}(p) \cos(px_{1}) dp$$

with $\chi^1_{Ai}(p) = 0$ for p < 0 as χ^1_{Ai} is even. And similarly we may evaluate a_i ,

$$a_{i} = \int_{M_{1}} \tilde{\chi}_{Ai}^{1}(p) \tilde{\phi}^{1}(p) dp + \int_{M_{2}} \tilde{\chi}_{Ai}^{1}(p) \tilde{\phi}^{1}(-p) dp .$$
(13)

So to ensure $a_i = b_i$ it is sufficient to demand that $\chi^1_{Ai}(x_1)$ satisfy $\tilde{\chi}^1_{Ai}(p) = \tilde{\chi}^1_{Bi}(p)$ for $p \in M_1$, $\tilde{\chi}^1_{Ai}(p) = -\tilde{\chi}^1_{Bi}(p)$ for $p \in M_2$, and $\tilde{\chi}^1_{Ai}(p) = \tilde{\chi}^1_{Bi}(p) = 0$ for p < 0.

We have left to show that $\{\chi_{Ai}^{1}\}$ is an orthonormal set of real even functions in H_1 . That $\chi_{Ai}^{1}(x_1)$ is even is obvious since it has a cosine expansion. That $\chi_{Ai}^{1}(x_1)$ is real is ensured by $\tilde{\chi}_{Ai}^{1}(p)$ being real which in turn comes from $\chi_{Bi}^{1}(x_1)$ being real. Using the above relationship between $\tilde{\chi}_{Ai}^{1}(p)$ and $\tilde{\chi}_{Bi}^{1}(p)$ it is easy to show that

$$(\chi^{1}_{Ai},\chi^{1}_{Ai}) = (\chi^{1}_{Bi},\chi^{1}_{Bi}) .$$
(14)

Thus $\chi_{Ai}^1(x_1)$ is square integrable for all *i* and $\{\chi_{Ai}^1(x_1)\}$ is an orthonormal set in H_1 . So D_2 is complex whenever *D* satisfies (B').

C. Superpositions

We will extend this result to superpositions, that is to states $D_{\langle \Phi \rangle}$ with

$$\Phi(x_1,x_2) = \sum_i \phi_i^1(x_1) \psi_i^2(x_2) \, ,$$

where

$$\phi_i^1(x_1) = \left(\frac{1}{2\pi}\right)^{1/2} \int_{-\infty}^{\infty} \exp(e_1 p_1 x_1) \tilde{\phi}_i^1(p) dp$$

and where $\tilde{\phi}_i^1(p) \neq 0$ for some *i* implies $\tilde{\phi}_j^1(-p) = 0$ for all *j*. We will say $\{\tilde{\phi}_i^1\}$ satisfies condition (C), and that *D* satisfies condition (C').

We may write $\Phi(x_1, x_2)$,

$$\Phi(x_{1},x_{2}) = \sum_{i} \left(\frac{1}{2\pi}\right)^{1/2} \int_{-\infty}^{\infty} \exp(e_{1}p_{1}x_{1})\tilde{\phi}_{i}^{1}(p)dp \,\psi_{i}(x_{2}) ,$$

$$= \left(\frac{1}{2\pi}\right)^{1/2} \sum_{i} \left(\int_{-\infty}^{\infty} \tilde{\phi}_{i}^{1}(p)\cos(px_{1})dp \,\psi_{i}(x_{2}) + e_{1} \int_{-\infty}^{\infty} \tilde{\phi}_{i}^{1}(p)\sin(px_{1})dp \,\psi_{i}(x_{2})\right)$$

$$= \sum_{i} A_{i}^{1}(x_{1})\psi_{i}(x_{2}) + \sum_{i} e_{1}B_{i}^{1}(x_{1})\psi_{i}(x_{2}) . \quad (15)$$

Then if we can write

$$B_i^1(x) = \sum_i \chi_{Bj}^1(x_1) b_{ij}$$

with $\{\chi_{B_j}^1\}$ an orthonormal set of real odd functions in H_1 and if we can also write

$$A_{i}^{1}(x_{1}) = \sum_{i} \chi_{Ai}^{1}(x_{1}) a_{ij}$$

with $\{\chi_{Aj}^1\}$ an orthonormal set of real even functions in H_1 and $a_{ij} = b_{ij}$ for all *i* and *j* we have

$$\Phi(x_1, x_2) = \sum_j \left(\frac{(\Sigma_i b_{ij} \psi_i^2(x_2))}{N_j} \chi_{Aj}^1(x_1) N_j + e_1 \frac{(\Sigma_i b_{ij} \psi_i^2(x_2))}{N_j} \chi_{Bj}^1(x_2) N_j \right), \quad (16)$$

with

$$N_j = \left| \left| \sum_i b_{ij} \psi_i(x_2) \right| \right|$$

So as $(\chi^1_{A_i}(x_1), \chi^1_{B_i}(x_1)) = 0$ for all *i* and *j* we have that

$$D_2 = \sum_j \left(P_{\langle \Sigma_i(a_{ij}\psi_i^2/N_j) \rangle} + P_{\langle e_i \Sigma_i(b_{ij}\psi_i^2/N_j) \rangle} \right), \qquad (17)$$

which, as before, is complex.

To show that we can find the required orthonormal sets we note that $\{\tilde{\phi}_i^1\}$ satisfying condition (C) implies that the non-negative real line may be split into two disjointed sets M_1 and M_2 with

$$M_1 \supseteq \{p: \tilde{\phi}_i^1(p) \neq 0, p \ge 0\}$$
 for all *i*

and

$$M_2 \supseteq \{p: \tilde{\phi}_i^1(-p) \neq 0, p \ge 0\}$$
 for all i

So given $\{\chi_{Bj}^{1}\}\$ an orthonormal set of real odd functions we may define $\{\chi_{Aj}^{1}\}\$ as before via M_{1} and M_{2} which we note are independent of *i*. So for each *i* the required conditions to show $a_{ij} = b_{ij}$ for all *j* are satisfied by this choice of $\{\chi_{Bj}^{1}\}\$ and $\{\chi_{Aj}^{1}\}$.

D. Momentum conditions

We will now show that a pure composite state satisfying condition (C') is equivalent to the probability density Pd of the momentum of particle 1 satisfying condition (B).

First take $D = P_{\langle \Phi \rangle}$ with

$$\Phi(x_1,x_2) = \sum_i \phi_i^1(x_1) \psi_i^2(x_2)$$

and

$$\psi_i^2(x_2) = \sum_j \xi_j^2(x_2) \lambda_{ji}$$
,

where $\{\xi_i^2\}$ is an orthonormal set of real functions and a basis for H_2 and $\lambda_{ji} \in Q$ for all *i* and *j*. Then

$$\Phi(x_1,x_2) = \sum_{ji} \phi_i^1(x_1) \lambda_{ji} \xi_j^2(x_2) \, .$$

Thus the component state

$$D_1 = \sum_j \alpha_j P_{\langle \Sigma_i \phi_i^1 \lambda_{ij} \rangle}$$

where

$$\alpha_j = \left| \left| \sum_i \phi_i^1(x_1) \lambda_{ji} \right| \right|$$

The particle 1 momentum probability density function Pd(p) is then

$$\sum_{j} \alpha_{j}^{2} \left(\sum_{i} \tilde{\phi}_{i}^{1}(p) \lambda_{ji} \right)^{\bullet} \left(\sum_{i} \tilde{\phi}_{i}^{1}(p) \lambda_{ji} \right).$$

Suppose D satisfies (C'). So condition (C) holds for $\{\bar{\phi}_i^1\}$. Take $p \in \mathbb{R}$ if $\mathrm{Pd}(p) \neq 0$, then $\tilde{\phi}_i^1(p) \neq 0$ for some *i*; thus we have that $\tilde{\phi}_k^1(-p) = 0$ for all k. It follows that Pd(-p) = 0. Therefore Pd satisfies condition (B).

On the other hand suppose Pd satisfies (B), i.e., $Pd(p) \neq 0$ implies Pd(-p) = 0. Then

$$\sum_i \tilde{\phi}_i^1(p) \lambda_{ji} \neq 0$$

for some *j* forces

$$\sum_{i} \tilde{\phi}_{i}^{1}(-p)\lambda_{ji} = 0$$

for all *j*. So letting

$$f_j^1(x_1) = \sum_i \phi^1(x_1) \lambda_{ji}$$

we have that

$$\Phi(x_1, x_2) = \sum_j f_j^1(x_1) \xi_j^2(x_2)$$

with $\{\tilde{f}_i^1\}$ satisfying condition (C) as

$$\tilde{f}_j^1(p) = \sum_i \tilde{\phi}_i^1(p) \lambda_{ji}$$
.

So D satisfies (C').

Finally consider a nonpure state $D = \sum_i \alpha_i D_i$, D_i pure, with the particle 1 momentum probability density function Pd(p) satisfying condition (B). As

$$\operatorname{Pd}(p) = \sum_{i} \alpha_{i}^{2} \operatorname{Pd}^{i}(p)$$
,

Pdⁱ the particle 1 momentum probability density function for state D^{i} also satisfies condition (B) for all *i*. It follows that D^{i} satisfies (C') and therefore that D_{2}^{i} is complex and, as convex combinations of complex states are complex, that D_2 is complex.

We have then that for any composite state D if the probability density function for the momentum of particle 1 satisfies condition (B) then the component state of particle 2, D_2 , is complex. Extending this result to three dimensions and to many particles is just a matter of introducing new parameters and may be derived as a corollary to the two-particle, one-dimensional case. The statement of the theorem then becomes, that for any many particle three-dimensional state D, if the probability density function for the component of the momentum in any direction of any particle satisfies condition (B) then the component system of the rest of the particles together is complex. That condition (B) is satisfied by a momentum probability density is a common situation. For instance, (B) is satisfied if any component of the momentum of a particle is entirely positive. This occurs whenever the momenta of all but one of the products of a decay are measured sufficiently accurately which, for the two-body decay, amounts to observing one of the particles to be within any hemisphere centered on the center of mass.

VI. CONCLUSION

We have found that the description of the component states for a given composite state follows along much the same lines as in complex quantum mechanics. Using this formalism we have shown that the component states are complex under weak conditions, much weaker than requiring one particle to be a momentum state. This means that if one can only measure component systems then it is plausible for the component system to be always in a complex state while the total system is in a quaternionic state, thus shielding quaternionic states from investigation.

- ²D. Finkelstein, J. M. Jauch, and D. Speiser, CERN Report 59-7, 1959. Reprinted in Logico Algebraic Approach to Quantum Mechanics II, edited
- by C. A. Hooker (Reidel, Dordrecht, 1979).
- ³A. Zecca, J. Math. Phys. 19, 1482 (1978). ⁴D. Aerts, J. Math. Phys. 21, 778, (1980).
- ⁵V. S. Varadarajan, Geometry of Quantum Theory (Van Nostrand, New York, 1970).
- ⁶E. G. Beltrametti and G. Cassinelli, The Logic of Quantum Mechanics (Addison-Wesley, Reading, MA, 1981), pp. 66-67, 237.

⁸C. G. Nash and G. C. Joshi, University of Melbourne preprint UM-P-86/ 73 (unpublished).

¹C. G. Nash and G. C. Joshi, University of Melbourne preprint UM-P-86/ 71 (unpublished).

⁷J. M. Jauch, Foundations of Quantum Mechanics (Addison-Wesley, Reading, MA, 1969), pp. 179-182.

Absolutely continuous spectra of quasiperiodic Schrödinger operators

Luigi Chierchia^{a)}

Forschungsinstitut für Mathematik, E. T. H. Zentrum, CH 8092 Zürich, Switzerland

(Received 22 July 1986; accepted for publication 13 August 1987)

Several aspects of the general and constructive spectral theory of quasiperiodic Schrödinger operators in one dimension are discussed. An explicit formula for the absolutely continuous (a.c.) spectral densities that yields an immediate proof of the fact that the Kolmogorov– Arnold–Moser (KAM) spectrum constructed by Dinaburg, Sinai, and Rüssmann [Funkt. Anal. Prilozen. 9, 8 (1975); Ann. Acad. Sci. 357, 90 (1980)] is a subset of the a.c. spectrum is provided. Some quasiperiodicity properties of the Deift–Simon a.c. eigenfunctions are proved, namely, that the normalized phase of such eigenfunctions is a quasiperiodic distribution. In the constructive part the Dinaburg–Sinai–Rüssmann theory is extended to quasiperiodic perturbations of periodic Schrödinger operators using a KAM Hamiltonian formalism based on a new treatment of perturbations of harmonic oscillators. Particular attention is devoted to the dependence upon the eigenvalue parameter and a complete control of KAM objects is achieved using the notion of Whitney smoothness.

I. INTRODUCTION

Let L_{θ} be a quasiperiodic Schrödinger operator in one dimension, ¹⁻⁶

$$L_{\theta} \equiv L(v_{\theta}) \equiv -\frac{d^2}{dx^2} + v_{\theta}(x),$$

$$v_{\theta}(x) \equiv V(T_x\theta), \quad T_x\theta \equiv \theta + \omega x,$$

where $x \in \mathbb{R}$, $\theta \in \mathbb{T}^d \equiv \mathbb{R}^d / 2\pi \mathbb{Z}^d$, $\omega \in \mathbb{R}^d$ is a rationally independent vector and V is a real function defined on \mathbb{T}^d . In this paper we discuss, from two points of view, the absolutely continuous (a.c.) spectrum of L_{θ} . First, continuing the analysis in Refs. 2, 3, and 7, we study some general problems such as characterization almost everywhere (with respect to Lebesgue measure on \mathbb{R} and/or Haar measure on \mathbb{T}^d) of the a.c. eigenfunctions and of spectral densities. Then we turn to the explicit construction of many (in the sense of Lebesgue measure) quasiperiodic a.c. eigenfunctions for a special class of potentials v. This second part should be regarded as a refinement of the theory in Refs. 8 and 9.

Our results in the general part are described by the following three theorems. Before describing them let us recall a few definitions. The spectral class measure $d\mu^{\theta}$ is given by one of the following mutually equivalent measures:

$$d\mu^{\theta} = \sum_{1}^{\infty} a_n d\mu^{\theta}_{\phi_n}, \quad a_n > 0, \quad \phi_n \in C_0^{\infty},$$

where $\Sigma a_n < \infty$, $\{\phi_n\}$ is an L^2 -dense set of C^{∞} functions with compact support and $d\mu_{\phi_n}^{\theta}$ denotes the standard spectral measure of L_{θ} based upon ϕ_n . Now let $d\mu_{a.c.}^{\theta}$ be the a.c. part of $d\mu^{\theta}$ in the Jordan–Lebesgue decomposition. The essential support S of $d\mu_{a.c.}^{\theta}$ is uniquely determined (modulo sets of zero Lebesgue measure) by the requirement that if $A \subset S$ is also a support for $d\mu_{a.c.}^{\theta}$ then meas(S - A) = 0. Finally, let $f_{\pm}(x,\theta,E)$ be the solution of

$$L_{\theta}f = Ef \tag{1.1}$$

with Im $E \neq 0$, $f \in L^2(\mathbb{R}_{\pm})$, $\mathbb{R}_{+} \equiv (0, \infty)$, $\mathbb{R}_{-} \equiv (-\infty, 0)$. Denoting by d/dE the Radon-Nicodym derivative with respect to Lebesgue measure, by [g,h] the Wronskian $gh' - g'h \equiv g(dh/dx) - (dg/dx)h$ and by (ϕ, f) the L^2 product $\int_{\mathbb{R}} \phi(x) \overline{f}(x) dx$, we have the following theorem.

Theorem 1.1: For any $\phi \in C_0^{\infty}$, for a.e. $(\theta, E) \in \mathbb{T}^d \times S$,

$$\frac{d\mu_{\phi,\text{a.c.}}^{\theta}}{dE} = \frac{1}{2\pi i} \frac{|(\phi,f)|^2 + |(\phi,\bar{f})|^2}{[f,\bar{f}]},$$

where $f(x,\theta,E) \equiv \lim_{\epsilon \downarrow 0} f_+(x,\theta,E+i\epsilon)$ and $i[f,\bar{f}] > 0$.

Throughout this paper a fundamental role is played by Bloch waves (or Floquet solutions). These are eigensolutions of the form $\psi = e^{i\beta x}\chi$ with $\beta \in \mathbb{R}$ and χ a quasiperiodic function with basic frequencies ω .

Theorem 1.2: Let $I \subseteq \mathbb{R}$ be a set of positive Lebesgue measure and assume that for a.e. (θ, E) in $\mathbb{T}^d \times I$ there exists a Bloch wave ψ . Then $I \subseteq S$, $[\psi, \overline{\psi}] \neq 0$ and, for any $\phi \in C_0^{\infty}$,

$$\frac{d\mu_{\phi,\text{a.c.}}^{\sigma}}{dE} = \frac{1}{2\pi} \frac{|(\phi,\psi)|^2 + |(\phi,\overline{\psi})|^2}{|[\psi,\overline{\psi}]|},$$

(θ,E) a.e. in $\mathbb{T}^d \times I$.

In Ref. 3 Deift and Simon showed that, for a.e. (θ, E) in $\mathbf{T}^d \times S$ there exist eigensolutions $g = e^{i(\alpha x + \beta(x))}r(x)$, with $[g,\overline{g}] = -2i$ and α being the Johnson-Moser rotation number,^{7,10} such that r is an L^2 quasiperiodic function, i.e., $r(x,\theta,E) = R(T_x\theta)$ with $R(\cdot,E) \in L^2(\mathbf{T}^d)$. However, no quasiperiodicity properties were proved for the phase β . Now assume that ω satisfies a Diophantine condition like

$$|\omega \cdot \nu| \equiv \left| \sum_{i=1}^{d} \omega_{i} \nu_{i} \right| \geq \frac{1}{c |\nu|^{\tau}},$$

$$|\nu| \equiv \sum |\nu_{i}| \quad (\text{any } \nu \in \mathbb{Z}^{d} - 0, \text{ some } c, \tau > 0), \qquad (1.2)$$

and denote by

$$\mathbf{0} \equiv \bigg\{ \Phi \in C^{\infty}(\mathbb{T}^d) \colon \int \Phi = 0 \bigg\}.$$

^{a)} Permanent address: Dipartimento di Matematica, IIa Università di Roma, 00173 Rome, Italy.

Then β is an (ω) quasiperiodic distribution on 0 in the sense of the following theorem.

Theorem 1.3: There exists a distribution B on $C^{\infty}(\mathbb{T}^d)$ such that for any $\Phi \in \mathbf{0}$,

$$\langle B, \Phi \rangle = \lim_{x \to \infty} \frac{1}{x} \int_0^x \beta(y, \theta, E) \overline{\Phi}(T_y \theta) dy,$$

(\theta, E) a.e. in $\mathbb{T}^d \times S.$ (1.3)

Remark 1.4: Equation (1.3) determines B uniquely on 0. Taking $\Phi = e^{i\nu\cdot\theta}$, (1.3) shows that all the quasiperiodic Fourier coefficients (with $\nu \neq 0$) of β are well defined and are equal to $\hat{B}_{\nu}e^{i\nu\cdot\theta}$, $\hat{B}_{\nu} \equiv \langle B, e^{i\nu\cdot\theta} \rangle$.

Remark 1.5: Theorems 1.1 and 1.2 can be trivially extended to the case of almost periodic Schrödinger operators. Theorem 1.3 is false if ω fails to satisfy any Diophantine condition, i.e., if ω is a "Liouville vector"; compare Ref. 1.

The problem of characterizing the a.c. spectrum in terms of genuine Bloch waves remains open but we will see that it is closely related to the analysis of regularity properties of a nonlinear partial differential equation (PDE) on \mathbb{T}^d , namely,

$$D_{\omega}^{2} F = \frac{1}{F^{3}} + (V - E)F, \quad F(\theta) > 0 \quad \text{for a.e. } \theta, \quad (1.4)$$

where

$$D_{\omega} \equiv \sum_{i=1}^{d} \omega_i \, \frac{\partial}{\partial \theta_i}.$$

Equation (1.4) will be shown to be satisfied, for a.e. E in S, by $R(\cdot, E)$ in the sense of distributions.

We pass now to the constructive part of the theory. The operators that we shall consider are of the form $L^{(\epsilon)} \equiv L(v + \epsilon w), \quad v + \epsilon w \equiv V(\omega_1 x) + \epsilon W(\omega_2 x, ..., \omega_d x),$ with V, W real analytic on, respectively, T, T^{d-1} and ϵ a positive number. The vector ω is assumed to satisfy a generalized Diophantine condition

$$|\omega \cdot v| \ge 1/c\Omega(|v|), \quad v \in \mathbb{Z}^d - 0, \ c > 0 \quad (\text{fixed}), \quad (1.5)$$

where $\Omega(r) \ge r^{d-1}$ is a monotone function growing not too fast as $r \uparrow \infty$ (see Ref. 9). Then, employing a Kolmogorov-Arnold-Moser (KAM) technique,¹¹⁻¹³ we will construct, for small $c\epsilon/\kappa$, a subset $\mathbf{E}^{(\epsilon)}$ of $\sigma(L^{(0)}) \cap \sigma(L^{(\epsilon)})$ and for each $E \in E^{(\epsilon)}$ a Bloch wave $e^{i\alpha x} \chi(\omega x)$ with (α, ω) rationally independent and $\chi(\theta)$ analytic on \mathbf{T}^d . The parameter κ is a function of E asymptotic to \sqrt{E} and, for some a, b > 0 and for any $E_0 > 0$, the set $\mathbf{E}^{(\epsilon)}$ satisfies

$$\operatorname{meas}\left\{\left(\sigma(L^{(0)}) - \mathbf{E}^{(\epsilon)}\right) \cap [E_{0}, \infty)\right\}$$
$$\leq \frac{a}{c} \left(\sum_{|\nu| > bE_{0}} \frac{|\nu| \log \log \Omega(|\nu|)}{\Omega(|\nu|)}\right).$$
(1.6)

The connection with the general part is then given by Theorem 1.2 which yields immediately $\mathbf{E}^{(\epsilon)} \subset \sigma_{a.c.} (L^{(\epsilon)})$.

Before constructing such Bloch waves we will explain that the existence of quasiperiodic eigenfunctions corresponds to quasiperiodic Hamiltonian flows on (d + 1)-dimensional tori; see, also, Refs. 14 and 15. In general, to any operator $L_{\theta_0}(u), u(x) \equiv U(T_x \theta_0)$, we can associate the (d + 1)-dimensional Hamiltonian

$$H_U(p,B,q,\theta; E) \equiv p^2/2 + \omega \cdot B + (q^2/2) [E - U(\theta)],$$

where $(p,B) \in \mathbb{R}^{d+1}$ denote the generalized momenta and $(q,\theta) \in \mathbb{R} \times \mathbb{T}^d$ denote the generalized coordinates. It is readily checked that the evolution equation for q [with initial data $q(0), p(0) \equiv q'(0), \theta(0) \equiv \theta_0$] is nothing but the eigenvalue equation $L_{\theta_0}q = Eq$. What we will see is that, for $E \in \mathbb{E}^{(\epsilon)}, H_{V+\epsilon W}$ is canonically conjugate to a system of harmonic oscillators with Hamiltonian $\alpha A_0 + \omega_1 A_1 + \cdots + \omega_d A_d$ in action-angle variables $(A, \theta) \in \mathbb{R}_+ \times \mathbb{R}^d \times \mathbb{T}^{d+1}$. This fact, from one side, clarifies the use of KAM techniques in the theory of quasiperiodic Schrödinger operators and, on the other side, gives a rather natural interpretation of spectral quantities such as a.c. eigenfunctions and the rotation number in terms of Hamiltonian H_U is integrable whenever $E \in \sigma_{a,c} (L(u))$.

In our treatment of these matters we refine some aspects of the Dinaburg-Sinai-Rüssmann theory. For example, we will see that KAM objects, such as rotation number and Bloch waves constructed on $\mathbf{E}^{(\epsilon)}$ are C^{∞} functions of $E \in \mathbf{E}^{(\epsilon)}$ in the sense of Whitney.¹⁶⁻¹⁸ Exploiting this fact it will be easy to give a self-contained and complete description of the KAM spectrum $\mathbf{E}^{(\epsilon)}$ that was still missing in the literature.

Since the basic KAM techniques are by now well known (see, e.g., Ref. 19), most of the proofs in this second part will be outlined without going into detail.

The content of the rest of the paper is the following: Sec. II, proof of Theorem 1.1; Sec. III, Bloch waves; Sec. IV, weak Bloch waves; Sec. V, periodic Schrödinger operators as harmonic oscillators; Sec. VI, quasiperiodic perturbations; Sec. VII, KAM Bloch waves; Sec. VIII, Whitney smoothness; Sec. IX, structure of KAM spectra; Appendix A: on a new condition in analytic KAM; Appendix B: Moser-Deift-Simon inequality on KAM spectra.

II. PROOF OF THEOREM 1.1

We need the following facts:

(a)
$$\frac{d\mu_{\phi,\text{a.c.}}}{dE} = \lim_{\epsilon \downarrow 0} \operatorname{Im}(R_{E+i\epsilon}\phi,\phi),$$

 $R_E \equiv (L-E)^{-1},$

for any $\phi \in C_0^\infty$ and a.e. *E* in *S*.

(b)
$$R_E(x,y) \equiv g(x,y;E)$$

$$=f_+(x)f_-(y)/[f_+,f_-], \text{ Im } E \neq 0,$$

for $x \ge y$ and symmetrically for x < y (f_{\pm} are the eigenfunctions introduced in Sec. I).

(c)
$$f_{\pm}(x,\theta,E) = \operatorname{const}(f_1(x,\theta,E))$$

$$\pm h_{\pm}(\theta,E)f_2(x,\theta E)),$$

where f_1 , f_2 solve (1.1) with $f_1(0) = f'_2(0) = 1$, $f'_1(0) = f_2(0) = 0$ and h_{\pm} are, for every θ , the Herglotz functions defined by $\lim_{x \to \pm \infty} \mp f_1/f_2$. We recall that a function h is Herglotz if it maps holomorphically the open upper half plane \mathbb{C}_+ into itself. We will denote the boundary value of h,

existing a.e. on **R**, by the same symbol. For more information see, e.g., Ref. 20.

(d) For a.e.
$$E$$
 in \mathbb{R} ,

$$-\lim_{\epsilon \downarrow 0} \operatorname{Re} \int_{\mathbb{T}^{d}} h_{+}(\theta, E + i\epsilon) d\theta$$

$$\equiv \gamma(E) = \gamma_{+}(E, \theta)$$

$$\equiv \text{(highest) Lyapunov exponent for } L_{e} - E$$

and

$$S = \{E: \gamma(E) = 0\} \subset \{E: h_{+} = -\bar{h}_{-}, \operatorname{Im} h_{+} > 0\}.$$

Equation (a) is a simple consequence of Stone's formula (see, e.g., Ref. 21). Equations (b) and (c) are the main results of Weyl's limit-point theory (see Ref. 22). Equation (d) is proved in Ref. 2.

Notice that, for a.e. $E, f_{\pm}(x, \theta, E + i\epsilon)$ converge, as $\epsilon \downarrow 0$, uniformly on compact x sets. Also, for a.e. E in S, (d) shows that $f_{+} = \overline{f}_{-}$ with $[f_{+}, f_{-}] = -2i \text{ Im } h_{+}$. These observations together with (a) and the evaluation

$$\operatorname{Re} \iint_{x > y} \psi(x) \overline{\psi}(y) \phi(x) \overline{\phi}(y) dx \, dy = \frac{1}{2} |(\overline{\psi}, \phi)|^2,$$

valid for any $\psi \in C(\mathbb{R})$ and $\phi \in C_0^{\infty}(\mathbb{R})$, make Theorem 1.1 plain.

III. BLOCH WAVES

In this section we prove some elementary properties of (genuine \equiv smooth) Bloch waves and Theorem 1.2.

Lemma 3.1: (i) If $\psi(x) = e^{i\beta x}\chi(\omega x)$ is a Bloch wave for $L_0 - E$ then $\psi(x,\theta) \equiv e^{i\beta x}\chi(T_x\theta)$ is a Bloch wave for $L_{\theta} - E$.

(ii) Let *I* be as in Theorem 1.2. Then ψ can be written (a.e. on *I*) in the form $e^{i\alpha x}\chi(T_x\theta)$ with (α,ω) rationally independent.

(iii) If $\psi = e^{i\alpha x} \chi(\omega x)$ is a Bloch wave (α, ω) rationally independent, then $[\psi, \overline{\psi}] \neq 0$ and $\min_{\mathbf{T}^d} |\psi| > 0$.

Proof: Since ψ solves (1.1) with $\theta = 0$, χ satisfies

$$D_{\omega}^{2}\chi + 2i\beta D_{\omega}\chi + (E - \beta^{2} - V)\chi = 0 \qquad (3.1)$$

at $\theta = \omega x$. But because $\{\theta = \omega x : x \in \mathbb{R}\}$ is dense in \mathbb{T}^d , (3.1) holds identically on \mathbb{T}^d . In particular, it holds at $\theta + \omega x$ and (i) is proved.

Property (ii) follows easily from (i) and the fact that $E \in \mathbb{R} \to \alpha(E) \in \overline{\mathbb{R}}_+$ is an increasing function, constant only on spectral gaps where it takes value in $\{\omega \cdot \nu/2, \nu \in \mathbb{Z}^d\}$; see Ref. 7.

If $[\psi, \bar{\psi}] = 0$ we would have $\chi = ae^{-2i\alpha x} \bar{\chi}$, for some $a \in \mathbb{C}$. But two quasiperiodic functions cannot be equal unless they have the same basic frequencies; see, e.g., Ref. 23. Thus $[\psi, \bar{\psi}] \neq 0$. If $|\psi|$ were not bounded away from 0, there would exist $x_n \uparrow \infty$ for which $\psi(\omega x_n) \to 0$, but this would imply $[\psi, \bar{\psi}] = 0$, a contradiction.

Proof of Theorem 1.2: From the above lemma $[\psi, \bar{\psi}] \neq 0$ (a.e.) on *I*. Thus the Lyapunov number vanishes a.e. on *I* and $I \subset S$ by Kotani's results [see (d), Sec. II]. Now fix *E* (a.e.) in *I* and let *g* be the Deift-Simon function described in Sec. I. Then, for a.e. θ and all *x*,

$$g(x,\theta) = a\psi(x,\theta) + b\overline{\psi}(x,\theta)$$

for some complex numbers a, b depending on θ . Taking absolute values one obtains

$$\begin{aligned} R^{2}(T_{x}\theta) + c|\chi(T_{x}\theta)|^{2} \\ &= de^{2i\alpha x}\chi^{2}(T_{x}\theta) + \bar{d}e^{-2i\alpha x}\bar{\chi}^{2}(T_{x}\theta), \end{aligned}$$

where $c = -(|a|^2 + |b|^2)$ and $d = 2a\overline{b}$. Now take y > 0 and $v \in \mathbb{Z}^d$. Multiply the above equation by $(1/y)\exp[-2i(t + \alpha x) - iT_x \theta \cdot v]$ and integrate it from 0 to y with respect to x. Since R^2 and $|\chi|^2$ belong to $L^1(\mathbb{T}^d)$ we can use the ergodic theorem to let $y \uparrow \infty$ and conclude

$$0 = de^{-2it}(\hat{\chi}^2)_{\nu}, \text{ for all } \nu, \text{ a.e. } (t,\theta) \in \mathbb{T}^{d+1},$$

where $(\hat{})_v$ denote Fourier coefficients. This shows that d = 0, i.e., either a = 0 or b = 0. Theorem 1.2 follows now from Theorem 1.1.

IV. WEAK BLOCH WAVES

1

Here we discuss the a.c. Deift-Simon eigenfunctions g on S and prove Theorem 1.3. Henceforth we will often omit the sentence (E,θ) a.e. in $S \times T^d$.

Since $[g,\overline{g}] = -2i$, r(x) never vanishes and the normalized phase β is a well-defined function from $\mathbb{R} \to \mathbb{R}$. The Schrödinger equation for g implies

$$w'' = 1/r^3 + (v_\theta - E)r, (4.1)$$

$$\beta' = 1/r^2 - \alpha, \tag{4.2}$$

with initial data $r(0) = R(\theta)$, $r'(0) = D_{\omega}R(\theta)$, and $\beta(0) = 0 \pmod{2\pi}$. [The initial value for β is explained by the identification $g(x,\theta,E) = R(\theta)f(x,\theta,E)$, cf. Ref. 3]. Deift and Simon in Ref. 3, extending to S a formula by Johnson and Moser, proved

$$\lim_{x \to \infty} \frac{1}{x} \int_0^x \frac{1}{r^2} = \int_{\mathbf{T}^d} \frac{1}{R^2} = \alpha.$$
 (4.3)

This, together with the Schrödinger equation for g, yields easily the finiteness of

$$\overline{\lim} \frac{1}{x} \int_0^x |g'|^2 \text{ and } \overline{\lim} \frac{|r'| + r + r^{-1}}{x}.$$
 (4.4)

Our next goal is to show that (4.1) and (4.3) imply $R^{-3} \in L^1$ and that R is a distributional solution of (1.4). Let $0 \leq \Phi \in C^{\infty}(\mathbb{T}^d)$ and write $\phi(x) = \phi(x,\theta) \equiv \Phi(T_x\theta)$. Then by the ergodic theorem, (4.4) and (4.1),

$$\int_{\mathbf{T}^{d}} RD_{\omega}^{2} \phi = \lim_{x \to \infty} \frac{1}{x} \int_{0}^{x} R(T_{y}\theta) (D_{\omega}^{2}\Phi) (T_{y}\theta) dy$$

= $\lim \frac{1}{x} \int_{0}^{x} r \phi''$
= $\lim \frac{1}{x} \Big\{ [r\phi']_{0}^{x} - [r'\phi]_{0}^{x} + \int_{0}^{x} r'' \phi \Big\}$
= $\lim \frac{1}{x} \int_{0}^{x} \Big(\frac{1}{r^{3}} + (v_{\theta} - E)r \Big) \phi.$ (4.5)

Another application of the ergodic theorem to positive random variables shows

$$\lim \frac{1}{x} \int_0^x \frac{\phi}{r^3} = \int_{\mathbf{T}^d} \frac{\Phi}{R^3}$$

But then from (4.5) we conclude, a fortiori, that

$$\int_{\mathbf{T}^d} \frac{\Phi}{R^3} = \int_{\mathbf{T}^d} (E - V) R \Phi + \int_{\mathbf{T}^d} R D_{\omega}^2 \Phi < \infty.$$
 (4.6)

In particular, by taking $\Phi \equiv 1$ we get

$$\int_{\mathbf{T}^{d}} \frac{1}{R^{3}} = \int_{\mathbf{T}^{d}} (E - V) R.$$
(4.7)

Now we can repeat the computation in (4.5) with an arbitrary $\Phi \in C^{\infty}(\mathbb{T}^d)$ and get back (4.6). This is the same as saying the *R* is a weak solution of (1.4).

Next we turn to the proof of Theorem 1.3. Since ω satisfies (1.2),

$$B(\theta) \equiv \sum_{\nu \neq 0} \frac{1}{i\omega \cdot \nu} \left(\frac{\hat{1}}{R^2}\right)_{\nu} e^{i\nu \cdot \theta}$$

is seen to be a distribution on $C^{\infty}(\mathbb{T}^d)$. In fact, if $t > \tau + d/2$,

$$\sum_{\nu \neq 0} |\widehat{B}_{\nu}|^{2} (1 + \nu \cdot \nu)^{t}$$
$$\equiv \sum_{\nu \neq 0} \frac{1}{|\omega \cdot \nu|^{2}} \left| \left(\frac{\widehat{1}}{R^{2}} \right)_{\nu} \right|^{2} \frac{1}{(1 + \nu \cdot \nu)^{t}}$$
$$\leq c^{2} \left(\int \frac{1}{R^{2}} \right)^{2} \sum \frac{|\nu|^{2m}}{(1 + \nu \cdot \nu)^{t}} < \infty$$

shows that $B \in H_{-t}(\mathbb{T}^d)$. Now denote by D_{ω}^{-1} the linear operator

$$D_{\omega}^{-1}: \Phi \in \mathbf{0} \to D_{\omega}^{-1} \Phi \equiv \sum_{\nu \neq 0} \frac{\widehat{\Phi}_{\nu}}{i\omega \cdot \nu} e^{i\nu \cdot \theta} \in \mathbf{0}.$$

Then by (4.3), the ergodic theorem, and (4.2) we have

$$\langle B, \Phi \rangle = \langle B, D_{\omega} D_{\omega}^{-1} \Phi \rangle$$

$$= - \langle D_{\omega} B, D_{\omega}^{-1} \Phi \rangle$$

$$= - \langle 1/R^{2} - \alpha, D_{\omega}^{-1} \Phi \rangle$$

$$= - \int \left(\frac{1}{R^{2}} - \alpha\right) \overline{D_{\omega}^{-1} \Phi}$$

$$= - \lim \frac{1}{x} \int_{0}^{x} \left(\frac{1}{R^{2}} - \alpha\right) \overline{D_{\omega}^{-1} \Phi} (T_{y}\theta) dy$$

$$= -\lim \frac{1}{x} \int_{0}^{x} \beta' \overline{D_{\omega}^{-1} \Phi} (T_{y}\theta) dy$$

$$= -\lim \frac{1}{x} \left[\beta \overline{D_{\omega}^{-1} \Phi} (T_{y}\theta)\right]_{0}^{x}$$

$$+ \lim \frac{1}{x} \int_{0}^{x} \beta \overline{\Phi} (T_{y}\theta)$$

$$= \lim \frac{1}{x} \int_{0}^{x} \beta \overline{\Phi} (T_{y}\theta),$$

in which the last equality holds because α is the rotation number of g so that $\lim(1/x)\beta(x) = 0$.

To connect the existence of smooth Bloch waves with regularity properties for (1.4), assume that V is of class $C^{\infty}(\mathbf{T}^d)$ and that R is a smooth solution of (1.4). Then, by

the equation, min R > 0 and $1/R^2 - \alpha$ belongs to $C^{\infty}(\mathbb{T}^d)$. Thus also *B*, as defined above, is a smooth function and we can identify $\beta(x,\theta)$ with the quasiperiodic solution of (4.2) $B(T_x\theta) - B(\theta)$. Unfortunately, regularity properties for such nonlinear equations on tori are difficult to obtain by general PDE methods. (See, however, Ref. 24.)

V. PERIODIC SCHRÖDINGER OPERATORS AS HARMONIC OSCILLATORS

From now on we will be concerned with the constructive part of the theory. In this section we look at periodic Schrödinger operators $L^{(0)}$ from the Hamiltonian point of view described in the Introduction. We show briefly that for each E in the interior $\overset{\circ}{\sigma}$ of the spectrum σ of $L^{(0)}$, the Hamiltonian H_V of Sec. I is conjugated to $\alpha_0 A_0 + \omega_1 A_1$, $(A_0,A_1) \in \mathbb{R}_+ \times \mathbb{R}, \alpha_0 \equiv$ rotation number for $L^{(0)} - E$. (We learned about the integrability of H_V in Ref. 15.) For more details on this and the following sections see Ref. 25.

From Floquet theory²⁶ one knows that, for each $E \in \sigma$, there exist two independent Bloch waves f_0 , \bar{f}_0 of the form

$$f_0(x) \equiv e^{i\alpha_0 x} \chi_0(\omega_1 x)$$

$$\equiv f_1(x) + \frac{e^{i\alpha_0(2\pi/\omega_1)} - f_1(2\pi/\omega_1)}{f_2(2\pi/\omega_1)} f_2(x),$$

$$\chi_0 \in C(\mathbb{T}),$$

with

$$a \equiv \frac{i}{2} [f_0, \bar{f}_0] = \operatorname{Im} f'_0(0) = \frac{\sin(\alpha_0(2\pi/\omega_1))}{f_2(2\pi/\omega_1)} > 0.$$

Now define

$$Q(\theta_0, \theta_1) \equiv \operatorname{Re} F_0(\theta_0, \theta_1), \quad P(\theta_0, \theta_1) \equiv \operatorname{Re} DF_0(\theta_0, \theta_1),$$

where

$$F_{0}(\theta_{0},\theta_{1}) \equiv e^{i(\theta_{0} - (\alpha_{0}/\omega_{1})\theta_{1})}f_{0}(\theta_{1}/\omega_{1}), \quad (\theta_{0},\theta_{1}) \in \mathbb{T}^{2};$$
$$D \equiv \alpha_{0} \frac{\partial}{\partial \theta_{0}} + \omega_{1} \frac{\partial}{\partial \theta_{1}}.$$

One recognizes easily that $x \to F_0(\theta_0 + \alpha_0 x, \theta_1 + \omega_1 x)$ is an eigensolution for $L(v_{\theta_1})$ and that

$$\frac{d}{dx}F_0(\theta_0+\alpha_0x,\theta_1+\omega_1x)=DF_0(\theta_0+\alpha_0x,\theta_1+\omega_1x).$$

Moreover, from

$$\frac{\partial Q}{\partial \theta_0} P - Q \frac{\partial P}{\partial \theta_0} = \frac{i}{2} [f, \bar{f}] = \frac{i}{2} [f_0, \bar{f}_0] = \kappa > 0,$$

it follows readily that the map

$$(r,B,\theta_0,\theta_1) \in \mathbb{R}_+ \times \mathbb{R} \times \mathbb{T}^2 \to (p,B,q,\theta_1)$$
$$\equiv (rP(\theta_0,\theta_1),B,rQ(\theta_0,\theta_1),\theta_1)$$

is a diffeomorphism onto the phase space of H_V , i.e., $\mathbb{R}^3 \times \mathbb{T} - (0,\mathbb{R},0,\mathbb{T})$. Now we can construct a diffeomorphism

C: $(p,B,q,\theta_1) \rightarrow (A_0,A_1,\theta_0,\theta_1) \in \mathbb{R}_+ \times \mathbb{R} \times \mathbb{T}^2$ by setting $A_0 = (r^2/2)\kappa$,

$$A_1 = B + \frac{r^2}{2} \left(\frac{\partial Q}{\partial \theta_1} P - Q \frac{\partial P}{\partial \theta_1} \right).$$

Straightforward computations will show first that $dp \wedge dq + dB \wedge d\theta_1 = dA_0 \wedge d\theta_0 + dA_1 \wedge d\theta_1$, so that **C** is canonical, and then

$$(p^2/2) + \omega_1 B + (q^2/2)(E - V) = \alpha_0 A_0 + \alpha_1 A_1,$$

confirming what we claimed above.

Remark 5.1: Even for V merely continuous, α_0 , f_1 , f_2 , and κ are real analytic functions of E.²⁶ Furthermore, α_0 maps $\sigma(L^{(0)})$ onto $[0, \infty)$, $d\alpha_0/dE > 0$ on $\overset{0}{\sigma}$ and, setting $\partial^0_{\sigma} \equiv \{E_0^0 < E_1^0 < \cdots\}$, $a_k^0 \equiv \alpha_0(E_k^0) = h\omega_1/2$ for some integer h. Now denoting by $e_0(a)$, $a \in R_+$, $a \neq a_k^0$, the inverse function of α_0 , one can easily show that, if $\rho < \omega_1/4$, e_0 admits a holomorphic extension to

 $D(\rho; \mathbf{A}_0) \equiv \bigcup_{a \in \mathbf{A}_0} \{a \in \mathbb{C} \colon |a - a_0| < \rho\},\$

with

$$\mathbf{A}_0 \equiv \bigcup_{k=0}^{\infty} \left[a_k^0 + \rho, a_{k+1}^0 - \rho \right].$$

This will be of later use.

VI. QUASIPERIODIC PERTURBATIONS

Now let
$$\epsilon > 0$$
. Under the canonical transformation

$$(p,B_1,...,B_d,q,\theta_1,...,\theta_d) \in \mathbb{R}^{d+2} \times \mathbb{T}^d - (0,\mathbb{R}^d,0,\mathbb{T}^d) \to (A,\theta)$$

$$\equiv (A_0,A_1,...,A_d,\theta_0,\theta_1,...,\theta_d) \in \mathbb{M} \equiv \mathbb{R}_+ \times \mathbb{R}^d \times \mathbb{T}^{d+1},$$

$$(A_0,A_1,\theta_0,\theta_1) \equiv \mathbb{C}(p,B,q,\theta_1), \quad A_i = B_i, \quad i \ge 2$$

(C as in Sec. V), the Hamiltonian $H_{V+eW}(\cdot;E), E \in \sigma(L^{(0)})$
takes the form

 $H_{\epsilon}(A,\theta;E) \equiv \omega^{(0)} \cdot A + \epsilon A_0 F(\theta)$ with

$$\omega^{(0)} \equiv (\alpha_0, \omega), F(\theta) \equiv - [Q^2(\theta_0, \theta_1)/\kappa] W(\theta_2, \dots, \theta_d).$$

In this section we describe an iterative scheme that will allow us to integrate H_{ϵ} for special values of the parameter Eand small ϵ . Henceforth, it will be more convenient to consider H_{ϵ} parametrized by the rotation number $\alpha_0 \equiv a$ rather than by the eigenvalue E. It will be only later that we shall express our result directly in terms of eigenvalues. We start by considering the *j*th order analog of H_{ϵ} . Let $j \ge 0$ and, for $(A, \theta) \in \mathbf{M}$, let

$$H^{(j)}(A,\theta;a,\epsilon) \equiv \omega^{(j)}(a;\epsilon) \cdot A + \epsilon^{2^{j}} A_{0} F^{(j)}(\theta;a,\epsilon),$$

$$\omega^{(j)} \equiv (\omega_{0}^{(j)},\omega_{1},...,\omega_{d}).$$

Assume that $\omega_0^{(j)}$ and $F^{(j)}$, as functions of a, are holomorphic in

$$D_j \equiv D(\rho_j; \mathbf{A}^{(j)}) \equiv \bigcup_{a_0 \in \mathcal{A}^{(j)}} \{a \in \mathbb{C} \colon |a - a_0| \leq \rho_j \}$$

for some $\mathbf{A}^{(j)} \subset \mathbb{R}$. Also, as a function of $\theta \in \mathbb{R}^{d+1}$, $F^{(j)}$ is required to have holomorphic extension to

$$S_j \equiv S^{d+1}(\xi_j) \equiv \{\theta \in \mathbb{C}^{d+1} \colon |\operatorname{Im} \theta_i| \leq \xi_j\}, \quad \xi_j > 0,$$

with

$$|F^{(j)}||_{\xi_j,\rho_j} \equiv \sup_{(\theta,a)\in S_j\times D_j} |F^{(j)}| \leq M_j$$

independently of ϵ . Notice that because of the analyticity assumptions on V and W, $H_{\epsilon}(A, \theta; e_0(a))$, A_0 as in Remark

5.1, satisfies the above hypothesis, thus we can set $H^{(0)} \equiv H_{\epsilon}$. Now, let $\delta_j < \xi_j/2$ and let us define the main recursive objects

$$\begin{split} \zeta(s) &\equiv 1 + \sum_{\nu \in \mathbb{Z}^{d+1} - 0} |\nu| \Omega(|\nu|) e^{-s|\nu|}, \ s > 0, \\ N_j &\equiv 2^{j+1} \delta_j^{-1} \log \epsilon^{-1}, \\ F_R^{(j)}(\theta) &\equiv \sum_{|\nu| > N_j} \hat{F}_{\nu}^{(j)} e^{i\nu \cdot \theta}, \\ \rho_{j+1} &\equiv \min \left\{ \left(2cN_j \Omega(N_j) \sup_{a \in D_j} \left| \frac{d\omega_0^{(j)}}{da} \right| \right)^{-1}, \frac{\rho_j}{2} \right\}, \\ \xi_{j+1} &\equiv \xi_j - 2\delta_j, \\ A^{(j+1)} &\equiv \{a \in A^{(j)} : |\omega^{(j)}(a) \cdot \nu| \ge 1/c\Omega(|\nu|), \\ \nu \in \mathbb{Z}^{d+1} - 0, \ |\nu| \le N_j \}, \\ D_{j+1} &\equiv D(\rho_{j+1}; A^{(j+1)}), \\ S_{j+1} &\equiv S^{d+1}(\xi_{j+1}). \end{split}$$

Lemma 6.1 (Inductive Lemma): If $a \in \mathbf{A}^{(j+1)}$ and ϵ is small enough, i.e.,

$$K_1 \zeta(\delta_j) \delta_j^{-1} c M_j \epsilon^{2^j} \leqslant 1, \qquad (6.1)$$

where K_1 is a universal constant, then the function

$$(A',\theta) \in \mathbf{M} \to A' \cdot \theta + \epsilon^{2'} A_0 \Phi_j(\theta; a, \epsilon)$$
$$\Phi_j \equiv \sum_{0 < |\nu| < N_j} \frac{\widehat{F}_{\nu}^{(j)}}{-i\omega^{(j)} \cdot \nu} e^{i\nu \cdot \theta}$$

is the generating function of a surjective canonical transformation, $(A,\theta) \rightarrow (A',\theta') = (A'(A,\theta),\theta'(\theta))$, that conjugates $H^{(j)}(A,\theta)$ to

$$H^{(j+1)}(A',\theta';a,\epsilon)$$

$$\equiv H^{(j)}(A(A',\theta'),\theta(\theta'))$$

$$= \omega^{(j+1)} \cdot A' + \epsilon^{2^{j+1}} A'_0 F^{(j+1)}(\theta';a,\epsilon),$$

where

$$\omega^{(j+1)} \equiv (\omega_0^{(j)} + \epsilon^{2^j} \widehat{F}_0^{(j)}, \omega),$$

$$F^{(j+1)}(\theta'(\theta)) \equiv \frac{\partial \Phi_j}{\partial \theta_0}(\theta) F^{(j)}(\theta) + \frac{F_R^{(j)}(\theta)}{\epsilon^{2^j}},$$

Furthermore, $a \in \mathbf{A}^{(j+1)} \to \omega_0^{(j)}(a)$ and $(\theta, a) \in \mathbb{R}^{d+1} \times \mathbf{A}^{(j+1)} \to F^{(j+1)}(\theta; a)$ have holomorphic extensions to, respectively, D_{j+1} and $S_{j+1} \times D_{j+1}$ with

$$\|F^{(j+1)}\|_{\xi_{j+1},\rho_{j+1}} \leq K_2 \zeta(\delta_j) \delta_j^{-(d+1)} c M_j^2 \equiv M_{j+1}, \quad (6.2)$$

in which K_2 is a second universal constant.

Applying this Lemma infinitely many times one can integrate H_{ϵ} for $a \in \mathbf{A}^{(\infty)} \equiv \bigcap_{j=0}^{\infty} \mathbf{A}^{(j)}$.

Theorem 6.2: Let $\{\delta_j\}$ be such that $\sum_{j=0}^{\infty} \delta_j < \xi/2$, let $a \in \mathbf{A}^{(\infty)}$ and let ϵ verify

$$(K_1/K_2)\epsilon\tau\leqslant 1\tag{6.3}$$

with

$$\tau \equiv K_2 \psi c M_0, \quad \psi \equiv \prod_{j=0}^{\infty} \left[\zeta(\delta_j) \delta_j^{-(d+1)} \right]^{1/2^j}.$$

Then the Hamiltonian $H^{(0)}$ is conjugate to

 $H^{(\infty)} \equiv \omega^{(\infty)} \cdot A,$

where
$$\omega^{(\infty)} \equiv (\omega_0^{(\infty)}, \omega)$$
 satisfies

$$c|\omega_{0}^{(\infty)}-a| \leq \frac{1}{K_{2}} \sum_{j=0}^{\infty} (\epsilon \tau)^{2^{j}},$$

$$|\omega^{(\infty)}\cdot \nu| \geq \frac{1}{c\Omega(|\nu|)}, \quad \nu \in \mathbb{Z}^{d+1} - 0.$$
 (6.4)

The (surjective) canonical transformation conjugating $H^{(0)}$ to $H^{(\infty)}$ has the form

$$(A',\theta') \in \mathbf{M} \to (A,\theta)$$

$$\equiv (S(\theta')A',\theta'_{0} + \epsilon \Delta(\theta'_{0}),\theta'_{1},...,\theta'_{d}) \in \mathbf{M}$$
(6.5)

with S a $(d+1) \times (d+1)$ matrix of the form

$$\begin{bmatrix} 1 + \epsilon s_0 & 0 & 0 & \cdots & 0 \\ \epsilon s_1 & 1 & 0 & \cdots & 0 \\ \vdots & & & & \\ \epsilon s_d & 0 & 0 & \cdots & 1 \end{bmatrix}.$$

Moreover, the vector $s \equiv (s_0, ..., s_d)$ and Δ have holomorphic extensions to $S^{d+1}(\xi_{\infty}), \xi_{\infty} \equiv \xi - 2 \sum_{j=0}^{\infty} \delta_j$, and

 $\max\{\|s\|_{\xi_{\infty}}, \|\Delta\|_{\xi_{\infty}}\} \leq (K_1/K_2)\tau.$

Remark 6.3: Examples of $\{\delta_j\}$ and Ω such that $\psi < \infty$ are displayed in Appendix A.

Remark 6.4: Perturbations of the Hamiltonian of the form $h(A,\theta) \equiv \tilde{\omega} \cdot A$ were investigated in Refs. 27 and 28 using Moser's idea of "modified systems."²⁹

Remark 6.5: An easy corollary of Theorem 6.2 is that all the eigensolutions of $L^{(\epsilon)}$ for $E \in \widetilde{\mathbf{E}}^{(\epsilon)} \equiv \alpha_0^{-1}(\mathbf{A}^{(\infty)})$ are quasiperiodic with basic frequencies $(\omega_0^{(\infty)}, \omega)$. Also, since all the transformations involved in the process are close to the identity it is easy to see that $\omega_0^{(\infty)}$ coincides with the rotation number α .

Remark 6.6: From an elementary asymptotic analysis $(E \ge 1)$ of the periodic case, one realizes that $||Q|^2||/K \sim 1/\sqrt{E}$ so that $M_0 \sim ||W||/\sqrt{E}$.

VII. KAM BLOCH WAVES

Even though we already obtained a complete description of the quasiperiodic eigenfunctions of $L^{(\epsilon)}$ for $E \in \widetilde{\mathbf{E}}^{(\epsilon)}$, it is not immediate from the above analysis that such eigenfunctions are of the form $e^{i\alpha x} \chi(\omega x)$. Since this representation is crucial in the application of Theorem 1.2, we proceed now with a direct construction of Bloch waves for values of Ein a set $\mathbf{E}^{(\epsilon)} \subset \sigma(L^{(0)})$, which a priori need not be identical to $\widetilde{\mathbf{E}}^{(\epsilon)}$.

The eigenvalue equation $L^{(\epsilon)}f = Ef$ is equivalent to the first-order system

$$y' = \begin{bmatrix} 0 & 1 \\ V(\theta_1) - E & 0 \end{bmatrix} y + \epsilon W(\theta_2, ..., \theta_d) \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix} y, \quad (7.1)$$
$$\theta' = \omega,$$

with $y = {f \choose f}$. A fundamental matrix for (7.1) at $\epsilon = 0$ is

$$Y = \begin{bmatrix} f_0 & \bar{f}_0 \\ f'_0 & \bar{f}'_0 \end{bmatrix} \quad (f_0 \text{ as in Sec. V}).$$

By setting

$$Y = Te^{Cx}, \quad T \equiv \begin{bmatrix} \chi_0(\omega_1 x) & \bar{\chi}_0 \\ i\alpha_0\chi_0 + \omega_1\chi'_0 & -i\alpha_0\bar{\chi}_0 + \omega_1\bar{\chi}'_0 \end{bmatrix},$$
$$C \equiv \begin{bmatrix} i\alpha_0 & 0 \\ 0 & -i\alpha_0 \end{bmatrix},$$

the system (7.1) becomes, under the change of variable y = Tz,

$$z' = Cz + \epsilon Pz, \quad \theta' = \omega \tag{7.2}$$

with

$$P \equiv \frac{W(\theta_2,\ldots,\theta_d)}{2\kappa} \begin{bmatrix} -i|\chi_0(\theta_1)|^2 & -i\overline{\chi}_0^2(\theta_1) \\ i\chi_0^2 & i|\chi_0|^2 \end{bmatrix}.$$

Notice that $P \in \mathbf{G}_0 \equiv \{G \in \mathbf{G} : \text{tr } \int_{\mathbf{T}^d} G = 0\}$ where **G** denotes the ring of matrix-valued functions on \mathbf{T}^d of the form

$$G = \begin{bmatrix} g & h \\ \overline{h} & \overline{g} \end{bmatrix}$$

Theorem 7.1: If ϵ satisfies the smallness condition

$$(K_1/K_2)\epsilon\tau\leqslant 1, \quad \tau\equiv K_2\psi cM_0,$$

where K_1 and K_2 are suitable universal constants, then one can construct a set $\mathbf{E}^{(\epsilon)} \subset \sigma(L^{(0)})$ and, for each $E \in \mathbf{E}^{(\epsilon)}$, a change of variables $z = (I + \epsilon U)w$, with $U \in G_0$, which transforms (7.2) into the trivial system

$$w' = \begin{bmatrix} i\alpha & 0 \\ 0 & -i\alpha \end{bmatrix} w, \quad \theta' = \omega.$$

Furthermore, U as a function of $\theta \in \mathbf{T}^d$ admits a holomorphic extension to $S^d(\xi_{\infty})$, for a suitable $\xi_{\infty} > 0$, with $||U||_{\xi_{\infty}} \leq (K_1/K_2)\tau$ and α verifies

$$\begin{aligned} |\alpha - |\omega \cdot \nu/2|| \ge \frac{1}{c\Omega(|\nu|)}, \quad \nu \in \mathbb{Z}^d - 0, \quad E \in \mathbb{E}^{(\epsilon)}, \\ \sup_{E \in \mathbb{E}^{(\epsilon)}} |\alpha - \alpha_0| < \epsilon \tau / c. \end{aligned}$$

Remark 7.2: Above we used the same symbols for quantities that are analogous, but not always identical, to the ones appearing in Sec. VI.

The proof of this result is based on a scheme very similar to the one described in Sec. VI: One removes infinitely many times the order of the perturbation of systems like

$$z_{j}' = \begin{bmatrix} i\alpha_{j} & 0\\ 0 & -i\alpha_{j} \end{bmatrix} z_{j} + \epsilon^{2^{j}} P^{(j)} z_{j}, \quad \theta' = \omega \quad (P^{(j)} \in \mathbf{G}_{0}),$$

by the aid of a change of variable $(I + \epsilon^{2^j} U_j(\theta)) z_{j+1} = z_j$. The set $\mathbf{E}^{(\epsilon)}$ will be given by $\alpha_0^{-1}(\mathbf{A}^{(\infty)})$ where $\mathbf{A}^{(\infty)} \equiv \cap \mathbf{A}^{(j)}$, where as in Sec. V, $\mathbf{A}^{(0)}$ is the positive half-line minus suitable intervals of length 2ρ and

$$\mathbf{A}^{(j+1)} \equiv \{ a \in \mathbf{A}^{(j)} : |\alpha_j(a) - \omega \cdot \nu/2| \ge 1/c\Omega(|\nu|), \\ \nu \in \mathbf{Z}^d, \ 0 < |\nu| \le N_j \}$$

with N_j denoting the *j*th cutoff in the Fourier expansion of $P^{(j)}$.

VIII. WHITNEY SMOOTHNESS

In this section we study the *E* dependence of the KAM limits. Following Ref. 16 we say that a function $f: A \subset \mathbb{R} \to \mathbb{R}$ belongs to $C_W^n(A)$ if there exist, on A, functions $f_k, 0 \le k \le n$,

 $f_0 \equiv f$, with the following property: For each $x_0 \in A$ and $\epsilon > 0$ there is a $\delta > 0$ s.t. if $x, x' \in \{y \in A : |y - x_0| < \delta\}$ then

$$\left|f_{k}(x)-\sum_{h=0}^{n-k}\frac{f_{h+k}(x')}{h!}(x-x')^{h}\right| \leq \epsilon |x-x'|^{n-k}.$$
 (8.1)

At interior points this definition coincides with the standard one but the next lemma shows how nontrivial $C_W^n(\mathbf{A})$ functions can arise.

Lemma 8.1: Let $A \subset \mathbb{R}$, $r_j \downarrow 0$, and $\{g_j\}$ be a sequence of holomorphic functions on $D(r_j, A)$ which are real on A. If

$$\sum \|g_j\|_{r_j}r_j^{-n}<\infty,$$

then $g \equiv \Sigma g_j$ belongs to $C_W^n(\mathbf{A})$. *Proof:* Since for any $k \leq n$

$$\sup_{\mathbf{A}}\left|\frac{d^{k}g}{dx^{k}}\right| \leq \sum \left| \left|\frac{d^{k}g_{j}}{dx^{k}}\right| \right|_{r/2} \leq 2^{k} \sum ||g_{j}||_{r_{j}}r_{j}^{-k} < \infty,$$

we can define

$$\frac{d^k g}{dx^k} \equiv \sum \frac{d^k g_j}{dx^k}$$

on *A*. To check that the $d^k g/dx^k$ are the Whitney derivatives of *g*, let *x*, $x' \in \mathbf{A}$, let s = s(|x - x'|) be such that $r_{s+1} \leq |x - x'| < r_s$, and consider the splitting $g = g^{\{s\}} + \tilde{g}^{\{s\}}$ with $g^{\{s\}} \equiv \Sigma^s g_j$. The lemma follows now from $g^{\{s\}} \in C^{\infty}$ $\times (D(r_s; \mathbf{A}) \cap \mathbf{R})$, the inequality

$$\sup_{\mathbf{A}} \left| \frac{d^k \tilde{g}^{[s]}}{dx^k} \right| \leq 2^n \sum_{j=s+1}^{\infty} ||g_j||_{r_j} r_j^{-n}$$

and from $\lim_{|x-x'| \to 0} s(|x-x'|) = \infty$.

The KAM limits of Secs. VI and VII are exactly of the above kind. For example,

$$\omega_0^{(\infty)} = a + \sum_0^\infty \epsilon^{2^j} \hat{F}_0^{(j)}(a),$$

with $\hat{F}_{0}^{(j)}$ holomorphic on $D(\rho_{j}; \mathbf{A}^{(\infty)})$ and one has the following theorem.

Theorem 8.2: If Ω and $\{\delta_i\}$ are such that

$$\prod_{j=0}^{\infty} \zeta(\delta_j)^{1/2^j} < \infty, \qquad (8.2)$$

$$\lim_{j \neq \infty} \frac{\log \Omega(2^{j} \delta_{j}^{-1})}{2^{j}} = 0, \qquad (8.3)$$

and

$$2\left[\epsilon M_0 \rho_0^{-1} + (\epsilon \tau)^2 \sum_{j=0}^{\infty} (\epsilon \tau)^{2j} N_j \Omega(N_j)\right] < 1, \qquad (8.4)$$

then $\omega_0^{(\infty)} \in C_W^{\infty}(\mathbf{A}^{(\infty)})$.

The proof follows easily after noticing that (8.4) yields

$$\sup_{D(\rho_{j+1};\mathbf{A}^{(j-1)})} \left| \frac{d\omega_0^{(j)}}{da} - 1 \right| < \frac{2}{3},$$
 (8.5)

so that

$$\rho_j^{-1} < 4N_{j-1}\Omega(N_{j-1})c. \tag{8.6}$$

For more details see Ref. 25, Sec. 2.6.

Remark 8.3: Whitney smoothness is obviously preserved under composition with smooth functions. Thus $\omega_0^{(\infty)}(\alpha_0(E)) \quad (=\alpha(E)) \quad \text{belongs} \quad \text{to} \quad C^{\infty}_{\mathcal{W}}(\widetilde{\mathbf{E}}^{(\epsilon)}), \, \widetilde{\mathbf{E}}^{(\epsilon)} \\ \equiv \alpha_0^{-1}(\mathbf{A}^{(\infty)}).$

Remark 8.4: While a condition analogous to (8.2) appears in the (analytic) KAM literature,⁸ condition (8.3) is new. This condition is necessary in order to be able to meet the smallness condition (8.4) and, as we shall see, to give a complete description of $A^{(\infty)}$. We also point out that (8.2) and (8.3) are independent (see Appendix A).

IX. STRUCTURE OF KAM SPECTRA

The main theorem in Ref. 16 is that any function $g \in C^n_W(A)$, A closed, can be extended to a $C^n(\mathbb{R})$ function which is real analytic on $\mathbb{R} - A$; a simple corollary of this and of the maximum principle imply

$$\sup_{R} \left| \frac{d^{k}g}{dx^{k}} \right| \leq \max_{A} \left| \frac{d^{k}g}{dx^{k}} \right|, \quad k \leq n.$$

Here we show how to use the above facts in order to give a precise description of the KAM spectrum $E^{(\epsilon)}$.

Denote by **R** the "resonant" set of $a = \alpha_0(E)$ for which we cannot apply the KAM scheme,

$$\mathbf{R} \equiv \mathbf{A}^{(0)} - \mathbf{A}^{(\infty)} \equiv \bigcup_{\substack{j=0\\ v \in \mathbb{Z}^d \\ 0 < |v| < N_j}} \bigcup_{\nu \in \mathbb{Z}^d} \mathbf{R}_{\nu}^{(j)},$$

where for $0 < |\nu| \leq N_j$,

$$\mathbf{R}_{\mathbf{v}}^{(j)} \equiv \{a \epsilon \mathbf{A}^{(j)} \colon |\alpha_j(a) - \omega \cdot \mathbf{v}/2| < 1/c \Omega(|\mathbf{v}|)\}.$$

A condition analogous to (8.4) implies easily that the Whitney extension of the α_i 's satisfy

$$\sup_{R} \left| \frac{d\alpha_j}{da} - 1 \right| \leq \frac{2}{3}.$$

Thus defining

$$a_{j,\nu} \equiv \alpha_j^{-1}(\omega \cdot \nu/2), \quad r_{\nu} \equiv 3/c\Omega(|\nu|),$$

we see that

$$\mathbf{R}_{v}^{(j)} \subset I_{v}^{(j)} \equiv \{ a \in \mathbf{A}^{(0)} : |a - a_{j,v}| < r_{v} \}.$$

This completes the description of $A^{(\infty)}$ and hence, via the smooth map α_0^{-1} , of $E^{(\epsilon)}$.

Finally it is not difficult to show that²⁵

$$\bigcup_{v=0}^{\infty} I_{v}^{(j)} \subset \{ |a - a_{v}| < r_{v}' \}, \quad r_{v}' \equiv \frac{7 + \log \log 3\Omega(|v|)}{c\Omega(|v|)}$$

and $a_v \sim \omega \cdot v/2$. These facts together with the asymptotic evaluation $\alpha_0(E) \sim \sqrt{E}$ yields (1.4).

ACKNOWLEDGMENTS

I gratefully acknowledge instructive discussions with P. Deift, S. Kotani, B. Simon, and T. Spencer. I am also deeply indebted to G. Gallavotti, who largely inspired the second part of this paper, and H. McKean for his continuous help and encouragement. I wish to thank the Mathematics Department of the University of Arizona for its kind hospitality during the period in which this work has been completed.

APPENDIX A: ON A NEW CONDITION IN ANALYTIC KAM

Here we show that conditions (8.2) and (8.3) are independent as announced in Remark 8.4. To do this we give two examples.

(1) Let
$$\Omega(r) = r^m$$
 for some m . Then
(8.2) $\Leftrightarrow \sum \frac{1}{2^j} \log \delta_j^{-1} < \infty$,
(8.3) $\Leftrightarrow (1/2^j) \log \delta_j^{-1} \rightarrow 0 (j \uparrow \infty)$.
(2) Let
 $\Omega(r) = \begin{cases} \exp(r/\log^{\sigma} r), & r \ge e^{\sigma}, \\ \Omega(e^{\sigma}), & 1 \le r \le e^{\sigma}, \end{cases} \quad \sigma > 1;$
(8.2) $\Leftrightarrow \delta_j^{-1}/j^{\sigma}$ is bounded;
(8.3) $\Leftrightarrow \delta_j^{-1}/j^{\sigma} \rightarrow 0 (j \uparrow \infty)$.

In the first example (8.2) is stronger than (8.3) but in the second one the opposite occurs.

Notice that since $\Omega(r) \ge r^{d-1}$ (8.2) implies easily the finiteness of ψ .

APPENDIX B: MOSER-DEIFT-SIMON INEQUALITY ON KAM SPECTRA

Deift-Simon,³ extending an idea of Moser,³⁰ showed that, for general, almost periodic potentials,

$$\lim_{\epsilon \downarrow 0} \frac{\alpha^2 (E+\epsilon) - \alpha^2 (E-\epsilon)}{2\epsilon} \ge 1, \quad E \text{ a.e. in } S.$$
(B1)

Here we want to discuss briefly the constructive version of (B1) for $L^{(\epsilon)}$, namely, we sketch the proof of

$$\frac{d}{dE} \alpha^2 \ge 1, \quad E \in \widetilde{\mathbf{E}}^{(\epsilon)}, \quad \frac{d}{dE} \equiv \text{Whitney derivative.}$$
(B2)

Without loss of generality we can assume that $W_0 = 0$ and, to simplify the Hamiltonian formalism, we consider $V \equiv 0$ in which case $H_{\epsilon W} = \sqrt{E}A_1 + \omega_2A_2 + \cdots + \omega_dA_d$ $- (\epsilon/\sqrt{E})A_1 \sin^2 \theta_1 W(\theta_2,...,\theta_d)$. Then we have

$$\alpha(E) = \sqrt{E} + \epsilon \hat{F}_0^{(0)} + \epsilon^2 \hat{F}_0^{(1)} + O(\epsilon^4)$$

with $F^{(0)} \equiv -(\sin^2 \theta_1/\sqrt{E}) W(\theta_2,...,\theta_d)$ and $F^{(1)}$ as in the inductive Lemma 6.1. Here $\hat{W}_0 = 0$ implies $\hat{F}_0 = 0$. Now, setting $\tilde{\omega} \equiv (\omega_2,...,\omega_d)$, a computation shows that

$$\hat{F}_{0}^{(1)} = \int \frac{\partial \Phi_{0}}{\partial \theta_{1}} F^{(0)} + O(\epsilon)$$

$$\equiv \int \left(\sum_{0 < |\nu| < N_{0}} \frac{\hat{F}_{\nu}^{(0)}}{-i\omega^{(0)} \cdot \nu} \right) F^{(0)} + O(\epsilon)$$

$$= \frac{1}{2} \sum_{\substack{\mu \in \mathbb{Z}^{d-1} \\ |\mu| < N_{0}}} \frac{|\widehat{W}_{\mu}|^{2}}{(\mu \cdot \widetilde{\omega})^{2} - 4E} + O(\epsilon),$$

so that

$$\frac{d\alpha^2}{dE} = 1 + \epsilon^2 \frac{\sqrt{E}}{2} \sum_{|\mu| \le N_0} \frac{|\widehat{W}_{\mu}|^2}{[(\mu \cdot \widetilde{\omega})^2 - 4E]^2} + \frac{1}{\sqrt{E}} O(\epsilon^3).$$

- ¹B. Simon, "Almost periodic Schrödinger operators: A review," Adv. Appl. Math. **3**, 463 (1982).
- ²S. Kotani, "Lyapunov indices determine absolutely continuous spectra of stationary random one-dimensional Schrödinger operators," Taniguchi Symp. SA Katata, 1982, p. 225.
- ³P. Deift and B. Simon, "Almost periodic Schrödinger operators. III. The absolutely continuous spectrum in one dimension," Commun. Math. Phys. **90**, 389 (1983).
- ⁴J. Moser and J. Pöschel, "An extension of a result by Dinaburg and Sinai on quasi-periodic potentials," Comment. Math. Helv. **59**, 39 (1984).
- ⁵Ya. G. Sinai, "Structure of the spectrum of the Schrödinger operator with almost-periodic potential in the vicinity of its left edge," Funkt. Anal. Prilozen. **19**, 42 (1985).
- ⁶S. Kotani, "One-dimensional Schrödinger operators and Herglotz functions," Kyoto University preprint, 1986.
- ⁷R. Johnson and J. Moser, "The rotation number for almost periodic potentials," Commun. Math. Phys. 84, 403 (1982).
- ⁸E. I. Dinaburg and Ya. G. Sinai, "On the one dimensional Schrödinger equation with quasi-periodic potential," Funkt. Anal. Prilozen. 9, 8 (1975).
- ⁹H. Rüssmann, "On the one-dimensional Schrödinger equation with a quasi-periodic potential," Ann. Acad. Sci. **357**, 90 (1980).
- ¹⁰J. Avron and B. Simon, "Almost periodic Schrödinger operators. II. The density of states," Duke Math. J. 50, 369 (1983).
- ¹¹A. N. Kolmogorov, "On the conservation of conditionally periodic motions under small perturbations of the Hamiltonian," Dokl. Akad. Nauk SSSR 98, 527 (1954).
- ¹²V. Arnold, "Proof of a theorem by A. N. Kolmogorov on the invariance of quasi-periodic motions under small perturbation of the Hamiltonian," Russ. Math. Surveys 18, 9 (1963).
- ¹³J. Moser, "On invariant curves of area-preserving mappings of an annulus," Nachr. Akad. Wiss. Gottingen Math. Phys. K1. II, 1 (1962).
- ¹⁴G. Gallavotti, Classical Mechanics and Renormalization Group. In Dynamics Systems, edited by G. Velo and A. S. Wightman (Reidel, Dordrecht, 1985).
- ¹⁵G. Gallavotti, "Quasi integrable mechanical systems," in *Critical Phenomena, Random Systems, Gauge Theories*, edited by K. Osterwalder and R. Stora (Elsevier, New York, 1986).
- ¹⁶H. Whitney, "Analytic extensions of differentiable functions defined in closed sets," Trans. Am. Math. Soc. 36, 63 (1934).
- ¹⁷J. Pöschel, "Integrability of Hamiltonian systems on Cantor sets," Commun. Pure Appl. Math. 35, 220 (1982).
- ¹⁸L. Chierchia and G. Gallavotti, "Smooth prime integrals for quasi-integrable Hamiltonian systems," Nuovo Cimento B 67, 227 (1982).
- ¹⁹G. Gallavotti, The Elements of Mechanics (Springer, Berlin, 1983).
- ²⁰H. Dym and H. P. McKean, Gaussian Processes, Function Theory and the Inverse Spectral Problem (Academic, New York, 1976).
- ²¹M. Reed and B. Simon, *Methods of Modern Mathematical Physics, Vol. I: Functional Analysis* (Academic, New York, 1972).
- ²²E. Coddington and N. Levinson, *Theory of Ordinary Differential Equa*tions (McGraw-Hill, New York, 1955).
- ²³A. Fink, "Almost periodic differential equations," Lecture Notes in Mathematics, Vol. 377 (Springer, Berlin, 1974).
- ²⁴H. Brezis and L. Nirenberg, "Some first-order nonlinear equations on a torus," Commun. Pure Appl. Math. 30, 1 (1977).
- ²⁵L. Chierchia, "Quasi-periodic Schrödinger operators in one dimension, absolutely continuous spectra, Bloch waves and integrable Hamiltonian systems," Ph.D. thesis, New York University, 1986.
- ²⁶W. Magnus and W. Winkler, *Hill's Equation* (Wiley-Interscience, New York, 1966).
- ²⁷H. Rüssmann, "Über die Normalform analytischer Hamiltonscher Differentialgleichungen in der Nahe einer Gleichgewichtslosungen," Math. Ann. **169**, 55 (1967).
- ²⁸G. Gallavotti, "A criterion of integrability for perturbed nonresonant harmonic oscillators. 'Wick ordering' of the perturbations in classical mechanics and invariance of the frequency spectrum," Commun. Math. Phys. 87, 365 (1982).
- ²⁹J. Moser, "Convergent series expansions for quasi-periodic motions," Math. Ann. 169, 136 (1967).
- ³⁰J. Moser, "An example of Schrödinger equation with almost periodic potential and nowhere dense spectrum," Comment. Math. Helv. 56, 198 (1981).

On the construction of perfect Morse functions on compact manifolds of coherent states

S. Berceanu and A. Gheorghe

Central Institute of Physics, Institute of Physics and Nuclear Engineering, Bucharest-Magurele, P. O. Box MG-6, Romania

(Received 14 January 1987; accepted for publication 30 July 1987)

Perfect Morse functions on the manifold of coherent states are effectively constructed. The case of a compact, connected, simply connected Lie group of symmetry, having the same rank as the stationary group of the manifold of coherent states, such that the manifold of coherent states is a Kählerian C-space, is considered. It is proved that the set of perfect Morse functions is dense in the set of energy functions for linear Hamiltonians in the elements of the Cartan algebra of the Lie algebra of the representation of the group considered. It is proved that the maximum number of orthogonal vectors on a coherent vector manifold is equal to the Euler–Poincaré characteristic of the manifold.

I. INTRODUCTION

Particular attention has been paid lately to the application of geometrical ideas and methods in physics. First, geometrical ideas, especially symplectic structures,¹ have an important role in classical mechanics.² Second, the geometry of classical phase space is the starting point of the geometrical quantization program.^{3,4} On the other hand, the right application of the variational principles in quantum physics is conditioned on the establishment of the topological and global geometrical properties of quantum state manifolds.⁵

Global variational methods are studied by means of the Morse theory. It is often useful to find the absolute minima of given functionals. In solving this difficult problem, Morse inequalities⁶⁻⁸ can be successfully used. Morse inequalities imply constraints on the stable and unstable critical manifolds, imposed by the topology of the spaces on which the variational problems are considered. Morse inequalities have been applied to classical mechanics,9 the Hartree-Fock problem,^{10,11} and in connection with supersymmetry theories.¹² The extended Morse inequalities have been used in connection with the symmetry breaking phenomena for a Lie group of symmetry for Higgs-Landau systems.¹³ Atiyah and Bott have applied Morse theory to determine the manifold of minima for the Yang-Mills functional in the equivariant case for Riemannian surfaces and the gauge group U(n).⁷ Another trend of applications of Morse theory in mathematical physics is the use of the Lusternick-Šchnirelman theory (see, e.g., Ref. 14).

Morse inequalities become equalities for perfect Morse functions. The number of critical points of a given index of a perfect Morse function is minimal and is a topological invariant (the Betti number) of the manifold on which the variational problem is based. In the quantum case, many quantum states (e.g., the coherent^{15,16} state manifolds for Lie groups of symmetry) are obtained by embedding of some symplectic manifolds in Hilbert spaces of state vectors.^{5,17} Generally, these manifolds have locally phase space structures, but do not admit global canonical coordinates.^{1,2} However, the corresponding quantum dynamical problems are global analysis problems. The perfect Morse functions just provide an economical method for the effective description of the geometry of quantum state manifolds. These functions exist only on manifolds that have torsion-free cohomology groups.^{18,19} They provide a cell decomposition of the manifold in Morse–Borel cells.²⁰

Many authors have been interested in the construction of perfect Morse functions. The ideas go back to papers of Bott^{18,19} (see also Ref. 21). Let G be a compact connected Lie group with Lie algebra g, t a Cartan subalgebra of g, and $X \in \mathfrak{g}$ a regular element such that t equals the centralizer of X. Then f: $O(Y) \rightarrow \mathbb{R}$, $f(Z) = \langle Z, X \rangle$ is a perfect Morse function and the number of critical points of f is equal to the Euler-Poincaré characteristic of t. Here O(Y) = Ad(G)Y, $Y \in g$, and \langle , \rangle is an Ad-invariant product in g. In our case the coherent state manifold is a compact, simply connected homogeneous Kähler manifold (Kählerian C-space²²) (cf. Refs. 1 and 23, see also Sec. III). The fact that these manifolds admit perfect Morse functions was proved by Bott^{18,19} (see also references to papers of Borel). Frankel has extended the results of Bott to some classes of Kähler manifolds, not necessarily homogeneous.²⁴ Also Frankel gives examples of Morse functions for classical groups using the trace function²⁵ (see also Refs. 26 and 27). Takeuchi²⁸ has studied large classes of real and complex symmetric Rspaces (in the complex case these are the irreducible Hermitian symmetric spaces) that admit perfect ("economical"²⁹) Morse functions, using (Nagano's) zonal spherical functions. Another idea is to embed the manifolds into suitable Euclidean spaces and apply Morse theory to the length function.^{27,30} Flag manifolds³¹⁻³³ are embedded in the projective space by means of Plücker embedding and the perfect Morse function on projective space is used.8

In this paper, perfect Morse functions are constructed on coherent state manifolds admitting a Kählerian C-space structure. These functions have the significance of energy functions associated to the Hamiltonian.

Section II comprises elements of Morse theory on a manifold of quantum states: Morse inequalities and properties of perfect Morse functions. In Sec. III, perfect Morse functions are effectively constructed as energy functions associated with linear quantum systems, described by coherent state manifolds for Lie groups of symmetry. The cellular structure of these manifolds and the maximal orthonormal systems of coherent vectors are deduced.

Some applications to the Slater determinant manifold are outlined in Sec. IV. The last section is devoted to concluding remarks.

II. MORSE INEQUALITIES ON A MANIFOLD OF QUANTUM STATES

(1) The conventional model of quantum mechanics attaches to every physical system a complex Hilbert space \mathcal{H} . To every wave vector $\phi \in \mathcal{H}^* = \mathcal{H} \setminus \{0\}$ the state $\tilde{\phi} = \{e^{i\varphi} \|\phi\|^{-1} \phi | \varphi \in \mathbb{R}\}$ is associated. The complex projective space of states is denoted by $\tilde{\mathcal{H}}$.

Let $\xi: \mathcal{H}^* \to \widetilde{\mathcal{H}}$ be the projection $\xi(\phi) = \tilde{\phi}, \phi \in \mathcal{H}^*$. The unit sphere in \mathcal{H} is defined as $\mathcal{S}(\mathcal{H}) = \{\phi | \phi \in \mathcal{H}; \|\phi\| = 1\}.$

Let Q be a C^{∞} -differentiable manifold and let η : $Q \rightarrow \mathscr{S}(\mathscr{H})$ be an injective and continuous mapping so that each function f^{ψ} : $Q \rightarrow \mathbb{C}$, $f^{\psi}(p) = \langle \psi, \eta(p) \rangle$, $p \in Q$, $\psi \in \mathscr{S}(\mathscr{H})$ is differentiable. Here \langle , \rangle is the symbol of the scalar product of \mathscr{H} . The manifold $M = \eta(Q)$ is called the quantum vector manifold. If the restriction of the projection ξ to M is injective, then $\widetilde{M} = \xi(M)$ is called the quantum state manifold. The structure of the differentiable manifold is canonically carried from Q onto M and \widetilde{M} .

Differentiable functions $f: M \to \mathbb{R}$ will be studied later. As a typical example, functions $f_A(\psi) = \langle \psi, A\psi \rangle$ will be considered in Sec. III, where $\psi \in M$ and A is a self-adjoint operator with M in the domain.

(2) The Morse inequalities used further will be now outlined. $^{6-8,34-36}$

Let $f: M \to \mathbb{R}$ be a smooth function on a compact, $C \simeq m$ dimensional manifold M. Let $\mathscr{C}_{\mathcal{F}}(f)$ be the set of critical points of f, that is, the set of points when df = 0. The Hessian H(f) of f becomes a well-defined quadric form on the restriction of the tangent bundle of M to $\mathscr{C}_{\mathcal{F}}(f)$. The number of negative (resp. 0) eigenvalues of the Hessian matrix at a critical point p is called the index of p (resp. the degeneracy degree) and is denoted by $\lambda_p(f)$. The connected submanifold $N \subset \mathscr{C}_{\mathcal{F}}(f)$ is called the *nondegenerate critical manifold* for f if the degeneracy degree of all $p \in N$ equals the dimension of N. The latter condition means that the restriction $H_N(f)$ of the Hessian to the normal bundle of N is nondegenerate.

A function f on M is called a *nondegenerate (Morse)* function in the extended sense if $\mathscr{C}_{\mathcal{N}}(f)$ is a union of nondegenerate critical manifolds. If N is a nondegenerate critical manifold of M, one may choose a subbundle $v_N^-(f)$ on the tangent bundle to M restricted to N on which $H_N(f)$ is negative definite. The fiber dimension of $v_N^-(f)$, denoted by λ_N , is called the index of N, and we say that N is orientable along f if and only if $v_N^-(f)$ is orientable. In the orientable case the extended Morse (counting) series for the Morse function f is defined, relative to a coefficient field K, as

$$M_t(f,K) = \sum_{N \subset \mathscr{C}_r(f)} t^{\lambda_N} P_t(N,K), \qquad (2.1)$$

whereas in the nonorientable case the field K is \mathbb{Z}_2 . Here $P_t(N,K)$ is the Poincaré series for N relative to a coefficient field K,

$$P_{t}(N,K) = \sum_{i=0}^{\dim N} t^{i} b_{i}(N,K), \qquad (2.2)$$

and $b_i(N,K)$ is the Betti number relative to a field K, i.e., $b_i(N,K) = \dim H^i(N,K)$, $H^i(N,K)$ being the *i*th cohomology group of N with coefficients in the field K.

For a nondegenerate (Morse) function f in the extended sense, the Morse inequalities can be written compactly,

$$M_t(f,K) - P_t(M,K) = (1+t)R(t), \qquad (2.3)$$

where R(t) is a polynomial with non-negative coefficients.

Note also that for a nondegenerate Morse function in the extended sense, Eq. (2.1) can be put in the form

$$M_{t}(f,K) = \sum_{j=0}^{m} t^{j}C_{j}(f,K), \qquad (2.4)$$

$$C_j(f,K) = \sum_{N \subset \mathscr{C}_A(f)} b_{j-\lambda_N}(N,K).$$
(2.5)

The Morse inequalities (2.3) can be written

$$C_{\lambda}(f,K) \ge b_{\lambda}(M,K), \quad 0 \le \lambda \le m, \tag{2.6a}$$

$$\sum_{i=0}^{n} (-1)^{\lambda-i} C_i(f,K)$$

$$\geq \sum_{i=0}^{\lambda} (-1)^{\lambda-i} b_i(M,K), \quad 0 \leq \lambda \leq m, \quad (2.6b)$$

$$\sum_{i=0}^{m} (-1)^{m-i} C_i(f,K)$$

= $\sum_{i=0}^{m} (-1)^{m-i} b_i(M,K) = \chi(M,K),$ (2.6c)

where $\chi(M,K) = P_{-1}(M,K)$ denotes the Euler-Poincaré characteristic of M.

The Morse function f in the extended sense is perfect⁷ (or economical²⁹) if in Eq. (2.3) $R(t) \equiv 0$. From Eqs. (2.6) it follows that if all odd coefficients $C_j(f,K)$ of $M_t(f,K)$ vanish, f is a perfect Morse function in the extended sense (Morse lacunary principle) and $C_j(f,K) = b_j(M,K)$.

When $\mathscr{C}_{\mathcal{F}}(f)$ consists of points alone, Eq. (2.5) reduces to

$$C_{j}(f,K) = \operatorname{card} \{ p \in \mathscr{C}_{\mathscr{F}}(f) | \lambda_{p}(f) = j \}, \quad 0 \leq j \leq m.$$

$$(2.7)$$

Note also that the set of Morse functions on a compact manifold is an open set, dense in the set of differentiable functions on M (relative to the C^2 topology³⁷).

The existence of perfect Morse functions on a compact manifold has other topological consequences. For example, perfect Morse functions can exist only on manifolds that have a torsion-free cohomology group.^{18,19}

The compact manifolds M appearing in this paper admit a Kählerian C-space structure (cf. Ref. 1, p. 168 and Ref. 23), so the field $K = \mathbb{R}$ can be used. However, because the manifolds here are Hodge, it will be sufficient to take $K = \mathbb{Z}$. The existence of perfect Morse functions on Kähler C-spaces is known.^{18,19} More precisely, the homogeneous manifold G/C(T), where G is a compact, connected, semisimple Lie group and C(T) the centralizer of a toral subgroup T of G, admits perfect Morse functions and also admits a homogeneous Kähler structure.^{18-20,38} Here we shall present a concrete realization of a perfect Morse function in local coordinates, indicating also the globalization for a compact manifold of coherent states admitting a Kählerian C-space structure. Also these manifolds have complex analytic cellular decomposition in the sense of Morse–Borel,^{20,28,39–41} and this cell decomposition is the one as the CW complexes, identical to the decomposition of certain classical groups like the Grassmann manifolds.⁴²

III. PERFECT MORSE FUNCTIONS ON COMPACT MANIFOLDS OF COHERENT STATES

(1) Lately, the coherent states have been intensively studied.^{15,16} Now we shall introduce in a convenient manner the strictly necessary elements used in this paper.

A quantum system with symmetry (in the sense of Wigner⁴³ and Bargmann⁴⁴) is characterized by a continuous homomorphism $\tilde{\pi}$ of a topological group G into a group of transformations \tilde{G} of the space $\tilde{\mathcal{H}}$ which leaves invariant the transition probabilities

 $((\tilde{\pi}(g)\tilde{\phi},\tilde{\pi}(g)\tilde{\psi})) = ((\tilde{\phi},\tilde{\psi})), \quad g \in G, \quad \tilde{\phi},\tilde{\psi} \in \widetilde{\mathcal{H}},$

where

 $\begin{aligned} &((\tilde{\phi},\tilde{\psi})) = \|\phi\|^{-1} \|\psi\|^{-1} |\langle \phi,\psi\rangle|^2, \quad \phi,\psi \in \mathcal{H}^*. \\ &\text{Let } \tilde{\psi}_0 \in \widetilde{\mathcal{H}} \text{ be a fixed state. The } \tilde{G} \text{ orbit containing } \tilde{\psi}_0, \\ &\widetilde{M} = \tilde{G}\tilde{\psi}_0 = \{\tilde{\pi}(g)\tilde{\psi}_0 | g \in G\}, \end{aligned}$ (3.2)

(3.1)

is called the manifold of coherent states, and every $\tilde{\phi} \in \widetilde{M}$ is called a coherent state.

The closed group

$$K = \{h \mid h \in G; \ \tilde{\pi}(h)\tilde{\psi}_0 = \tilde{\psi}_0\}$$
(3.3)

is also considered. Then $\tilde{\pi}(K)$ is a stationary group of the state $\tilde{\psi}_0$ and there exists the bijection $\tilde{\xi}: G/K \to \tilde{M}$, defined by $\tilde{\xi}(g') = \tilde{\pi}(g)\tilde{\psi}_0$, where $g' = gK \in G/K$. Let the notation $\tilde{\psi}(g') = \tilde{\xi}(g')$. Evidently, $\tilde{\xi}(\theta') = \tilde{\psi}_0$, where θ is the unity element of the group G.

Further, G will be taken as a compact, connected Lie group. The fact that rank $G = \operatorname{rank} K$ will result in the next paragraph from general considerations, when restrictions on the fixed state $\tilde{\psi}_0$ and representation $\tilde{\pi}$ will be imposed. Also, further it will result that K is the centralizer of a toral subgroup of G and, moreover, is connected. The manifold \tilde{M} will be endowed with the canonical differentiable (even real analytic) structure induced by $\tilde{\xi}$ from the homogeneous space G/K, hence \tilde{M} is a quantum state manifold diffeomorphic with G/K. On the other hand, according to a theorem of Wigner and Bargmann,^{43,44} there exists a continuous, unitary representation π of the group G onto the complex Hilbert space \mathcal{H} , such that

$$\widetilde{\pi(g)\psi} = \widetilde{\pi}(g)\widetilde{\psi}, \quad g \in G, \quad \psi \in \mathscr{H}^*.$$
(3.4)

Then, there exists the cross section $\sigma: \widetilde{M} \to \mathscr{S}(\mathscr{H})$, where σ is an injective mapping such that $\xi(\sigma(\widetilde{\psi}(g'))) = \widetilde{\psi}(g'), g' \in G/K$. Let the notations $M = \sigma(\tilde{M}), \psi(g') = \sigma(\tilde{\psi}(g')), \psi_0 = \sigma(\tilde{\psi}_0)$. It follows that M is a differentiable manifold relative to the structure induced by σ from \tilde{M} . The manifold M is named a *coherent vector manifold*, and every $\phi \in M$ is called a *coherent vector*. However, M is not a submanifold of \mathcal{H} .

Let us also introduce the mapings $\xi_0: M \to \widetilde{M}$ and $\eta: G/K \to M$ by $\xi_0(\psi(g')) = \widetilde{\psi}(g') = \widetilde{\xi}(g')$, $g' \in G/K$, and $\eta(g') = \sigma^0 \widetilde{\xi}(g') = \sigma(\widetilde{\psi}(g')) = \psi(g')$. Evidently, ξ_0 , η , and $\widetilde{\xi} = \xi_0^0 \eta$ are diffeomorphisms. It follows that the coherent state manifold \widetilde{M} , the coherent vector manifold M, and the homogeneous space G/K are diffeomorphic. It can be noted that M is a system of coherent vectors of type (π, ψ_0) in the sense of Perelomov¹⁵ [local sections of the holomorphic line bundle $\xi^{-1}(\widetilde{M}) \to \widetilde{M}$].

The previous construction is natural from the point of view of the interpretation of symmetries in quantum mechanics. Moreover, this construction is important for establishing the global properties of coherent state and coherent vector manifolds.

(2) Further, we need some elements of the theory of finite-dimensional representations of compact, connected, simply connected Lie groups (see, for example, Ref. 45). Because we deal with coset spaces G/K (rank K = rank G), there is no loss in generality in assuming that G is semisimple and simply connected (see, e.g., Ref. 46, p. 490).

Now, let T be a Cartan subgroup of the group G, and suppose that ψ_0 is a j-dominant weight vector relative to the representation π . Let \mathcal{H}_j denote the complex linear covering of the manifold M. Let π_j denote the restriction of the representation π to \mathcal{H}_j . Hence π_j is a finite-dimensional unitary irreducible representation of the group G onto the complex linear space \mathcal{H}_j . There exists an isomorphism π'_j of the Lie algebra g of the group G onto the Lie algebra $\pi'_j(g)$ of the group $\pi_i(G)$ such that

$$\pi_i(e^X) = \exp(\pi_i(X)), \quad X \in \mathfrak{g}, \tag{3.5}$$

where $e: \mathfrak{g} \to G$ and $exp: \pi'_j(\mathfrak{g}) \to \pi_j(G)$ are exponential mappings.

Let us also fix a Cartan–Weyl base⁴⁷ of the complexification g^c of the Lie algebra g, with elements h_i , e_{α} , $1 \le i \le r$, $\alpha \in \Delta$, where r is the rank of the group G and Δ is a system of nonzero roots satisfying the commutation relations

$$[h_i,h_j] = 0, \quad [h_i,e_\alpha] = \alpha(h_i)e_\alpha, \quad (3.6a)$$

$$[e_{\alpha}, e_{-\alpha}] = \sum_{i=1}^{r} \alpha(h_i)h_i, \qquad (3.6b)$$

$$[e_{\alpha}, e_{\beta}] = 0, \quad \alpha + \beta \notin \Delta \cup \{0\}, \tag{3.6c}$$

$$[e_{\alpha}, e_{\beta}] = N_{\alpha\beta} e_{\alpha+\beta}, \quad \alpha + \beta \in \Delta, \tag{3.6d}$$

where $1 \le i \le j \le r$, α , $\beta \in \Delta$, and $\alpha(h_i)$, $N_{\alpha\beta}$ are real structure constants (cf. Ref. 47, pp. 166–171). If $\alpha + \beta \ne 0$, then the roots e_{α} , e_{β} are orthogonal relative to the Killing form $B(\cdot, \cdot)$, and the relation $\alpha(h) = B(h, h_{\alpha})$, where h_{α} $= [e_{\alpha}, e_{-\alpha}]$ was taken into account in Eq. (3.6b). The root system Δ is included in the dual t* of the Cartan algebra t of T, and by means of the mapping $\alpha \rightarrow h_{\alpha}$, Δ can be embedded in t.

The elements h_i $(1 \le i \le r)$ form a base of the complexification t^c of the Cartan algebra t.

Let Π denote the set of simple roots. The simple roots can be chosen such that

$$\alpha_i(h_j) = \delta_{ij}, \quad \alpha_i \in \Pi, \quad 1 \leq i, \quad j \leq r.$$
(3.7)

Every root $\alpha \in \Delta$ is a linear combination of simple roots from Π with integer coefficients of the same sign. If these coefficients are non-negative, the root α is called a positive root. Let Δ_+ denote the set of positive roots.

Let \mathscr{C} be the fundamental Weyl chamber

$$\mathscr{C} = \left\{ w \middle| w = \sum_{i=1}^{r} \lambda_{i} w_{i}, \ \lambda_{i} \in \mathbb{Z}_{+} \right\},$$
(3.8)

where the fundamental weights $w_i \in t^*$ verify the relations

$$2(w_i,\alpha_j) = \delta_{ij}(\alpha_j,\alpha_j), \quad 1 \leq i, \quad j \leq n.$$
(3.9)

Here (\cdot, \cdot) denotes the Euclidean scalar product in t^* .

The representations π_j and π'_j can be uniquely extended to the group homomorphism π^*_j : $G^c \to \pi_j(G^c)$ and, respectively, Lie algebra isomorphism π^*_i : $g^c \to \pi^*_i$ (g^c) by

$$\pi_j^*(e^Z) = \exp(\pi_j^{*'}(Z)), \quad Z \in \mathfrak{g}^c,$$
 (3.5')

where $\pi_j^{*'}(g^c)$ is the complexification of the Lie algebra $\pi_j'(g)$, but G^c and $\pi_j^{*}(G^c)$ denote the complexification of the groups G and, respectively, $\pi_j(G)$. Of course, $\pi_j^{*}(G^c)$ and $\pi_i^{*'}(g^c)$ are the sets of linear operators on \mathcal{H}_j . Also let

$$H_i = \pi_j^{*'}(h_i), \quad E_\alpha = \pi_j^{*'}(e_\alpha),$$
 (3.10)

where $1 \le i \le n$ and $\alpha \in \Delta$. According to the theory of compact representations,³⁸ the *j*-dominant weight can be chosen to belong to the Weyl chamber \mathscr{C} and

$$H_{i}\psi_{0} = j_{i}\psi_{0}, \quad 1 \leq i \leq r,$$

$$E_{-\alpha}\psi_{0} \neq 0, \quad \alpha \in \Delta',$$

$$E_{-\alpha}\psi_{0} = 0, \quad \alpha \in \Delta \setminus \Delta',$$
(3.11)

where $j = (j_1, ..., j_r), j_i = (\omega_i, j), i = 1, ..., r$, and

$$\Delta' = \{ \alpha | \alpha \in \Delta; \ (j, \alpha) < 0 \}. \tag{3.12}$$

The base of the real Lie algebra g [resp. $\pi'_j(g)$] is made of elements ih_k , $i(e_{\alpha} + e_{-\alpha})$, $e_{\alpha} - e_{-\alpha}$ [resp., the anti-Hermitian operators iH_k , $i(E_{\alpha} + E_{-\alpha})$, $E_{\alpha} - E_{-\alpha}$, $1 \le k \le r, \alpha \in \Delta$].

Also, the unitarity of the representation π_i implies

$$H_i^+ = H_i, \quad E_\alpha^+ = E_{-\alpha}, \quad 1 \leq i \leq r, \quad \alpha \in \Delta, \qquad (3.13)$$

where A^+ denotes the adjoint of the operator A.

Let \mathfrak{p} denote the complex Lie algebra with base h_i , $e_{-\alpha}$, $1 \leq i \leq r$, $\alpha \in \Delta \setminus \Delta'$. The complex Lie subgroup $P = e^{\mathfrak{p}}$ of the group G^c is a parabolic group corresponding to the parabolic subset $-(\Delta \setminus \Delta')$ of Δ (cf. Ref. 45, p. 54) (Borel group if $\Delta' = -\Delta_+$). The representation $\tilde{\pi}_j$ is irreducible, $K = G \cap P$, and

$$P = \{ g | g \in G^c; \ \pi_j(g) \psi_0 = \tilde{\psi}_0 \}.$$
(3.14)

It follows that there is a diffeomorphism of homogeneous spaces $c: G/K \to G^c/P$, defined by $c(gK) = gP, g \in G.^{38}$ The following facts are known about the homogeneous space G^c/P strictly associated to the irreducible representation $\tilde{\pi}_j$ of dominant weight $j.^{1,20,22,38}$ The space G^c/P is algebraic. The isotropy group $K = G \cap P$ is connected, with the same rank as G and it is the centralizer of a torus in G (K is a maximal torus if P is a Borel subgroup). The principal bundle $P \rightarrow G^c \rightarrow G^c/P$ is holomorphic. Details on the geometric realization are provided by the Bott-Borel-Weil theorem (see, e.g., Ref. 45, p. 201). Following the denomination of Wang,²² G^c/P is a C-space, endowed with a Kählerian structure.

Let W(G) denote the Weyl group⁴⁷ associated with G, defined as the quotient W(G) = N(T)/C(T) of the normalizer

$$N(T) = \{g | g \in G; \ gTg^{-1} = T\}$$
(3.15)

of the Cartan group T by the corresponding centralizer $C(T) = \{g | g \in G; gt = tg, any t \in T\}$. Similarly, W(K) denotes the Weyl group associated with the group K. Let $\Sigma \subset N(T)$ be a set of elements such that the quotient space W(G)/W(K) is made of the coset classes (sC(T))W(K), $s \in \Sigma$. Let us define the mappings $\tilde{s}: \Delta \to \Delta$, for every $s \in \Sigma$, by

$$f_{\alpha}(h) = \alpha(s^{-1}hs), \quad h \in \mathfrak{t}, \quad \alpha \in \Delta.$$
 (3.16)

The action of \tilde{s}_{α} on an element k of the space generated by Δ is the reflection

$$\tilde{s}_{\alpha}(k) = k - 2(\alpha, k)\alpha/(\alpha, \alpha), \qquad (3.17)$$

and for every $\alpha \in \Delta$, there exists $\beta \in \Delta$ and $\gamma \in \Pi$ such that $\alpha = \tilde{s}_{\beta}(\gamma)$.

(3) With the previous facts, the Kählerian structure of the homogeneous space G^{c}/P will be carried onto the manifold M of coherent vectors. Moreover, the Kählerian structure of the homogeneous space G^{c}/P will be also transported onto the homogeneous space G/K. To get the Kählerian structure on G^{c}/P , here we use local coordinates. The effective construction of a Kähler metric on a compact manifold of coherent states in the case of nonsingular highest weight was pointed out in Ref. 23. The same argument as in Ref. 23 provides the globalization in the present construction.

Let us introduce the vectors

$$\Phi_{z} = \exp\left(\sum_{\alpha \in \Delta'} z_{\alpha} E_{-\alpha}\right) \psi_{0}, \qquad (3.18)$$

$$\psi_{z} = \|\Phi_{z}\|^{-1} \Phi_{z} = \psi(g') \in \mathcal{M}, \qquad (3.19)$$

for

$$c(g') = \exp\left(\sum_{\alpha \in \Delta'} z_{\alpha} E_{-\alpha}\right) P, \qquad (3.20)$$

where z belongs to the *m*-dimensional Euclidean complex space \mathbb{C}^m and

$$2m = \dim \tilde{M} = \dim M = \dim G / K = \dim G^{c} / P.$$
(3.21)
Let the notation

$$\mathscr{V}_0 = \{\psi_z | z \in \mathbb{C}^m\}$$

and let us consider the homeomorphism $h: \mathscr{V}_0 \to \mathbb{C}^m$, defined by $h^{-1}(z) = \psi_z$. Let also the notation $\mathscr{V}_s = \pi_j(s) \mathscr{V}_0$ and $h_s = h^{\circ} \pi_j(s)^+$ for every $s \in \Sigma$. Then $(\mathscr{V}_s)_{s \in \Sigma}$ is an open, finite covering of the manifold M, and the collection of local charts $(\mathscr{V}_s, h_s)_{s \in \Sigma}$ generates an atlas of a Kählerian manifold with the fundamental two-form on \mathscr{V}_0

$$\omega = -i \sum_{\alpha,\beta\Delta} g_{\alpha\beta} \, dz_{\alpha} \wedge d\overline{z}_{\beta},$$

where

$$g_{\alpha\bar{\beta}} = \frac{\partial^2 F(z,\bar{z})}{\partial z_{\alpha} \, \partial \bar{z}_{\beta}}, \qquad (3.22)$$

$$F(z,\bar{z}) = \ln \langle \Phi_z, \Phi_z \rangle.$$
(3.23)

Note that ω is nonsingular everywhere.²³

The Kählerian structure of the manifold M is induced by the mapping ξ_0 onto the coherent state manifold \widetilde{M} .

To every linear operator A on \mathcal{H}_j , a function $f_A: \tilde{M} \to \mathbb{C}$ can be associated by

$$f_{\mathcal{A}}(\tilde{\psi}) = \langle \psi, \mathcal{A}\psi \rangle, \quad \psi \in \mathcal{M}, \tag{3.24}$$

called the *covariant symbol.*⁴⁸ If A is a Hermitian operator associated to an observable, then f_A is a real function, and $f_A(\tilde{\psi})$ can be interpreted as an expectation value of the considered observable. If H is a Hamiltonian, then f_H is called an energy function. The manifold \tilde{M} (being Kählerian) is, of course, also symplectic, hence the symbol f_A associated to a Hermitian operator admits the interpretation of a classical observable.

Let us now introduce the notation

$$\lambda_{\alpha} = \sum_{i=1}^{r} \epsilon_{i} \alpha(h_{i}), \quad \epsilon = (\epsilon_{1}, \dots, \epsilon_{r}) \in \mathbb{R}^{r}.$$
(3.25)

After this long preparation, we are ready to state the main theorem of the paper.

Theorem 1: Let M be the homogeneous compact manifold of coherent states, diffeomorphic with G/K and the Kählerian C-space G^{c}/P . Here G is a compact, connected, simply connected Lie group and K is a closed subgroup (connected and rank $K = \operatorname{rank} G$). Then the energy function f_{H} associated to the Hamiltonian

$$H = \sum_{i=1}^{r} \epsilon_i H_i, \quad \epsilon = (\epsilon_1, \dots, \epsilon_r) \in \mathbb{R}^r, \tag{3.26}$$

is a perfect Morse function in the extended sense. The nondegenerate critical manifolds of \widetilde{M} are described by $\tilde{\psi}_{z,er}^s = \pi_j(s)\psi_{z,er}$, $s \in \Sigma$, where

$$\psi_{z,er} = \exp\left(\sum_{\alpha \in \Delta' \setminus \Delta'} z_{\alpha} E_{-\alpha}\right) \psi_{0}, \qquad (3.27)$$

$$\Delta'' = \{\gamma | \gamma \in \Delta', \ \lambda_{\gamma} \neq 0\}.$$
(3.28)

Every nondegenerate critical manifold has even index

$$\lambda_{s,c*} = 2 \operatorname{card} \{ \alpha | \alpha \in \Delta'', \ \lambda_{\bar{s}_{\alpha}} > 0 \}, \quad s \in \Sigma.$$
(3.29)

Moreover, if

$$\lambda_{\tilde{s}_{\alpha}} = \sum_{i=1}^{r} \epsilon_{i} \tilde{s}_{\alpha_{i}}(h_{i}) \neq 0, \quad s \in \Sigma, \quad \alpha \in \Delta', \quad (3.30)$$

then the associated energy function f_H is a perfect Morse function.

Proof: Let us consider the function $f: \mathbb{C}^m \to \mathbb{R}$ defined by

$$f(z,\bar{z}) = \langle \psi_z, H\psi_z \rangle, \quad z \in \mathbb{C}^m.$$
(3.31)

The function f can be put into the form

$$f(z,\overline{z}) = \sum_{i=1}^{r} \epsilon_{i} j_{i} - \sum_{\gamma \in \Delta'} \lambda_{\gamma} z_{\gamma} \frac{\partial F(z,\overline{z})}{\partial z_{\gamma}}.$$
 (3.32)

Indeed, from Eqs. (3.18) and (3.19) it follows that

$$f(z,\overline{z}) = \|\Phi_z\|^{-2} \langle \Phi_z, He^X \psi_0 \rangle, \qquad (3.33)$$

where

$$X = \sum_{\alpha \in \Delta'} z_{\alpha} E_{-\alpha}.$$
 (3.34)

But

$$He^{X} = e^{X} \sum_{n>0} (n!)^{-1} (-\operatorname{ad} X)^{n} H, \qquad (3.35)$$

where, taking into account Eqs. (3.26), (3.25), (3.34), (3.10), and (3.6a),

$$(\operatorname{ad} X)H = [X,H] = \sum_{\alpha \in \Delta'} z_{\alpha} \lambda_{\alpha} \frac{\partial X}{\partial z_{\alpha}},$$
 (3.36)

$$(\operatorname{ad} X)^{n}H = \sum_{\alpha \in \Delta'} z_{\alpha} \lambda_{\alpha} (\operatorname{ad} X)^{n-1} \frac{\partial X}{\partial z_{\alpha}}, \quad n \ge 2. \quad (3.37)$$

Equation (3.35) becomes

$$He^{X} = e^{X}H - \sum_{\alpha \in \Delta'} z_{\alpha}\lambda_{\alpha}e^{X}$$

$$\times \sum_{n>0} (-1)^{n} ((n+1)!)^{-1} (\operatorname{ad} X)^{n} \frac{\partial X}{\partial z_{\alpha}}$$

$$= e^{X}H - \sum_{\alpha \in \Delta'} \lambda_{\alpha}z_{\alpha} \frac{\partial e^{X}}{\partial z_{\alpha}}.$$
(3.38)

From Eqs. (3.11) and (3.26),

$$H\psi_0 = \sum_{i=1}^r \epsilon_i j_i \psi_0, \qquad (3.39)$$

and also

$$He^{X}\psi_{0} = e^{X}\sum_{i=1}^{r} \epsilon_{i} j_{i}\psi_{0} - \sum_{\alpha \in \Delta^{\prime}} z_{\alpha}\lambda_{\alpha} \frac{\partial \Phi_{z}}{\partial z_{\alpha}}.$$
 (3.40)

Introducing the latter expression in Eq. (3.33) and taking into account Eq. (3.23), the relation (3.32) is proved. Note that if the subspaces generated by $E_{-\alpha}$, where $\alpha \in \Delta'$ and $\alpha \in \Delta \setminus \Delta'$ are Abelian, all terms with n > 1 in Eq. (3.35) are 0. This situation corresponds to Hermitian symmetric manifolds (cf. Ref. 49, p. 1191).

The point $z \in \mathbb{C}^m$ is a critical point of the function f if and only if

$$\frac{\partial f}{\partial \overline{z}_{\beta}} = -\sum_{\gamma \in \Delta'} \lambda_{\gamma} z_{\gamma} \frac{\partial^2 F}{\partial z_{\gamma} \partial \overline{z}_{\beta}} = 0, \quad \beta \in \Delta'.$$
(3.41)

Since the fundamental two-form ω is nondegenerate, the matrix $\Gamma = (g_{\gamma \overline{\beta}})_{\gamma, \beta \in \Delta'}$ is nonsingular. Then Eq. (3.41) is equivalent to the conditions

$$\lambda_{\gamma} z_{\gamma} = 0, \quad \gamma \in \Delta'. \tag{3.42}$$

The manifold of critical points of f is

$$\mathscr{E}_{\mathcal{F}}(f) = \{ z | z \in \mathbb{C}^m; \ z_{\gamma} = 0, \ \gamma \in \Delta^n \}, \tag{3.43}$$

where Δ'' is given by Eq. (3.28).

4

If the point $z_0 \in \mathscr{C}_{\mathcal{F}}(f)$, then (3.41) and (3.42) imply

$$\left(\frac{\partial^2 f}{\partial z_{\alpha} \partial z_{\beta}}\right)_{z_0} = \left(\frac{\partial^2 f}{\partial \overline{z}_{\alpha} \partial \overline{z}_{\beta}}\right)_{z_0} = 0, \quad \alpha, \beta \in \Delta', \quad (3.44)$$

$$(W_{z_0}(f))_{\alpha\beta} = \left(\frac{\partial^2 f}{\partial z_{\alpha} \ \partial \overline{z}_{\beta}}\right)_{z_0} = -\lambda_{\alpha} g_{\alpha\overline{\beta}} = -g_{\alpha\overline{\beta}}\lambda_{\beta}.$$
(3.45)

From Eq. (3.45) it results that the positive definite matrix Γ and the matrix $\Lambda = (\lambda_{\alpha} \delta_{\alpha\beta})_{\alpha,\beta \in \Delta'}$ are simultaneously diagonalizable. So, the Hessian matrix of the function $f \text{ in } z_0$,

$$H_{z_{0}}(f) = 2U \begin{pmatrix} W & 0 \\ 0 & W' \end{pmatrix} U^{+},$$

$$U = 2^{-1/2} \begin{pmatrix} \mathbf{1}_{m} & \mathbf{1}_{m} \\ i\mathbf{1}_{m} & -i\mathbf{1}_{m} \end{pmatrix},$$
(3.46)

admits λ_{\perp} (resp. λ_{0}) negative (resp. 0) eigenvalues, where

$$\lambda_{-} = 2 \operatorname{card} \{ \gamma | \gamma \in \Delta'; \ \lambda_{\gamma} > 0 \}, \lambda_{0} = 2 \operatorname{card} \{ \gamma | \gamma \in \Delta'; \ \lambda_{\gamma} = 0 \},$$
(3.47)

and 1_n denotes the unit matrix of the group GL(*n*,C) and W^t denotes the transpose of matrix W. In Eq. (3.46) the Hessian is expressed in the real coordinates (x_{α}, y_{α}) , where $z_{\alpha} = x_{\alpha} + iy_{\alpha}$,

$$(H(f))_{\alpha\beta} = \begin{pmatrix} \frac{\partial^2 f}{\partial x_{\alpha} \partial x_{\beta}} & \frac{\partial^2 f}{\partial x_{\alpha} \partial y_{\beta}} \\ \frac{\partial^2 f}{\partial y_{\alpha} \partial x_{\beta}} & \frac{\partial^2 f}{\partial y_{\alpha} \partial y_{\beta}} \end{pmatrix}, \quad \alpha, \beta \in \Delta'.$$
(3.48)

Equations (3.44)–(3.47) imply that the Hessian matrix (3.48) has the nonzero determinant $det(H(f))_{\alpha,\beta\in\Delta^{-}}\neq 0$, hence the manifold (3.43) is a nondegenerate critical manifold. A point (vector) of this manifold has the expression (3.27).

Now, choosing $f^s = f_H \circ \xi_0 \circ \pi_j(s) \circ h^{-1}$, $s \in \Sigma$, it follows that the nondegenerate critical manifolds of \tilde{M} for f^s are described by $\tilde{\psi}^s_{z,cr} = \pi_j(s)\psi_{z,cr}$. Every nondegenerate critical manifold has even index (3.29).

It follows that in Eq. (2.1) the fiber of the nondegenerate critical manifold has even dimension, and, moreover, the coefficients $C_i(f)$ (2.5) are all even, hence the lacunary principle of Morse in the extended sense is applicable.

The first part of Theorem 1 was proved.

If $\Delta' = \Delta''$, then $z_0 = 0$ is the only critical point of the function *f*. This point is a nondegenerate one and has an even index (3.47).

If the conditions (3.30) are fulfilled, then it results that the critical points of the function f_H are the distinct states $\tilde{\psi}^s$, $s \in \Sigma$, where $\psi^s = \pi_j(s)\psi_0$. Every critical state $\tilde{\psi}^s$ is nondegenerate and has an even index

$$\lambda_s = 2 \operatorname{card} \{ \alpha | \alpha \in \Delta'; \ \lambda_{s_{\alpha}} > 0 \}.$$
(3.49)

Now, the lacunary principle of Morse is applied and the Theorem is proved.

Remark 1: The set of perfect Morse functions is dense in the set of energy functions associated to Hamiltonians which are linear in H_i , i = 1,...,r.

Remark 2: As in the formulation of Bott's theorem²¹ presented in the Introduction, Theorem 1 is also true for Hamiltonians H such that *iH* belongs to the Lie algebra $\pi'_j(g)$. Note also that the case $iH \in \pi'_j(g)$ corresponds to coherent state preserving Hamiltonians for semisimple Lie groups.⁴¹ We also remember that Frankel has proved the existence of perfect Morse functions for linear Hamitonians in the generators of the Lie algebra of a group of isometries of a compact connected Kähler manifold.²⁴ Here the Kählerian structure of the compact coherent manifold follows, as was already pointed out, from the fact the \tilde{M} is the orbit of the *j*-dominant weight (see also Ref. 1, p. 168). The existence of perfect Morse functions on Kählerian C-spaces was proved by Bott.^{18,19,21}

Remark 3: The perfect Morse function f_H is related to the spherical zonal function used by Takeuchi in the case of irreducible symmetric complex and real *R*-spaces.²⁸ The complex *R*-spaces in the paper²⁸ of Takeuchi are the *C*spaces of Wang.²² In the complex case, the function of Takeuchi is constructed only for Hermitian symmetric spaces, whereas the proof presented here is valid for Kählerian *C*spaces. In fact, in the case of complex Grassmann manifold Takeuchi²⁹ claims that the economical spherical function reduces to the "nice" function (of Hattori), which is a particular case of the function constructed in Sec. IV [see Eq. (4.11)]. On the other hand, the proof presented here in the case of the compact coherent state manifold is more direct.

As an application to Theorem 1, a description follows of a maximal system of coherent vectors.

Theorem 2: If $\mathcal{T} \subset M$ is a maximal orthogonal system of coherent vectors, then there exists an element $g \in G$ such that

$$\mathcal{T} = \{\pi_j(g)\psi^s|s\in\Sigma\}$$
(3.50)

and the number of vectors in \mathcal{T} is equal to the Euler-Poincaré characteristic $\chi(M)$.

Proof: The perfect Morse function f_H appearing in Theorem 1 induces a cellular structure onto the manifold of coherent states \widetilde{M} . For every $s \in \Sigma$ there exists in the cell $\widetilde{\mathcal{V}}_s$ = $\xi_0(\mathcal{V}_s)$ one and only one critical state $\widetilde{\psi}^s$ of the function f_H . Observing that $\{\widetilde{\mathcal{V}}_s\}_{s\in\Sigma}$ is an open covering of the manifold \widetilde{M} , that the group \widetilde{G} acts transitively on M, and that

$$\langle \psi^{s}, \psi^{s'} \rangle = \delta_{ss'}, \langle \psi_{0}, \psi_{z} \rangle \neq 0, \qquad (3.51)$$

for $s, s' \in \Sigma$, $z \in \mathbb{C}^m$, Eq. (3.50) follows. On the other hand, from the Morse equality (2.6c) with all odd Betti numbers equal to 0, it can be deduced that Σ and \mathcal{T} have exactly $\chi(M)$ elements.

IV. APPLICATIONS TO THE MANIFOLD OF SLATER DETERMINANTS

From the preceding section, it can be noted that the results outlined there are applicable to large classes of quantum systems with symmetry, where G is compact and rank- $G = \operatorname{rank} K$, and admit a Kählerian C-space structure. The situation when the manifold \widetilde{M} is endowed with a homogeneous complex structure includes the Hermitian symmetric spaces [in fact, this situation occurs when dim(center K) = 1, and there are only two invariant structures (see, e.g., Ref. 46, p. 505). This situation also corresponds exactly to the case when the subspaces generated by $E_{-\alpha}$, where $\alpha \in \Delta'$ and $\alpha \in \Delta \setminus \Delta'$, both can be chosen as Abelian subalgebras of $\pi_j^{\sharp'}(g^c)$.⁴⁹

The irreducible Hermitian symmetric spaces (see Ref. 47, p. 518) appear frequently in different branches of physics. For example, $SU(p+q)/S(U(p) \times U(q))$ appears in connection with Hartree-Fock problems¹⁰ or Grassmann nonlinear σ models,⁵⁰ SO(2n)/U(n) is present in fermion models¹⁶ or in quantum field theoretical models as the Gross-Neveu type model,⁴⁸ SO(p + 2)/SO(p) × SO(2) is used in models with spins with SO(p) symmetry,⁵¹ and Sp(n)/U(n) is the manifold of Lagrangian subspaces of the phase space.⁵² Also, Fordy and Kulish have studied the generalization of nonlinear Schrödinger equation by getting the field components to take values in irreducible Hermitian symmetric spaces.⁵³ Here we sketch a short application of the general Theorem 1 to the Hartree–Fock case. Details will be presented elsewhere.

The time-dependent variational principle of Hartree– Fock is based on the Slater determinant manifold.⁵⁴ A geometrical description of this manifold will be achieved by means of perfect Morse functions and topological constraints onto the energy function will also be established.

Let us consider a fermion Fock space \mathcal{H} with vacuum state $\widetilde{\Omega}$. Let \mathcal{H}_j denote the linear complex subspace of \mathcal{H} with a base formed by the *n*-particle vectors

$$a_{p_1}^+ a_{p_2}^+ \cdots a_{p_n}^+ \Omega, \quad 1 \leq p_1 < p_2 \cdots < p_n \leq n',$$
 (4.1)

where a_p^+ , a_p^- (p = 1,...,n') are the usual fermion creation and annihilation operators, respectively.

The Hartree–Fock Hamiltonian is a Hermitian operator on \mathcal{H}_j , realized as a second degree polynomial in bifermion operators

$$C_{pq} = a_p^+ a_q, \quad 1 \le p, \le n'. \tag{4.2}$$

From the usual anticommutation relations of the annihilation and creation operators, it follows that

$$[C_{pq}, C_{p'q'}] = \delta_{qp'} C_{pq'} - \delta_{pq'} C_{p'q}, \quad 1 \le p, q \le n'.$$
(4.3)

The Lie algebra of anti-Hermitian operators on \mathscr{H}_j , which are linear combinations with complex coefficients $x_{pq}, x_{pq} = -\bar{x}_{qp}$, of the operators C_{pq} , is isomorphic with the Lie algebra u(n') of the group U(n'). But the group U(n') is canonically isomorphic with the product of groups $U(1) \times SU(n')$, being, of course, connected, simply connected. Hence it can be chosen that G = SU(n'). Let us also fix the initial vector

$$\psi_0 = a_1^+ a_2^+ \cdots a_n^+ \Omega. \tag{4.4}$$

Since

$$C_{pq}\psi_{0} = \delta_{pq}\psi_{0}, \quad 1 \le p, q \le n,$$

$$C_{pq}\psi_{0} = 0, \quad 1 \le p \le n', \quad n+1 \le q \le n',$$

$$C_{pq}\psi_{0} \ne 0, \quad n+1 \le p \le n', \quad 1 \le q \le n,$$
(4.5)

 ψ_0 is a vector of dominant weight $j = (1^n 0^{n'-n})$ of the unitary irreducible representation π_i of SU(n') on \mathcal{H}_i .

A manifold of coherent states of dimension m = 2n(n' - n) is obtained,

$$\widetilde{M} = \{ \widetilde{\psi} | \widetilde{\psi} = \widetilde{\pi_j(g)} \psi_0; \ g \in \mathrm{SU}(n') \},$$
(4.6)

and is diffeomorphic with the complex Grassmann manifold $G_n(\mathbb{C}^{n'}) \approx U(n')/U(n) \times U(n'-n)$.^{11,55} The manifold of coherent vectors M is diffeomorphic with \widetilde{M} , as in Sec. III. Here we introduce the notation

$$\Phi_{z} = \exp\left(\sum_{p=n+1}^{n'} \sum_{q=1}^{n} \bar{z}_{qp} C_{pq}\right) \psi_{0},$$

$$\Sigma = \{\Delta^{\sigma} | \sigma \in S(n,n') \},$$

$$\Delta^{\sigma} = (\delta_{i\sigma(j)})_{1 \le i, j \le n'},$$

(4.7)

where $z = (z_{pq})_{n+1 and the set <math>S(n,n')$ of $C_n^{n'}$ Schubert symbols comprises all permutations σ : {1,2,...,n'} \rightarrow {1,2,...,n'} with the property that its restrictions to the subsets {1,2,...,n} and {n + 1,...,n'} are increasing. The atlas of the manifold M is generated by the open covering $\{\mathscr{V}_{g}\}_{g\in S(n,n')}$, where

$$\mathscr{V}_{\sigma} = \{\pi_j(\Delta^{\sigma})\psi_z | z \in \mathbb{C}^{n(n'-n)}\}.$$
(4.8)

Here the local coordinate mappings $h_{\sigma}: \mathscr{V}_{\sigma} \to \mathbb{C}^{n(n'-n)}$ are

$$h_{\sigma}(\psi_{z\sigma}) = z, \quad \pi_{j}(\Delta^{\sigma})^{+}\psi_{z\sigma} = \psi_{z}, \quad \sigma \in S(n,n').$$
(4.9)

The manifold *M* is called the *Slater determinant mani*fold (in the second quantization).

According to Theorem 1, the energy function f_{H_0} associated with the Hamiltonian

$$H_0 = \sum_{i=1}^n c_i C_{ii}, \quad c_1 < c_2 < \cdots < c_n,$$
(4.10)

is a perfect Morse function.

By direct calculation¹¹ it is obtained, for $f = f_{H_0} \circ \xi_0$ $\circ \pi_i(\Delta^{\sigma}) \circ h_{\sigma}^{-1}$,

$$f(z) = \operatorname{Tr}[(U + zVz^{+})(\mathbf{1}_{n} + zz^{+})^{-1}],$$

$$U = (c_{p}\delta_{pq})_{1 < p, q < n},$$

$$V = (c_{p}\delta_{pq})_{n+1 < p, q < n'}.$$
(4.11)

Then, the function $f: \mathbb{C}^{n(n'-n)} \to \mathbb{R}$ has a unique critical point z = 0. This point is nondegenerate, and the Hessian matrix has double degenerate eigenvalues: $c_q - c_p$, $1 \le p \le n \le q \le n'$.

The critical sets of the energy function f_{H_0} consists of the critical states $\tilde{\psi}_{\sigma}(\sigma \in S(n,n'))$, where

$$\psi_{\sigma} = a_{\sigma(1)}^+ a_{\sigma(2)}^+ \cdots a_{\sigma(n')}^+ \Omega, \qquad (4.12)$$

and the critical state
$$\tilde{\psi}_{\sigma}$$
 has the index

$$\lambda_{\sigma} = 2 \operatorname{card}\{(p,q) | 1 \leq p \leq n < q \leq n'; \sigma(p) > \sigma(q) \}.$$
(4.13)

It is funny to recover directly the Betti numbers of the manifold M of Slater determinants⁵⁵

$$b_{2\lambda+1} = 0, \quad 0 \leq \lambda \leq n(n'-n),$$

$$b_{2\lambda} = \operatorname{card}\{(\omega_1, \dots, \omega_n \in \mathbb{Z}^n_+ \mid 0 \leq \omega_1 \leq \dots \leq \omega_n \leq n'-n;$$

$$\omega_1 + \omega_2 + \dots + \omega_n = \lambda\}, \quad (4.14)$$

and also the Euler-Poincaré characteristic $\chi(M) = C_n^{n'}$.

Remark 4: Observing that the uniparticle states are eigenstates with eigenvalues c_i , i = 1,...,n, it can be noted that perfect Morse functions are obtained when the uniparticle space is nondegenerate.

Remark 5: If the Hartree-Fock energy function f_H admits only nondegenerate critical states, then the Morse inequalities (2.6) are satisfied, where $0 < \lambda < m = 2n(n' - n)$, $C_{\lambda}(f)$ is the number of critical states of index λ for f_H , and the Betti numbers are given by (4.14). The problem of existence of Hartree-Fock states for Hamiltonians that do not have spurious states was analyzed in Ref. 10. The Hamiltonians that give the minimum number of Hartree-Fock states were effectively constructed here. It can also be shown that if the eigenvalues of the uniparticle states c_i are not all distinct, then the set of critical states are Grassmann submanifolds of the Grassmann manifold.

Remark 6: Perfect Morse functions of the type (4.11)

have been constructed in the mathematical literature for the Grassmann manifold (see, e.g., Refs. 31–33, where in the last reference, a calculation with components of matrices is performed). Note that the proof from Ref. 11 is more direct than the proofs of Refs. 32 and 33. Also in the case of Grassmann manifolds there are direct proofs that the Morse-Borel cells are identical to the Schubert cells.⁴⁰

V. CONCLUSION AND DISCUSSION

The Morse inequalities in the classical and extended sense have been used for the effective construction of perfect Morse functions on a manifold of quantum states and quantum vectors. Actually, the case of a compact Lie group and of a compact manifold of coherent states having the structure of a Kählerian C-space was considered. It was proved that the set of perfect Morse functions is dense in the set of energy functions for linear Hamiltonians in the elements of the Cartan algebra of the Lie algebra of the representation of the considered group, which was chosen compact, connected, simply connected, and having the same rank as the stationary group of the manifold of coherent states. By the construction of perfect Morse functions, the cellular decomposition of the manifold of coherent states was found. For every energy function, the Morse inequalities are satisfied, the odd Betti numbers being in this case 0. Particularly, the energy function admits a number of critical points at least equal to the Euler-Poincaré characteristic of the manifold of coherent states. It is also proved that the Euler-Poincaré characteristic is equal to the maximum number of orthogonal vectors. The existence of perfect Morse functions for Kähler C-spaces is well established.^{18,19,21} Here we have presented a construction of a perfect Morse function with physical significance as an energy function on the manifold of compact coherent states, diffeomorphic with Kählerian C-spaces.

The results obtained permit a correct approach to different problems of the classical limit and variational principles on manifold of coherent states from the point of view of global analysis and geometry. In this context, the cellular decomposition of the coherent state manifold induced by the perfect Morse functions is useful for solving asymptotical problems of actual interest such as the following: the classical limit of quantum collective models, ^{56,57} 1/N expansions in quantum field theory, ⁴⁸ semiclassical behavior of functional integral based on coherent states, ⁵⁸ Lagrangian analysis, ⁵⁹ and the connection between the geometric quantization method and the functional integral. ⁵²

- ¹V. Guillemin and S. Sternberg, *Symplectic Techniques in Physics* (Cambridge U. P., London, 1984).
- ²R. Abraham and J. E. Marsden, *Foundations of Mechanics* (Benjamin, London, 1978), 2nd ed.

- ³B. Konstant, "Quantization and unitary representation," in *Modern Analysis and Applications, Lecture Notes in Mathematics,* Vol. 170 (Springer, New York, 1970), p. 87.
- ⁴J. M. Souriau, Structure des systèmes dynamique (Dunod, Paris, 1970).
- ⁵T. W. B. Kibble, Commun. Math. Phys. 65, 189 (1974).
- ⁶M. Morse, *The Calculus of Variations in the Large* (Am. Math. Soc., Providence, RI, 1934), Vol. XVIII.
- ⁷M. F. Atiyah and R. Bott, Philos. Trans. R. Soc. London Ser. A **308**, 523 (1982).
- ⁸J. Milnor, "Morse theory," Ann. Math. Stud. 51 (1963).
- ⁹S. P. Novikov, Usp. Mat. Nauk 37, 3 (1982) (in Russian).
- ¹⁰G. Rosensteel and E. Ihring, J. Math. Phys. 21, 2297 (1980).
- ¹¹S. Berceanu, "Lectures on Morse theory," Institute of Physics and Nuclear Engineering, preprint FT-23-1984, Bucharest, 1984.
- ¹²E. Witten, J. Differ. Geom. **17**, 661 (1982); "Holomorphic Morse inequalities," Princeton preprint.
- ¹³P. J. Houston and S. Sen, J. Phys. A: Math. Gen. 17, 1163 (1984).
- ¹⁴C. H. Taubes, Commun. Math. Phys. 86, 257, 299 (1982); 91, 235 (1983); 97, 473 (1985).
- ¹⁵A. M. Perelomov, Commun. Math. Phys. 26, 222 (1972).
- ¹⁶A. M. Perelomov, "Generalized coherent states. General case," Institute of Theoretical and Experimental Physics preprint ITEP-144, Moscow, 1983.
- ¹⁷J. P. Prevost and G. Vallée, Commun. Math. Phys. 76, 289 (1980).
- ¹⁸R. Bott, Proc. Natl. Acad. Sci. USA 40, 586 (1954).
- ¹⁹R. Bott, Bull. Soc. Math. France 84, 251 (1956).
- ²⁰A. Borel, Proc. Natl. Acad. Sci. USA 40, 1147 (1954).
- ²¹R. Bott, "The geometry and representation theory of compact Lie groups," in Proceedings of the SRL/LMS Research Symposium on Representations of Lie groups, London Mathematics Society, Lecture Notes Series 34 (Cambridge U. P., Cambridge, 1979).
- ²²H. C. Wang, Am. J. Math. 76, 1 (1954).
- ²³E. Onofri, J. Math. Phys. 16, 1087 (1975); "Quantization theory for homogeneous Kähler manifolds," IFPR-T-038, Parma preprint, 1974.
 ²⁴T. Frankel, Ann. Math. 70, 1 (1959).
- ²⁵T. Frankel, "Critical submanifolds of the classical groups and Stiefel manifolds," in *Differential and Combinatorial Topology, A Symposium in Honour of M. Morse*, edited by S. S. Cairns (Princeton U. P., Princeton, NJ, 1965).
- ²⁶S. Ramanujam, J. Indian Math. Soc. 32, 243 (1968).
- ²⁷S. Ramanujam, Tohoku Math. J. 21, 343 (1969).
- ²⁸M. Takeuchi, J. Fac. Sci. Univ. Tokyo, Sec. 1, 12, 81 (1965).
- ²⁹M. Takeuchi, Osaka J. Math. 6, 283 (1969).
- ³⁰S. Ramanujam, Osaka J. Math. 6, 243 (1969).
- ³¹T. Hangan, J. Differ. Geom. 2, 363 (1968); "Sur la géometrie différentielle des varietés de drapeaux," *Symposia Math. X*, 211, Bologna, 1972.
 ³²J. Alexander, Ill. J. Math. 15, 672 (1971).
- ³³T. Sugawara, Mem. Fac. Sci. Kyushu Univ. Ser. A: **34**, 107 (1980).
- ³⁴R. Bott, Ann. Math. **60**, 248 (1954).
- ³⁵R. Bott, "Morse theory and Yang-Mills equations," in Differential Geometrical Methods in Mathematical Physics, Lecture Notes in Mathematics, Vol. 836 (Springer, Berlin, 1980), p. 269.
- ³⁶R. Bott, Bull. Am. Math. Soc. 7, 331 (1982).
- ³⁷J. Milnor, Lecture on h-cobordism Theory (Princeton U.P., Princeton, NJ, 1965).
- ³⁸J. P. Serre, "Répresentations linéaires et espaces homogènes kählériens des groupes de Lie compacts," in *Sém Bourbaki*, 6^{-ème} année, 1953/1959, Exp. 100 (Secretariat mathematiques, Paris, 1959), 2nd edition.
- ³⁹B. Kostant, Ann. Math. 77, 72 (1963).
- ⁴⁰B. Parker, Indiana Univ. Math. J. 24, 95 (1974).
- ⁴¹G. D'Ariano, M. Rasetti, and M. Vadacchino, J. Phys. A: Math. Gen. 18, 1295 (1985).
- ⁴²C. Ehresmann, Ann. Math. 35, 396 (1934).
- ⁴³E. P. Wigner, Group Theory and Its Applications to Quantum Mechanics of Atomic Spectra (Academic, London, 1959).
- 44V. Bargmann, J. Math. Phys. 5, 862 (1964).
- ⁴⁵G. Warner, Harmonic Analysis on Semi-Simple Lie Groups (Springer, Berlin, 1972), Vol. 1.
- ⁴⁶A. Borel and F. Hirzebruch, Am. J. Math. 80, 458 (1958).
- ⁴⁷S. Helgason, *Differential Geometry, Lie Groups and Symmetric Spaces* (Academic, New York, 1978).
- ⁴⁸F. A. Berezin, Commun. Math. Phys. 63, 131 (1978).
- ⁴⁹J. A. Wolf, Bull. Am. Math. Soc. 75, 1121 (1969).
- ⁵⁰A. M. Perelomov, Physica D 4, 1 (1981).
- ⁵¹W. Fuller and A. Lenard, Commun. Math. Phys. 67, 69 (1979).

- ⁵²V. Guillemin and S. Sternberg, Geometric Asymptotics, Mathematical Surveys, No. 14 (Am. Math. Soc., Providence, RI, 1977).
- ⁵³A. D. Fordy and P. P. Kulish, Commun. Math. Phys. 89, 427 (1983).
- ⁵⁴B. G. Giraud and D. J. Rowe, J. Phys. Lett. 40, 177 (1979).
- ⁵⁵S. S. Chern, Complex Manifolds (University of Chicago, Chicago, 1956).
- ⁵⁶E. H. Lieb, Commun. Math. Phys. **31**, 327 (1973).
- ⁵⁷B. Simon, Commun. Math. Phys. 71, 247 (1980).
- ⁵⁸L. G. Yaffe, Rev. Mod. Phys. **54**, 407 (1982).
- ⁵⁹J. Leray, Analyse lagrangienne et mecanique quantique: une structure mathematique apparentée aux developement asymptotiques et a l'indice de Maslov, Séminaire du College de France 1976–1977 (R.C.P. 25, I.R.M.A., Strassbourg, France, 1978).

Exponential time-evolution operator for the time-dependent harmonic oscillator

Francisco M. Fernández

Instituto de Investigaciones Fisicoquímicas Teóricas y Aplicadas (INIFTA), División Química Teórica, Sucursal 4, Casilla de Correo 16, 1900 La Plata, Argentina

(Received 15 April 1987; accepted for publication 22 July 1987)

The time-evolution operator for the time-dependent harmonic oscillator $H = \frac{1}{2} \{\alpha(t)p^2 + \beta(t)q^2\}$ is exactly obtained as the exponential of an anti-Hermitian operator. The method is

based on the equations of motion for the coordinate and momentum operators in the

Heisenberg representation. The problem is reduced to solving the classical equations of motion.

I. INTRODUCTION

The time-dependent Schrödinger equation (units are used so that $\hbar = 1$),

$$i\frac{\partial}{\partial t}U = HU, \quad U(t=0) = 1,$$
 (1)

can be exactly solved for a number of simple problems. The solution is most conveniently written as a product of unitary operators.^{1,2} However, in some cases it is preferable to look for a solution of the form

$$U=e^{-iA}, \qquad (2)$$

where A is a Hermitian operator. When no exact solution is available exponential perturbation theories may be used. One of them is the celebrated Magnus expansion³ that enables one to write A as $A_1 + A_2 + A_3 + \cdots$. The convergence of this series was extensively discussed.⁴⁻⁶

The applicability of the Magnus expansion to spin systems in periodic magnetic fields was investigated by Fel'dman⁵ and Salzman.⁶ To this end exactly solvable problems prove to be very useful.⁵

II. THE TIME-DEPENDENT HARMONIC OSCILLATOR

The purpose of this paper is to obtain the exponential time-evolution operator for the time-dependent harmonic oscillator

$$H = \frac{1}{2} \{ \alpha(t) p^2 + \beta(t) q^2 \},$$
(3)

where α and β are real functions of t and [q,p] = i. Since the operators

$$H_1 = \frac{1}{2}q^2, \quad H_2 = \frac{1}{2}(qp + pq), \quad H_3 = \frac{1}{2}p^2,$$
 (4)

span a three-dimensional Lie algebra there is a solution of (1) of the form (2) with¹

$$A = b_1(t)H_1 + b_2(t)H_2 + b_3(t)H_3,$$
(5)

where b_j , j = 1,2,3 are real functions of t and $b_j(0) = 0$. It remains to determine the form of the b's.

The coordinate and momentum operators in the Heisenberg representation $q_t = U^+ qU$ and $p_t = U^+ pU$, respectively, satisfy

$$\frac{d}{dt}p_t = -\beta q_t, \quad \frac{d}{dt}q_t = \alpha p_t. \tag{6}$$

Therefore they can be written

$$p_t = P_p p + P_q q, \quad q_t = Q_p p + Q_q q, \tag{7}$$

where the P's and Q's obey the classical equations of motion

$$\frac{d}{dt}P_{u} = -\beta Q_{u}, \quad \frac{d}{dt}Q_{u} = \alpha P_{u}, \quad u = p,q, \quad (8)$$

with the boundary conditions $P_p(0) = Q_q(0) = 1$ and $P_q(0) = Q_p(0) = 0$. Only three of the four P's and Q's are independent since $[q_i,p_i] = i$ which leads to $P_pQ_q - P_qQ_p = 1$.

On using the well-known Baker-Campbell-Hausdorff formulas it is not difficult to prove that

$$P_{p} = \cos \omega - b_{2} \omega^{-1} \sin \omega, \quad P_{q} = -b_{1} \omega^{-1} \sin \omega,$$

$$Q_{p} = b_{3} \omega^{-1} \sin \omega, \quad Q_{q} = \cos \omega + b_{2} \omega^{-1} \sin \omega,$$
(9)

where $\omega^2 = b_1 b_3 - b_2^2$. When $\omega^2 < 0$ it is convenient to write $\omega = i\xi$, $\omega^{-1} \sin \omega = \xi^{-1} \sinh \xi$, and $\cos \omega = \cosh \xi$. Clearly, the form of the *b*'s is completely determined by Eqs. (8) and (9) and the problem of finding *U* is solved.

Pechukas and Light² showed that U can also be written $U = \exp(ia_1H_1)\exp(ia_2H_2)\exp(ia_3H_3)$, where $a_1 = P_q/Q_q$, $a_2 = -\frac{1}{2}\ln Q_q$, and $a_3 = -Q_p/2Q_q$. Therefore the results of this paper also reveal the quite complex relationship between the two forms of writing U.

Apparently, the exponential time-evolution operator for the model just discussed was not obtained before. The method proposed here can be applied to other problems where the Hamiltonian operator can be written as a linear combination of the operators in a Lie algebra. Although the resulting expressions for U appear to be difficult to disentangle, they may be useful, for instance, in checking the applicability of the Magnus expansion.^{2,3}

Spin systems in periodic magnetic fields^{5,6} can be treated exactly in the same way if it is taken into account that

$$M_{x} = \frac{1}{2}(q^{2} + p^{2}), M_{y} = \frac{1}{2}i(q^{2} - p^{2}),$$

$$M_{z} = -\frac{1}{2}i(qp + pq),$$
(10)

and the Pauli spin matrices σ_x , σ_y , and σ_z obey the same commutation rules.

¹J. Wei and E. Norman, J. Math. Phys. 4, 575 (1963).

- ²P. Pechukas and J. C. Light, J. Chem. Phys. 44, 3897 (1966).
- ³W. Magnus, Commun. Pure Appl. Math. 7, 649 (1954); D. W. Robinson, Helv. Phys. Acta. **36**, 140 (1963).

⁴M. M. Maricq, Phys. Rev. B 25, 6622 (1982).

⁵E. B. Fel'dman, Phys. Lett. A 104, 479 (1984).

⁶W. R. Salzman, Chem. Phys. Lett. **124**, 531 (1986).

Quadratic zeros of Racah 6/ coefficients: A geometrical approach

J. J. Labarthe

Laboratoire Aimé Cotton, Université de Paris XI, F91405 Orsay Cedex, France

(Received 19 May 1987; accepted for publication 19 August 1987)

It is shown that the projective symmetries of the polynomial Φ of quadratic 6*j* coefficients form the symmetrical group S_6 . Nonlinear rational symmetries of Φ are found. Partial parametrizations of the zeros of Φ are presented.

I. INTRODUCTION

Recently some progress has been made in finding zeros of 6*j* coefficients for the quadratic case.^{1,2} Unlike the linear case^{3,4} all solutions are not known.

In this paper we study the quadratic case from the point of view of geometry. All concepts of algebraic geometry used in this paper (projective coordinates, multiple points, base points, etc.) can be found in Semple and Roth.⁵ Beyer, Louck, and Stein⁶ have recently found that the polynomial of 6*j* coefficients is invariant under a symmetrical group S₆. In Sec. II, by studying some geometrical properties of the surface associated to the polynomial we also arrive at this symmetry group and show moreover that there is no other projective symmetry of the polynomial in the quadratic case. In Sec. III, we present rational nonlinear symmetries of the polynomial that we conjecture to generate an infinite group. In Sec. IV, two-dimensional cubic surfaces of zeros are found from geometrical considerations. These cubic surfaces, like those introduced by Bremner⁷ for the linear case, can be rationally parametrized. However these parametrizations do not provide a complete parametrization of the fourdimensional surface of all zeros. By considering a family of cubic surfaces we arrive at a parametrization of a three-dimensional subvariety of zeros.

We take a quadratic 6j coefficient in the form

$$\begin{cases} (T+A+C-2)/2 & (S+A+B-2)/2 & (S+T+B+C-4)/2 \\ (T+B)/2 & (S+C)/2 & (S+T+A-2)/2 \\ \end{cases}$$

where A, B, C, S, and T are integers ≥ 2 .

This 6*j* coefficient has exactly three decompositions in terms of extremal elements⁴:

$$(2-k)e_1 + (S-k)e_2 + (T-k)e_3 + (A-2+k)e_4 + (B-2+k)e_5 + (C-2+k)e_6 + ke_7,$$

with k = 0, 1, or 2 in the notations of Ref. 4.

The polynomial part of this 6*j* is given by

 $\Phi(A,B,C,S,T)$

$$= A(A-1)B(B-1)C(C-1) + 2ABCSTU + S(S-1)T(T-1)U(U-1), (1)$$

where

U=2-A-B-C-S-T.

Using the notation $A^{(k)} = A(A-1)\cdots(A-k+1)$ and considering it as a symbolic exponential we can write Eq. (1) as

 $\Phi = (ABC + STU)^{(2)}.$

II. SOME GEOMETRIC PROPERTIES OF THE SURFACE $\Phi = 0$

We consider A, B, C, S, and T as belonging to C, the field of complex numbers, and view the solutions of $\Phi = 0$ as an algebraic surface (which we also call Φ) of complex dimension 4 in the projective space $P^{5}(C)$. More precisely, we introduce the homogenizing coordinate z, and the homogenization of Φ , which is of degree 6: $\Phi_H(A,B,C,S,T,z) = \Phi(A/z,B/z,C/z,S/z,T/z)z^6.$

The projective coordinates A, B, C, S, T, z of a point of $P^{5}(\mathbb{C})$ are defined apart from a multiplicative factor. Points at infinity are those for which z = 0.

Let us first study the multiple points of the surface Φ . To determine if a point M of the surface Φ is a multiple point we proceed in the following way. We take any other point $Q(\neq M)$ in $P^{5}(\mathbb{C})$ and parametrize the points of line MQ as

$$R(t) = Mt + Q(1-t) \quad (t \in \mathbb{C}).$$

The equation of degree 6 in t

$$\varphi_{M,Q}(t) = \Phi_H(R(t)) = 0,$$

for M and Q fixed, gives the intersection of line MQ with the surface Φ . The point M is multiple of multiplicity k if t = 0 is a root of multiplicity k of $\varphi_{M,Q}(t) = 0$ for any point Q.

The only multiple points of the surface Φ are triple and double points. There are 31 triple points, ten of which are at infinity (Table I, where the coordinates will be defined a little below). We use a notation for the triple points that makes easy the description of the system L of lines joining these points (Table II). The double points of Φ form a set of 195 lines that is also the set of lines of types c and d of system L. We can classify the triple points of Φ accordingly to the type of lines of system L going through them. The triple points fall in three classes (Table III) that also correspond to the difference of notations of the triple points.

TABLE I. The 31 triple points. The indices k, l, m, i take the values 1, 2, ..., 6. Note that there are two notations for the 15 points like $T_{12} = T_{21}$ and 12 notations for the ten points at infinity like $T_{123} = T_{456} = T_{321} = \cdots$.

Number of points	Notation	Coordinates
6	T _k	$x_i = \delta_{ik}, \ z = 1$
15	T_{kl}	$x_i = \frac{1}{2} - \delta_{ik} - \delta_{il}, \ z = 1$
	$k \neq l$	
10	T _{kim}	$x_i = \frac{1}{2} - \delta_{ik} - \delta_{il} - \delta_{im}, \ z =$
	$k \neq l \neq m \neq k$	

The six triple points $(T_k)_{1 < k < 6}$ are independent and form one of these classes. It is then very natural to introduce new projective coordinates $(x_i)_{1 < i < 6}$ such that the triple point T_k has coordinates $x_i = \delta_{i,k}$. The transformation from projective coordinates A, B, C, S, T, U (we use coordinate U = 2z - A - B - C - S - T instead of z to make the transformation simpler) is given by

$$\begin{pmatrix} A \\ B \\ C \\ S \\ T \\ U \end{pmatrix} = \begin{pmatrix} 0 & 1 & 1 & 0 & 0 & 0 \\ 1 & 0 & 1 & 0 & 0 & 0 \\ 1 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 1 \\ 0 & 0 & 0 & 1 & 0 & 1 \\ 0 & 0 & 0 & 1 & 1 & 0 \end{pmatrix} \begin{pmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \\ x_5 \\ x_6 \end{pmatrix}.$$
 (2)

The transformation matrix in (2) has been chosen so that its column k represents the A, B, C, S, T, U coordinates of the triple point T_k with the determination corresponding to z = 1.

We recall that here, the projective coordinates A, B, C, S, T, U, or $(x_i)_{1 \le i \le 6}$ of a given point of $P^5(\mathbb{C})$ are defined apart from a multiplicative factor. Thus if we want that A, B, C, S, T, U correspond to our earlier definition of Eq. (1) we have to fix this multiplicative factor by z = 1. Since $z = \sum_{i=1}^{6} x_i$, this condition corresponds to the relation

$$\sigma_1 = \sum_{i=1}^{6} x_i = 1 \tag{3}$$

on the new coordinates.

TABLE II. System L of lines. Classification of the lines joining the triple points.

Type of lines	Number of lines	Lines	
a	30	$T_k T_{kl}$	meet Φ only at T_k and T_{kl}
b	120	$T_k T_l$	lines of simple points of Φ
c	60	$\frac{T_{kl}T_{pq}}{T_kT_{lm}T_{klm}}$	lines of double points of Φ passing through three triple points
đ	135	$T_{klm} T_{pq} (\neq T_{klm} T_{kl}) T_{klm} T_{pqr}$	lines of double points of Φ passing through two triple points

TABLE III. Classification of triple points. The three types of points, A, B, and C are characterized by different values of the numbers n_{α} , n_b , n_c , n_d of lines of types a, b, c, d of the system L through one triple point ($n_a + n_b$ $+ 2n_c + n_d = 30$).

Type of points	Number of points	Points	n _a	n _b	n _c	n _d
A	6	T _k	5	5	10	0
В	15	T_{kl}	2	14	4	6
С	10	T_{klm}	0	0	6	18

Transforming polynomial Φ_H by Eq. (2) gives, after dehomogenizing by $\sigma_1 = 1$ [Eq. (3)],

$$\Phi = \frac{1}{9}(\sigma_3 + \frac{1}{2})^2 - \frac{1}{5}(\sigma_5 + \frac{1}{4}), \tag{4}$$

where

$$\sigma_k = \sum_{i=1}^6 x_i^k.$$

In this form it is clear that the polynomial Φ is invariant under any permutations of the x_i .

This symmetric group S_6 has also been found by Beyer, Louck, and Stein⁶ in a more general setting (for 6*j* coefficients of any degree). However, with our approach we can now show in addition that there is no other projective symmetry of Φ . We intend by *projective symmetry* a projective transformation [in other words, this is a linear transformation of the $(x_i)_{1 \le i \le 6}$ coordinates] that leaves the surface Φ unchanged.

The system L of lines together with their types (a-d)and the triple points together with their classification are, of course, invariant under any projective symmetry ϑ of Φ . By considering the triple points of class A of Table III, we see that ϑ permutes the six triple points T_k ,

$$\vartheta(T_k) = T_{\sigma(k)},$$

where σ is a permutation of 1,...,6.

Then by considering the lines of types a and c we have $\vartheta(T_{kl}) = T_{\sigma(k)\sigma(l)}$ and $\vartheta(T_{klm}) = T_{\sigma(k)\sigma(l)\sigma(m)}$. Since T_1 , $T_2, ..., T_6$, and T_{12} form a basis of the projective space $P^5(\mathbb{C})$, the transformation ϑ is entirely determined by $\vartheta(T_1)$, ..., $\vartheta(T_6)$ and $\vartheta(T_{12})$, and so by the permutation σ . This shows that there is no other projective symmetry than the permutations of the x_k .

We now mention another remarkable property of the surface Φ which will be useful in Sec. IV. The 45 projective three-spaces $E_{(kl)(mn)}$ (where k, l, m, n are different integers ≤ 6), defined by

$$x_k + x_l = 0, \quad x_m + x_n = 0,$$

are on the surface Φ . Moreover, there is no other projective three-space on Φ . Each of these spaces contains ten triple points, for example, $E_{(12)(34)}$ contains the points T_5 , T_6 , T_{13} , T_{14} , T_{23} , T_{24} , T_{135} , T_{136} , T_{145} , and T_{146} .

Let us point out that 27 of these spaces can be obtained at once from Eq. (1). For example, each term of Eq. (1) is zero if we set A = S = 0 (giving $E_{(23)(56)}$) or if we set A = S - 1 = 0 (giving $E_{(23)(14)}$ since $1 - A - S = x_1 + x_4$ for dehomogenized x_i).

III. THE RATIONAL SYMMETRY Δ OF Φ

Equation (1) is invariant under two transformations R of coordinates A, B, C, S, T of the form

$$R(A,B,C,S,T) = (S,T,C+w,A,B).$$

Indeed forming $D = \Phi(A,B,C,S,T) - \Phi(S,T,C + w,A,B)$ we obtain a polynomial in w of degree 2:

$$D = a(w - w_1)(w - w_2).$$

But one root of D, w_1 , corresponds to the permutation $A \leftrightarrow S$, $B \leftrightarrow T$, $C \leftrightarrow U$, which is an obvious rational symmetry of Eq. (1). It results that the symmetry corresponding to the other root of D, w_2 , is also rational. More precisely we have

$$w_1 = 2 - A - B - S - T - 2C$$

and

$$w_{2} = (1 - A - B - S - T) \times [(A^{(2)}B^{(2)} - S^{(2)}T^{(2)})/(AB - ST)^{(2)}].$$
(5)

We shall denote by Δ the symmetry corresponding to root w_2 ,

$$\Delta(A,B,C,S,T) = (S,T,C + w_2,A,B).$$
 (6)

Some properties of Δ are as follows.

(a) Δ is an involution, $\Delta^2 = 1$.

(b) The base points of Δ , that is, the points where Δ is not defined (giving homogeneous transformed coordinates all equal to zero), are the points canceling both the numerator and denominator of Eq. (5).

The base points can be obtained by making use (for k = 2) of the first part of the following identity:

$$(AB - ST)^{(k)} = \begin{cases} (A + S)^{(k)} (B + S)^{(k)}, & \text{if } A + B + S + T = k - 1 \\ (A + S - 1)^{(k-1)} (B + S - 1)^{(k-1)} ((A + S)(B + S) - kS), & \text{if } A + B + S + T = k. \end{cases}$$

It turns out that the base points consist of the 18 threespaces of Φ given in Table IV. The neighborhood of a base point is transformed by Δ into a line (i.e., when we let A, B, C, S, T approach a base point from all directions, the corresponding transforms by Δ tend to points forming a line). The neighborhoods of points in $E_{(13)(24)}$ are lines that generate $E_{(15)(46)}$, so we write $\Delta_N(E_{(13)(24)}) = E_{(15)(46)}$. We have a similar property for the 18 three-spaces of base points: $\Delta_N(E_{(25)(46)}) = E_{(13)(25)}, \quad \Delta_N(E_{(13)(46)}) = E_{(13)(46)},...,$ which is also displayed in Table IV.

(c) We know very little about the group of rational symmetries generated by Δ and the permutations of the x_i . We conjecture that this group is infinite. Calculating orbits of $\Delta_{13} = (13) \circ \Delta$ (Δ followed by the permutation of x_1 and x_3) in $P^5(F_p)$ (the five-dimensional projective space over the field F_p of integers modulo a prime number p), we get divisors (the number of points of an orbit) of the order k of Δ_{13} if this order is finite. By calculating a few orbits we got that if k exists, it is a multiple of a very large number $(>10^{80})$.

IV. SOME CUBICS ON THE SURFACE Φ

If we put S = A and T = B in Eq. (1) we observe that Φ factorizes in three linear factors and one cubic factor,

TABLE IV. The 18 three-spaces of base points of Δ . Two three-spaces on the same line in the first and second columns are exchanged by Δ_N . The three-spaces in the third column are self-transformed by Δ_N .

$E_{(13)(24)}$	$E_{(15)(46)}$	$E_{(13)(46)}$
$E_{(13)(25)}$	$E_{(25)(46)}$	$E_{(14)(25)}$
$E_{(13)(56)}$	$E_{(23)(46)}$	$E_{(14)(36)}$
$E_{(14)(23)}$	$E_{(14)(56)}$	$E_{(15)(24)}$
$E_{(15)(23)}$	$E_{(24)(56)}$	$E_{(23)(56)}$
$E_{(15)(36)}$	$E_{(24)(36)}$	$E_{(25)(36)}$

$$\Phi = AB(A + B - 1)F_1,$$

$$F_1 = C^2 + 2(A + B - 1)C$$

$$- (A - 1)(B - 1)(2A + 2B - 1).$$
(7)

A similar factorization also occurs by setting S = A and T = B + 1,

$$\Phi = AB(A+B)F_2,$$

$$F_2 = C^2 + (2B+1)C$$

$$- (A-1)(B+1)(2A+2B-1).$$
(8)

The geometrical interpretation of these factorizations is very simple. For example, relations S = A and T = B define a projective subspace P_1 of $P^5(\mathbb{C})$ of dimension 3. Equation (7) describes the intersection of P_1 and Φ . The factorization means that this intersection consists of three planes and one cubic surface. The three planes are in fact, the intersections of P_1 with the three-spaces $E_{(23)(56)}$, $E_{(13)(46)}$, and $E_{(15)(24)}$. Similarly, Eq. (8) describes the intersection of the subspace P_2 (S = A, T = B + 1) with Φ that consists of three planes, intersections of P_2 with the three-spaces $E_{(23)(56)}$, $E_{(13)(25)}$, and $E_{(15)(36)}$, and of the cubic surface F_2 (we also denote by F_2 the surface of P_2 defined by the equation $F_2 = 0$).

In $P^{5}(\mathbb{C})$ the intersection of two projective subspaces of dimension 3 is, in general, a line. So the intersections of a generic projective subspace P of dimension 3 with the spaces $E_{(kl)(mn)}$ consist of lines. It is only for particular positions of P that some of the intersections of P with $E_{(kl)(mn)}$ become planes giving a factorization of linear factors in the equation of the intersection. It happens that there is an interesting family of such three-spaces:

$$P_a \begin{cases} x_1 + x_2 + x_3 + x_4 - a(x_1 + x_2) = 0, \\ x_2 + x_4 + a(x_3 + x_5) = 0. \end{cases}$$
(9)

The space P_a intersects each of the spaces $E_{(12)(34)}$, $E_{(13)(25)}$, and $E_{(24)(36)}$ along a plane. We can take any value

for a, excepted a = 0 for which P_a coincides with $E_{(13)(24)}$, and a = 1 for which P_a coincides with $E_{(34)(25)}$. Algebraically we use the projective coordinates u, v, w, and z of P_a :

$$x_{1} = u - v,$$

$$x_{2} = v,$$

$$x_{3} = (a - 1)u + v - w,$$

$$x_{4} = -v + w,$$

$$x_{5} = (1 - a)u - v + (1 - 1/a)w,$$

$$x_{6} = -u + v - (1 - 1/a)w + z.$$

Substituting into Φ , we obtain a factorization of the form

 $\Phi = uv(au - v)F_a,$

where F_a describes a cubic surface. Rather than writing the complicated expression of F_a in terms of u, v, w, and z we use the classical theory as exposed in Chap. VII of Semple and Roth⁵ to obtain a simpler form. The cubic F_a has one double point only when $a = -1, \frac{1}{2}$, or 2. For these values of a, there is a permutation of the x_i coordinates such that the space P_a and the cubic F_a coincide with the space P_2 and the cubic F_2 of Eq. (8), respectively. For other values of a, F_a has no double point and possesses a system of 27 lines that are rational in a. There are 45 tritangent planes that are also rational in a and cut the cubic along three lines. The method for simplifying the equation of the cubic consists in finding a pair of associated Steiner trihedrals. This is two sets (X_1, X_2, X_3) , (Y_1, Y_2, Y_3) of three tritangent planes (of equations $X_1 = 0, X_2 = 0, ...$) such that the mutual intersections as shown in the array

$$\begin{array}{cccc} X_1 & X_2 & X_3 \\ Y_1 & l_1 & l_2 & l_3 \\ Y_2 & l_4 & l_5 & l_6 \\ Y_3 & l_7 & l_8 & l_9 \end{array}$$

are nine different lines $l_1,...,l_9$ of the cubic F_a . There are 120 such pairs of Steiner trihedrals, but it is enough to find one of these. It is then possible to choose $X_1, X_2, X_3, Y_1, Y_2, Y_3$ such that the equation of the cubic takes the form $X_1X_2X_3$ $= Y_1Y_2Y_3$ and such that the following relation is verified: X_1 $+ X_2 + X_3 = Y_1 + Y_2 + Y_3$. Then by setting $X_1 = y_2 + y_3$, $X_2 = y_1 + y_3, X_3 = y_1 + y_2, Y_1 = -y_5 - y_6, Y_2 = -y_4$ $- y_6, Y_3 = -y_4 - y_5$ the equation of the cubic can be written as

$$\det \begin{vmatrix} 0 & y_2 + y_3 & y_5 + y_6 \\ y_4 + y_6 & 0 & y_1 + y_3 \\ y_1 + y_2 & y_4 + y_5 & 0 \end{vmatrix} = 0
 (10)$$

or as

$$\sum_{i=1}^{6} y_i^3 = 0.$$
 (11)

Carrying out these calculations, we obtained that a possible choice for the y_i is given by

$$y_1 = -(a-1)a^2u + (a-1)aw + (a^3 - a^2/2 - 3a/2 + 1/2)z,$$

$$y_{2} = -(a-1)^{2}au + (a-1)^{2}w + (a^{3} - 5a^{2}/2 + a/2 + 1/2)z, y_{3} = (a-1)a^{2}u - (a-1)^{2}w - (a^{3} - 3a^{2}/2 + 3a/2 - 1/2)z, y_{4} = -(a-1)^{2}au - 2(a-1)av + (a-1)aw + (a^{2} + a - 1)z/2, y_{5} = (a-1)^{2}au + 2(a-1)av - (a-1)aw - (a^{2} - 3a + 1)z/2, y_{6} = (a-1)^{2}au - (a-1)aw - (a^{3} - 3a^{2}/2 - a/2 + 1/2)z.$$
(12)

These variables are linked by two linear relations:

$$\sum_{i=1}^{6} y_i = 0,$$
(13)

$$2(a-1)y_1 - 2ay_2 - (a^2 - a + 1)(y_4 + y_5) = 0.$$

The sum of cubes of y_i is then related to the cubic factor F_a by

$$\sum_{i=1}^{6} y_i^3 = 3(2a-1)(1-1/a)^3 F_a,$$

so that for $a \neq \frac{1}{2}$ the equation of the cubic F_a is given by Eq. (10) or (11).

The cubic F_a can be parametrized rationally by homogeneous variables b, c, d by solving in terms of u, v, w the set of linear equations

$$(y_2 + y_3)c + (y_5 + y_6)d = 0,$$

$$(y_4 + y_6)b + (y_1 + y_3)d = 0,$$

$$(y_1 + y_2)b + (y_4 + y_5)c = 0.$$

(14)

Indeed the equation of the cubic [Eq. (10)] is the result of eliminating b, c, and d from Eq. (14), and as shown in Semple and Roth⁵ the correspondence between the projective plane b, c, d and the cubic F_a is birational (i.e., bijective and rational).

Dehomogenizing by c = 1, for example, and z = 1, we obtain six rational functions $R_i(a,b,c)$ that give the parametrization $x_i = R_i(a,b,d)$ of the cubic F_a for fixed a. We omit giving the complicated expressions of the R_i . We can interpret this result by considering the surface Q (of dimension 4) generated by the spaces P_a by varying a. The equation of Q is obtained by eliminating a from Eqs. (9),

$$(x_1 + x_2 + x_3 + x_4)(x_3 + x_5) + (x_1 + x_2)(x_2 + x_4) = 0,$$

so that Q is a quadric. The functions R_i are thus giving a rational parametrization of the intersection of Q and Φ in terms of the three independent parameters a, b, and d.

The cubic F_1 can be studied by proceeding as in the case of cubic F_a . We introduce in this case the following y_i :

$$y_{1} = -\frac{3}{4}, \quad y_{2} = -A - B - C + \frac{5}{4}$$
$$y_{3} = A + B + C - \frac{3}{4}, \quad y_{4} = B - \frac{1}{4},$$
$$y_{5} = A - \frac{1}{4}, \quad y_{6} = -A - B + \frac{3}{4},$$

which are linked by

$$\sum_{i=1}^{6} y_i = 0 \tag{15}$$

and

$$2y_1 + 3(y_2 + y_3) = 0.$$

The expression F_1 , Eq. (7) then takes the form

$$F_1 = \frac{2}{3} \sum_{i=1}^{6} y_i^3.$$
 (16)

The cubic F_1 then can be parametrized by solving Eq. (14). This gives, setting d = 1 and $c = b^2 - a$, a full rational parametrization of F_1 as

$$A = (b^{4} + b^{2} - 2b)/2a + a/2 - b^{2},$$

$$B = (b^{2} - a + 1)/2,$$

$$C = (b - 1)A + 1 - (b^{2} + b - a)/2.$$
(17)

In the case of the cubic F_2 we define similarly

$$y_{1} = \frac{1}{4}, \quad y_{2} = -B - C - \frac{1}{4},$$

$$y_{3} = B + C + \frac{3}{4}, \quad y_{4} = B + \frac{1}{4},$$

$$y_{5} = A - \frac{5}{4}, \quad y_{6} = -A - B + \frac{1}{4},$$

which are linked by

and

$$\sum_{i=1}^{\infty} y_i = 0$$

 $3y_1 + y_4 + y_5 + y_6 = 0.$

The expression F_2 , Eq. (8), then takes the form

$$F_2 = \frac{2}{3} \sum_{i=1}^{6} y_i^3.$$
 (19)

For the cubic F_2 there is a very simple way of obtaining a rational parametrization. We have already mentioned that F_2 has one double point Ω . It is $(A = 1, B = -\frac{1}{2}, C = 0)$. The line (for fixed a and b the line is parametrized by t),

A = a(2t-1) + 1, B = t - 1, C = b(2t - 1), (20)intersects F_2 twice at Ω (for $t = \frac{1}{2}$) and at another point given by

$$t = b(b+1)/a(2a+1),$$
(21)

as can be verified by substituting Eq. (20) into F_2 . Equations (20) and (21) give a rational parametrization of F_2 in terms of the two parameters a and b. Let us note that this method cannot be applied to F_1 or F_a , which have no double points.

It is clear that Eqs. (18) and (19) describing the cubic F_2 are invariant in the permutations of the triplet (y_4, y_5, y_6) and the pair (y_2, y_3) . The corresponding symmetry expressed in terms of A, B, and C is described in Table V. In fact, the 12 transformations are the only projective transformations of P_2 that leave F_2 invariant. This can be shown by considering the system of Steiner trihedrals and the double

TABLE V. Symmetries of $F_2(A,B,C)$. The cubic polynomial F_2 takes the same value for the 12 sets of A, B, C.

A	B	С
A	В	-2B-C-1
А	-A-B	A - C - 1
A	-A-B	A+2B+C
$B + \frac{3}{2}$	$A-\frac{3}{2}$	$-A-B-C+\frac{1}{2}$
$B + \frac{3}{2}$	$A-\frac{3}{2}$	$-A+B+C+\frac{3}{2}$
$B + \frac{3}{2}$	-A-B	A-C-1
$B + \frac{3}{2}$	-A - B	A+2B+C
$-A-B+\frac{3}{2}$	$A-\frac{3}{2}$	$-A-B-C+\frac{1}{2}$
$-A-B+\frac{3}{2}$	$A - \frac{3}{2}$	$-A+B+C+\frac{3}{2}$
$-A-B+\frac{3}{2}$	B	-2B-C-1
$-A-B+\frac{3}{2}$	В	С

point of F_2 that have to be invariant under any projective symmetry.

Similarly, Eqs. (15) and (16) for the cubic F_1 are also invariant under the same 12 permutations of the y_i as in the case of F_2 . However unlike the case of F_2 , the corresponding transformations are simply permutations of the x_i coordinates.

V. CONCLUDING REMARKS

(18)

(1) It is still an open question whether there exists or not a full rational parametrization of the surface Φ (in terms of four independent parameters). The parametrizations that we have found (in terms of three or two independent parameters) really represent a very small part of the surface.

(2) Once a rational parametrization is found, one would like to know how to find all the values of the parameters that give genuine zeros of 6j, that is, for which the corresponding A, B, C, S, T are integers ≥ 2 . This is, however, a difficult problem in itself, even for a simple parametrization like Eqs. (20) and (21) for the cubic F_2 . Indeed, if it is quite easy to generate integer A, B, C [for example setting b = a(2a + 1), and taking any integer for parameter a, in Eqs. (20) and (21)], it seems difficult to characterize the parameters a and b that give integers A, B, C without calculating A, B, and C. Note that a and b are not necessarily integers (for example, $a = \frac{1}{3}, b = \frac{5}{3}$ give A = 6, B = 7, C = 25).

- ²S. Brudno and J. D. Louck, J. Math. Phys. 26, 1125 (1985).
- ³A. Bremner and S. Brudno, J. Math. Phys. 27, 2613 (1986).
- ⁴J. J. Labarthe, J. Math. Phys. 27, 2964 (1986).
- ⁵J. G. Semple and L. Roth, Introduction to Algebraic Geometry (Oxford U. P., New York, 1986).
- ⁶W. A. Beyer, J. D. Louck, and P. R. Stein, J. Math. Phys. 28, 497 (1987).
- ⁷A. Bremner, J. Math. Phys. 27, 1181 (1986).

¹S. Brudno, J. Math. Phys. 28, 124 (1987).

On discrete Schrödinger equations and their two-component wave equation equivalents

Alfred M. Bruckstein^{a)} and Thomas Kailath Information Systems Laboratory, Stanford University, Stanford, California 94305

(Received 23 October 1986; accepted for publication 24 June 1987)

An approach to inverse scattering problems for discrete Schrödinger equations, which are discrete three-term recursions, is presented by systematically transforming them into discrete two-component wave-propagation equations. The wave-propagation equations permit the immediate application of certain computationally efficient and physically insightful "layer-peeling" algorithms for inverse scattering. The mapping of three-term recursions to two-component evolution equations is one to many, because the relation between the "potential" sequence parametrizing Schrödinger equations and the "reflection coefficient" sequence determining local wave interaction is a nonlinear difference equation. This mapping is examined in some detail and it is used to study both direct and inverse scattering problems associated with discrete Schrödinger equations.

I. INTRODUCTION

A highly nonstandard approach to the inverse scattering problem of quantum mechanics is due to Krein (see, e.g., Chadan and Sabatier,¹ p. 123). It is based on a transformation of the Schrödinger equation, parametrized by a potential function P(x), into a two-component system of firstorder differential equations parametrized by a local reflectivity function K(x).

The point of this transformation is that the corresponding inverse scattering problem for the two-component system leads to an integral equation (the Krein equation) different from the classical equations due to Gel'fand and Levitan and to Marchenko. One can formally show in a rather straightforward way that, to a given one-dimensional second-order nonhomogeneous wave equation

$$\left(\frac{\partial}{\partial x^2} - \frac{\partial^2}{\partial t^2}\right) V(x,t) - P(x) V(x,t) = 0$$
(1.1)

(which by Fourier transforming w.r.t. the time variable becomes a Schrödinger equation), we can associate a two-component first-order differential equation [relating right and left propagating waves $\{W_R(x,t), W_L(x,t)\}\]$ of the following form:

$$\frac{\partial}{\partial x} \begin{bmatrix} W_R(x,t) \\ W_L(x,t) \end{bmatrix} = \begin{bmatrix} -\partial/\partial t & -K(x) \\ -K(x) & \partial/\partial t \end{bmatrix} \begin{bmatrix} W_R(x,t) \\ W_L(x,t) \end{bmatrix}.$$
(1.2)

Indeed, if the reflectivity function K(x) is related to the potential P(x) via the Riccati equation

$$-\frac{d}{dx}K(x) + K^{2}(x) = P(x)$$
(1.3)

and $V(x_0,t) = W_R(x_0,t) + W_L(x_0,t)$ at some point x_0 , it follows that

$$V(x,t) = W_{R}(x,t) + W_{L}(x,t)$$
(1.4)

holds everywhere (see Chadan and Sabatier,¹ Bruckstein

and Kailath,² and Bruckstein, Levy, and Kailath³). Note that if one starts with wave-propagation equations as in (1.2), then it is immediate to determine the potential function of the corresponding second-order (Schrödinger) equation; however, the reverse mapping is more problematic. From (1.3) we expect it to be nonunique, since we also need an initial condition, say $K(x_0)$, in order to determine K(x) given the potential P(x).

Several interesting questions now arise in the context of Krein's approach to inverse scattering: For which potentials can we find reflectivity functions obeying (1.3)? Are there potentials for which the reflectivity function is uniquely determined from the corresponding Riccati equation? Are there potentials for which no reflectivity function obeying (1.3) exists? The answer obviously depends on the interval over which we wish to determine a model as well as on the properties of the potential P(x). If the potential is identically 0 for $x \in [0, \infty)$ then reflectivity functions obeying (1.3) are of the form K(0)/(1 - K(0)x), and, to be well defined on the positive axis, K(0) can be any negative value. However, had we required a solution over the entire real axis, the conclusion would have been that K(x) = 0 is the only one that does not blow up at any point. This simple example shows that the mapping from potentials to reflectivity functions may sometimes be problematic.

We shall analyze the problems raised above for a discrete version of the wave (or Schrödinger) equation. Doing a straightforward discretization of Eq. (1.1), with both spatial and temporal quantization intervals chosen to have length Δ , one arrives at the equation

$$V(x + \Delta, t) + V(x - \Delta, t) = G(x) \{ D + D^{-1} \} V(x, t) + O(\Delta^4) , \qquad (1.5)$$

where G(x) is defined as

$$G(x) = \exp\{\Delta^2[P(x)/2]\}$$
(1.6)

and D acts on time sequences as a Δ /unit delay operator, i.e.,

$$Df(t) = f(t - \Delta)$$
 and $D^{-1}f(t) = f(t + \Delta)$. (1.7)

For details of the discretization see, e.g., Case and Kac.⁴ If, for integer values of n and τ , $V(n\Delta, \tau\Delta)$ is rewritten as

^{*}Presently with the Faculty of Electrical Engineering, Technion, IIT, 32000, Haifa, Israel.
$V(n,\tau)$ and all functions of the (discrete) time index are replaced by formal two-sided power series in D, i.e.,

$$f(\tau) \rightarrow f(D) = \sum_{\tau \in \mathbb{Z}} f(\tau) D^{\tau},$$

Eq. (1.5) becomes a discrete wave equation

$$V(n+1,D) + V(n-1,D) = G(n)\{D+D^{-1}\}V(n,D),$$
(1.8)

which has the form of a "classical" discrete Schrödinger equation with eigenvalue $2\lambda = D + D^{-1}$ (see, e.g., Case and Kac⁴ and Case⁵). Note that by (1.6) the potential sequence G(n) is always positive. Since in the continuous case it is assumed that P(x) tends to 0 as x goes to infinity (in each direction), we have that G(n) tends to 1 as $|n| \to \infty$. It is usually further assumed that convergence to these limits is quite fast, therefore our discussion will often refer to the case of G(n) = 1 for |n| exceeding a certain value N.

We shall also consider here a generalized version of (1.8) parametrized by two sequences of numbers $\{F(n), G(n)\},\$

$$F(n)V(n+1,D) + F(n-1)V(n-1,D) = G(n)\{D+D^{-1}\}V(n,D), \qquad (1.9)$$

where F(n) is never zero and tends to 1 as $|n| \to \infty$.

Our interest in discrete Schrödinger equations arose from recent research on efficient algorithms for solving nested (linear) systems of equations, or integral equations, recursively for increasing dimensions. Such algorithms provide lattice filter solutions for linear prediction/estimation of stochastic processes and are of crucial importance in digital signal processing (see, e.g., the discussion in Kailath⁶ and the references therein). It gradually became clear that there are close connections between inverse scattering problems for transmission-line models and estimation theory, and in both fields the problems are solved once one has an algorithm for recursively determining a layered medium, or a cascade filter, from impulse responses or equivalent spectral data. Moreover, in both fields the solutions may be found either by first deriving and solving sets of linear equations (or integral equations, in the continuous case) or in a direct way, by exploiting the causality of signal propagation and structure of the layered medium, i.e., the assumed structure of the elementary processors in a cascade filter. This research also showed that the more efficient way of finding the solution of inverse scattering problems is via a recursive layer identification and peeling process, in which signals are propagated through the already identified layers to provide the scattering data for the deeper and yet unidentified layers of the medium (see, e.g., Bruckstein, Levy, and Kailath³ and Bruckstein and Kailath^{2,7}). It was thus natural to look for similar algorithmic solutions to the classical inverse scattering problem, which starts with the Schrödinger equation as the propagation model. This paper shows that, in the discrete case previously analyzed by Case and Kac,⁴ and Case^{5,8} and several others, direct layer-peeling solutions are always possible via a mapping of the Schrödinger equation to a twocomponent wave-propagation model, and hence to an equivalent discrete transmission line. The mapping then provides a new algorithmic inverse scattering procedure that does not proceed via classical systems of equations (see, e.g., Case⁵) corresponding to the solutions of Gel'fand-Levitan, Marchenko, or Krein (see, e.g., Chadan and Sabatier¹).

This paper is organized as follows. The next section deals with the formal mapping of the discrete Schrödinger equation into a two-term wave evolution equation, and proves that such a mapping is always possible. In Sec. III we then discuss some properties of the mapping between potentials and the reflection coefficients that parametrize twoterm evolution equations. In Sec. IV we deal with direct and inverse scattering problems and derive layer-peeling algorithms for inverse scattering based on inverse scattering for transmission-line models; an example from Case⁵ is then reexamined from this point of view.

II. FROM THREE-TERM TO TWO-COMPONENT EQUATIONS

Equations (1.8) and (1.9) relate discrete functions of *time*, in their formal power series representations, at three consecutive points in *space*. They may be regarded as recursive ways of defining V(n,D), given the "interaction" parameters $\{F(\cdot),G(\cdot)\}$ and some initial conditions V(0,D) and V(1,D), say.

Suppose that, over a certain interval in the space coordinate $n \in [N_L, N_R]$, we have F(n) = 1 and G(n) = 1. It is then straightforward to verify that, over $[N_L, N_R]$, the solution of the three-term recursion is of the form

$$V(n,D) = D^{n}\phi_{R}(D) + D^{-n}\phi_{L}(D). \qquad (2.1)$$

To determine the functions $\phi_{R,L}(D)$ we of course need the signals V(n,D) at two points in space. In the time domain, (2.1) means that

$$V(n,\tau) = \phi_R(\tau - n) + \phi_L(\tau + n) , \qquad (2.2)$$

i.e., $V(n,\tau)$ is a sum of two *noninteracting* right and left propagating waves over $n \in [N_L, N_R]$. Regarding $V(n,\tau)$ as a sum of waves propagating in opposite directions can give simple physical interpretations to relations induced between time signals at various points in space, as a consequence of linear and *causal* wave interaction. One of the major aims of this paper is to show that a decomposition of signals $V(n,\tau)$ into *interacting* waves propagating in opposite directions can be done in general: The potential sequences $\{F(n), G(n)\}$ determine the local interactions between the right and left propagating wave components of $V(n,\tau)$ via a local reflection coefficient sequence $\{K_n\}$, corresponding to the continuous reflectivity function.

To do this, we start by writing, for a general Schrödinger equation, that V(n,D) is the sum of two components as follows:

$$V(n,D) = [1 \ 1] \begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix}.$$
 (2.3)

Now, since V(n,D) obeys (1.8) or (1.9), we have to determine a sequence of operators, $\{\Theta(n,D)\}$, that will yield wave components at n + 1 from the ones at n, and also ensure that (2.3) will hold at all points in space. This sequence of 2×2 matrix operators will, of course, be determined by

the potential sequences, i.e., G(n) and F(n), that parametrize the three-term evolution equation. Since we wish to consider both forward and backward evolution operators we require the $\Theta(n,D)$ to be invertible for every n. Writing

$$\begin{bmatrix} W_R(n+1,D) \\ W_L(n+1,D) \end{bmatrix} = \Theta(n,D) \begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix}, \quad (2.4)$$

some algebra shows that we can determine V(n + 1,D) from V(n,D) and V(n - 1,D) as follows. Here V(n + 1,D) is given by

$$V(n+1,D) = [1 \ 1]\Theta(n,D) \begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix}$$
(2.5)

and using (2.4) and (2.5) we also have

$$\begin{bmatrix} V(n,D) \\ V(n-1,D) \end{bmatrix} = \begin{bmatrix} 1 & 1 \\ [1 & 1]\Theta^{-1}(n-1,D) \end{bmatrix} \begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix}.$$
(2.6)

 $[\theta_{11}(n,D) + \theta_{21}(n,D) \quad \theta_{12}(n,D) + \theta_{22}(n,D)]$

$$V(n+1,D) = \begin{bmatrix} 1 & 1 \end{bmatrix} \Theta(n,D) \begin{bmatrix} 1 & 1 \\ [1 & 1] \Theta^{-1}(n-1,D) \end{bmatrix}^{-1} \\ \times \begin{bmatrix} V(n,D) \\ V(n-1,D) \end{bmatrix}, \qquad (2.7)$$

a three-term recursion for V(n,D) that should be made identical to (1.9). For this the sequence of matrices $\Theta(n,D)$ must obey

$$[1 \ 1][\Theta(n,D)] = \left[\frac{G(n)}{F(n)} \{D + D^{-1}\} - \frac{F(n-1)}{F(n)}\right] \\ \times \left[\begin{array}{c} 1 \ 1 \\ [1 \ 1]\Theta^{-1}(n-1,D) \end{array} \right].$$
(2.8)

If we set

$$\Theta(n,D) = \begin{bmatrix} \theta_{11}(n,D) & \theta_{12}(n,D) \\ \theta_{21}(n,D) & \theta_{22}(n,D) \end{bmatrix},$$
(2.9)

then (2.8) reads

$$= \left[\frac{G(n)}{F(n)} \{D + D^{-1}\} - \frac{F(n-1)}{F(n)}\right] \left[\frac{\theta_{22}(n-1,D) - \theta_{21}(n-1,D)}{\det \Theta((n-1,D)} \frac{\theta_{11}(n-1,D) - \theta_{12}(n-1,D)}{\det \Theta(n-1,D)}\right].$$
 (2.10)

There are many ways of choosing operator sequences that will make (2.10) into an identity. We shall be interested, for reasons to become clear soon, in nonsingular matrices having the following structure:

$$\Theta(n,D) = \Theta_n \begin{bmatrix} D & 0 \\ 0 & D^{-1} \end{bmatrix}$$

= $\gamma_n \begin{bmatrix} 1 & -K_n \\ -K_n & 1 \end{bmatrix} \begin{bmatrix} D & 0 \\ 0 & D^{-1} \end{bmatrix},$ (2.11)

where we require $|K_n| \neq 1$, for all n. Thus it will be assumed that the propagation operators are composed of a relative shift operator, delaying $W_R(n,D)$ one unit of time and advancing $W_L(n,D)$ by the same amount, followed by a symmetric pure-gain matrix mixing the two sequences, $DW_R(n,D)$ and $D^{-1}W_L(n,D)$, pointwise in time. We assume this form for the operators $\Theta(n,D)$ because their socalled scattering domain representation $\Sigma(n,D)$, relating $W_R(n+1,D)$ and $W_L(n,D)$ to $W_R(n,D)$ and $W_{I}(n+1,D)$, is then the cascade connection of a causal scattering matrix that delays the waves propagating in opposite directions and a wave-interaction matrix that transmits part of the signal and reflects part of it adding the reflected part to the wave propagating in the opposite direction (see Fig. 1). Indeed, writing

$$\begin{bmatrix} W_R(n+1,D) \\ W_L(n,D) \end{bmatrix} = \Sigma(n,D) \begin{bmatrix} W_R(n,D) \\ W_L(n+1,D) \end{bmatrix}$$
(2.12)

a few steps of algebra show [using (2.11)] that

$$\Sigma(n,D) = \begin{bmatrix} 1 & 0 \\ 0 & D \end{bmatrix} \times \begin{bmatrix} (\gamma_n^2 - K_n^2) 1/\gamma_n & -K_n \\ K_n & 1/\gamma_n \end{bmatrix} \begin{bmatrix} D & 0 \\ 0 & 1 \end{bmatrix}.$$
(2.13)

(a)





FIG. 1. (a) Transmission and (b) scattering representations of wave-propagation operators.

The above assumed structure for the signal transfer operators, and their scattering equivalents, leads to transmissionline models that are discrete-case analogs of (1.2), see, e.g., Bruckstein and Kailath.⁷

Using (2.11) for the $\Theta(n,D)$ operators (2.10) becomes

$$\begin{bmatrix} \gamma_n (1 - K_n) D & \gamma_n (1 - K_n) D^{-1} \end{bmatrix} = \begin{bmatrix} \frac{G(n)}{F(n)} \{D + D^{-1}\} - \frac{F(n-1)}{F(n)} \end{bmatrix} \begin{bmatrix} 1 & 1 \\ \frac{D^{-1} \gamma_{n-1}^{-1} + D \gamma_{n-1}^{-1} K_{n-1}}{1 - K_{n-1}^2} & \frac{D^{-1} \gamma_{n-1}^{-1} K_{n-1} + D \gamma_{n-1}^{-1}}{1 - K_{n-1}^2} \end{bmatrix}$$
(2.14)

and the equation becomes an identity provided

$$\gamma_n = F(n)/G(n+1)(1-K_n^2), \qquad (2.15a)$$

(1+K_n)(1-K_{n-1}) = F²(n)/G(n)G(n+1).
(2.15b)

Here we have used the fact that delay and advance operators commute with scalar gains and have equated the coefficients of D and D^{-1} in (2.14). Thus to have that $V(n,D) = W_R(n,D) + W_L(n,D)$, the wave components should evolve according to (2.4) with the parameters of the symmetric delay-and-interaction operators $\Theta(n,D)$ being determined by (2.15).

To summarize, suppose that we are dealing with a threeterm recursion (1.9) defined over $n \in (-\infty, \infty)$. Given the potential sequences $\{F(n), G(n)\}$, we have the following result: If a sequence K_n can be determined so that it obeys (2.15b) and is, for all *n*, different in absolute value from 1, then we can associate to the three term recursion (1.9) the following transmission-line type wave propagation model

$$\begin{bmatrix} W_{R}(n+1,D) \\ W_{L}(n+1,D) \end{bmatrix} = \frac{F(n)}{G(n+1)(1-K_{n}^{2})} \begin{bmatrix} 1 & -K_{n} \\ -K_{n} & 1 \end{bmatrix} \times \begin{bmatrix} D & 0 \\ 0 & D^{-1} \end{bmatrix} \begin{bmatrix} W_{R}(n,D) \\ W_{L}(n,D) \end{bmatrix}.$$
 (2.16)

We now show that we can always determine infinite sequences K_n , so that $|K_n| \neq 1$ for all *n*, and also satisfying (2.15b). Let us further analyze the relation (2.15b). Define the strictly positive sequence of numbers

$$o_n = F^2(n)/G(n)G(n+1) . \qquad (2.17)$$

It is clear that the nonlinear difference equation (2.15b) can be read either as a right, or forward propagating recursion, yielding K_n from K_{n-1} ,

$$K_n = \rho_n / (1 + K_{n-1}) - 1$$
, (2.18a)

or as a left, or backward propagation equation, yielding K_n from K_{n-1} ,

$$K_{n-1} = 1 - \rho_n / (1 + K_n)$$
 (2.18b)

To determine a $\{K_n\}$ sequence, we thus need to choose an "anchoring" value at some point, say n = 0, i.e., to set an initial condition to the forward and backward recursions (2.18). Clearly we need to have K_n different from 1 in abso-

lute value, since ρ_n in (2.15b) is strictly positive. The following lemma shows that we can *always* choose an initial condition that determines a $\{K_n\}$ sequence that has $|K_n| \neq 1$ for all n.

Lemma: Given an arbitrary sequence of numbers $\{\rho_n\}$, $\rho_n > 0$ for all $n \in \mathbb{Z}$, there exist uncountably many sequences $\{K_n\}$, such that, for all n,

$$(1+K_n)(1-K_{n-1})=\rho_n$$

Proof: We shall show that there are uncountably many values for K_0 for which both forward and backward recursions yield sequences K_n having $|K_n| \neq 1$. The forward recursions will never yield a -1 since ρ_n is always positive, however, they cannot be propagated past a point where the K-sequence hits a value of 1. The backward recursions never hit the value 1, however, they may at some point yield a - 1, beyond which we cannot proceed. Assume that at all points n > 0 we start backward recursions, i.e., (2.18b), with the initial value of 1, and at all points n < 0 we start forward recursions with $K_n = -1$. We proceed, if it is possible, with these recursions until n reaches 0 and exclude the values that are attained at this point from the set of possible anchoring values for K_0 . But, since any real value can be an anchor point for K_0 and the above described process rules out at most countably many of them, it will be possible to find uncountably many initial values K_0 that will yield solutions of the nonlinear difference equation (2.15b) on the entire line. Hence there always exist infinitely many different $\{K_n\}$ sequences that correspond to any given ρ_n sequence. Q.E.D.

We have proved that to any three-term recursion of the form (1.9) there correspond many wave propagation models of the form (2.16). We call these models wave-propagation equations because, as pointed out above, (2.16) implies that the signals $W_R(n,D)$ and $W_L(n,D)$ may indeed be regarded as waves propagating in opposite directions in a layered scattering medium that determines their causal interaction. The wave-interaction $\Sigma(n,D)$ show that the medium layers operate causally on the incoming waves $W_R(n,D)$ and $W_L(n+1,D)$ to generate the outgoing waves $W_R(n+1,D)$ and $W_L(n,D)$. As we see from (2.13) the pointwise interaction matrix of the layers' scattering representations have as right and left reflection gains K_n and $-K_n$, respectively. This is the reason for calling the K_n 's local reflection coefficients.

The causal picture of wave propagation proves to be a most intuitive physical interpretation of signal interactions described by the discrete Schrödinger equation: Given a potential sequence $\{G(n)\}$, or generally, a pair of sequences $\{F(n),G(n)\}$ we shall investigate the properties of the discrete Schrödinger equation via corresponding reflection coefficient sequences $\{K_n\}$.

III. POTENTIALS AND LOCAL REFLECTION COEFFICIENTS

It is usually assumed that the sequences $\{F(n), G(n)\}$ tend to 1 as $|n| \to \infty$. In this case the corresponding ρ_n sequences defined by (2.17) also converge to 1 with increasing |n|. We have from the forwards and backwards recursions (2.18) that, if K_n tends to a limit, then

$$\lim_{|n|\to\infty}K_n=0.$$
(3.1)

However, the convergence of K_n to 0 is conditioned on the asymptotic behavior of the sequence ρ_n . From (2.15) we see that, in order to have a convergent K_n sequence for $|n| \to \infty$, it is necessary that $\rho_n - 1$ approaches 0 as fast as $K_n - K_{n-1} - K_n K_{n-1}$ ($= \rho_n - 1$), for some K_n sequence convergent to 0. It is not difficult to show that a sufficient condition for having a convergent K_n sequence is that $\delta_n = 1 - \rho_n$ approaches 0 from above, i.e., so that $\rho_n < 1$ for all *n* big enough in their absolute value. This means, for example, that the potential in Eq. (1.8), G(n), should obey G(n)G(n+1) > 1 for all *n*, a condition trivially met if G(n) is always greater than 1.

To avoid dealing with convergence conditions we often assume that the potentials are already at their limiting value, 1, for $|n| > N_0$. In this case the iteration of the functions

$$\mathbf{F}_{f}(x) = 1/(1-x) - 1$$
 and $\mathbf{F}_{b}(x) = 1 - 1/(1-x)$
(3.2)

yields K_n sequences that converge to 0 as 1/|n|. To see this we can either check directly that sequences converging to 0 as 1/|n| satisfy the recursions induced by (3.2) or transform (2.18) into two-component recursions for A_n and B_n , where

$$K_n = A_n / B_n . \tag{3.3}$$

It is straightforward to show that

$$\begin{bmatrix} A_n \\ B_n \end{bmatrix} = \begin{bmatrix} 1 & \rho_n - 1 \\ -1 & 1 \end{bmatrix} \begin{bmatrix} A_{n-1} \\ B_{n-1} \end{bmatrix}$$
(3.4)

implying that, for n > 0, A_n and B_n are given by

$$\begin{bmatrix} A_n \\ B_n \end{bmatrix} = \prod_{i=n}^{0} \begin{bmatrix} 1 & \rho_i - 1 \\ -1 & 1 \end{bmatrix} \begin{bmatrix} K_0 \\ 1 \end{bmatrix}.$$
 (3.5)

Now if $\rho_n = 1$ for $n > N_0$ we obtain

$$K_n = A_{N_0} / - (n - N_0) A_{N_0} + B_{N_0} \to 0.$$
 (3.6)

and a similar exercise shows that for $n \to -\infty$, K_n also tends to zero as $1/(|n - N_0|)$.

Note that if we had F(n) = G(n) = 1 for all $n \in \mathbb{Z}$, the sequences K_n that would correspond to an anchoring value of K_0 would be

$$K_n = K_0 / (1 - nK_0) , \qquad (3.7)$$

the excluded values for K_0 being $\{1/n | n \in \mathbb{Z}\}$.

A. Conditions for losslessness

Suppose that we can find a sequence K_n corresponding to the given potential sequence(s) so that

$$|K_n| < 1 \quad \text{for all } n \,. \tag{3.8}$$

The evolution equations (2.16) into which the three-term recursion (1.9) is mapped then yield sequences proportional to those provided by the following associated wave equation:

$$\begin{bmatrix} W_{R}^{*}(n+1,D) \\ W_{L}^{*}(n+1,D) \end{bmatrix}$$

$$= \frac{1}{(1-K_{n}^{2})^{1/2}} \begin{bmatrix} 1 & -K_{n} \\ -K_{n} & 1 \end{bmatrix}$$

$$\times \begin{bmatrix} D & 0 \\ 0 & D^{-1} \end{bmatrix} \begin{bmatrix} W_{R}^{*}(n,D) \\ W_{L}^{*}(n,D) \end{bmatrix}.$$
(3.9)

These equations describe wave propagation on a lossless transmission line, as discussed in Bruckstein and Kailath.⁷ The interaction (gain) matrices are J lossless in the transmission representation, i.e.,

$$\Theta_n^* J \Theta_n^{*T} = J \quad \text{with} \ J = \begin{bmatrix} 1 & 0 \\ 0 & -1 \end{bmatrix}$$
(3.10)

and this implies that they are unitary (lossless) in their scattering representation.

Let us determine conditions under which a given threeterm recursion may be mapped into a lossless model, with all reflection coefficients less than 1 in absolute value. From (2.19) we see that necessary conditions for this are

$$0 < \rho_n < 4 = \max\{(1 + K_n)(1 - K_{n-1})\}. \quad (3.11)$$

To obtain sufficient conditions, too, let us study the forward and backward recursions providing the $\{K_n\}$ sequence from the anchoring point K_0 , say. A simple analysis of functions of the form

$$\mathbf{F}_{f}(x;\rho) = \rho/(1-x) - 1$$
 and $\mathbf{F}_{b}(x;\rho) = 1 - \rho/(1+x)$
(3.12)

shows that, if $\rho < 1$, then $\mathbf{F}_{f}(x;\rho)$ maps the interval (-1,0) into itself, whereas $\mathbf{F}_{b}(x;\rho)$ maps (0,1) into itself. If we therefore ensure that K_{1} is in (-1,0), then $K_{n} \in (-1,0)$ for all positive *n*, and similarly if $K_{-1} \in (0,1)$ then K_{n} will remain in that interval for all n < 0. Thus, choosing $K_{0} = 0$ [or any value in the interval $(\rho_{0} - 1, 1 - \rho_{1})$], we shall have that $|K_{n}| < 1$ for all *n*, provided $\rho_{n} < 1$, for all *n*. This proves that a sufficient condition for the three-term recursion (1.9) to correspond to a lossless two-component evolution equation is that all the positive values ρ_{n} , defined by (2.17), are less than 1 in magnitude.

In the special case of the three-term recursion (1.8), which is parametrized by the single sequence $\{G(n)\}$, we realize that the above condition for losslessness is satisfied automatically provided we have all G(n) > 1. This corresponds to a continuous model for which the potential is always positive, a well-known condition for the nonexistence of so-called "bound-state" solutions for Schrödinger equations. Losslessness should intuitively correspond to nobound-state situations, since bound states correspond to energy trapped in the medium, which is impossible in lossless

B. Local mapping between reflection coefficients and potentials

Suppose we are given a local reflection coefficient sequence $\{K_n\}$ and we ask for the potential sequences $\{F(n), G(n)\}$. These sequences should obey

$$F^{2}(n)/G(n)G(n+1) = \rho_{n} = (1+K_{n})(1-K_{n-1})$$
(3.13)

and they are not uniquely determined by (3.13). If F(n) = 1for all n and $G^*(n)$ obeys the relation (3.13), then clearly $G(n) = G^*(n)Z^{(-1)*}$ is another valid potential sequence for any $Z \neq 0$. To determine the potential sequence uniquely we need to anchor it at a given point, say n = 0. Suppose, however, that we wish to have a local functional dependence of the potentials on the reflection coefficients. If we try to set $G(n) = \Gamma(K_{n-1})$, then we see that it is necessary for F(n)to be different from 1. Writing (3.13) out in terms of the $\Gamma(K_n)$ we obtain

$$F^{2}(n) = (1+K_{n})\Gamma(K_{n})(1-K_{n-1})\Gamma(K_{n-1}).$$
(3.14)

(3.15)

Requiring further F(n) to be independent of K_{n-1} leads to

and

$$F(n) = \left[(1+K_n)/(1-K_n) \right]^{-1/2}$$

 $\Gamma(K_{n-1}) = G(n) = 1/(1 - K_{n-1})$

and we see that, to have positive G(n) and real F(n), it is necessary for all K_n to be less than unity in magnitude. In this case the evolution equations (2.16) become identical to the lossless propagation model (3.9). Three-term recursions having F(n) and G(n) given by (3.15) were first derived in (Bruckstein and Kailath²) from the corresponding lossless two-component equations as recursions that describe the evolution of the *voltage* on a transmission line with piecewise constant impedance function. The associated *current* signals are defined as $I(n,D) = W_R(n,D) - W_L(n,D)$, and obey a complementary pair of three-term recursions, with G(n) replaced by $\tilde{G}(n) = 1/[1 + K_n]$.

We note that associated *wave difference* or current variables can be defined for arbitrary three-term recursions and then we can write that

$$\begin{bmatrix} V(n,D)\\ I(n,D) \end{bmatrix} = \begin{bmatrix} 1 & 1\\ 1 & -1 \end{bmatrix} \begin{bmatrix} W_R(n,D)\\ W_L(n,D) \end{bmatrix}$$
(3.16)

and we readily obtain two-component evolution equations of the following form:

$$\begin{bmatrix} V(n+1,D) \\ I(n+1,D) \end{bmatrix} = \frac{1}{2} \begin{bmatrix} 1 & 1 \\ 1 & -1 \end{bmatrix} \times \Theta(n,D) \begin{bmatrix} 1 & 1 \\ 1 & -1 \end{bmatrix} \begin{bmatrix} V(n,D) \\ I(n,D) \end{bmatrix}.$$
(3.17)

After some simple algebraic manipulation these equations can be put in the form

$$\begin{bmatrix} V(n+1,D) \\ I(n+1,D) \end{bmatrix} = \frac{F(n)}{2G(n+1)[1+K_n]} \begin{bmatrix} 1 & 0 \\ 0 & (1+K_n)/(1-K_n) \end{bmatrix} \\ \times \begin{bmatrix} D+D^{-1} & D-D^{-1} \\ D-D^{-1} & D+D^{-1} \end{bmatrix} \begin{bmatrix} V(n,D) \\ I(n,D) \end{bmatrix}.$$
(3.18)

This form of two-component equations associated to a transmission-line model was recently found useful in deriving new and computationally efficient algorithms for matrix factorization and estimation applications. It turns out that, by parametrizing the media in terms of $\lambda_n = (1 + K_n)/(1 - K_n)$, these algorithms require about half the number of multiplications when compared to conventional transmission line based algorithms see, e.g., Bistritz, Lev-Ari, and Kailath.¹⁰

IV. DIRECT AND INVERSE SCATTERING PROBLEMS

Up to this point we discussed the basic three-term recursions (1.18) and (1.9) and showed that we can always associate to them two-component propagation equations of the form (2.16) or (3.18). The basic three-term recursions (1.9) can be written out in the space-time coordinates (n,τ) as a second-order system of partial linear difference equations (see, e.g., Fort,¹¹)



FIG. 2. (a) Computable regions of $V(n,\tau)$ from given sequences, and (b) a diagram of influence propagation from a certain data point.

$$a_{r}(n,\tau)V(n+1,\tau) + a_{l}(n,\tau)V(n-1,\tau) + a_{u}(n,\tau)V(n,\tau+1) + a_{d}(n,\tau)V(n,\tau-1) + a_{c}(n,\tau)V(n,\tau) = 0,$$
(4.1)

where

$$a_r(n,\tau) = F(n+1), \quad a_l(n,\tau) = F(n), a_u(n,\tau) = a_d(n,\tau) = -G(n), \quad a_c(n,\tau) = 0.$$
(4.2)

Notice that the $a_*(n,\tau)$ are independent of the time index τ , which made possible the transform analysis, i.e., the representation of (4.1) as a recursion relating formal power series via three-term propagation equations having constants or time-delay/advance operators as coefficients.

To determine $V(n,\tau)$ from (4.1), or the three-term recursions (1.9), we need to be given two initial condition sequences, say $V(0,\tau)$ and $V(1,\tau)$, or in general any pair $V(m,\tau)V(m+1,\tau)$. Then we can forwards or backwards propagate the three-term recursion to obtain the time sequence $V(n,\tau)$ at any point in space *n*. Furthermore, the sequence at $V(n,\tau)$ depends linearly on the initial given initial sequences. To analyze graphically the influence propagation of given data sequences we can plot space-time diagrams as depicted in Fig. 2. We see in these diagrams that given a contiguous portion of the data sequences, it will be possible to determine $V(n,\tau)$ in a certain influence region that can readily be mapped.

If we pass to the equivalent two-component domain, where we are given interlocked recursions for the right and left propagating signals then we need as initial conditions the two wave components at a certain point in space, say $W_R(m,\tau)$ and $W_L(m,\tau)$. The waves at any point in space will obviously be given by (in the transform domain),

$$\begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix} = M(m:n,D) \begin{bmatrix} W_R(m,D) \\ W_L(m,D) \end{bmatrix},$$
(4.3)

where the operator M(m:n,D) is given by

$$M(m:n,D) = \begin{cases} \left[M_{m}^{n}(D) \right] = \prod_{i=n-1}^{m} \frac{F(i)}{G(i+1)\left[1-K_{i}^{2}\right]} \begin{bmatrix} 1 & -K_{i} \\ -K_{i} & 1 \end{bmatrix} \begin{bmatrix} D & 0 \\ 0 & D^{-1} \end{bmatrix}, & \text{if } n > m, \\ \left[M_{n}^{m}(D) \right]^{-1} = \prod_{i=m-1}^{n} \frac{G(i+1)}{F(i)} \begin{bmatrix} D^{-1} & 0 \\ 0 & D \end{bmatrix} \begin{bmatrix} 1 & K_{i} \\ K_{i} & 1 \end{bmatrix}, & \text{if } n < m. \end{cases}$$
(4.4)

Given a legal choice of "anchor" value for the reflection coefficient sequence, at m, K_m , and the data pair $\{V(m,D), V(m+1,D)\}$, we can determine the wave components $\{W_R(m,D), W_L(m,D)\}$ from

 $W_{R}(m,D) + W_{L}(m,D) = V(m,D),$ $\{F(m)/G(m+1)[1+K_{m}]\}(DW_{R}(m,D) + D^{-1}W_{L}(m,D)) = V(m+1,D),$ (4.5)

i.e., we have

$$\begin{bmatrix} W_{R}(m,D) \\ W_{L}(m,D) \end{bmatrix} = E(m,D) \begin{bmatrix} V(m,D) \\ V(m+1,D) \end{bmatrix}$$

$$= \begin{bmatrix} 1 & 1 \\ \frac{F(m)}{G(m+1)[1+K_{m}]} D & \frac{F(m)}{G(m+1)[1+K_{m}]} D^{-1} \end{bmatrix}^{-1} \begin{bmatrix} V(m,D) \\ V(m+1,D) \end{bmatrix}.$$

$$(4.6)$$

Now to determine V(n,D) for any n we can write that

$$V(n,D) = [1 \ 1] \begin{bmatrix} W_R(n,D) \\ W_L(n,D) \end{bmatrix}$$

= [1 \ 1] $M(m:n,D)E(m,D) \begin{bmatrix} V(m,D) \\ V(m+1,D) \end{bmatrix}$
(4.7)

and, since the choice of the reflection coefficient sequence, and thus of K_m , clearly does not enter in the determination of V(n,D) from the initial conditions $\{V(m,D), V(m+1,D)\}$, we have that the row vectors

$$[\Phi(m:n,D)\Psi(m:n,D)] = [1 \ 1]M_{m:n}(D)E(m,D)$$
(4.8)

are independent of K_m , and in fact of the two-component medium altogether.

We have shown that there exist transfer functions relat-

ing the initial conditions to the signals at any point in space, and they are seen to have a very special structure. Let us analyze further the properties of these transfer (Green's) functions.

A. Transfer function properties

Let us first analyze the transfer matrices $M_m^n(D)$, for m < n. We have by the definition (4.4) that

$$M_{m}^{n}(D) = \prod_{i=n-1}^{m} \Theta(i,D)$$
$$= \begin{bmatrix} [M_{m}^{n}(D)]_{11} & [M_{m}^{n}(D)]_{12} \\ [M_{m}^{n}(D)]_{21} & [M_{m}^{n}(D)]_{22} \end{bmatrix}$$
(4.9)

and from the symmetry and invertibility of the elementary layer transfer functions, we obtain that $M_m^n(D)$ is invertible and obeys

$$\tilde{I}M_{m}^{n}(D)\tilde{I} = M_{m}^{n}(D^{-1}) \text{ with } \tilde{I} = \begin{bmatrix} 0 & 1 \\ 1 & 0 \end{bmatrix}.$$
 (4.10)

Furthermore, the entries of $M_m^n(D)$ are polynomials in Dand D^{-1} of degree not more than n - m. The determinant of the matrix operator $M_m^n(D)$ is given by the product of the determinants of the $\Theta(i,D)$ appearing in (4.9). We have

$$\det\{\Theta(i,D)\} = F(i)^2 / G(i+1)^2 [1-K_i^2]$$
(4.11)

yielding

$$\det\{M_{m}^{n}(D)\} = \prod_{i=m}^{n-1} \frac{F(i)^{2}}{G(i+1)^{2} [1-K_{i}^{2}]}$$
$$= \frac{[1-K_{m-1}]G(m)}{[1-K_{n-1}]G(n)}.$$
(4.12)

Note that det $\{M_m^n(D)\}\$ approaches 1 as $m \to -\infty$ and $n \to +\infty$, if the potentials approach 1 as the space index increases. Equation (4.12) also shows that the inverse of the transfer matrix from m to n is given by

$$\begin{bmatrix} M_{m}^{n}(D) \end{bmatrix}^{-1} \\ = \frac{[1 - K_{n-1}]G(n)}{[1 - K_{m-1}]G(m)} \\ \times \begin{bmatrix} [M_{m}^{n}(D)]_{22} & -[M_{m}^{n}(D)]_{12} \\ -[M_{m}^{n}(D)]_{21} & [M_{m}^{n}(D)]_{11} \end{bmatrix} \\ = \frac{[1 - K_{n-1}]G(n)}{[1 - K_{m-1}]G(m)} \\ \times \begin{bmatrix} [M_{m}^{n}(D^{-1})]_{11} & -[M_{m}^{n}(D)]_{12} \\ -[M_{m}^{n}(D)]_{21} & [M_{m}^{n}(D^{-1})]_{22} \end{bmatrix}.$$
(4.13)

Finally, we shall note that the matrices $M_m^n(D)$, when formally regarded as functions of a complex variable z = D, obey on the unit circle

$$\left[M_{m}^{n}\left(e^{j\theta}\right)\right]J\left[M_{m}^{n}\left(e^{j\theta}\right)\right]^{*} = \det\left\{M_{m}^{n}\left(D\right)\right\}J$$
(4.14)

showing that the transfer matrix approaches (as the boundaries tend to $\pm \infty$) a *J*-unitary matrix (provided it converges, i.e., if its entries have limits as functions of a complex variable z).

The transfer matrices $M_m^n(D)$ yield the wave variables at depth *n* from those at depth *m*. It is natural to inquire about the relationship between the waves impinging on a portion of the medium extending from depth *m* to depth *n* and the outgoing ones. If the medium is originally at rest, the emerging waves are the causally generated response of the medium layers, since, as we have seen, the scattering descriptions of elementary medium layers are the causal, linear and time-invariant operators (2.13). A few steps of algebra show that the scattering representation corresponding to $M_m^n(D)$ is

$$S_{m}^{n}(D) = \frac{1}{[M_{m}^{n}(D)]_{22}} \times \begin{bmatrix} \det\{M_{m}^{n}(D)\} & [M_{m}^{n}(D)]_{12} \\ - [M_{m}^{n}(D)]_{21} & 1 \end{bmatrix}.$$
(4.15)

Important properties of the scattering representations $S_m^n(D)$ are causality, inherited from the causality of elementary layers the medium is composed of, and asymptotic losslessness due to the J losslessness of the corresponding transfer representation on the unit circle.

We shall see that the data assumed to be available for inverse scattering, a problem that requires the recovery of the local parametrization of a scattering medium described by either a three-term recursion (or partial difference equation) or a two-component evolution equation, are signals that are causally generated in these media. In the case of the wave-propagation description the causality has immediate meaning, however, if we are dealing with the V(n,D) signals, the concept of a causal signal has to be properly defined. We shall show that the Jost solutions as defined in the classical theory of inverse scattering (Case⁵) do indeed correspond to causally generated waves in the equivalent wave propagation media.

B. Causal solutions and inverse scattering

Suppose first that we are dealing with a two-component system corresponding to the discrete Schrödinger equation (1.8), and that we know the medium parametrization up to depth m, i.e., we have G(m) and K_{m-1} . Assume that we are also given an input-response pair $\{W_R(m,D), W_L(m,D)\}$ at m = 0, that is right causal, i.e., we know that the $W_{R/L}(n,\tau)$ were zero prior to the time $\tau = 0$ when the first nonzero lag of the sequence $W_R(0,\tau)$ is sent into the medium towards the right. This implies that $W_L(0,\tau)$ will be the causal response elicited by right propagating input signal, and a little thought will show that we have, at depth n,

$$W_R(n,\tau) = 0$$
, for $\tau < n$,
 $W_L(n,\tau) = 0$, for $\tau < n+2$, (4.16)

simply due to the delay structure of the medium. Also we shall have, immediately (see Bruckstein and Kailath⁷), that

$$W_L(n,\tau = n+2) = K_n W_R(n,\tau = n)$$
. (4.17)

This observation is the basis of a straightforward inverse scattering process that yields the medium parameters from the data. Indeed, (4.17) readily yields K_0 from the given data, then (2.15) provides G(1), and then we can use (2.16) to determine the sequences $\{W_R(1,D), W_L(1,D)\}$. These sequences are a synthesized set of causal scattering data for the medium starting at depth 2 and extending to $+\infty$, and we can proceed to determine K_1 and G(2), and so on. We thus have an immediate recursive layer-peeling procedure that recovers the medium parameters from the scattering data. For more detail on such procedures see Bruckstein and Kailath.⁷ We note that the scattering data assumed to be available is equivalent to having the left reflection function of S_0^{∞} , the scattering representation for the medium portion extending from 0 to $+\infty$. We have, from the structure of the medium, that the nestedness property

$$[S_m^{\infty}(D)]_{21} = [S_m^n(D)]_{21} + D^{2(n-m)}$$

$$\times \{\text{a bilinear function of } [S_n^{\infty}(D)]_{21}\}$$

$$(4.18)$$

holds, showing that the medium up to depth n can be recovered from the first 2n lags of its impulse response alone. There are interesting connections of the above discussed inverse scattering method to an algorithm by Schur for testing for boundedness inside the unit circle, analytic functions of a complex variable (see Schur,² Kailath,⁶ and Bruckstein and Kailath⁷).

Suppose now that we are dealing with scattering associated to the original three-term recursion (1.8), and assume that the potential sequence G(n) is different from 1 only somewhere between n = 0 and n = N > 0. In this case we know that solutions of (1.8) in the "outer" regions $(-\infty.0]$ and $[N, +\infty),$ are of the form $\phi_R(0,D)D^n + \phi_L(0,D)D^{-n}$ $\phi_R(N,D)D^n$ and $+\phi_L(N,D)D^{-n}$, respectively. Only two of the four $\phi_{R/L}$ (0/N,D) functions are independent, and those may be determined if we are given the signals $V(\cdot, D)$ at any two points in space. If we consider the Jost-type solution (see Case⁵), for which $V^{J}(n,D) = D^{n}$ for n > N, then it is easy to see from the influence propagation diagrams (Fig. 3) that we have for all *n* that

$$V^{J}(n,D) = \sum_{i=n}^{\infty} D^{i} \Omega(n,i) .$$
 (4.19)

Since by definition the sequences $V^{J}(n,D)$ obey the recursion (1.8), we readily obtain that the potential can be computed from the function $\Omega(\cdot, \cdot)$ as

$$G(n) = \Omega(n - 1, n - 1) / \Omega(n, n), \qquad (4.20)$$

therefore, an inversion algorithm could be based on first determining the kernels $\Omega(\cdot, \cdot)$. The classical approach to inverse problems was to derive integral/matrix equations relating the kernels $\Omega(n,i)$ to the scattering (or so-called spectral) data, and then using (4.20) to recover the potentials. We shall outline here an alternative approach, which exploits the structure of the problem directly, and recovers the potentials by propagating a nonlinear difference equation. The insight that yields immediate derivations of these results follows from the equivalent two-component evolution equations that can always be associated to three-term Schrödinger recursions.

The scattering data that we shall assume available are the functions $\phi_{R/L}^{J}(D)$ that specify the behavior of $V^{J}(n,D)$ in the region $(-\infty,0]$. First let us see what is the meaning of a Jost-type solution in the equivalent wave propagation representation. We have at depth n > N, that $V(n,D) = D^{n}$ and also that $V(n + 1,D) = D^{n+1}$, and using (4.6) we obtain

$$W_{R}(n,D) = D^{n} \{ 1 - [1 + K_{n}]D^{2} \}$$

$$\times \{ 1 + D^{2} + D^{4} + D^{6} + \cdots \}, \qquad (4.21)$$

$$W_{L}(n,D) = D^{n+2}K_{n} \{ 1 + D^{2} + D^{4} + D^{6} + \cdots \},$$

where K_n is the local reflection coefficient at depth *n* in some equivalent two-component model, and the formal expansion $1/(1 - D^2) = 1 + D^2 + D^4 + D^6 + \cdots$ was also used. An interesting fact becomes clear from this exercise: the waves corresponding to the Jost solutions are *causal* pairs in the region n > N. Let us look at what happens at the other edge of



FIG. 3. Causality of Jost solutions.

the medium, at n = 0. Assuming without loss of generality that G(0) = 1, we have that $V^{J}(0,D) = \phi_{R}(D) + \phi_{L}(D)$ and $V^{J}(-1,D) = \phi_{R}(D)D^{-1} + \phi_{L}(D)D$. Using (4.6) again we obtain the following expressions for the waves: $W_{R}(0,D) = 1/(1-K_{-1})\{\phi_{R}(D)[[1-K_{-1}] - D^{2}] - \phi_{L}(D)K_{-1}\}\{1 + D^{2} + D^{4} + D^{6} + \cdots\},$ $W_{L}(0,D) = 1/(1-K_{-1})\{\phi_{R}(D)D^{2}K_{-1}$ (4.22) $+ \phi_{L}(D)[1 - [1-K_{-1}]D^{2}]\}$ $\times \{1 + D^{2} + D^{4} + D^{6} + \cdots\}.$

It can be seen that, provided $\phi_L(D)$ has a D^2 factor and is causal, the waves at n = 0 will constitute a causal pair of scattering data. It is relatively easy to show, using the structure of the wave-propagation medium that we shall have a causal pair at all depth *n*. This follows from the relation (4.3) and the form of $[M_n^m(D)]^{-1}$.

Therefore the Jost solutions are causal, in the sense of corresponding to causal waves for any choice of equivalent reflection coefficient sequence. The straightforward layer-peeling algorithm can now be invoked to recover the potential over the region [0,N]. If we would have chosen $K_{-1} = 0$, and assuming that this choice yields a legal two-component model, we would have that

$$W_R(1,D) = \phi_R(D)$$
 and $W_L(1,D) = \phi_L(D)$ (4.23)

and we could start propagating the layer-peeling algorithm on this data. Note that we can never recover both F(n) and G(n) in the general equation (1.9) since the recovery of the local reflection coefficient sequence provides recursions for a nonlinear combination of these parameters. Therefore we can use this method to recover the parameters of either the classical discrete Schrödinger equation (1.8), or of a variant for which only F(n) are different from 1, and there exists a mapping between such equations based on a renormalization, see, e.g., Case.⁸ Also it is important to note that if we assumed $K_{-1} = 0$ and the algorithm yields a $K_{n_0} = 1$ at some point, we can continue the medium recovery from this point on by recomputing the scattering data under the assumption of another K_n . This can be done by obtaining $V(n_c - 1,D)$ and $V(n_c - 2,D)$, say, and recomputing via (4.5) the waves with a different K_{n_c-1} .

It is clear that if we are given the Jost functions, we can obtain alternative causal pairs by assuming that the behavior in the region n > N is $V^{C}(n,D) = D^{n}$ (a causal function), since this corresponds to convolving the input and output waves at n = 0 with the same causal function. Sometimes the scattering data is given by an equivalent set of functions, for which, at n < 0 we have

$$V^{C}(n,D) = D^{n} + s(D)D^{-n} = V^{J}(n,D)/\phi_{L}(0,D) .$$
(4.24)

Here $s(D) = [\phi_L(0,D)]/[\phi_R(0,D)]$ and we consider its causal expansion, starting with D^2 , a property inherited from $\phi_L(0,D)$. The pair $W_R(0,D) = 1$ and $W_L(0,D) = s(D)$ form a causal impulse response pair for the scattering medium associated with the Schrödinger equation, having $K_{-1} = 0$ as the "anchor" reflection coefficient and we shall be able to recover the sequence G(n) from this data, via layer peeling.

In his paper on one-dimensional inverse scattering Case⁵ tests an inversion method, based on the discrete version of the classical theory that proceeds via Gel'fand-Levitan or Marchenko systems of equations, on an example that corresponds to the following one. Assume s(D) of (4.24) is given as

$$s(D) = D^{2} \left\{ \frac{D^{2} - 1}{(2 - g)D^{2} - g} - 1 \right\}$$

= $\left(\frac{1}{g} - 1 \right) D^{2} + \left(\frac{2 - g}{g^{2}} - \frac{1}{g} \right) D^{4}$
+ $\left(\frac{(2 - g)^{2}}{g^{3}} - \frac{2 - g}{g^{2}} \right) D^{6} + \cdots$ (4.25)

Applying the layer-peeling inversion algorithm on the pair of sequences $\{1,s(D)\}$ is seen to yield

$$K_{0} = \frac{1}{g} - 1, \quad K_{1} = \frac{(1 - g)}{-(1 - g) + \frac{1}{2}},$$

$$K_{2} = \frac{(1 - g)}{-2(1 - g) + \frac{1}{2}},$$
(4.26)

which provide, via (2.17) and assuming G(n) = 1 for $n \leq 0$, the potential sequence

$$G(1) = g, \quad G(2) = 1, \quad G(3) = 1,$$
 (4.27)

a result that coincides, up to a shift in space, with that obtained by Case.⁵

V. CONCLUDING REMARKS ON BOUND STATES

We have presented a rather direct approach to the analysis of three-term recursions by mapping them into twocomponent wave-propagation equations having a certain delay-and-interaction structure. Note that we have assumed that the propagating signals are general time sequences and used their representation as formal power series, or generating functions. This immediately puts us into the domain of so-called scattering solutions of Schrödinger equations, therefore we did not discuss issues pertaining to the existence of so-called bound states. Indeed note that once Eqs. (1.8) or (1.9) is written as an eigenvalue problem

$$F(n+1)V(n+1,\lambda) + F(n)V(n-1,\lambda)$$

= $G(n)2\lambda V(n,\lambda)$, (5.1)

it receives more mathematical content than that implied by its interpretation as a wave-propagation equation. The spectrum of values for λ where solutions exist consists of the values $\lambda = \cos \theta \in [-1,1]$, for which $D = e^{j\theta}$, and can have the interpretation of a delay operator in the harmonic analysis of time sequences, but we may also ask what happens for values of λ outside the itnerval [-1,1]. If a solution to (5.1) exists for such a value of λ , then we shall have that $D \in \mathbb{R}$, i.e., it will be a real value, since

$$D + D^{-1} = 2\lambda \tag{5.2}$$

implies that

$$D_{1,2} = \lambda \pm [\lambda^2 - 1]^{1/2}$$
,

which also shows, incidentally, that $D_1 = D_2^{-1}$.

In our two-component interpretation, the waves then disappear in a rather mysterious way and are replaced by single values at each point in space (or rather, time sequences will be have a constant value at all times). The Doperators become pure gains, instead of delays, and the solutions of (5.1) effectively become sequences in only the space dimension. If we want these sequences to be bounded and have finite energy, it turns out that there is only a discrete spectrum of values of λ for which such solutions exist. Suppose that we have a medium described by a sequence of potentials for which the reflection coefficients are very close to zero outside an interval [0,N]. Then we have that $M_0^N(D)$ describes the interaction of signals that exist at n = 0 and those at n = N. In order to have bounded solutions corresponding to some real value $D = d_i < 1$, we need to have, see Fig. 4,

$$W_R(N,d_i) \neq 0$$
 and $W_L(0,d_i) \neq 0$ (5.3a)

together with

$$W_R(0,d_i) = 0$$
 and $W_L(N,d_i) = 0$ (5.3b)

proving that in this case we shall have $[M_0^N(d_i)]_{22} = 0$. Therefore, to have a bound-state solution, $[M_0^N(D)]_{22}$ has to have zeros inside the unit circle. The alternative analysis of $d_j > 1$ leads to the same conclusion, viz. $[M_0^N(D)]_{11}$ needs to have zeros outside the unit circle. This condition however implies that the scattering matrix (4.15) has poles



FIG. 4. Conditions for the existence of bound states.

inside the unit circle, which means that, from a system theoretical point of view, the wave-scattering system is unstable, i.e., some inputs will elicit exponentially growing causal responses. This did not bother us much, since all along we were not concerned with the boundedness of our functions, or with the convergence of the generating functions to legal and analytic functions in the complex variable D. We, however, obtained sufficient conditions under which such problems do not arise: indeed it is easy to realize that if the medium can be associated to a lossless structure we shall never have bound state solutions. Also if we stay in the scattering domain and deal with sequences represented by purely formal generating functions, we can apply the straightforward analysis without any problems. If we want to deal with sequences of numbers that are bounded and summable in some sense, we should only use inputs that do not excite the unstable modes of the scattering medium. These issues underlie much of the effort in the rigorous treatment of discrete inverse scattering problems, however, we feel that the simplicity of the approach outlined in this paper explains the mechanism of inverse scattering in a way that most directly exploits the wave interaction model assumed.

ACKNOWLEDGMENTS

This work was supported in part by the U.S. Army Research Office, under Contract No. DAAG-79-C-0215, and by the Air Force Office of Scientific Research, Air Force Systems Command, under Contract AF49-620-79-C-0058.

- ¹K. Chadan and P. Sabatier, "Inverse problems in quantum scattering theory," *Physics Monographs* (Springer, New York, 1977).
- ²A. M. Bruckstein and T. Kailath, "Spatio-temporal scattering and inverse problems," ISL Report, EE, Stanford University, 1983.
- ³A. M. Bruckstein, B. C. Levy, and T. Kailath, "Differential methods in inverse scattering," SIAM J. Appl. Math. **45**, 312 (1985).
- ⁴K. M. Case and M. Kac, "A discrete version of the inverse scattering problem," J. Math. Phys. 14, 594 (1973).
- ⁵K. M. Case, "The discrete inverse scattering problem in one dimension," J. Math. Phys. **15**, 143 (1974).
- ⁶T. Kailath, "A theorem of I. Schur and its impact on modern signal processing," Operator Theory, Adv. Appl. 18, 9 (1986).
- ⁷A. M. Bruckstein and T. Kailath, "Inverse scattering for discrete transmission-line models," SIAM Rev. 29, 359 (1987).
- ⁸K. M. Case, "On discrete inverse scattering problems II," J. Math. Phys. **14**, 916 (1973).
- ⁹M. J. Ablowitz and H. Segur, "Solitons and the inverse scattering transform," *SIAM Studies 4* (SIAM, Philadelphia, PA, 1981).
- ¹⁰Y. Bistritz, H. Lev-Ari, and T. Kailath, "Immitance-domain Levinson algorithms," ISL preprint, 1986.
- ¹¹T. Fort, "Partial linear difference equations," Am. Math. Mon. 1957, 161.
- ¹²I. Schur, "On power series which are bounded in the interior of the unit circle I," J. Reine Angew. Math. 147, 205 (1917) [reprinted in Operator Theory, Adv. Appl. 18 (1986)].

Time-dependent canonical formalism of thermally dissipative fields and renormalization scheme

I. Hardman, H. Umezawa, and Y. Yamanaka

Theoretical Physics Institute, Department of Physics, University of Alberta, Edmonton, Alberta T6G 2J1, Canada

(Received 3 February 1987; accepted for publication 5 August 1987)

The canonical formalism of thermally dissipative semifree fields in the time-dependent situation is presented. The use of thermal covariant derivatives simplifies the formulation considerably. With this formalism one can unambiguously obtain the interaction Hamiltonian under any thermal situation which together with the free propagator enables perturbative calculations to be performed. The "on-shell" renormalization condition in the time-dependent case is also discussed. The model of a system with a thermal reservoir illustrates how the present formalism works in time-dependent situations.

I. INTRODUCTION

Thermo field dynamics (TFD) is a real time quantum field theory with thermal degrees of freedom. In a series of papers¹⁻⁷ we have been extending TFD to describe any thermal situation, including nonequilibrium phenomena. This extended TFD has been shown to be equivalent to the density matrix formalism with the Liouville equation.¹ The purpose of this paper is to present this extended TFD in a systematic form placing a strong emphasis on the time dependence of thermal situations.

Before starting on the extended TFD we summarize here the equilibrium TFD^{8,9} that is now well established. The basic concepts in equilibrium TFD are the following: (i) the doubling of each degree of freedom through the *tilde conjugation rules*, (ii) the thermal Bogoliubov transformation taking care of the freedom in choice of the temperature, and (iii) the total Lagrangian density is given by

$$\hat{\mathscr{L}} = \mathscr{L} - \tilde{\mathscr{L}},\tag{1.1}$$

where \mathscr{L} is the usual Lagrangian density and $\widetilde{\mathscr{L}}$ is obtained from \mathscr{L} by the tilde conjugation rules.

According to (i), with an operator A we associate with it its tilde conjugate \tilde{A} . The tilde-conjugation rules are summarized in the next section. In TFD, therefore, any field has its tilde conjugate, forming a thermal doublet. A similar situation of the doubling of the degrees of freedom is observed in other formulations of real-time quantum field theory at finite temperature such as the C^* -algebraic formalism,¹⁰ the path-ordering formalism of the Schwinger-Keldysh type,¹¹ and the superoperator formalism¹² of the Liouville equation. The relation between TFD and the C^* -algebraic formalism was beautifully analyzed by Ojima in Ref. 13. The relation between TFD and the path-ordering method was clarified in Ref. 14. In the theory of the density matrix there appear right- and left-operating operators which form thermal doublets and which are called the superoperators.¹² The relationship between the superoperator formalism and TFD was given in Ref. 1. Intuitively speaking, the nontilde operators take care of the usual quantum excitations, while the tilde operators describe the thermal excitations.8

Since each field is a thermal doublet, there appears a

mixing of the doublet components through the Bogoliubov transformations. In the equilibrium TFD the different choices of temperature correspond to different choices of these Bogoliubov transformations. Thus we were led to the concept (ii), i.e., the thermal Bogoliubov transformation.

The concept (iii) for the total Lagrangian in TFD exhibits the following beautiful feature of TFD. Since \mathscr{L} does not depend on the temperature, neither does \mathscr{G} . The temperature appears only through the thermal Bogoliubov transformations. Since these features play a fundamental role in the extension of TFD to nonequilibrium situations let us elaborate on them. As is well known the language of quantum field theory has a dual structure.⁸ The basic entities such as the Lagrangian, the Heisenberg equations, and the basic canonical commutation relations are expressed in terms of the basic fields called the Heisenberg operators, while the phenomenological language is constructed in terms of the "quasiparticle" operators, which are the particles including all the renormalization effects. The state-vector space is the Fock space associated with these quasiparticles. When a Lagrangian is given, an essential task in quantum field theory is to obtain the expressions for the Heisenberg operators written in terms of the quasiparticle operators. This expression, which has been frequently called the dynamical map,⁸ determines the realization of the Heisenberg operators. The different choices of the thermal Bogoliubov transformation correspond to different choices of the realization of the Heisenberg operators. Thus concepts (i) and (ii) imply that different temperatures correspond to the different realization of the Heisenberg operators that are determined by the total Lagrangian $\hat{\mathscr{L}}$. Since the state-vector space is the Fock space associated with the quasiparticles, the best choice for the unperturbed Hamiltonian is the Hamiltonian of the quasiparticles. Denoting the annihilation and creation operators by $(\xi_k, \tilde{\xi}_k)$ and $(\xi_k^{\dagger}, \tilde{\xi}_k^{\dagger})$, respectively (i.e., $\xi_k |0\rangle = \tilde{\xi}_k |0\rangle = 0$ and $\langle 0|\xi_k^{\dagger} = \langle 0|\tilde{\xi}_k^{\dagger} = 0$ with the thermal vacua $|0\rangle$ and $\langle 0|$, where k represents any quantum number), the Fock space is constructed by the cyclic operation of creation operators on $|0\rangle$ and of annihilation operators on $\langle 0 |$. Denoting the quasiparticle energy by ω_k , the temporal behavior of the creation and annihilation operators

are required to be of the form

$$\xi_k(t) = \xi_k E_k(t), \quad \tilde{\xi}_k(t) = \tilde{\xi}_k E_k^*(t)$$
 (1.2)

with $E_k(t) = \exp(-i\omega_k t)$. Then, the feature (ii) means that the oscillator operators $(a_k(t), \tilde{a}_k(t))$ of the fields are related to $(\xi_k(t), \tilde{\xi}_k(t))$ through the Bogoliubov transformation B,

$$\begin{bmatrix} a_k(t) \\ \tilde{a}_k^{\dagger}(t) \end{bmatrix} = B \begin{bmatrix} \xi_k(t) \\ \tilde{\xi}_k^{\dagger}(t) \end{bmatrix},$$
(1.3)

where B is a 2×2 matrix.

A remarkable fact in equilibrium TFD is that the loop corrections to the self-energy of the quasiparticles create an imaginary term, making the quasiparticles dissipative (spontaneous creation of dissipation). This dissipative behavior is caused by the presence of the background tilde field whose quanta carry negative energy. The real quanta described by the nontilde operators live in a world with the background tilde field of negative energy, which easily absorbs the energy of real quanta and makes them dissipative. An essential reason why the quasiparticles easily become dissipative, irrespective of the fact that the Lagrangian $\hat{\mathscr{L}}$ is real, is that the quantum field has an infinite number of degrees of freedom. When we denote the number of degrees of freedom by N, the (1/N) expansion provides us with a clear picture for the spontaneous creation of dissipation. The leading term in the (1/N) expansion exhibits the dissipative behavior. Since the Lagrangian is real when N is finite the dissipative behavior disappears when all of the terms in the (1/ N) expansion are summed up. However, when N is infinite and only the leading term in the expansion is picked up, the dissipative behavior may arise from a real Lagrangian. This consideration will be explicitly demonstrated in the analysis of the reservoir model in Sec. V. The appearance of a dissipative effect in the equilibrium situation is not surprising. Although the equilibrium situation is stationary, a short-time measurement induces excited states which approach the ground state dissipatively. Thus although the thermal average of the observable is independent of time, the dissipative effect appears in the multipoint functions such as Green's functions and correlation functions.

The same dissipative effect becomes apparent even in the thermal average of observables when the thermal situation becomes nonstationary. Even a brief consideration² of a time-dependent transition between a disordered state and an ordered state indicates that the renormalization effects due to loop corrections become dependent on time, making most of the physical quantities (such as energy, order parameter, dissipative coefficient, particle number, etc.) dependent on time. This analysis also indicated a need for time-dependent renormalization.² This motivates us to extend TFD to timedependent nonequilibrium phenomena.

The task of extending TFD began with the construction of the Hamiltonian for dissipative quasiparticle.^{1,2,4} The idea is that once this Hamiltonian, say \hat{H}^0 , becomes known the interaction Hamiltonian is given by $\hat{H} - \hat{H}^0$, where \hat{H} is the Hamiltonian obtained from the total Lagrangian $\hat{\mathscr{L}}$. This provides us with the Feynman-diagram method for computation. In the density matrix formalism the description of an open system in terms of quasiparticles was obtained by eliminating the variables of the reservoir through the coarse graining process. Our first approach¹ was to reformulate this coarse graining process in terms of the TFD formalism. In Ref. 3 this approach was reformulated in terms of the Green's function formalism and the result was compared with the path-ordering method. Our next step, made in Ref. 4, was to construct \hat{H}^0 by making the so-called thermal state condition time dependent [this is equivalent to making the thermal Bogoliubov transformation (1.3) dependent on time]. It turned out that the \hat{H}^0 thus obtained depends on time explicitly. A remarkable fact is that this \hat{H}^{0} at the longtime limit $(t \to \infty)$ coincides with \hat{H}^0 obtained from the coarse graining process. Another significant point in the analysis in Ref. 4 was to point out that the spontaneous creation of dissipation can be formulated in terms of the renormalization technique. The basic idea here is that since the unperturbed quasiparticles include all of the self-energy corrections, and since the self-energy corrections frequently make quasiparticles dissipative, the unperturbed Hamiltonian contains dissipative terms. Thus we are led to the selfconsistent renormalization condition which states that the interaction Hamiltonian $(\hat{H} - \hat{H}^0)$ should not create any on-shell self-energy. Since the on-shell self-energy is a 2×2 thermal matrix this condition gives four complex equations, which are equivalent to eight real equations. Although not all of these equations may be independent of each other, this condition does contain more information than the energy ω_k and the dissipative coefficient κ_k . However, the precise formulation of the self-consistent renormalization condition required more analysis and was not presented in Ref. 4. To learn more from this formalism in Ref. 4 we applied it to two models in the stationary case in Ref. 5, one of which was an exactly solvable reservoir model. Since TFD treats the reservoir as a part of the quantum field system we do not need to eliminate the reservoir variables. This analysis clearly exhibits the previously mentioned argument based on the (1/N)expansion with the limit $N \rightarrow \infty$. The other model was a real scalar field model without a reservoir. Since, as was pointed out above, TFD treats a reservoir as a part of a quantum field system, it is readily applicable to any quantum field system without any reservoir. Here it is important to note that a quantum field has an infinite number of degrees of freedom as the reservoir does and also that a stochastic effect is caused by the background tilde field, which has negative energy. In any application of the formalism to a quantum field system it is important to put the theory of dissipative quasiparticle in the canonical form. This was done in Ref. 6. However, this just dealt with the stationary case.

With all of the results mentioned above, we are now ready to put the foundations of nonequilibrium TFD in a systematic form. This is the purpose of this paper. The exactly solvable reservoir model in a time-dependent situation will be treated as an example. The foundations of nonequilibrium TFD are the previously-mentioned basic concepts, (i), (ii), and (iii). We now state that any (even time-dependent) Bogoliubov transformation, which is consistent with other basic concepts, is permitted and that *the entire set of the Bogoliubov transformations cover all possible thermal phe*- nomena. In other words, the general TFD considers all possible realizations of the Heisenberg fields controlled by the Lagrangian $\hat{\mathscr{L}}$. In the equilibrium TFD we considered only a particular set of Bogoliubov transformations. To cover the time-dependent phenomena, we consider time-dependent Bogoliubov transformation. Furthermore, since the quasiparticle is dissipative we might assume that $E_k(t)$ in (1.2) is $\exp(-i\omega_k t - \kappa_k t)$ with unknown parameters ω_k and κ_k . However, we find that when a time-dependent thermal situation is considered the higher-order loop corrections may make ω_k and κ_k dependent on time. We thus state that the dissipative quasiparticles are defined by $(\xi_k(t), \xi_k^{-1}(t))$ with

$$E_k(t) = \exp\left[\int_0^t ds \{-i\omega_k(s) - \kappa_k(s)\}\right].$$

This dissipative quasiparticle is called the "semifree" quasiparticle. The appearance of κ_k in $E_k(t)$ and the time-dependent Bogoliubov transformation are the basic features in the extended TFD. Our unperturbed particles are the fully renormalized semifree quasiparticles. Since these particles are fully renormalized the interaction Hamiltonian (which contains the renormalization counter terms) should not create any on-shell correction. This leads to the self-consistent renormalization condition which, in general, depends on time.

The extended TFD distinguishes itself from the equilibrium one by the fact that the unperturbed Hamiltonian \hat{H}^{0} is not the usual free Hamiltonian, but the semifree Hamiltonian which has an imaginary term responsible for the dissipative effects. Then we need the full knowledge of the semifree field in order to perform a perturbative calculation formulated in the terminology of the interaction representation. We first formulate the extended TFD in terms of the oscillator operators $a(\mathbf{k})$ and $a^{\dagger}(\mathbf{k})$. It is not an easy task to formulate the extended TFD in terms of field operators $\psi(x)$ and $\psi(x)$ when the formalism is required to include any time-dependent thermal situation. In order to construct such a formalism one requires a method simpler and more systematic than those that have previously been presented. In this paper the formulation of the extended TFD begins with the time-dependent Bogoliubov transformation applied to a semifree field. This leads us to the thermal covariant derivatives that simplify the construction of the extended TFD.

The reconstruction of the generalized semifree oscillator operators⁴ is given in Sec. II by use of the thermal covariant derivatives. Then by using the same method we formulate a theory of semifree fields which is applicable to time-dependent thermal phenomena. Although the use of the time-dependent Bogoliubov transformation and the covariant derivatives simplified the formulation of semifree fields, the consideration in Sec. III is still limited to semifree fields of type 1 (i.e., the semifree fields with a positive frequency part only). Study of semifree fields of type 2 (i.e., those with both the positive and negative frequency parts) is limited to time-independent situations only. This is presented in the Appendix. The works in the past^{4,7} showed that there exists a flexibility in the formulation of TFD: TFD carries an arbitrary parameter denoted by α . In equilibrium TFD it has been shown¹⁴ that the freedom in choice of α corresponds to the freedom in choice of path in the pathordering method. This might lead us to anticipate that this freedom might disappear in nonequilibrium TFD. Contrary to this expectation, *this freedom remains in nonequilibrium* TFD. This means that the choice of α is not intrinsically related to the choice of path in the path-ordering method. Furthermore, the consideration in Sec. II will show that the formulation of TFD has a wider choice of arbitrary parameters. The entire consideration in this paper is made without specifying these parameters.

Once we have a formulation of the semifree field we try to develop the perturbative calculation in the extended TFD including the time-dependent situation by using a semifree quasiparticle field for the unperturbed field. To do this we need the knowledge of the interaction Hamiltonian. A method for identifying the interaction Hamiltonian is presented in Sec. IV. The argument there begins with the fact that the TFD Lagrangian density is given by $\hat{\mathscr{L}} = \mathscr{L} - \tilde{\mathscr{L}}$ and the Hamiltonian \hat{H} in TFD follows from $\hat{\mathscr{L}}$ through the usual route in the canonical theory. The Hamiltonian \hat{H}^0 of the semifree quasiparticle field is given in Sec. III. Then the interaction Hamiltonian in the extended TFD is found to be $\hat{H}_{int} = \hat{H} - \hat{H}^0$.

In the perturbative calculation in the extended TFD a vital step is the renormalization procedure which leads to a set of so-called self-consistent equations.^{4,5} The latter equations determine not only the renormalized energy and the dissipative coefficient, but also the temporal behavior of the average number density when the initial value of the average number density is given. Any renormalization procedure is based on the on-shell renormalization condition which is associated with the renormalization point. In previous papers the on-shell renormalization condition has been given only for the time-independent case, so in Sec. V we formulate the on-shell renormalization condition for time-dependent thermal situations. This renormalization method is applied to a simple solvable model to illustrate how the energy and dissipative coefficients as well as temporal behavior of the average number density can be determined. We find a nonvanishing dissipative coefficient (spontaneous creation of dissipation).

Section VI is devoted to a brief description of the physical mechanism for the appearance of the dissipative coefficient in an isolated quantum field system. There, the importance of the infinite degrees of freedom and the presence of the negative energy background field (tilde quanta) is explained in terms of the (1/N)-power expansion method. A similarity between the mechanism of spontaneous breakdown of symmetries and the one for spontaneous creation of dissipation is also pointed out. A discussion on the relation between the spontaneous creation of dissipation and the spontaneous breakdown of time-translation symmetry is also presented. It will be pointed out what kinds of problems still remain in order to complete the extended TFD.

II. RECONSTRUCTION OF THE GENERAL SEMIFREE HAMILTONIAN IN TFD

We start by introducing certain quasiparticle annihilation and creation operator $\xi(\mathbf{k})$ and $\xi^{\dagger}(\mathbf{k})$ in the interaction representation with the thermal vacua $|0\rangle$, $\langle 0|$:

$$\boldsymbol{\xi}(\mathbf{k})|0\rangle = \mathbf{0},\tag{2.1a}$$

$$\langle 0|\boldsymbol{\xi}^{\dagger}(\mathbf{k}) = 0. \tag{2.1b}$$

These operators obey the commutation relation

$$[\xi(\mathbf{k}),\xi^{\dagger}(\mathbf{l})]_{\sigma} = \delta(\mathbf{k} - \mathbf{l}), \qquad (2.2)$$

where the σ commutator is defined by

$$[A,B]_{\sigma} = AB - \sigma BA, \qquad (2.3)$$

with $\sigma = 1$ for boson operators and $\sigma = -1$ for fermion operators. The other commutators vanish. We suppressed all indices representing spin and any other internal degrees of freedom other than the momentum k. For simplicity we will drop the momentum variable in the section below unless it is required.

It is now widely known that in TFD with an operator A we associate with it its tilde conjugate \tilde{A} according to the rules

$$(AB)^{\tilde{}} = \widetilde{A}\widetilde{B}, \qquad (2.4a)$$

 $(c_1A + c_2B)^{\tilde{}} = c_1^*\tilde{A} + c_2^*\tilde{B}$ (c_1,c_2 are c numbers), (2.4b)

$$(A^{\dagger})^{\tilde{}} = \tilde{A}^{\dagger}, \qquad (2.4c)$$

$$(\widetilde{A})^{\tilde{}} = \sigma A,$$
 (2.4d)

$$|0\rangle^{\tilde{}} = |0\rangle, \quad \langle 0|^{\tilde{}} = \langle 0|.$$
 (2.4e)

Taking the tilde conjugate of (2.1) and (2.2) we get

$$\tilde{\xi} \left| 0 \right\rangle = 0, \tag{2.5a}$$

$$\langle 0|\tilde{\xi}^{\dagger} = 0, \qquad (2.5b)$$

$$[\tilde{\boldsymbol{\xi}}(\mathbf{k}), \tilde{\boldsymbol{\xi}}^{\dagger}(\mathbf{l})]_{\sigma} = \delta(\mathbf{k} - \mathbf{l}).$$
(2.6)

The σ -commutation relations between tilde and nontilde operators are assumed to vanish. Equations (2.1) and (2.5) are called the thermal state conditions.

For later convenience we introduce the thermal doublet notation

$$\xi^{\mu} = \begin{bmatrix} \xi \\ \bar{\xi}^{\dagger} \end{bmatrix}^{\mu}, \quad \bar{\xi}^{\mu} = [\xi^{\dagger} - \sigma \tilde{\xi}]^{\mu} \quad (\mu = 1, 2). \quad (2.7)$$

Then we write (2.2) and (2.6) compactly as

$$\left[\xi^{\mu}, \bar{\xi}^{\nu}\right]_{\sigma} = \delta^{\mu\nu}.$$
(2.8)

We now introduce $\xi(t)^{\mu}$ and $\overline{\xi}(t)^{\mu}$ with the form

$$\xi(t)^{\mu} = E_0(t)^{\mu\nu}\xi^{\nu}, \qquad (2.9a)$$

$$\bar{\xi}(t)^{\mu} = \bar{\xi}^{\nu} E_0^{-1}(t)^{\nu\mu}, \qquad (2.9b)$$

where

$$E_0(t) = \exp\left[-i \int_0^t ds \{\omega(s) - i\kappa_0(s)\tau_3\}\right].$$
 (2.10)

We call $\xi(t)^{\mu}$ and $\overline{\xi}(t)^{\mu}$ the semifree quasiparticle operators.

Our state vector space, called the thermal space, is the linear space spanned by the set of bra and ket state vectors that are generated by cyclic operations of the annihilation operators ξ^{1} and $\overline{\xi}^{2}$ on the thermal vacuum $\langle 0|$, and of the creation operators ξ^{2} and $\overline{\xi}^{1}$ on $|0\rangle$.

Let us now identify how the quasiparticle creation and annihilation operators are related to the variables describing a quantum field system. A usual choice for variables is harmonic oscillator-type operators, a and a^{\dagger} , satisfying $[a(\mathbf{k}), a^{\dagger}(\mathbf{l})]_{\sigma} = \delta(\mathbf{k} - \mathbf{l})$. In TFD these operators also form thermal doublets. Introducing the notations

$$a(t) = \hat{S}_{a}^{-1}(t)a\hat{S}_{a}(t), \qquad (2.11a)$$

$$a^{\dagger\dagger}(t) = \widehat{S}_{a}^{-1}(t)a^{\dagger}\widehat{S}_{a}(t),$$
 (2.11b)

where the time-translation operator $\hat{S}_a(t)$ will be specified later, the thermal doublets are

$$a(t)^{\mu} = \begin{bmatrix} a(t) \\ \tilde{a}^{\dagger\dagger}(t) \end{bmatrix}^{\mu}, \quad \bar{a}(t)^{\mu} = \begin{bmatrix} a^{\dagger\dagger}(t) - \sigma \tilde{a}(t) \end{bmatrix}^{\mu}.$$
(2.12)

The thermal state conditions for $a(t)^{\mu}$ and $\bar{a}(t)^{\mu}$ in TFD (cf. Ref. 4) can be rephrased by the statement that these operators in the interaction representation are related to $\xi(t)^{\mu}$ and $\bar{\xi}(t)^{\mu}$ through a generalized Bogoliubov transformation

$$a(t)^{\mu} = b^{-1}(t)^{\mu\nu}\xi(t)^{\nu}, \qquad (2.13a)$$

$$\bar{a}(t)^{\mu} = \bar{\xi}(t)^{\nu} b(t)^{\nu \mu}.$$
(2.13b)

In this paper we use these as basic relations in TFD, rather than the thermal state conditions for a and \bar{a} . The above relations give the commutation relation

$$[a(t)^{\mu},\bar{a}(t)^{\nu}] = \delta^{\mu\nu}.$$
(2.14)

For the consistency of the tilde conjugation rules (2.4) for both a and ξ , we must have

$$let b(t) = 1.$$
(2.15)

The observable average number density n(t) is given in terms of a(t) and $a^{\dagger\dagger}(t)$ by

$$n(t) = \langle 0 | a^{\dagger\dagger}(t) a(t) | 0 \rangle.$$
(2.16)

Let us express the matrix b by

$$b(t) = \begin{bmatrix} b_{11} & b_{12} \\ b_{21} & b_{22} \end{bmatrix}.$$
 (2.17)

Using (2.13) together with (2.9) and considering the definitions of the quasiparticle creation and annihilation operators, (2.1), (2.2), (2.5), and (2.6), together with the unimodular nature of b, (2.15), we find that

$$b_{12}b_{21} = \sigma n(t), \tag{2.18a}$$

$$b_{11}b_{22} = 1 + \sigma n(t).$$
 (2.18b)

From (2.18) it follows that the matrix b(t) contains two free parameters. We can therefore write b(t) as

$$b(t) = W_L^{-1}(t)B_0(t)W_R(t), \qquad (2.19)$$

where

$$B_0(t) = \begin{bmatrix} 1 + \sigma n(t) & -\sigma n(t) \\ -1 & 1 \end{bmatrix},$$
 (2.20a)

$$W_L(t) = \exp[\tau_3 \ln b_L(t)],$$
 (2.20b)

$$W_R(t) = \exp[\tau_3 \ln b_R(t)].$$
 (2.20c)

The two arbitrary parameter b_L and b_R , which generally depend on time and momentum, etc., are related to b_{21} and b_{22} by

$$b_L^2 = -b_{21}b_{22}, \qquad (2.21a)$$

$$b_R^2 = -b_{21}/b_{22}.$$
 (2.21b)

The parameter b_R is related to the thermal state condi-

tions (2.1) and (2.5), rewritten in terms of
$$a(t)^{\mu}$$
 and $\bar{a}(t)^{\mu}$,

$$[a(t)^{1} - F(t)a(t)^{2}]|0\rangle = 0, \qquad (2.22a)$$

$$[a(t) + F(t)a(t)]|0\rangle = 0,$$
 (2.220)

$$(0|[a(t)^{2} + \sigma f(t)F^{-1}(t)a(t)^{2}] = 0, \qquad (2.22c)$$

$$\langle 0|[a(t)^2 - \sigma f(t)F^{-1}(t)a(t)^1] = 0, \qquad (2.22d)$$

where

$$f(t) = n(t) / [1 + \sigma n(t)], \qquad (2.23)$$

$$F(t) = \sigma f(t) b_{R}^{-2}(t).$$
 (2.24)

The tilde conjugation of (2.22a) together with (2.21b) shows that F(t), and therefore, $b_R(t)$ should be real. Note that the parameter b_L does not appear in (2.22). In comparison with Refs. 4 and 7 we obtain the previous thermal state conditions if we choose

$$F(t) = f^{\alpha}(t) \tag{2.25}$$

with $0 \le \alpha \le 1$. This implies that the arbitrariness of b_R includes the freedom of choice of α in the previous papers.^{4,7} However, the present formulation is wider than the previous one since b_R and F(t) are quite arbitrary as far as they are real.

On the other hand, the flexibility of choice of b_L has not been discussed in the previous papers. To take into account this flexibility, we introduce

$$E(t) = W_L(t)E_0(t)$$
 (2.26)

$$= \exp\left[-i\int_0^t ds\{\omega(s) - i\kappa(s)\tau_3\}\right]$$
(2.27)

with

$$\kappa(t) = \kappa_0(t) - \partial_t \ln b_L(t). \qquad (2.28)$$

Now (2.13) read as

$$a(t)^{\mu} = B^{-1}(t)^{\mu\rho} E(t)^{\rho\nu} \xi^{\nu}, \qquad (2.29a)$$

$$\bar{a}(t)^{\mu} = \bar{\xi}^{\nu} E^{-1}(t)^{\nu\rho} B(t)^{\rho\mu}, \qquad (2.29b)$$

with

$$B(t) = B_0(t) W_R(t).$$
(2.30)

This motivates us to introduce the *thermal covariant deriva*tives,

$$D^{+}(t) = B^{-1}(t)E(t)\partial_{t}E^{-1}(t)B(t), \qquad (2.31a)$$

$$\hat{D}^{+}(t) = B^{-1}(t)E(t)\hat{\partial}_{t}E^{-1}(t)B(t).$$
 (2.31b)

The equations of motion for a(t) and $\bar{a}(t)$ are

$$iD^{+}(t)^{\mu\nu}a(t)^{\nu} = 0,$$
 (2.32a)

$$i\bar{a}(t)^{\nu}\overline{D}^{+}(t)^{\nu\mu} = 0.$$
 (2.32b)

We obtain from (2.31)

$$D^{+}(t) = \partial_{t} + i[\omega(t) - iP(t)], \qquad (2.33a)$$

$$\overleftarrow{D}^{+}(t) = \overleftarrow{\partial}_{t} - i[\omega(t) - iP(t)], \qquad (2.33b)$$

$$P(t) = P_1(t) + P_2(t), \qquad (2.34)$$

$$P_1(t) = \kappa(t)A(t) + \sigma \dot{n}(t)\tau(t), \qquad (2.35)$$

$$P_2(t) = W_R^{-1}(t)\partial_t W_R(t), \qquad (2.36)$$

$$=\partial_t \{\ln b_R(t)\}\tau_3,\tag{2.37}$$

where the various matrices are defined by

$$A(t) = W_R^{-1}(t)A_0(t)W_R(t) = B^{-1}(t)\tau_3 B(t), \qquad (2.38)$$

$$\mathbf{A}_{0}(t) = \mathbf{B}_{0}^{-1}(t)\tau_{3}\mathbf{B}_{0}(t)$$

$$= \begin{bmatrix} 1 + 2\sigma n(t) & -2\sigma n(t) \\ 2(1 + \sigma n(t)) & -(1 + 2\sigma n(t)) \end{bmatrix},$$
 (2.39)

$$\tau(t) = W_R^{-1}(t)\tau_0 W_R(t), \qquad (2.40)$$

$$\tau_0 = \begin{bmatrix} 1 & -1 \\ 1 & -1 \end{bmatrix}.$$
 (2.41)

The equations in (2.32) show that the semifree Hamiltonian for a and \overline{a} is

$$\hat{H}^{0}_{a}(t) = \bar{a}(t)^{\mu} [\omega(t) - iP(t)]^{\mu\nu} a(t)^{\nu}.$$
(2.42)

This Hamiltonian becomes the one given in Ref. 4 when (2.25) is assumed. It is easy to prove that $\hat{H}_{a}^{0}(t)$ is tildian,

$$\left[i\widehat{H}_{a}^{0}(t)\right]^{\tilde{}}=i\widehat{H}_{a}^{0}(t). \tag{2.43}$$

We have

$$a(t)^{\mu} = \widehat{S}_{a}^{-1}(t)a^{\mu}\widehat{S}_{a}(t), \qquad (2.44a)$$

$$\bar{a}(t)^{\mu} = S_{a}^{-1}(t)\bar{a}^{\mu}S_{a}(t), \qquad (2.44b)$$

with

$$\partial_t \hat{S}_a(t) = -i\hat{H}_a^0(t)\hat{S}_a(t), \qquad (2.45a)$$

$$(S_a(t)) = S_a(t),$$
 (2.450)

$$S_a(0) = 1.$$
 (2.45c)

The construction of the semifree Hamiltonian $\hat{H}_{a}^{0}(t)$ in this paper is considerably simpler than the one in the previous papers. This is because our consideration in this paper is based on the existence of quasiparticle creation and annihilation operators which form the thermal doublets ξ^{μ} and $\bar{\xi}^{\mu}$, and also on the use of the generalized Bogoliubov transformation. In the next two sections we formulate the semifree field theory, not in terms of oscillator variables (a,\bar{a}) but in terms of field variables. There we find the thermal covariant derivatives extremely useful.

III. A GENERAL THEORY FOR SEMIFREE FIELDS

In this section we formulate the semifree field theory in terms of the field variables. Our consideration is confined to fields of type 1 (i.e., fields that contain only the positive frequency part). We have not made the extension to fields of type 2 (i.e., fields which contains both positive and negative frequency parts) in time-dependent situations. In the timeindependent case fields of type 2 will be treated in the Appendix.

In this section we consider a Bose field without spin or any other internal degrees of freedom, the extension to general multicomponent fields being straightforward.

We consider thermal doublet semifree fields, say $\psi(x)^{\mu}$ and $\overline{\psi}(x)^{\mu}$, which are linear in the semifree operators $a(t,\mathbf{k})^{\mu}$ and $\overline{a}(t,\mathbf{k})^{\mu}$, considered in the last section. Following the assumption in the usual quantum field theory we require that the time development of ψ and $\overline{\psi}$ is controlled by the operator $\widehat{S}_a(t)$ in (2.44),

$$\psi(\mathbf{x})^{\mu} = \widehat{S}_{a}^{-1}(t)\psi(t=0,\mathbf{x})^{\mu}\widehat{S}_{a}(t), \qquad (3.1a)$$

$$\overline{\psi}(\mathbf{x})^{\mu} = \widehat{S}_{a}^{-1}(t)\overline{\psi}(t=0,\mathbf{x})^{\mu}\widehat{S}_{a}(t), \qquad (3.1b)$$

where x stands for (t,x). Therefore we expand $\psi(x)$ and $\overline{\psi}(x)$ as

$$\psi(\mathbf{x})^{\mu} = \int d^{3}k \ U(\mathbf{x},\mathbf{k})^{\mu\nu}a(t,\mathbf{k})^{\nu}, \qquad (3.2)$$

$$= \int d^{3}k \, u(\mathbf{x},\mathbf{k})^{\mu\nu}a(\mathbf{k})^{\nu}, \qquad (3.3)$$

$$\bar{\psi}(\mathbf{x})^{\mu} = \int d^{3}k \,\bar{a}(t,\mathbf{k})^{\nu} \overline{U}(\mathbf{x},\mathbf{k})^{\nu\mu}, \qquad (3.2')$$

$$= \int d^{3}k \,\bar{a}(\mathbf{k})^{\nu} \bar{u}(x,\mathbf{k})^{\nu\mu}, \qquad (3.3')$$

where U, \overline{U}, u , and \overline{u} are certain c number 2×2 matrix wave functions. According to (2.29), u and \overline{u} are related to U and \overline{U} , respectively, in the following way:

$$u(\mathbf{x},\mathbf{k}) = U(\mathbf{x},\mathbf{k})B^{-1}(t,\mathbf{k})E(t,\mathbf{k})B(0,\mathbf{k}), \qquad (3.4a)$$

$$\overline{u}(x,\mathbf{k}) = B^{-1}(0,\mathbf{k})E^{-1}(t,\mathbf{k})B(t,\mathbf{k})\overline{U}(x,\mathbf{k}). \quad (3.4b)$$

Here $E(t,\mathbf{k})$ depends on \mathbf{k} through (2.27) with $\omega(s)$ and $\kappa(s)$ being replaced with $\omega(s,\mathbf{k})$ and $\kappa(s,\mathbf{k})$, respectively.

Now, in order for ψ and $\overline{\psi}$ to be canonical, the wave functions u and \overline{u} have to form a canonical complete set⁶ that will be defined shortly. In the case of type 1 field, the simplest choice of U and \overline{U} ,

$$U(\mathbf{x},\mathbf{k})^{\mu\nu} = [\delta^{\mu\nu}/(2\pi)^{3/2}]e^{i\mathbf{k}\cdot\mathbf{x}}, \qquad (3.5a)$$

$$\overline{U}(\mathbf{x},\mathbf{k})^{\mu\nu} = [\delta^{\mu\nu}/(2\pi)^{3/2}]e^{-i\mathbf{k}\cdot\mathbf{x}}, \qquad (3.5b)$$

turns out to give such a canonical set $\{u, \overline{u}\}$. We are going to show this below.

Since the thermal covariant derivatives $D^+(t)$ and $\overline{D}^+(t)$ in (2.31) depend on the momentum **k** we denote them by $D^+(t,\mathbf{k})$ and $\overline{D}^+(t,\mathbf{k})$. The field equations for ψ and $\overline{\psi}$ and those for u and \overline{u} follow from (3.2)–(3.5) and (2.32),

$$iD^{+}(t, -i\nabla)^{\mu\nu}\psi(x)^{\nu} = 0,$$
 (3.6a)

$$i\psi(x)^{\nu}D^{+}(t,i\nabla)^{\nu\mu} = 0,$$
 (3.6b)

$$iD^{+}(t,\mathbf{k})^{\mu\lambda}u(x,\mathbf{k})^{\lambda\nu}=0, \qquad (3.6c)$$

$$i\overline{u}(\mathbf{x},\mathbf{k})^{\nu\lambda}\overline{D}^{+}(t,\mathbf{k})^{\lambda\mu} = 0.$$
(3.6d)

Using (3.4) and (3.5), we can prove that u and \overline{u} satisfy the orthonormalization relation

$$\int d^{3}x \,\overline{u}(x,\mathbf{k})u(x,l) = \delta(\mathbf{k}-l)$$
(3.7)

at any time t.

Furthermore, we can explicitly prove the sum rule

$$G^{+}(\mathbf{x},\mathbf{x}')^{\mu\nu} = \int d^{3}k \ u(\mathbf{x},\mathbf{k})^{\mu\lambda} \overline{u}(\mathbf{x}',\mathbf{k})^{\lambda\nu}, \qquad (3.8)$$

where

$$G^{+}(x,x')^{\mu\nu} = \int \frac{d^{3}k}{(2\pi)^{3}} e^{i\mathbf{k}\cdot(\mathbf{x}-\mathbf{x}')} \times [B^{-1}(t,\mathbf{k})E(t,\mathbf{k})E^{-1}(t',\mathbf{k})B(t',\mathbf{k})]^{\mu\nu}.$$
 (3.9)

The function $G^+(x,x')^{\mu\nu}$ has the following properties:

$$G^{+}(x,x')^{\mu\nu}\delta(t-t') = \delta^{\mu\nu}\delta^{(4)}(x-x'), \qquad (3.10a)$$

$$iD^{+}(t, -i\nabla)^{\mu\lambda}G^{+}(x, x')^{\lambda\nu} = 0,$$
 (3.10b)

$$iG^{+}(x,x')^{\nu\lambda}\overleftarrow{D}^{+}(t',i\overleftarrow{\nabla}')^{\lambda\mu} = 0.$$
(3.10c)

When u and \overline{u} satisfy the orthonormalization conditions (3.7) and the sum rule (3.8) in which the function $G^+(x,x')$ has the properties in (3.10), the set $\{u,\overline{u}\}$ is called the *canonical complete set*. Using the properties of the canonical complete set, we obtain the equal-time canonical commutation relation

$$[\psi(t,\mathbf{x})^{\mu},\bar{\pi}(t,\mathbf{x}')^{\nu}] = i\delta^{\mu\nu}\delta(\mathbf{x}-\mathbf{x}'), \qquad (3.11)$$

where

$$\bar{\pi}(x) = i\bar{\psi}(x) \tag{3.12}$$

is the canonical conjugate of $\psi(x)$. This is the reason for the name "canonical complete set."

The semifree field equations in (3.6) can be derived from the following Lagrangian density:

$$\widehat{\mathscr{L}}^{0}(x) = i\overline{\psi}(x)^{\mu}D^{+}(t, -i\nabla)^{\mu\nu}\psi(x)^{\nu}.$$
(3.13)

The momentum field $\overline{\pi}$ conjugate to ψ , already mentioned above, is indeed given by

$$\bar{\pi}(x)^{\mu} = \partial \hat{\mathscr{L}}^{0}(x) / \partial \dot{\psi}(x)^{\mu} = i \bar{\psi}(x)^{\mu}, \qquad (3.14)$$

which is consistent with (3.12). The usual Legendre transformation gives us the semifree field Hamiltonian \hat{H}_{a}^{0} ,

$$\begin{aligned} \hat{H}^{0}_{a}(t) &= \int d^{3}x \left[\bar{\pi} \dot{\psi} - \hat{\mathscr{L}}^{0} \right] \\ &= \int d^{3}x \, \bar{\psi}(x)^{\mu} \left[\omega(t, -i\nabla) - iP(t, -i\nabla) \right]^{\mu\nu} \psi(x)^{\nu}, \end{aligned}$$

$$(3.15)$$

where use was made of (2.33) in which $\omega(t)$ and P(t) depend on k as $\omega(t,\mathbf{k})$ and $P(t,\mathbf{k})$, respectively. Needless to say, \hat{H}_a^0 is reduced to (2.42) when it is rewritten in terms of oscillator variables, $a(t,\mathbf{k})$ and $\bar{a}(t,\mathbf{k})$.

Here some comments on the semifree fields in the timedependent case may be in order. As it was pointed out in Sec. II the generalized Bogoliubov transformation contains two arbitrary parameters, $b_L(t,\mathbf{k})$ and $b_R(t,\mathbf{k})$. Since the physical results are independent of these quantities we expect that these quantities do not cause any ambiguity in the choice of the Lagrangian. When we change these quantities ψ changes into ψ' and, at the same time, the expression for the Lagrangian density changes from \mathcal{L} to \mathcal{L}' . However, the Lagrangian density is really unique in the sense that

$$\mathscr{L}'(\psi') = \mathscr{L}(\psi). \tag{3.16}$$

Then the canonical momenta of ψ and ψ' are given by $\bar{\pi} = \partial \mathcal{L} / \partial \dot{\psi}$ and $\bar{\pi}' = \partial \mathcal{L}' / \partial \psi'$, respectively. Thus the Hamiltonian densities for ψ and ψ' are $[\bar{\pi}\psi - \mathcal{L}]$ and $[\bar{\pi}'\psi' - \mathcal{L}']$. The transformation $(\psi,\bar{\pi}) \rightarrow (\psi',\bar{\pi}')$ is a canonical transformation. For example, when $b_R(t)$ is changed to $b'_R(t)$, we have

$$\hat{H}_{a}^{0\prime}(t) - \hat{H}_{a}^{0}(t)$$

$$= -i \int d^{3}k \,\bar{a}(t,\mathbf{k})^{\mu} \left[\partial_{t} \left\{ \ln \frac{b_{R}^{\prime}(t,\mathbf{k})}{b_{R}(t,\mathbf{k})} \right\} \tau_{3} \right]^{\mu\nu} a(t,\mathbf{k})^{\nu}.$$
(3.17)

The causal two-point function has the form

$$\Delta_{c}(x,x')^{\mu\nu} = -i\langle 0|T[\psi(x)^{\mu}\overline{\psi}(x')^{\nu}]|0\rangle \qquad (3.18)$$
$$= [B^{-1}(t,-i\nabla)_{\mathscr{G}_{c}}(x,x')B(t',i\overline{\nabla}')]^{\mu\nu},$$
$$(3.19)$$

with

$$\varphi_{c}(\mathbf{x},\mathbf{x}')^{11} = -i\theta(t-t')\int \frac{d^{3}k}{(2\pi)^{3}}e^{i\mathbf{k}\cdot(\mathbf{x}-\mathbf{x}')}$$
$$\times \exp\left[-i\int_{t'}^{t} ds\{\omega(s,\mathbf{k}) - i\kappa(s,\mathbf{k})\}\right],$$
(3.20a)

$$\varphi_{c}(\mathbf{x},\mathbf{x}')^{22} = i\theta(t'-t) \int \frac{d^{3}k}{(2\pi)^{3}} e^{i\mathbf{k}\cdot(\mathbf{x}-\mathbf{x}')} \\ \times \exp\left[-i \int_{t'}^{t} ds \{\omega(s,\mathbf{k}) + i\kappa(s,\mathbf{k})\}\right],$$
(3.20b)

$$\varphi_c(x,x')^{12} = \varphi_c(x',x)^{21} = 0.$$
 (3.20c)

In practical computations using the Feynman-diagram method we need the knowledge of the Feynman function and the interaction Hamiltonian in the interaction representation. The Feynman function is given by $\Delta_c(x,x')$ in (3.19). The interaction Hamiltonian will be studied in the next section.

IV. THE INTERACTION HAMILTONIAN

Assuming a given system, we denote its Lagrangian density by \mathscr{L} . Then we can construct $\widetilde{\mathscr{L}}$ by means of the tilde conjugation rules. It has been shown that the total Lagrangian density in TFD is

$$\hat{\mathscr{L}} = \mathscr{L} - \tilde{\mathscr{L}}. \tag{4.1}$$

Different thermal situations are covered by the different realizations of operators satisfying the field equation which follows from $\hat{\mathscr{L}}$. As it was pointed out above, when we deal with time-dependent thermal situations the Hamiltonian depends on the choice of field variables, while the Lagrangian does not. In other words $\hat{\mathscr{L}}$ is independent of the thermal situation. Therefore use of the Lagrangian is more suitable for the unambiguous definition of the interaction than use of the Hamiltonian.

Let us consider the following Lagrangian density for an unrenormalized bosonic complex field ψ_0 :

$$\mathscr{L}(\psi_0) = \psi_0^* [i \,\partial_t - \omega_0] \psi_0 - g_0 W(\psi_0^* \psi_0). \tag{4.2}$$

Here W(x) is a polynomial function of x and the subscript 0 means unrenormalized quantities. According to the tilde conjugation rules we have the total Lagrangian density $\hat{\mathscr{L}}$ in TFD as follows:

$$\hat{\mathscr{L}} = \mathscr{L} - \tilde{\mathscr{L}} = \bar{\psi}_0(x)^{\mu} [i \partial_t - \omega_0(i\nabla)] \psi_0(x)^{\mu} - g_0 \{ W(\bar{\psi}_0(x)^1 \psi_0(x)^1) - W(-\bar{\psi}_0(x)^2 \psi_0(x)^2) \}, (4.3)$$

where the thermal doublet notation is made use of. The bare quantities should be related to renormalized ones through the renormalization transformations. In addition to the energy renormalization

$$\omega_0 = \omega - \delta \omega, \tag{4.4}$$

where ω is the renormalized energy and $\delta \omega$ is the self-energy, we consider also the wave function renormalization. In TFD the wave function renormalization factor can be a 2×2 matrix that will be denoted by V, with V being a 2×2 matrix. Thus the wave function renormalization reads as

$$\psi_0(x)^{\mu} = V(t)^{\mu\nu} \psi(x)^{\nu}, \qquad (4.5a)$$

$$\bar{\psi}_0(x)^{\mu} = \bar{\psi}(x)^{\nu} V^{-1}(t)^{\nu\mu}.$$
(4.5b)

We consider this wave function renormalization by requiring that V is diagonal, i.e.,

$$V(t)^{\mu\nu} = V(t;\tau_3)^{\mu\nu}.$$
(4.6)

The reason for this will become clear shortly. In this paper we do not discuss the coupling constant renormalization and the overall wave function renormalization other than V.

Substituting (4.4) and (4.5) into (4.3) we have $\hat{\mathscr{L}}$ expressed in terms of the renormalized quantities

$$\hat{\mathscr{L}} = \hat{\mathscr{L}}^0 + \hat{\mathscr{L}}_{\text{int}}, \qquad (4.7)$$

$$\begin{aligned} \widehat{\mathcal{L}}^{0} &= \overline{\psi}^{\mu} [i \partial_{t} - \omega + iP]^{\mu\nu} \psi^{\nu}, \qquad (4.8) \\ \widehat{\mathcal{L}}_{i,1} &= -g_{0} \{ W(Z_{2} \overline{\psi}^{1} \psi^{1}) - W(-Z_{2} \overline{\psi}^{2} \psi^{2}) \} \end{aligned}$$

$$+ \bar{\psi}^{\mu} [iV^{-1}\dot{V} + \delta\omega - iP]^{\mu\nu} \psi^{\nu}, \qquad (4.9)$$

where P(t) was given in (2.34). Comparing (4.8) with (3.13) and (2.33a) we see that the unperturbed Lagrangian density $\hat{\mathscr{L}}^0$ in (4.7) is the semifree field Lagrangian density in (3.13). Note that *iP* in $\hat{\mathscr{L}}^0$ and *iP* in $\hat{\mathscr{L}}_{int}$ cancel one another. The total Hamiltonian follows from (4.7) as

$$\widehat{H} = \widehat{H}_{a}^{0} + \widehat{H}_{int}, \qquad (4.10)$$

where \hat{H}_{a}^{0} was given in (3.15) and

$$\begin{aligned} \widehat{H}_{int} &= \int d^{3}x \left[g_{0} \{ W(\overline{\psi}^{1}\psi^{1}) - W(-\overline{\psi}^{2}\psi^{\nu}) \} \\ &- \overline{\psi}^{\mu} \left[iV^{-1}\dot{V} + \delta\omega - iP \right]^{\mu\nu}\psi^{\nu} \right] \end{aligned} \tag{4.11} \\ &= \int d^{3}x \left[g_{0} \{ W(\overline{\psi}^{1}\psi^{1}) - W(-\overline{\psi}^{2}\psi^{2}) \} \\ &- \overline{\psi}^{\mu} \left[iV^{-1}\dot{V} + \delta\omega - i(\kappa A + \dot{n}\tau) \\ &- iW_{R}^{-1}\dot{W}_{R} \right]^{\mu\nu}\psi^{\nu} \right], \end{aligned} \tag{4.12}$$

where (2.34)-(2.36) are considered. Without loss of generality we can choose

$$V = W_R. \tag{4.13}$$

This gives

$$V^{-1}\dot{V} - W_R^{-1}\dot{W}_R = 0. \tag{4.14}$$

According to (2.20c), this makes V diagonal. Now the inter-

action Hamiltonian is

$$\hat{H}_{int} = \int d^{3}x [g_{0}\{W(\bar{\psi}^{1}\psi^{1}) - W(-\bar{\psi}^{2}\psi^{2})\} - \bar{\psi}^{\mu}[\delta\omega(t) - i\{\kappa(t)A(t) + \dot{n}(t)\tau(t)\}]^{\mu\nu}\psi^{\nu}],$$
(4.15)

$$= \int d^{3}x [g_{0}\{W(\bar{\psi}^{1}\psi^{1}) - W(-\bar{\psi}^{2}\psi^{2})\} \\ - [\bar{\psi}W_{R}^{-1}(t)]^{\mu} [\delta\omega(t) - i\{\kappa(t)A_{0}(t) \\ + \dot{n}(t)\tau_{0}\}]^{\mu\nu} [W_{R}(t)\psi]^{\nu}], \qquad (4.16)$$

where (2.38) and (2.40) were used. Note that A_0 and τ_0 were given in (2.39) and (2.41), respectively.

With the knowledge of this interaction Hamiltonian and the casual two-point function of the semifree field in (3.19), we can formulate the perturbative computation in terms of the Feynman diagrams. A vital step in this formulation is the renormalization procedure which leads to the self-consistent equations for ω , κ , and n. This will be illustrated in the next section by means of a simple model. There, an explicit form of the on-shell renormalization condition is presented. We will see also how the self-consistent equation determines the time dependence of n(t) when the initial value n (t = 0) is given.

V. EXAMPLE OF TIME-DEPENDENT RENORMALIZATION—THE THERMAL RESERVOIR MODEL

In this section we illustrate the self-consistent renormalization method by taking as an example the well-known thermal reservoir model, which is a system with a thermal reservoir. We treat the entire system including the reservoir as one system. Therefore this example shows explicitly how dissipation is created spontaneously. The explicit calculation below will also indicate how the self-consistent renormalization method should be formulated in time-dependent situations.

The model consists of a simple harmonic oscillator (a,a^{\dagger}) and a reservoir $(R_k, R_k^{\dagger}; k = 1, 2, ..., N)$ interacting with each other. These operators satisfy

$$[a,a^{\dagger}]_{\sigma} = 1, \tag{5.1a}$$

$$\begin{bmatrix} R_k, R_l^{\dagger} \end{bmatrix}_{\sigma} = \delta_{kl}. \tag{5.1b}$$

The total Hamiltonian H is assumed to take the simple bilinear form

$$H = \omega_0 a^{\dagger} a + \sum_{k=1}^{N} \Omega_{0k} R_k^{\dagger} R_k$$
$$+ g \sum_{k=1}^{N} (R_k^{\dagger} a + a^{\dagger} R_k), \quad \text{with } N \to \infty.$$
(5.2)

According to the usual notion of a reservoir, g^2 is of order 1/N,

$$\bar{g}^2 = Ng^2. \tag{5.3}$$

The spacing of reservoir energy levels is of order 1/N,

$$\Omega_{0k} = \overline{\Omega}_0 + k\delta; \quad \delta = O(1/N), \tag{5.4}$$

and the bandwidth Δ of the reservoir energy spectrum is

given by

$$\Delta = N\delta. \tag{5.5}$$

Note that the limit $N \to \infty$ should be performed at the end of computation, because otherwise the (R_k, R_k^{\dagger}) system cannot act as a thermal reservoir. At this limit the energy spectrum Ω_{0k} becomes a continuum. We have previously treated this model in the stationary case.⁵ There the (1/N)power expansion was made. With each finite order in this expansion a dissipation does appear. When N is finite we should sum up all of the (1/N)-power expansion terms and then the dissipation simply disappears (i.e., $\kappa = 0$). However, since N is infinite only the leading term in the (1/N)power expansion should be picked up. Then a dissipative effect remained (i.e., $\kappa \neq 0$).

Physically, the spontaneously created dissipation is a result of decay processes of the oscillator into the reservoir quanta through infinite channels, the decay probability for each channel being infinitesimal. This decay is the thermal instability that is encouraged by the negative energy of the tilde quanta. Thus the infinite degrees of freedom together with the negative energy of the tilde quanta is the origin of the spontaneous creation of dissipation. The role played by the infinite degrees of freedom is the same as the one in the phenomena of spontaneous breakdown of symmetries which require the infinite degrees of freedom.⁸ We might say that the system behaves like an "open" system, not because of the elimination of reservoir variables (which we do not do), but because each decay product appearing with infinitesimal probability is not observable. The situation is very similar to the case of impurity scattering because the unobservable tilde quanta act like the impurities. In other words, the dissipation is caused by the communication of nontilde quanta with the unobservable background field which is the tilde field. Intuitively this tilde field effect represents the effect of thermally excited particles. It could be that this process may act as the coarse graining.

In previous papers⁵ we studied only the time-independent situation. The result agreed fully with the well-known result of the projection operator method applied to the Liouville equation for the density matrix. Although the thermal averages are time independent, any two-point function or correlation function has a dissipative effect, which is a manifestation of the contributions from the excited states dissipatively approaching the equilibrium state. The number distribution of the oscillator was found to be $n = 1/[e^{\beta \omega} - \sigma]$ with $\beta = 1/k_B T$, with T being the temperature of the reservoir. This was a result of the self-consistent equations obtained by the renormalization method. Therefore we may expect that when we prepare the initial value of n(t) different from $1/[e^{\beta\omega} - \sigma]$, *n* should become dependent on time. The temporal behavior of n(t) should also be determined by the self-consistent renormalization method. Thus our consideration in this section presents the renormalization condition for time-dependent thermal phenomena which has never previously been presented. With the help of the considerations in the previous sections we are now ready to treat the model (5.2) in a time-dependent situation.

Since the (R_k, R_k^{\dagger}) system is a reservoir it should be in

an equilibrium state and have the number distribution

$$n_k = \left[e^{\beta\Omega_k} - \sigma\right]^{-1},\tag{5.6}$$

with Ω_k being the renormalized energy. On the other hand, the initial value (i.e., the value of t = 0) of the number distribution n(t) of the (a,a^{\dagger}) oscillator is not specified.

According to (2.42) the unperturbed semifree Hamiltonian is

$$\widehat{H}^{0}(t) = \overline{a}(t)^{\mu} [\omega - iP(t)]^{\mu\nu} a(t)^{\nu} + \sum_{k=1}^{N} \overline{R}_{k}(t)^{\mu} [\Omega_{k} - iP_{k}]^{\mu\nu} R_{k}(t)^{\nu}.$$
(5.7)

Since Ω_k and P_k are time independent (2.34) together with (2.20), (2.38), and (2.39) gives

$$P_k = \kappa_k A_k \tag{5.8}$$

with

$$A_{k} = W_{Rk}^{-1} A_{0k} W_{Rk}, (5.9)$$

$$A_{0k} = \begin{bmatrix} 1 + 2\sigma n_k & -2\sigma n_k \\ 2(1 + \sigma n_k) & -(1 + 2\sigma n_k) \end{bmatrix},$$
 (5.10)

$$W_{Rk} = e^{\tau_3 \ln b_{Rk}}.$$
 (5.11)

The P(t) was given in (2.34).

We assume that the wave function renormalization (4.5) with $V = W_R$ [cf. (4.13)] has already been performed. Then the interaction Hamiltonian is given by (4.16),

$$\begin{aligned} \hat{H}_{int} &= g \sum_{k=1}^{N} \{ \overline{R}_{k}(t)^{\mu} [W_{Rk}^{-1} W_{R}(t)]^{\mu\nu} a(t)^{\nu} \\ &+ \overline{a}(t)^{\mu} [W_{R}^{-1}(t) W_{Rk}]^{\mu\nu} R_{k}(t)^{\nu} \} \\ &- [\overline{a}(t) W_{R}^{-1}(t)]^{\mu} [\delta \omega - i \{ \kappa A_{0}(t) + \sigma \dot{n}(t) \tau_{0} \}]^{\mu\nu} \\ &\times [W_{R}(t) a(t)]^{\nu} \\ &- \sum_{k=1}^{N} [\overline{R}_{k}(t) W_{Rk}^{-1}]^{\mu} [\delta \Omega_{k} - i \kappa_{k} A_{0k}]^{\mu\nu} \\ &\times [W_{Rk} R_{k}(t)]^{\nu}. \end{aligned}$$
(5.12)

Note that according to (2.35) $\dot{n}(t)$ should carry the factor σ .

We first assume that ω and κ are independent of time. We are going to justify this assumption by showing that it is consistent with the self-consistent equation which will be derived shortly.

The casual two-point functions are

$$\Delta_{c}(t_{1},t_{2})^{\mu\nu} = -i\langle 0|T[a(t_{1})^{\mu}\overline{a}(t_{2})^{\nu}]|0\rangle, \qquad (5.13)$$

$$= [B^{-1}(t_1)\varphi(t_1-t_2)B(t_2)]^{\mu\nu}, \qquad (5.14)$$

$$\Delta_{ck}(t_1,t_2)^{\mu\nu}\delta_{kl} = -i\langle 0|T[R_k(t_1)^{\mu}\overline{R}_l(t_2)^{\nu}]|0\rangle,$$
(5.13')

$$= \left[B_{k}^{-1} \varphi_{k} (t_{1} - t_{2}) B_{k} \right]^{\mu \nu} \delta_{kl}, \quad (5.14')$$

where

29

$$\varphi(t_1 - t_2)^{11} = -i\theta(t_1 - t_2)\exp[-i(\omega - i\kappa)(t_1 - t_2)], \quad (5.15a)$$

$$\varphi(t_1 - t_2)^{22} = i\theta(t_2 - t_1)\exp[-i(\omega + i\kappa)(t_1 - t_2)], \quad (5.15b)$$

$$\varphi_{k}(t_{1}-t_{2})^{11} = -i\theta(t_{1}-t_{2})\exp[-i(\Omega_{k}-i\kappa_{k})(t_{1}-t_{2})],$$
(5.15c)

 $\varphi_k(t_1-t_2)^{22}$

$$= i\theta(t_2 - t_1) \exp[-i(\Omega_k + i\kappa_k)(t_1 - t_2)], \quad (5.15d)$$

$$\varphi(t_1 - t_2)^{\mu\nu} = \varphi_k (t_1 - t_2)^{\mu\nu} = 0, \text{ for } \mu \neq \nu.$$
 (5.15e)

It can be seen from (5.12) that the computation is very much simplified when we choose the arbitrary matrix W_R to be independent of time. This choice corresponds to taking a time-dependent α because n(t) or f(t) change with time [see (2.20c), (2.23)-(2.25)]. Then the proper self-energy diagram for (a,\bar{a}) gives

$$\Sigma(t_{1},t_{2})^{\mu\nu} = W_{R}^{-1\mu\lambda} \bigg[g^{2} \bigg\{ \sum_{k=1}^{N} W_{Rk} \Delta_{ck} (t_{1} - t_{2}) W_{Rk}^{-1} \bigg\} - \{ \delta\omega - i(\kappa A_{0} (t_{1}) + \sigma \dot{n}(t_{1}) \tau_{0}) \} \delta(t_{1} - t_{2}) \big]^{\lambda\rho} W_{R}^{\rho\nu},$$
(5.16)

while the one for (R_k, \overline{R}_k) is

$$\Sigma_{k}(t_{1},t_{2})^{\mu\nu} = W_{Rk}^{-1\mu\lambda} [g^{2} \{W_{R}\Delta_{c}(t_{1},t_{2})W_{R}^{-1}\} - \{\delta\Omega_{k} - i\kappa_{k}A_{0k}\}\delta(t_{1}-t_{2})]^{\lambda\rho}W_{Rk}^{\rho\nu}.$$
 (5.17)

The connected full propagators of a and R_k (i.e., G and G_k) satisfy the following Dyson equations:

$$G(t_1,t_2) = \Delta_c(t_1,t_2) + \int ds_1 \, ds_2 \, \Delta_c(t_1 \cdot s_1) \Sigma(s_1,s_2) G(s_2,t_2), \quad (5.18a)$$

$$G_k(t_1,t_2) = \Delta_{ck}(t_1,t_2) + \int ds_1 \, ds_2 \, \Delta_{ck}(t_1-s_1) \Sigma_k(s_1,s_2) G_k(s_2,t_2).$$

(5.18b)

The self-energies above are functions of two times, t_1 and t_2 . As a result of this we cannot directly follow the same procedure for the on-shell renormalization as in the stationary case or in the usual quantum field theory.

In the present model, the on-shell renormalization conditions of Σ and Σ_k are stated as follows:

 $\Sigma(\omega, k_0) = 0, \tag{5.19a}$

$$\Sigma(k_0,\omega) = 0, \tag{5.19b}$$

$$\Sigma_k(\Omega_k, k_0) = 0, \qquad (5.19c)$$

$$\Sigma_k(k_0, \Omega_k) = 0, \tag{5.19d}$$

for arbitrary real k_0 . Here the Fourier transform of any twotime function [say $f(t_1,t_2)$] is defined by

$$f(k_0^1, k_0^2) = \int \int dt_1 dt_2 e^{ik_0^{1}t_1} f(t_1, t_2) e^{-ik_0^{2}t_2}.$$
 (5.20)

These on-shell renormalization conditions are equivalent to

$$\Sigma(\omega,t) = 0, \tag{5.21a}$$

$$\Sigma(t,\omega) = 0, \tag{5.21b}$$

$$\Sigma_k(\Omega_k, t) = 0, \tag{5.21c}$$

$$\Sigma_k(t,\Omega_k) = 0, \tag{5.21d}$$

at any time t.

When (5.16) and (5.17) are considered, (5.21a) and (5.21c) read as

$$W_{R}\Sigma(\omega,t)W_{R}^{-1} = e^{i\omega t} \left[\frac{\overline{g}^{2}}{N}\sum_{k=1}^{N}\frac{1}{(\omega-\Omega_{k})+i\kappa_{k}A_{0k}} -\delta\omega+i(\kappa A_{0}(t)+\sigma \dot{n}(t)\tau_{0})\right] = 0, \qquad (5.22a)$$

$$W_{L}\Sigma_{k}(\Omega_{k},t)W_{k}^{-1}$$

$$W_{Rk} \Sigma_{k} (\Omega_{k}, t) W_{Rk}^{-1}$$

$$= \frac{\overline{g}^{2}}{N} W_{R} \left[\int ds \, e^{i\Omega_{k}s} B^{-1}(s) g(s-t) B(t) \right] W_{R}^{-1}$$

$$- e^{i\Omega_{k}t} \left[\delta\Omega_{k} - i\kappa_{k}A_{0k} \right] = 0, \qquad (5.22b)$$

with (5.3) and (5.14).

Equation (5.22b) is the self-consistent equation for $\delta\Omega_k$ and κ_k . This leads to the vanishing of $\delta\Omega_k$ and κ_k when N is taken to infinite with fixed \bar{g}^2 ,

$$\delta\Omega_k = O(1/N), \tag{5.23a}$$

$$\kappa_k = O(1/N). \tag{5.23b}$$

This situation is the same as the one in the stationary case studied previously. On the other hand, dividing another self-consistent equation (5.22a) into real and imaginary parts, we have the matrix equations,

$$\kappa A_0(t) + \sigma \dot{n}(t) \tau_0 = \frac{\overline{g}^2}{N} \sum_{k=1}^N \frac{\kappa_k}{(\omega - \Omega_k)^2 + \kappa_k^2} A_{0k},$$
(5.24a)

$$\delta\omega = \frac{\bar{g}^2}{N} \sum_{k=1}^{N} \frac{(\omega - \Omega_k)}{(\omega - \Omega_k)^2 + \kappa_k^2},$$
(5.24b)

using the property $A_{0k}^2 = 1$. Equation (5.24b) again has the same expression as the one in the stationary case studied in Ref. 5.

Let us now solve (5.24a) at the limit $N \to \infty$. Then Ω_k becomes a continuous parameter denoted by Ω and summation becomes integration,

$$\kappa A_{0}(t) + \sigma \dot{n}(t)\tau_{0} = \frac{\bar{g}^{2}}{\Delta} \int_{\Delta} d\Omega \ \pi \delta(\omega - \Omega) A_{0R}(\Omega)$$
$$= \frac{\pi \bar{g}^{2}}{\Delta} A_{0R}(\omega), \qquad (5.25)$$

where $A_{0R}(\Omega)$ is A_{0k} in (5.10) with n_k being replaced with [cf. (5.6)],

$$n_R(\Omega) = [e^{\beta\Omega} - \sigma]^{-1}.$$
 (5.26)

Let us introduce \bar{n} by

$$\overline{n} = n_R(\omega) = [e^{\beta\omega} - \sigma]^{-1}.$$
(5.27)

Then

$$A_{0R}(\omega) = \begin{bmatrix} 1 + 2\sigma\bar{n} & -2\sigma\bar{n} \\ 2(1 + \sigma\bar{n}) & -(1 + 2\sigma\bar{n}) \end{bmatrix}.$$
 (5.28)

Now recall (2.39) and (2.41), i.e.,

$$A_0(t) = \begin{bmatrix} 1 + 2\sigma n(t) & -2\sigma n(t) \\ 2(1 + \sigma n(t)) & -(1 + 2\sigma n(t)) \end{bmatrix}, \quad (5.29)$$

$$\tau_0 = \begin{bmatrix} 1 & -1 \\ 1 & -1 \end{bmatrix}.$$
(5.30)

The four matrix elements of the self-consistent equation (5.25) give the following three equations:

$$\kappa \{1 + 2\sigma n(t)\} + \sigma \dot{n}(t) = (\pi \bar{g}^2 / \Delta) \{1 + 2\sigma \bar{n}\},$$
 (5.31a)

$$2\sigma\kappa n(t) + \sigma \dot{n}(t) = (\pi \bar{g}^2 / \Delta) 2\sigma \bar{n}, \qquad (5.31b)$$

$$2\kappa\{1+\sigma n(t)\}+\sigma \dot{n}(t)=(\pi \bar{g}^2/\Delta)2\{1+\sigma \bar{n}\}.$$
 (5.31c)

However the addition of (5.31b) and (5.31c) leads to (5.31a), which implies that the independent equations are any two of (5.31a)-(5.31c). They give

$$\kappa = \pi \bar{g}^2 / \Delta, \tag{5.32a}$$

$$\dot{n}(t) = -2\kappa[n(t) - \bar{n}]. \qquad (5.32b)$$

The dissipative factor κ is found to be the same as in the stationary case.⁵ When the initial value of n(t) is given by n(0), (5.32b) determines the temporal behavior of n(t),

$$n(t) = [n(0) - \bar{n}]e^{-2\kappa t} + \bar{n}.$$
 (5.33)

This leads to

$$n(\infty) = \bar{n}. \tag{5.34}$$

Then (5.27) indicates that n(t) in (5.33) starts with the initial value n(0) and approaches the equilibrium value $n_R(\omega)$ at $t \to \infty$.

The fact that $\kappa \neq 0$ indicates the spontaneous creation of dissipation.

Note that the substitution of the solutions of the selfconsistent equation rewrite $\hat{H}^{0}(t)$ in (5.7) as the following form for the final equilibrium situation:

$$\hat{H}^{0}(t) = \bar{a}(t)^{\mu} [\omega - i\overline{P}]^{\mu\nu} a(t)^{\nu} + \sum_{k=1}^{N} \overline{R}_{k}(t)^{\mu} \Omega_{k} R_{k}(t)^{\mu}, \qquad (5.35)$$

where

$$\overline{P} = \kappa A_{0R}(\omega). \tag{5.36}$$

Thus the reservoir model is described by the time-independent Hamiltonian for equilibrium. But this may not be true when the higher-order corrections are taken account of in general nonlinearly interacting systems.

When we happen to choose $n(0) = \overline{n}$ then n(t) becomes independent of time and we have the equilibrium situation. This is the case studied in Ref. 5.

We now comment on the Heisenberg operators. When we express the Heisenberg operators a_H^{μ} in terms of the interaction representation operator a^{μ} and R_k^{μ} , we find that

$$a_{H}(t)^{\mu} = e(t,0)a(t=0)^{\mu}$$

- $ig \sum_{k=1}^{N} \int_{0}^{t} dt_{1} \,\theta(t-t_{1})e(t,t_{1})R_{k}(t_{1})^{\mu},$
(5.37a)

$$\bar{a}_{H}(t)^{\mu} = \bar{a}(t=0)^{\mu}e^{*}(0,t) + ig \sum_{k=1}^{N} \int_{0}^{t} dt_{1} \,\overline{R}_{k}(t_{1})^{\mu}e^{*}(t_{1},t)\theta(t-t_{1}),$$
(5.37b)

where $e(t,t_1)$ is a certain *c*-number function.¹⁵ When $|a\rangle$ and $|b\rangle$ denote any two vectors in our Fock space and $g^2 = \overline{g}^2/N$

is considered,

$$\lim_{N \to \infty} \langle a | a_H(t)^{\mu} | b \rangle = \langle a | e(t,0) a(t=0)^{\mu} | b \rangle, \qquad (5.38a)$$

$$\lim_{N \to \infty} \langle a | \bar{a}_H(t)^{\mu} | b \rangle = \langle a | \bar{a}(t=0)^{\mu} e^*(0,t) | b \rangle, \qquad (5.38b)$$

which means that $a_{H}^{\mu}(t)$ is weakly equal to $e(t,0)a^{\mu}(t=0)$,

$$a_{H}^{\mu}(t) \stackrel{w}{=} e(t,0)a^{\mu}(t=0), \text{ etc.}$$
 (5.39)

However, this does not mean that $\bar{a}_{H}^{\mu}(t)a_{H}^{\nu}(t)$ is weakly equal to $|e(t,0)|^{2}\bar{a}^{\mu}(0)a^{\nu}(0)$ because the R_{k}^{μ} terms in (5.37) contribute to $\bar{a}_{H}^{\mu}(t)a_{H}^{\nu}(t)$ even at the limit $N \to \infty$.¹⁶ It can be shown that $e(t,t_{1})$ satisfies

$$[i\partial_t - \omega_0]e(t,t_1) + i\frac{\kappa}{\pi}\int_0^t dt_2 \int_\Delta d\Omega \ e^{-i\Omega(t-t_2)}\theta(t_2-t_1)e(t_2,t_1) = 0,$$
(5.40a)

with

$$e(t,t) = 1.$$
 (5.40b)

Since

$$F(t-t_1) = \int_{\Delta} d\Omega \ e^{-i\Omega(t-t_1)}$$
(5.41)

is a function that has a peak at $t = t_1$ and becomes very small for $|t - t_1| \ge 1/\Delta$, $e(t, t^1)$ can be approximately written as

$$e(t,t^{1}) \simeq e(t)e^{-1}(t_{0}),$$
 (5.42)

when κ is small (i.e., Δ is large). Here e(t) is a solution of the equation

$$(i\partial_t - \omega + i\kappa)e(t) = 0 \tag{5.43a}$$

with

$$e(0) = 1.$$
 (5.43b)

Note that ω_0 is equal to ω in this approximation. Thus

$$e(t) = e^{-i\omega t - \kappa t}.$$
(5.44)

A calculation shows that

$$n_{H}(t) \simeq \langle 0|\bar{a}_{H}(t)^{1}a_{H}(t)^{1}|0\rangle$$

= $|e(t)|^{2} \left[n(0) + \frac{\kappa}{\pi} \int_{0}^{t} d\Omega \, n_{R}(\Omega) \left| \int_{0}^{t} dt_{1} \, e^{-i\Omega t_{1}} e^{-1}(t_{1}) \right|^{2} \right].$
(5.45)

Use of (5.45) gives

$$n_{H}(t) \simeq e^{-2\kappa t} \bigg[n(0) + \int d\Omega \,\rho(\Omega) n_{R}(\Omega) |e^{\kappa t - i(\omega - \Omega)t} - 1|^{2} \bigg],$$
(5.46)

where

$$\rho(\Omega) = \frac{1}{\pi} \frac{\kappa}{(\omega - \Omega)^2 + \kappa^2}.$$
 (5.47)

We can rewrite (5.46) as

$$n_{H}(t) \simeq \bar{n}_{R} + e^{-2\kappa t} [n(0) - \bar{n}_{R}] + 2e^{-\kappa t} \left[\bar{n}_{R} e^{-\kappa t} - \int_{\Delta} d\Omega \rho(\Omega) n_{R}(\Omega) \cos(\omega - \Omega) t \right], \quad (5.48)$$

where

$$\bar{n}_R = \int d\Omega \,\rho(\Omega) n_R(\Omega). \tag{5.49}$$

Since $\rho(\Omega)$ is a function whose peak is at $\Omega = \omega$ and whose width is κ , \bar{n}_R is very close to \bar{n} when κ is small. Equation (5.48) indicates that $n_H(t = \infty) = \bar{n}_R$ and $n_H(t = 0) = n(0)$, behaving like n(t) at $t = \infty$ and t = 0. The last bracket term in (5.48) gives the difference between $n_H(t)$ and n(t) when t is finite and nonvanishing. Since $\Delta \gg \kappa$, the presence of $\rho(\Omega)$ makes the last integral term vanish quickly at $t \gg 1/\kappa$. When Δ is large and κ is small, $n_H(t)$ and n(t) behave similarly. At present, we are not sure which one should correspond to the observable value. We might consider also the quantity

$$N_{H}(t) = {}_{H} \langle 0 | \bar{a}_{H}(t)^{1} a_{H}(t)^{1} | 0 \rangle_{H}$$

with

$$|0\rangle_{H} = u(0, -\infty) |0\rangle / [\langle 0|u(\infty, -\infty)|0\rangle]^{1/2}$$

and

$$_{H}\langle 0| = \langle 0|u(\infty,0)/[\langle 0|u(\infty,-\infty)|0\rangle]^{1/2}.$$

We then find

$$N_H(t) = \langle 0 | u(\infty, t) \overline{a}(t)^1 a(t)^1 u(t, -\infty) | 0 \rangle$$

$$\times \langle 0 | u(\infty, -\infty) | 0 \rangle^{-1}.$$
 (5.50)

Since $u(0, -\infty)|0\rangle$ may make $|0\rangle_H$ equilibrium, $N_H(t)$ is the average number in the equilibrium state. This can be confirmed by means of an explicit computation. It is still an important open question as to what really corresponds to each observable quantity. We are planning to continue our study by assuming that n(t) describes the behavior of number in a reasonable approximation.

We discussed the Heisenberg operators in the reservoir model, but we should be careful not to extend the arguments to general nonlinearly interacting systems in which the dynamical maps may be very complicated. The problem of the Heisenberg operator in general models of time-dependent TFD is still unsolved.

VI. SUMMARY

We have presented a general formalism for semifree field theory which includes the time-dependent thermal situation. The formalism expressed in terms of oscillator operators was given in Sec. II. The formalism expressed in terms of field operators was presented in Sec. III. It is due to a complication caused by the time-dependent nature of the thermal behavior that the consideration in Sec. III was limited to the semifree field of type 1. In the Appendix we discuss the semifree field of type 2 in a stationary situation. Using the knowledge of the total Lagrangian in TFD and choosing the semifree field Lagrangian for the unperturbed Lagrangian, we determined the general form of the interaction Hamiltonian, in the interaction representation, for any time-dependent situation in Sec. IV. This is the basis for the Feynman-type perturbative calculation in TFD in time-dependent situations. A set of self-consistent equations determine ω_k , κ_k , and the temporal behavior of the average number density $n_k(t)$. An example of the derivation of the self-consistent equations in a time-dependent situation was studied in Sec. V. A remarkable point in this consideration is the explicit expression of the *on-shell renormalization condition* in the time-dependent situation which leads to the self-consistent equations. The latter equations were explicitly solved, and κ and n(t) were determined.

Since $n(t) = \langle 0|\bar{a}(t)^1 a(t)^1|0\rangle$, the explicit t dependence of n indicates that the time-translational symmetry (based on $\hat{H}|0\rangle = 0$) is spontaneously broken. This is a result of the communication between the nontilde quanta with the background tilde field which is a manifestation of the effects of the thermally excited quanta which have infinite degrees of freedom. This communication seems to be playing a role similar to the coarse graining. The negative sign of the energy of the tilde quanta encourage the thermal instability, leading to the dissipation. Since there are an infinite number of dissipative channels, each of which participates with infinitesimal probability, the instability takes the form of dissipation. This becomes clear through the example in Sec. V when the (1/N)-power expansion method (N: number of degrees of freedom) with the limit $N \to \infty$.

The spontaneous breakdown of the time-translation symmetry suggests that, at any finite time, there are no eigenvectors of the basic total Hamiltonian $\hat{H} = H - \tilde{H}$ in our representation space, which is the Fock space of $\xi(0)^{\mu}$ and $\bar{\xi}(0)^{\mu}$. The diagonal property of the time-translation matrix $E_0(t)$ in (2.10) implies that the realization space remains unchanged during the time evolution. However, the dynamical map of \hat{H} expressed in terms of quasiparticle operators depends on time explicitly. In this sense, the realization of \hat{H} depends on time. As a matter of fact, when a system reaches an equilibrium state, the thermal vacuum may become an eigenvector of \hat{H} at $t = \infty$. Then the time-translation symmetry may be recovered at $t \to \infty$, although κ does not vanish at $t = \infty$. (Remember that κ is independent of time in Sec. V.)

Here a comment about the quasiparticle number may be in order. Although $n(t) = \langle 0 | \bar{a}(t)^{1} a(t)^{1} | 0 \rangle$ is the number of the unperturbed particles, *this includes all of the loop corrections*, because the unperturbed particle is the quasiparticle that is *fully normalized*. Each loop correction in the selfenergy diagram modifies the parameters such as ω , κ , and n(t) in the unperturbed Hamiltonian through the self-consistent renormalization condition.

Although the consideration in Sec. V is the first explicit treatment of the on-shell time-dependent renormalization condition, this consideration is still simple in the sense that ω and κ are independent of time. When we study a transition from temperature T to T', and when the quantum energy depends on temperature, it is obvious that the energy ω changes in time. A study of such a case requires the most complete form of the on-shell time-dependent renormalization condition. We are planning an application of the formalism in this paper to a case in which ω and κ depend on time. A strong merit of TFD presented in this paper is that it can be applied to any thermal situation including an isolated dynamical system. Thus TFD is suitable for treating the thermal behavior of the Universe.

ACKNOWLEDGMENTS

This work was supported by the Natural Sciences and Engineering Research Council, Canada, and the Dean of Science, Faculty of Science, the University of Alberta, Edmonton, Alberta, Canada. One of the authors (I.H.) would like to thank the Canadian Commonwealth Scholarship and Fellowship Committee for scholarship support.

APPENDIX: THE SEMIFREE FIELDS OF TYPE 2 IN THE TIME-INDEPENDENT SITUATION

In Sec. II we considered the wave functions u(x,k) with the positive frequency. The equation for u(x,k) was summarized in (3.6),

$$iD^{+}(t,\mathbf{k})^{\mu\nu}u(x,\mathbf{k})^{\nu}=0, \qquad (A1a)$$

$$i\overline{u}(\mathbf{x},\mathbf{k})^{\nu}\overleftarrow{D}^{+}(t,\mathbf{k})^{\nu\mu}=0.$$
 (A1b)

The thermal covariant derivatives were defined in (2.31),

$$D^{+}(t,\mathbf{k}) = B^{-1}(t,\mathbf{k})E(t,\mathbf{k})\partial_{t}E^{-1}(t,\mathbf{k})B(t,\mathbf{k}), \qquad (A2a)$$

$$\overline{D}^{+}(t,\mathbf{k}) = B^{-1}(t,\mathbf{k})E(t,\mathbf{k})\overline{\partial}_{t}E^{-1}(t,\mathbf{k})B(t,\mathbf{k}).$$
(A2b)

Recall the form of $E(t,\mathbf{k})$ in (2.27).

In a similar manner, we can write the thermal doublet wave function $v(t,\mathbf{k})$ with the negative frequency

$$iD^{-}(t,\mathbf{k})^{\mu\nu}v(t,\mathbf{k})^{\nu}=0, \qquad (A3a)$$

$$i\overline{v}(t,\mathbf{k})^{\nu}D^{-}(t,\mathbf{k})^{\nu\mu}=0, \qquad (A3b)$$

where

$$D^{-}(t,\mathbf{k}) = B^{-1}(t,\mathbf{k})E^{-1}(t,\mathbf{k})\partial_{t}E(t,\mathbf{k})B(t,\mathbf{k}), \qquad (A4a)$$

$$\hat{D}^{-}(t,\mathbf{k}) = B^{-1}(t,\mathbf{k})E^{-1}(t,\mathbf{k})\partial_t E(t,\mathbf{k})B(t,\mathbf{k}).$$
(A4b)

A difficulty in constructing a semifree field by making a linear superposition of $u(x,\mathbf{k})$ and $v(x,\mathbf{k})$ is due to the fact that $D^+(t,\mathbf{k})$ and $D^-(t,\mathbf{k})$ do not commute with each other when we consider a time-dependent situation. Therefore, in this Appendix, we restrict our considerations to a time-independent situation only. Then $B(t,\mathbf{k})$, $\omega(t,\mathbf{k})$, and $k(t,\mathbf{k})$ become independent of time. These quantities will be written simply as $B(\mathbf{k})$, $\omega(\mathbf{k})$, and $\kappa(\mathbf{k})$. Then according to (2.27), $E(t,\mathbf{k})$ takes the following simple form:

$$E(t,\mathbf{k}) = \exp[-i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})\tau_3\}t].$$
(A5)

According to (2.33), we have

$$D^{+}(t,\mathbf{k}) = \partial_{t} + i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\}, \qquad (A6a)$$

$$\overline{D}^{+}(t,\mathbf{k}) = \overline{\partial}_{t} - i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\}.$$
(A6b)

The matrix $A(\mathbf{k})$ was given in (2.38). Similarly, we obtain

$$D^{-}(t,\mathbf{k}) = \partial_{t} - i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\}, \qquad (A7a)$$

$$\overleftarrow{D}^{-}(t,\mathbf{k}) = \overleftarrow{\partial}_{t} + i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\}.$$
(A7b)

We also construct the wave functions in such a manner

$$u(x,\mathbf{k}) = u(t,\mathbf{k})e^{i\mathbf{k}\cdot\mathbf{x}},$$
 (A8a)

$$\bar{u}(x,\mathbf{k}) = \bar{u}(t,\mathbf{k})e^{-i\mathbf{k}\cdot\mathbf{x}},\tag{A8b}$$

$$v(\mathbf{x},\mathbf{k}) = v(t,\mathbf{k})e^{-i\mathbf{k}\cdot\mathbf{x}},$$
 (A9a)

$$\overline{v}(x,\mathbf{k}) = \overline{v}(t,\mathbf{k})e^{i\mathbf{k}\cdot\mathbf{x}}.$$
(A9b)

Then (A1), (A3), (A6), and (A7) lead to the orthogonality condition at any time t,

$$\int d^{3}x \,\overline{u}(x,\mathbf{k}) \,\Gamma v(x,\mathbf{l}) = 0, \qquad (A10a)$$

$$\int d^{3}x \,\overline{v}(x,\mathbf{k}) \Gamma u(x,\mathbf{l}) = 0, \qquad (A10b)$$

where

$$\Gamma = i \overleftrightarrow{\partial_t} = i(\partial_t - \overleftrightarrow{\partial_t}).$$
 (A11)

Therefore we define the inner product of two wave functions by

$$\int d^{3}x \,\overline{g}(x,) \Gamma f(x). \tag{A12}$$

It is obvious from (A8) and (A9) that we can choose the wave functions to satisfy

$$\int d^{3}x \, \overline{u}(x,\mathbf{k}) \Gamma u(x,\mathbf{l}) = \rho_{u} \delta(\mathbf{k} - \mathbf{l}), \qquad (A13a)$$

$$\int d^{3}x \,\overline{v}(x,\mathbf{k}) \Gamma v(x,\mathbf{l}) = \rho_{v} \delta(\mathbf{k}-\mathbf{l}).$$
 (A13b)

Here $\rho_u = 1$ or -1 and $\rho_v = 1$ or -1. Thus $\{u(x,\mathbf{k}), v(x,\mathbf{k})\}$ forms an orthonormalized complete set.

In order to relate u to v, we note that

$$D^{+}(t,\mathbf{k})^{T} = B^{T}(\mathbf{k})E^{-1}(t,\mathbf{k})\overline{\partial}_{t}E(t,\mathbf{k})B^{-1T}(\mathbf{k}), \quad (A14)$$
$$= B^{T}(\mathbf{k})B(\mathbf{k})\overline{D}^{-}(t,\mathbf{k})B^{-1}(\mathbf{k})B^{-1T}(\mathbf{k}),$$

(A15) $\overleftarrow{D}^{+}(t,\mathbf{k})^{T} = B^{T}(\mathbf{k})E^{-1}(t,\mathbf{k})\partial_{t}E(t,\mathbf{k})B^{-1T}(\mathbf{k}), \quad (A14')$

$$= B^{T}(\mathbf{k})B(\mathbf{k})D^{-}(t,\mathbf{k})B^{-1}(\mathbf{k})B^{-1T}(\mathbf{k}),$$
(A15')

which follow from (A2) and (A4). Here T means the transposition. Thus (A1) leads to (A3) when we choose

$$\overline{v}(t,\mathbf{k}) = \overline{v}_0(\mathbf{k})u^T(t,\mathbf{k})B^T(\mathbf{k})B(\mathbf{k}), \qquad (A16a)$$

$$v(t,\mathbf{k}) = B^{-1}(\mathbf{k})B^{-1T}(\mathbf{k})\overline{u}^{T}(t,\mathbf{k})v_{0}(\overline{k}), \qquad (A16b)$$

where v_0 and \overline{v}_0 satisfy

$$\overline{v}_0 v_0 = 1 \tag{A16c}$$

and are arbitrary otherwise.

According to (3.4) we have

$$u(\mathbf{x},\mathbf{k}) = U(\mathbf{x},\mathbf{k})B^{-1}(\mathbf{k})E(t,\mathbf{k})B(\mathbf{k}), \qquad (A17a)$$

$$\overline{u}(\mathbf{x},\mathbf{k}) = B^{-1}(\mathbf{k})E^{-1}(t,\mathbf{k})B(\mathbf{k})\overline{U}(\mathbf{x},\mathbf{k}).$$
(A17b)

Since the normalization condition in (A13a) differs from the one in (3.7) by the presence of Γ , we should modify the definition of $U(\mathbf{x},\mathbf{k})$ and $\overline{U}(\mathbf{x},\mathbf{k})$. Since $A(\mathbf{k})$ commutes with $D^{\pm}(t,\mathbf{k})$ according to (A6) and (A7) we choose

$$U(\mathbf{x},\mathbf{k}) = [1/(2\pi)^{3/2}]F[A(\mathbf{k})]e^{i\mathbf{k}\cdot\mathbf{x}}, \qquad (A18a)$$

$$\overline{U}(\mathbf{x},\mathbf{k}) = [1/(2\pi)^{3/2}]F[A(\mathbf{k})]e^{-i\mathbf{k}\cdot\mathbf{x}}, \qquad (A18b)$$

where $F[A(\mathbf{k})]$ is a function of the matrix $A(\mathbf{k})$. Equations (2.38) and (A5) show that $B^{-1}(\mathbf{k})E(t,\mathbf{k})B(\mathbf{k})$ is also a function of $A(\mathbf{k})$, implying that $A(\mathbf{k})$ commutes with $B^{-1}(\mathbf{k})E(t,\mathbf{k})B(\mathbf{k})$. Now (A16) lead to

$$v(\mathbf{x},\mathbf{k}) = [1/(2\pi)^{3/2}]B^{-1}(\mathbf{k})E^{-1}(t,\mathbf{k})B^{-1T}(\mathbf{k})$$
$$\times F[A^{T}(\mathbf{k})]v_{0}(\mathbf{k})e^{-i\mathbf{k}\cdot\mathbf{x}}, \qquad (A19a)$$
$$\bar{v}(\mathbf{x},\mathbf{k}) = [1/(2\pi)^{3/2}]\bar{v}_{0}(\mathbf{k})F[A^{T}(\mathbf{k})]$$

$$\times B^{T}(\mathbf{k})E(t,\mathbf{k})B(\mathbf{k})e^{i\mathbf{k}\cdot\mathbf{x}}.$$
 (A19b)

Feeding (A17) with (A18) into (A13a), we obtain

$$2F^{2}[A(\mathbf{k})]\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\} = \rho_{u}.$$
 (A20)

When we choose

$$\rho_u = 1, \tag{A21}$$

we have

$$F[A(\mathbf{k})] = (1/\sqrt{2}) \{ \omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k}) \}^{-1/2}.$$
 (A22)

On the other hand, (A13b) together with (A19) and (A16c) gives

$$\rho_{v} = -2F^{2}[A^{T}(\mathbf{k})]\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A^{T}(\mathbf{k})\} = -1.$$
(A23)

We now construct the field

$$\psi(\mathbf{x})^{\mu} = \int d^{3}k \left[u(\mathbf{x},\mathbf{k})^{\mu\nu}a(\mathbf{k})^{\nu} + v(\mathbf{x},\mathbf{k})^{\mu\nu}\overline{b}^{T}(\mathbf{k})^{\nu} \right],$$
(A24a)

$$\overline{\psi}(x)^{\mu} = \int d^{3}k \left[\overline{a}(\mathbf{k})^{\nu}\overline{u}(x,\mathbf{k})^{\nu\mu} + b^{T}(\mathbf{k})^{\nu}\overline{v}(x,\mathbf{k})^{\nu\mu}\right],$$
(A24b)

where we introduced two sets of thermal doublet oscillators, $\{a(\mathbf{k}), \overline{a}(\mathbf{k})\}\$ and $\{b(\mathbf{k}), \overline{b}(\mathbf{k})\}\$. Note that the positive- and negative-frequency parts of the fields ψ^{μ} and $\overline{\psi}^{\mu}$ are transformed separately under the tilde operation (2.4) if v_0 and \overline{v}_0 in (A16) are kept arbitrary. Therefore we choose v_0 and \overline{v}_0 as

$$v_0(\mathbf{k}) = \bar{v}_0(\mathbf{k}) = B^T(\mathbf{k})\tau_3 B(\mathbf{k}), \qquad (A25)$$

so that ψ^{μ} and $\overline{\psi}^{\mu}$ have the definite transformation property of the tilde conjugation:

$$[\psi(x)^1]^{\sim} = -\bar{\psi}(x)^2,$$
 (A26a)

$$[\psi(x)^2]^{\sim} = \bar{\psi}(x)^1,$$
 (A26b)

$$[\bar{\psi}(x)^1]^{\sim} = \psi(x)^2,$$
 (A26c)

$$[\bar{\psi}(x)^2]^{\sim} = -\psi(x)^1.$$
 (A26d)

The semifree field equations are

$$D^{+}(t, -i\nabla)D^{-}(t, -i\nabla)\psi(x) = 0, \qquad (A27a)$$

$$\overline{\psi}(x)\overline{D}^{-}(t, i\overline{\nabla})\overline{D}^{+}(t, i\overline{\nabla}) = 0, \qquad (A27b)$$

$$\left[\partial_t^2 + \{\omega(-i\nabla) - i\kappa(-i\nabla)A(-i\nabla)\}^2\right]^{\mu\nu}\psi(x)^{\nu} = 0,$$
(A28a)

$$\bar{\psi}(x)^{\nu} \left[\overleftarrow{\partial}_{i}^{2} + \{ \omega(i\nabla) - i\kappa(i\nabla)A(i\nabla) \}^{2} \right]^{\nu\mu} = 0.$$
 (A28b)

These equations of motion can be derived from the Lagrangian density

$$\hat{L}^{0} = -\bar{\psi}(x)^{\mu} [\partial_{t}^{2} + \{\omega(-i\nabla) - i\kappa(-i\nabla)A(-i\nabla)\}^{2}]^{\mu\nu}\psi(x)^{\nu}.$$
(A29)

This gives the canonical conjugate momentum field of ψ as

$$\bar{\pi} = \frac{\partial}{\partial t} \,\bar{\psi}.\tag{A30}$$

We can easily derive the following sum rules:

$$i\Delta^+ (\mathbf{x} - \mathbf{x}')^{\mu\nu} = \int d^3k \ u(\mathbf{x}, \mathbf{k})^{\mu\lambda} \overline{u}(\mathbf{x}', \mathbf{k})^{\lambda\nu}, \qquad (A31a)$$

$$i\Delta^{-}(x-x')^{\mu\nu} = -\int d^{3}k \, v(x,\mathbf{k})^{\mu\lambda} \overline{v}(x',\mathbf{k})^{\lambda\nu}, \quad (A31b)$$

where

Δ

$$= \mp i \int \frac{d^{3}k}{(2\pi)^{3}} \times \frac{\exp[\mp i\{\omega(\mathbf{k}) - i\kappa(\mathbf{k})A(\mathbf{k})\}t]\exp[\pm i\mathbf{k}\cdot\mathbf{x}]}{2\{\omega(\mathbf{k}) - \kappa(\mathbf{k})A(\mathbf{k})\}}.$$
(A32)

The functions Δ^{\pm} have the following properties:

$$\Delta^+(-x) = -\Delta^-(x), \qquad (A33a)$$

$$\delta(t) \frac{\partial}{\partial t} \Delta^{\pm}(\mathbf{x}) = -\frac{1}{2} \delta(\mathbf{x}) \delta(t), \qquad (A33b)$$

$$\left[\partial_{i}^{2} + \left\{\omega(-i\nabla) - i\kappa(-i\nabla)A(-i\nabla)\right\}^{2}\right]\Delta^{\pm}(x) = 0.$$
(A33c)

The sum rules (A31) lead to the commutation relation

$$[\psi(x)^{\mu}, \overline{\psi}(x')^{\nu}] = i\Delta(x - x')^{\mu\nu},$$
(A34)

where

$$\Delta(x) \equiv \Delta^+(x) + \Delta^-(x). \tag{A35}$$

We call (A31) the canonical sum rules, since (A33b), (A34), (A30), and (A24) give the equal-time canonical commutation relation

$$[\psi(x)^{\mu},\bar{\pi}(x')^{\nu}]\delta(t-t') = i\delta^{\mu\nu}\delta^{(4)}(x-x').$$
(A36)

Thus we have explicitly constructed the canonical formalism for the semifree field of type 2 in the time-independent thermal situation. Note that the parameter b_R was never fixed in this formulation. This freedom includes the freedom in choice of the α parameter which was discussed in several previous papers.^{4,7} We have not succeeded in extend-

ing this formulation of type 2 semifree fields to a time-dependent situation.

¹T. Arimitsu and H. Umezawa, Prog. Theor. Phys. 74, 429 (1985); 77, 53 (1987); J. Phys. Soc. Jpn. 55, 1475 (1986); H. Umezawa and T. Arimitsu, in *Path Integrals from meV to MeV*, edited by M. C. Gutzwiller, A. Inomata, J. R. Klauder, and L. Streit (World Scientific, Singapore, 1986), p. 217.

- ²H. Umezawa and T. Arimitsu, Prog. Theor. Phys. Suppl. **86**, 243 (1986); H. Umezawa, *Symmetry in Science II*, edited by B. Gruber and R. Lenczewski (Plenum, New York, 1986), p. 517; T. Arimitsu, Y. Sudo, and H. Umezawa, "Dynamical rearrangement of the thermal vacuum in TFD," to be published in Physica A.
- ³T. Arimitsu, J. Pradko, and H. Umezawa, Physica A 135, 487 (1986).
- ⁴T. Arimitsu, M. Guida, and H. Umezawa, Europhys. Lett. **3**, 277 (1987); "Dissipative quantum field theory—thermo field dynamics—," to be published in Physica A; T. Arimitsu and H. Umezawa, *Advances on Phase Transitions and Disordered Phenomena*, edited by G. Busiello *et al.* (World Scientific, Singapore, 1987), p. 483.
- ⁵H. Umezawa and T. Arimitsu, *Proceedings of the 2nd International Symposium on Foundations of Quantum Mechanics*, edited by M. Namiki *et al.* (Physical Society of Japan, Tokyo, 1987), p. 79; T. Arimitsu, H. Umezawa, Y. Yamanaka, and N. J. Papastamatiou, "Spontaneous creation of dissipation in thermo field dynamics and its examples," to be published in Physica A.
- ⁶H. Umezawa, Y. Yamanaka, I. Hardman, and T. Arimitsu, *Wandering in the Fields*, edited by K. Kawarabayashi and A. Ukawa (World Scientific, Singapore, 1987).
- ⁷T. Arimitsu, H. Umezawa, and Y. Yamanaka, J. Math. Phys. 28, 2741 (1986).
- ⁸H. Umezawa, H. Matsumoto, and M. Tachiki, *Thermo Field Dynamics and Condensed States* (North-Holland, Amsterdam, 1982).
- ⁹N. P. Landsman and Ch. G. van Weert, Phys. Rep. 145, 141 (1987), and references therein.
- ¹⁰H. Araki and E. J. Woods, J. Math. Phys. 4, 637 (1963); H. Araki and W. Wyss, Helv. Phys. Acta 37, 136 (1964); R. Haag, N. M. Hugenbold, and M. Winnihk, Commun. Math. Phys. 5, 215 (1967).
- ¹¹J. Schwinger, J. Math. Phys. 2, 407 (1961); L. V. Keldysh, Sov. Phys. JETP 20, 1018 (1965); R. A. Craig, J. Math. Phys. 9, 605 (1968); R. Mills, *Propagators for Many-Particle Systems* (Gordon and Breach, New York, 1969).
- ¹²U. Fano, Rev. Mod. Phys. 29, 74 (1957); *Lectures on the Many-Body Problem*, edited by E. R. Caianiello (Academic, New York, 1964), p. 217;
 J. A. Crawford, Nuovo Cimento 10, 698 (1958); M. Schmutz, Z. Phys. B 30, 97 (1978).
- ¹³I. Ojima, Ann. Phys. (NY) 137, 1 (1981).
- ¹⁴H. Matsumoto, Y. Nakano, H. Umezawa, F. Mancini, and M. Marinaro, Prog. Theor. Phys. **70**, 599 (1983); H. Matsumoto, Y. Nakano, and H. Umezawa, J. Math. Phys. **25**, 3076 (1984).
- ¹⁵We are indebted to Dr. K. Nakamura for deriving the expressions in (5.37).
- ¹⁶Note that we can preserve the g terms in (5.37), treating k as a continuous variable. Then $g \sum_{k=1}^{N} \cdots R_k(t)^{\mu}$ in (5.37a), e.g., now reads as $(\overline{g}/N\Delta) \int_{\Delta} d\Omega \cdots R(t;\Omega)^{\mu}$, where $R_k(t)^{\mu}$ is replaced by $(1/\sqrt{N}) R(t;\Omega)^{\mu}$.

There is no isolated *p-p* wave

P.F. Yip

Department of Mathematics, University of California, San Diego, La Jolla, California 92093

(Received 20 January 1987; accepted for publication 24 June 1987)

A theorem is presented that basically states that there are no nontrivial well-behaved, spatially asymptotically flat space-times which are *p*-*p* waves at infinity.

I. INTRODUCTION

Space-times with a constant null vector (p-p waves) have been studied extensively in the literature.¹ Almost always, these space-times are interpreted physically as infinite plane gravitational waves. There are, however, attempts to construct p-p waves to represent, instead of waves, particles moving at the speed of light.^{2,3} But any notion of isolated particles would naturally restrict the space-times under consideration to the class of spatially asymptotically flat spacetimes. Of all the solutions proposed, only one³ is clearly spatially asymptotically flat (this solution is also vacuum and axisymmetric). Perhaps, the most intriguing aspect of the solution is that it has a vanishing ADM momentum, and yet it is not flat. On closer examination, we see that it gets around the positive mass theorem^{4,5} because it does not have a complete spacelike surface. A natural question thus arises: can one construct a p-p wave that is well behaved (spatially complete) and spatially asymptotically flat? In this paper, we present a theorem which basically states that a spatially complete space-time satisfying the dominant energy condition and that is a p-p wave at least at infinity cannot be asymptotically flat unless it is flat. Thus we see that no wellbehaved space-times based on the p-p waves can be constructed to represent an isolated particle.

In Sec. II we will briefly look at the solution proposed in Ref. 3. Section III contains the precise statement and the proof of the theorem mentioned above.

II. A METRICAL SOLITON

The space-time (M,g_{ab}) of interest reported in Ref. 3, representing a "metrical soliton," has the metric in the following form:

$$ds^{2} = \frac{\rho^{2}}{\rho^{2} + \chi} d\rho^{2} + (\rho^{2} + \chi) d\psi^{2} + \left[1 - \frac{(\partial \chi / \partial z)^{2}}{4(\rho^{2} + \chi)}\right] dz^{2} - \left[1 + \frac{(\partial \chi / \partial z)^{2}}{4(\rho^{2} + \chi)}\right] (dx^{4})^{2} - \frac{1}{2} \frac{\alpha (\partial \chi / \partial z)^{2}}{(\rho^{2} + \chi)} dz dx^{4},$$
(1)

where (ρ, ψ, z, x^4) are the standard cylindrical coordinates. Here χ is a compact supported function of $(z + \alpha t)$, with $\alpha = \pm 1$. The basic facts concerning this metric (axisymmetric, *p*-*p* wave) can be easily verified. Of special interest is that the space-time admits a spatially asymptotically flat hypersurface Σ . In fact, the desired surface is given by $x^4 = 0$.

To see this more clearly, introduce on $\Sigma = \{x^4 = 0\}$ the usual rectangular coordinates (x,y,z) with $x = \rho \cos \psi$,

 $y = \rho \sin \psi$. Then on Σ the components of the first and second fundamental forms, h_{ij} and π_{ij} , respectively, in the rectangular coordinates, have the following asymptotic properties:

$$h_{ij} = \delta_{ij} + O(1/r^2), \quad \pi_{ij} = O(1/r^3).$$

Thus Σ is asymptotically flat. Moreover, since the falloffs are one power of 1/r faster than the standard ones, the ADM four-momentum of M on Σ has to vanish.

To understand how the space-time can be reconciled with the positive mass theorem, it suffices to examine the induced metric h_{ij} on Σ . It has the form, in the (ρ, ψ, z) coordinates,

spatial metric =
$$\left(\frac{\rho^2}{\rho^2 + \chi}\right) d\rho^2 + (\rho^2 + \chi) d\psi^2$$

+ $\left[1 - \frac{(\partial \chi / \partial z)^2}{4(\rho^2 + \chi)}\right] dz^2.$

To ensure that Σ is spacelike, the conditions $\chi \ge 0$ and $4(\rho^2 + \chi) \ge (\partial x/\partial z)^2$ must hold. Even with these provisions, the spatial metric is degenerate at the line $\rho = 0$, unless $\chi \equiv 0$, in which case the space-time metric is flat. To remove this irregularity, one can, for example, cut out the line $\rho = 0$ from Σ . But then Σ will be incomplete. In conclusion, it is obvious that the only well-behaved (nondegenerate and complete) space-time metric of the form (1) is flat.

III. THERE IS NO ISOLATED p-p WAVE

Of course it would be tempting to patch up the metric (1) so that it becomes complete and still retains the *p*-*p* wave nature of the solution. However, any attempt in this direction, without violating the dominant energy condition, would be unsuccessful. The following theorem elucidates this situation.

Theorem: Let (M,g_{ab}) be a spatially asymptotically flat space-time with Σ as an asymptotically flat spacelike complete surface. Suppose (a) there exists a covariantly constant null vector k^a in a neighborhood of $(\Sigma - K)$, where K is a compact set of Σ , and (b) the dominant energy condition holds on Σ ; then (M,g_{ab}) is flat along Σ .

The proof of the theorem follows easily from the following two lemmas. The conditions in the theorem are assumed in the following discussion.

Lemma: There exists a covariant constant spinor ξ^{A} in a neighborhood of $(\Sigma - B)$, where B is a compact set of Σ .

Proof: Consider the covariant null vector k^a , in the

theorem. Anticommuting second derivatives of k^a leads to $R_{abcd} k^d = 0$, and by contracting, $R_{ad} k^d = 0$. The dominant energy condition further implies that $R_{ab} = fk_a k_b$ for some f. Finally, it is easy to conclude that $C_{abcd} k^d = 0$.

Now, let ξ^{A} be the spinor such that $\xi^{A}\xi^{A'} = k^{a}$. Here $\nabla_{a}k_{b} = 0$ implies (in a neighborhood of $\Sigma - K$) that

$$\nabla_a \xi_B = i v_a \xi_B,$$

where v_a is a real vector field. Anticommuting second derivatives gives

 $\nabla_{[a}\nabla_{b]}\xi_{C}=i\nabla_{[a}v_{b]}\xi_{C}.$

The left-hand side of the above equation actually vanishes. To see this, consider the standard spinor representation of the curvature tensors (see, e.g., Ref. 6),

$$C_{abcd} = \psi_{ABCD} \epsilon_{A'B'} \epsilon_{C'D'} + \text{c.c.},$$

$$R_{ab} = -2\Phi_{A'B'AB} + 6\Lambda \epsilon_{A'B'} \epsilon_{AB},$$

and

 $R=-24\Lambda.$

Then

$$\nabla_{[a} \nabla_{b} \xi_{C} = \psi^{D}_{ABC} \epsilon_{A'B'} \xi_{D} + \Phi^{D}_{A'B'C} \epsilon_{AB} \xi_{D} + \Lambda(\epsilon_{AC} \epsilon^{D}_{B} + \epsilon_{BC} \epsilon^{D}_{A}) \epsilon_{A'D'} \xi_{D}.$$

Since $C_{abcd}k^a = 0$ and $R_{ab} = fk_ak_b$, the right-hand side of the above equation vanishes. Thus

 $\nabla_{[a} v_{b]} = 0.$

Now take a compact set $B(\supseteq K)$ of Σ such that $\Sigma - B$ is simply connected (for instance, take B to be the coordinate R ball, B_R for R sufficiently large). Therefore, $v_b = \nabla_b g$, for some g in $\Sigma - B$. Finally, the following rephasing of ξ^A : $\xi^A \to \exp(-ig)\xi^A$, yields $\nabla_a \xi_B = 0$ in $\Sigma - B$. Q.E.D. The following definition definition (can also Pafe 5.7 and 8) is

The following definition (see also Refs. 5, 7, and 8) is relevant for the second lemma.

Definition: A spinor field ξ^A on $\Sigma - B$, where B is a compact set of Σ , is called asymptotically constant if

$$\xi^{A}_{r\to\infty} \xi^{A}_{0} + O(1/r),$$

where r is the asymptotically flat radial coordinate of Σ , and ξ_0^A is a constant spinor as defined by a tetrad associated with the asymptotically flat coordinates on Σ .

Lemma: A covariantly constant spinor is asymptotically constant.

Proof: Introduce a norm (see, e.g., Ref. 9) on the spinors by $\langle \xi^A, \eta_A \rangle = \xi^A \eta^{A'} t_{AA'}$, where $t_{AA'}$ is the unit normal to Σ . Choose two constant spinors ξ^A and η^A at infinity such that $\xi^A_0 \eta_A = 1$, and they are orthonormal with respect to \langle , \rangle . Now

$$\xi^{A} = \alpha \xi^{A} + \beta \eta^{A}$$
, for some α and β .

For ξ^A to be asymptotically constant, α and β should have the correct asymptotic properties, namely, $\alpha \rightarrow_{r \rightarrow \infty} \alpha_0 + O(1/r)$ and $\beta \rightarrow_{r \rightarrow \infty} \beta_0 + O(1/r)$, with α_0 and β_0 constant. We will first establish that the norm

$$|\xi^A| = \langle \xi^A, \xi^A \rangle^{1/2} = (\bar{\alpha}\alpha + \bar{\beta}\beta)^{1/2}$$

has the right asymptotic properties.

Taking a derivative and noting that $\nabla_a \xi^A = 0$, give $\nabla_b |\xi^A|^2 = \xi^A \xi^A \nabla_b t_a$.

Projecting by h_{h}^{c} , we obtain

$$D_b |\xi^A|^2 = s^a \pi_{ba},$$

with $s^a = h_b^a k^b$. Now let v^b be any unit vector of Σ (with respect to h_{ab}); then

$$\begin{split} v^{b}D_{b} |\xi^{A}|^{2} &= s^{a}v^{b}\pi_{ba}, \\ |v^{b}D_{b}|\xi^{A}| | &= |s_{c}h^{cd}v^{b}\pi_{bd}| \\ &\leq |s^{a}| |v^{b}\pi_{b}{}^{d}|, \end{split}$$

where $|m^a|$, for instance, is $(-h_{ab}m^am^b)^{1/2}$ for any m^a on Σ . [Note that since the space-time metric has signature (+, -, -, -) in accordance with the spinor convention h_{ab} , the induced metric on Σ is negative definite; hence the negative sign in the above inner product for vectors on Σ .] Now

$$k^{a}k_{a} = \xi^{A}\xi^{A'}g_{ab}\xi^{B}\xi^{B'}$$

= $\xi^{AA'}(h_{ab} + t_{a}t_{b})\xi^{A}\xi^{B'}$
= $-|s^{a}|^{2} + |\xi^{A}|^{4}$.

Since k^a is null,

 $|\mathbf{x}^a| = |\boldsymbol{\xi}^A|^2.$

Thus

$$|v^b D_b|\xi^A|^2| \leq |\xi^A|^2|v\pi^a|.$$

By the assumption of asymptotic flatness, $|v^a \pi_a{}^b| = O(1/r^2)$. Therefore, there exists positive constants C and r_0 such that

$$|v^b D_b \ln|\xi^A| | \leq C/r^2 \quad \text{for } r \geq r_0.$$
⁽²⁾

Now, suppose $v^b = r^a$, the vector $(\partial /\partial r)^a$. Though r^a is not strictly a unit vector, its norm differs from unity only by terms of order (1/r) and higher. Therefore we can still use the above inequality, which yields

$$\left|\frac{\partial}{\partial r}\ln|\xi^A|\right| \leqslant \frac{C}{r^2}.$$

Upon integration, the above gives

$$C/r - C/r_0 \leq \ln |\xi^A(r,\theta,\phi)| - \ln |\xi^A(r_0,\theta,\phi)|$$
$$\leq C/r_0 - C/r.$$

It is then clear that $\lim_{r\to\infty} \ln|\xi^A(r,\theta,\phi)|$ exists. Let

 $\lim_{r\to\infty} \ln|\xi^{A}(r,\theta,\phi)| = \ln N(\theta,\phi),$

which implies $\lim_{r\to\infty} |\xi^A(r,\theta,\phi)| = N(\theta,\phi)$. The above inequality further asserts

$$|\xi^{A}(r,\theta,\phi)| \rightarrow N(\theta,\phi) + O(1/r).$$

The following argument establishes that $N(\theta, \phi)$ is, in fact, independent of θ and ϕ . Define M(r) > 0 such that

$$\ln M(r) = \frac{1}{4\pi} \int \ln |\xi^{A}(r,\theta,\phi)| \sin \theta \, d\theta \, d\phi.$$

Now,

 $\max \left| \ln M(r) - \ln \left| \xi^{A}(r, \theta, \phi) \right| \right|$

 $\langle C'r \max | v^a D_a \ln | \xi^A(r,\theta,\phi) | |,$

where C' is a positive constant and v^a is any unit vector on the sphere S_r ; and

 $\max |\ln M(r) - \ln |\xi^{A}(r,\theta,\phi)| | \leq C''/r,$

where C'' is a positive constant. Consequently for sufficiently large r,

 $\max |\ln M(r) - \ln N(\theta,\phi)| \leq C''/r,$

C''' is a positive constant.

Now it is obvious that $N(\theta,\phi)$ is independent of (θ,ϕ) up to order 1/r. Finally we conclude that

$$|\xi^{A}(r,\theta,\phi)| \xrightarrow[r\to\infty]{} N + O(1/r),$$

where N is a constant.

Similar considerations lead to the desired results $\alpha \rightarrow_{r \rightarrow \infty} \alpha_0 + O(1/r)$, and $\beta \rightarrow_{r \rightarrow \infty} \beta_0 + O(1/r)$. Q.E.D.

The previous two lemmas can now be employed to establish the theorem. Consider the following integral:

$$S = \int_{S^2} ds^a t_b \eta^B \nabla_a \eta^B$$

with ds^a the surface element of S^2 , a sphere at infinity, t^a the unit normal to Σ , and η^A a spinor. If η^A is an asymptotically constant spinor satisfying the Witten equation, $D_{AA'} \eta^{A'} = 0$, where $D_a = h_{ab} \nabla_b$, then^{5,7}

$$S = \kappa P^{a}_{ADM} \eta_{A} \eta_{A'},$$

 P_{ADM}^{a} is the ADM four-momentum of (M,g_{ab}) along Σ, η^{A}

is the constant spinor to which η_A approaches, and κ is a constant. But by the previous lemmas, there exists outside a compact set of Σ , a covariantly constant spinor ξ^A (thus satisfying the Witten equation), which is also an asymptotically constant spinor. Here S must vanish for this spinor ξ^A , and thus

$$P^a_{ADM} \xi_0^A \xi_0^{A'} = 0.$$

But since P_{ADM}^{a} is strictly timelike, ^{10,11} it therefore vanishes. In conclusion, by the positive mass theorem, (M,g_{ab}) is flat along Σ . Q.E.D.

IV. CONCLUSION

Thus we have shown that a truly well-behaved isolated object which is a p-p wave at infinity cannot exist. We would like to point out in closing that if one were to relax any of the conditions in the theorem, one could easily obtain p-p waves which are spatially asymptotically flat.³ Thus the conditions in the theorem are the weakest one can assume for the theorem to hold.

ACKNOWLEDGMENT

The research was supported in part by NSF Grant DMS84-09447.

- ¹D. Kramer, H. Stephani, M. MacCallum, and E. Herlt, *Exact Solutions of Einstein's Field Equations* (Cambridge U. P., Cambridge, 1980).
- ²I. Robinson, "On plane wave and nullides," in *From SU(3) to Gravity*, edited by E. Gotsman and G. Tauber (Cambridge U. P., Cambridge, 1985).

³G. Lessner, "Axially symmetric *pp*-waves and their interpretation as extended massless particles," Gen. Relativ. Gravit. **18**, 899 (1986).

⁴R. Schoen and S.-T. Yau, "The proof of the positive mass theorem II," Commun. Math. Phys. 9, 231 (1981).

- ⁵E. Witten, "A new proof of the positive energy theorem," Commun. Math. Phys. **80**, 381 (1981).
- ⁶R. M. Wald, *General Relativity* (University of Chicago Press, Chicago, 1984).

⁷T. Parker and C. H. Taubes, "On Witten's proof of the positive energy theorem," Commun. Math. Phys. 84, 223 (1982).

⁸O. Reula, "Existence theorem for solutions of Witten's equation and nonnegativity of total mass," J. Math. Phys. 23, 810 (1982).

⁹A. Sen, "Quantum theory of spin ³/₂ field in Einstein spaces," Int. J. Theor. Phys. 21, 1 (1982).

- ¹⁰A. Ashtekar and G. Horowitz, "Energy-momentum of isolated systems cannot be null," Phys. Lett. A 89, 181 (1982).
- ¹¹P. F. Yip, "A strictly-positive mass theorem," Commun. Math. Phys. **108**, 653 (1987).

Some solutions of Einstein's equations with shock waves

A. M. Anile and G. Moschetti Dipartimento di Matematica Universita di Catania, Italy

O. I. Bogoyavlenski Steklov Mathematical Institute of Moscow, Moscow, USSR

(Received 27 October 1986; accepted for publication 11 March 1987)

The Einstein field equations in the presence of a polytropic fluid performing self-similar motion are reduced to a dynamical system. Qualitative properties of the dynamical system are investigated in the case when the fluid motion is with shock waves.

I. INTRODUCTION

Solutions of Einstein's equations with shock waves can be relevant in several areas of astrophysics. In particular, relativistic fluid motions with shocks occur in (1) nonlinear evolution of an initial adiabatic perturbation in the radiation dominated phase of the universe (in the standard scenario, neglecting the presence of collisionless particles)¹; (2) nonlinear collapse of an adiabatic perturbation in the matter dominated phase¹; (3) explosions in the intergalactic medium and hydrodynamical models of galaxy formation²; and (4) bubble growth in the phase transition assumed to occur in some models of the early universe.³

The problem of the evolution of cosmic thin shells has been tackled by using a general relativistic numerical code by Sato and co-workers.⁴⁻⁶ A thin shell is to be distinguished from a shock wave, because in the former there is a source of energy-momentum within the surface layer (and this is modeled as a δ -function discontinuity in the energy-momentum tensor). Thin shells could be reasonable models for propagating detonation waves.

Here we shall limit ourselves to shock waves which correspond to the approximation of neglecting the source of energy-momentum within the surface layer compared to the total explosion energy.

Cases (3) and (4) have been investigated by looking for similarity solutions in the framework of Newtonian fluid dynamics^{2,7,8} and special relativistic fluid dynamics, respectively.³

From the mathematical point of view it is very difficult to study general relativistic fluid motion without any symmetry. Therefore we choose to work in spherical symmetry (which is a very good approximation in the case when the source of the motion can be considered a point explosion). Even with this limitation the resulting equations are partial differential equations (PDE's) and a full analysis is still very difficult and in general one has PDE's to resort to numerical calculations.

In order to obtain a mathematically tractable problem we restrict ourselves to self-similar motion (which should be a good approximation asymptotically far from the source) and in this way the system of PDE's is reduced to a dynamical system to which we can apply the qualitative theory.⁹ In a previous paper¹⁰ one of us talked about the problem of the behavior of a self-similar shock wave in an expanding universe in the case of a barotropic fluid. In the present paper, among other things we also provide analytical proofs of some features of the fluid motion which were discovered by numerical integration.

The plane of the paper is the following: In Sec. II the basic equations in the case of a polytropic state equation are derived. In Sec. III the barotropic fluid case is revisited and a Lyapunov function for the dynamical system is found. In Sec. IV qualitative properties of the expanding self-similar shock wave in a barotropic fluid are studied. In Sec. V configurations with two expanding shock waves are analyzed. In Sec. VI some qualitative properties of the dynamical system in the politropic case are discussed. In Sec. VII still in the polytropic case a numerical integration is performed for an expanding shock wave in a Friedmann dust universe.

II. BASIC EQUATIONS

In Ref. 11 it is shown that the most general self-similar spherically symmetric space-time, in a fluid comoving reference frame, is

$$ds^{2} = e^{2\varphi(z)} dt^{2} - e^{2\psi(z)} dR^{2} - R^{2} \mathscr{R}^{2}(z) d\Omega^{2},$$

$$d\Omega^{2} = d\theta^{2} + \sin^{2} \theta d\chi^{2},$$
 (1)

where z = R/t, the fluid velocity is $U^a = (e^{-\varphi}, 0, 0, 0)$ and the vector $V^a = (t, R, 0, 0)$ is a homothetic Killing vector, that is, $L_V g_{ab} = 2g_{ab}$, where L_V is the Lie derivative along V.

In a reference frame adapted to the vector field V the metric (1) takes the following conformally static form⁹:

$$ds^{2} = e^{2\tau} (e^{\nu(r)} d\tau^{2} - e^{\lambda(r)} dr^{2} - r^{2} d\Omega^{2}), \qquad (2)$$

and the vector fields U and V become, respectively,

$$U^{a} = (e^{-\tau - \nu/2} / \sqrt{1 - u^{2}},$$

$$u e^{-\tau - \lambda/2} / \sqrt{1 - u^{2}}, 0, 0), \qquad (3)$$

$$V^a = (1, 0, 0, 0). \tag{4}$$

where u = u(r) is the spatial velocity of the fluid with respect to the hypersurfaces r = const.

In the following, the energy momentum tensor of a perfect fluid will be considered, that is,

$$T_{ab} = (p + \epsilon) U_a U_b - pg_{ab}, \qquad (5)$$

where p and ϵ denote, respectively, the pressure and the total energy density, which by self-similarity have the following form:

$$p(\tau,r) = e^{-2\tau} \bar{p}(r), \quad \epsilon(\tau,r) = e^{-2\tau} \bar{\epsilon}(r). \tag{6}$$

0022-2488/87/122942-07\$02.50

Furthermore, it will be assumed the state equation

$$p = (\gamma - 1)(\epsilon - c\rho), \tag{7}$$

where $\rho(\tau, r) = e^{-2\tau}\overline{\rho}(r)$ is the total density of matter, and γ and c are constants such that $1 < \gamma < 2$ and c = 0 or c = 1. In the first case Eq. (6) is the barotropic state equation and in the last one is the polytropic state equation.

In both cases the sound velocity *a* is given by

$$a^2 = \gamma p/(p+\epsilon). \tag{8}$$

From Eqs. (7) and (8) it follows that

$$\bar{\epsilon} = (\gamma - a^2)(\bar{p} + \bar{\epsilon})/\gamma, \tag{9}$$

$$\bar{p} = a^2 (\bar{p} + \bar{\epsilon}) / \gamma, \qquad ($$

and if c = 1,

$$\bar{\rho} = (\gamma - 1 - a^2)(\bar{p} + \bar{\epsilon})/(\gamma - 1). \tag{11}$$

By letting $z = \ln r$, = d/dz, $Q = re^{(\lambda - \nu)/2}$, $w = \dot{\nu} = r d\nu/dr$, Einstein's equations $R_0^0 - R_1^1 = \lambda (T_0^0 - T_1^1)$, $R_2^2 - \frac{1}{2}R = \lambda T_2^2$, the energy-momentum conservation laws $T_{0;a}^a = 0$, $T_{1;a}^a = 0$, and the baryon number conservation law $(\rho U^a)_{;a} = 0$ give the following evolution equations:

$$Q = Q(1 - w - Q^{2} - wQ(1 + u^{2})/(2u)), \qquad (12)$$

$$\dot{w} = -w(1+w+Q^2-(Q/u)((1+(w/2))(1+u^2))$$

$$+ (2a^2/\gamma)(1-u^2)), \qquad (13)$$
$$\dot{u} = - (1-u^2)(Q(u^2 - 3a^2 + 2(a^2/\gamma)(1-u^2)))$$

+
$$(w/2)u(1-a^2) - 2ua^2)/(u^2-a^2)$$
, (14)

$$\dot{a} = a(\gamma - 1 - a^2)((w/4)(1 - u^2) - u^2 - (Q/u))$$

$$\times (u^2 - (a^2/\gamma)(1-u^2)))/(u^2 - a^2).$$
(15)

By letting $L = e^{\lambda}$, Einstein's equation $R_{1}^{1} - \frac{1}{2}R = \lambda T_{1}^{1}$, gives the constraint equation

$$L = 1 + w - Q^{2} + (wQ/u)(u^{2} + (a^{2}/\gamma)(1 - u^{2})) > 0,$$
(16)

and the equation $R_{01} = \mathscr{L}T_{01}$ gives

$$\bar{p} + \bar{\epsilon} = -wQ(1-u^2)/(\varkappa r^2 uL).$$
(17)

From Eqs. (9)-(11) and (17) one obtains the functions $\bar{p}, \bar{\epsilon}, \bar{\rho}$ when a solution of the dynamical system (12)-(15) is known. Furthermore, Eq. (17) gives the constraint

$$\operatorname{sgn}(u) = -\operatorname{sgn}(w). \tag{18}$$

The mass conservation law $(\rho U^a)_{;a} = 0$ may be written in the following form:

$$(\bar{\rho}/\bar{\rho}) + (Q/u) + (w/2) + 2 + \dot{u}/(u(1-u^2)) = 0.$$
(19)

By deriving the expression for $\bar{\rho}$ which follows from Eqs. (11) and (17), one obtains

$$w = -(\bar{\rho}/\bar{\rho}) + (\dot{w}/w) - (\dot{Q}/Q) - 2a\dot{a}/(\gamma - 1 - a^2) - (1 + u^2)\dot{u}/(u(1 - u^2)),$$
(20)

and by Eqs. (14) and (15)

$$\dot{u}/(u(1-u^2)) + 2\dot{a}/(a(\gamma-1-a^2))$$

$$= -2 - (w/2) - 3(Q/u).$$
(21)
From Eq. (10) (21) one obtains

From Eqs. (19)-(21) one obtains

$$4 + 2(\bar{\rho}/\bar{\rho}) - (Q/Q) + (\dot{u}/u) + (\dot{w}/w) - 2(1 + a^2)\dot{a}/(a(\gamma - 1 - a^2)), \qquad (22)$$

which gives the following first integral:

$$Qw^{3}a^{-2/(\gamma-1)}(1-u^{2})^{2}(\gamma-1-a^{2})^{(3\gamma-2)/(\gamma-1)}/(uL^{2})$$

= c₁<0, (23)

where c_1 is an integration constant.

The invariant manifold $a^2 = \gamma - 1$ ($c_1 = 0$) corresponds to the barotropic case. On this invariant manifold equation (15) is identically satisfied and the dynamical system becomes⁹

$$\dot{Q} = Q(1 - w - Q^2 - wQ(1 + u^2)/(2u)),$$
 (12')

$$\dot{w} = -w(1+w+Q^2-(Q/u))((1+(w/2))(1+u^2))$$

$$+2k(1-u^{2})/(1+k))), (13')$$

$$u = -(1-u)((1-k)u(w/2) - 2ku + Q) \times (u^2(1-k) - k(1+3k))/(1+k))/(u^2-k),$$
(14')

with the constraint equations

$$L = 1 + w - Q^{2} + wQ(k + u^{2})/(u(1 + k)) > 0, \qquad (16')$$

$$\hat{\epsilon} = -wQ(1-u^2)/(k(1+k)ur^2L), \qquad (17')$$

$$\operatorname{sgn}(u) = -\operatorname{sgn}(w), \tag{18'}$$

where $k = \gamma - 1$.

10)

III. COMMENTS ON THE BAROTROPIC CASE

The qualitative features of the dynamical system in the barotropic case has been studied extensively in Ref. 9. Here, after an outline of the behavior of the solutions, we prove a new result. The main features of the dynamical system in the region u < 0 are pictured in Fig. 1 (in the region u > 0 there are no physical solutions because they cannot be continued until the center even if a discontinuity is introduced).

There are two kinds of solutions in the subsonic region $(-\sqrt{k} < u < 0)$: solutions which cross the plane $u = -\sqrt{k}$ in a point of the line *I*, and solutions which arrive in the region *L'* of the plane $u = -\sqrt{k}$. These latter solutions cannot be continued in the supersonic region $(u < -\sqrt{k})$ because on *L'* the dynamical system change direction; the only way to continue these solutions is to join them across a discontinuity (shock wave) to a solution in the supersonic region.

We recall that a shock wave is a timelike discontinuity hypersurface such that the first and second fundamental forms are continuous and such that the following continuity conditions hold true:

$$[T^{a}_{\ b}N^{b}] = 0, \quad [\rho U^{a}N_{a}] = 0, \quad (24)$$

where the brackets indicate the jump across the discontinuity. The continuity of the first and second fundamental form is equivalent to the continuity of the variables Q, w, and L, and these latter together with Eq. (16') give the following jump conditions^{9,10}:

$$[Q] = [w] = [(k + u^2)/(u(1 + k))] = 0.$$
(25)

The first of Eqs. (24) does not give further constraints, and



FIG. 1. Phase space of the dynamical system (12')-(14') (barotropic state equation).

the second is an identity in the barotropic case.

In the subsonic region the solutions in the interior of phase space which start from some singular point are the following.

(a) Solutions filling the two-dimensional separatrix Z of the singular point Z_1 (it is possible to identify the separatrix Z in Fig. 1 by taking into account that it intersects the component of the boundary u = 0, w = 0, and L = 0, respectively, in the line L_1, L_2 , and L_3).

(b) The limit line X of the separatrix Z which starts from the singular point Z_3 and corresponds to the Oppenheimer-Volkov solution. All other trajectories come from the unphysical region u > 0 across the line N (u = w = 0, Q > 0).

The behavior of the latter solutions has been studied in Ref. 10 and it has been proved that it is necessary to join them across a discontinuity at u = 0 to some solution of the full spherically symmetric Einstein equations and if this latter solution contains the center of symmetry it must possess a trapped surface and therefore a space-time singularity.

However, it is possible to avoid the singularity by making a change in the topology of solution in such way that the center of symmetry does not belong to the space-time. A trivial example in which this last solution may be realized is obtained by stopping the self-similar solution at u = 0 and then by continuing it with its mirror image.

The qualitative behavior of the solution obtained in Ref. 9 is based on the assumption that all solutions start (end) from (to) some singular point or from the characteristic plane $u = -\sqrt{k}$, that is, it is assumed that there are no compact invariant manifolds near which solutions wind around.

Now we shall prove that there are no invariant compact manifolds. Because on a compact manifold any continuous function has a maximum, to prove the statement it is sufficient to show the existence of a Lyapunov function, that is a C^{1} function F(Q,w,u) defined in the interior of the dynamical system which is increasing on the solutions of Eqs. (12')-(14').

By Eqs. (12'), (13'), (14'), and (16'),

$$\dot{L} = -L(w + 2Q^2 + wQ(1 + u^2)/u).$$
 (26)

$$L = -L(w + 2Q^{2} + wQ(1 + u^{2})/u).$$
 (7)

By deriving the function

$$F = L^{1-k}Q^{-1-k}w^{3k-1}u^{-2k}(1-u^2)^{k-1} > 0 \quad (27)$$

and by using Eqs. (12'), (13'), (14'), and (26), one obtains

$$F = (1-k)(1+3k)FQ(1-u^2)/(u(1+k)), \quad (28)$$

which is always positive in the region u > 0 and negative in the region u < 0. Therefore, Eq. (27) gives a Lyapunov function in the regions u > 0 and u < 0 separately.

In particular, if k = 1 one obtains the first integral

$$w/(Qu) = c_2 < 0.$$
 (29)

IV. EXPANDING SELF-SIMILAR SHOCK WAVE IN THE FLAT FRIEDMANN UNIVERSE (BAROTROPIC CASE)

The exact solution F of the dynamical system (12')-(14') corresponding to the flat Friedmann solution has the form⁹

$$Q = u/(\bar{\beta} - 1 - \bar{\beta}u^2),$$

$$w = -2\bar{\beta}u^2/(\bar{\beta} - 1 - \bar{\beta}u^2),$$

$$u = (\bar{\beta} - 1)\mu,$$

$$r(\mu) = C_2 \mu (1 - (\bar{\beta} - 1)^2 \mu^2)^{1/(2\bar{\beta} - 2)},$$

$$\bar{\beta} = 2/(3(1 + k)).$$

(30)

This solution belongs to the separatrix Z and crosses the plane L of nonextensibility of solutions $(u = -\sqrt{k})$ in a point F_1 of the line of singular points I.

Proposition 1: The intersection of the separatrix Z with the plane L in the neighborhood of the point F_1 at $0 < k < \frac{1}{3}$ coincides with a segment on the line I.

Proof: In coordinates α , β , and parameter τ_1 such that

$$\alpha = \frac{1}{Q}, \quad \beta = \frac{(1-k)w}{2Q} - \frac{2k}{Q},$$

$$\frac{d\tau_1}{dz} = -\frac{Q}{u^2 - k},$$
(31)

the eigenvalues of singular points of the segment I are⁹

$$\lambda_{\pm} = \sqrt{k} (1-k)(-1 + (1 \pm 2Z_0(1-k)^{-1})^{1/2}),$$

$$\lambda_3 = 0,$$
(32)

with

$$Z_0 = (2k(1+3k)/(1-k))\alpha^2 - (2\sqrt{k}/(1-k^2))$$
$$\times ((1+k)^3 + 2k(3+k^2))\alpha + 6k$$
$$+ 16k^2(1+k^2)/((1-k)(1+k)^2).$$

After substituting in Eqs. (32) the value of parameter $\alpha = (1+5k)/(3\sqrt{k}(1+k))$, corresponding to the singular point F_1 , one obtains, in the case $0 < k < \frac{1}{3}$, the formulas

$$\lambda_{+} = -2\sqrt{k} (1+3k)(1-k)/(3(1+k)),$$

$$\lambda_{-} = -4\sqrt{k} (1-k)/(3(1+k)).$$
(33)

The corresponding eigenvectors are

$$V_{+} = (1/3k, -4/(3(1+k)), 1),$$

$$V_{-} = ((1+3k)/6k, -2(1+3k)/(3(1+k)), 1).$$
(34)

From the formula (33) we have $\lambda_+ > \lambda_-$, so the qualitative behavior of the dynamical system near the singular point F_1 is like in Fig. 2, where L_+ and L_- denote the invariant manifolds filled by the separatrix of the singular points of the segment *I*.

By calculating the tangent to the trajectory F in the point F_1 it is easily seen that F corresponds to the eigenvalue λ_+ , so the trajectory F is tangent to the stable two-dimensional invariant manifold L_+ filled with the separatrices of singular points of the line I corresponding to the greatest eigenvalues λ_+ .

Therefore all trajectories on the two-dimensional separatrix Z, which are close to the trajectory F, are tangent to the invariant manifold L_+ . Hence all of them enter a segment I_Z of the line I in a neighborhood of the point F_1 . This proves the statement.

In the following we consider self-similar solutions with an expanding shock wave such that in front of the shock wave there is the Friedmann solution with state equation $p = k_1 \epsilon$ and inside there is the solution, with state equation $p = k_2 \epsilon (k_1 < k_2 < \frac{1}{3})$, corresponding to the initial data determined by Eqs. (25).

The initial data are not uniquely determined but depend on the radial velocity $-u_2$ of the shock wave which respect to the interior fluid. Therefore there is a line G in the phase space such that the solutions defined previously (but with different Mach numbers) arrive at different points of G before the jump.



FIG. 2. Qualitative behavior of the dynamical system (12')-(14') near the singular point F_1 .

The line G intersects the plane L of nonextensibility of solutions $u = -\sqrt{k_2}$ in some point G_0 in the region L', which is at a finite distance from the segment I_Z .

Because G_0 depends continuously on parameter k_2 and tends to Z_1 when k_2 tends to k_1 , it follows that for $k_2 \approx k_1$ there exists some segment G_X on the line G (beginning from the point G_0) which does not intersect the separatrix Z (see Ref. 12). From this fact and from the presence of the Lyapunov function, it follows that all trajectories of the dynamical system (12')-(14') starting from the line G_X , with decreasing parameter z, arrive to the singular line N. This proves the following.

Corollary: All self-similar solutions, which may be matched with the Friedmann solution across a shock wave at $k_2 \approx k_1$ and small enough Mach number, belong to the class of solutions which come from the region u > 0 (see Sec. III).

This corollary gives an analytical proof of the numerical results obtained in Ref. 10.

V. SELF-SIMILAR SOLUTIONS WITH TWO EXPANDING SHOCK WAVES

The existence of self-similar solutions with two shock waves in general relativity is connected with the presence of segment I_1 of saddle singular points of the line *I*. For these points there exists a two-dimensional separatrix L_1 filled with the trajectories *T* passing through the segment I_1 from supersonic to subsonic domain (see Fig. 3).

We consider two points K_1 , K_2 on such trajectory T, at the point K_1 we have $u < -\sqrt{k}$, at the point k_2 , $u > -\sqrt{k}$. The points JK_1 and JK_2 correspond to the points K_1 , K_2 by means of jump conditions (25). We denote T_1 and T_2 , the trajectories passing through the points JK_1 and JK_2 for decreasing and increasing self-similar parameter r, respectively.²

The whole self-similar solution is described by three trajectories T_1 , T, T_2 , and has two shock waves, corresponding to the jumps at the points $JK_1 \rightarrow K_1$ and $K_2 \rightarrow JK_2$. In general



FIG. 3. Solutions with two expanding shock waves.

the trajectory T_1 , for decreasing values of parameter r, tends to some point on the segment N and hence its behavior is described like in the Sec. III. But if the point JK_1 belongs to the separatrix Z then the trajectory T_1 tends to the point Z_1 for $r \rightarrow 0$ and the corresponding self-similar solution is continued to the center of symmetry without singularity. The trajectory T_2 for increasing values of the parameter r tends to some point on segment DE (see Fig. 1), then it may be continued without singularity in the domain u > -1. This trajectory describes the solutions in front of the first expanding shock wave.

These solutions depend on six parameters: one parameter determines the position of the singular point on the segment I_1 , two parameters determine the positions of the points K_1 , K_2 on the trajectory T, and three parameters k_1 , k_2 , k_3 determine the state equation $p = k\epsilon$ for three trajectories T_1 , T, T_2 . By means of variations of these six parameters one may try to obtain a self-similar solution with some extra properties. Some of these extra properties may be that the point JK_2 belongs to the Friedmann trajectory or to the Oppenheimer-Volkov trajectory, and the point JK_1 belongs to the separatrix Z.

The problem of the existence of the points JK_1 and JK_2 with these properties may be solved by means of numerical calculations or by analytic methods at small Mach numbers $M \approx 1$. These new solutions with two expanding shock waves are analogs of the known Landau solution with two shock waves in classical gas dynamics.¹³

VI. SOME QUALITATIVE REMARKS IN THE POLYTROPIC CASE

In this section some qualitative results related to the general dynamical system (12)-(15), which are helpful for the investigation of the solutions described in the next section, will be found. Because of Eqs. (16) and (23) the dynamical system is defined in the region -1 < u < 1, Q > 0, L > 0, $0 < a < \sqrt{\gamma - 1}$, which may be divided in two regions A: -1 < u < 0 and B: 0 < u < 1.

In region A we have the following results.

(i) Equation (16) can be written in the following form:

 $-w + Q^{2} - (wQ/u)(u^{2} + (a^{2}/\gamma)(1 - u^{2})) < 1$ (16")

and Eq. (18) will give the following bounds for the region A:

$$0 < Q < 1, -1 < w < 0, 0 < u < 1, 0 < a < \sqrt{\gamma - 1}.$$

(ii) The dynamical system changes direction on the hyperplane u = a, in fact it is easily seen that the numerator of the right-hand side of Eq. (14),

$$-(1-u^2)(Q(u^2-a^2+2a^2(1-u^2-\gamma)/\gamma) + (w/2)u(1-a^2) - 2ua^2),$$
(35)

is positive in the region u < a ($\dot{u} < 0$) and in a neighborhood of the hyperplane u = a in the region u > a ($\dot{u} > 0$). Therefore there are no solutions that cross the hyperplane u = abut all solutions exit in both sides of this hyperplane (Fig. 4).

(iii) It is easy to see that in the region u < a there are no singular points and in the region u > a there are only saddle singular points with separatrices belonging to the boundary of phase space. This means that there are no interior solutions starting from some singular point, hence all interior solutions start with a positive value of parameter r from the hyperplane u = a.

From (ii) and (iii), as in the barotropic case, the following holds.

Proposition: All trajectories lying in the interior of region u > 0 cannot be continued until the center even if a discontinuity (shock wave) is introduced.

Solutions starting from the hyperplane u = a in the side u < a arrive to the region u < 0 across the hyperplane u = 0 with a finite value r_0 of parameter r, because in this hyperplane there are no singular points of the dynamical system. For these solutions it must be $w(r_0) = a(r_0) = 0$. The first equation follows immediately from Eq. (18), the second one follows from the first and from Eq. (23) by taking into account that by Eq. (16) the function wQ/u cannot tend to $-\infty$ when r tends to r_0 .

Following this section we will study the behavior of the dynamical system in the neighborhood of the line N (w = u = a = 0). Before making this study it is necessary to regularize the dynamical system on N by taking the new parameter z_1 such that $dz/dz_1 = u(u^2 - a^2)$.

In this way the points of N become singular points for the new dynamical system, but unfortunately they are degenerate (all eigenvalues are zero). This difficulty may be avoided by considering the following new coordinates $x = (Q - \alpha)/a$, y = w/a, z = u/a, a in a neighborhood of the singular point $Q = \alpha$, w = u = a = 0. In these new coordinates the dynamical system becomes

$$\dot{x} = (z^2 - 1)(xa + \alpha)(z(1 - ya - (xa + \alpha)^2) - (y/2)(xa + \alpha)(1 + z^2a^2)) - xF_a/a, \qquad (12'')$$



FIG. 4. Behavior of the dynamical system (12)-(15) near the hyperplane u = a.

$$\dot{y} = -y((z^2 - 1)(za(1 + ya + (xa + \alpha)^2) + (xa + \alpha)((1 + (ya/2))(1 + z^2a^2) + (2a^2/\gamma)(1 - z^2a^2)) + F_a/a), \qquad (13'')$$

$$\dot{z} = -z((1 - z^2a^2)((xa + \alpha)(z^2 - 3 + (2/\gamma)(1 - z^2a^2)))$$

$$+ (yz/2)(1-a^{-}) - 2za) + F_{a}/a, \qquad (14')$$

$$\dot{a} = F_{-}(x, y, z, a), \qquad (15'')$$

 $\dot{a} = F_a(x, y, z, a),$

$$F_a = a(\gamma - 1 - a^2)((yz/4)(1 - z^2a^2) - z^3a) - (xa + \alpha)(z^2 - (1/\gamma)(1 - z^2a^2))).$$

The system (12'')-(15'') after linearization has the following eigenvalues:

$$\lambda_{x} = -\alpha(\gamma - 1)/\gamma, \quad \lambda_{y} = \alpha/\gamma, \lambda_{z} = \alpha(2\gamma - 1)/\gamma, \quad \lambda_{a} = \alpha(\gamma - 1)/\gamma.$$
(36)

This means that for any α and therefore for any point P of the line N there exists a three-dimensional unstable separatrix that is an invariant three-dimensional manifold filled by trajectories exiting from P. Furthermore for $r \rightarrow r_0$, u, $a^{(3\gamma-2)/(\gamma-1)}$, $w^{(3\gamma-2)/\gamma}$ are infinitesimal of the same order. From this and from Eqs. (9), (10), and (17) it follows that for such trajectories, for $r \rightarrow r_0$, the pressure tends to a positive constant and the density of energy tends to infinity. By defining the total mass enclosed within the sphere of radius $e^{\tau}r$ as

$$m(\tau,r) = e^{\tau} r (1 + r^2 e^{-\nu} - e^{-\lambda})/2, \qquad (37)$$

it follows that the function

$$M(r) = 2m/(e^{r}r)$$

= $(w + (Qw/u)(u^{2} + (a^{2}/\gamma)(1 - u^{2})))/L$ (38)

must be negative on r_0 because the ratio wa^2/u tends to a negative constant. In Ref. 10 it has been proved that the condition $M(r_0) < 0$ implies the existence of a trapped hypersurface in the interior solution $(0 < r < r_0)$ holding the center of symmetry.¹⁴

Thus for trajectories coming from the region u > 0 we obtain the following conclusions.

(1) As in the barotropic case the only possibility of obtaining a complete solution is to join these solutions at $r = r_0$ with non-self-similar solutions containing the center of symmetry, across a shock wave [contact discontinuity because $u(r_0) = 0$] or to change the topology of interior solution; in the first case the interior solution possesses a trapped surface.

(2) The behavior of these solutions near the contact discontinuity at $r = r_0$ is different from the corresponding barotropic solutions,¹¹ because in this case the density of energy tends to infinity.

VII. EXPANDING SELF-SIMILAR SHOCK WAVE IN THE FLAT FRIEDMANN UNIVERSE (POLYTROPIC CASE)

In Sec. IV we have considered solutions of the dynamical system (12')-(14'), which are matched with flat Friedmann solution across a shock wave with a small Mach num-



FIG. 5. Behavior of energy density and pressure for the solution of the dynamical system (12)-(15) with $\gamma = \frac{4}{3}$, matched to the Friedmann solution with dust for u = -0.3.

ber. In this section we will look for solutions of the general dynamical system (12)-(15) which may be joined to the flat Friedmann solution with dust (p = 0) across a shock wave, without restriction on Mach number.

In the polytropic case the continuity of Q, w, and L give the following jump conditions:

$$[Q] = [w] = [u + a^2(1 - u^2)/(\gamma u)] = 0.$$
(39)

As in the barotropic case the first of Eqs. (24) does not give further constraint, but the second becomes



FIG. 6. Like Fig. 5 with u = -0.2.



FIG. 7. Like Fig. 5 with u = -0.05.

$$(\gamma - 1 - a^2)\sqrt{(1 - u^2)}/(\gamma - 1) = 0.$$
 (40)

Equations (39) and (40) are the jump conditions on a shock wave. The Friedmann solution with dust, is obtained from Eqs. (30) by letting k = 0:

$$Q_F = \frac{9r\sigma}{9\sigma^2 + 2r^2}, \quad w_F = \frac{4r^2}{9\sigma^2 + 2r^2}, \quad u_F = -\frac{r}{3\sigma},$$
(41)

where $\sigma(r)$ is defined implicitly by the equation

$$\sigma^2 = c_3 (9\sigma^2 - r^2)^{3/2} \tag{42}$$

and c_3 is an integration constant.

By calculating the density of matter (= density of energy) and by taking into account that a = 0 it follows that the matching conditions (39) and (40) become

$$Q(r_0) = Q_F(r_0), \quad w(r_0) = w_F(r_0),$$
 (43)

$$u + a^2(1 - u^2)/(\gamma u) = u_F,$$
 (44)

$$(\gamma - 1 - a^2)\sqrt{1 - u^2}/(\gamma - 1) = \sqrt{1 - u_F^2},$$
 (45)

where it is understood that the variable in Eqs. (44) and (45) are calculated at $r = r_0$ (equation of shock wave).

By using the third equation of (41) and Eq. (42), Eqs. (44) and (45) give

$$a^{2} = 2\gamma(\gamma - 1)u^{2}/((\gamma - 1)^{2} + (2\gamma - 1)u^{2}),$$
(46)

$$r_0 = -u(\gamma^2 - 1 - u^2)((\gamma - 1)^2 + (2\gamma - 1)u^2)^2/(9(1 - u^2)^{3/2}((\gamma - 1)^2 - u^2)^3)$$
(47)

and

$$u_F = u(\gamma^2 - 1 + u^2) / ((\gamma - 1)^2 + (2\gamma - 1)u^2).$$
(48)

Equations (43), (46), and (47) give the initial data for the interior solution for any choice of u.

From Eq. (46) it follows that $0 < a^2 < \gamma - 1$ if and only if *u* belongs to the interval $(1 - \gamma, 0)$. And from Eq. (48) it follows that $-u_F$ (= velocity of the shock wave with respect to the exterior fluid) is nonzero and less than the velocity of light (= 1) if *u* is different from 0 and $1 - \gamma$.

Therefore solutions in the interior of the shock wave are parametrized by the variable u belonging to the interior of the interval $(1 - \gamma, 0)$. For some value of u in the previous interval and for $\gamma = \frac{4}{3}$, the corresponding numerical solutions of the system (12)-(15) have been found. These solutions come from the region u > 0, then qualitative analysis made in the previous section may be used.

The numerical behavior of the pressure and energy density for any solution is pictured in Figs. 5–7.

ACKNOWLEDGMENTS

This work was in part supported by the Italian Ministry of Education.

O. I. Bogoyavlenski and G. Moschetti gratefully acknowledge the Agreement of Cooperation between the Italian, Consiglio Nazionale delle Ricerche (CNR) and the Soviet Academy of Sciences.

- ¹P. J. E. Peebles, *The Large Scale Structure of the Universe* (Princeton U. P., Princeton, NJ, 1980).
- ²S. Ikeuchi, K. Tomisaka, and J. P. Ostriker, Astrophys. J. 265 (1983).
- ³P. J. Steinhardt, Phys. Rev. D 25, 2074 (1982).
- ⁴K. Maeda and H. Sato, Prog. Theor. Phys. 70 (1983).
- ⁵K. Maeda and H. Sato, Prog. Theor. Phys. 71 (1984).
- ⁶Y. Suto, K. Sato, and H. Sato, "Nonlinear evolution of negative density perturbations in a radiation dominated universe," preprint.
- ⁷E. Bertschinger, Astrophys. J. 268 (1983).
- ⁸E. Bertschinger, Astrophys. J. 295 (1985).
- ⁹O. I. Bogoyavlenski, *Methods of Qualitative Theory of Dynamical Systems in Astrophyics and Gas Dynamics* (Nauka, Moscow, 1980) (in Russian) [English translation (Springer, New York 1985)].
- ¹⁰G. Moschetti, Gen. Relativ. Gravit. 19, 155 (1987).
- ¹¹M. E. Cahill and A. H. Taub, Commun. Math. Phys. 21, 1 (1971).
- ¹²Rigorously speaking we make the reasonable hypothesis that the separatrix Z is non-self-intersecting in the point F_1 . The opposite case seems pathological and unstable with respect to the variations of the parameter k_2 .
- ¹³L. D. Landau, J. Phys. (Moscow) 9, 496 (1945).
- ¹⁴In this case it is not possible to apply the singularity theorems because these are proved by taking into account only Einstein's equations, but now we have also used the equation of conservation of mass $(\rho U^a)_{;a} = 0$.

General exact solutions of Einstein equations for static perfect fluids with spherical symmetry

Sonia Berger Instituto de Física, Universidad Austral de Chile, Casilla 567, Valdivia, Chile

Roberto Hojman Departamento de Física, Facultad de Ciencias, Universidad de Chile, Casilla 653, Santiago, Chile and Centro de Física El Trauco, Fanor Velasco 41D, Santiago, Chile

Jorge Santamarina^{a)} International Centre for Theoretical Physics, Trieste, Italy

(Received 7 April 1986; accepted for publication 3 June 1987)

The gravitational field equations for a spherical symmetric perfect fluid are completely solved. The general analytical solution obtained depends on an arbitrary function of the radial coordinate. As illustrations of the proposed procedure the exterior and interior Schwarzschild solutions are regained.

I. INTRODUCTION

In this paper we derive the whole set of exact spherically symmetric solutions of Einstein gravitational field equations (with cosmological constant) when a perfect fluid is assumed to be the source of the gravitational field.

An extension of the solution to plane and hyperbolic symmetries can easily be obtained and our previous result¹ for the (exact solution of the) plane symmetric case is regained.

In what follows we pose the mathematical problem and then we mimic the procedure used in our recent work¹ to reduce the problem to mere quadratures.

The line element considered here is

$$ds^{2} = g^{2}(x)dt^{2} - dx^{2} - r^{2}(x)[d\theta^{2} + \sin^{2}\theta d\varphi^{2}], \quad (1)$$

which represents the most general static line element admitting spherical transformations.

Einstein field equations with cosmological constant are

$$G_{\mu\nu} + \Lambda g_{\mu\nu} = T_{\mu\nu} \tag{2}$$

with

$$G_{\mu\nu} = R_{\mu\nu} - \frac{1}{2}g_{\nu\mu}R.$$
 (3)

In Ref. 1 we proved that if the right-hand side of Eq. (2) models a perfect fluid with pressure p and energy density ρ ,

$$T_{\mu\nu} = (\rho + p)u_{\mu}u_{\nu} - pg_{\mu\nu}, \qquad (4)$$

and whose flow lines are tangent to the unit vector $u^{\mu} = (1/g)\delta_0^{\mu}$, the change of variable

$$\tilde{p} = p - \Lambda, \quad \tilde{\rho} = \rho + \Lambda,$$
 (5)

transforms Eq. (2) into an equivalent system with $\Lambda = 0$. Therefore, twirls can now be dropped keeping in mind that the $\Lambda \neq 0$ case is already included. The field equations (2) for the metric given by (1) are (see, for instance, Ref. 1)

$$2r''/r + r'^2/r^2 - 1/r^2 = -\rho,$$

$$r'^2/r^2 + 2r'g'/rg - 1/r^2 = p,$$

$$r''/r + g''/g + r'g'/rg = p,$$
(6)

and the equation of hydrostatic support (Bianchi identities) is

$$(\rho + p)(g'/g) + p' = 0, \tag{7}$$

where the prime denotes x differentiation.

II. INTEGRATION

The system of equations can be reaccommodated if the first equation of (6) is multiplied by r^2r' and integrated over x. Then

$$r'^2 = 1 - 2m(r)/r,$$
 (8)

where

$$\frac{dm}{dr} = \frac{1}{2}\rho r^2.$$
(9)

Replacing Eqs. (7) and (8) in the second equation of (6) one obtains

$$(r-2m)\left(\frac{1}{r^3} - \frac{2}{r^2}\frac{dp}{dr}\frac{1}{\rho+p}\right) = p + \frac{1}{r^2}.$$
 (10)

Defining

$$G \equiv -(r-2m)/(p+1/r^2)$$
(11)

Eq. (10) becomes

$$r^{3}G(G-r^{3})\frac{dp}{dr} + r^{3}(G+r^{3})\left(\frac{dG}{dr} + r^{2}\right)p + (G+r^{3})\left(r^{3} + r\frac{dG}{dr} - 2G\right) = 0.$$
 (12)

Equation (12) can be integrated at once for p(r) if G(r) is a given function, in fact,

$$p(r) = \exp\left[\int \frac{(G+r^3)(dG/dr+r^2)}{G(r^3-G)}\right] \\ \times \left\{ p_0 - \int \frac{(G+r^3)(r^3+r(dG/dr)-2G)}{r^3G(G-r^3)} \\ \times \exp\left[-\int \frac{(G+r^3)(dG/dr+r^2)}{G(r^3-G)} dr \right] dr \right\}, (13)$$

where p_0 is an integration constant.

The function $\rho(r)$ can be obtained with the help of Eq. (9) and the definition of G

$$\rho = \frac{1}{r^2} \left(G \frac{dp}{dr} + \frac{dG}{dr} p - \frac{2G}{r^3} + \frac{dG/dr}{r^2} + 1 \right), \quad (14)$$

^{a)} Permanent address: Instituto de Física, Universidad Austral de Chile, Casilla 567, Valdivia, Chile.

where p(r) is given by Eq. (13). The metric coefficient g can be found by direct integration of Eq. (7) and then using Eq. (10) and the definition (11)

$$g^{2}(r) = g_{0}^{2} \exp\left(-2\int \frac{dp/dr}{\rho+p} dr\right)$$
$$= \frac{g_{0}^{2}}{r} \exp\left(-\int \frac{r^{2}}{G} dr\right).$$
(15)

To complete the integration we can recover the link between the metric coefficient r and the original variable x. From Eq. (8)

$$x = \int \frac{dr}{\sqrt{1 - 2m/r}} \tag{16}$$

with m(r) constructed from Eq. (11).

As we have remarked in our previous work the crucial step to get the general solutions (13)-(16) is the definition (11) of the arbitrary function G(r). It is worthwhile to emphasize that any choice of G(r) does provide a solution to field equations (6), (7) and no spurious solutions are introduced anywhere as it can be straightforwardly shown by replacing p, ρ , g, and r given, respectively, by expressions (13), (14), (15), and (16) in the original field equations. It is rather simple and illustrative to see how the prescription works.

By using the plane symmetric version of the above-depicted procedure,¹ the most general function G that produces a γ law equation of state was found.² Also, the associated fluid pressure, energy density, and metric coefficients were explicitly written down. In particular, after a change of variables, the solutions given by Tabensky and Taub,³ and by Teixeira, Wolk, and Som⁴ were reencountered.

III. EXAMPLES

A. Exterior Schwarzschild solution

In the present case, choose for instance

$$G(r) = -r^2(r - 2M)$$
(17)

(where M is a constant) and

$$p_0 = 0.$$
 (18)

Under such conditions

$$r^3 + r\frac{dG}{dr} - 2G = 0 \tag{19}$$

and consequently [see Eq. (13)]

 $p(r) = 0. \tag{20}$

Equations (14), (19), and (20) imply

$$\rho(r) = 0. \tag{21}$$

$$m(r) = M$$

Thus, Eq. (16) can be written as

$$dx^2 = \frac{dr^2}{1 - 2M/r}.$$
 (23)

$$g^2 = (1 - 2M/r).$$
 (24)

As it can be readily recognized, Eqs. (20), (21), (23), and

(24) represent the well known exterior Schwarzschild solution.

It is perhaps worthwhile to notice that in this case $\rho + p = 0$ and Eq. (10) seems to loose its meaning. However, by virtue of Eq. (7) the finiteness of the quotient $p'/(p+\rho)$ is ensured if we assume g'/g to be finite everywhere (except for some r).

B. Interior Schwarzschild solution

Now take⁵

$$G = -r^{3} \frac{A\sqrt{1 - r^{2}/R^{2}} - B(1 - r^{2}/R^{2})}{A\sqrt{1 - r^{2}/R^{2}} - B(1 - 3r^{2}/R^{2})}, \quad (25)$$

where A, B, and R are constants.

By choosing $p_0 = 0$ in the expression (13) for p(r), it is found after a lengthy calculation that

$$p(r) = \frac{1}{R^2} \frac{3B\sqrt{1 - r^2/R^2} - A}{A - B\sqrt{1 - r^2/R^2}}.$$
 (26)

Also, from Eq. (14)

$$\rho(r) = 3/R^2.$$
(27)

Finally Eqs. (15) and (16) give

$$g^{2}(r) = g_{0}^{2} (A - B\sqrt{1 - r^{2}/R^{2}})^{2},$$
 (28)

$$h^{2}(r) \equiv \frac{1}{1 - 2m/r} = \frac{1}{1 - r^{2}/R^{2}}.$$
 (29)

Equations (26)–(29) represent the interior Schwarzschild solution.⁶

As a final comment, let us say that any other solution can be cast in the above scheme as well. In fact, given any metric that solves Eqs. (6) and (7), the associated function G can be obtained from Eq. (15) as

$$G = -r^{3} \frac{g^{2}}{(d/dr)(rg^{2})}.$$
 (30)

The other relevant functions (p,ρ,h^2) are related to G through the above stated formulas.

In Ref. 5 the generating functions G leading to Tolman, Einstein, Schwarzschild-de Sitter, many other known⁸ and some new explicit solutions are exhibited.

ACKNOWLEDGMENTS

(22)

Two of us (R. H. and J. S.) are indebted to Professor Abdus Salam, the International Atomic Energy Agency, and UNESCO for hospitality at the International Centre for Theoretical Physics, Trieste. One of us (R. H.) gratefully acknowledges the support of the Fondo Nacional de Ciencias, Comisión Nacional de Investigaciones Científicas y Tecnológicas (Chile). The work of another of us (J. S.) was partially supported by Grant No. S-8235 of the Dirección de Investigación y Desarrollo, Universidad Austral de Chile.

²C. Collins, J. Math. Phys. 26, 2268 (1985).

- ³R. Tabensky and A. H. Taub, Commun. Math. Phys. 29, 61 (1973).
- ⁴A. F. da F. Teixeira, I. Wolk, and M. M. Som, J. Phys. A 10, 1679 (1977).

- ⁶K. Schwarzschild, Sitzungsber. Preuss. Akad. Wiss. 1919, 424.
- ⁷R. C. Tolman, Phys. Rev. 55, 364 (1939).

⁸D. Kramer, H. Stephani, M. MacCallum, and E. Hertl, *Exact Solutions of Einstein's Field Equations* (Cambridge U.P. Cambridge, 1980).

¹R. Hojman and J. Santamarina, J. Math. Phys. 25, 1973 (1984).

⁵P. Gaete, M.Sc. Thesis, U. de Chile, 1987 (unpublished).
Colliding gravitational plane waves with noncollinear polarization. II

Frederick J. Ernst Department of Mathematics, Clarkson University, Potsdam, New York 13676

Alberto García D. Departamento de Física, Cinvestav del Instituto Politécnico Nacional, Apdo. 14-740, 07000 México, D. F., Mexico

Isidore Hauser 1615 Cottonwood Drive, #8, Louisville, Colorado 80027

(Received 10 July 1987; accepted for publication 30 July 1987)

A simple criterion for colliding gravitational plane waves is developed. This colliding wave condition is preserved by a new realization of the Geroch group augmented by a Kramer-Neugebauer involution. A three-parameter generalization of a two-parameter family of solutions with noncollinear polarization discovered recently by Ferrari, Ibañez, and Bruni is presented, and two additional solutions are derived that demonstrate that much larger families are likely to be constructed in the near future.

I. INTRODUCTION

In an earlier paper¹ we discussed the derivation of new colliding wave solutions of the vacuum Einstein solutions from old ones through the use of coordinate and Ehlers transformations. As an example, we gave explicit results for the metric and the curvature tensor of the first new solution obtained by applying such transformations to the famous Nutku-Halil solution.² In the collinear limit the new solution reduced to the n = 2 solution of the Ferrari-Ibañez family³ of solutions. (The Kasner parameter n was defined in paper I.) In principle, additional coordinate and Ehlers transformations would yield noncollinear generalizations of all the even *n* Ferrari-Ibañez solutions. Similarly, starting with the Chandrasekhar–Xanthopoulos n = 1 solution,⁴ a series of such transformations would in principle yield noncollinear generalizations of all the odd n Ferrari-Ibañez solutions.

As we were going to press we learned that Ferrari, Ibañez, and Bruni⁵ had actually obtained closed form expressions for a two-parameter family of colliding wave solutions with noncollinear polarization. Interestingly, for even n their solutions agreed with ours, while for odd n their solutions were distinct. In particular, their n = 1 solution was not the Chandrasekhar-Xanthopoulos (Kerr) solution. We soon discovered that the Ferrari-Ibañez-Bruni family could easily be extended to a three-parameter family such that all the known solutions were special members of this family. That resolved the mystery, for it became apparent that the Ferrari-Ibañez-Bruni n = 1 solution bears the same relation to the Schwarzschild-NUT solution as the Chandrasekhar-Xanthopoulos n = 1 solution bears to the Kerr solution. In our three-parameter family the n = 1 solution corresponds to Kerr-NUT!

In Sec. II, Eq. (2.30) of this paper we shall describe a simple criterion for distinguishing a colliding wave solution from other solutions. That section will culminate in explicit metrical expressions, Eqs. (2.45)–(2.48), for our three-parameter generalization of the Ferrari–Ibañez–Bruni family of solutions.

It is quite apparent that our three-parameter family of

solutions can be further enlarged. In Sec. II we shall describe a simpler realization $G_0(\Sigma_1)$ of a Geroch group⁶ of transformations which transform one colliding wave solution into another. That section will culminate in the description of a new three-parameter n = 2 solution, Eqs. (3.29)-(3.31), obtained from the Nutku-Halil n = 0 solution by employing transformations which are members of our realization of the Geroch group. The new parameter in this solution is denoted by Δ .

In principle, one should be able to extend the three-parameter family of solutions described in Sec. II to a fourparameter family with the additional parameter Δ . For integral values of *n* one can proceed as follows. One can employ a Kramer-Neugebauer involution⁷ to obtain from the threeparameter n = 2 solution a new three-parameter n = 3 solution. The augmentation of our realization of the Geroch group by the Kramer-Neugebauer involution is formulated in Sec. IV, which culminates in the new three-parameter n = 3 solution, Eqs. (4.12) and (4.13). Additional coordinate and Ehlers transformations can, of course, be used to derive analogous three-parameter solutions for arbitrary integer values of *n*.

It remains to find an effective way to construct the fourparameter generalization (including Δ) of our own threeparameter generalization of the Ferrari–Ibañez–Bruni solutions. In a subsequent paper we shall formulate a homogeneous Hilbert problem, the solution of which may bring this objective within our grasp.

II. A COLLIDING WAVE CONDITION

A. The set CW₁ of vacuum metrics

We shall be considering certain vacuum solutions of the Einstein field equations for which there exist coordinates x^1 , x^2 , u, v such that the line element has the form

$$g_{ab}(u,v)dx^{a}dx^{b} + 2g_{uv}(u,v)du \, dv \quad (a,b=1,2) , \qquad (2.1)$$

 $\partial/\partial x^1$ and $\partial/\partial x^2$ are Killing vectors, g_{ab} has signature + +, $g_{uv} < 0$, and

$$p: = [g_{11}g_{22} - (g_{12})^2]^{1/2} > 0$$
(2.2)

over the domain of the chart which consists of all (x^1,x^2,u,v) such that $(x^1,x^2)\in\mathbb{R}^2$ and (u,v) is a member of a union of four contiguous regions

 $I \cup II \cup III \cup IV$,

where

$$I := \{(u,v) \in \mathbb{R}^2: u \leq 0, v \leq 0\},\$$

$$II := \{(u,v) \in \mathbb{R}^2: u \leq 0, 0 \leq v < 1\},\$$

$$III := \{(u,v) \in \mathbb{R}^2: 0 \leq u < 1, v < 0\},\$$

$$IV := \{(u,v) \in \mathbb{R}^2: 0 < \rho(u,v) \leq 1, 0 \leq u, 0 \leq v\}.$$
(2.3)

The intersection of these four regions of \mathbb{R}^2 is the singlet set whose element is the origin (u,v) = (0,0). The coordinates x^a have been scaled so that $\rho(0,0) = 1$.

One requires, of course, that the metric be continuous over the entire domain. In region I, it is assumed that g_{ab} and g_{uv} are uniform fields; i.e.,

$$g_{ab}(u,v) = g_{ab}(0,0), \quad g_{uv}(u,v) = g_{uv}(0,0),$$

$$\rho(u,v) = \rho(0,0) = 1 \quad \text{for all } (u,v) \in I.$$

In other words, the space-time domain in which $(u,v) \in I$ is a closed subregion of Minkowski space whose boundary consists of two null hypersurfaces that intersect at the Euclidean two-surface u = v = 0. One can always subject the ignorable coordinates x^a to an SL(2,R) transformation and the null coordinates u, v to a scaling transformation to make

$$g_{ab}(u,v) = \delta_{ab}, \quad g_{uv}(u,v) = -1 \quad \text{for all } (u,v) \in I.$$

(2.4)

This option will be kept open but not imposed now.

It is next assumed that the metric components in region II depend only on v, while the metric components in region III depend only on u. Thus

$$g_{ab}(u,v) = g_{ab}(0,v),$$

$$g_{uv}(u,v) = g_{uv}(0,v) \text{ for all } (u,v) \in \text{II},$$
(2.5)

and

 $g_{ab}(u,v) = g_{ab}(u,0),$

$$g_{uv}(u,v) = g_{uv}(u,0)$$
 for all $(u,v) \in III$. (2.6)

Thus regions II and III are each occupied by a *p*-*p* type N gravitational field, i.e., a plane-fronted gravitational wave. Consider the timelike coordinate $t = \frac{1}{2}(u + v)$ and spacelike coordinate $s = \frac{1}{2}(u - v)$. One defines "the future" as being in the direction of increasing *t*. For fixed t < 0, the two wave fronts are at s = t and at s = -t, respectively. As *t* increases and approaches 0, the spatial "interval" |2t| between the wave fronts decreases until they collide at t = 0.

We now introduce an assumption which holds only for a restricted set of p-p type N waves in II and III. So far, we have made no particular choice of the null coordinates u and v. We now assume that if we define u and v by the equations

$$\rho(0,v) = 1 - v^2 \quad \text{in II},
\rho(u,0) = 1 - u^2 \quad \text{in III},$$
(2.7)

then the metric components in these regions are C^2 -differentiable⁸ functions over the intervals $0 \le v < 1$ and $0 \le u < 1$, respectively. We shall adhere to the choices (2.7) in this paper.

We now come to the region occupied by the scattered

waves; viz., IV. The field equation which governs ρ is

$$\rho_{uv}:=\frac{\partial^2\rho}{\partial u\,\partial v}=0\,.$$

Hence, from Eqs. (2.7),

$$\rho = 1 - u^2 - v^2$$
 in IV. (2.8)

We next seek a C^2 -differentiable solution in IV of the vacuum field equations for $g_{ab}(u,v)$ and $g_{uv}(u,v)$. This must satisfy the continuity conditions for the metric at the bounding null hypersurface at u = 0 and v = 0, respectively; i.e., the solution must match the prescribed $g_{ab}(0,v)$, $g_{uv}(0,v)$ over the interval $0 \le v < 1$ and the prescribed $g_{ab}(u,0)$, $g_{uv}(u,0)$ over $0 \le u < 1$. Moreover, the solution must satisfy the requirement $g_{uv}(u,v) < 0$ at all (u,v) in IV. Such solutions do not exist in all cases.

Definition: CW_1 will denote the set of all vacuum metrics that satisfy all of the conditions prescribed above in Sec. II.

Our main interest in this paper will be the members of CW_1 for which colliding wave polarizations are not collinear. So far, every explicitly known vacuum metric regarded as representing the collision of two plane-fronted gravitational waves with noncollinear polarizations is a member of CW_1 .

B. The field equations

The field equations governing the metrical components g_{ab} can be reduced to the Ernst equation

$$F(d * \rho dE) = \rho dE(* dE), \qquad (2.9)$$

where E is the complex potential

$$E = F + i\omega := (g_{22})^{-1} (\rho + ig_{12}) ,$$

$$F := \operatorname{Re} E ,$$
(2.10)

and * is a two-dimensional duality operator such that

$$= du, \quad *dv = -dv \,. \tag{2.11}$$

In terms of u and v, Eq. (2.9) becomes

$$F[E_{uv} - \rho^{-1}(u E_v + v E_u)] = E_u E_v \qquad (2.12)$$

over the region IV.

*du

The only remaining independent field equations are those that can be used to compute g_{uv} when E is given:

$$2u\Gamma_{u} = 1 - \rho |E_{u}/2F|^{2},$$

2v $\Gamma_{v} = 1 - \rho |E_{v}/2F|^{2},$ (2.13)

where Γ is the real field defined by

$$-(\rho)^{1/2}g_{uv} = \exp(2\Gamma) . \qquad (2.14)$$

Equation (2.12) guarantees that the integrability condition for Eqs. (2.13) is satisfied. The solution of Eqs. (2.13) in an instructive form will be given later in Sec. II.

C. The field equations in terms of ${\mathscr S}$

The Ernst potential

$$\mathscr{E} = f + i\chi, \quad f := \operatorname{Re} \mathscr{E} := -g_{22}$$
 (2.15)

may be defined in terms of E by the equations

$$fF = -\rho, \quad f^{-1} d\chi = -F^{-1} * d\omega.$$
 (2.16)

The Ernst equation (2.9) guarantees that the integrability condition for the second Eq. (2.16) is satisfied. As is well known, \mathscr{C} is also a solution of the Ernst equation, and Eqs. (2.13) are expressible as

$$2u \gamma_{u} = 1 - \rho |\mathscr{C}_{u}/2f|^{2},$$

$$2v \gamma_{v} = 1 - \rho |\mathscr{C}_{v}/2f|^{2},$$
(2.17)

where γ is defined by

$$-g_{22}g_{uv} = \exp(2\gamma) . (2.18)$$

The reasons for introducing the potential \mathscr{C} as well as the potential E are to be found in various methods that have been developed for transforming given solutions of the Ernst equation into new solutions of the equation. In particular, as we shall discuss in detail in Sec. III, new colliding wave solutions with any desired finite number of parameters can be constructed by alternately subjecting E and \mathscr{C} to certain rational linear transformations. This is equivalent to a method originally devised by Geroch.⁶

D. The determination of the colliding wave solution by E(u,0) and E(0,v)

As is known from the theory of quasilinear hyperbolic equations such as Eq. (2.12), the solution E(u,v) in the scattered wave region is uniquely determined by its values E(u,0), E(0,v) on the respective null hypersurfaces which constitute the interfaces between the colliding waves and the scattered wave. In particular, the first partial derivatives on the interfaces; viz., $E_v(u,0)$, $E_u(0,v)$ are uniquely determined by E(u,0) and E(0,v), respectively. Specific equations which give these first partial derivatives in terms of E(u,0) and E(0,v) are of some interest and will now be derived.

The derivations proceed by first setting v = 0 in Eq. (2.12). The result is an ordinary differential equation that can be expressed in the form

$$\frac{d}{du}\left\{\ln\left[\frac{UE_{v}(u,0)}{F(u,0)}\right]\right\}=\frac{i\omega_{u}(u,0)}{F(u,0)},$$

where the capital letters U and V will be used to designate the non-negative valued fields

$$U:=(1-u^2)^{1/2}, \quad V:=(1-v^2)^{1/2}.$$
 (2.19)

By setting u = 0 in Eq. (2.12) a similar ordinary differential equation is obtained for $E_u(0,v)$. Integration of these ordinary differential equations gives us our final results,

$$\frac{UE_{v}(u,0)}{F(u,0)} = \frac{E_{v}(0,0)}{F(0,0)} \left\{ \exp \int_{0}^{u} da \left[\frac{i \,\omega_{a}(a,0)}{F(a,0)} \right] \right\},$$

$$\frac{VE_{u}(0,v)}{F(0,v)} = \frac{E_{u}(0,0)}{F(0,0)} \left\{ \exp \int_{0}^{v} db \left[\frac{i \,\omega_{b}(0,b)}{F(0,b)} \right] \right\}.$$
(2.20)

Equations similar to the above hold for \mathscr{C} ; these are simply obtained by replacing E by \mathscr{C} , F by f and ω by χ in Eqs. (2.20). Moreover, from Eq. (2.16) one obtains

$$\frac{\mathscr{C}_u}{f} + \frac{E_u}{F} = -\frac{2u}{\rho}, \quad \frac{\mathscr{C}_v}{f} + \frac{E_v *}{F} = -\frac{2v}{\rho}.$$
 (2.21)

Therefore,

$$\mathscr{E}_{u}(0,v)/f(0,v) = -E_{u}(0,v)/F(0,v),$$

$$\mathscr{E}_{v}(u,0)/f(u,0) = -E_{v}*(u,0)/F(u,0).$$

(2.22)

From Eqs. (2.20) and (2.22) we obtain the following relations, which will be used later:

$$k: = |E_{v}(0,0)/2F(0,0)|^{2} = |UE_{v}(u,0)/2F(u,0)|^{2}$$

$$= |U\mathscr{C}_{v}(u,0)/2f(u,0)|^{2},$$

$$l: = |E_{u}(0,0)/2F(0,0)|^{2} = |VE_{u}(0,v)/2F(0,v)|^{2}$$

$$= |V\mathscr{C}_{u}(0,v)/2f(0,v)|^{2}.$$

(2.23)

The particular case of colliding wave with *collinear* polarization is defined by the statement that there exists a choice of the ignorable coordinates x^a such that g_{12} vanishes at all points. In this case, $\omega = 0$ and one can select the arbitrary additive constants in χ so that $\chi = 0$ as well. Upon introducing the real scalar field ψ defined by

$$-\mathscr{E} = g_{22} = \exp(2\psi)$$
, (2.24)

one finds that the Ernst equations for E and for \mathscr{C} are each equivalent to the single linear equation

$$\psi_{uv} - \rho^{-1}(u\psi_v + v\psi_u) = 0. \qquad (2.25)$$

Moreover, Eqs. (2.23) become

$$k = [\psi_v(0,0)]^2 = [U\psi_v(u,0)]^2,$$

$$l = [\psi_u(0,0)]^2 = [V\psi_u(0,v)]^2.$$
(2.26)

E. A necessary and sufficient condition for a solution of the Ernst equation to yield a member of CW_1

Let E by any solution of the Ernst equation over the domain

 $\{(u,v)\in\mathbb{R}^2: 0\leq u^2+v^2<1, 0\leq u, 0\leq v\},\$

such that E is of differentiable class C^2 and such that $F(u,v) := \operatorname{Re} E(u,v) > 0$ for all (u,v) in the domain. From the Ernst equation (2.12), one can readily prove that

$$\frac{\partial}{\partial u} \left(\rho F^{-2} |E_v|^2 \right) = v F^{-2} (E_u E_v * + E_u * E_v) ,$$

$$\frac{\partial}{\partial v} \left(\rho F^{-2} |E_u|^2 \right) = u F^{-2} (E_u E_v * + E_u * E_v) .$$
(2.27)

Now let us introduce the scalar field

$$\Lambda(u,v) := \frac{1}{4} \int_0^u da \int_0^v db \, [F(a,b)]^{-2} [E_a(a,b)E_b * (a,b) + E_a * (a,b)E_b(a,b)] .$$
(2.28)

Then, from Eqs. (2.23) and (2.27), we obtain

$$\frac{1}{4}\rho F^{-2}|E_{v}|^{2}-l=v\Lambda_{v},$$

$$\frac{1}{4}\rho F^{-2}|E_{u}|^{2}-k=u\Lambda_{u}$$

Substitution of the above into Eqs. (2.13) yields

$$\Gamma_u = (1-k)/u - \Lambda_u, \quad \Gamma_v = (1-l)/v - \Lambda_v.$$

Hence, from Eqs. (2.14), one obtains the following result for g_{uv} :

$$f_{uv} = -(\text{const})(\rho)^{-1/2} u^{1-k} v^{1-l} e^{-2\Lambda}.$$
 (2.29)

The positive constant factor in g_{uv} is arbitrary unless one imposes an auxiliary condition to fix it.

It is apparent from Eq. (2.29) that not every E yields a g_{uv} which fits our definition of the class CW₁ of colliding wave solutions. The following key theorem results.

Theorem: E determines a metric in CW₁ by the relation

$$E = (g_{22})^{-1}(\rho + ig_{12})$$

if and only if

$$k = l = 1. (2.30)$$

Let us consider examples.

F. Some old examples of Ernst potentials that yield colliding wave solutions in CW₁ and other examples that fail in that respect

An Ernst potential which satisfies the conditions k = l = 1 is the Kerr metric \mathscr{C} which is given by

$$-\mathscr{C}^* = (1-\xi)(1+\xi)^{-1}, \qquad (2.31)$$

where

$$\begin{aligned} \xi &:= px + iqy, \\ x &:= uV + vU, \quad y = uV - vU, \end{aligned} \tag{2.32}$$

and where the real parameters q and p are related by

$$p^2 + q^2 = 1. (2.33)$$

The colliding wave solution corresponding to this Ernst potential wave analyzed in depth by Chandrasekhar and Xanthopoulos.⁴

Another example is the Nutku-Halil solution,² for which Chandrasekhar and Ferrari⁹ observed that

$$E = (1 - \xi)(1 + \xi)^{-1}, \qquad (2.34)$$

which equals the Kerr $-\mathscr{C}^*$ given by Eq. (2.31).

The potential given by Eqs. (2.34) is one member of a larger family of Ernst equation solutions called the *Tomimatsu-Sato* (TS) family¹⁰ and which involves an additional real parameter δ . $\delta = 0$ yields Minkowski space; $\delta = 1$ yields the potential (2.34); and $\delta = -1$ yields (2.34) with ξ replaced by $-\xi$. The fact is that the only members of the TS family for which k = l = 1 are those for which $\delta^2 = 1$.

The TS family Ernst potentials are very complicated. Let us, therefore, look at the Zipoy–Voorhees^{11,12} (ZV) family, which is the collinear version of the TS family. For the ZV family,

$$\psi = \frac{\delta}{2} \ln \frac{1+x}{1-x}$$
 or $\psi = \frac{\delta}{2} \ln \frac{1-y}{1-y}$. (2.35)

A brief calculation employing Eqs. (2.26) and (2.32) yields

$$k=l=\delta^2.$$

Thus the only members of the ZV family in CW_1 are the Khan–Penrose¹³ (collinear specialization of Nutku–Halil) and Schwarzschild solutions, for which

$$E = \exp(2\psi)$$
 and $\mathscr{C} = -\exp(2\psi)$

respectively, where ψ is given by Eqs. (2.35) with $\delta = \pm 1$.

As an example which does not satisfy the colliding wave condition, consider the Kasner solutions, for which

$$\mathscr{C}=-\rho^{1-n}=-g_{22}.$$

It is clear that k = l = 0. In fact, any Ernst potential that is expressible as a function of the Weyl canonical coordinates

$$z = u^2 - v^2, \quad \rho = 1 - u^2 - v^2$$
 (2.36)

and is an analytic function of z and ρ in a neighborhood of $(z,\rho) = (0,1)$ satisfies k = l = 0 and cannot, therefore, yield a member of CW₁. The reader can easily prove this assertion.

G. Recent examples of members of CW₁: A new threeparameter family of colliding wave solutions

The Nutku-Halil solution is itself a member of a twoparameter family of noncollinear members of CW_1 which has been obtained by Ferrari, Ibañez, and Bruni.⁵ In paper I we reported our discovery of a three-parameter extension of their family, the derivation of which will be given in a sequel to the present paper. The Ernst potentials E and \mathscr{C} of our three-parameter family are given by

$$E(n,\nu,\nu') = \rho^n T(n+1,\nu',\nu) / T(n-1,\nu',\nu) , \qquad (2.37)$$

$$\mathscr{C}(n,\nu,\nu') = \rho^{1-n} [T(n-2,\nu,\nu')/T(n,\nu,\nu')]^*, \qquad (2.38)$$

where n, v, v' are any real numbers, where we let

$$p_{i} = \cos v, \quad q_{i} = \sin v, p' = \cos v', \quad q'_{i} = \sin v',$$
(2.39)

and where

$$T(n,v,v') := \frac{1}{2} X \left[(p+p') \left(\frac{1-x}{1+x}\right)^{n/2} + (p-p') \left(\frac{1+x}{1-x}\right)^{n/2} \right] + \frac{1}{2} Y \left[(q+q') \left(\frac{1-y}{1+y}\right)^{n/2} + (q-q') \left(\frac{1+y}{1-y}\right)^{n/2} \right], \qquad (2.40)$$

where x and y were defined by Eqs. (2.32), and

$$X:=(1-x^2)^{1/2}, \quad Y:=(1-y^2)^{1/2}.$$
 (2.41)

When q' = 0 and p' = 1, i.e., v' = 0, the above family reduces to the one found by Ferrari *et al.*, except that we should caution that their parameter *p* differs from ours.

When n = 0 in Eqs. (2.37), (2.38), and (2.40), one obtains the Nutku-Halil solution. The parameter v' is not essential for the Nutku-Halil solution, since it can be removed by an SL(2,R) transformation of the coordinates x^a . However, both v and v' are essential parameters for all $n \neq 0$.

When |n| = 1 in Eqs. (2.37), (2.38), and (2.40), one obtains the Kerr-NUT family, except that we should warn the reader that it is v which is the NUT parameter in this case. (Therefore the n = 1 member of the solutions of Ferrari *et al.* is not the Kerr metric; it is the Schwarzschild-NUT metric.)

Finally, when n = 2 and v' = 0 in Eqs. (2.37), (2.38), and (2.40), one obtains a solution reported in paper I. When n = 2, one obtains a specialized version of a three-parameter family which we shall give in Sec. IV.

In evaluating the metric tensor it is convenient to express Eq. (2.37) in the form

$$E = AB^{-1}, (2.42)$$

where

$$A = \rho^{n(n+2)/4} T(n+1,\nu',\nu) ,$$

$$B = \rho^{n(n-2)/4} T(n-1,\nu',\nu) ,$$
(2.43)

and T(n,v,v') is defined by Eq. (2.40).

The metric tensor components g_{ab} , g_{uv} are computed from Eqs. (3.7) and (3.9) in paper I. The components g_{ab} in region IV are there expressed in the form

$$g_{11} = \rho K / N, \quad g_{12} = \rho L / N, \quad g_{22} = \rho M / N.$$
 (2.44)

We obtain

 $L = \operatorname{Im}(AB^*)$

$$N = \operatorname{Re} (AB^{*})$$

$$= \rho^{n^{2}/2} \frac{1}{4} \left\{ X^{2} \left[(p+p')^{2} \left(\frac{1-x}{1+x} \right)^{n} + (p-p')^{2} \left(\frac{1+x}{1-x} \right)^{n} \right] + Y^{2} \left[(q+q')^{2} \left(\frac{1-y}{1+y} \right)^{n} + (q-q')^{2} \left(\frac{1+y}{1-y} \right)^{n} \right] + 2(q^{2}-q'^{2})(x^{2}-y^{2}) \right\}$$

$$= \rho^{n^{2}/2} |T(n + y)|^{2} \qquad (2.45)$$

$$= p ||T(n,v,v)||,$$

$$K = |A|^2 = o^{(n^2/2) + n} |T(n+1,v',v)|^2.$$
(2.46)

$$M = |R|^{2} - o^{(n^{2}/2) - n} |T(n - 1v', v)|^{2}, \qquad (2.47)$$

$$M = |B|^{2} = \rho^{(n+1)} ||||(n-1, v, v)||^{2}, \qquad (2.47)$$

$$= \rho^{n^{2}/2} \frac{1}{2} \left\{ (x-y) \left[(p+p')(q+q') \left(\frac{1-x}{1+x} \right)^{n/2} \right. \\ \left. \times \left(\frac{1-y}{1+y} \right)^{n/2} - (p-p')(q-q') \left(\frac{1+x}{1-x} \right)^{n/2} \right. \\ \left. \times \left(\frac{1+y}{1-y} \right)^{n/2} \right] + (x+y) \left[(p-p')(q+q') \right. \\ \left. \times \left(\frac{1+x}{1-x} \right)^{n/2} \left(\frac{1-y}{1+y} \right)^{n/2} - (p+p')(q-q') \right. \\ \left. \times \left(\frac{1-x}{1+x} \right)^{n/2} \left(\frac{1+y}{1-y} \right)^{n/2} \right] \right\}.$$
(2.48)

To obtain the explicit expression for $|T(n + 1, v', v)|^2$ in Eq. (2.46), simply replace *n* by n + 1 and interchange *p* with p'

and q with q' in that expression for $|T(n,v,v')|^2$, which is given by Eq. (2.45). A like rule yields $|T(n-1,v',v)|^2$ in Eq. (2.47).

The remaining metrical component g_{uv} is

$$g_{uv} = -\rho^{-1/2} N / (UV) ,$$
 (2.49)

where N is given by Eq. (2.45).

Interesting relations between members of the family corresponding to different n are given in Secs. III and IV.

III. A REALIZATION OF A GEROCH GROUP

A. Objectives

Regarding the set of metrics CW_1 , our ultimate goal is to find an effective way of constructing an exact solution for the metric in the scattered wave region IV when the metrics of the two plane-fronted waves in regions II and III are given. Equivalently, we would like to be able to obtain $\mathscr{C}(u,v)$ when $\mathscr{C}(0,v)$ and $\mathscr{C}(u,0)$ are given.

If the polarizations of the *p*-*p* waves are collinear, a general solution of this problem is obtained by applying Riemann's method to the hyperbolic linear equation (2.25), and this has already been done under varying assumptions by Szekeres¹⁶ and by Xanthopoulos.¹⁷ When the polarizations are not collinear, the quasilinear Ernst equation cannot be converted by any known way to an equivalent linear partial differential equation involving only a finite number of dependent and independent variables. One must seek a different approach.

The approach which we are seeking is a generalization of methods that several people, including two of the present authors, have applied to stationary axisymmetric gravitational fields in the Einstein theory. There, the problem is to determine the Ernst potential $\mathscr{C}(z,\rho)$ when its values $\mathscr{C}(z,0)$ on the axis are given.

In the particular method developed by Hauser and Ernst,¹⁸ a 2×2 matrix function u(t) of a complex variable t is constructed by algebraic means from $\mathscr{C}(z,0)$. The function u(t) is an element in a realization of a group central to our entire discussion. This group was first effectively formulated and applied by Kinnersley and Chitre¹⁵ and is an extension of one originally conceived by Geroch.⁶

The method continues with the use of u(t) to construct the kernel of a Fredholm equation of the second kind or an equivalent homogeneous Hilbert problem whose solution is a 2×2 matrix function $F(t,z,\rho)$. The Ernst potential $\mathscr{C}(z,\rho)$ is one of the matrix elements of $\partial F/\partial t$ at t = 0. [Also, the elements of the real part of $\partial F/\partial t$ at t = 0 are the negatives of the metric components $g_{ab}(z,\rho)$.]

Thus the problem of determining $\mathscr{C}(z,\rho)$ from $\mathscr{C}(z,0)$ is reduced to the (not necessarily easy) problem of solving a linear integral equation or homogeneous Hilbert problem. This method may be described as a synthesis of a group theoretical theme of Geroch and a type of inverse scattering technique. We wish to formulate a similar synthesis for the colliding wave problem.

To approach the ultimate goal we have described above, we must first go through several stages. In this section we shall present the first stage which is to provide a realization of the original Geroch group, specifically one suitable for our present purposes. The Geroch group is infinite dimensional and can be used to construct colliding wave solutions with any desired finite number of parameters. In fact, we shall give a simple example of the way in which the group realization is used to build a new solution out of a given known one. However, we stress that our central objective here is to provide the first link in a conceptual chain. We are not concerned in this paper with providing any effective modes of calculation; that will come in sequels.

B. A realization of SL(2,*R*) that induces rational linear transformations of *E*

The Geroch group is constructed by taking the free product of two groups of transformations, both of which are realizations of SL(2,R), and both of which transform a vacuum metric with a line elment of the form (2.1) into the same vacuum metric or into another one with a line element of the form (2.1). In this subsection we shall describe one of these realizations of SL(2,R).

We start by considering the group of SL(2,R) transformations of the ignorable coordinates, or, equivalently, the set of transformations

$$\begin{pmatrix} \frac{\partial}{\partial x^{1'}} \\ \frac{\partial}{\partial x^{2'}} \end{pmatrix} = w \begin{pmatrix} \frac{\partial}{\partial x^1} \\ \frac{\partial}{\partial x^2} \end{pmatrix}, \qquad (3.1)$$

where w is any member of SL(2,R),

$$w = w^* = \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix}, \quad \alpha \delta - \beta \gamma = 1.$$
 (3.2)

Correspondingly, the metric component g_{uv} remains invariant while the metric components g_{ab} undergo the transformation into

$$h' = whw^T$$
, where $h: = \begin{pmatrix} g_{11} & g_{12} \\ g_{21} & g_{22} \end{pmatrix}$. (3.3)

As a result, E as defined by Eq. (2.10) becomes

$$E' = i(\alpha E + i\beta)(\gamma E + i\delta)^{-1}, \qquad (3.4)$$

where

$$E' := (g'_{22})^{-1}(\rho + ig'_{12})$$

Thus we obtain, in general, a new solution of the Ernst equation, though, of course, the transformation has not changed the physical field; i.e., the transformation has not altered the value of the line element at any point of the manifold. It is the same space-time.

What about the way \mathscr{C} transforms under this group? From the defining relations (2.6) between E nd \mathscr{C} , one deduces that

$$\mathscr{C}'(u,v) = -\left[\rho/F'(u,v)\right] + i\chi'(0,0) + i\left\{\int_0^u da \left[\frac{(1-a^2)\omega_a'(a,0)}{F'(a,0)^2}\right] + \int_0^v db \left[\frac{(1-u^2-b^2)\omega_b'(u,b)}{F'(u,b)^2}\right]\right\}, \quad (3.5)$$

where $F' = \operatorname{Re} E'$ and $\omega' = \operatorname{Im} E'$ are given by Eq. (3.4) and where $\chi'(0,0)$ is arbitrary. Thus the transformation of \mathscr{C} is nonlocal. Incidentally, we shall henceforth adopt the convention

$$\chi'(0,0) = \chi(0,0) \tag{3.6}$$

in Eq. (3.5). [However, $\omega'(0,0)$ does not generally equal $\omega(0,0)$ in Eq. (3.4).]

Before we proceed with our discussion of the group, let us pause to introduce two helpful notations.

Definition: Σ will denote the set of all ordered pairs (E,\mathscr{C}) of C^2 -differentiable solutions of the Ernst equation such that (1) region IV is the domain of E and \mathscr{C} ; (2) F(u,v): = Re E(u,v) > 0 and f(u,v): = Re $\mathscr{C}(u,v) < 0$ for all (u,v) in IV; (3) the relations (2.6), where $E = F + i\omega$ and $\mathscr{C} = f + i\chi$, hold throughout IV.

For each (E,\mathscr{C}) in Σ , there is exactly one metric (2.1) such that

 $E = (g_{22})^{-1}(\rho + ig_{12}) .$

Definition: For each w in SL(2,R), let $\gamma_1(w)$ denote that mapping whose domain is Σ such that

$$\gamma_1(w)(E,\mathscr{E}) = (E',\mathscr{E}'), \qquad (3.7)$$

where E' is given by Eqs. (3.2) and (3.4) and where \mathscr{C}' is given by Eqs. (3.5) and (3.6).

We shall also find it helpful to use the mnemonic symbols

 $\gamma_1(w)E$ and $\gamma_1(w)\mathscr{C}$

to stand for the first member E' and the second member \mathscr{C}' in the ordered pair (3.7).

Let us look at two examples of $\gamma_1(w)$. A much used set of generators of SL(2,R) consists of its elements

$$w_{\Delta} := \begin{pmatrix} 1 & \Delta \\ 0 & 1 \end{pmatrix}$$
 for all $\Delta \in \mathbb{R}^{1}$, (3.8a)

$$\boldsymbol{\epsilon} := \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}. \tag{3.8b}$$

From the definition of $\gamma_1(w)$ given by Eqs. (3.7), (3.4), (3.2), (3.5), and (3.6), one can see that

$$\gamma_1(w_{\Delta})(E,\mathscr{C}) = (E + i\Delta,\mathscr{C}), \qquad (3.9a)$$

$$\gamma_1(\epsilon)(E,\mathscr{E})(E^{-1},\mathscr{E}') . \tag{3.9b}$$

As regards $\mathscr{C}' = \gamma_1(\epsilon) \mathscr{C}$ in Eqs. (3.9b), the expression for it is not trivial and is not required here. Note, by the way, that

 $\gamma_1(\pm I)(E,\mathscr{C}) = (E,\mathscr{C});$

i.e., $\gamma_1(\pm I)$ is the identity mapping on Σ .

In the following theorem, \circ denotes the mapping composition symbol, and the factors in a composition are read from right to left.

Theorem: (1) $\gamma_1(w)$ maps Σ onto Σ and is one-to-one. (2) For all w and v in SL(2,R),

$$\gamma_1(w) \circ \gamma_1(v) = \gamma_1(wv) . \tag{3.10}$$

(3)
$$F^{-2}(dE \otimes dE^*)$$
 is invariant under $\gamma_1(w)$. (3.11)

We shall not supply any details of the simple proof of the above well-known statements. Equation (3.11) is most easily proved by showing that it is true for $w = w_{\Delta}$ and ϵ . Note that $|E_u|^2$ and $|E_v|^2$ are components of the tensor product in Eq. (3.11). Therefore, the following corollary is implied by Eqs. (2.23) and (3.11).

Corollary: The colliding wave condition k = l = 1 is preserved by $\gamma_1(w)$.

 $\gamma_1(w)$ is our first realization of SL(2,R). We next consider the second realization of SL(2,R) which will be used to construct a realization of the Geroch group.

C. A realization of SL(2, $\ensuremath{\mathcal{R}}$) that induces rational linear transformations of $\ensuremath{\mathscr{C}}$

Let us start by inspecting the simple transformation

$$\mathscr{E} \to \mathscr{E} + i\Delta . \tag{3.12}$$

 $\mathscr{C} + i\Delta$ is also a solution of the Ernst equation. Note that $E \rightarrow E$ under this transformation.

Next consider

$$\mathscr{E} \to \mathscr{E}^{-1} \,. \tag{3.13}$$

The reader can easily verify that $\mathscr{C}' = \mathscr{C}^{-1}$ is also a solution of the Ernst equation such that Re $\mathscr{C}' < 0$. This is the transformation which was applied to the Nutku-Halil \mathscr{C} potential in paper I in order to construct a new solution. It is the first transformation, among those explicitly considered so far in Sec. III, which actually changes the space time into a new one.

Repeated alternate applications of the transformations (3.12) and (3.13), with different values of the parameters Δ , generate the group of rational linear transformations

$$\mathscr{C} \to \mathscr{C}'' = i(\alpha \mathscr{C} + i\beta)(\gamma \mathscr{C} + i\delta)^{-1},$$
 (3.14)

where $\alpha\delta - \beta\gamma = 1$. It is apparent at this point that much of what we did in Sec. III B is applicable when E and \mathscr{C} are interchanged. Therefore, we proceed onwards to a formal development that parallels what we did in Sec. III B.

Definition: For each w in SL(2,R), let $\gamma_2(w)$ denote that mapping whose domain is Σ such that

$$\gamma_2(w)(E,\mathscr{E}) = (E'',\mathscr{E}''), \qquad (3.15)$$

where \mathscr{C}'' is given by Eq. (3.14) and where

$$E''(u,v) = -\left[\rho/f''(u,v)\right] + iw(0,0) + i\left\{\int_0^u da\left[\frac{(1-a^2)\chi_a''(a,0)}{f''(a,0)^2}\right] + \int_0^v db\left[\frac{(1-u^2-b^2)\chi_b''(u,b)}{f''(u,b)^2}\right]\right\}.$$
 (3.16)

 $f'' = \operatorname{Re} \mathscr{C}''$ and $\chi'' = \operatorname{Im} \mathscr{C}''$ are given by Eq. (3.14). We have fixed the constant of integration in E'' by the requirement that

 $\omega''(0,0) = \omega(0,0)$.

The reader should compare Eqs. (3.14) and (3.16) with Eqs. (3.4)-(3.6) and with Eq. (3.7).

We shall find it helpful to use the *mnemonic symbols* $\gamma_2(w)E$ and $\gamma_2(w)\mathscr{C}$ to stand for the first member E'' and the second member \mathscr{C}'' in the ordered pair (3.15).

The transformations (3.12) and (3.13) are now expressible as

$$\gamma_2(w_{\Delta})(E,\mathscr{C}) = (E,\mathscr{C} + i\Delta), \qquad (3.17a)$$

$$\gamma_2(\epsilon)(E,\mathscr{E}) = (E'',\mathscr{E}^{-1}), \qquad (3.17b)$$

which should be compared with Eqs. (3.9). Furthermore,

we have the following theorem and corollary corresponding to the ones given in Sec. III B.

Theorem: (1) $\gamma_2(w)$ maps Σ onto Σ and is one-to-one. (2) For all w and v in SL(2,R),

$$\gamma_2(w) \circ \gamma_2(v) = \gamma_2(wv) . \tag{3.18}$$

$$(3) f^{-2}(d\mathscr{C} \otimes d\mathscr{C}^*) \tag{3.19}$$

is invariant under $\gamma_2(w)$.

Corollary: The colliding wave condition k = l = 1 is preserved by $\gamma_2(w)$.

We now have two realizations of SL(2,R); viz., $\gamma_1(w)$ and $\gamma_2(w)$. The key idea of Geroch, when expressed in the language of our SL(2,R) realizations, was to construct a new group by alternate applications of $\gamma_1(v)$ and $\gamma_2(w)$ with different choices of v and w in successive transformations. This creates a nontrivial infinite-dimensional group.

D. The realizations $G_0(\Sigma)$ and $G_0(\Sigma_1)$ of the Geroch group

Definition: Let $G_0(\Sigma)$ denote the set of all compositions of the form

$$\gamma_{r_m}(w_m) \circ \cdots \circ \gamma_{r_1}(w_1)$$

for all choices of $m \ge 1$, the SL(2,R) members $w_1, ..., w_m$ and the subscript values $r_1, r_2, ..., r_m = 1$ or 2 (taken in any order).

The following theorem is an immediate consequence of the ones presented in Secs. III B and III C.

Theorem: $G_0(\Sigma)$ is a group of one-to-one mappings of Σ onto Σ .

Definition: Let Σ_1 denote the set of all members of Σ that satisfy the colliding wave condition k = l = 1 as defined by Eqs. (2.23).

Definition: Let $G_0(\Sigma_1)$ be the set of restrictions to Σ_1 of all members of $G_0(\Sigma)$.

It follows from the preceding theorem and the corollaries in Secs. III B and III C that $G_0(\Sigma_1)$ is a group of oneto-one mappings of Σ_1 onto Σ_1 .

E. An example of the generation of a new colliding wave solution by employing members of $G_0(\Sigma_1)$

We shall start with the Nutku–Halil solution. The Ernst potential $E^{\rm NH} = E_0$ was given by Eq. (2.34) and $\mathscr{C}^{\rm NH}$ is given by

$$\mathscr{C}_{0} = XY - 2(pY - iqX - xY)(pX - iqY)^{-1}, \qquad (3.20)$$

where x, y, X, Y are defined by Eqs. (2.32) and (2.41). We want to obtain

$$(E_2,\mathscr{C}_2):=\gamma_2(\epsilon)\circ\gamma_2(w_{\Delta})\circ\gamma_1(w)(E_0,\mathscr{C}_0),\qquad(3.21)$$

where w_{Δ} and ϵ are defined by Eqs. (3.8), and w is any member of SO(2,R), which we parametrize as follows:

$$w = \begin{pmatrix} \cos(\nu'/2) & \sin(\nu'/2) \\ -\sin(\nu'/2) & \cos(\nu'/2) \end{pmatrix}.$$
 (3.22)

Let

$$(E'_{0}, \mathscr{E}'_{0}) := \gamma_{1}(w) (E_{0}, \mathscr{E}_{0}) . \qquad (3.23)$$

Then Eqs. (3.4), (3.7), and (3.22) yield

$$E'_{0} = \gamma_{1}(w)E_{0} = (1 - \xi')(1 + \xi')^{-1}, \qquad (3.24)$$

where

$$\xi' := (p' - iq')\xi$$

and

 $p':=\cos\nu', \quad q':=\sin\nu'.$ (3.25)

The corresponding \mathscr{C} potential is

$$\mathscr{C}_{0}' = \gamma_{1}(w) \mathscr{C}_{0}$$

= $XY - 2[pY - iqX - (p'x - iq'y)Y]$
 $\times (pX - iqY)^{-1}.$ (3.26)

We shall comment on the mode of calculation later.

Next, from Eq. (3.17a) we obtain

$$\gamma_2(w_{\Delta})(E'_0,\mathscr{E}'_0) = (E'_0,\mathscr{E}'_0 + i\Delta), \qquad (3.27)$$

where E'_0 and \mathscr{C}'_0 are given by Eqs. (3.24)–(3.26). It is easy to see that $\gamma_2(w_{\Delta})E'_0 = E'_0$ since adding an imaginary constant to \mathscr{C}'_0 does not alter the metric components. That does not mean, however, that this imaginary constant will not show up in the metric after further transformations. In fact, from Eqs. (3.17b) and (3.18),

$$\gamma_{2}(\epsilon)(E_{0}',\mathscr{E}_{0}'+i\Delta) = (\gamma_{2}(\epsilon)E_{0}',(\mathscr{E}_{0}'+i\Delta)^{-1})$$
$$= (E_{2},\mathscr{E}_{2}), \qquad (3.28)$$

where we have combined (3.24), (3.27), and (3.28) to give us (E_2, \mathscr{C}_2) as defined by Eq. (3.21). The computation of $E_2 = \gamma_2(\epsilon) E'_0$ is not trivial. The result is

$$E_2 = AB^{-1}, (3.29)$$

where

$$B = (p' + iq') - (px + iqy), \qquad (3.30)$$

$$A = -B\rho^{2} + 2[(p' - 2px + p'x^{2})Y^{2} + i(q' - 2qy + q'y^{2})X^{2}] + 2i\Delta[py(x^{2} - 2) + iqx(y^{2} - 2) + xy(p' + iq')] + \Delta^{2}(p' + iq' + px + iqy). \qquad (3.31)$$

When $\Delta = 0$, q' = 0, and p' = 1, the above result reduces to the solution given by Eqs. (3.7)-(3.9) in paper I. When $\Delta = 0$, the result reduces to the n = 2 solution given by Eqs. (2.37)-(2.41) in the present paper.

For $\Delta = 0$, q' = 0, and p' = 1, we computed the results given above by several means, including direct integration, as in Eqs. (3.5) and (3.16). However, to obtain the solution with all parameters, we employed a more powerful and efficient method which involves solving a homogeneous Riemann-Hilbert problem to effect the transformation $(E_0, \mathscr{C}_0) \rightarrow (E'_0, \mathscr{C}'_0)$. This method will be detailed in a subsequent paper.

From the A and B given by Eqs. (3.30) and (3.31), both g_{ab} and g_{uv} can be computed by simple algebraic means with the aid of equations given by Ernst.¹⁹ Also, one can employ Eqs. (3.7) and (3.9) in paper I.

Of course, g_{ab} can always be algebraically computed from E by using the definition (2.10) of E.

As regards the computation of g_{uv} , we shall now see that there is a striking shortcut that is generally applicable to the transformation of a metric induced by $\gamma_r(w)$ for any w in SL(2,R).

F. The transformations of g_{uv} induced by $\gamma_1(w)$ and $\gamma_2(w)$

First consider $\gamma_1(w)$. Since Eq. (3.11) implies that $|E_u|^2 F^{-2}$ and $|E_v|^2 F^{-2}$ are invariant under $\gamma_1(w)$, Eqs. (2.13) and (2.14) imply that

$$g'_{uv} = g_{uv} , \qquad (3.32)$$

where $(E', \mathscr{C}') = \gamma_1(w)(E, \mathscr{C})$. This is to be expected, since $E \rightarrow \gamma_1(w)E$ is equivalent to an SL(2,R) transformation of the ignorable coordinates.

Consider $\gamma_2(w)$. In this case, it is $|E_u|^2 f^{-2}$ and $|\mathscr{C}_v|^2 f^{-2}$ that are invariant, since $\mathscr{C} \to \gamma_2(w) \mathscr{C}$. Hence, from Eqs. (2.17) and (2.18),

$$g_{22}' g_{uv}' = g_{22} g_{uv} , \qquad (3.33)$$

where $(E', \mathscr{E}') = \gamma_2(w)(E, \mathscr{E})$.

G. Some interesting relations involving $\gamma_1(\epsilon)$ and $\gamma_2(\epsilon)$

For the Kasner metrics of index n,

$$E = E_n = \rho^n, \quad \mathscr{E} = \mathscr{E}_n = -\rho^{1-n}. \tag{3.34}$$

Therefore

$$\gamma_{1}(\epsilon)(E_{n},\mathscr{C}_{n}) = (E_{-n},\mathscr{C}_{-n}),$$

$$\gamma_{2}(\epsilon)(E_{n},\mathscr{C}_{n}) = (E_{-n+2},\mathscr{C}_{-n+2}),$$

$$\gamma_{2}(\epsilon)\circ\gamma_{1}(\epsilon)(E_{n},\mathscr{C}_{n}) = (E_{n+2},\mathscr{C}_{n+2}).$$

(3.35)

The interesting thing about the above relations is that they essentially carry over to other families of metrics that are obtained by applying a significant class of Kinnersley– Chitre transformations to the Kasner metrics.

For example, consider the family given by Eqs. (2.37)–(2.41). From Eqs. (2.39) and (2.40),

$$T(-n,v,v' + \pi) = T(n,v,v'),$$

$$T(n,v,v' \pm 2\pi) = T(n,v,v'),$$

$$T(n,v + \pi,v' + \pi) = -T(n,v,v').$$

(3.36)

Therefore, from Eqs. (2.37) and (2.38),

$$\gamma_{1}(\epsilon)(E(n,v,v'),\mathscr{E}(n,v,v')) = (E(-n,v+\pi,v'),\mathscr{E}(-n,v+\pi,v')), \qquad (3.37a)$$

$$\gamma_{2}(\epsilon)(E(n,v,v'),\mathscr{E}(n,v,v'))$$

$$= (E(-n+2,\nu,\nu'+\pi),\mathscr{E}(-n+2,\nu,\nu'+\pi)),$$
(3.37b)

$$\gamma_{2}(\epsilon) \circ \gamma_{1}(\epsilon) (E(n, \nu, \nu'), \mathscr{E}(n, \nu, \nu'))$$

= $(E(n+2, \nu, \nu'), \mathscr{E}(n+2, \nu, \nu')).$ (3.37c)

The general theory behind these remarkable results will be given in a subsequent paper.

IV. AUGMENTATION OF THE GEROCH GROUP BY THE KRAMER-NEUGEBAUER INVOLUTION

A. The augmented Geroch group realization $G_1(\Sigma)$

The set $G_0(\Sigma)$ does not contain two obvious idempotent mappings of Σ onto Σ which we shall denote by γ_{cc} and γ_{KN} and which are defined by

$$\gamma_{cc}(E,\mathscr{C}) := (E^*,\mathscr{C}^*),$$

$$\gamma_{KN}(E,\mathscr{C}) := (-\mathscr{C}^*, -E^*).$$
(4.1)

Note that $(\gamma_{cc})^2 = (\gamma_{KN})^2$ is the identity mapping on Σ , and

$$\gamma_{cc}\circ\gamma_{\mathrm{KN}}(E,\mathscr{C})=(-\mathscr{C},-E).$$

Here γ_{KN} will be called the *Kramer-Neugebauer involution* and it is central to the discussion in Sec. IV.

Definition: $G_1(\Sigma)$ will denote the group of all compositions

 $f_m \circ \cdots \circ f_1$,

for all $m \ge 1$, such that f_1, \dots, f_m are any members of

 $G_0(\Sigma) \cup \{\gamma_{cc}, \gamma_{KN}\}.$

Since Eqs. (2.2) imply that the colliding wave condition k = l = 1 is preserved by both γ_{cc} and γ_{KN} , we obtain the following theorem.

Theorem: γ_{cc} and γ_{KN} map Σ_1 onto Σ_1 ; i.e., $G_1(\Sigma_1)$ is a group of one-to-one mappings Σ_1 onto Σ_1 .

B. The transformation of g_{uv} induced by γ_{KN}

A relation we shall now derive supplies the new g_{uv} resulting from a Kramer-Neugebauer involution *in terms of the old* g_{22} and g_{uv} . Thus it resembles Eq. (3.33), which does the same for the transformation $\gamma_2(w)$.

We start with Eqs. (2.21), from which we derive

$$\frac{1}{u} \left[1 - \frac{\rho |\mathscr{E}_u|^2}{4f^2} \right] = \frac{1}{u} \left[1 - \frac{\rho |E_u|^2}{4F^2} \right] - \frac{u}{\rho} - \frac{F_u}{F}.$$
(4.2)

From Eqs. (2.17), (2.18), and (4.2) we obtain

$$(g'_{22})^{1/2}g'_{uv} = (g_{22})^{1/2}g_{uv}, \qquad (4.3)$$

where $(E',\mathscr{E}') = \gamma_{KN}(E,\mathscr{E})$. However, $g'_{22} = \rho(g_{22})^{-1}$. Therefore

$$g'_{uv} = \rho^{-1/2} g_{22} g_{uv} , \qquad (4.4)$$

which is our final result.

C. The effects of γ_{KN} on the Kasner index *n*

From the expressions for E_n and \mathscr{C}_n given by Eqs. (3.34) for the Kasner family and from Eqs. (3.35),

$$\gamma_{\mathrm{KN}}(E_n,\mathscr{C}_n) = (E_{-n+1},\mathscr{C}_{-n+1}),$$

$$\gamma_{\mathrm{KN}}\circ\gamma_1(\epsilon)(E_n,\mathscr{C}_n) = (E_{n+1},\mathscr{C}_{n+1}).$$
(4.5)

This effect on n carries over to other families which are obtained from the Kasner one by Kinnersley–Chitre transformations. As an example, for the family given by Eqs. (2.37)– (2.41), with the aid of Eqs. (3.36) and (3.37) one finds

$$\gamma_{\rm KN}(E(n,v,v'),\mathscr{E}(n,v,v')) = (E(-n+1,v,v'+\pi),\mathscr{E}(-n+1,v,v'+\pi)), \gamma_{\rm KN} \circ \gamma_1(\epsilon)(E(n,v,v'),\mathscr{E}(n,v,v'))$$
(4.6)
$$= (E(n+1,v,v'),\mathscr{E}(n+1,v,v')).$$

Note that, from Eq. (3.37c),

$$\gamma_2(\epsilon) \circ \gamma_1(\epsilon) = \left[\gamma_{\rm KN} \circ \gamma_1(\epsilon) \right]^2. \tag{4.7}$$

The generalization of the above relations for arbitrary Kin-

nersley-Chitre transformations will be proven in a subsequent paper.

D. A new n=3 solution

We shall now apply the transformation $\gamma_{KN} \circ \gamma_1(\epsilon)$ to the n = 2 solution of Eqs. (3.28)-(3.31). We first compute

$$\gamma_1(\epsilon)(E_2,\mathscr{C}_2) = (E_2^{-1},\gamma_1(\epsilon)\mathscr{C}_2), \qquad (4.8)$$

where E_2 was given explicitly by Eqs. (3.29)-(3.31), and

$$\mathscr{E}_2 = (\mathscr{E}'_0 + i\Delta)^{-1},$$
 (4.9)

where \mathscr{C}_0 was given explicitly by Eq. (3.26). The result for $\gamma_1(\epsilon)\mathscr{C}_2$, i.e., for the second member of the right-hand side of Eq. (4.8), was obtained from the solution of a homogeneous Riemann-Hilbert problem and is given by

$$\gamma_{1}(\epsilon) \mathscr{C}_{2} = -4z^{2}\rho - \rho^{3} + (4 - i\Delta\rho)(\rho - i\Delta) + AB^{-1} \{ [2(\rho - i\Delta)^{3}(pY + iqX + p'xY + iq'yX) + 8z\rho(\rho - i\Delta)(p'yX + iq'xY)](pX + iqY)^{-1} - 4z^{2}\rho^{2} - 4(\rho - i\Delta)^{2} \}, \qquad (4.10)$$

where B and A are given explicitly by Eqs. (3.30) and (3.31), and

$$p^{2} + q^{2} = p'^{2} + q'^{2} = 1.$$

Next we apply γ_{KN} to obtain
 $(E_{3}, \mathscr{C}_{3}):= \gamma_{KN} \circ \gamma_{1}(\epsilon) (E_{2}, \mathscr{C}_{2}),$ (4.11)

where

$$E_3 = -\left[\gamma_1(\epsilon)\mathscr{C}_2\right]^*,\tag{4.12}$$

$$\mathscr{E}_{3} = - [AB^{-1}]^{*}; \qquad (4.13)$$

and $\gamma_1(\epsilon) \mathscr{C}_2$ is given by Eq. (4.10), while A and B are given by Eqs. (3.31) and (3.30).

V. FUTURE DEVELOPMENTS

Even when other more elementary methods suffice in principle, we have often found it convenient to solve a homogeneous Hilbert problem¹⁴ (HHP) instead. In the next paper of this series we shall present the formulation of the HHP that we have utilized in deriving new colliding wave solutions. In particular, we shall provide a detailed derivation of our three-parameter generalization of the Ferrari–Ibañez– Bruni two-parameter family of solutions. The HHP is certainly an important piece of artillery to have in one's arsenal as one attempts to enlarge further the family of known colliding wave solutions of the Einstein equations.

ACKNOWLEDGMENT

One of the authors (FJE) was supported in part by Grant Nos. PHY-8605958 and PHY-8306684 from the National Science Foundation.

¹F. J. Ernst, A. García-D., and I. Hauser, J. Math. Phys. 28, 2155 (1987), henceforth cited as paper I.

²Y. Nutku and M. Halil, Phys. Rev. Lett. 39, 1379 (1977).

³V. Ferrari and J. Ibañez, Gen. Relativ. Gravit. 19, 405 (1987).

⁴S. Chandrasekhar and B. C. Xanthopoulos, Proc. R. Soc. London Ser. A **408**, 175 (1986).

- ⁵V. Ferrari, J. Ibañez, and M. Bruni, "Colliding gravitational waves with noncollinear polarization: A class of soliton solutions," preprint, 1987. ⁶R. Geroch, J. Math. Phys. 12, 918 (1971); 13, 394 (1972).
- ⁷D. Kramer and G. Neugebauer, Commun. Math. Phys. 10, 132 (1968).
- ${}^{8}C^{2}$ differentiability is sufficient for the purposes of the present paper. In practice, of course, analytic functions are usually considered. The derivative of f(v) at v = 0 is defined as the limit of $[f(v) - f(0)]v^{-1}$ as $v \to 0$ through positive values.
- 9S. Chandrasekhar and V. Ferrari, Proc. R. Soc. London Ser. A 396, 55 (1984).
- ¹⁰A. Tomimatsu and H. Sato, Phys. Rev. Lett. 29, 1344 (1972); Prog. Theor. Phys. 50, 95 (1973).
- ¹¹D. M. Zipoy, J. Math. Phys. 7, 1137 (1966).
- ¹²B. H. Voorhees, Phys. Rev. D 2, 2119 (1970).

- ¹³K. Khan and R. Penrose, Nature 229, 1575 (1971).
- ¹⁴I. Hauser and F. J. Ernst, J. Math. Phys. 21, 1126 (1980).
- ¹⁵W. Kinnersley and D. M. Chitre, J. Math. Phys. 18, 1538 (1977); 19, 1926, 2037 (1978).
- ¹⁶P. Szekeres, J. Math. Phys. 13, 286 (1972).
- ¹⁷B. C. Xanthopoulos, J. Math. Phys. 27, 2129 (1986).
- ¹⁸I. Hauser and F. J. Ernst, J. Math. Phys. 22, 1051 (1981).
- ¹⁹F. J. Ernst, "Derivation of Nutku-Halil colliding plane wave solution from isotropic Kasner metric using double-Harrison transformation," in Gravitational Collapse and Relativity, edited by H. Sato and T. Nakamura (World Scientific, Singapore, 1987), pp. 141-149. [Note that there is a misprint in Eq. (3.6b), which can be corrected by taking the complex conjugate of the expression on the right.]

The generalized Weyl correspondence and time-dependent stochastic processes

M. Gadella, L. M. Nieto, and J. M. Noriega

Departamento de Física Teórica, Atómica Molecular y Nuclear, Facultad de Ciencias, Universidad de Valladolid, Valladolid 47005, Spain

E. Santos

Facultad de Ciencias, Universidad de Santander, Santander, Spain

(Received 13 August 1986; accepted for publication 22 July 1987)

The definition and some general properties of the generalized Weyl correspondence between stochastic processes and operator-valued real functions on a Hilbert space plus a trace class operator are given. The relation between the derivatives of an operator-valued function and the derivative of the corresponding stochastic process are studied. When the operator-valued function is the position (or momentum) in the Heisenberg picture, a condition for the positivity of the joint distribution functions of the corresponding process is given, provided that the evolution Hamiltonian be quadratic in the position and momentum. Finally, the case of an arbitrary Hamiltonian evolution for the position operator is studied and the two-dimensional density functions of the process is related to the Wigner function associated to some state $\hat{\rho}$, and a necessary condition for the positivity of the densities is given.

I. INTRODUCTION

The idea of a generalized Weyl correspondence (GWC) has already been presented in previous papers.¹⁻³ This correspondence shows a relation between random variables and linear operators on a separable Hilbert space \mathcal{H} as well as between signed measures on a certain sample space and trace class operators on \mathcal{H} . We here recall its definition.

Let $(X_j)_{j \in J}$ be a set of real random variables and $(\hat{x}_j)_{j \in J}$ a set of linear operators on \mathcal{H} , where J is an arbitrary index set. If there exists a trace class operator $\hat{\rho}$ such that for any finite set of indices $\{j_i,...,j_n\} \subset J$ and any real numbers $\alpha_{j_1},...,\alpha_{j_n}$,

$$E\{\exp[i(\alpha_{j_{1}}X_{j_{1}} + \dots + \alpha_{j_{n}}x_{j_{n}})]\}$$

= tr{ $\hat{\rho} \exp[i(\alpha_{j_{1}}\hat{x}_{j_{1}} + \dots + \alpha_{j_{n}}\hat{x}_{j_{n}})]\}$ (1)

holds, then we say that there exists a GWC between the random variables and the operators. Here $E\{\cdots\}$ means expectation value. Each of the random variables acts on a sample space Ω which is endowed with a finite measure such that $\mu(\Omega) = 1$. Here μ is not necessarily non-negative [i.e., there may exist a measurable set $A \subset \Omega$ such that $\mu(A) < 0$]. Usually, Ω will be the Cartesian product of the ranges of the variables $(X_j)_{j \in J}$ and X_j the projection of Ω on its *j*th coordinate, although this construction of Ω is not always necessary.

References 1-3 deal with the general properties of this GWC as well as with particular examples. On the other hand, it is a generalization of the classical Weyl correspondence (CWC) which assigns a Hermitian operator to a real function of the coordinates of the position Q and the momentum P in phase space.⁴⁻⁶ Here Q and P have the role of the random variables X_j . The CWC also assigns a finite measure on the phase space to any quantum state (positive operator on \mathcal{H} with trace 1). As in the general case, this measure is not non-negative in general. Here, the index set J contains

2N elements, where N is the dimension of the configuration space.

A stochastic process is a set of random variables.⁷ Usually, this set is indexed by an interval J of the real line which eventually is the whole real line R. For any $t \in J$, we have a random variable we call X(t). Here X(t) is a measurable mapping from a certain sample space Ω into R.

To adapt the original definition of GWC to this particular case is now simple. Let $\hat{x}(t)$ be a set of linear operators on \mathcal{H} , where $t \in J$. Here X(t) and $\hat{x}(t)$ are related through a GWC if there exists a trace operator $\hat{\rho}$ on \mathcal{H} and a signed measure μ on Ω such that

$$E\{\exp[i(\alpha_1 X(t_1) + \dots + \alpha_n X(t_n))]\}$$

= $\int_{\Omega} \exp[i(\alpha_1 X(t_1) + \dots + \alpha_n X(t_n))]d\mu$
= $\operatorname{tr} \{\hat{\rho} \exp[i(\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n))]\}$ (2)

for any finite set of times $\{t_1,...,t_n\}$ and whatever real numbers of $\alpha_1,...,\alpha_n$, where *n* is also arbitrary.

We define a stochastic vector as an N-dimensional vector whose components are stochastic processes. For any $t \in J$, $(X_1(t),...,X_N(t))$ is a measurable mapping from Ω onto a subset of \mathbb{R}^N .

A stochastic field is a further generalization of a stochastic process. Let A be any set. A stochastic field is a set of stochastic process indexed by A. In particular, a stochastic vector is a stochastic field with $A = \{1, 2, ..., N\}$ and a stochastic process is also a stochastic field with $A = \{\phi\}$ (or any set containing one element only). The denomination of a stochastic field is properly used when $A \subset R^3$. Then the mapping $a \to X(t,a)$ assigns a stochastic process to any point of the region A which is the physical space in which the phenomenon described by the stochastic field $X(t,a), a \in A$, occurs.

For a stochastic process X(t), we define the finite-di-

mensional distribution function as

$$F(x_{1},t_{1}) = \mu \{X(t_{1}) \leqslant x_{1}\},$$

:

$$F(x_{1},t_{1}; \cdot ;x_{n},t_{n}) = \mu \{X(t_{1}) \leqslant x_{1};...;X(t_{n}) \leqslant x_{n}\},$$
(3)

for all $t_1, t_2, ..., t_n$ in J, and the joint densities as

$$\rho(x_1,t_1) = \frac{\partial F(x_1,t_1)}{\partial x_1},$$

$$\vdots$$

$$\rho(x_1,t_1;...;x_n,t_n) = \frac{\partial^n F(x_1,t_1;...;t_n,t_n)}{\partial x_1\cdots\partial x_n}.$$
(4)

The partial derivatives in (4) should be understood in the sense of the derivatives of distributions, which coincides with the usual sense of the derivative when the finite-dimensional distribution functions are absolutely continuous. If μ is a *positive* probability measure all the distribution functions are non-negative.

The finite-dimensional distribution functions do not fix the process X(t) uniquely. There are many examples of two different stochastic processes having the same finite-dimensional distribution functions.

Let us go back to Eq. (2). It can be written as

$$\int_{\mathbb{R}^{n}} \exp\{i(\alpha_{i}x_{1} + \dots + \alpha_{n}x_{n})\}$$
$$\times \rho(x_{1},t_{1};\dots;x_{n},t_{n})dx_{1}\cdots dx_{n}$$
$$= \operatorname{tr}\{\hat{\rho}\exp[i(\alpha_{1}\hat{x}(t_{1}) + \dots + \alpha_{n}\hat{x}(t_{n}))]\}, \qquad (5)$$

whence

$$\rho(x_1,t_1;...;x_n,t_n) = \frac{1}{(2\pi)^n} \int_{\mathbb{R}^n} \exp\{-i(\alpha_1 x_1 + \dots + \alpha_n x_n)\} \\ \times \operatorname{tr}\{\hat{\rho} \exp[i(\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n))]\} \\ \times d\alpha_1 \cdots d\alpha_n.$$
(6)

Equation (6) is a consequence of the properties of the Fourier transform, and gives us the following information: Since the finite-dimensional distribution functions and hence their corresponding densities do not fix a process uniquely, in general we shall have more than one process associated to the set of operators $\hat{x}(t)$ and $\hat{\rho}$ through a GWC.

If we have a stochastic vector, the measure μ on Ω determines the following distribution functions:

$$F(x_{11},...,x_{N1},t_{1};...;x_{1s},...,x_{Ns},t_{s}) = \mu\{X_{1}(t_{1}) \leqslant x_{11},...,X_{N}(t_{1}) \leqslant x_{N1};...;X_{1}(t_{s}) \\ \leqslant x_{1s},...,X_{N}(t_{s}) \leqslant x_{Ns}\}$$
(7)

and the corresponding densities:

$$\rho(x_{11},...,x_{N1}t_{1};...;x_{1s};...;x_{Ns},t_{s}) = \frac{\partial^{s \times N} F(x_{11},...,x_{Ns})}{\partial x_{11} \cdots \partial x_{Ns}}.$$

The definition of GWC can also be used to associate stochastic vectors to operator-valued real functions with Ncomponents, as for example the position or momentum operators on $L^2(\mathbb{R}^N)$ in the Heisenberg picture. Let $\overline{X}(t)$ = $(X_1(t),...,X_N(t))$ be a stochastic vector and $\overline{\hat{x}}(t)$ = $(\hat{x}_1(t),...,\hat{x}_N(t))$ be an N-dimensional operator-valued real function, where t runs over an interval J of the real line. Here $\overline{X}(t)$ and $\overline{\hat{x}}(t)$ are related through the GWC if there exists a signed measure μ on Ω with $\mu(\Omega) = 1$ and a trace class operator $\hat{\rho}$ on \mathcal{H} such that

$$E\{\exp[i(\alpha_{11}X_{1}(t_{1}) + \dots + \alpha_{N1}X_{n}(t_{1}) + \dots + \alpha_{1s}X_{1}(t_{s}) + \dots + \alpha_{Ns}X_{N}(t_{s}))]\}$$

$$= \int_{R_{N\times s}} \exp[i(\alpha_{11}x_{11} + \dots + \alpha_{N1}x_{N1} + \dots + \alpha_{1s}x_{1s} + \dots + \alpha_{Ns}x_{Ns})], \qquad (8)$$

$$\int \rho(x_{11},...,x_{N1},t_{1};...;x_{1s},...,x_{Ns},t_{s}) dx_{11}\cdots dx_{Ns}$$

= tr{ $\hat{\rho} \exp[i(\alpha_{11}\hat{x}_{1}(t_{1}) + \cdots + \alpha_{N1}\hat{x}_{N}(t_{1}) + \cdots + \alpha_{1s}\hat{x}_{1}(t_{s}) + \cdots + \alpha_{Ns}\hat{x}_{N}(t_{s}))] \}$ (9)

for any $\{t_1,...,t_s\} \subset J$ and any real numbers $\alpha_{11},...,\alpha_{Ns}$. From (9), it follows that

$$\rho(x_{11},...,x_{N1},t_{1};...;x_{1s},...,x_{Ns},t_{s}) = \frac{1}{(2\pi)^{N\times s}} \int_{\mathbb{R}^{N\times s}} d\alpha_{11}\cdots d\alpha_{Ns} \\ \times \exp\{-i(\alpha_{11}x_{11}+\cdots+\alpha_{Ns}x_{Ns}) \\ + \operatorname{tr}\{\hat{\rho}\exp[i(\alpha_{11}\hat{x}_{1}(t_{1})+\cdots+\alpha_{Ns}\hat{x}_{N}(t_{s}))]\}.$$
(10)

The present paper contains three main sections. In Sec. II, we study the relation between the derivative of the process and the derivative of the operator-valued function $\hat{x}(t)$. Section III starts with a one-dimensional problem. Here, $\hat{x}(t)$ is the position operator of a particle on the Heisenberg picture. The time evolution of $\hat{x}(t)$ is controlled by a quadratic time independent Hamiltonian. In this case, the density function of each of the random variables in the process X(t) can be interpreted as the classical probability density of finding the particle in the configuration space at the time t, provided that all the joint densities of the process are positive. We give a sufficient condition for this positivity: The Wigner function of $\hat{\rho}$ must be positive at t = 0. This result is easily generalized to the case of N dimensions. In Sec. IV, we discuss a necessary condition for the positivity of μ when $\hat{\rho}$ is a pure state and we are in the situation described in Sec. II.

II. GENERAL CONSIDERATIONS

We start with the assumption of the existence of a GWC for a stochastic process X(t) or a stochastic vector $(X_1(t),...,X_n(t))$. This means that Eqs. (2) and (9) make sense and, in particular, their right-hand sides are well defined, which in (2) happens if $\exp\{i(\alpha_1\hat{x}(t_1) + \cdots + \alpha_n\hat{x}(t_n))\}$ is a bounded operator for all the α_i and all the values of the time parameter. There are two interesting situations in which this occurs: (1) all the $\hat{x}(t)$ are bounded for all $t \in J$, and (2) all the operators of the form $\alpha_1\hat{x}(t_1) + \cdots + \alpha_n\hat{x}(t_n)$ are essentially self-adjoint on a common domain dense in \mathcal{H} . This occurs, in particular, when $\hat{x}(t) = h(t)\hat{x}(0) + g(t)\hat{p}(0) + \eta(t)I$, where $\hat{x}(0)$ and $\hat{p}(0)$ are, respectively, the position and momentum operators in one dimension, I is the identity operator, and h(t), g(t), and $\eta(t)$ are real functions. This case and its generalizations to N dimensions is studied in this paper.

In order to simplify our notation, we mainly deal here with stochastic processes X(t). However, most of our results are trivially extensible to stochastic vectors.

We can look at the random variable $X(t_0)$ for each $t_0 \in J$ in two ways. First, $X(t_0)$ can be properly considered as something which "transports" measure from Ω to R. Under this point of view, what really matters of $X(t_0)$ is its probability distribution and its joint distributions with the other random variables in the process. The other possibility is to look at $X(t_0)$ as a measurable function from Ω into R. Both points of view are quite different, since the first one depends critically on the measure μ on Ω while the second is independent on μ . To visualize the difference, one can imagine a situation in which $X(t) = X(0) \quad \forall t \in J$ in measure (this means that $\mu\{w/[X(t)](w)\neq [X(0)](w)$ for some $t \in J$ = 0 if μ is positive; if μ is not positive and $\mu = \mu_1 - \mu_2$ is the Jordan decomposition of μ , we have to require that $\mu_i \{ w / [X(t)](w) \neq [X(0)](w), \text{ for some } t \in J \} = 0 \text{ with}$ i = 1,2) and the measurable functions are all different. Such a process can be easily constructed.⁸

Henceforth, a stochastic process will be a set of measurable functions X(t) on a sample space Ω ; where t belongs to an open interval J of the real line; plus a bounded signed measure μ on Ω with $\mu(\Omega) = 1$ (a signed measure is bounded if $\mu = \mu_1 - \mu_2$, where μ_1 and μ_2 are positive and finite). If μ is a probability measure, we say that the process is veritable. All the finite-dimensional distribution functions and their corresponding densities are well defined provided the boundedness of μ . A process is veritable if and only if all the densities are positive (a.e. with respect to the Lebesgue measure).

The stochastic relation among the X(t) can appear in many other ways. An interesting case arises when the identity $X(t) = h_1(t)X_1 + \cdots + h_n(t)X_n$ holds in measure, where $h_1(t),...,h_n(t)$ are real functions on J and $X_1,...,X_n$ are *n* random variables on Ω . The interest of this kind of process will be evident later.

For any $t \in J$, the GWC maps X(t), considered as a measurable function on Ω , into $\hat{x}(t)$, provided that certain conditions are satisfied.^{1,2} Assume now that X(t) and $\hat{x}(t)$ admit a derivative with respect to t in some sense. Are these derivatives related through a GWC? If the derivative for $\hat{x}(t)$ exists at some point, does it for X(t) and vice versa? Our next objective is to give an answer to these questions. In order to make our procedures and results comprehensible to a wide audience, we start with the basic definitions.

III. DEFINITIONS

(1) Let X(t) be a veritable stochastic process. If all the second moments existed, we define a norm for the random variables in the process as

$$||X(t)|| = \{E\{X^{2}(t)\}\}^{1/2} = \left\{\int_{\Omega} X^{2}(t)d\mu\right\}^{1/2}.$$
 (11)

If X(t) were not veritable, we still could define a norm for each of the random variables X(t), provided that their second moments existed with respect to the measures μ_1 and μ_2 , where $\mu = \mu_1 - \mu_2$ is the Jordan decomposition of μ . The square of this norm is given by

$$|X(t)||^{2} = \int_{\Omega} X^{2}(t) d\mu_{1} + \int_{\Omega} X^{2}(t) d\mu_{2}$$
(12)

and the second moment of X(t) with respect to μ is

$$E\{X^{2}(t)\} = \int_{\Omega} X^{2}(t) d\mu_{1} - \int_{\Omega} X^{2}(t) d\mu_{2}$$
$$= \int_{\Omega} X^{2}(t) d\mu.$$
(13)

In the two cases considered here, we have that $E\{X(t)X(t')\} < \infty$ for any $t, t' \in J$ and

$$E\{X^{2}(t)\} = -\frac{d^{2}}{d\alpha^{2}} E\{e^{i\alpha X(t)}\}|_{\alpha=0}.$$
 (14)

(2) X(t) is continuous at $t = t_0$ if $||X(t) - X(t_0)|| \to 0$ when $t \to t_0$. Here X(t) is continuous if it is so at any $t \in J$.

(3) X(t) admits a derivative at t_0 if there exists a random variable $\dot{X}(t_0)$ with finite second moment such that

$$\lim_{s \to 0} \| [X(t_0 + s) - X(t_0)] / s - \dot{X}(t_0) \| = 0.$$

If X(t) admits a derivative at any $t \in J$, we have a stochastic process $\dot{X}(t)$ that we call the derivative of X(t).

Definitions (2) and (3) are independent of whether the process is veritable or not.

(4) Let Y(t,s) be a collection of stochastic processes indexed by a continuous real index s. A process X(t) is the limit of Y(t,s) when s goes to zero if

$$\lim_{t \to 0} ||Y(t,s) - X(t)|| = 0.$$
(15)

If all the Y(t,s) are veritable, its limit X(t) is also veritable. In this case, the joint characteristic function

 $E\{\exp\{\{\alpha_1 Y(t_1,s) + \cdots + \alpha_n Y(t_n,s)\}\}\}$

$$=\varphi_{s,t_1,\ldots,t_n}(\alpha_1,\ldots,\alpha_n)$$
(16)

converges pointwise to the characteristic function

$$E\{\exp(i\{\alpha_1 X(t_1) + \cdots + \alpha_n X(t_n)\})\}.$$
(17)

This result is also true even if the Y(t,s) were not veritable.

(5) Now assume that all the operators in the family $\hat{x}(t)$ are defined in a common domain \mathscr{D} dense in the Hilbert space \mathscr{H} . Here $\hat{x}(t)$ admits a strong derivative or simply a derivative at $t_0 \in J$ if there exists a linear mapping $\dot{x}(t_0)$ from \mathscr{D} to \mathscr{H} such that

$$\lim_{t \to 0} \|\{[\hat{x}(t_0 + s) - \hat{x}(t_0)]/s - \hat{x}(t_0)\}\varphi\| = 0 \quad (18)$$

 $\forall \varphi \in \mathscr{D}$. In (18) we have used the standard Hilbert space norm. If $\hat{x}(t)$ admits a derivative at any $t \in J$, we say that it is derivable.

When all the $\hat{x}(t)$ are bounded and there is a bounded operator $\dot{\hat{x}}(t_0)$ such that in the norm operator sense

$$\lim_{s \to 0} \| [\hat{x}(t_0 + s) - \hat{x}(t_0)] / s - \dot{x}(t_0) \| = 0,$$
(19)

we say that $\dot{x}(t_0)$ is the uniform derivative of x(t) at t_0 . The existence of the uniform derivative implies that of the derivative but the converse is not true.

Definitions (1)-(5) are standard and appear in the textbooks.⁹⁻¹²

Our next goal is to prove that, under certain conditions, if X(t) and $\hat{x}(t)$ are related through a GWC so are their derivatives. Henceforth, we shall assume that for any $t \in J$ the operators $\hat{x}(t)$ and $\dot{x}(t)$ and their linear combinations are essentially self-adjoint on a common domain \mathcal{D} dense in \mathcal{H} . In this case $\exp\{i(\alpha_1\hat{x}(t_1) + \cdots + \alpha_n\hat{x}(t_n))\}\)$ and $\exp\{i(\alpha_1\hat{x}(t_1) + \cdots + \alpha_n\hat{x}(t_n))\}\)$ are well defined unitary operators on \mathcal{H} for any real values of $\alpha_1, \dots, \alpha_n$ and arbitrary t_1, \dots, t_n in J. For our purpose, we need to begin with some results in functional analysis.

Theorem I: Let $\{A_n\}_{n \in \mathbb{N}}$, A be self-adjoint operators and

$$\mathscr{D}$$
 a common core for them. If $A_n \to A$ for any $\varphi \in \mathscr{D}$, then
 $e^{it A_n} \varphi \to e^{it A} \varphi$, $\forall \varphi \in \mathscr{H}$ and $\forall t \in \mathbb{R}$.
The proof can be seen in Ref. 12.
Corollary I: Assume that $\forall \varphi \in \mathscr{D}$,

$$\{\alpha_1 \hat{x}(t_1 + s) + \dots + \alpha_n \hat{x}(t_n + s)\}\varphi$$

$$\rightarrow \{\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n)\}\varphi, \qquad (20)$$

wherever $s \to 0 \forall \varphi \in \mathscr{D}$. Then

$$\exp(i\{\alpha_1 \hat{x}(t_1+s) + \alpha_n \hat{x}(t_n+s)\})\varphi$$

$$\rightarrow \exp(i\{\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n)\})\varphi$$
(21)

 $\forall \varphi \in \mathcal{H} \text{ and any } \alpha_1, \dots, \alpha_n \text{ in } R.$

Corollary II:

$$\lim_{s\to 0} \exp\left(i\left\{\alpha_1\left[\frac{\hat{x}(t_1+s)-\hat{x}(t_1)}{s}\right]+\cdots+\alpha_n\left[\frac{\hat{x}(t_n+s)-\hat{x}(t_n)}{s}\right]\right\}\right)\varphi = \exp(i\{\alpha_1\dot{x}(t_1)+\cdots+\alpha_n\dot{x}(t_n)\})\varphi.$$
(22)

Corollary III: Assume that $\hat{\rho}$ is a trace class normal operator (in particular $\hat{\rho}$ can be positive). Then

$$\lim_{s\to 0} \operatorname{tr}\left\{\hat{\rho} \exp\left(i\left\{\alpha_1\left[\frac{\hat{x}(t_1+s)-\hat{x}(t_1)}{s}\right]+\cdots+\alpha_n\left[\frac{\hat{x}(t_n+s)-\hat{x}(t_n)}{s}\right]\right\}\right)\right\} = \operatorname{tr}\left\{\hat{\rho} \exp\left(i\left\{\alpha_1\dot{x}(t_1)+\cdots+\alpha_n\dot{x}(t_n)\right\}\right)\right\}.$$
 (23)

ſ

Proof: Since $\hat{\rho}$ is normal, the following spectral decomposition works¹²:

$$\hat{\rho} = \sum_{k=1}^{\infty} \lambda_k |\psi_k\rangle \langle \psi_k|, \qquad (24)$$

where $\hat{\rho}\psi_k = \lambda_k \psi_k$, $\sum_{k=1}^{\infty} |\lambda_k| < \infty$, and $||\psi_k|| = 1$. Assume that all the λ_k but a finite number are zero. Then, Corollary III follows from Corollary II trivially. The proof in the general case is a little more involved. In order to simplify the notation, we shall here demonstrate (23) for n = 1. The case of $n \neq 1$ does not require more sophistication.

Therefore consider

$$\operatorname{tr}\left\{\hat{\rho}\exp\left\{i\alpha\left[\frac{\hat{x}(t+s)-\hat{x}(t)}{s}\right]\right\}\right\}$$
and used $\hat{y}(s) = [\hat{x}(t+s)-\hat{x}(t)]/s$. Thus
$$(25)$$

$$\operatorname{tr}\{\rho e^{i\alpha \hat{\mathfrak{P}}(s)}\} = \sum_{k=1}^{\infty} \lambda_{k} \langle \psi_{k} | e^{i\alpha \hat{\mathfrak{P}}(s)} | \psi_{k} \rangle$$
$$= \sum_{k=1}^{\infty} f_{k}(s), \qquad (26)$$

$$|f_{k}(s)| = |\lambda_{k}| |\langle \psi_{k} | e^{i\alpha \hat{y}(s)} | \psi_{k} \rangle|$$

$$\leq |\lambda_{k}| ||\psi_{k}|| ||e^{i\alpha \hat{y}(s)} \psi_{k}||.$$
(27)

2964 J. Math. Phys., Vol. 28, No. 12, December 1987

Since $\hat{y}(s)$ is self-adjoint and α is a real number, $e^{i\alpha\hat{y}(s)}$ is unitary. Thus $|f_k(s)| \leq |\lambda_k|$ and

$$|\mathrm{tr}\{\hat{\rho}e^{i\alpha\hat{y}(s)}\}| \leq \sum_{k=1}^{\infty} |f_k(s)| \leq \sum_{k=1}^{\infty} |\lambda_k| < \infty.$$
(28)

Hence the Weierstrass M criterion¹³ says that the series (26) converges uniformly in s. Since

$$\lim_{s\to 0} \langle \psi_k | e^{i\alpha \hat{y}(s)} | \psi_k \rangle = \langle \psi_k | e^{i\alpha \hat{x}(t)} | \psi_k \rangle$$

exists for any $k = 1, 2, ..., the limit of the series exists for <math>s \to 0$ and is tr{ $\hat{\rho} \exp\{i\alpha \hat{x}(t)\}$ }.

Theorem II: Assume the following.

(a) X(t) is a stochastic process on Ω and μ is a bounded signed measure on Ω .

(b) The derivative X(t) exists at each $t \in J$.

(c) $\hat{\rho}$ is a normal trace class operator on \mathcal{H} .

(d) The operators $\hat{x}(t)$ and their real linear combinations are e.s.a. on a common domain \mathcal{D} dense in \mathcal{H} .

(e) The derivative $\dot{\hat{x}}(t)$ exists for any $t \in J$ and any real linear combination of these operators is e.s.a. on \mathcal{H} .

(f) There exists a GWC relating X(t) to $\hat{x}(t)$ and μ to $\hat{\rho}$. Then, the GWC relates $\dot{X}(t)$ and $\dot{x}(t)$ with μ and $\hat{\rho}$ being the same.

Proof: After Corollary III, the following limit is well defined:

$$\lim_{s \to 0} \operatorname{tr} \hat{\rho} \exp i \left\{ \alpha_1 \frac{x(t_1 + s) - x(t_1)}{s} + \dots + \alpha_n \frac{x(t_n + s) - x(t_n)}{s} \right\}$$

= $\operatorname{tr} \left\{ \hat{\rho} \exp \left\{ i [\alpha_1 \dot{\hat{x}}(t_1) + \dots + \alpha_n \dot{\hat{x}}(t_n)] \right\} \right\}$
= $\lim_{s \to 0} E \left\{ \exp \left\{ i_1 \left[\frac{X(t_1 + s) - X(t_1)}{s} + \dots + \alpha_n \frac{X(t_n + s) - X(t_n)}{s} \right] \right\} \right\},$ (29)

the derivative of X(t) exists by hypothesis. Thus after the comments in Definition 4, this limit is

(29) =
$$E \left[\exp \left[i (\alpha_1 \dot{X}(t_1) + \dots + \alpha_n \dot{X}(t_n)) \right] \right].$$
 (30)

The next question we want to answer is if the existence of $\hat{X}(t)$ says something about the existence of the derivative of X(t). In the next theorem we do not assume the existence of the moments of X(t).

Theorem III: Assumptions (c)-(f) in Theorem II are also valid here, but X(t) is veritable. Then the functions

$$\lim_{s \to 0} E\left\{ \exp\left[i \left(\alpha_1 \frac{X(t_1 + s) - X(t_1)}{s} + \dots + \alpha_n \frac{X(t_n + s) - X(t_n)}{s} \right) \right] \right\} = \varphi_{t_1, \dots, t_n}(\alpha_1, \dots, \alpha_n)$$
(31)

are the characteristic functions of a certain process.

Proof: Since $\hat{\rho}$ is a normal operator, the limit (31) equals to

$$\operatorname{tr}\{\hat{\rho}\exp\{i(\alpha_{1}\dot{\hat{x}}(t_{1})+\cdots+\alpha_{n}\dot{\hat{x}}(t_{n}))\}\}=\sum_{k=1}^{\infty}\lambda_{k}\langle\psi_{k}|\exp[i(\alpha_{1}\dot{\hat{x}}(t_{1})+\cdots+\alpha_{n}\dot{\hat{x}}(t_{n}))]|\psi_{k}\rangle,\tag{32}$$

where the λ_k are the eigenvalues of $\hat{\rho}$ and the ψ_k their corresponding eigenvectors. If we prove that (32) is continuous on $\overline{\alpha} = (\alpha_1, ..., \alpha_n)$, Theorem III would be proved.

The series in (32) converges uniformly on $\overline{\alpha}$. To show this assertion, we use the Weierstrass *M* criterion as we did it in the proof of Corollary III. Furthermore, we know after the Stone theorem that

$$\langle \psi_k | \exp\{i\{\alpha_1 \hat{x}(t_1) + \cdots + \alpha_n \hat{x}(t_n)\}\} | \psi_k \rangle$$

is continuous on $\overline{\alpha}$. Therefore the series in (32) is continuous on $\overline{\alpha}$, since it is the limit of a sequence of continuous functions on $\overline{\alpha}$ converging uniformly on $\overline{\alpha}$.

This result cannot be extended to the case in which μ is nonpositive. In this case the limit (31) could not be the difference between two characteristic functions.

Obviously, if X(t) is derivable, (31) are the joint characteristic functions of the derivative. However, we cannot claim that we have proved the existence of the derivative of X(t).

Its existence requires a further assumption. We start the discussion with four lemmas.

Lemma I: Assume that we are in the conditions of Theorem III and that any real combination of the form $\alpha_1 \hat{x}(t_1) + \cdots + \alpha_n \hat{x}(t_n) + \beta_1 \dot{\hat{x}}(t_1) + \cdots + \beta_n \dot{\hat{x}}(t_n)$ is self-adjoint. Then the limit

$$\lim_{s \to 0} E\left[\exp i\left\{\alpha_1 X(t_1) + \beta_1 \frac{X(t_1 + s) - X(t_1)}{s} + \dots + \beta_n \frac{X(t_n + s) - X(t_n)}{s}\right\}\right]$$
(33)

is well defined and represents the characteristic function of a stochastic vector with two components.

Proof: It is simple to realize that

$$\lim_{s \to 0} \operatorname{tr} \left\{ \hat{\rho} \exp i \left\{ \alpha_i \hat{x}(t_1) + \beta_1 \frac{\hat{x}(t_1 + s) - \hat{x}(t_1)}{s} + \dots + \beta_n \frac{\hat{x}(t_n + s) - \hat{x}(t_n)}{s} \right\} \right\}$$

= tr { $\hat{\rho} \exp i (\alpha_1 \hat{x}(t_1) + \beta_1 \dot{\hat{x}}(t_1) + \dots + \alpha_n \hat{x}(t_n) + \beta_n \dot{\hat{x}}(t_n)) \}$ (34)

is continuous at zero function of $(\alpha_1, \beta_1, ..., \alpha_n, \beta_n)$.

If X(t) admits a derivative, this stochastic vector is $(X(t), \dot{X}(t))$.

Remark: If $(X(t), \pi(t))$ is some stochastic vector having (33) as joint characteristic functions, we can write

$$E\{\exp(i\{\alpha_{1}X(t_{1}) + \beta_{1}\pi(t_{1}) + \dots + \alpha_{n}X(t_{n}) + \beta_{n}\pi(t_{n})\})\}$$

= tr{ $\hat{\rho} \exp i\{\alpha_{1}\hat{x}(t_{1}) + \beta_{1}\dot{x}(t_{1}) + \dots + \alpha_{n}\hat{x}(t_{n}) + \beta_{n}\dot{x}(t_{n})\}\}.$ (35)

If for any $t \in J$, X(t), and $\pi(t)$ admitted moments of second order, we would have

(1)
$$E\left\{\left(\frac{X(t+s)-X(t)}{s}-\pi(t)\right)^2\right\}$$
, (36)

(2)
$$-\frac{d^2}{d\alpha^2} E\left\{ \exp\left(i\alpha \left[\frac{X(t+s) - X(t)}{s} - \pi(t)\right]\right) \right\} \Big|_{\alpha = 0} = E\left\{ \left(\frac{X(t+s) - X(t)}{s} - \pi(t)\right)^2 \right\}.$$
 (37)

Gadella et al. 2965

Lemma II: Let A be a self-adjoint operator and $\hat{\rho}$ a trace class normal operator with $\hat{\rho} = \sum_{k=1}^{\infty} \lambda_k |\psi_k\rangle \langle \psi_k|$ with $\psi_k \in \mathscr{D}(A)$, the domain of A. If tr{ $\hat{\rho}|A|^n$ } < ∞ ,

$$\frac{d^n}{d\alpha^n}\operatorname{tr}\{\hat{\rho}e^{i\alpha A}\}|_{\alpha=0}=(i)^n\operatorname{tr}\{\hat{\rho}A^n\}.$$
(38)

Proof: See Appendix A.

Lemma III: Let $\hat{\rho}$ be as in Lemma II, with ψ_k in the domains of two self-adjoint operator A and B and $B\psi_k \in \mathscr{D}(A)$. If $\operatorname{tr}\{\rho A^2\} < \infty$ and $\operatorname{tr}\{\hat{\rho} B^2\} < \infty$, then $\operatorname{tr}\{\hat{\rho} AB\} < \infty$.

Proof: See Appendix B.

Lemma IV: Under the conditions in Lemma I, if $tr{\hat{\rho}\hat{x}^2(t)}$ and $tr{\hat{\rho}\hat{x}^2(t)}$ are both finite and $\hat{\rho}$ is a normal trace class operator, then

$$\operatorname{tr}\left\{\hat{\rho}\left[\frac{(\hat{x}(t+s)-\hat{x}(t))}{s}-\dot{x}(t)\right]^{2}\right\}$$
(39)

for any $t \in J$ and s in a neighborhood of zero.

Proof: It is obvious after Lemmas II and III.

Theorem IV: We are under the conditions of Theorem III and the two following hypothesis: (a) X(t) and $\pi(t)$ (see remark after Lemma I) have finite second moments $\forall t \in J$; and (b) either

$$\hat{
ho} = \sum_{k=1}^{N} \lambda_k |\psi_k\rangle \langle \psi_k|$$

with N finite and the ψ_k belong to proper domains so that Lemmas II and III can be used, or the operators $\hat{x}(t)$ and $\dot{x}(t)$ are bounded and the derivative exists in the norm operator sense.

Then, $\dot{X}(t) = \pi(t)$.

Proof: A proper choice of the constants in (35) yields

$$E\left\{\exp\left(i\alpha\left[\frac{\dot{X}(t+s)-\dot{X}(t)}{s}-\pi(t)\right]\right)\right\}$$
$$=\operatorname{tr}\left\{\hat{\rho}\exp\left(i\alpha\left[\frac{\hat{x}(t+s)-\hat{x}(t)}{s}-\dot{x}(t)\right]\right)\right\}.$$
 (40)

If we take second derivatives on α at the point $\alpha = 0$, we have

$$E\left\{\left(\frac{X(t+s) - X(t)}{s} - \pi(t)\right)^{2}\right\}$$
$$= \operatorname{tr}\left\{\hat{\rho}\left[\frac{\hat{x}(t+s) - \hat{x}(t)}{s} - \dot{x}(t)\right]^{2}\right\}.$$
(41)

Hypothesis (b) implies that the lhs of (41) has a limit when $s \rightarrow 0$ and that this limit is zero. Since the derivative of a process is unique, Theorem IV follows.

Theorem V: We are under the conditions of Theorem IV, but we ignore that $\pi(t)$ have finite second moments.

Then, the derivative of X(t) exists.

Proof: A sufficient condition for the existence of the derivative is the existence of the limit¹⁰

$$\lim_{s,\tau \to 0} \frac{1}{s\tau} \left\{ E \left[X(t+s)X(t'+\tau) \right] - E \left[X(t+s)X(t') \right] - E \left[X(t)X(t') \right] \right\} \\ - E \left[X(t)X(t'+\tau) \right] + E \left[X(t)X(t') \right] \right\} \\ = \lim_{s,\tau \to 0} E \left\{ \frac{X(t+s) - X(t)}{s} \frac{X(t'+X) - X(t')}{\tau} \right\}.$$
(42)

The existence of a GWC between X(t) and $\hat{x}(t)$ implies that

$$E\{\exp(i\alpha\{\alpha_1 X(t_1) + \alpha_2 X(t_2)\})\}$$

= tr{ $\hat{\rho} \exp(i\alpha\{\alpha_1 \hat{x}(t_1) + \alpha_2 \hat{x}(t_2)\})\},$ (43)

where α , α_1 , and α_2 are arbitrary real numbers and t_1 , t_2 belong to J.

Deriving twice with respect to α yields

$$E\{(\alpha_{1}X(t_{1}) + \alpha_{2}X(t_{2}))^{2}\}$$

$$= \alpha_{1}^{2}E(X^{2}(t_{1})) + 2\alpha_{1}\alpha_{2}E(X(t_{1})X(t_{2}))$$

$$+ \alpha_{2}^{2}E(X^{2}(t_{2}))$$

$$= \operatorname{tr} \{ \hat{\rho}(\alpha_{1}\hat{x}(t_{1}) + \alpha_{2}\hat{x}(t_{2}))^{2} \}$$

$$= \alpha_{1}^{2} \operatorname{tr} \{ \hat{\rho}(\hat{x}(t_{1}))^{2} \} + \alpha_{1}\alpha_{2}$$

$$\times \operatorname{tr} \{ \hat{\rho}(\hat{x}(t_{1})\hat{x}(t_{2}) + \hat{x}(t_{2})\hat{x}(t_{1})) \}$$

$$+ \alpha_{2}^{2} \operatorname{tr} \{ \hat{\rho}(\hat{x}(t_{2}))^{2} \}.$$
(44)

The arbitrary character of α_1 and α_2 gives

$$E\{X(t_1)X(t_2)\} = \frac{1}{2} \operatorname{tr} \left[\hat{\rho}\{\hat{x}(t_1)\hat{x}(t_2) + \hat{x}(t_2)\hat{x}(t_1)\}\right].$$
(45)

From (45), one obtains

$$(42) = \lim_{2 \to \tau \to 0} \operatorname{tr} \left\{ \hat{\rho} \left\{ \left[\frac{\hat{x}(t+s) - \hat{x}(t)}{s} \right] \right] \\ \times \left[\frac{\hat{x}(t'+\tau) - \hat{x}(t')}{\tau} \right] \\ + \left[\frac{\hat{x}(t'+\tau) - \hat{x}(t')}{\tau} \right] \left[\frac{x(t+s) - x(t)}{s} \right] \right\}.$$

$$(45')$$

Under the stated conditions this limit exists and is

$$\frac{1}{2} \operatorname{tr} \left[\hat{\rho} \{ \hat{x}(t) \hat{x}(t') + \hat{x}(t') \hat{x}(t) \} \right].$$
(46)

One may ask whether an operator-valued real function $\hat{x}(t)$ can exist fulfilling all the required conditions. We have already mentioned that if \hat{x} and \hat{p} are the position and momentum operators in one dimension, respectively, and $\hat{x}(t) = h(t)\hat{x} + g(t)\hat{p} + \eta(t)I$, where h(t), g(t), and $\eta(t)$ are real functions, any real linear combination of the $\hat{x}(t)$ is e.s.a. on the Schwartz space S(R).¹⁴ Furthermore, if h(t), g(t), and $\eta(t)$ are derivable, $\dot{\hat{x}}(t) = \dot{h}(t)\hat{x} + \dot{g}(t)\hat{p} + \dot{\eta}(t)I$ and any real linear combination of the $\hat{x}(t)$ is e.s.a. on S(R). Moreover, the $\hat{x}(t)$ and the derivatives to all orders (if they existed) leave S(R) invariant.

Remark: We know that the CWC in one dimension assigns to $\hat{\rho}$ a signed measure μ on the phase space R^2 . Here R^2 can naturally be viewed as the Cartesian product of the ranges of the measurable functions representing, respectively, the position Q and the momentum P of a one-dimensional classical free particle. Here μ represents the quantum state of the particle given by $\hat{\rho}$ in the Hilbert space formalism. If μ were positive definite, the quantum mechanical state $\hat{\rho}$ would be equivalent to the classical statistical state given by μ on the phase space R^2 . The CWC gives the following relation:

$$E\{e^{i(\alpha Q + \beta P)}\} = \int_{\Omega} e^{i(\alpha Q + \beta P)} d\mu$$
$$= \int_{\Omega} e^{i(\alpha q + \beta p)} \rho(q, p) dq dp$$
$$= \operatorname{tr}\{\hat{\rho}e^{i(\alpha \hat{x} + \beta \hat{p})}\}.$$
(47)

The density $\rho(q,p)$ is called the Wigner function⁵ associated to the operator $\hat{\rho}$.

Assume now that $\hat{x}(t) = h(t)\hat{x} + g(t)\hat{p} + \eta(t)I$ and take $Y(t) = h(t)Q + g(t)P + \eta(t)\mathscr{I}$, where \mathscr{I} maps R^2 into 1. Since we know $\rho(q,p)$, the joint distribution functions of the Y(t) can be calculated, provided that μ were bounded. Let X(t) be any stochastic process having the same distribution functions as Y(t). The X(t) and $\hat{x}(t)$ are related through a GWC.

The proof is based on the idea according to which the joint distribution functions fix the joint characteristic functions uniquely, if two stochastic processes have the same distribution functions, they have the same characteristic functions. Hence

$$E\{\exp(i\{\alpha_1 X(t_1) + \dots + \alpha_n X(t_n)\})\}$$

$$= E\{\exp(i\{\alpha_1 [h(t_1)Q + g(t_1)P + \eta(t_1)\mathscr{I}] + \dots + \alpha_n [h(t_n)Q + g(t_n)P + \eta(t_n)\mathscr{I}]\}\}$$

$$= \operatorname{tr}\{\hat{\rho}\exp(i\{\alpha_1 [h(t_a)\hat{x} + g(t_1)\hat{\rho} + \eta(t_1)I] + \dots + \alpha_n [h(t_n)\hat{x} + g(t_n)\hat{\rho} + \eta(t_n)I]\})\}$$

$$= \operatorname{tr}\{\hat{\rho}\exp(i\{\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n)\})\}. \quad (48)$$

IV. QUADRATIC HAMILTONIANS AND VERITABLE PROCESSES

This section deals with a particular case. Here $\hat{x}(t)$ represents the position operator in the Heisenberg picture. The evolution from $\hat{x}(0)$ to $\hat{x}(t)$ is given by a time-independent Hamiltonian H. We assume that H is a quadratic function of the position and the momentum, i.e., $H = \alpha \hat{p}^2 + \beta x^2 + \delta(\hat{p}\hat{x} + \hat{q}\hat{x}) + \gamma \hat{p} + \eta \hat{x} + \epsilon$. Here $\hat{\rho}$ is usually a positive trace class operator on \mathcal{H} , although its positivity is not necessarily required. The question we want to pose here is the following: If $\hat{x}(t)$ is related to a stochastic process X(t) through the GWC, when is this process veritable? The answer obviously depends on $\hat{\rho}$.

If X(t) were veritable, we could give a physical meaning to it whenever $\hat{\rho}$ is positive. For any $t \in J$, the distribution function of X(t) would give the probability of finding a classical particle in some region of the configuration space. This classical particle has at any $t \in J$ the same expectation values for the position as a quantum particle evolving under the action of the Hamiltonian H and whose state in the Heisenberg picture is given by the positive trace class operator $\hat{\rho}$.

If X(t) were not veritable or $\hat{\rho}$ were not positive, we could not give a physical meaning to X(t).

The process X(t) is veritable if and only if its joint density functions are positive definite. We shall calculate these joint densities for the case of a quadratic Hamiltonian. We start with the most simple cases including those of the free particle and the harmonic oscillator before solving the problem in its full generality. We finally find a sufficient condition for these density functions to be positive.

In the case of N dimensions, the position operator has N components $(\hat{x}_1(t),...,\hat{x}_N(t))$ and the condition for the positivity of the joint densities is the same as in the one-dimensional case.

A. The one-dimensional case

1. Calculus of ρ(x,t)

In order to find the densities of the random variables in the process X(t), we assume that $\hat{\rho}$ is positive first as in (2). Thus we have

$$\hat{
ho} = \sum_{k=1}^{\infty} \lambda_k |\psi_k
angle \langle \psi_k |$$

with $\lambda_k \ge 0$, and (2) gives

$$E\{e^{i\alpha X(t)}\} = \int_{-\infty}^{\infty} e^{i\alpha x} \rho(x,t) dx = \operatorname{tr}\{\hat{\rho}e^{i\alpha x(t)}\}$$
$$= \sum_{k=1}^{\infty} \lambda_k \langle \psi_k | e^{i\alpha \hat{x}(t)} | \psi_k \rangle.$$
(49)

In the Schrödinger representation for states and observables, ψ_k is a function in $L^2(R)$ and $\hat{x}(0)$ is the multiplication by the variable x. Furthermore, if we use the Schrödinger picture, we have that

$$\langle \psi_k | e^{ia\hat{x}(t)} | \psi_k \rangle = \int_{-\infty}^{\infty} |\psi_k(x,t)|^2 e^{iax} \, dx.$$
 (50)

Hence

$$\int_{-\infty}^{\infty} e^{i\alpha x} \rho(x,t) dx = \sum_{k=1}^{\infty} \lambda_k \int_{-\infty}^{\infty} |\psi_k(x,t)|^2 e^{i\alpha x} dx.$$
(51)

Since $\lambda_k \ge 0$ and $\sum_{k=1}^{\infty} \lambda_k = 1$, we can apply the monotonic convergence theorem so as to find the following result:

$$\int_{-\infty}^{\infty} e^{i\alpha x} \rho(x,t) dx = \int_{-\infty}^{\infty} \sum_{k=1}^{\infty} \lambda_k |\psi_k(x,t)|^2 e^{i\alpha x} dx.$$
(52)

Hence

$$o(x,t) = \sum_{k=1}^{\infty} \lambda_k |\psi_k(x,t)|^2.$$
 (53)

Note that this result is independent on the evolution Hamiltonian H. The case in which $\hat{\rho}$ is normal but not positive is not more involved and yields the same result. However, $\rho(x,t)$ is positive if $\hat{\rho}$ is positive, as is clear from (53).

2. Calculus of the joint densities for the free particle

In the present case the Hamiltonian is $H = \hat{p}^2/2m$ and the position and the momentum operators in the Heisenberg picture are

$$\hat{x}(t) = \hat{x}(0) + \hat{p}(0)(t/m), \quad \hat{p}(t) = \hat{p}(0).$$
 (54)

To find the joint densities we make use of Eq. (6). Relations (54) give

$$tr\{\hat{\rho} \exp(i\{\alpha_1 \hat{x}(t_1) + \dots + \alpha_n \hat{x}(t_n)\})\}$$

= $tr\{\hat{\rho} \exp(i\{(\alpha_1 + \dots + \alpha_n) \hat{x} + (1/m)(\alpha_1 t_1 + \dots + \alpha_n t_n) \hat{p}\})\},$ (55)

where $\hat{x} = \hat{x}(0)$ and $\hat{p} = \hat{p}(0)$ for simplicity. If we write

$$\alpha = \sum_{k=1}^{n} \alpha_k \quad \text{and} \quad \beta = \frac{1}{m} \sum_{k=1}^{n} \alpha_k t_k, \quad (56)$$

then

$$(55) = \operatorname{tr}\{\hat{\rho} e^{i\{\alpha \hat{x} + \beta \hat{p}\}}\}.$$
(57)

Therefore, (6) yields

$$\rho(x_1,t_1;...; ;x_n,t_n) = \frac{1}{(2\pi)^n} \int_{\mathbb{R}^n} d\alpha_1,...,d\alpha_n \times \exp(-i(\alpha_1 x_1 + \cdots + \alpha_n x_n)) \operatorname{tr}\{\hat{\rho} e^{i(\alpha \hat{x} + \beta \hat{\rho})}\},$$
(58)

where β contains the dependence on $t_1, ..., t_n$.

To perform the integration in (58), we have to change variables:

$$\alpha_{1}x_{1} + \dots + \alpha_{n}x_{n}$$

$$= \alpha A + \beta B + \alpha_{3}C_{3} + \dots + \alpha_{n}C_{n}$$

$$= \alpha_{1}A + \dots + \alpha_{n}A + (\alpha_{1}t_{1}/m)B + \dots$$

$$+ (\alpha_{n}t_{n}/m)B + \alpha_{3}C_{3} + \dots + \alpha_{n}C_{n}.$$
(59)

Since $\alpha_1, ..., \alpha_n$ are arbitrary real numbers, we obtain the following equations:

Hence

$$\rho(x_{1},t_{1},...;x_{n},t_{n}) = \frac{1}{(2\pi)^{n}} \frac{m}{|t_{2}-t_{1}|} \int_{\mathbb{R}^{n}} d\beta_{1},...,d\beta_{n} \exp(-i(\beta_{1}S_{1}+\cdots+\beta_{n}S_{n}))\operatorname{tr}\{\hat{\rho}\exp(i(\beta_{1}\hat{x}+\beta_{2}\hat{p}))\}$$

$$= \frac{1}{(2\pi)^{2}} \frac{m}{|t_{2}-t_{1}|} \int_{\mathbb{R}^{2}} d\beta_{1} d\beta_{2} \exp(-i(\beta_{1}S_{1}+B_{2}S_{2}))\operatorname{tr}\{\hat{\rho}\exp(i(\beta_{1}\hat{x}+\beta_{2}\hat{p}))\}$$

$$\times \frac{1}{2\pi} \int_{-\infty}^{\infty} d\beta_{3} e^{-i\beta_{3}S_{3}} \times \cdots \times \frac{1}{2\pi} \int_{-\infty}^{\infty} d\beta_{n} e^{-i\beta_{n}S_{n}}$$

$$= \frac{1}{(2\pi)^{2}} \frac{m}{|t_{2}-t_{1}|} W_{\hat{\rho}}(S_{1},S_{2},0) \prod_{k=3}^{n} \delta(S_{k}), \qquad (64)$$

where

$$W_{\hat{\rho}}(S_{1},S_{2},0) = \int_{R^{2}} d\beta_{1} d\beta_{2} \exp(-i(\beta_{1}S_{1} + \beta_{2}S_{2})) \\ \times \operatorname{tr}\{\hat{\rho} \exp(i(\beta_{1}\hat{x} + \beta_{2}\hat{p}))\}$$
(65)

is the Wigner function associated to the state $\hat{\rho}$.

From (64) and (65) we conclude that the process X(t) is veritable if and only if the following propositions are fulfilled:

(1) the Wigner function for $\hat{\rho}$ is positive at t = 0, and (2) the sum in (53) is positive.

This happens in particular if $\hat{\rho}$ is a Gaussian pure state.¹⁵

$$\begin{aligned} A + t_1 B / m &= x_1, \\ A + t_2 B / m &= x_2, \\ A + t_3 B / m + C_3 &= x_3, \\ \vdots \end{aligned}$$
(60)

$$A + t_n B / m + C_n = x_n$$

from which we find the values of A, B, and C_k :

$$A = x_{1} - t_{1}[(x_{2} - x_{1})/(t_{2} - t_{1})],$$

$$B = m[(x_{2} - x_{1})/(t_{2} - t_{1})],$$

$$C_{k} = x_{k} - x_{1} - (t_{k} - t_{1})$$

$$\times [(x_{2} - x_{1})/(t_{2} - t_{1})], \quad k \ge 2.$$
(61)

Here B is the mean value of the momentum of a classical free particle with mass m moving from x_1 to x_2 in a time interval $t_2 - t_1$, and A and C_k represent positions of this particle.

We relabel the variables as follows:

$$\alpha = \beta_1, \quad \beta = \beta_2, \quad \alpha_3 = \beta_3, \quad \dots, \quad \alpha_n = \beta_n; \\ A = S_1, \quad B = S_2, \quad C_3 = S_3, \quad \dots, \quad C_n = S_n.$$
 (62)

In order to write the integral in (58) in terms of these new variables, we first need to find the Jacobian corresponding to the change of variables (62). The inverse of this Jacobian is

$$\begin{vmatrix} 1 & 1 & 1 & \cdots & 1 \\ t_1/m & t_2/m & t_3/m & \cdots & t_n/m \\ 0 & 0 & 1 & 0 \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & 0 & 0 & 1 \end{vmatrix} = (t_2 - t_1)/m.$$
(63)

3. The case of the harmonic oscillator

Here
$$H = \hat{p}^2/2m + (mw^2/2)\hat{x}^2$$
 and
 $\hat{x}(t) = \hat{x}(0)\cos wt + \hat{p}(0)(\sin wt)/mw,$
 $\hat{p}(t) = -\hat{x}(0)mw \sin wt + \hat{p}(0)\cos wt.$
(66)

The procedures to find $\rho(x_1,t_1;...;x_n,t_n)$ are always the same. After a more or less lengthy calculation, we find

$$\rho(x_{1},t_{1};...;x_{n},t_{n}) = \frac{1}{(2\pi)^{2}} \frac{mw}{|\sin[w(t_{2}-t_{1})]|} \prod_{k=3}^{n} \delta(S_{k}) W_{\hat{\rho}}(S_{1},S_{2},0),$$
(67)

which is well defined provided that $w(t_2 - t_1) \neq \pi k$ with k integer and also that the S_k be well defined. We see again that a sufficient condition for X(t) to be veritable is the positivity of the Wigner function associated to $\hat{\rho}$ at t = 0. Here, the S_k have the following form:

$$S_{1} = \frac{x_{1}}{\cos(wt_{1})} \left\{ \frac{\tan(wt_{2})}{\tan(wt_{2}) - \tan(wt_{1})} \right\}$$
$$-\frac{x_{2}}{\cos(wt_{2})} \left\{ \frac{\tan(wt_{1})}{\tan(wt_{2}) - \tan(wt_{1})} \right\},$$
$$S_{2} = \frac{mw}{\tan(wt_{2}) - \tan(wt_{1})} \left\{ \frac{x_{2}}{\cos(wt_{2})} - \frac{x_{1}}{\cos(wt_{1})} \right\},$$
$$S_{k} = x_{k} + x_{1} \left\{ \frac{\sin(wt_{k})}{\cos(wt_{1})} - \frac{\cos(wt_{k})}{\cos(wt_{1})} \tan(wt_{2}) \right\}$$

$$\times \frac{1}{\tan(wt_2) - \tan(wt_1)}$$

$$+ x_2 \left\{ \frac{\cos(wt_k)}{\cos(wt_2)} \tan(wt_1) - \frac{\sin(wt_k)}{\cos(wt_2)} \right\}$$

$$\times \frac{1}{\tan(wt_2) - \tan(wt_2)}, \quad k \ge 2.$$
(68)

If the Hamiltonian was of the form $H = \hat{p}^2 - \hat{x}^2$ or $H = \hat{p}^2 + \alpha \hat{x}$, we would obtain a similar result. Thus we could conjecture that if H were any quadratic Hamiltonian we would arrive at the same conclusion. This conjecture is true and we shall prove it in the following two subsections.

4. The general quadratic Hamiltonian in one dimension

Assume that $H = \alpha \hat{p}^2 + \beta \hat{x}^2 + \gamma (\hat{x}\hat{p} + \hat{p}\hat{x}) + \delta \hat{x} + \eta \hat{p}$ + ϵ . This Hamiltonian gives us a simple behavior for $\hat{x}(t)$. After having solved the equations of motion, we obtain

$$\hat{x}(t) = f(t)\hat{x}(0) + g(t)\hat{p}(0) + h(t)I,$$
(69)

where f(t), g(t), and h(t) are real functions on t, and I is the identity on \mathcal{H} .

In this situation we arrive at

$$\rho(x_1t_1;...;x_n,t_n) = \frac{1}{(2\pi)^2} \frac{1}{|g(t_2)f(t_1) - g(t_1)f(t_2)|} \prod_{k=3}^n \delta(S_k) W_{\hat{\rho}}(S_1,S_2,0),$$
(70)

where

$$S_1 = \frac{[x_2 - h(t_2)]g(t_1) - [x_1 - h(t_1)]g(t_2)}{g(t_1)f(t_2) - g(t_2)f(t_1)},$$

$$S_{2} = \frac{[x_{1} - h(t_{1})]f(t_{2}) - [x_{2} - h(t_{2})]f(t_{1})}{g(t_{1})f(t_{2}) - g(t_{2})f(t_{1})},$$
(71)

$$S_{k} = x_{k} - h(t_{k}) - f(t_{k}) \left\{ \frac{[x_{2} - h(t_{2})]g(t_{1}) - [x_{1} - h(t_{1})]g(t_{2})}{g(t_{1})f(t_{2}) - g(t_{2})f(t_{1})} \right\} - g(t_{k}) \left\{ \frac{[x_{1} - h(t_{1})]f(t_{2}) - [x_{2} - h(t_{2})]f(t_{1})}{g(t_{1})f(t_{2}) - g(t_{2})f(t_{1})} \right\}.$$

Equation (70) leads to the same conclusion: if $\hat{\rho}$ is positive and $g(t_2)f(t_1) \neq g(t_1)f(t_2)$, the finite-dimensional distribution functions are positive if and only if the Wigner function for $\hat{\rho}$ is positive at t = 0 ($n \ge 2$).

B. The N-dimensional case

Here we want to obtain the density given by formula (10). First, we have to find a new expression for the trace,

$$tr\{\hat{\rho} \exp\{i(\alpha_{11}\hat{x}_{1}(t_{1}) + \cdots + \alpha_{Ns}\hat{x}_{N}(t_{s}))\}\}, \qquad (72)$$

for which we need to write $\hat{x}_k(t)$ as a function of the coordinates of the position $(\hat{x}_1(0),...,\hat{x}_n(0))$ and the momentum $(\hat{p}_1(0),...,\hat{p}_n(0))$ at t = 0.

The transition from $\hat{x}(0)$ to $\hat{x}(t)$ is due to a dynamics

given by a quadratic Hamiltonian which is $H = \mu' B \mu + C' \mu + D$, where

$$\mu = \begin{pmatrix} \hat{x}_1(t) \\ \vdots \\ \hat{x}_N(t) \\ \hat{p}_1(t) \\ \vdots \\ \hat{p}_N(t) \end{pmatrix}.$$

Here B is a symmetric $2N \times 2N$ real matrix, C is a $2N \times 1$ real matrix, and D is a real function. The equations of the motion are $(\hbar = 1)$

$$i\dot{\hat{x}}_{k}(t) = [\hat{x}_{k}(t), H], \quad i\dot{\hat{p}}_{k}(t) = [p_{k}(t), H].$$
 (73)

Equations (73) yield the following linear system:

$$\begin{pmatrix} \hat{x}_{1}(t) \\ \vdots \\ \hat{x}_{N}(t) \\ \hat{p}_{1}(t) \\ \vdots \\ \hat{p}_{N}(t) \end{pmatrix} = M \begin{pmatrix} \hat{x}_{1}(t) \\ \vdots \\ \hat{x}_{N}(t) \\ \hat{p}_{1}(t) \\ \vdots \\ \hat{p}_{N}(t) \end{pmatrix} + T.$$
(74)

Here M is a $2N \times 2N$ real matrix and T is a $2N \times 1$ real matrix. The system in (74) can be easily integrated. Its solutions are of the form

$$\hat{x}_{i}(t) = \sum_{k=1}^{N} \left\{ f_{ik}(t) \hat{x}_{k}(0) + g_{ik}(t) \hat{p}_{k}(0) \right\} + h_{i}(t) I$$
(75)

and (72) becomes

$$tr\{\hat{\rho} \exp(i\{\gamma_{1}\hat{x}_{1}(0) + \dots + \gamma_{N}\hat{x}_{N}(0) + \delta_{1}\hat{p}_{1}(0) + \dots + \delta_{N}\hat{p}_{N}(0) + \xi I\})\},$$
(76)

where $\gamma_1,...,\gamma_N$, $\delta_1,...,\delta_N$, depend on $t_1,...,t_s$ and $\alpha_{11},...,\alpha_{Ns}$. This dependence is given by a suitable change of variables transforming (10) into the following formula for the joint densities:

$$\rho(x_{11},...,x_{N1},t_{1};...;x_{1s},...,x_{Ns},t_{s}) = \frac{|J|}{(2\pi)^{N\times s}} \int dr_{1}\cdots dr_{Ns} \exp(-i(r_{1}q_{1}+\cdots+r_{Ns}q_{Ns})) \times \operatorname{tr}\{\hat{\rho}\exp(i\{r_{1}\hat{x}_{1}(0)+\cdots+r_{N}\hat{x}_{N}(0)+r_{N+1}\hat{p}_{1}(0)+\cdots+r_{2N}\hat{p}_{N}(0)+r_{2N+1}I\})\} = \frac{|J|}{(2\pi)^{2N}} \delta(1-q_{2N+1}) \sum_{k=2N+2}^{N\times s} \delta(q_{k}) W_{\hat{\rho}}(q_{1,q_{N+1}};...;q_{N},q_{N+N};0), r_{i} = \gamma_{i}, \quad i = 1,...,N, \quad r_{k} = \delta_{k-N}, \quad k = N+1,...,2N, \quad r_{2N+1} = \xi,$$
(77)

where

$$W_{\hat{\rho}}(q_{1},q_{N+1};\ldots;q_{N},q_{N+N};0)$$

$$= \int d\gamma_{1},\ldots,d\gamma_{N}d\delta_{1},\ldots,d\delta_{N}$$

$$\times \exp(-i(\gamma_{1}q_{1}+\cdots+\delta_{N}q_{2N}))$$

$$\times \operatorname{tr}\{\hat{\rho}\exp(i\{\gamma_{1}\hat{x}_{1}(0)+\cdots+\delta_{N}\hat{\rho}_{N}(0)\})\}$$
(78)

is the N-dimensional Wigner function at t = 0. Here |J| is the Jacobian of the change of the variables. It depends on t_1, \dots, t_s and on the Hamiltonian H as well as the q_k .

A detailed calculation of the q_k is not necessary to conclude that, after (77), the joint densities are positive if and only if the N-dimensional Wigner function is positive at t = 0.

C. Joint densities in phase space

Assume that the stochastic vector (X(t),P(t)) and $(\hat{x}(t),\hat{p}(t))$ are related through a GWC, where $\hat{x}(t)$ and $\hat{p}(t)$ are again the position and momentum, respectively, of a quantum Hamiltonian. The stochastic vector (X(t),P(t)) is veritable if and only if the joint densities $\rho(x_1,p_1,t_1;...;x_m,p_m,t_m)$ are positive. Performing the same kind of calculations, we find out that the positive of the densities is equivalent to the positivity of the corresponding Wigner function at t = 0. This result is immediately extended to the N-dimensional case.

V. TIME-INDEPENDENT HAMILTONIANS IN ONE DIMENSION

So far, we have studied the case in which H is a quadratic Hamiltonian. The objective of the present section is the study of the situation produced by a general time-independent Hamiltonian in one dimension and of the form $H = p^2/2m + V(x)$. Quantum mechanics gives us the time development for the observable position as

$$\hat{x}(t) = e^{iHt} \hat{x}(0) e^{-iHt}.$$
(79)

Assume that $\hat{x}(0)$ and H have a common dense domain \mathscr{D} stable under the action of these operators. From (79) we conclude that $\hat{x}(t)$ is strongly differentiable as a function of t and the following equations make sense on \mathscr{D} :

$$i\hat{x}(t) = [\hat{x}(t),H],$$

$$\hat{x}(t + \Delta t) = x(t) - i[\hat{x}(t),H]\Delta t + O(\Delta t).$$
Since $[\hat{x}(t),H] = i\hat{p}(t)/m$, (80) yields
$$\hat{x}(t + \Delta t) = \hat{x}(t) + (\Delta t/m)\hat{p}(t) + O(\Delta t).$$
(81)

Now, we are seeking a necessary condition for X(t) to be veritable. Equation (6) gives us the following equation:

$$\rho(x,t;x',t + \Delta t)$$

$$= \frac{1}{(2\pi)^2} \int d\alpha_1 \, d\alpha_2 \exp(-i(\alpha_1 x + \alpha_2 x'))$$

$$\times \operatorname{tr} \{ \hat{\rho} \exp(i\{\alpha_1 \hat{x}(t) + \alpha_2 \hat{x}(t + \Delta t)\}) \}$$

$$= \frac{1}{(2\pi)^2} \int d\alpha_1 \, d\alpha_2 \exp(-i(\alpha_1 x + \alpha_2 x'))$$

$$\times \operatorname{tr} \{ \hat{\rho} \exp(i\{(\alpha_1 + \alpha_2) \hat{x}(t) + (\alpha_2 \Delta t / m) \hat{\rho}(t) + O(\Delta t)\}) \}, \qquad (82)$$

where we have made use of (81). Now, we perform the change of variables

$$\alpha = \alpha_1 + \alpha_2, \quad \beta = \alpha_2(\Delta t / m),$$
 (83)
whose Jacobian is $\Delta t / m$. Thus

 $\rho(x,t;x',t+\Delta t)$

$$= \frac{1}{(2\pi)^2} \frac{m}{|\Delta t|} \int d\alpha \, d\beta$$

$$\times \exp(-i\{\alpha x + \beta \left[(m/\Delta t) (x' - x) \right] \})$$

$$\times tr\{\hat{\rho} \exp(i\{\alpha \hat{x}(t) + \beta \hat{p}(t) + O(\Delta t)\})\}.$$
(84)

From a physical point of view, if X(t) is veritable, $\rho(x,t;x',t + \Delta t)$ is the joint probability density of finding the corresponding classical particle at x at the time t and x' at the time $t + \Delta t$. Therefore, the joint density $\rho(x,t;x',t)$ does not have physical meaning when $x \neq x'$. In addition, if $\Delta t \rightarrow 0$ the point x' has to approach x, if we want to be consistent with this physical interpretation. Thus $p(\Delta t) = m(x' - x)/\Delta t$ goes to the classical momentum of the particle at x when $\Delta t \rightarrow 0$.

Obviously, if $\rho(x,t;x',t + \Delta t)$ has to be positive for any values of t, Δt , x, and x', it has also to be positive at the limit when $\Delta t \rightarrow 0$. Hence

$$0 \leq \lim_{\Delta t \to 0} \int d\alpha \, d\beta \, \exp(-i\{\alpha x + \beta m/\Delta t(x' - x)\})$$

$$\times \operatorname{tr}\{\hat{\rho} \, \exp(i\{\alpha \hat{x}(t) + \beta \hat{p}(t) + O(\Delta t)\})\}$$

$$= \int d\alpha \, d\beta \, \exp(-i\{\alpha x + \beta p\})$$

$$\times \operatorname{tr}\{\hat{\rho} \, \exp(i\{\alpha \hat{x}(t) + \beta \hat{p}(t)\})\}$$

$$= W_{\hat{\rho}}(x, p, t).$$
(85)

Thus a necessary condition for X(t) to be veritable is that the Wigner function for $\hat{\rho}$ be positive at any value of the time.

When $\hat{\rho}$ is a pure state, Soto and Claverie¹⁶ have proved the following statement: The positivity of the Wigner function at t = 0 guarantees its positivity at any t > 0 if H is a quadratic Hamiltonian of the type $\alpha p^2 + \beta q^2 + \gamma q + \delta$. Conversely, if H is not of this kind, the two-dimensional joint functions are not positive definite for any value of time, and, therefore, X(t) cannot be veritable, if $\hat{\rho}$ is a pure state.

To close this section, we give here a new characterization of the Wigner function. Consider the following function:

$$F(x,p,t) = \lim_{\Delta t \to 0^+} \int_{-\infty}^{x} dx'$$
$$\times \int_{-\infty}^{x'+p\Delta t/m} dx'' \rho(x',t;x'',t+\Delta t)$$
$$= \lim_{\Delta t \to 0^+} F_{\Delta t}(x,p,t).$$
(86)

Under sufficient regularity conditions, we can take on $F_{\Delta t}(x,p,t)$ partial derivatives with respect to the variables x and p, so as to have

$$\frac{\partial^2 F_{\Delta t}(x,p,t)}{\partial x \, \partial p} = \frac{\Delta t}{m} \rho \left(x,t; x + p \, \frac{\Delta t}{m} , t + \Delta t \right)$$
$$= \frac{1}{(2\pi)^2} \frac{\Delta t}{|\Delta t|} \int d\alpha \, d\beta \, e^{-i\{\alpha x + \beta p\}}$$
$$\times \operatorname{tr} \{ \hat{\rho} \exp(i\{\alpha \hat{x}(t) + \beta \hat{p}(t) + O(\Delta t)\}) \}.$$
(87)

2971 J. Math. Phys., Vol. 28, No. 12, December 1987

Therefore

$$\lim_{\Delta t \to 0^+} \frac{\partial F_{\Delta t}(x,p,t)}{\partial x \, \partial p} = \frac{1}{(2\pi)^2} W_{\hat{\rho}}(x,p,t).$$
(88)

Note that, if the two-dimensional joint densities are positive, the Wigner function is also positive.

ACKNOWLEDGMENTS

We are gratefully in debt to Professor Marí Cruz Valsero and Professor César Palencia for helpful discussions.

APPENDIX A: PROOF OF LEMMA II

Let us give the proof for n = 1. For higher values of n, this proof does not involve new arguments. Thus assume that

$$\operatorname{tr}\{\hat{\rho}|A|\} < \infty. \tag{A1}$$

Since A is self-adjoint, the spectral theorem gives a positive measure μ_k for each $\psi_k \in \mathscr{D}(A)$ such that

$$\langle \psi_k | A \psi_k \rangle = \int_{-\infty}^{\infty} \lambda \, d\mu_k \tag{A2}$$

and

$$\langle \psi_k | |A|\psi_k \rangle = \int_{-\infty}^{\infty} |\lambda| d\mu_k.$$
 (A3)

Let

$$\hat{\rho} = \sum_{k=1}^{\infty} \lambda_k |\psi_k\rangle \langle \psi_k|$$

with $\psi_k \in \mathscr{D}(A)$ and assume first that $\hat{\rho}$ is positive. Then, we have

$$\operatorname{tr}\{\hat{\rho}|A|\} = \sum_{k=1}^{\infty} \lambda_k \int_{-\infty}^{\infty} |\lambda| d\mu_k.$$
 (A4)

Since $\sum_{k=1}^{\infty} \lambda_k < \infty$, the Lebesgue theorem gives

$$(\mathbf{A4}) = \int_{-\infty}^{\infty} |\lambda| \sum_{k=1}^{\infty} \lambda_k d\mu_k = \int_{-\infty}^{\infty} |\lambda| d\mu, \quad (\mathbf{A5})$$

where $d\mu = \sum_{k=1}^{\infty} \lambda_k d\mu_k$ is a finite measure on R. Now take

$$\operatorname{tr}\{\hat{\rho}e^{i\alpha A}\} = \sum_{k=1}^{\infty} \lambda_{k} \langle \psi_{k} | e^{i\alpha A} \psi_{k} \rangle$$
$$= \sum_{k=1}^{\infty} \lambda_{k} \int_{-\infty}^{\infty} e^{i\alpha \lambda} d\mu_{k} = \int_{-\infty}^{\infty} e^{i\alpha \lambda} d\mu. \quad (A6)$$

Our next goal is to obtain the derivative of (A6) with respect to α at zero. This derivative can be introduced inside the last integral in (A6) if

(1)
$$\int_{-\infty}^{\infty} |i\lambda|| e^{i\alpha\lambda} |d\mu < \infty, \qquad (A7)$$

(2)
$$\left|\frac{\partial}{\partial \alpha}e^{i\alpha\lambda}\right| \leq f(\lambda)\in L^{1}(d\mu).$$
 (A8)

Gadella et al. 2971

Here (1) and (2) are fulfilled, since $f(\lambda) = \lambda$ and $\int_{-\infty}^{\infty} |\lambda| d\mu < \infty$. Thus

$$\frac{d}{d\alpha} \operatorname{tr} \hat{\rho} e^{i\alpha A} \Big|_{\alpha = 0}$$

$$= \frac{d}{d\alpha} \int_{-\infty}^{\infty} e^{i\alpha \lambda} d\mu \Big|_{\alpha = 0}$$

$$= \int_{-\infty}^{\infty} i\lambda \ d\mu = i \sum_{k=1}^{\infty} \lambda_k \int_{-\infty}^{\infty} d\mu_k = i \operatorname{tr} \{ \hat{\rho} A \}.$$
(A9)

If $\hat{\rho}$ were any normal trace class operator, the procedures of the demonstration would be identical. The only peculiarity to be noted in such a case is that μ would then be a complex bounded measure.

APPENDIX B: PROOF OF LEMMA III

Assume that $\hat{\rho}$ is positive. Then,

$$\operatorname{tr}\{\hat{\rho}A^{2}\} = \sum_{k=1}^{\infty} \lambda_{k} \langle \psi_{k} | A^{2} \psi_{k} \rangle = \sum_{k=1}^{\infty} \lambda_{k} \| A \lambda_{k} \|^{2}, \qquad (B1)$$

$$tr\{\hat{\rho}B^{2}\} = \sum_{k=1}^{\infty} \lambda_{k} \|B\psi_{k}\|^{2},$$
(B2)

$$\operatorname{tr}\{\hat{\rho}AB\} = \sum_{k=1}^{\infty} \lambda_k \langle \psi_k | AB\psi_k \rangle = \sum_{k=1}^{\infty} \lambda_k \langle A\psi_k | B\psi_k \rangle,$$
(B3)

$$|\mathrm{tr}\{\hat{\rho}AB\}| \leq \sum_{k=1}^{\infty} \lambda_k ||A\psi_k|| ||B\psi_k||. \tag{B4}$$

We want to find a upper bound for the rhs in (B4). This can be easily done by noting that if a and b are positive numbers, we have

$$0 \le ab \le \frac{1}{2}(a^2 + b^2).$$
 (B5)

Thus,

$$0 \leq \sum_{k=1}^{\infty} \lambda_{k} \|A\psi_{k}\| \|B\psi_{k}\|$$

$$\leq \frac{1}{2} \left\{ \sum_{k=1}^{\infty} \lambda_{k} (\|A\psi_{k}\|^{2} + \|B\psi_{k}\|^{2}) \right\}$$

$$= \frac{1}{2} \left\{ \sum_{k=1}^{\infty} \lambda_{k} \|A\psi_{k}\|^{2} + \sum_{k=1}^{\infty} \lambda_{k} \|B\psi_{k}\|^{2} \right\}.$$
(B6)

If $\hat{\rho}$ were not positive, we would have to assume that

$$\sum_{k=1}^{\infty} |\lambda_k| \|A\psi_k\|^2$$

and

$$\sum_{k=1}^{\infty} |\lambda_k| \|B\psi_k\|^2$$

are both finite.

¹M. Gadella and E. Santos, J. Math. Phys. 22, 1651 (1981).

²M. Gadella, J. Math. Phys. 24, 534 (1983).

³M. Gadella, Nuovo Cimento B 63, 510 (1981).

⁴H. Weyl, *Theory of Groups and Quantum Mechanics* (Dover, New York, 1931).

⁵S. de Groot, La Transformation de Weyl et la Fonction de Wigner: Une forme Alternative de la Mécanique Quantique (Université de Montreal Press, Montreal, 1974).

⁶G. S. Agarwal and E. Wolf, Phys. Rev. D 2, 161 (1970).

⁷J. L. Doob, Stochastic Processes (Wiley, New York, 1953).

⁸Let $\Omega_t = R$, $\forall t \in J$ and consider $\Omega = \Pi_{t \in J} \Omega_t$, with the usual product topology. Let X(t) be the t th projection from Ω onto Ω_t . It is easy to prove that

$$\Delta = \{w \in \Omega / [X(t)](w) = [X(t_0)](w) \ \forall t \in J\}$$

is a closed set in Ω . Therefore if we consider in Ω its Borel σ algebra β , Δ is a measurable set. On (Ω,β) we can give a probability measure such that $P(\Delta) = 1$. The process X(t) fulfills the required condition.

- ⁹R. B. Ash, Real Analysis and Probability (Academic, New York, 1972).
- ¹⁰T. T. Soong, Random Differential Equations in Science and Engineering (Academic, New York, 1973).

¹¹A. Friedman, Stochastic Differential Equations and Applications (Academic, New York, 1975).

- ¹²M. Reed and B. Simon, *Functional Analysis* (Academic, New York, 1972).
- ¹³J. C. Burkill and H. Burkill, A Second Course in Mathematical Analysis (Cambridge U.P., Cambridge, 1970).
- ¹⁴A. O. Barut and R. Rayczka, Theory of Group Representations and Applications (Polish Scientific, Warsaw, 1977).
- ¹⁵F. Soto and P. Claverie, J. Math. Phys. 24, 97 (1983).
- ¹⁶F. Soto and P. Claverie, J. Math. Phys. 24, 1104 (1983).

Effective potential determining one-dimensional Slater sum in independentparticle theory

N. H. March

Theoretical Chemistry Department, University of Oxford, Oxford OX1 3TG, England

(Received 16 January 1987; accepted for publication 19 August 1987)

Hilton, March, and Curtis [Proc. R. Soc. London Ser. A **300**, 391 (1967)] have earlier focused on the utility of the effective potential $U(\mathbf{r},\beta)$ in determining the Slater sum $Z(\mathbf{r},\beta) = Z_0(\beta)\exp(-\beta U(\mathbf{r},\beta))$. Here an explicit, though highly nonlinear, partial differential equation is derived for determining the effective potential $U(x,\beta)$ in one-dimensional problems. Direct solution of this equation by power series expansion in β leads readily to Husimi's results obtained from off-diagonal density matrix calculations. Perturbation theory in the potential is also developed, and thereby it is shown that an infinite subseries of the Husimi expansion is readily summed, and that a scaling property is exhibited.

I. INTRODUCTION

This work is concerned with some formally exact results for the effective potential¹ $U(x,\beta)$ which determines the one-dimensional Slater sum

$$Z(x,\beta) = Z_0(\beta) \exp(-\beta U(x,\beta)).$$
(1)

In terms of the eigenfunctions $\psi_i(x)$ and the corresponding eigenvalues ϵ_i of the one-particle Hamiltonian

$$H = -(\hbar^2/2m)\nabla^2 + V(x), \qquad (2)$$

the Slater sum is defined by

2973

$$Z(x,\beta) = \sum_{i} \psi_i^*(x) \psi_i(x) \exp(-\beta \epsilon_i), \quad \beta = (k_{\rm B} T)^{-1},$$
(3)

 $Z_0(\beta)$ in Eq. (1) being its free-particle value corresponding to V(x) = 0.

As was proved by March and Murray² in their treatment of central field problems, which, of course, corresponds to dealing with one-dimensional Schrödinger equations, the Bloch equation satisfied by the off-diagonal density matrix $C(x,x_0,\beta)$, defined by replacing $\psi_i(x)$ in Eq. (3) by $\psi_i(x_0)$,

$$HC = -\frac{\partial C}{\partial \beta},\tag{4}$$

can be expanded about its diagonal $x_0 = x$ to yield the following partial differential equation for the Slater sum $Z(x,\beta)$:

$$\frac{\hbar^2}{8m}Z''' - \frac{\partial Z'}{\partial \beta} - VZ' - \frac{1}{2}V'Z = 0.$$
⁽⁵⁾

Though Eq. (5) is linear in Z, it has so far only proved possible to solve it for specific potentials V(x), e.g., the harmonic oscillator with $V(x) = \frac{1}{2}m\omega^2 x^2$ in closed form. Since, for many purposes, one must have recourse to approximate methods, Ref. 1 will be followed here in seeking to determine $U(x,\beta)$. While in Ref. 1, and also in earlier work by Husimi,³ $U(x,\beta)$ was calculated by expansions either in V or in β , but employing off-diagonal knowledge of $C(x,x_0,\beta)$, current interests in density functional theory prompt the investigation reported below of $U(x,\beta)$ from a purely diagonal approach.

II. DIAGONAL DIFFERENTIAL EQUATION FOR THE EFFECTIVE POTENTIAL

Returning to Eq. (1), and noting that the free-particle Slater sum $Z_0(\beta)$ can be found by inserting free-particle waves $\exp(ikx)$ and energies $\frac{\pi^2 k^2}{2m}$ into Eq. (3) to yield

$$Z_0(\beta) = (m/2\pi\hbar^2\beta)^{1/2},$$
 (6)

one can insert Eqs. (1) and (6) into Eq. (5) to obtain the defining equation for $U(x,\beta)$ itself as

$$\frac{\hbar^2}{8m}\beta U''' - \beta \frac{\partial U'}{\partial \beta} + \frac{1}{2}V' - \frac{1}{2}U'$$

$$= \beta U'(V-U) - \beta^2 U' \frac{\partial U}{\partial \beta}$$

$$+ \frac{3\hbar^2}{8m}\beta^2 U'U'' - \frac{\hbar^2}{8m}\beta^3 U'^3.$$
(7)

This purely diagonal equation for the effective potential $U(x,\beta)$ can be viewed as the diagonal counterpart of the offdiagonal equation derived by Husimi³ from the Bloch equation (4), and solved by him via a power series expansion in β .

III. APPROXIMATE SERIES SOLUTIONS OF THE DIAGONAL EQUATION FOR $U(x,\beta)$

Notwithstanding the highly nonlinear character of Eq. (7), let us first demonstrate, in establishing its usefulness, that all of Husimi's results are readily regained from it. Writing

$$U(x,\beta) = u_0(x) + \beta u_1(x) + \beta^2 u_2(x) + \cdots, \qquad (8)$$

one finds immediately from the O(1) terms in Eq. (7) that

$$u_0'(x) = V'(x),$$
 (9)

showing that the Thomas-Fermi approximation $U(x,\beta) \simeq V(x)$ is regained in this lowest-order treatment.

Proceeding to terms of $O(\beta)$, one finds similarly

$$(\hbar^2/8m)V''' - \frac{3}{2}u_1' = 0, \tag{10}$$

which reproduces immediately Husimi's first-order result

$$u_1(x) = (\hbar^2/12m) V''(x).$$
(11)

Continuing to $O(\beta^2)$ one has

$$(\hbar^2/8m)u_1''' - \frac{5}{2}u_2' = -2V'u_1 + (3\hbar^2/8m)V'V'' \qquad (12)$$

with solution, using Eq. (11) for u_1 , which is readily verified to be

$$u_2 = \frac{1}{240} (\hbar^2/m)^2 V^{iv}(x) - (\hbar^2/24m) \{V'(x)\}^2.$$
(13)

The next useful step is to note that the leading terms in u_0, u_1 , and u_2 , etc., are all first order in the potential V. These are readily summed to all orders in β by returning to the diagonal equation (7) and noting that the right-hand side is of second and higher order in V(x). Hence denoting the solution to first order in V by $U_1(x,\beta)$ one needs to solve

$$\frac{\hbar^2}{8m}\beta U_1'' - \beta \frac{\partial U_1}{\partial \beta} + \frac{1}{2}V - \frac{1}{2}U_1 = 0, \qquad (14)$$

after inserting appropriate physical boundary conditions. Taking the Fourier transform with respect to x, to yield $\tilde{U}_1(p,\beta)$, one finds for potentials V(x), which have a Fourier transform $\tilde{V}(p)$,

$$-\frac{\hbar^2 p^2 \beta}{8m} \tilde{U}_1 - \beta \frac{\partial \tilde{U}_1}{\partial \beta} + \frac{1}{2} \tilde{V} - \frac{1}{2} \tilde{U}_1 = 0.$$
(15)

Writing

$$\widetilde{U}_{1}(p,\beta) = \widetilde{V}(p)\widetilde{f}(p,\beta)$$
(16)

yields

$$\frac{\partial \tilde{f}}{\partial \beta} + \left[\frac{1}{2\beta} + \frac{\hbar^2 p^2}{8m}\right] \tilde{f} = \frac{1}{2\beta} \,. \tag{17}$$

Employing the integrating factor $\beta^{1/2} \exp((\hbar^2 p^2/8m)\beta)$, one finds the explicit form of $\tilde{f}(p,\beta)$ to be

$$\tilde{f}(p,\beta) = \beta^{-1/2} \exp\left(\frac{-\hbar^2 p^2 \beta}{8m}\right) \int^{\beta} \frac{1}{2} \beta^{1/2} \\ \times \exp\left(\frac{\hbar^2 p^2 \beta}{8m}\right) d\beta.$$
(18)

In coordinate space, the product form (16) evidently corresponds to a convolution,¹

$$U_{1}(x,\beta) = \int_{-\infty}^{\infty} V(x_{1})f(x_{1} - x_{3}\beta)dx_{1}, \qquad (19)$$

where $f(x_1 - x_1\beta)$ can be found from Eq. (18) as

$$f(x_1 - x_{\beta}) = [1/2\beta Z_0(\beta)] \operatorname{erfc}\{\sqrt{2}|x_1 - x|/\sqrt{\beta}\},$$
(20)

which, it is to be emphasized, has been derived by purely diagonal arguments. This result (19), with f given by (20), sums up the Husimi series in β to infinite order, to first order in V.

Evidently, returning to Eq. (7), one can determine the "correction" U_2 of $O(V^2)$ to U_1 in Eq. (19). The appropriate

differential equation is

$$\frac{\hbar^{2}}{8m}\beta U_{2}^{'''} - \beta \frac{\partial U_{2}'}{\partial \beta} - \frac{1}{2}U_{2}'$$

$$= \beta U_{1}'(V - U_{1}) - \beta^{2}U_{1}'\frac{\partial U_{1}}{\partial \beta} + \frac{3\hbar^{2}}{8m}\beta^{2}U_{1}'U_{1}''.$$
(21)

This can, in fact, be solved, but the detail proliferates. The only case, so far, in which a closed solution to Eq. (7) has been achieved is for the harmonic oscillator example referred to above. Here it is readily verified that the form

$$Z(x,\beta) = \left(\frac{m}{2\pi\hbar^2\beta}\right)^{1/2} \left(\frac{\beta\hbar\omega}{\sinh(\beta\hbar\omega)}\right)^{1/2} \\ \times \exp\left(-x^2 \frac{m\omega}{\hbar} \tanh\left(\frac{1}{2}\beta\hbar\omega\right)\right), \qquad (22)$$

derived by Uhlenbeck and Gropper⁴ by directly performing the sum (3) using the known wave functions and energy levels corresponds to

$$U(x,\beta) = \frac{m\omega^2 x^2}{\hbar\omega\beta} \tanh\left(\frac{1}{2}\,\hbar\omega\beta\right) - \frac{1}{2\beta}\ln\left(\frac{\hbar\omega\beta}{\sinh(\hbar\omega\beta)}\right)$$
(23)

and that this is an exact solution of Eq. (7) to all orders in V.

To summarize, Eq. (7) is the main result of the present work: it establishes a diagonal method for generalizing the Thomas-Fermi approximation $U(x,\beta) \simeq V(x)$, which is thereby demonstrated to be true in the limits of (a) sufficiently small β from Eqs. (8) and (9) for sufficiently slow spatial variations in V(x) from Eq. (7). In the latter connection, it is worth emphasizing that Eq. (13), with f given by Eq. (14), shows that in the expression for U_1 , the combination $V^{(2n)}(x)\beta^n$ always appears, with $V^{(n)}$ representing the *n*th derivative of V(x). Presumably, similar scaling will be derivable in each order of the perturbation series and this may point to further progress in solving the nonlinear equation (7) for the effective potential.

ACKNOWLEDGMENTS

This work was completed during sabbatical leave from Oxford spent at UCSB. It is a pleasure to thank Professor D. Hone for making this stay possible and for kind hospitality.

The Leverhulme Trustees are to be thanked for partial financial support for the visit.

⁴G. E. Uhlenbeck and L. Gropper, Phys. Rev. 41, 79 (1932).

¹D. Hilton, N. H. March, and A. R. Curtis, Proc. R. Soc. London Ser. A **300**, 391 (1967).

²N. H. March and A. M. Murray, Phys. Rev. 120, 830 (1960).

³K. Husimi, Proc. Phys. Math. Soc. Jpn. 22, 264 (1940).

Electron density and Slater sum of a semi-infinite electron gas in *d* dimensions

N. H. March

Theoretical Chemistry Department, University of Oxford, 1 South Parks Rd., Oxford OX1 3TG, England

(Received 4 February 1987; accepted for publication 19 August 1987)

By taking the electron densities of semi-infinite electron gases in one and in three dimensions, and forming the Slater sum by the Laplace transform, it is shown that the Slater sum is the classical partition function in d dimensions, times a function independent of dimensionality. The electron density is thereby calculated for general dimensionality, as is the kinetic energy density. As a by-product, the dimensionality dependence of Friedel oscillations emerges in general form.

I. INTRODUCTION

There is currently considerable interest in describing many-electron systems by their electron density ρ . It seemed, therefore, to be of interest to study the inhomogeneous semi-infinite electron gases as a functional of dimensionality.

To introduce the study, consider the simplest wave mechanical problem: electrons moving freely within a one-dimensional box of length l, in the region 0 < z < l. Then with Nelectrons singly filling the lowest levels, the electron density is evidently

$$\rho_1(z) = \left(\frac{2}{l}\right) \sum_{1}^{N} \sin^2 \frac{n\pi z}{l} \,. \tag{1}$$

One can calculate the sum exactly, and then pass to the semiinfinite electron gas limit $l \rightarrow \infty$, $N \rightarrow \infty$, $N/l \rightarrow \rho_{10}$, the bulk electron gas number of electrons per unit length. Or one can replace the summation over levels by an integration, and writing the highest (or Fermi) energy *E* as $k^2/2$ one finds readily

$$\rho_{1}(z) = \rho_{10}[1 - [\sin(2kz)]/(2kz)]$$
$$= \rho_{10}[1 - j_{0}(2kz)], \quad \rho_{10} = k/\pi, \quad (2)$$

where $j_l(x)$ denotes the *l* th-order spherical Bessel function.

As utilized extensively by March and Murray,¹ one can construct the Slater sum, which weights the square of the wave functions in Eq. (1) with the Boltzmann factor $\exp(-\beta\epsilon_n)$, by using the Laplace transform relation

$$Z(z,\beta) = \beta \int_0^\infty \rho(z,E) \exp(-\beta E) dE.$$
(3)

Inserting Eq. (2), and using the properties of Laplace transforms set out below, one finds

$$Z_1(z,\beta) = (2\pi\beta)^{-1/2} (1 - \exp(-2z^2/\beta)).$$
(4)

Next, let us turn to the three-dimensional case, in which electrons are confined to the semi-infinite space $0 < z < \infty$ by an infinite barrier in the (x, y) plane. This model was studied by Bardeen,² as a description of the electron distribution at a planar metal surface. His result for the electron density, written for comparison with Eq. (2) in terms of singly filled levels, is

$$\rho_3(z) = \rho_{30}(1 - 3j_1(2kz)/2kz), \quad \rho_{30} = k^3/6\pi^2.$$
(5)
Brown *et al.*³ pointed out that, for this problem, the canoni-

cal density matrix could be derived by solving the Bloch equation and on the diagonal they obtained^{3,4}

$$Z_3(z,\beta) = (2\pi\beta)^{-1} Z_1(z,\beta).$$
(6)

In fact, by taking the Bloch equation in d dimensions, it is readily shown that

$$Z_{d}(z,\beta) = (2\pi\beta)^{(1-d)/2} Z_{1}(z,\beta),$$
(7)

confirming that one can write for the d-dimensional case a factor related to the free-particle partition function per unit volume times a factor independent of dimensionality.

Starting from the form (7), one can now invert the Laplace transform relation (3), using the following identities:

$$\mathscr{L}^{-1}(\beta^{-n-1/2}) = \frac{\pi^{-1/2}}{\frac{1}{2}\cdot\frac{1}{2}\cdots(n-\frac{1}{2})} \left(\frac{k^2}{2}\right)^{n-1/2}$$
(8)

and

$$\mathscr{L}^{-1} \frac{\exp(-2z^2/\beta)}{\beta^{(l+3/2)}} = \frac{(2\pi)^{3/2}}{2\pi^2} k^{2l+1} \frac{j_l(2kz)}{(2kz)^l}.$$
 (9)

One needs to find

$$\rho_d(z) = \left[\frac{1}{(2\pi)^{(d-1)/2}} \right] \mathcal{L}^{-1} \left[\frac{1}{\beta}^{(d+1)/2} \right] Z_1(z_*\beta),$$
(10)

and from Eqs. (8) and (9) the result follows readily as

$$\rho_{d}(z) = \frac{k^{d}}{2^{d} \pi^{(d+1)/2} \frac{1}{2 \cdot 2} \cdots d/2} - \left(\frac{2}{\pi}\right)^{1/2} \frac{1}{(2\pi)^{d/2}} k^{d} \frac{j_{(d-1)/2}(2kz)}{(2kz)^{(d-1)/2}}.$$
 (11)

This result (11), valid for d odd, is easily shown to include Eqs. (2) and (5) for the special cases d = 1 and d = 3, respectively.

The case of d even can similarly be handled; one merely notes here the result

$$\mathscr{L}(1/\beta)e^{-2z^2/\beta} = K_0(2kz), \qquad (12)$$

which enables $\rho_2(z)$, etc. to be calculated in terms of the appropriate Bessel functions of a purely imaginary argument.

II. KINETIC ENERGY DENSITY IN d DIMENSIONS

Returning to the case of d odd, one can also calculate the kinetic energy density. Using the Bloch equation, this is easily written for the case of classical statistics as

$$t_{d}^{(c)}(z) = -\frac{\partial Z_{d}}{\partial \beta} = -\frac{1}{(2\pi\beta)^{(d-1)/2}} \frac{\partial Z_{1}}{\partial \beta} -\frac{((1-d)/2)Z_{1}}{(2\pi)^{(d-1)/2}\beta^{(d+1)/2}}.$$
 (13)

Utilizing the expression for $\partial Z_1 / \partial \beta$,

$$\frac{\partial Z_1}{\partial \beta} = -\frac{1}{2(2\pi)^{1/2}\beta^{3/2}} - \frac{2z^2}{(2\pi)^{1/2}} \times \frac{\exp(-2z^2/\beta)}{\beta^{5/2}} + \frac{\exp(-2z^2/\beta)}{2(2\pi)^{1/2}\beta^{3/2}}, \quad (14)$$

one can calculate the degenerate kinetic energy $t_d(z)$ corresponding to the electron density (11) as the inverse Laplace transform of $t_d^{(c)}(z)/\beta$. One then obtains

$$t_{d}(z) = \frac{1}{2(2\pi)^{d/2}} \mathscr{L}^{-1} \left[\frac{d}{\beta^{(d+4)/2}} - \frac{d \exp(-2z^{2}/\beta)}{\beta^{(d+4)/2}} + \frac{4z^{2} \exp(-2z^{2}/\beta)}{\beta^{(d+6)/2}} \right].$$
(15)

Again utilizing Eqs. (8) and (9), one finds, after a short calculation,

$$t_{d}(z) = \frac{d}{2\pi^{1/2}(2\pi)^{d/2} \frac{1}{2} \frac{1}{2} \cdots (d+2)/2} \left(\frac{k^{2}}{2}\right)^{(d+2)/2} + \frac{k^{d+2}}{2\pi^{2}(2\pi)^{(d-3)/2}} \times \left[\frac{j_{(d+1)/2}(2kz)}{(2kz)^{(d+1)/2}} - \frac{2j_{(d-1)/2}(2kz)}{(2kz)^{(d-1)/2}}\right], \quad (16)$$

where use has been made of the relations⁵

$$\frac{d}{dx} \left[x^{-l} j_l(x) \right] = -x^{-l} j_{l+1}(x) \tag{17}$$

and

$$j_{l-1}(x) + j_{l+1}(x) = (2l+1)/x j_l(x).$$
 (18)

Equation (16) represents the d-dimensional kinetic energy corresponding to the electron density (11). As a final step, it

is worth substituting in Eq. (16) for $j_{(d-1)/2}(2kz)$ from Eq. (11), to obtain the desired result

$$t_{d}(z) = t_{d0} + \frac{k^{2}}{2} \left[\rho_{d}(z) - \rho_{d0} \right] + \frac{k^{d+2}}{2\pi^{2} (2\pi)^{(d-3)/2}} \frac{j_{(d+1)/2}(2kz)}{(2kz)^{(d+1)/2}}.$$
 (19)

In Eq. (19), the first term on the right-hand side is the kinetic energy density of the bulk electron gas. In the second term, $\rho_d(z) - \rho_{(d0)}$ is the charge displaced in the electron gas by the infinite barrier, ρ_{d0} as usual being the bulk density. This term is evidently reflecting the kinetic energy due to adding $\int \left[\rho_d(z) - \rho_{d0}\right] d\tau$ electrons at the Fermi energy $k^2/2$. The final term in Eq. (19) is the physically interesting kinetic energy change associated with the induced inhomogeneity of the d-dimensional electron gas. In summary, Eqs. (11) and (17) for the electron density and the corresponding kinetic energy are the main results of this work, both stemming directly from the form (7) of the *d*-dimensional Slater sum of the noninteracting semi-infinite electron gas. As a by-product of Eq. (11), use of the large z asymptotic form of the spherical Bessel function shows immediately the dimensionality dependence of the long-range oscillations induced in the electron gas by the "perturbing" barrier.

ACKNOWLEDGMENTS

This work was completed during sabbatical leave from Oxford at the University of California at Santa Barbara. Thanks are due to Professor D. Hone for making this stay possible and for kind hospitality.

The Leverhulme trustees are thanked for partial financial support.

³J. S. Brown, R. C. Brown, and N. H. March, Phys. Lett. A 46, 463 (1974).

D. Moore and N. H. March, Ann. Phys. (NY) 97, 136 (1976).
 See, for example, L. I. Schiff, *Quantum Mechanics* (McGraw-Hill, New York, 1955), 2nd ed., p. 79.

2976 J. Math. Phys., Vol. 28, No. 12, December 1987

¹N. H. March and A. M. Murray, Phys. Rev. 120, 830 (1960).

²J. Bardeen, Phys. Rev. 49, 653 (1936).

Topologies for point distributions

E. Elizalde

Departament de Física Teòrica, Universitat de Barcelona, Diagonal 647, 08028 Barcelona, Spain

(Received 13 November 1986; accepted for publication 13 August 1987)

The concepts of void and cluster for an arbitrary point distribution in a domain \mathscr{D} are defined and characterized by some parameters such as volume, density, number of points belonging to them, shape, etc. After assigning a weight to each void and cluster—which is a function of its characteristics—the concept of distance between two point configurations S_1 and S_2 in \mathscr{D} is introduced, both with and without the help of a lattice in the domain \mathscr{D} . This defines a topology for the point distributions in \mathscr{D} , which is different for the different characterizations of the voids and clusters.

I. INTRODUCTION

The spatial distributions of points in one, two, or a higher number of dimensions (spatial processes, point processes, spatial patterns, spatial point patterns) constitute a very interesting field of research, not only in pure mathematical statistics but also in its innumerable applications, which range from biometrics to astrophysics and includes such diverse fields as agriculture, econometrics, ecology, traffic problems, and medical sciences. In fact, such point distributions may correspond, for instance, to plants of a given species, to cars along a road, to seeds in a field, to microorganisms in a living body, or to stars—or even galaxies or clusters of galaxies—in our Universe. An important mathematical aspect of spatial processes is the study of the geometrical and topological properties of point distributions.

Although in biometry the study of such distributions has always been very popular during the last years-as can be seen through the large number of research articles and even books which have been issued,¹ in the last couple of months the interest about this field of research has grown very rapidly, due in great part to the remarkable discoveries of de Lapparent, Geller, and Huchra about the spatial distribution of galaxies in our Universe.² These authors made an optical red shift survey of all 1099 galaxies brighter than magnitude 15.5 of a thin slice of sky and came to the conclusion that galaxies are concentrated on the surfaces of contiguous bubble like structures with very large typical diameters of about 25 h^{-1} Mpc. The large void in Boötes of 60 h^{-1} Mpc, discovered in 1981 by Kirshner et al.,³ has been thereby proved to be no peculiarity but a very common feature. Too often, the analysis of the point distributions of galaxies, with their voids and clusters, is done simply by looking at pictures and plates with the naked eye, a very primitive procedure which in general is not that bad. Nevertheless, the important discoveries we have just mentioned stress once more the necessity for a more profound understanding of spatial point distributions and, in particular, of the still unsolved problem concerning the construction of a mathematical "measure" for quantifying how far away are two of such point distributions (characterized by the number and magnitude of the voids and clusters, their forms and spatial distribution, etc.).

II. DEFINITIONS OF (SPHERICAL) VOID AND CLUSTER

The very large numbers of points one has to deal with makes it almost a necessity to introduce definitions which are suitable to be treated with a numerical algorithm. This has been pointed out in several previous papers on the subject⁴ and will be considered later in detail (Sec. VI). However, I do not think that discrete algorithms alone can solve these problems satisfactorily, and it is much better to play at a time both with discrete and with continuous concepts.

Let S be a set of points in a given domain \mathcal{D} of volume V in d-dimensional Euclidean space. Let N be the number of points in S. For any point $p \in \mathcal{D}$ and any positive real number $r \in \mathbb{R}^+$, the density of points in a ball around p of radius r is given by

$$\rho_{p}(r) = n_{p}(r) / V_{p}(r), \qquad (2.1)$$

where $n_p(r)$ is the number of points of S inside the ball, and $V_p(r)$ is the volume of the ball. By definition, there is a void around p of radius bigger than r if the density $\rho_p(r)$ verifies

$$\rho_{p}(r) < \lambda N / V, \qquad (2.2)$$

where $\lambda \leq 1$ must be fixed (we may take, for instance, $\lambda = \frac{1}{2}$). The radius of the void around p is defined to be the value r_p such that

$$\rho_{p}(r_{p}) = \lambda N / V. \tag{2.3}$$

In this way the density of any void will be the same. Alternatively, one could define the radius of the void as the value of rat which the slope of $\rho_p(r)$ is maximum.

On the other hand, there exists, by definition, a cluster around p if

$$\rho_p(r) > N/\lambda V. \tag{2.4}$$

The radius of the cluster may be defined to be the value r_p such that

$$o_p(r_p) = N/\lambda V. \tag{2.5}$$

As before, one could alternatively define the radius of the cluster as the value of r at which the slope of $-\rho_p(r)$ is maximum.

Until now we have studied only what happens at some given place p. A global analysis has to distinguish between the different voids and clusters, so that we do not count the

same point of the distribution twice: as belonging to a void (or cluster) and to another one which intersects the first. Moreover, the preceding definitions are best suited for spherical voids or clusters only and when p is the center of them. These difficulties will be taken care of in the subsequent sections.

III. EFFICIENT SEARCH FOR VOIDS AND CLUSTERS

Let us now introduce a lattice \mathscr{L} of lattice site *a* in the domain \mathscr{D} . For a given set of $r_k \in \mathbb{R}^+$, k = 1, 2, ..., m, select the set S_1 of vertices of the lattice corresponding to the *s* smallest and to the *h* highest values of $\rho_{p_i}(r_k)$, for all k = 1, 2, ..., m, for all lattice vertices p_i on \mathscr{L} . Improve now the set of points p_i in S_1 to a set S_2 coming from the *s* smallest and *h* highest values of $\rho_{q_i}(r_k)$ for all k = 1, 2, ..., m, and for all q_i of the form

$$p_i + \lambda \left[\prod_{k=1}^m \rho_{p_i}(r_k)\right]^{-1/2m} \hat{e}, \qquad (3.1)$$

where λ is a constant that we can adjust at will (for instance, $\lambda = \frac{1}{2}$ for voids and $\lambda = 2$ for clusters), while \hat{e} sweeps all unitary directions of the form

$$\hat{e}_{i_1} \pm \cdots \pm \hat{e}_{i_l}, \quad 1 \leq i_1 < \cdots < i_l \leq d, \tag{3.2}$$

where \hat{e}_i is the unitary vector along the *i* axis of \mathbb{R}^d . Notice that, in general, the points q_i are not vertices of the lattice. In fact, the vertices of the lattice serve only as starting points in order to begin the search for the best centers of the voids and clusters.

The procedure is then repeated until it stabilizes. In this way we obtain the positions of the centers of a desired number of the less dense voids and of the more dense clusters in the point distribution S.

IV. WEIGHTS OF THE INDIVIDUAL VOIDS AND CLUSTERS

The weight of a spherical void of radius r and density ρ with the center at the point p is given by the following expression:

$$W_{v} = k_{v} \left[V_{p}(r) / \rho \right] = k_{v} \left[V_{p}(r)^{2} / n \right], \qquad (4.1)$$

where k_v is a constant (independent of the void), $V_p(r)$ the volume of a sphere of radius r in d dimensions, and n is the number of points of S inside the sphere. That this expression is correct can be seen through the following argument. For a given density ρ , increase of W_v in (4.1) is proportional to the volume of the void, while for fixed volume, increase of W_v is proportional to decrease of ρ_i , as it should be by intuition. Alternatively, at fixed n increase of W_v is proportional to the volume and *also* to the decrease of density, i.e., proportional to the volume squared.

The weight of a spherical cluster of radius r and density ρ centered at p is given by

$$W_{c} = k_{c} n\rho = k_{c} \left[n^{2} / V_{p}(r) \right] = k_{c} \rho^{2} V_{p}(r), \qquad (4.2)$$

where k_c is a constant independent of the cluster. Expression (4.2) can be understood by reasoning as follows. At fixed n, W_c is proportional to increase of ρ (or to decrease of volume). At fixed ρ , W_c is proportional to increase of n (or to increase of volume). Alternatively, with full generality, in-

crease of W_c is due both to increase of n (it would also be true for fixed ρ , i.e., letting volume increase) and to increase of ρ .

V. THE SET OF ALL POINT DISTRIBUTIONS AS A METRIC SPACE

Once the spherical voids and clusters have been constructed by the procedures described in Secs. II and III, and just before their individual weights (Sec. IV) are calculated, one has to look for superpositions of them which may result in nonspherical voids and clusters. The idea is very simple: to consider as a unique void (resp. cluster) the union of all of them which are connected by a chain of intersections (Fig. 1). In this way, the numbers of voids and clusters, s and h, respectively, diminish and, at the same time, they are no more spherical but acquire a form of the type depicted in Fig. 2. It is now immediate to modify the formulas (4.1) and (4.2) accordingly: in both cases the volume $V_p(r)$ of a ddimensional sphere of radius r centered at p must be substituted by the volume of the void or cluster considered. Of course, the density ρ and the number of points *n* will also correspond now to the whole, nonspherical void or cluster.

Once all the voids and all the clusters have been constructed, the remaining region of the domain \mathcal{D} is filled up with a (under ideal conditions) sensibly uniform distribution of points of S with a density almost equal to $\rho_0 = N/V$. In practice this must be checked *a posteriori* and if it were not true, the free parameters introduced in the definitions and construction of the voids and clusters (s, h,...) ought to be changed accordingly. For instance, if the density of the remaining region were smaller than ρ_0 , then the number s of voids should be increased. On the other hand, if the homogeneity of the remaining region were not very good then both s and h ought to be augmented.

Let us now consider the plane (V,ρ) and the points (V_i,ρ_i) in it, where the index *i* goes through all the different voids and clusters, with one value of the index corresponding to the intermediate, remaining region. Introduce a regular lattice in this plane and denote the different cells by $(V_j,\rho_j), j \in \mathcal{J}$. Define now the function $f(V_j,\rho_j)$ which as-



FIG. 1. Examples for distributions of spherical voids and clusters. They can superpose in a variety of ways.



FIG. 2. Voids and clusters constructed from the distributions of Fig. 1 in order to avoid overcounting.

signs to each cell (V_j,ρ_j) the number of clusters plus the number of voids which belong to this cell, each of them multiplied by the corresponding weight (4.2) and (4.1), respectively, i.e.,

$$f(V_{j},\rho_{j}) = n_{c}(V_{j},\rho_{j})W_{c}(V_{j},\rho_{j})$$

+ $n_{v}(V_{j},\rho_{j})W_{v}(V_{j},\rho_{j}), \quad j \in \mathcal{J}.$ (5.1)

Here the intermediate region is to be counted as an additional void or cluster depending on its density ρ being $\rho \leq \rho_0$ or $\rho > \rho_0$, respectively. Notice that $f(V_j, \rho_j)$ is different from zero only in a finite number of cells (V_j, ρ_j) . The following step is to construct with these points the minimal triangulated surface with vertices at these points. Thus, we get a continuous function $f(V,\rho)$ defined on the plane (V,ρ) . Now, given another point configuration on the domain \mathcal{D} , we define the distance between these two configurations S_1 and S_2 by

$$d(S_1,S_2)^2 = \iint [f_1(V,\rho) - f_2(V,\rho)]^2 \, dV \, d\rho, \quad (5.2)$$

where f_1 and f_2 are the functions corresponding to the point configurations S_1 and S_2 , respectively.

The problem we are dealing with is not so standard. No wonder, therefore, that definition (5.2) is not a *usual* measure of the configuration space. However, it is important to observe that d, as given by (5.2), can be easily implemented to yield a true distance by the usual mathematical procedures. Let us be completely rigorous.

The set which is going to turn into a metric space is $\mathscr{S} = \text{set of all finite point distributions in the domain } \mathscr{D}$. The "distance" defined by (5.2) is actually only a semidistance. In fact, it satisfies (i) $d(S_1,S_2) = 0$, (ii) $d(S_2,S_1) = d(S_1,S_2)$, and (iii) $d(S_1,S_2) \leqslant d(S_1,S_3) + d(S_3,S_2)$, for any $S_1,S_2,S_3 \in \mathscr{S}$. All we have to do is to define the coset $\overline{\mathscr{S}} = \mathscr{S}/\sim$, where $S_1 \sim S_2$ iff $d(S_1,S_2) = 0$, in order to obtain a metric space $\widetilde{\mathscr{S}}$ with the distance d given by

$$\vec{d}(\vec{S}_1, \vec{S}_2) = d(S_1, S_2), S_1 \in \vec{S}_1, S_2 \in \vec{S}_2.$$
(5.3)

In fact, this is a consistent definition for, let us consider two

other configurations $S'_1 \in \overline{S}_1$ and $S'_2 \in \overline{S}_2$. Then, we have

$$d(S_1,S_2) \leq d(S_1,S_1') + d(S_1',S_2') + d(S_2',S_2).$$

But $d(S_1, S'_1) = 0$ and $d(S_2, S'_2) = 0$, so that $d(S_1, S_2) \le d(S'_1, S'_2)$. Moreover,

$$d(S'_1,S'_2) \leq d(S'_1,S_1) + d(S_1,S_2) + d(S_2,S'_2),$$

and we get $d(S'_1, S'_2) \leq d(S_1, S_2)$. Therefore, $d(S'_1, S'_2) = d(S_1, S_2)$. Equation (5.3) defines a distance \overline{d} on $\overline{\mathscr{S}}$. In fact, \overline{d} satisfies the axioms (i)-(iii) above and, moveover, the additional one, (i') $\overline{d}(\overline{S}_1, \overline{S}_2) = 0$ implies $\overline{S}_1 = \overline{S}_2$. This is immediate from (5.3) and from the definition of the coset $\overline{\mathscr{S}}$.

Summing up, $\overline{\mathscr{S}}$ is a metric space endowed with the distance \overline{d} . This constructive procedure is very well known to mathematicians, in fact, it is the standard way to proceed. This allows one to be a little loosely in the notation and speak of the metric space \mathscr{S} and of the distance d, as given by (5.2). The alternative definitions of distance which will follow have to be compared with (5.2). Actually all of them ought to be submitted to the same procedure as given above in order that they become true distances \overline{d} .

A metric space is readily made into a topological space, the topology being provided by the distance, much as in the standard example of the metric space \mathbb{R}^n . The neighborhoods of the basis of this topology are open balls of the form $B_p(\bar{S}) = \{\bar{S}' \in \mathcal{F} | \bar{d}(\bar{S}, \bar{S}') < p\}, p$ being any rational number $p \in \mathbb{Q}$. Being again a little loosely with the notation we may say that the set \mathscr{S} of all finite point distributions in \mathscr{D} is a topological space, the topology being given through the distance d in (5.2).

This is by no means the only possibility to define a distance between two point configurations. But the definition which has just been given above is quite a sensible one. An example of a different, more simple definition is the following. Consider the weights (4.1) and (4.2) and place them at the negative and positive semiaxis x, respectively (Fig. 3). Then discretize this axis by considering intervals of a given length *l*. For each interval of the x axis, on the y axis set the number of voids (resp. clusters) with a value of W_v (resp. W_c) which belongs to this interval. Now consider the segment-wise curve constructed with the resulting points (Fig. 3). Let us call this curve g(x). The distance between two point configurations S_1 and S_2 can then by defined by

$$d(S_1,S_2)^2 = \int \left[g_1(x) - g_2(x)\right]^2 dx.$$
 (5.4)

Notice, however, that on taking the weights from the beginning we have implicitly introduced in this last case an equivalence relation among voids (and among clusters). In some cases this can actually be convenient in order to simplify the problem from the beginning, but in other situations a finer definition such as the first one will have to be adopted.

VI. POINT DISTRIBUTIONS IN A DOMAIN WITH A LATTICE

In order to treat all the preceding questions in a way better suited for numerical manipulations, one can carry all these definitions to a lattice \mathcal{L} of certain site *a* on the do-



FIG. 3. Segment-wise curve g(x) constructed using a discretization of the weights corresponding to voids (negative x axis) and clusters (positive x axis). In the y axis the numbers of voids and clusters that fall into each interval of x are represented.

main \mathcal{D} . That is, a discretization of the methods which have been elaborated above has to be developed.

One must start by counting the number of points of the distribution S in \mathcal{D} that fall into each of the elementary cells of the lattice. The voids will consist of all those cells whose number of points does not reach a value n_1 fixed in advance, while the clusters are made up of cells with a number of points above a second number n_2 also fixed in advance. Of course, one must have

$$n_1 < N/N_c < n_2,$$
 (6.1)

where $N_c = V/a^d$ is the number of cells of the lattice \mathcal{L} . (To begin with, we consider all cells equal and obviate the small modifications in these definitions which had to be made for cells touching the border of the domain \mathcal{D} .) In this way, extended voids and clusters made up of cells will arise, in general. A huge void (cluster) will consist of several contiguous cells with a small (big) number of points. Figure 2 will be almost the same, only that the curved contour will be substituted by a segment-wise one, with segments of longitude proportional to a. Formulas (4.1) and (4.2) will remain unchanged: only $V_p(r)$ will be substituted by the volume V_v or V_c of the void or cluster under consideration (a volume always proportional to a^d , the volume of an elementary cell).

Notice that this procedure is less time consuming than the former one when it is carried out in practice. However, it is not so sensible to detect the voids and clusters with precision. In fact, once the lattice \mathscr{L} has been fixed, a given cell can participate at the same time of a void and of a cluster so that the total number of points in it may compensate (Fig. 4), thus hiding this fact completely. Clearly, everything becomes better as *a* is made smaller (continuum limit). However, with a (discrete) point distribution this cannot be done indefinitely: for *a* small enough every cell contains at most one point only and for such small cells the whole procedure ceases to be of much use (this was the difficulty with the topology of discrete point distributions in the first place).

Once the weights (4.1) and (4.2) have been adapted to the lattice voids and clusters, the definitions (5.2)-(5.4) for the distance between two point configurations S_1 and S_2 go through immediately. Thus, we complete the treatment of the lattice case and define a topology for point configurations on the domain \mathscr{D} . One could think, in principle, that in order to proceed in accordance with the discretization of \mathscr{D} , the plane $(V_s \rho)$ ought also to be discretized, i.e., divided into rectangles of sides a_V and a_ρ , and the minimal triangulated surfaces constructed using the vertices corresponding to the centers of these cells. However, it must be pointed out that this last discretization is completely independent from the one of the domain \mathscr{D} .

Finally, notice that in our definition only the volume and the density (we may substitute one of these by the number of points inside) of the void or cluster have been taken into account in the definition of the distance $d(S_1,S_2)$. A more elaborate definition should also include other parameters such as some characterizing the shape of the void or cluster (for instance, a combination of the diameters along each of the axes, as the sum or the product of these diameters). The function $f(V,\rho)$ given in (5.1) and the distance (5.2) have to be redefined accordingly. That is (we drop the subindex *j* for convenience)



FIG. 4. Some cells (here the one in the middle) of a lattice in \mathscr{D} may participate both from some void and from some cluster. They may compensate and give a deceptive mean density approximately equal to $\rho_0 = N/V$.



FIG. 5. A huge void or cluster (the points have not been depicted) displaying the gradient of density: ρ_1 is the density of the most inner cell; ρ_2 that of the surrounding crown of cells (here 12 cells); ρ_3 is the density of the most exterior crown (here 24 cells).

$$f(V,\rho,k) = n_c(V,\rho,k) W_c(V,\rho,k) - n_v(V,\rho,k) W_v(V,\rho,k),$$
(6.2)

where k is the new parameter, and

$$d(S_1,S_2)^2 = \iiint [f_1(V,\rho,k) - f_2(V,\rho,k)]^2 \, dV \, d\rho \, dk,$$
(6.3)

respectively. In an analogous way, we may introduce other parameters, such as the gradient of density for large voids or clusters, as one proceeds from inside to the border (Fig. 5). We may define, for instance,

$$h = [(\rho_2 - \rho_1)^2 + (\rho_3 - \rho_2)^2 + \cdots]^{1/2}$$
(6.4)

and include h besides k as a new parameter. All these param-

eters and others one may think of improve the definition of the topology (6.3) and may be introduced at ease into our formalism in the way we have just shown.

VII. OUTLOOK

The procedures introduced here for the first time (to our knowledge) are currently being applied to the point distributions that correspond to the analysis of galaxies of de Lapparent *et al.*² and also to other related results. Moreover, simulation methods are being developed with the purpose of checking the reliability of the distance between point distributions as defined here compared with the only one which is presently available, namely, the "distance" that our naked eye would *grosso modo* assign to them. The investigation is in progress. Its partial results are pretty good and will be published elsewhere with a detailed account of the analysis involved.

ACKNOWLEDGMENTS

Useful conversations with E. Gaztañaga are gratefully acknowledged.

This paper has been partially supported by the CAICYT.

¹M. S. Bartlett, *The Statistical Analysis of Spatial Pattern* (Chapman and Hall, London, 1975); A. D. Cliff and J. K. Ord, *Spatial Processes: Models and Applications* (Pion, London, 1981); D. R. Cox and V. Isham, *Point Processes* (Chapman and Hall, London, 1980); P. J. Diggle, *Statistical Analysis of Spatial Point Patterns* (Academic, London, 1983); B. D. Ripley, *Spatial Statistics* (Wiley, New York, 1981); A. Rogers, *Statistical Analysis of Spatial Dispersion* (Pion, London, 1974).

²V. de Lapparent, M. J. Geller, and J. P. Huchra, Astrophys. J. Lett. **302**, L1 (1986).

³R. P. Kirshner, A. Oemler, Jr., P. L. Schechter, and S. A. Schectman, Astrophys. J. Lett. 248, L57 (1981).

⁴F. R. Bouchet and M. Lachièze-Rey, Astrophys. J. Lett. **302**, L37 (1986);
J. E. Moody, E. L. Turner, and J. R. Gott, III, Astrophys. J. **273**, 16 (1983);
S. Otto, H. D. Politzer, and M. B. Wise, Phys. Rev. Lett. **56**, 1878 (1986);
H. D. Politzer and J. P. Preskill, *ibid*. **56**, 99 (1986);
S. Otto, H. D. Politzer, Astrophys. J. **304**, 62 (1986).

Comment on "On a property of a classical solution of the nonlinear mass transport equation $u_t = u_{xx}/1 + u_x^2$. II" [J. Math. Phys. 28, 536 (1987)]

Akihiko Kitada

Liberal Arts, Nippon College of Physical Education, Fukazawa, Setagaya-ku, Tokyo 158, Japan

(Received 17 March 1987; accepted for publication 5 August 1987)

A mechanism of smoothing due to evaporation condensation of the roughly perturbed surface of a solid is formulated as a Cauchy problem in the real line \mathbb{R}^1 for the equation $u_t = u_{xx}/1$ $+ u_x^2$, which describes the evolution of the profile u(x,t) of the surface. In the preceding paper [A. Kitada and H. Umehara, J. Math. Phys. 28, 536 (1987)], it was demonstrated that, if the solution u(x,t) of the Cauchy problem is obtained in the classical sense with an additional restriction $u \in C^3(\mathbb{R}^1 \times (0, \infty))$, each peak in the surface decreases in height with time in the strict sense. In the present paper, by modifying the proof, it is shown that the additional restriction to the classical solution is excessive.

I. INTRODUCTION

In the preceding paper,¹ using the maximum principle due to Nierenberg,² well known in the theory of linear parabolic equations, we obtained the estimate (1),

$$u(x_2,t_2) < u(x_1,t_1), \quad (x_i,t_i) \in C \quad (i=1,2), \quad t_1 < t_2, \quad (1)$$

describing the strictly monotone decrease with time t, due to evaporation condensation, in the height of a peak in a roughly perturbed surface of solid. In the estimate (1), u(x,t) is such a classical solution³ that $u \in C^3(\mathbb{R}^1 \times (0, \infty))$ (Ref. 4) of the Mullins' model (P) (Ref. 5) in the real line \mathbb{R}^1 ,

$$u_t = u_{xx}/1 + u_x^2, \quad (x,t) \in \mathbb{R}^1 \times (0,\infty),$$

$$u(x,0) = \alpha(x), \quad x \in \mathbb{R}^1,$$
(P)

and C is a trajectory in the (x,t) plane drawn by the migration with time of a peak top in $\alpha(x)$; that is, the set C is characterized by

$$C = \{(x,t); x = g(t), g \in C^{1}([0,t_{f}])\},$$
(2a)

$$u_x(x,t) = 0, \quad u_{xx}(x,t) < 0, \quad (x,t) \in C.$$
 (2b)

In the present paper, we show that the estimate (1) holds for the ordinary classical solution of the problem (P) without the additional restriction $u \in C^3(\mathbb{R}^1 \times (0, \infty))$; that is, the restriction $u \in C^3(\mathbb{R}^1 \times (0, \infty))$ is shown to be excessive.

II. A DEMONSTRATION OF THE STRICTLY MONOTONE DECREASE IN HEIGHT OF A PEAK IN THE CLASSICAL SOLUTION

We will make general discussions on the following Cauchy problem (P*) which is a generalization of the Mullins' model (P):

$$u_{t} = F(u_{x}, u_{xx}),$$

$$[F \in C^{1}(\mathbb{R}^{2}), F_{q}(p,q) > 0 \text{ (Ref. 6)}, F(0,0) = 0],$$

$$(x,t) \in \mathbb{R}^{1} \times (0, \infty),$$

$$u(x,0) = \alpha(x), \quad x \in \mathbb{R}^{1}.$$
(P*)

As is pointed out in our previous paper,⁷ the solution of (P^*) must satisfy the following problem (LP):

$$u_{t} - u_{xx} \int_{0}^{1} F_{q}(hu_{x}(x,t),hu_{xx}(x,t))dh - u_{x} \int_{0}^{1} F_{p}(hu_{x}(x,t),hu_{xx}(x,t))dh = 0,$$

(x,t) $\in \mathbb{R}^{1} \times (0, \infty)$, (LP)
 $u(x,0) = \alpha(x), \quad x \in \mathbb{R}^{1},$

where the function F(p,q) is what is given in the problem (P^*) . In the preceding paper,¹ we showed that if the maximum principle due to Nierenberg is applicable to the problem (LP), the estimate (1) holds, and in order for the maximum principle to be applicable to (LP), it is sufficient that all the coefficients of the linear equation in (LP) are continuous at any point of the set $\mathbb{R}^1 \times (0, \infty)$. In the preceding paper,¹ we used the mean value theorem in the differential calculus to show the continuity of the coefficients

$$\int_0^1 F_q(hu_x(x,t),hu_{xx}(x,t))dh$$

and

$$\int_0^1 F_p(hu_x(x,t),hu_{xx}(x,t))dh.$$

In this context, the additional restriction $u \in C^3(\mathbb{R}^1 \times (0, \infty))$ was required.⁸ In the present paper, we replace the procedure of the demonstration by the following one which does not require this restriction.

Proof of the continuity of the coefficients: We will show the continuity of the coefficient of u_{xx} ,

$$\int_0^1 F_q(hu_x(x,t),hu_{xx}(x,t))dh$$

at the arbitrarily fixed point $(x_0,t_0) \in \mathbb{R}^1 \times (0,\infty)$. Let Ω be a bounded open convex set in the real plane \mathbb{R}^2 , for example, an open disk, which contains two points (p_0,q_0) and (0,0). Here, $\underline{p}_0 = u_x(x_0,t_0)$ and $q_0 = u_{xx}(x_0,t_0)$. Since the closure of Ω , $\overline{\Omega}$, is compact and the function $F_q(p,q)$ is continuous as is indicated in (\mathbb{P}^*) , $F_q(p,q)$ is uniformly continuous on $\overline{\Omega}$. Therefore, for any $\epsilon > 0$, there exists $\delta > 0$ such that

$$F_{q}(hp,hq) - F_{q}(hp',hq')|$$

< $\epsilon[0 \le h \le 1, (p,q) \in \overline{\Omega}, (p',q') \in \overline{\Omega}],$

ļ

whenever $|p - p'| < \delta$ and $|q - q'| < \delta$. As both functions $u_x(x,t)$ and $u_{xx}(x,t)$ are continuous at the point (x_0,t_0) , for the above $\delta > 0$, there exists $\eta > 0$ such that

$$|u_x(x,t) - u_x(x_0,t_0)| < \delta, \quad |u_{xx}(x,t) - u_{xx}(x_0,t_0)| < \delta,$$

whenever $|x - x_0| < \eta$ and $|t - t_0| < \eta$. Combining the above two estimates, we have

$$\begin{aligned} \left| \int_{0}^{1} F_{q}(hu_{x}(x,t),hu_{xx}(x,t)) dh \right| \\ &- \int_{0}^{1} F_{q}(hu_{x}(x_{0},t_{0}),hu_{xx}(x_{0},t_{0})) dh \right| \\ &\leq \int_{0}^{1} |F_{q}(hu_{x}(x,t),hu_{xx}(x,t)) \\ &- F_{q}(hu_{x}(x_{0},t_{0}),hu_{xx}(x_{0},t_{0}))| dh < \epsilon, \end{aligned}$$

for any (x,t) such that $|x - x_0| < \eta$ and $|t - t_0| < \eta$. This means the continuity of the coefficient of u_{xx} at the point (x_0,t_0) . As the same is true for the coefficient of u_x , all the coefficients of the linear equation in (LP) are continuous. \Box

We can conclude that the estimate (1) holds for the classical solution without the restriction $u \in C^3(\mathbb{R}^1 \times (0, \infty))$.

ACKNOWLEDGMENT

I am grateful to Dr. M. Taniguchi for helpful discussions.

- ¹A. Kitada and H. Umehara, J. Math. Phys. 28, 536 (1987).
- ²L. Nirenberg, Commun. Pure. Appl. Math. 6, 167 (1956).
- ³O. A. Ladyzenskaja, V. A. Solonikov, and N. N. Ural'ceva, *Linear and Quasilinear Equations of Parabolic Type* (Am. Math. Soc., Providence, RI, 1968), p. 12.
- ⁴The symbol $C^{m}(\Omega)$ denotes the set of all functions defined on Ω whose partial derivatives of order $\leq m$ are all continuous.
- ⁵W. W. Mullins, J. Appl. Phys. 28, 333 (1957); 30, 77 (1959).
- ⁶The partial derivative $\partial F/\partial q$ is abbreviated as F_q . In the same manner, we write $\partial F/\partial p$ as F_q .
- ⁷A. Kitada, J. Math. Phys. 27, 1391 (1986).
- ⁸Moreover, in the preceding paper,¹ the extent of the generalization (P^{*}) of the Mullins' model was also limited by the excess restriction $F \in C^2(\mathbb{R}^2)$. In the present paper, this restriction is so softened as $F \in C^1(\mathbb{R}^2)$.

Relativistic Coulomb Sturmian matrix elements and the Coulomb Green's function of the second-order Dirac equation

Levere Hostler Physics Department, Wilkes College, Wilkes Barre, Pennsylvania 18766

(Received 20 January 1987; accepted for publication 13 August 1987)

Matrix elements with respect to a discrete "Sturmian" basis set of the operator $r^{1/2}gr^{1/2}$, where g is the Coulomb Green's function of the "second-order" Dirac equation, are investigated. Closed expressions, involving Gaussian hypergeometric functions, can be obtained for the matrix elements. This calculation illustrates contour integration techniques capable of yielding closed expressions for a number of overlap integrals associated with the relativistic Kepler problem.

I. INTRODUCTION

Discrete Coulomb Sturmian basis sets have been used successfully for some time in nonrelativistic atomic calculations.¹ One of the first examples of a Sturmian basis set appeared in a discrete expansion of the nonrelativistic Coulomb Green's function in momentum space due to Schwinger.² Sturmian functions in this discrete expansion entered as a set of four-dimensional spherical harmnics connected with the O(4) symmetry of the nonrelativistic Kepler problem. A coordinate space form of Schwinger's expansion was investigated in which the Sturmian functions arose in a different way as "coupling constant eigenfunctions."³ Subsequently, it was found that the coupling constant eigenfunction concept could yield a discrete expansion also for the relativistic Coulomb Green's functions. The Coulomb Green's functions for both the "second-order" Dirac equation⁴ (defined below) and the conventional linear Dirac equation⁵ have been studied from this point of view.

The Sturmian basis sets seem convenient for atomic calculations, since through the use of the Sturmian basis sets one can avoid the introduction of continuum states, and this without sacrificing completeness. In perturbation calculations using a Coulomb Sturmian basis set there will in general be a need for Sturmian matrix elements of the Coulomb Green's function. Such matrix elements will be investigated here for the second-order Dirac equation,

$$\{\Pi \cdot (1 + i\sigma) \cdot \Pi + m^2\} \Phi = 0,$$

$$\Pi_{\mu} \equiv -i \partial_{\mu} - eA_{\mu},$$

(1.1)

in which the wave function Φ is a 2×1 Pauli spinor, and $\sigma_{\mu\nu}$ is a self-dual Lorentz spin tensor, defined by the equation

$$\sigma_{\mu\nu} \equiv \begin{bmatrix} 0 & \sigma_3 & -\sigma_2 & \sigma_1 \\ -\sigma_3 & 0 & \sigma_1 & \sigma_2 \\ \sigma_2 & -\sigma_1 & 0 & \sigma_3 \\ \hline -\sigma_1 & -\sigma_2 & -\sigma_3 & 0 \end{bmatrix}, \quad (1.2)$$

 $\sigma_{1,2,3}$ being the ordinary 2×2 Pauli spin matrices. Equation (1.1) has been discussed by a number of authors^{4,6-18} and is known to be equivalent to the usual linear Dirac equation. The second-order Dirac equation (1.1) brings out a close parallel between the quantum theory of a Dirac particle and the quantum theory of a simple scalar particle. For this reason, and because of the small dimension of the matrices in-

volved, calculations in quantum electrodynamics are expected to simplify when the second-order form of the Dirac equation is used. These expectations have been borne out in a recent study of the mass operator for an electron in an external Coulomb potential.¹⁹

To carry the study of the mass operator referred to above further and turn it into a new Lamb shift calculation, matrix elements with respect to a Sturmian basis set of the Coulomb Green's function of the second-order Dirac equation are required. Since these matrix elements will be useful, in general, for a variety of perturbation calculations, it was decided to publish the investigation of the matrix elements of the Green's function separately. The results presented here generalize known results for the nonrelativistic Coulomb Green's function.²⁰ Also, this calculation illustrates contour integration techniques capable of yielding closed expressions for a number of overlap integrals associated with the relativistic Kepler problem.

In Sec. II the relativistic Coulomb Sturmian basis sets are discussed and the derivation of the overlap integral between two different such basis sets is indicated. Although this material is not especially new, the derivation of the overlap integral is considered because some intermediate steps in the derivation provide a point of departure for the later work with the Coulomb Green's function. The Coulomb Green's function material is presented in Sec. III. Armed with the contour integration technique used for the overlap integral, it is shown how to obtain a closed analytic expression, involving Gaussian hypergeometric functions, for the operator $r^{1/2}gr^{1/2}$, where g is the Coulomb Green's function of the second-order Dirac equation. The factors $r^{1/2}$ supplied on each side of the Green's function before taking matrix elements are essential to obtain this relatively simple result. Because of these factors, the most difficult integrals encountered in Sec. III are of the Gaussian hypergeometric type.

Sturmian basis sets and the Coulomb Green's function of the second-order Dirac equation are discussed in detail in the literature cited above. In the interest of brevity the reader is referred to this earlier work, especially Ref. 4, for background material.

II. COULOMB STURMIAN BASIS SETS

The reader will find a more heuristic introduction to Sturmian basis sets in the literature cited above. Here a purely mathematical approach seems best. Relativistic Coulomb Sturmian functions will be defined as the eigenfunctions of the self-adjoint operator

$$O \equiv \left[\frac{r}{2}\right]^{1/2} \left[-\frac{1}{r^2}\frac{\partial}{\partial r}r^2\frac{\partial}{\partial r} + \eta^2 - \frac{1 - (2\gamma + 1)^2}{4r^2}\right] \left[\frac{r}{2}\right]^{1/2}$$
(2.1)

acting on the space of spinor functions $\Phi(\mathbf{r})$. Here $\eta > 0$ is an arbitrary real constant, and γ is a self-adjoint operator acting on only the angular and spin degrees of freedom of $\Phi(\mathbf{r})$ and having the eigenvalues $\gamma = ((J + \frac{1}{2})^2 - (Z\alpha)^2)^{1/2}$ if $L = J + \frac{1}{2}$, and $\gamma = ((J + \frac{1}{2})^2 - (Z\alpha)^2)^{1/2} - 1$ if $L = J - \frac{1}{2}$. The corresponding eigenfunctions $\langle \mathbf{r} | \eta, n, L, J, M \rangle$ form a discrete basis. These eigenfunctions are proportional to the spinor spherical harmonics $Y_{LJM}(\mathbf{r})$:

$$\langle \mathbf{r} | \eta, n, L, J, M \rangle = \langle r | | \eta, n, L, J \rangle Y_{LJM}(\mathbf{r}),$$
 (2.2)

$$\langle r | |\eta, n, L, J \rangle = (2\eta)^{3/2} [(n-1)!/(n+2\gamma)!]^{1/2} \\ \times (2\eta r)^{\gamma - (1/2)} e^{-\eta r} L_{n-1}^{2\gamma + 1}(2\eta r) .$$

$$(2.3)$$

The corresponding eigenvalues of O are $O' = \eta(\gamma + n)$. The double bar notation $\langle r | |\eta, nL, J \rangle$ is used to signify the radial part of the wave function of the particle. The radial function $\langle r | |\eta, n, L, J \rangle$ is without spinor structure. The angular momentum quantum numbers L, J, M are the familiar ones, and n runs over the range 1,2,3,..., independently of the other quantum numbers. The Sturmian functions (2.2) obey the orthogonality and completeness relations $\langle \eta, A | \eta, B \rangle = \delta_{A,B}$, $1 = \sum_{A} |\eta, A \rangle \langle \eta, A |$. Here and subsequently when convenient the shorthand notation A, B, etc. is used to signify the whole set of quantum numbers n, L, J, M.

In applications the Coulomb Sturmian functions (2.2) are needed sometimes with one, sometimes with another value of η . Accordingly, the overlap integral¹⁹

$$\langle \eta_{0}A \mid |\eta,B \rangle = \langle \eta,B \mid |\eta_{0}A \rangle \equiv \int_{0}^{\infty} r^{2} dr \langle \eta_{0}A \mid |r\rangle \langle r \mid |\eta,B \rangle$$

$$= (-1)^{n-1} \left[\frac{(n_{>} + 2\gamma)!}{(n_{>} - 1)!} \frac{(n_{<} + 2\gamma)!}{(n_{<} - 1)!} \right]^{1/2} \frac{(2\zeta - \zeta^{2})^{\gamma + 1}}{(2\gamma + 1)!} (1 - \zeta)^{n_{>} - n_{<}}$$

$$\times_{2} F_{1}(-(n_{<} - 1), 2\gamma + 1 + n_{>}, 2\gamma + 2; 2\zeta - \zeta^{2}),$$

$$(2.4)$$

is sometimes encountered. In Eq. (2.4) $n_{>} \equiv \min(n_{A} = n_{0}, n_{B} \equiv n), n_{>} \equiv \max(n_{A}, n_{B}), \text{ and } \zeta$ is the parameter

$$\zeta \equiv 2\eta_0 / (\eta + \eta_0). \tag{2.5}$$

The contour integration technique needed in Sec. III will be illustrated here in a simpler context by indicating the derivation of the overlap integral (2.4). Also, the first steps in this derivation provide the starting point for the more involved calculation of Sec. III. As noted in the Introduction, the final result, Eq. (2.4), of this section is not new.

Proceeding with the derivation of the overlap integral (2.4), the Laguerre polynomials under the integral sign are replaced using the integral representation²¹

$$L_{n}^{\mu}(z) = \frac{1}{2\pi i} \oint_{(0+)} dt \, e^{-tz} \, \frac{(1+t)^{n+\mu}}{t^{n+1}} \,. \tag{2.6}$$

The r integral then goes over into a special case of the formula

$$\int_0^\infty dr \, r^{\nu-1} e^{-ar} = \frac{\Gamma(\nu)}{a^\nu},$$

$$\operatorname{Re}(\nu) > 0, \quad \operatorname{Re}(a) > 0,$$

$$|\operatorname{arc}(a)| < \pi/2. \qquad (2.7)$$

At this point the formula

$$\langle \eta_0, \eta_0 | | \eta, \eta_0 \rangle$$

$$= \left[\frac{2\eta_0}{2\eta} \right]^{\gamma+1} \left[\frac{(n_0 - 1)!(2\gamma + 1)!}{(n_0 + 2\gamma)!} \right]^{1/2}$$

$$\times \left[\frac{(n - 1)!(2\gamma + 1)!}{(n + 2\gamma)!} \right]^{1/2}$$

$$\times \frac{1}{2\pi i} \oint_{(0+)} du \, \frac{(1+u)^{n_0+2\gamma}}{u^{n_0}} \\ \times \frac{1}{2\pi i} \oint_{(0+)} dw \, \frac{(1+w)^{n+2\gamma}}{w^n} \frac{1}{(w-w_p)^{2\gamma+2}}, \\ w_p \equiv -\frac{\eta+\eta_0+u2\eta_0}{2\eta},$$
(2.8)

is obtained. In order to obtain the formula (2.8) the convergence condition $\eta + \eta_0 + 2\eta_0 \operatorname{Re}(u) + 2\eta \operatorname{Re}(w) > 0$, corresponding to $\operatorname{Re}(a) \ge 0$ in Eq. (2.7), is required to allow performing the r integral first. Since $\eta + \eta_0 > 0$, this inequality can be arranged simply by taking the loop (0 +) about the origins in the u and w planes sufficiently small. The convergence condition then implies that $\eta + \eta_0$ $+2\eta_0 \operatorname{Re}(u) > -2\eta \operatorname{Re}(w),$ $-(\eta + \eta_0 + 2\eta_0 \operatorname{Re}(u))$ $< 2\eta \operatorname{Re}(w)$, or $\operatorname{Re}(w_p) < \operatorname{Re}(w)$ for all w on the contour of integration. If follows that the singular point w_p in the integrand of the w integral in Eq. (2.8) lies outside the integration contour, when the contours are chosen so as to permit performing the r integral first.

The next step is to perform the *w* integral. This is facilitated by the fact that the sum of the exponents in the integrand is equal to minus two, $(n + 2\gamma) + (-n)$ $+ (-(2\gamma + 2)) = -2$. Such an integral is said to be of the Gaussian hypergeometric type. For an integral of the Gaussian hypergeometric type a linear transformation $w \equiv (At + B)/(Ct + D)$ can be performed and the transformed integral will have the same general structure as the original integral, having the same number of factors, each of which has the same exponent as before the transformation.²² For example, to the factor $(w - w_p)^{-(2\gamma + 2)}$ of the original integral, there will correspond a factor proportional to $(t - t_p)^{(2\gamma + 2)}$ in the transformed integral, where t_p is the image of w_p . It is true that powers of a new factor (Ct + D) are introduced by the transformation, but for an integral of the Gaussian hypergeometric type the sum of all powers of (Ct + D) that are introduced by the transformation just cancels a corresponding contribution from the Jacobian dw/dt. This "covariance" property of a Gaussian hypergeometric integral under linear transformations can be used to eliminate a particular factor of the integrand, thereby greatly simplifying the integral. This is accomplished for the factor $(w - w_p)^{-(2\gamma + 2)}$ in Eq. (2.8) by mapping the point w_p onto the point at infinity, $t_p = \infty$. The linear transformation needed for this is

$$w = tw_p / (t + 1 + w_p) .$$
 (2.9)

The t plane is mapped into the w plane in such a way that the points 0, -1, w_p in the w plane go over into corresponding points in the t plane as follows:

$$\frac{w=0 \quad w=-1 \quad w=w_p}{t=0 \quad t=-1 \quad t=\infty}.$$
 (2.10)

Accordingly, the w integral takes the form

$$\frac{1}{2\pi i} \oint_{(0+)} dw \frac{(1+w)^{n+2\gamma}}{w^n} \frac{1}{(w-w_p)^{2\gamma+2}} = (-1)^{n-1} \frac{(1+w_p)^{n-1}}{(-w_p)^{n+2\gamma+1}} \frac{1}{2\pi i} \\ \times \oint_{(0+)} dt \frac{(1+t)^{n+2\gamma}}{t^n} = (-1)^{n-1} \frac{(1+w_p)^{n-1}}{(-w_p)^{n+2\gamma+1}} \frac{(n+2\gamma)!}{(n-1)!(2\gamma+1)!},$$
(2.11)

when transformed into the t plane. The final t integral in Eq. (2.11) was evaluated by use of the following integral representation of a binomial coefficient:

$$(2\pi i)^{-1} \oint dt (1+t)^{n+2\gamma} t^{-n} = \frac{(n+2\gamma)!}{(n-1)!(2\gamma+1)!}.$$
(2.12)

When the final result (2.11) for the w integral is substituted into the expression (2.8) for the overlap integral and the factors $(-w_p)$ and $(1 + w_p)$ are written out in terms of u, the overlap integral $\langle \eta_0, n_0 | | \eta, n \rangle$ goes over into the form

$$\langle \eta_{0}, \eta_{0} | |\eta, n \rangle$$

$$= (4\eta_{0}\eta)^{\gamma+1} \left[\frac{(n_{0}-1)!}{(n_{0}+2\gamma)!} \right]^{1/2} \left[\frac{(n+2\gamma)!}{(n-1)!} \right]^{1/2}$$

$$\times \frac{1}{2\pi i} \oint_{(0+)} du \frac{(1+u)^{n_{0}+2\gamma}}{u^{n_{0}}}$$

$$\times \frac{(\eta_{0}-\eta+2u\eta_{0})^{n-1}}{(\eta_{0}+\eta+2u\eta_{0})^{n+2\gamma+1}}.$$

$$(2.13)$$

Notice that the final u integral is again of the Gaussian hypergeometric type. Accordingly one final linear transformation is all that is needed to evaluate the overlap integral. This

transformation chosen time the is to map $u_p = -(\eta_0 + \eta)/(2\eta_0)$ into the point at infinity, while leaving u = 0 and u = -1 invariant. The factor $(\eta_0 + \eta + 2u\eta_0)^{n+2\gamma+1}$ is thereby eliminated from the integrand. The factor $(\eta_0 - \eta + 2u\eta_0)^{n-1}$ that remains is expanded by the binomial theorem, and the integral performed term by term using Eq. (2.12). The result is the expression (2.4) for the overlap integral. These final steps in the derivation need not be presented here in detail, since the results needed for the Green's function calculation of Sec. III are already in place: the representation (2.13) of the overlap integral forms the starting point for the Green's function calculation of Sec. III.

III. STURMIAN REPRESENTATION OF THE COULOMB GREEN'S FUNCTION

The discrete representation

$$g = S \left[\frac{r}{2} \right]^{1/2} \sum_{nLJM} \frac{|\eta, nLJM\rangle \langle \eta, nLJM |}{(\gamma + n)\eta - EZ\alpha} \left[\frac{r}{2} \right]^{1/2} S^{-1},$$

$$S \equiv \cosh(\theta/2) + i\sigma \cdot \hat{r} \sinh(\theta/2),$$

$$\theta \equiv \tanh^{-1}(Z\alpha/K), \quad K \equiv \sigma \cdot L + 1, \qquad (3.1)$$

of the Coulomb Green's function of the second-order Dirac equation has been derived earlier.⁴ The derivation exploits the Coulomb Sturmian basis set (2.2), with $\eta \equiv (m^2 - E^2)^{1/2}$. The energy is for the moment assumed to be a real number lying in the range $|E| < E_{\text{ground state}}$. This restriction can be lifted later by means of an analytic continuation argument.

As indicated in the Introduction, Coulomb Sturmian matrix elements will be investigated not for g itself, but for $r^{1/2}gr^{1/2}$. Accordingly, Eq. (3.1) is rewritten as

$$r^{1/2}gr^{1/2} \equiv (rS)g(rS^{-1}),$$

$$g = \frac{1}{2\eta} \sum_{nLJM} \frac{|\eta, nLJM\rangle \langle \eta, nLJM|}{(\gamma + n) - EZ\alpha/\eta},$$
 (3.2)

and the matrix elements of rS, rS⁻¹, and g will be sought. The calculation of the matrix elements of rS and rS⁻¹ is of a rather trivial nature, and is discussed in the Appendix. Here a closed expression for $\langle \eta_0, n_2, L, J, M | \mathbf{g} | \eta_0, n_1, L, J, M \rangle$, in terms of Gaussian hypergeometric functions, will be obtained. Note that, due to the diagonal nature of g in the angular momentum quantum numbers, the matrix elements $\langle \eta_0, n_2, L, J, M | \mathbf{g} | \eta_0, n_1, L, J, M \rangle$ are the only nonzero ones. The real parameter $\eta_0 > 0$ can be quite arbitrary. From Eq. (3.2) the desired matrix element can be written in terms of the radial parts of the Sturmian functions as

$$\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle$$

= $\sum_{n=1}^{\infty} \frac{1}{2\eta} \frac{\langle \eta_0 n_2 LJ | | \eta_n LJ \rangle \langle \eta_n LJ | | \eta_0 n_1 LJ \rangle}{\gamma + n - \kappa},$
 $\kappa \equiv EZ\alpha/\eta.$ (3.3)

Next, the overlap integrals in Eq. (3.3) are represented by contour integrals according to Eq. (2.13),
$$= \frac{(4\eta_0\eta)^{2\gamma+2}}{2\eta} \left[\frac{(n_2-1)!(2\gamma+1)!}{(n_2+2\gamma)!} \right]^{1/2} \left[\frac{(n_1-1)!(2\gamma+1)!}{(n_1+2\gamma)!} \right]^{1/2} \frac{1}{2\pi i} \oint du \frac{(1+u)^{n_2+2\gamma}}{u^{n_2}} \frac{1}{(\eta_0+\eta+2u\eta_0)^{2\gamma+2}} \\ \times \frac{1}{2\pi i} \oint dw \frac{(1+w)^{n_1+2\gamma}}{w^{n_1}} \frac{1}{(\eta_0+\eta+2w\eta_0)^{2\gamma+2}} \sum_{n=1}^{\infty} \frac{(n+2\gamma)!}{(n-1)!(2\gamma+1)!} \frac{\Xi^{n-1}}{\gamma+n-\kappa}, \\ \Xi = \frac{(\eta_0-\eta+2u\eta_0)(\eta_0-\eta+2w\eta_0)}{(\eta_0+\eta+2u\eta_0)(\eta_0+\eta+2w\eta_0)}.$$
(3.4)

The infinite series in Eq. (3.4) can be summed by means of the binomial expansion

$$(1-z)^{-\nu} = \sum_{n=0}^{\infty} \frac{(\nu-1+n)!}{(\nu-1)!n!} z^n, \quad |z| < 1.$$
(3.5)

In order to achieve this, the identity $\int_0^1 dt t^{n+\gamma-1-\kappa} = 1/(n+\gamma-\kappa)$ is employed to replace the denominator $1/(n+\gamma-\kappa)$ by a power. There is an assumption $n+\gamma-\kappa>0$ needed here to ensure convergence of the *t* integral. This convergence condition is obeyed under the restriction noted above that $|E| < E_{\text{ground state}}$. When these changes are incorporated the matrix element takes the form

 $\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle$

$$= (2\eta)^{2\gamma+1} \left[\frac{(n_2-1)!(2\gamma+1)!}{(n_2+2\gamma)!} \right]^{1/2} \left[\frac{(n_1-1)!(2\gamma+1)!}{(n_1+2\gamma)!} \right]^{1/2} \int_0^1 dt \ t^{\gamma-\kappa} \frac{1}{2\pi i} \oint du \ \frac{(1+u)^{n_2+2\gamma}}{u^{n_2}} (\eta_0 + \eta + 2u\eta_0)^{-2\gamma-2} \right] \\ \times \left[1 - t \frac{\eta_0 - \eta + 2u\eta_0}{\eta_0 + \eta + 2u\eta_0} \right]^{-2\gamma-2} \frac{1}{2\pi i} \oint dw \frac{(1+w)^{n_1+2\gamma}}{w^{n_1}} (w - w_p)^{-2\gamma-2} , \\ w_p \equiv -\frac{\eta_0 + \eta}{2\eta_0} \left[1 - t \left[\frac{\eta_0 - \eta + 2u\eta_0}{\eta_0 + \eta + 2u\eta_0} \right] \left[\frac{\eta_0 - \eta}{\eta_0 + \eta} \right] \right] \left[1 - t \left[\frac{\eta_0 - \eta + 2u\eta_0}{\eta_0 + \eta + 2u\eta_0} \right] \right]^{-1} .$$
(3.6)

The w integral encountered in Eq. (3.6) is of the Gaussian hypergeometric type, and is in fact identical to the integral (2.11) already evaluated, aside from the different meaning of w_p . Using Eq. (2.11), and writing out all terms involving w_p in terms of u, gives

$$\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle = (-1)^{n_1 - 1} \frac{(4\eta\eta_0)^{2\gamma + 2}}{2\eta} \left[\frac{(n_2 - 1)!}{(n_2 + 2\gamma)!} \right]^{1/2} \left[\frac{(n_1 + 2\gamma)!}{(n_1 - 1)!} \right]^{1/2} \int_0^1 dt \, t^{\gamma - \kappa} \frac{1}{2\pi i} \oint du \frac{(1 + u)^{n_2 + 2\gamma}}{u^{n_2}} \\ \times \frac{\left[(\eta_0 + \eta + 2u\eta_0) (\eta_0 - \eta) - (\eta_0 - \eta + 2u\eta_0) (\eta_0 + \eta)t \right]^{n_1 - 1}}{\left[(\eta_0 + \eta + 2u\eta_0) (\eta_0 + \eta) - (\eta_0 - \eta + 2u\eta_0) (\eta_0 - \eta)t \right]^{n_1 + 2\gamma + 1}}.$$

$$(3.7)$$

The u integral in Eq. (3.7) is again of the hypergeometric type. The evaluation of the u integral begins with the linear transformation

$$u = \frac{su_p}{s+1+u_p}, \quad u_p \equiv -\frac{1}{\zeta} \frac{1-t(1-\zeta)^2}{1+t(1+\zeta)}, \quad \zeta \equiv \frac{2\eta_0}{\eta_0+\eta}, \quad (3.8)$$

which has the effect of eliminating the factor

 $[(\eta_0 + \eta + 2u\eta_0)(\eta_0 + \eta) - (\eta_0 - \eta + 2u\eta_0)(\eta_0 - \eta)t]^{-(n_1 + 2\gamma + 1)}$

from the integrand, while preserving the general structure

$$u^{-n_2}(1+u)^{n_2+2\gamma}$$

The result of the linear transformation to the complex s plane is

$$\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle$$

$$= \frac{1}{2\eta_0} \xi^{2\gamma+3} (2-\xi)^{2\gamma+1} (1-\xi)^{n_2-n_1} \left[\frac{(n_2-1)!}{(n_2+2\gamma)!} \right]^{1/2} \left[\frac{(n_1+2\gamma)!}{(n_1-1)!} \right]^{1/2} \int_0^1 dt \, t^{\gamma-\kappa} \frac{(1-t)^{n_2-n_1}}{(1-t(1-\xi)^2)^{n_2+n_1+2\gamma}} \\ \times \frac{1}{2\pi i} \oint ds \, \frac{1+s)^{n_2+2\gamma}}{s^{n_2}} \left[(1-\xi)^2 (1-t)^2 + ts\xi^2 (2-\xi)^2 \right]^{n_1-1}.$$
(3.9)

The result of the s integration is a hypergeometric function that breaks off to form a polynomial of order $\min(n_2n_1)$:

 $\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle$

$$=\frac{1}{2\eta_{0}}\xi^{2\gamma+3}(2-\xi)^{2\gamma+1}(1-\xi)^{n_{2}-1+n_{1}-1}\frac{1}{(2\gamma+1)!}\left[\frac{(n_{2}+2\gamma)!}{(n_{2}-1)!}\right]^{1/2}\left[\frac{(n_{1}+2\gamma)!}{(n_{1}-1)!}\right]^{1/2} \times \int_{0}^{1}dt \, t^{\gamma-\kappa}\frac{(1-t)^{n_{2}-1+n_{1}-1}}{(1-t(1-\xi)^{2})^{n_{2}+n_{1}+2\gamma}}{}_{2}F_{1}(-(n_{2}-1),-(n_{1}-1),2\gamma+2;\Lambda), \quad \Lambda \equiv \frac{t}{(1-t)^{2}}\frac{\xi^{2}(2-\xi)^{2}}{(1-\xi)^{2}}.$$
(3.10)

It is now an easy step to complete the calculation and obtain a closed expression for the matrix element. The hypergeometric polynomial $_2F_1(-(n_2-1), -(n_1-1), 2\gamma + 2; \Lambda)$ is expanded in a series, and the integral over t is performed term by term, with the result

$$\langle \eta_0 n_2 LJM | \mathbf{g} | \eta_0 n_1 LJM \rangle$$

$$= \frac{1}{2\eta_0} \xi^{2\gamma+3} (2-\xi)^{2\gamma+1} (1-\xi)^{n_2-1+n_1-1} \frac{1}{(2\gamma+1)!} \left[\frac{(n_2+2\gamma)!}{(n_2-1)!} \right]^{1/2} \left[\frac{(n_1+2\gamma)!}{(n_1-1)!} \right]^{1/2}$$

$$\times \sum_{p=0}^{n_{\varsigma}-1} \frac{(-(n_2-1))_p (-(n_1-1))_p}{(2\gamma+2)_p} \frac{[\xi^2 (2-\xi)^2/(1-\xi)^2]^p}{p!} \frac{\Gamma(\gamma+p+1-\kappa)(n_1+n_2-2-2p)!}{\Gamma(\gamma+n_1+n_2-p-\kappa)}$$

$$\times {}_2F_1(n_1+n_2+2\gamma,\gamma+p+1-\kappa,\gamma+n_1+n_2-p-\kappa;(1-\xi)^2).$$

$$(3.11)$$

The result (3.11) has been confirmed by an independent calculation in which the matrix elements are derived from the coordinate space representation of the Coulomb Green's function,²³ and using integral representations from the theory of special functions.²⁴

ſ

ACKNOWLEDGMENTS

Part of this work was carried out in the summer of 1986 while the author was a visiting fellow in the Floyd R. Newman Laboratory of Nuclear Studies at Cornell University. I am grateful to the Physics Department at Cornell University for their hospitality.

This research was supported by the National Science Foundation under Grant No. PHY-8603769.

APPENDIX: MATRIX ELEMENTS OF THE FACTORS rSAND rS^{-1}

Matrix elements of the operators

$$rS = r\cosh(\theta/2) + ri\sigma \cdot \hat{r}\sinh(\theta/2)$$
(A1)

and

$$rS^{-1} = r \cosh(\theta/2) - ri\sigma \hat{r} \sinh(\theta/2),$$

$$\theta \equiv \tanh^{-1}(Z\alpha/K), K = \sigma \mathbf{L} + 1,$$
(A2)

will be considered briefly. The fact that θ is diagonal in the angular momentum quantum numbers L, J, M makes the calculation relatively easy: $\langle \eta_0, n_2, L, J, M | r | \eta_0, n_1, L, J, M \rangle$ is the key matrix element needed for the $\cosh(\theta/2)$ terms of rS and rS^{-1} . The result

$$\langle \eta_0 n_2 LJM | 2\eta_0 r | \eta_0 n_1 LJM \rangle$$

= $2(\gamma + n_2) \delta_{n_2 n_1}$
- $(n_2(n_2 + 1 + 2\gamma))^{1/2} \delta_{n_2 + 1, n_1}$
- $((n_2 - 1)(n_2 + 2\gamma))^{1/2} \delta_{n_2 - 1, n_1}$ (A3)

for this matrix element incorporates the identities²⁵

$$xL_n^{\mu}(x) = (\mu + 1 + 2n)L_n^{\mu}(x) - (n + \mu)L_{n-1}^{\mu}(x)$$

$$-(n+1)L_{n+1}^{\mu}(x)$$
 (A4)

$$\int_0^\infty dx \, x^\mu e^{-x} L^\mu_n(x) L^\mu_m(x) = \frac{(n+\mu)!}{n!} \delta_{n,m} \,. \quad (A5)$$

Similar results can be obtained for the terms of rS and rS^{-1} involving $\sinh(\theta/2)$. This time the operator is diagonal in J and M, but not in L. The identity $\sigma \cdot \hat{r}Y_{LJM}(\hat{r}) = Y_{L'JM}(\hat{r})$, where L + L' = 2J, provides selection rules according to which the key nonzero matrix elements needed are now of the type

$$\langle \eta_0, n_2, J - \frac{1}{2}, JM | 2\eta_0 r | \eta_0, n_1, J + \frac{1}{2}, JM \rangle,$$

or

$$\langle \eta_0, n_2, J + \frac{1}{2}, JM | 2\eta_0 r | \eta_0, n_1, J - \frac{1}{2}, JM \rangle.$$

The equation

$$\langle \eta_{0}, n_{2}, L_{2} = J - \frac{1}{2} JM | 2\eta_{0}r | \eta_{0}, n_{1}, L_{1} = J + \frac{1}{2} JM \rangle$$

$$= \delta_{n_{2}, n_{1}} [(n_{1} + 2\gamma_{1})(n_{1} + 2\gamma_{1} - 1)]^{1/2}$$

$$- 2\delta_{n_{2}, n_{1} + 1} [n_{1}(n_{1} + 2\gamma_{1})]^{1/2}$$

$$+ \delta_{n_{2}, n_{1} + 2} [n_{1}(n_{1} + 1)]^{1/2}$$
(A6)

and the analogous equation for

$$\langle \eta_{0}, n_{2}, J + \frac{1}{2}, JM | 2\eta_{0}r | \eta_{0}, n_{1}, J - \frac{1}{2}, JM \rangle$$

can be obtained using the identity²⁷

$$L_{n-1}^{2\gamma-1}(x) = L_{n-1}^{2\gamma+1}(x) - 2L_{n-2}^{2\gamma+1}(x) + L_{n-3}^{2\gamma+1}(x),$$
(A7)

and Eq. (A5).

¹A. Maquet, Phys. Rev. A 15, 1088 (1977).

²J. Schwinger, J. Math. Phys. 5, 1606 (1964).

³L. Hostler, J. Math. Phys. 11, 2966 (1970).

⁴L. C. Hostler, J. Math. Phys. 24, 2366 (1983).

⁵N. L. Manakov, L. P. Rapoport, and S. A. Zapryagaev, Phys. Lett. A 43, 139 (1973); N. L. Manakov, L. P. Rapoport, and S. A. Zapryagaev, J. Phys. B 7, 1076 (1974); Ya. I. Granovskii and V. I. Nechet, Teor. Mat.

and²⁶

- ⁶O. Laporte and G. E. Uhlenbeck, Phys. Rev. 37, 1380 (1931).
- ⁷R. P. Feynman and M. Gell-Mann, Phys. Rev. 109, 193 (1958).
- ⁸L. M. Brown, Phys. Rev. 111, 957 (1958).
- ⁹M. Tonin, Nuovo Cimento 14, 1108 (1959).
- ¹⁰W. R. Theis, Fortschr. Phys. 7, 559 (1959).
- ¹¹H. Pietschmann, Acta Phys. Austriaca 14, 63 (1961).
- ¹²L. M. Brown, "Two-component fermion theory," in *Lectures in Theoretical Physics* (Interscience, New York, 1962), Vol. IV.
- ¹³L. M Brown, "Quantum electrodynamics at high energy," in Topics in Theoretical Physics, Proceedings of the Liperi Summer School in Theoretical Physics 1967, edited by C. Cronstrom (Gordon and Breach, New York, 1969), p. 113.
- ¹⁴L. C. Hostler, J. Math. Phys. 23, 1179 (1982).
- ¹⁵L. C. Hostler, J. Math. Phys. 26, 124 (1985).
- ¹⁶L. C. Hostler, J. Math. Phys. 26, 1348 (1985); 27, 2208(E) (1986).
- ¹⁷L. C. Hostler, J. Math. Phys. 27, 2423 (1986).
- ¹⁸L. C. Biedenharn and L. P. Horwitz, Found. Phys. 14, 953 (1984).
- ¹⁹L. C. Hostler, J. Math. Phys. 28, 720 (1987).

- ²⁰R. N. Hill and B. D. Huxtable, J. Math. Phys. 23, 2365 (1982).
- ²¹H. Buchholz, *Die Konfluente Hypergeometrische Function* (Springer, Berlin, 1953), p. 135, Eq (2).
- ²²F. Klein, Vorlesungen ueber die Hypergeometrische Funktion (Springer, Berlin, 1933).
- ²³See Ref. 4, Eq. (2.21).
- ²⁴An integral representation for a product of two Whittaker functions with different arguments is needed [Ref. 21, Eq. (5c), p. 86; the condition $\text{Re}(\mu) > 0$ stated in Buchholz is a misprint, and should read Re(t) > 0]; as well as an integral representation for the Bessel function, $I_v(z)$ [G. N. Watson, A Treatise on the Theory of Bessel Functions (Cambridge U. P., London, 1962), 2nd ed., p. 177, Eq. (8), and the relation on p. 77: $I_v(z) = e^{-\pi i v/2} J_v(z e^{\pi i/2}), -\pi < \arccos(z) < \pi/2$].
- ²⁵See Ref. 21. p. 137, Eq. (10b).
- ²⁶See Ref. 21, p. 136, Eq. (9).
- ²⁷See Ref. 21, p. 137, Eq. (10g) implies the relationship $nL_n^{\mu}(x) (\mu + n)L_{n-1}^{\mu}(x) = -\mu L_n^{\mu}(x) + (\mu + n)L_n^{\mu-1}(x)$, or $L_n^{\mu}(x) L_{n-1}^{\mu}(x) = L_n^{\mu-1}(x)$. This last identity used two times leads to Eq. (A7) in the Appendix.

A note on the electron self-energy

Alireza Abbasabadi and Wayne W. Repko Department of Physics and Astronomy, Michigan State University, East Lansing, Michigan 48824

(Received 1 October 1986; accepted for publication 30 July 1987)

Using dimensional regularization, the lowest-order electron self-energy function in an arbitrary covariant gauge is derived. For off-mass-shell electrons, the usual expression for the finite portion Σ_f of the self-energy as the number of space-time dimensions *n* approaches 4 is recovered. In the case of on-mass-shell electrons, the condition $(\not p - m)\Sigma_f \rightarrow 0$ as $\not p \rightarrow m$, which is necessary to make the usual separation of the renormalization constant unambiguous, requires that *n* approach 4 from above, i.e., $n \rightarrow 4_+$. This necessary condition on Σ_f is not satisfied by the off-mass-shell expression in the limit $p^2 \rightarrow m^2$ due to a branch point in the self-energy operator.

I. INTRODUCTION

The separation of renormalization constants from finite quantities in renormalizable field theories is a time-honored process which is facilitated by the use of regularization techniques.¹⁻³ In quantum electrodynamics, the necessity for a regularization scheme to respect gauge invariance was emphasized by Gupta.² Consistency with gauge invariance together with ease of implementation makes the dimensional regularization technique³ a particularly attractive alternative to regulator schemes involving covariant cutoffs. While applying dimensional regularization for certain radiative corrections to positronium energy levels and decays,⁴ we encountered a question, which may be of general interest, about how the renormalization constant associated with the electron self-energy $\Sigma(p)$ is separated from the finite part.

Recall that regardless of the method one may use for regularization, it is convenient to express the regularized $\Sigma(p)$ as

$$\Sigma(p) = A + (p - m)B + (p - m)^{2}\Sigma_{f}(p), \qquad (1)$$

where A and B are independent of the electron four-momentum p, and

$$(\not p - m) \Sigma_f(p) \big|_{\not p \to m} = 0, \qquad (2)$$

in order to have a unique decomposition in Eq. (1).

Evaluation of A is straightforward, whereas for B and Σ_f , some care must be exercised.⁵ We should mention that the Σ_f , which is usually found in literature,⁶ does not satisfy the condition given in Eq. (2).

In this paper we calculate the $\Sigma(p)$ in an arbitrary covariant gauge using dimensional regularization and express it in the form of Eq. (1). Assuming that the electron is offmass-shell, we recover the usual expression for Σ_f as the number of dimensions $n \rightarrow 4$. For an on-mass-shell electron, our general expression for Σ_f satisfies Eq. (2) if we assume that the dimension of space-time is larger than 4 $(n \rightarrow 4_+)$.

Throughout our paper, we use the notation and conventions of Bjorken and Drell.⁷ We also use the natural units $\hbar = c = 1$.

II. ELECTRON SELF-ENERGY

Using dimensional regularization, the lowest-order electron self-energy in an arbitrary covariant gauge can be

written as8

$$\Sigma(p) = -ie^{2} \int \frac{d^{n}k}{(2\pi)^{4}} \gamma^{\mu} \frac{1}{\not p - \not k - m + i\eta}$$
$$\times \gamma^{\nu} \frac{1}{k^{2}} \left(g_{\mu\nu} + \epsilon \frac{k_{\mu} k_{\nu}}{k^{2}} \right). \tag{3}$$

Employing Feynman parametrization and performing the integrals on k, we find

$$\Sigma(p) = (\alpha/4\pi)\Gamma(-a)I_1(p) - (\alpha\epsilon/4\pi)m^{-2}\Gamma(1-a)I_2(p), \qquad (4)$$

where

$$I_{1}(p) = \sum_{\beta=0}^{2} g_{\beta}(p) \rho^{a} \int_{0}^{1} dx \, x^{a+\beta} \left(1 - \frac{\rho - 1}{\rho} x\right)^{a},$$

$$I_{2}(p) = \sum_{\beta=0}^{2} h_{\beta}(p) \rho^{a-1} \qquad (5)$$

$$\times \int_{0}^{1} dx \, x^{a+\beta+1} \left(1 - \frac{\rho - 1}{\rho} x\right)^{a-1},$$

with

$$\rho = (-p^2 + m^2)/m^2 - i\eta \quad (\eta > 0),$$

$$a = n/2 - 2 \quad (a \neq 0, 1, 2, ...),$$
(6)

and we disregarded any overall multiplicative constant of the form C^{α} . The coefficients g_{β} and h_{β} are

$$g_0(p) = -(2 + \epsilon + 2a + \epsilon a)\not p + (4 + 2\epsilon + 2a + \epsilon a)m,$$

$$g_1(p) = (2 - 2\epsilon + 2a)\not p - (2\epsilon + \epsilon a)m,$$
 (7)

$$g_2(p) = (3\epsilon + \epsilon a)\not p,$$

and

$$h_{0}(p) = p^{2} \not p + p^{2} m ,$$

$$h_{1}(p) = -2p^{2} \not p - p^{2} m ,$$

$$h_{2}(p) = p^{2} \not p .$$
(8)

The integrals in I_1 and I_2 can be performed and expressed in terms of hypergeometric functions (Ref. 9, Eq. 3.194 1),

$$I_{1}(p) = \sum_{\beta=0}^{2} g_{\beta}(p) \frac{\rho^{a}}{a+\beta+1} \times F\left(-a, a+\beta+1; a+\beta+2; \frac{\rho-1}{\rho}\right),$$

$$I_{2}(p) = \sum_{\beta=0}^{2} h_{\beta}(p) \frac{\rho^{a-1}}{a+\beta+2} \qquad (9) \times F\left(-a+1, a+\beta+2; a+\beta+3; \frac{\rho-1}{\rho}\right),$$

with $a \neq -2, -3, ...,$ and we have used the notation $F \equiv_2 F_1$.

By using the analytic continuation of hypergeometric functions for the region $|\rho| < 1$, and expanding them in terms of *a* (see the Appendix), we obtain

$$I_1(p) = \sum_{\beta=0}^{2} g_{\beta}(p) \lambda_{\beta}(\rho) ,$$

$$I_2(p) = \sum_{\beta=0}^{2} h_{\beta}(p) \Lambda_{\beta}(\rho) ,$$
(10)

where, for $0 < |a| < \frac{1}{2}$ and $|\rho| < 1$,

$$\lambda_{0}(\rho) = 1 - 2a + (\rho^{2a} - 1)[\rho/2(1 - \rho)] \\ \times \{-1 + [2 + \ln(1 - \rho)]a\} + O(a^{2}), \\ \lambda_{1}(\rho) = \frac{1}{2} + \frac{2\rho - 1}{2(1 - \rho)}a + (\rho^{2a} - 1)\frac{\rho^{2}}{4(1 - \rho)^{2}} \\ \times \{1 + [-2 - \ln(1 - \rho)]a\} + O(a^{2}), \quad (11)$$

$$\lambda_{2}(\rho) = \frac{1}{3} + \frac{-10\rho + 10\rho - 4}{18(1-\rho)^{2}} a + (\rho^{2a} - 1)\frac{\rho^{3}}{6(1-\rho)^{3}} \times \left\{ -1 + \left[\frac{13}{6} + \ln(1-\rho)\right] a \right\} + O(a^{2}) ,$$

and

$$\begin{split} \Lambda_{0}(\rho) &= \frac{1}{1-\rho} + (\rho^{2a}-1)\frac{\rho}{2(1-\rho)^{2}} \\ &\times \left[\frac{1}{a}-1-\ln(1-\rho)\right] + O(a) ,\\ \Lambda_{1}(\rho) &= \frac{1-3\rho}{2(1-\rho)^{2}} + (\rho^{2a}-1)\frac{\rho^{2}}{2(1-\rho)^{3}} \\ &\times \left[-\frac{1}{a}+\frac{3}{2}+\ln(1-\rho)\right] + O(a) ,\\ \Lambda_{2}(\rho) &= \frac{11\rho^{2}-7\rho+2}{6(1-\rho)^{3}} + (\rho^{2a}-1)\frac{\rho^{3}}{2(1-\rho)^{4}} \\ &\times \left[\frac{1}{a}-\frac{11}{6}-\ln(1-\rho)\right] + O(a) . \end{split}$$
(12)

By inserting these expansions into Eq. (4) and rearranging terms, we find

$$\Sigma(p) = \frac{\alpha m}{4\pi} \Gamma(-a)(3-4a) + \frac{\alpha}{4\pi} \Gamma(-a)(p-m)$$

$$\times \left[-1 - \epsilon + \left(2 + \frac{5}{3}\epsilon\right)a \right] + \frac{\alpha}{4\pi}\Gamma(1-a)$$
$$\times (\not p - m)\left(\frac{2-\epsilon}{a} - 2 + \frac{5}{3}\epsilon\right) + \widetilde{\Sigma}(p), \qquad (13)$$

where

$$\widetilde{\Sigma}(p) = \frac{\alpha}{4\pi} \left[(\not p - m) \left(\frac{\epsilon - 2}{a} + 2 - 2\epsilon \right) - \frac{(\epsilon + 1)\rho \not p}{1 - \rho} \right] \\ \times \Gamma(1 - a) + \frac{\alpha}{4\pi} (\rho^{2a} - 1) \frac{\rho}{2(1 - \rho)} \\ \times \left\{ - (2 + \epsilon)m + \left[\frac{(1 + \epsilon)(2 - \rho)\not p}{1 - \rho} \right] \\ - (4 + \epsilon)m \right] \left[-\frac{1}{a} + 1 + \ln(1 - \rho) \right] \right\} \\ \times \Gamma(1 - a).$$
(14)

The term 1/a reflects the infrared divergence, while the factor (p - m) in the second term of Eq. (13) comes from (p - m) dependence of $g_{\beta}(p)$, and its coefficient gives the ultraviolet divergence.

We write Eq. (13) in the following compact form:

$$\Sigma(p) = A + (\not p - m)B + (\not p - m)^2 \Sigma_f(p),$$
(15)

where

B

$$A = (\alpha m/4\pi)\Gamma(-a)(3-4a),$$
 (16)

is the (gauge independent) self-mass,

$$= \frac{\alpha}{4\pi} \Gamma(-a)(-1-\epsilon+4a) + \frac{\alpha}{4\pi} \Gamma(1-a)\frac{2-\epsilon}{a}, \qquad (17)$$

is related to the wave function renormalization constant, and we have used the following definition:

$$\widetilde{\Sigma}(p) = (\not p - m)^2 \Sigma_f(p) .$$
(18)

Using the identities

$$(\not p - m) = [-(\not p + m)/m^2 \rho](\not p - m)^2,$$

$$1 = [(2\not p - m\rho + 2m)/m^3 \rho^2](\not p - m)^2,$$
(19)

and Eqs. (14) and (18), we find

$$\Sigma_{f}(p) = \frac{\alpha}{4\pi m} \Gamma(1-a) \left\{ \frac{1+\epsilon}{1-\rho} + (2+\epsilon)C + (\epsilon-2+3\rho)D - \frac{p+m}{m\rho} \left[\frac{\epsilon-2}{a} + \frac{4+(\epsilon-3)\rho}{1-\rho} + (4+2\epsilon)C + (2\epsilon-4+(4-2\epsilon)\rho+(1+\epsilon)\rho^{2})D \right] \right\},$$
(20)

where

$$C = (\rho^{2a} - 1)/2(1 - \rho), \qquad (21)$$

$$D = [(\rho^{2a} - 1)/2(1 - \rho)^2][1/a - 1 - \ln(1 - \rho)]. \quad (22)$$

Henceforth, we restrict the values of a to $0 < a < \frac{1}{2}$ [this is a necessary condition in order to satisfy Eq. (2), see Eq. (25)].

The expression for $\Sigma_f(p)$ can be simplified (in terms of *a*) if we write it in two separate forms, one for $\rho = 0$, and the other for $0 < |\rho| < 1$. In the following we use the notation

$$\Sigma_f(p) \Big|_{p=m} \equiv \Sigma_f^{\text{on}}, \qquad (23)$$

$$\Sigma_f(p) \Big|_{p \neq m} \equiv \Sigma_f^{\text{off}} \,. \tag{24}$$

For $\rho = 0$ (p = m), we have $\lim_{n \to \infty} \rho^{2a} \to 0$, and find

$$\Sigma_{f}^{\text{on}} = \frac{\alpha}{4\pi m} \left[\left(-3 + \frac{3}{2}\epsilon \right)/a + (1+\epsilon) -\rho^{2a-1} \left((2\epsilon - 4)/a + 8 \right) \right] \Gamma(1-a) .$$
 (25)

For $0 < |\rho| < 1$ $(\not p \neq m)$, we use the expansion $\rho^{2a} = 1 + 2a \ln \rho$, and find

$$\Sigma_{f}^{\text{off}}(p) = \frac{\alpha}{4\pi m} \left\{ \frac{1+\epsilon}{1-\rho} + \frac{3\rho+\epsilon-2}{(1-\rho)^{2}} \ln \rho - \frac{\not{p}+m}{m\rho} \left[\frac{(\epsilon-3)\rho+4}{1-\rho} + \frac{(1+\epsilon)\rho^{2}+(4-2\epsilon)\rho+(2\epsilon-4)}{(1-\rho)^{2}} \ln \rho + \frac{\epsilon-2}{a} \right] \right\} \Gamma(1-a) .$$
(26)

Notice that $\lim_{p\to m} (p - m) \Sigma_f^{on}(p) \to 0$ (this is not true for Σ_f^{off}). This relation assures that our decomposition of $\Sigma(p)$ in Eq. (15) is unique. It also guarantees that after mass renormalization, the electron propagator (near-mass-shell) will be modified, due to the self-energy correction, by a multiplicative factor. We can extend the domain of Σ_f^{off} by analytic continuation. It will have the same form for $|\rho| \ge 1$.

If we are not interested in a single expression for $\Sigma_f(p)$ for whole region $|\rho| < 1$, we can find Σ_f^{on} by keeping only those terms in expansions of hypergeometric functions [see Eq. (A3)] which are up to the second order in ρ . For Σ_f^{off} , it is much easier to use the original Eq. (9), which can be expanded in terms of ρ in region $\operatorname{Re}(\rho) > \frac{1}{2}$, and by analytic continuation it can be shown that it has the same form for all ρ (except at $\rho = 0$).

III. SUMMARY AND CONCLUSIONS

Our main objective for representing $\Sigma_f(p)$ by a single expression is to show, in a clear way, that the inequality $\lim_{\rho \to m} \Sigma_f^{\text{off}}(p) \neq \Sigma_f^{\text{on}}$ is due to the use of the expansion $\rho^{2a} = 1 + 2a \ln \rho$ in finding Σ_f^{off} , which is justified only for $\rho \neq 0$.

For one-loop corrections, the difference between Σ_f^{on} and $\lim_{p \to m} \Sigma_f^{\text{off}}$ does not present a problem since ρ is a nonvanishing constant in these cases. However, when dealing with multiloop corrections, some thought must be given to the values of the integration variable encountered in expressions containing Σ_f . In such circumstances, it is entirely possible that the surface $\rho = 0$ could make a significant contribution which is sensitive to the difference between Σ_f^{on} and $\lim_{p \to m} \Sigma_f^{\text{off}}$. If this is the case, then one should employ Eq. (20) for $\Sigma_f(p)$, which is correct for all values of p.

ACKNOWLEDGMENT

This research was supported in part by the National Science Foundation under Grant No. PHY-83-05722.

APPENDIX: REDUCTION OF $I_1(\rho)$ AND $I_2(\rho)$

The analytic continuation of $F(a,b;b+1;(\rho-1)/\rho)$ to the region $|\rho| < 1$ can be found by the relation¹⁰

$$F\left(a,b;b+1;\frac{\rho-1}{\rho}\right)$$

= $\frac{b\rho^{a}}{b-a}F(a,1;a-b+1;\rho)$
+ $\frac{\Gamma(b+1)\Gamma(a-b)}{\Gamma(a)}\left(\frac{\rho}{1-\rho}\right)^{b}$, (A1)

with condition $|\arg(1/\rho)| < \pi$, which can be satisfied [even for real values of four-momentum p, since $\rho = (-p^2 + m^2)/m^2 - i\eta$, $\eta > 0$].

In this Appendix we work out $F(-a+1,a+2; a+3; (\rho-1)/\rho)$; the other F's can be found in a similar way.

From Eq. (A1) we obtain

$$F\left(1 - a, a + 2; a + 3; \frac{\rho - 1}{\rho}\right)$$

= $\frac{(a+2)\rho^{1-a}}{2a+1}F(1 - a, 1; -2a;\rho)$
+ $\frac{\Gamma(a+3)\Gamma(-1-2a)}{\Gamma(1-a)}\left(\frac{\rho}{1-\rho}\right)^{a+2}$. (A2)

Using the formula

$$F(a,1;c;\rho) = \frac{\Gamma(c)}{\Gamma(a)} \sum_{m=0}^{\infty} \frac{\Gamma(a+m)}{\Gamma(c+m)} \rho^m, \quad |\rho| < 1,$$
(A3)

we find

$$F(-a+1,1; -2a;\rho) = 1 + \frac{1}{-2a} \frac{\Gamma(-2a+1)}{\Gamma(-a+1)} \times \sum_{m=1}^{\infty} \frac{\Gamma(-a+1+m)}{\Gamma(-2a+m)} \rho^{m}.$$
 (A4)

Notice that in the closed region of an annular ring which is bounded by two concentric circles centered at origin of the *a* plane, with radii ϵ_1 and $\frac{1}{2} - \epsilon_2$ (ϵ_1 and ϵ_2 are positive but much smaller than 1), $\Gamma(-2a+1)/\Gamma(-a+1)$ and $\Gamma(-a+1+m)/\Gamma(-2a+m)$ are analytic functions of *a*, and the series in Eq. (A4) is uniformly convergent $(|\rho| < 1)$ and consequently its sum is an analytic function of *a*. Therefore $F(-a+1,1; -2a;\rho)$ itself is an analytic function of *a* and we can expand it about a = 0,

$$F(-a+1,1;-2a;\rho) = 1 + \frac{1}{-2a} \left\{ \sum_{m=1}^{\infty} m\rho^m + a \sum_{m=1}^{\infty} \frac{\partial}{\partial a} \left[\frac{\Gamma(-2a+1)}{\Gamma(-a+1)} \right]_{a=0} \rho^m + O(a^2) \right\}$$
$$= \frac{2-3\rho}{2(1-\rho)^2} + \frac{\rho}{2(1-\rho)^2} \ln(1-\rho)$$
$$+ \frac{-\rho}{2(1-\rho)^2} \frac{1}{a} + O(a) .$$
(A5)

In derivation of Eq. (A5) we have used the following formulas (Ref. 9. Eqs. 8.36 and 1.513 6):

$$\frac{\Gamma'(m+1)}{\Gamma(m+1)} - \frac{\Gamma'(1)}{\Gamma(1)} = \sum_{k=1}^{m} \frac{1}{k}, \quad m = 1, 2, 3, \dots,$$
 (A6)

$$\sum_{m=1}^{\infty} \left(\sum_{k=1}^{m} \frac{1}{k} \right) \rho^{m} = \frac{-1}{1-\rho} \ln(1-\rho) , \qquad (A7)$$

$$\sum_{m=0}^{\infty} \rho^{m} = \frac{1}{1-\rho} \,. \tag{A8}$$

with their derivatives with respect to ρ (after multiplying by appropriate powers of ρ).

For the second term of $F(-a+1,a+2;a+3; (\rho-1)/\rho)$ in Eq. (A2), we obtain

$$\frac{\Gamma(a+3)\Gamma(-2a-1)}{\Gamma(-a+1)} \left(\frac{\rho}{1-\rho}\right)^{a+2}$$

$$= \frac{1}{(-2a-1)(-2a)} \frac{\Gamma(a+3)\Gamma(-2a+1)}{\Gamma(-a+1)}$$

$$\times \frac{\rho^{a+2}}{(1-\rho)^2} [1-a\ln(1-\rho)+O(a^2)]$$

$$= \frac{\rho^{a+2}}{(1-\rho)^2} \left[\frac{1}{a} - \frac{1}{2} - \ln(1-\rho)\right] + O(a), \quad (A9)$$

where we have used the expansion

$$(1-\rho)^{-a} = 1 - a \ln(1-\rho) + O(a^2), \quad |\rho| < 1.$$
(A10)

Therefore, combining (A2), (A5), and (A9), we find, for $|\rho| < 1$,

$$F\left(-a+1,a+2;a+3;\frac{\rho-1}{\rho}\right)$$

= $\frac{\rho^{a+2}}{(1-\rho)^2} \left[\frac{1}{a} - \frac{1}{2} - \ln(1-\rho)\right] + \frac{\rho^{2-a}}{(1-\rho)^2}$
 $\times \left[-\frac{1}{a} - \frac{3}{2} + \frac{2}{\rho} + \ln(1-\rho)\right] + O(a).$
(A11)

¹W. Pauli and F. Villars, Rev. Mod. Phys. 21, 434 (1949).

- ²S. N. Gupta, Proc. Phys. Soc. London Ser. A 66, 129 (1953).
 ³G. 't Hooft and M. Veltman, Nucl. Phys. B 44, 189 (1972); C. G. Bollini and J. J. Giambiagi, Nuovo Cimento B 12, 20 (1972); J. F. Ashmore, Lett. Nuovo Cimento 4, 289 (1972); for a review, see G. Leibbrandt, Rev. Mod. Phys. 47, 849 (1975).
- ⁴A. Abbasabadi and W. W. Repko, "Bound state perturbation theory and annihilation effects in positronium," Nuovo Cimento (in press).
 ⁵This was called to out attention by D. Heckathorn (private communica-
- tion); see also, Y. Tomozawa, Ann. Phys. (NY) **128**, 491 (1980).
- ⁶J. M. Jauch and F. Rohrlich, *The Theory of Photons and Electrons* (Springer, Berlin, 1976), 2nd ed.
- ⁷J. D. Bjorken and S. D. Drell, *Relativistic Quantum Fields* (McGraw-Hill, New York, 1965).
- ⁸W. J. Marciano and A. Sirlin, Nucl. Phys. B 88, 86 (1975), and references therein.
- ⁹I. S. Gradshteyn and I. M. Ryzhik, *Tables of Integrals, Series and Products* (Academic, New York, 1980), corrected and enlarged edition.
- ¹⁰M. Abramowitz and I. A. Stegun, *Handbook of Mathematical Functions* (Dover, New York, 1965), Chaps. 6 and 15.

Chiral symmetry breakdown. III. Delbourgo's gauge technique

D. Atkinson, A. Hulsebos, and P. W. Johnson^{a)} Institute for Theoretical Physics, P. O. Box 800, 9700 AV Groningen, The Netherlands

(Received 7 July 1987; accepted for publication 30 July 1987)

The quark propagator in massless quantum chromodynamics (QCD) is analyzed using the gauge technique. In both the Feynman and Landau gauges with a Pauli-Villars cutoff, a chirally symmetric solution is found, while a nonsymmetric solution appears at a critical coupling $\lambda_c > 0$. As the cutoff is removed, λ_c tends to 0 but the nonsymmetric solution vanishes in the continuum limit, so that chiral symmetry is then restored.

I. INTRODUCTION

Spontaneous dynamical breakdown of chiral symmetry represents an interesting and potentially important means through which particles become massive at a certain critical coupling strength because of interactions with the quantum field.¹ In QCD, this phenomenon provides a mechanism for quarks to acquire constituent masses in a formally massless theory through nonperturbative effects. The occurrence of dynamical symmetry breaking is investigated by analysis of Dyson–Schwinger equations for the fermion propagator. Such analyses are hampered by a degree of arbitrariness, in that the effect of unavoidable truncation of Dyson– Schwinger equations is difficult to assess.

Some time ago Maskawa and Nakajima showed that, in the context of quantum electrodynamics (QED), when the fermion-fermion-photon vertex function and the photon propagator are replaced by their free values, the solutions of the Dyson-Schwinger equation exhibit spontaneous chiral symmetry breaking.² In a recent paper we demonstrated that this spontaneous symmetry breaking occurs when one introduces both an infrared and an ultraviolet cutoff.³ In a more detailed analysis of the problem, this truncation procedure was found to be unacceptably gauge dependent. We showed that chiral symmetry was indeed broken in the theory with cutoffs present, but that the symmetry was restored in the continuum limit in certain gauges.⁴ It was found that chiral symmetry breaking was very sensitive to the infrared and ultraviolet behavior of the gauge-dependent free photon propagator. Furthermore, we showed that this sensitivity to choice of gauge remained even when the truncation procedure was modified to maintain consistency with the "transversality" constraint coming from gauge invariance in the ultraviolet.⁵ It is our conclusion that the free vertex approximation cannot provide a reliable indication as to the occurrence of spontaneous chiral symmetry breaking, and that other truncation schemes should be explored.

Truncation of Dyson-Schwinger equations is certainly an *ad hoc* procedure based primarily upon expediency, but one must make use of physically motivated conditions or requirements whenever possible. Several authors have chosen the quark-quark-gluon vertex in QCD to decrease as $1/\log q^2$ at large q^2 , in correspondence with the ultraviolet behavior imposed by asymptotic freedom.⁶⁻⁸ We have shown in Ref. 4 that chiral symmetry remains unbroken in certain gauges in the continuum limit in this case as well, because of formal divergence of the loop integral in the Dyson-Schwinger equation. We conclude that this truncation scheme has essentially the same difficulties as the free vertex approximation, and thus that one must look further to obtain a resolution of the ambiguity associated with gaugedependent results.

The free vertex truncation, as well as its renormalization-group-improved counterpart discussed in Ref. 4, is not consistent with the Slavnov–Taylor identity, and therefore stands in direct conflict with the requirements of gauge invariance. The gauge dependence of the results on chiral symmetry breaking is a manifestation of the gauge dependence of the truncation scheme. By contrast, in the truncation scheme proposed some time ago by Salam and Delbourgo,⁹ and studied by a number of people,^{10,11} one maintains consistency with the Slavnov–Taylor identity for the vertex function, and thus the effects of gauge dependence should be reduced.

Here we will study chiral symmetry breaking in the Salam-Delbourgo scheme, which is based upon spectral Ansätze for the propagator and vertex functions. We replace the gluon propagator by a free massive one, and discuss both Feynmann and Landau gauges in the Dyson-Schwinger equation in Sec. II. The formal divergence of the loop integral is removed by introducing a Pauli-Villars cutoff; and the Dyson-Schwinger equation reduces to a homogeneous linear equation for the spectral density function, which has nontrivial solutions.

We find that when the cutoff is present, only a chirally symmetric solution occurs below a critical coupling $\lambda_c > 0$, whereas above λ_c nonsymmetric solutions are also present. In the continuum limit as the cutoff is removed, λ_c goes formally to 0. However, in both gauges only the chirally symmetric solution survives. Therefore in the Salam-Delbourgo formalism chiral symmetry is restored in the continuum limit.

II. SALAM-DELBOURGO ANSATZ

The Dyson-Schwinger equation for the quark propagator $S'_F(p)$ can be written

$$pS'_{F}(p) - \Sigma(p)S'_{F}(p) = 1, \qquad (2.1)$$

where the self-energy is given by

^{a)} Permanent address: Physics Department, Illinois Institute of Technology, Chicago, Illinois 60616.

$$\Sigma(p) = \frac{ig^2}{(2\pi)^4} \int d^4k \, \gamma_\mu S'_F(p-k) \Gamma_\nu(p-k,p) D'_F^{\mu\nu}(k),$$
(2.2)

where Γ_{v} is the gluon-quark vertex function and D'_{F} the gluon propagator. Here g is proportional to the SU(3) coupling constant.

The Lehmann representation for the quark propagator can be written

$$S'_F(p) = \int_{-\infty}^{\infty} dw' \frac{\rho(w')}{\not p - w' + i\epsilon\epsilon(w')}, \qquad (2.3)$$

where $\epsilon(w') = \operatorname{sgn}(w')$, and ρ is an unknown spectral function. If $\rho(w)$ is even in w, $S'_F(p)$ has the form p times a function of p^2 , and so chiral symmetry is then unbroken. The Salam-Delbourgo Ansatz assumes that

$$S'_{F}(p-k)\Gamma_{\nu}(p-k_{\nu}p)S'_{F}(p) = \int_{-\infty}^{\infty} dw'\rho(w') \frac{1}{\not p - \not k - w'} \gamma_{\nu} \frac{1}{\not p - w'}, \qquad (2.4)$$

where the $i\epsilon\epsilon(w')$ is to be understood in both denominators. It is easy to see that (2.4) is consistent with the Slavnov-Taylor identity for Γ_{ν} .

With this Ansatz, the Dyson-Schwinger equation (2.1) can be rewritten in the form

$$\int_{-\infty}^{\infty} dw' \,\rho(w') \,[\not\!p - \Sigma(\not\!p - w')] \frac{1}{\not\!p - w'} = 1, \qquad (2.5)$$

where

$$\Sigma(\mathbf{p}, w') = \frac{ig^2}{(2\pi)^4} \int d^4k \, \gamma_\nu \, \frac{1}{\mathbf{p} - \mathbf{k} - w'} \, \gamma_\nu D_F^{\prime \mu\nu}(k).$$
(2.6)

In this paper, we propose to consider the Feynman and the Landau gauges. Moreover, we shall remove the divergence of the loop integral (2.6) by introducing a Pauli–Villars cutoff. We write

$$D_{F}^{\prime\mu\nu}(k) = D^{\mu\nu}(k,m) - D^{\mu\nu}(k,\Lambda) , \qquad (2.7)$$

where the cutoff Λ is much greater than *m*, the gluon mass, and where, in the Feynman gauge,

$$D^{\mu\nu}(k,m) = -g^{\mu\nu}/(k^2 - m^2 + i\epsilon), \qquad (2.8a)$$

and in the Landau gauge,

$$D^{\mu\nu}(k,m) = \left[-g^{\mu\nu} + \frac{k^{\mu}k^{\nu}}{k^{2} + i\epsilon} \right] \frac{1}{k^{2} - m^{2} + i\epsilon}.$$
(2.8b)

The corresponding expressions for Σ can be calculated:

$$\Sigma(\mathbf{p}, w') = \lambda \int_{-\infty}^{\infty} dw \, \frac{\Omega(w, w', m) - \Omega(w, w', \Lambda)}{\mathbf{p} - w} \, .$$
(2.9)

Here $\lambda = g^2/(16\pi^2)$ and

$$\Omega(w,w',m) = [\epsilon(w)/w^3]\theta(|w| - |w'| - m) \\ \times [w^2 - 4ww' + w'^2 - m^2] \\ \times \{[w^2 - (w' + m)^2][w^2 - (w' - m)^2]\}^{1/2}$$
(2.10a)

in the Feynman gauge, and

 $\Omega(w,w',m)$

$$= \frac{3}{2} [\epsilon(w)/w^{3}] \theta(|w| - |w'| - m) [(w - w')^{2} - m^{2}] \\ \times \{ [w^{2} - (w' + m)^{2}] [w^{2} - (w' - m)^{2}] \}^{1/2}$$
(2.10b)

in the Landau gauge. The expression (2.9) for $\Sigma(p)$ involves a convergent integral in both gauges, because of the Pauli– Villars cutoff.

In each gauge we substitute (2.9) into (2.5) and partially fractionate, obtaining

$$1 = \int_{-\infty}^{\infty} dw' \rho(w') \frac{\not p}{\not p - w'} - \lambda \int_{-\infty}^{\infty} dw' \rho(w')$$
$$\times \int_{-\infty}^{\infty} dw'' \frac{\Omega(w'', w', m) - \Omega(w'', w', \Lambda)}{w'' - w'}$$
$$\times \left[\frac{1}{\not p - w''} - \frac{1}{\not p - w'}\right]. \tag{2.11}$$

The imaginary part of this equation is

$$\phi(w)\rho(w) = \lambda \int_{-\infty}^{\infty} dw' \rho(w') \\ \times \frac{\Omega(w,w',m) - \Omega(w,w',\Lambda)}{w - w'}, \quad (2.12)$$

where

$$\phi(w) = w + \lambda \int_{-\infty}^{\infty} dw' \frac{\Omega(w', w, m) - \Omega(w', w, \Lambda)}{w' - w}.$$
(2.13)

The function $\phi(w)$ may be calculated in closed form for either gauge. Equation (2.12) will be analyzed, in these gauges, in the next section.

III. ANALYSIS OF THE EQUATION

Equation (2.12) can be written

 $\phi(w)\rho(w)$

$$=\lambda\theta(|w|-m)\int_{-|w|+n}^{|w|-m}dw'\rho(w')\frac{\Omega(w,w',m)}{w-w'}$$
$$-\lambda\theta(|w|-\Lambda)\int_{-|w|+\Lambda}^{|w|-\Lambda}dw'\rho(w')\frac{\Omega(w,w',\Lambda)}{w-w'}.$$
(3.1)

The θ functions are already implied by those in (2.10), they are included explicitly above to emphasize that the right-hand side of (3.1) vanishes when |w| < m.

First note that the denominator w - w' never vanishes, and that the kernel in continuous. The function $\phi(w)$ is also bounded and continuous, and we may write

$$\phi(w)/w = 1 - \lambda \left[\psi(w,m) - \psi(w,\Lambda) \right], \qquad (3.2)$$

where

$$\psi(w,m) = 2 \int_{|w|+m}^{M} \frac{dw'}{w'^{3}} P(w'^{2},w^{2},m^{2}) \\ \times \{ [w'^{2} - (w+m)^{2}] [w'^{2} - (w-m)^{2}] \}^{1/2}$$
(3.3)

with

 $P(w^{\prime 2},w^2,m^2) = 3 + (2w^2 + m^2)/(w^{\prime 2} - w^2) \quad (3.4a)$

in the Feynman gauge, and

$$P(w^{\prime 2}, w^{2}, m^{2}) = \frac{3}{2}(1 + m^{2}/(w^{\prime 2} - w^{2}))$$
(3.4b)

in the Landau gauge. It is possible to express $\psi(w,m)$ in terms of elementary functions in both gauges. We have introduced a cutoff parameter M in the integral (3.3), which must be taken to infinity in (3.2). The integral in (3.3) diverges logarithmically as $M \to \infty$, but the function $\phi(w)$ in (3.2) is well defined in this limit. In particular,

$$\phi(w)/w \to 1 - b\lambda \log(\Lambda/m) \tag{3.5}$$

as $w \rightarrow 0$, where b = 6 in Feynman gauge and b = 3 in Landau gauge. In both gauges, $\phi(w)/w$ is monotonically increasing for w > 0 and

$$\phi(w)/w \to 1 \tag{3.6}$$

as $w \to \pm \infty$.

The odd function $\phi(w)$ is zero at w = 0. In addition, under the condition

$$\lambda > \lambda_c \left(\Lambda \right) = [b \log(\Lambda/m)]^{-1}, \qquad (3.7)$$

it has a pair of zeros at $w = \pm w_0$, where w_0 is positive by convention. The function $\phi(w)$ has no positive zero for $\lambda < \lambda_c$. Note that the critical coupling λ_c (Λ) goes to 0 as the Pauli-Villars cutoff parameter tends to infinity.

Let us consider Eq. (3.1). We know that any solution $\rho(w)$ vanishes for |w| < m. Furthermore, for $m \le |w| < 2m$, the values of w' in the integrals on the right-hand side of (3.1) satisfy |w'| < m, for which $\rho(w')$ is 0. Consequently, $\rho(w)$ actually vanishes for |w| < 2m. In fact, one may iterate this procedure to show that $\rho(w)$ vanishes for all w. The conclusion is unavoidable if one requires the spectral density $\rho(w)$ to be a bounded continuous function. However, it is also possible for $\rho(w)$ to have delta distributions at those values of w for which $\phi(w)$ is 0; $\phi(w)\rho(w)$ and the integrals in (3.1) would then be ordinary functions.¹² Therefore, location of the zeros of $\rho(w)$ is a crucial ingredient in solving (3.1).

Let us set the mass scale by taking m = 1, and define the function

$$\lambda_0(w_0, \Lambda) = [\psi(w_0, 1) - \psi(w_0, \Lambda)]^{-1}.$$
 (3.8)

For coupling strength $\lambda = \lambda_0(w_0, \Lambda)$ with cutoff Λ , the function $\phi(\pm w_0)$ is 0, provided that λ_0 is greater than the critical coupling $\lambda_c(\Lambda)$; cf. Eq. (3.2). We find that $\lambda_0(w_0, \Lambda)$ is monotonically decreasing in w_0 for fixed Λ . The function $\lambda_0(w_0, \Lambda)$ is plotted against w_0 in Figs. 1(a) (Feynman gauge) and 1(b) (Landau gauge). Note that w_0 goes to 0 as λ approaches the critical coupling $\lambda_c(\Lambda)$ from above.

In Figs. 2(a) (Feynman gauge) and 2(b) (Landau gauge), w_0 is plotted against the cutoff parameter Λ for various choices of coupling strength λ_0 . There is an approximate linear relation between Λ and w_0 , which can be understood by examining the integral (3.3) in the parameter regime $M^2 \gg \Lambda^2 \gg w_0^2 \gg m^2 = 1$, to obtain

$$\psi(w_0, 1) - \psi(w_0, \Lambda) \simeq 6 \log(\Lambda/w_0) + 2$$
 (3.9a)

in Feynman gauge, and

$$\psi(w_0, 1) - \psi(w_0, \Lambda) \simeq 3 \log(\Lambda/w_0) - \frac{3}{4}$$
 (3.9b)

in Landau gauge. Correspondingly, we get



FIG. 1. λ_0 vs w_0 for various values of the cutoff Λ : (a) Feynman gauge, (b) Landau gauge.

$$\Lambda \simeq w_0 \exp\left[\frac{1}{b\lambda} + \frac{9-2b}{12}\right],\tag{3.10}$$

where b is 6 in Feynman gauge and 3 in Landau gauge, as in Eq. (3.5) above. These relations are valid to within a few percent for both gauges. Note, in particular, that for fixed λ , w_0 tends to infinity with Λ .

Let us look for solutions of Eq. (3.1) of the form



FIG. 2. w_0 versus cutoff parameter Λ for various values of coupling strength: (a) Feynman gauge, (b) Landau gauge.

$$\rho(w) = A\delta(w - w_0) + B\delta(w + w_0) + \sigma(w), \quad (3.11)$$

for constants A and B chosen arbitrarily, with the function $\sigma(w)$ to be determined. We obtain

$$\phi(w)\sigma(w) = \lambda\theta(|w| - m - w_0) \left[A \frac{\Omega(w, w_0, m)}{w - w_0} + B \frac{\Omega(w, - w_0, m)}{w + w_0} \right]$$
$$+ \lambda\theta(|w| - m) \int_{-|w| + m}^{|w| - m} dw'$$
$$\times \sigma(w') \frac{\Omega(w, w', m)}{w - w'} - [m \to \Lambda].$$
(3.12)

The terms involving A and B vanish when $|w| < m + w_0$, and so $\sigma(w) = 0$ for $|w| < m + w_0$, by the argument given before. Hence we may replace $\theta(|w| - m)$ in Eq. (3.12) by $\theta(|w| - m - w_0)$. Indeed, it may be sharpened to $\theta(|w| - 2m - w_0)$, since |w'| can exceed $m + w_0$ only when $|w| > 2m + w_0$. Now $\sigma(w)$ may be evaluated in successive steps of width m; for $m + w_0 < |w| < 2m + w_0$, it is given by just the first line of Eq. (3.12), for $2m + w_0|w| < 3m + w_0$ the integral contributes, but it involves only the domain $m + w_0 < |w'| < 2m + w_0$, for which $\sigma(w')$ is already known, and so on. In short, Eq. (3.12) is not a true equation for $\sigma(w)$; it is rather a progressive algorithm for evaluating $\sigma(w)$ to arbitrarily large w values, and the existence of a solution, parametrized by A and B, is assured.

The solution implies the following Lehmann representation for the quark propagator:

$$S'_{F}(p) = \frac{(A+B)\not p + (A-B)w_{0}}{p^{2} - w_{0}^{2}} + \int_{\infty}^{\infty} dw' \frac{\theta(|w'| - m - w_{0})\sigma(w')}{\not p - w'}.$$
(3.13)

There is a pole in the quark propagator at $p^2 = w_0^2$, but chiral symmetry actually remains unbroken when A = B. To verify the latter point, note that $\phi(w)$ is odd, the inhomogeneous term in (3.12) is an odd function of w when A = B, and the solution $\sigma(w)$ comes out as an even function of w. Thus the quark propagator (3.13) is proportional to p and chiral symmetry is preserved. The choice $A \neq B$ leads to a propagator that breaks chiral symmetry.

In the continuum limit $(\Lambda \rightarrow \infty)$, the pole in the quark propagator at $p^2 = w_0^2$ disappears to infinity. However, the parameters A and B can also be taken to depend upon Λ . In the continuum limit, the spectral density σ vanishes, and the renormalized quark propagator is of the form

$$S'_{F}(p) = ap + b.$$
 (3.14)

The subtraction constants a and b, while not determined in the Salam-Delbourgo formalism, must be set to 0 on physical grounds. Consequently, only the trivial solution $S'_F(p) = 0$ survives in the continuum limit.

Finally, we turn to the zero of $\phi(w)$ at w = 0. There is another solution of (3.1) of the form

$$\rho(w) = \delta(w) + \sigma(w), \qquad (3.15)$$

where $\sigma(w)$ satisfies Eq. (3.12) with replacements $w_0 = 0$ and A + B = 1. The inhomogeneous term in (3.12) is proportional to

$$\lambda \theta(|w|-m)\epsilon(w)(1-m^2/w^2)^2 \qquad (3.16)$$

in both gauges, and the solution $\sigma(w)$ is an even function of w. Consequently, the quark propagator becomes

$$S'_{F}(p) = \frac{p}{p^{2}} + 2p \int_{m}^{\infty} dw' \frac{\sigma(w')}{p^{2} - w'^{2}}.$$
 (3.17)

The quark propagator has a pole at $p^2 = 0$, and chiral symmetry remains unbroken at finite Λ , as well as in the continuum limit.

In summary, we have found two classes of solutions: in one class there may be a breaking of chiral symmetry with a cutoff present, but such solutions trivialize in the continuum limit; whereas the other class corresponds to unbroken chiral symmetry. Chiral symmetry thus remains unbroken in the continuum limit in the Salam–Delbourgo formalism. These conclusions, which are established here in both Feynman and Landau gauges, are shown in Ref. 13 to apply also in the "Landau-like" gauge proposed by Maskawa and Nakajima in Ref. 2 and treated by us in Ref. 4.

ACKNOWLEDGMENTS

This work has been partially supported by the National Science Foundation. One of us (PWJ) would like to thank the Stichting FOM (Fundamenteel Onderzoek der Materie), financially supported by the Nederlandse Organisatie voor Zuiver Wetenschappelijk Onderzoek, for its support.

- ¹Y. Nambu and G. Jona-Lasinio, Phys. Rev. 122, 345 (1961).
- ²T. Maskawa and H. Nakajima, Prog. Theor. Phys. **52**, 1376 (1974); **54**, 860 (1975).
- ³D. Atkinson and P. W. Johnson, Phys. Rev. D 35, 1943 (1987).
- ⁴D. Atkinson and P. W. Johnson, J. Math. Phys. 28, 2488 (1987).
- ⁵D. Atkinson, J. Math. Phys. 28, 2494 (1987).
- ⁶P. Fomin and V. Miransky, Sov. J. Part. Nucl. 16, 203 (1985).
- ⁷K. Higashijima, Phys. Rev. D 29, 1228 (1984).
- ⁸D. Atkinson, P. W. Johnson, and M. Koopmans, Z. Phys. C 34, 99 (1987).
- ⁹A. Salam, Phys. Rev. **130**, 1287 (1963); R. Delbourgo and A. Salam, *ibid*. **135**, 1398 (1963).
- ¹⁰D. Atkinson and D. W. E. Blatt, Nucl. Phys. B **151**, 342 (1979); D. Atkinson and H. Slim, Nuovo Cimento A **50**, 555 (1979).
- ¹¹R. Delbourgo and B. W. Keck, J. Phys. G. 6, 275 (1980).
- ¹²G. R. Bart and R. L. Warnock, SIAM J. Math. Anal. 4, 609 (1973).
- ¹³A. Hulsebos, "Chiral symmetry breakdown and the Salam-Delbourgo Ansatz," Groningen internal report No. 219, 1987.

The concept of streamfunctions in the Wien effect on a weak electrolyte

Douglas K. McIlroy and David P. Mason 6 Brandy Court, Calamvale 4116, Australia

(Received 14 January 1987; accepted for publication 13 August 1987)

It is shown that streamfunctions $g^{(k)}$, k = 1,2, can be defined to describe the two ionic states of a weakly dissociated electrolyte in the presence of a uniform applied electric field. These states comprise the dissociated state and the associated state and the $g^{(k)}$ are shown to be governed by a second-order partial differential equation. By consideration of characteristic numbers describing the two ionic states this equation is solved using similarity solutions. The association rate constant and relative increase in the dissociation constant due to the applied electric field are shown to have simple expressions in terms of the $g^{(k)}$. It is suggested that the streamfunctions are more natural functions with which to describe the mathematical properties of a weak electrolyte than the corresponding distribution functions. Some important mathematical properties of the streamfunctions are discussed.

I. INTRODUCTION

Existing mathematical theories of the association and dissociation of a weak electrolyte in the presence of a uniform applied electric field X (Refs. 1 and 2), i.e., of the so-called Wien effect, are in terms of the distribution functions $f^{(k)}(r,\theta)$, k = 1,2, describing oppositely charged pairs of ions (i, j pairs), where r is the distance of separation of the ions in a pair and θ is the angle between \mathbf{r} and \mathbf{X} . The superscripts refer to the two ionic states that exist in weak electrolytes. These are the state of complete dissociation denoted by superscript 1 and the associated state denoted by superscript 2. In these theories solutions for the $f^{(k)}$ are sought from

$$\operatorname{div} f^{(k)} \mathbf{v}^{(k)} = 0 \tag{1.1}$$

that are the equations of continuity in the steady state of ionic flux, where $\mathbf{v}^{(k)}$ is the mean relative velocity of an *i* ion relative to the *j* ion of an *i*, *j* pair.

In Sec. II we discuss how the above-mentioned theories do not take full advantage of Eq. (1.1) by showing how these equations permit the introduction of the concept of streamfunctions $g^{(k)}(r,\theta)$ to describe the two ionic states of the weak electrolyte. Simple expressions for the association rate constant A and the relative increase in the dissociation constant K(X)/K(0) are deduced in terms of the $g^{(k)}$. We derive the partial differential equation governing these streamfunctions in Sec. III and in Sec. IV we deduce possible characteristic numbers for the two ionic states of the weak electrolyte. These numbers are employed in Sec. V in solving by similarity solutions the governing partial differential equation for the $g^{(k)}$. Finally both A and K(X)/K(0) are evaluated.

The originality of this work lies in the concept of the $g^{(k)}$ to describe the mathematical properties of a weak electrolyte, in the introduction of their governing partial differential equation, and in the solutions of this equation employing numbers characteristic of the weak electrolyte.

II. STREAMFUNCTIONS AND STREAMLINES

The law of mass action gives the equation for the time rate of change of associated ions in a weak electrolyte as

$$\frac{dv_{ij}}{dt} = An_i n_j - KAv_{ij}, \qquad (2.1)$$

where v_{ij} is the concentration of associated ions and n_i , n_j are the concentrations of dissociated ions. For a weak electrolyte $n_i, n_j \ll v_{ij}$, which implies, that although n_i , n_j are functions of the applied field intensity X, v_{ij} is independent of X.

In the steady state $dv_{ij}/dt = 0$ and the equation of continuity of ionic flux leads to Eq. (1.1).³ The mean relative velocity of an *i* ion relative to a *j* ion is given by

$$\mathbf{v}^{(k)} = -\operatorname{grad} \Phi - (1/f^{(k)})\operatorname{grad} f^{(k)}, \quad k = 1, 2,$$
 (2.2)
where

$$\Phi = -1/r - 2\epsilon r \cos\theta \tag{2.3}$$

is the total electric potential of an *i*, *j* pair, $\epsilon = 2\beta q$,

$$\beta = \frac{(\omega_i e_i - \omega_j e_j)X}{2kT(\omega_i + \omega_j)}, \quad q = -\frac{e_i e_j}{2DkT} > 0,$$

where ω_i , ω_j are the mobility coefficients of the *i*, *j* ions, e_i , e_j their charges, *k* is Boltzmann's constant, *T* is absolute temperature, and *D* is the dielectric constant of the medium. To be specific we take $e_i > 0$ and $e_j < 0$ so that $\beta > 0$ and we take the origin at the position of the *j* ion of an *i*, *j* pair. In Eq. (2.3) the coordinate *r* and potential Φ have been nondimensionalized by division by the characteristic length 2q and by the characteristic potential kT, respectively. The *i*, *j* ions are assumed to be point ions and we further assume that the screening effects of the ionic atmospheres can be neglected in the region of interest which is the innermost region of the field of the ions. This is a valid assumption if n_i and n_j are sufficiently small to ensure that $\kappa^{-1} \ge q$, where

$$\kappa^{-1} = \{DkT/4\pi (n_i e_i^2 + n_j e_i^2)\}^{1/2}.$$
(2.4)

Here κ^{-1} is the Debye-Hückel radius of the ionic atmospheres and q is the Bjerrum association distance.

Since the flow pattern of ions in a weak electrolyte is clearly axisymmetric about the direction of X we suppose that there are functions $g^{(k)}$, k = 1,2, such that

$$r^{2}\sin\theta f^{(k)}v_{r}^{(k)} = -\frac{\partial g^{(k)}}{\partial\theta}, \quad r\sin\theta f^{(k)}v_{\theta}^{(k)} = \frac{\partial g^{(k)}}{\partial r}.$$
(2.5)

Evidently Eq. (1.1) is satisfied by this set and we call the $g^{(k)}(r,\theta)$ the streamfunctions. Similarly to the well-known definition of a streamfunction from fluid mechanics, $g^{(k)}$ has the following definition as a physical quantity: join a point Q, not coincident with the origin O, situated on the axis of symmetry Z'OZ to a point P whose cordiantes are r,θ by a curve that lies in the plane of P and the axis. (See Fig. 1). In the steady state, the value of $2\pi g^{(k)}(r,\theta)$ at P is defined to be the average flux measured relative to O per unit time of i ions flowing across the surface of revolution generated by rotating the curve QP about Z'OZ. The positive direction of the flux is away from the axis.

The functions $g^{(k)}$ are clearly independent of the choice of the axial curve connecting Q to P and hence are functions of positions only. Furthermore, there is no average flux of iions across the axis of symmetry and therefore the $g^{(k)}$ are independent of the position of Q on Z'OZ. When P coincides with Q the average flux of i ions relative to O will vanish and therefore

$$g(Q) = 0.$$
 (2.6)

We will choose Q to be on the ray $\theta = \pi, r > 0$ and since there is no average flux of *i* ions across this ray, the $g^{(k)}$ are constant along $\theta = \pi$ and therefore from Eq. (2.6),

$$g^{(k)}(r,\pi) = 0, \quad 0 < r < \infty.$$
 (2.7)

The streamfunctions lead naturally to the concept of *streamlines* for dissociated ions and associated ions: a streamline is a line in an axial plane that at any instant is everywhere parallel to the average relative velocity vector $\mathbf{v}^{(k)}$. The average flux of *i* ions across a streamline therefore vanishes and so along a streamline $g^{(k)}(r,\theta)$ can at most be a function of time. For a steady state this function of time must be a constant and the equation of a streamline reduces to

$$g^{(k)}(r,\theta) = \text{const}, \quad k = 1,2.$$
 (2.8)

In the steady state the average flow field is the same for all time and a streamline defined by Eq. (2.8) therefore coincides with the average trajectory of an *i* ion relative to a *j* ion at *O*. The surface of revolution generated by a streamline rotated about the axis of symmetry is called a *streamsurface*. On a streamsurface, $g^{(k)} = \text{const.}$ It is readily shown from Eq. (2.5) that

$$\mathbf{v}^{(k)} \cdot \operatorname{grad} g^{(k)} = 0, \quad k = 1, 2,$$
 (2.9)

which confirms that the flow field lies on the streamsurface.



FIG. 1. Streamfunctions in a weak electrolyte.

$$An_{i}(X)n_{j}(X) = -\int_{S} f^{(1)}\mathbf{v}^{(1)} d\mathbf{S}, \qquad (2.10)$$

where S is any closed surface completely surrounding the j ion at the origin. But the distribution function describing dissociated ions is the constant distribution¹ $f^{(1)}(r,\theta) = n_i(X)n_j(X)$ so on normalizing with respect to the characteristic distribution $n_i(X)n_j(X)$, $f^{(1)}(r,\theta) = 1$ and Eq. (2.10) gives

$$A = -\int_{S} \mathbf{v}^{(1)} \cdot d\mathbf{S}.$$
 (2.11)

The association constant is independent of the choice of $S^{3,4}$. If we therefore choose for S a closed surface of revolution generated by any axial curve joining any point Q not coincident with O on the ray $\theta = \pi$ and any point P(r,0) with r > 0on the ray $\theta = 0$, it follows immediately from the definition of $g^{(1)}$ that

$$\mathbf{A} = -2\pi g^{(1)}(\mathbf{r}, 0). \tag{2.12}$$

Again from Eq. (2.1) on normalizing $f^{(2)}(r,\theta)$ by dividing by the constant distribution function $n_i(0)n_i(0)$,

$$\frac{K(X)}{K(0)} = \frac{1}{A} \int_{S} f^{(2)} \mathbf{v}^{(2)} \cdot d\mathbf{S},$$
(2.13)

where S is any closed surface surounding the origin. From Eq. (1.1) it follows that K(X)/K(0) is independent of the choice of S. We choose for S the closed surface of revolution chosen above for the association process. It follows from the definition of streamfunction that

$$K(X)/K(0) = (2\pi/A)g^{(2)}(r,0).$$
(2.14)

The simplicity of both Eqs. (2.12) and (2.14) lends strength to the view that the natural functions to describe the mathematical properties of the Wien effect on weak electrolytes are the streamfunctions $g^{(k)}$ rather than the distribution functions.

III. THE GOVERNING PARTIAL DIFFERENTIAL EQUATIONS

We next derive the partial differential equation governing $g^{(k)}(r,\theta)$ k = 1,2. From Eqs. (2.2), (2.3), and (2.5)

$$r^{2}\sin\theta \frac{\partial f^{(k)}}{\partial r} - \frac{\partial g^{(k)}}{\partial \theta} + \sin\theta (1 - 2\epsilon r^{2}\cos\theta) f^{(k)} = 0,$$
(3.1)

$$\sin\theta \frac{\partial f^{(k)}}{\partial\theta} + \frac{\partial g^{(k)}}{\partial r} + 2\epsilon r \sin^2\theta f^{(k)} = 0.$$
(3.2)

On eliminating $f^{(k)}$ from these equations we obtain

$${}^{2}\frac{\partial^{2}g^{(k)}}{\partial r^{2}} + \frac{\partial^{2}g^{(k)}}{\partial \theta^{2}} + (1 - 2\epsilon r^{2}\cos\theta)\frac{\partial g^{(k)}}{\partial r} + (2\epsilon r\sin\theta - \cot\theta)\frac{\partial g^{(k)}}{\partial \theta} = 0, \quad k = 1,2.$$
(3.3)

The boundary conditions for dissociated ions are

f'

$$^{(1)}(r,\theta) = 1,$$
 (3.4)

$$g^{(1)}(r,\pi) = 0, \tag{3.5}$$

from Eq. (2.7) and for associated ions

$$\lim_{r \to 0} e^{-1/r} f^{(2)}(r,\theta) = 1,$$
(3.6)

$$\lim_{r \to \infty} f^{(2)}(r,\theta) = 0, \tag{3.7}$$

$$g^{(2)}(r,\pi) = 0, (3.8)$$

from Eq. (2.7). Equation (3.6) is a manifestation of the Maxwell-Boltzmann equilibrium distribution at the distance of closest approach of an *i*, *j* pair and Eq. (3.7) represents the requirement that an *i*, *j* pair must be separated to infinity for complete dissociation.

IV. CHARACTERISTIC NUMBERS FOR WEAK ELECTROLYTES

The total electrical potential of an *i*, *j* pair is the sum of the Coulomb potential and the potential due to the applied electric field and is given by Eq. (2.3) that has the reference direction, θ_0 , for potential, $\theta_0 = \pi/2$. For given values of ϵ and θ , Φ will attain a maximum value at $r = r_{\text{max}}$, say, which is given by $r_{\text{max}} = (2\epsilon \cos \theta)^{-1/2}$. The length r_{max} is finite $0 \le \theta < \pi/2$ but is complex for $\pi/2 < \theta \le \pi$ and would therefore not lead to a satisfactory characteristic length for the weak electrolyte.

This difficulty may be overcome by introducing an effective potential Φ_e given by

$$\Phi_e(r,\theta) = -1/r - \epsilon r(\cos\theta + 1), \qquad (4.1)$$

which has the same value as Φ at $\theta = 0$ and has a maximum value for given values of ϵ and θ at $r = r_{max}$, where

$$r_{\max} \triangleq (\epsilon(\cos\theta + 1))^{-1/2}. \tag{4.2}$$

The reference direction for Φ_e is $\Phi_e = \pi$, which, unlike the reference direction for Φ , has the virtue that it corresponds to a streamline for both associated and dissociated ions. With this definition r_{\max} is finite for $0 \le \theta < \pi$ having the minimum value of $(2\epsilon)^{-1/2}$ at $\theta = 0$ and increases monotonically as θ increases from 0. Thus Eq. (4.2) may lead to a suitable radial characteristic length $2qr_{\max}$ for the weak electrolyte. Accordingly we next transform the *r* coordinate

$$\bar{r} \triangleq r/r_{\rm max} = r(\epsilon y)^{1/2}, \tag{4.3}$$

where $y \triangleq 1 + \cos \theta$.

For associated ions we shall find that the function to work with is not $g^{(2)}(r,\theta)$ but is that obtained from this function by removing the factor $\exp[\epsilon y(y-2)]$. We therefore define G(r,y) such that

$$G(r,y) \triangleq g^{(2)}(r,\theta) \exp[-\epsilon r(y-2)]. \tag{4.4}$$

If we similarly define

$$F(r,y) \triangleq f^{(2)}(r,\theta) \exp[-\epsilon r(y-2)], \qquad (4.5)$$

then from Eq. (2.2)

$$\mathbf{v}^{(2)} = -\operatorname{grad} \Phi_e - (1/F) \operatorname{grad} F, \qquad (4.6)$$

which shows that F(r,y) is the distribution function associated with the potential Φ_e . Furthermore, from the first of Eqs. (2.5),

$$F\frac{\partial\Phi_e}{\partial r} + \frac{\partial F}{\partial r} = \frac{-\epsilon}{r}G - \frac{1}{r^2}\frac{\partial G}{\partial y}.$$
 (4.7)

If we make the assumption that F is an equilibrium distribution $F \propto \exp(-\Phi_e)$, and Eq. (4.7) gives $\partial G / \partial y = -G / (y/\epsilon)^{1/2}$ when $r = r_{\max}$. This implies that a suitable characteristic number for the y coordinate may be

$$y_{\rm ch} = (y/\epsilon)^{1/2} \tag{4.8}$$

and thus for dissociating ions we obtain a coordinate companion to \overline{r} as

$$\bar{y} \triangleq y/y_{\rm ch} = (\epsilon y)^{1/2}. \tag{4.9}$$

V. SOLUTIONS FOR THE STREAMFUNCTIONS

On setting $g^{(k)}(r,\theta) = h^{(k)}(r,y)$, Eq. (3.3) becomes

$$r^{2} \frac{\partial^{2} h^{(k)}}{\partial r^{2}} + y(2-y) \frac{\partial^{2} h^{(k)}}{\partial y^{2}} + (1 - 2\epsilon r^{2}(y-1)) \frac{\partial h^{(k)}}{\partial r} - 2\epsilon r y(2-y) \frac{\partial h^{(k)}}{\partial y} = 0, \quad k = 1, 2.$$
(5.1)

A. Dissociated ions

The boundary conditions (3.4) and (3.5) become

$$f^{(1)}(\mathbf{r},\theta) = 1,$$
 (5.2)

$$h^{(1)}(r,0) = 0.$$
 (5.3)

We shall solve Eq. (5.1) for $h^{(1)}(r,y)$ using the similarity transformation indicated by Eq. (4.3). By writing

$$h^{(1)}(r,y) = H^{(1)}(\bar{r},y),$$
 (5.4)
Eq. (5.1) gives

$$\frac{\overline{r}^{2}(2+3y)}{4y} \frac{\partial^{2} H^{(1)}}{\partial \overline{r}^{2}} + \left(\frac{-\overline{r}(2-y)}{4y} + (\epsilon y)^{1/2}(1-\overline{r}^{2})\right) \frac{\partial H^{(1)}}{\partial \overline{r}} + \overline{r}(2-y) \frac{\partial^{2} H^{(1)}}{\partial \overline{r} \partial y} + y(2-y) \frac{\partial^{2} H^{(1)}}{\partial y^{2}} - 2(\epsilon y)^{1/2} \overline{r}(2-y) \frac{\partial H^{(1)}}{\partial y} = 0.$$
(5.5)

This equation is similar to the example discussed by Hansen⁵: if we set

$$H^{(1)}(\bar{r}, y) = F(\bar{r})(2 - y), \qquad (5.6)$$

Eq. (5.5) gives

$$\frac{\overline{r}^{2}(2+3y)}{4y}F'' + \left(\frac{-\overline{r}(2+3y)}{4y} + (\epsilon y)^{1/2}(1-\overline{r}^{2})\right)F' + 2(\epsilon y)^{1/2}\overline{r}F = 0,$$
(5.7)

which possesses only one solution, $F(\bar{r}) = 1 - \bar{r}^2$, analytic at the irregular singular point $\bar{r} = 0$. Thus $H^{(1)}(\bar{r}, y)$ $= (1 - \bar{r}^2)(2 - y)$ is a solution of Eq. (5.5) and the required solution of this equation is therefore

$$H^{(1)}(\bar{r}, y) = B_1(1 - \bar{r}^2)(2 - y) + B_2,$$
(5.8)

where B_1 and B_2 are constants. The corresponding solution of Eq. (5.1) is

$$h^{(1)}(r,y) = B_1(1 - \epsilon r^2 y)(2 - y) + B_2.$$
 (5.9)

The boundary condition (5.3) requires that $B_2 = -2B_1$, and from Eq. (3.1) or (3.2) and Eq. (5.2) we find that $B_1 = 1$ and so

and

 $g^{(1)}(r,\theta) = -(1+\cos\theta)(1+\epsilon r^2(1-\cos\theta)) \quad (5.11)$

is the streamfunction describing dissociated ions.

 $h^{(1)}(r,y) = (1 - \epsilon r^2 y)(2 - y) - 2$

Next consider the way in which association depends on the angle of association. From the definition, $2\pi g^{(1)}(r,\theta)$ is the averge flux of *i* ions in a direction away from the axis of symmetry across the surface of revolution generated by any axial curve joining the reference point Q on the ray $\theta = \pi$ to the point $P(r,\theta)$. This flux depends on *r* and θ since the surface is not closed. Let us constrain *P* to vary only along a streamline that intersects the origin. Then since $g^{(1)}$ is constant along a streamline $g^{(1)}(r,\theta) = g^{(1)}(0,\alpha)$, where α is the angle which the streamline makes with X at $r = 0, 0 \le \alpha \le \pi$. If we define $\overline{A}(\alpha)$ to be the average association flux across the surface of revolution generated by any axial curve joining Q to any point *P* on the streamline making angle α with X at r = 0, then

$$\overline{A}(\alpha) = -2\pi g^{(2)}(0,\alpha) = 2\pi (1 + \cos \alpha)$$
 (5.12)

from Eqs. (2.12) and (5.11). Here $\overline{A}(\alpha)$ increases monotonically as α decreases from π to 0, which is consistent with the existence of a sink singularity at r = 0. We note that $\overline{A}(\alpha)$ is independent of ϵ , which is a generalization of Langevin's theorem⁴ that states that the association constant A is independent of ϵ . From Eqs. (2.12) and (5.11), or Eq. (5.12),

$$A = \overline{A}(0) = 4\pi. \tag{5.13}$$

B. Associated ions

We have to solve Eq. (5.1) with k = 2 subject to boundary conditions (3.6), (3.7), and Eq. (3.8) becomes

$$h^{(2)}(r,0) = 0.$$
 (5.14)

We first observe that if we define p(r,y) by

$$h^{(2)}(r,y) = p(r,y) \exp[\epsilon r(y-1)], \qquad (5.15)$$

Eq. (5.1) is rendered in separable form,

$$r^{2} \frac{\partial^{2} p}{\partial r^{2}} + \frac{\partial p}{\partial r} - \epsilon^{2} r^{2} p$$

= $-y(2-y) \frac{\partial^{2} p}{\partial y^{2}} - \epsilon(y-1)p,$ (5.16)

which on setting p(r,y) = R(r)Y(y) gives

$$r^{2} \frac{d^{2}R}{dr^{2}} + \frac{dR}{dr} - (\epsilon^{2}r^{2} + \lambda)R = 0,$$

$$y(2 - y) \frac{d^{2}y}{dv^{2}} + (\epsilon(y - 1) + \lambda)Y = 0,$$
(5.17)

where λ is the separation constant. Thus

$$\frac{d^2R}{dr^2} + \frac{1}{r^2}\frac{dR}{dr} - \epsilon^2 R = 0 \quad \text{as } r \to \infty, \qquad (5.18)$$

from which

3002

. . _

$$R \sim C_1 e^{\epsilon r} + C_2 e^{-\epsilon r} \quad \text{as } r \to \infty, \tag{5.19}$$

where C_1 and C_2 are constants. But $\lim_{r\to\infty} rf^{(2)}(r,\theta)$ = exp[$\epsilon r(\cos \theta - 1)$] (Ref. 6) and so from Eq. (3.2) C_1 must be 0 giving

$$h^{(2)}(r,y) \sim \exp[\epsilon r(y-2)]$$
 as $r \to \infty$. (5.20)

Since this function depends *explicitly* on ϵ when we transform to the coordinates \bar{r}, \bar{y} defined by Eqs. (4.3) and (4.9), we define $H^{(2)}(\bar{r}, \bar{y})$ by

$$h^{(2)}(r,y) \triangleq H^{(2)}(\bar{r},\bar{y}) \exp[\epsilon r(y-2)],$$
 (5.21)

whence Eq. (5.1) gives

(5.10)

$$\frac{\overline{r}^{2}}{\overline{y}^{2}} \frac{\partial^{2} H^{(2)}}{\partial \overline{r}^{2}} - \left(\frac{\overline{r}}{2\overline{y}^{2}} + \frac{2\overline{r}^{2}}{\overline{y}}\right) \frac{\partial H^{(2)}}{\partial \overline{r}} + \frac{\overline{r}}{\overline{y}} \frac{\partial^{2} H^{(2)}}{\partial \overline{r} \partial \overline{y}} - \frac{1}{2\overline{y}} \frac{\partial H^{(2)}}{\partial \overline{y}} + \frac{1}{2} \frac{\partial^{2} H^{(2)}}{\partial \overline{y}^{2}} - 2H^{(2)} + \epsilon^{-1} \left\{\frac{3}{4} \overline{r}^{2} \frac{\partial^{2} H^{(2)}}{\partial \overline{r}^{2}} + \left(\overline{y} + \frac{\overline{r}}{4}\right) \frac{\partial H^{(2)}}{\partial \overline{r}} - \frac{\overline{r} \overline{y}}{2} \frac{\partial^{2} H^{(2)}}{\partial \overline{r} \partial \overline{y}} + \frac{\overline{y}}{4} \frac{\partial H^{(2)}}{\partial \overline{y}} - \frac{\overline{y}^{2}}{4} \frac{\partial^{2} H^{(2)}}{\partial \overline{y}^{2}} + \overline{y}^{2} H^{(2)} \right\} = 0.$$
(5.22)

If we further assume that $H^{(2)}(\bar{r},\bar{y})$ is not an explicit function of ϵ then we must have

$$\frac{\overline{r}^{2}}{2\overline{y}^{2}}\frac{\partial^{2}H^{(2)}}{\partial\overline{r}^{2}} - \left(\frac{\overline{r}}{2\overline{y}^{2}} + \frac{2\overline{r}^{2}}{\overline{y}}\right)\frac{\partial H^{(2)}}{\partial\overline{r}} + \frac{\overline{r}}{\overline{y}}\frac{\partial^{2}H^{(2)}}{\partial\overline{r}\partial\overline{y}} - \frac{1}{2\overline{y}}\frac{\partial H^{(2)}}{\partial\overline{y}} + \frac{1}{2}\frac{\partial^{2}H^{(2)}}{\partial\overline{y}^{2}} - 2H^{(2)} = 0$$
(5.23)

and

$$\frac{3}{4}\overline{r}^{2}\frac{\partial^{2}H^{(2)}}{\partial\overline{r}^{2}} + \left(\overline{y} + \frac{\overline{r}}{4}\right)\frac{\partial H^{(2)}}{\partial\overline{r}} - \frac{\overline{r}\overline{y}}{2}\frac{\partial^{2}H^{(2)}}{\partial\overline{r}\partial\overline{y}} + \frac{\overline{y}}{4}\frac{\partial H^{(2)}}{\partial\overline{y}} - \frac{\overline{y}^{2}}{4}\frac{\partial^{2}H^{(2)}}{\partial\overline{y}^{2}} + \overline{y}^{2}H^{(2)} = 0.$$
(5.24)

First consider Eq. (5.23), which is a parabolic equation whose characteristic directions are given by

$$\frac{\overline{r}^2}{\overline{y}^2} \left(\frac{d\overline{y}}{d\overline{r}}\right)^2 - \frac{2\overline{r}}{\overline{y}} \frac{d\overline{y}}{d\overline{r}} + 1 = 0, \qquad (5.25)$$

and which has the solution $\overline{y} = \text{const } \overline{r}$. We therefore define the canonical coordinates

$$\xi(\bar{r},\bar{y}) = \bar{y}/\bar{r}, \quad \eta = \eta(\bar{r},\bar{y}), \quad (5.26)$$

and $q(\xi,\eta) = H^{(2)}(\overline{r},\overline{y})$ and Eq. (5.24) becomes

$$\begin{pmatrix} \frac{\bar{r}^2}{2\bar{y}^2} \left(\frac{\partial\eta}{\partial\bar{r}}\right)^2 + \frac{\bar{r}}{\bar{y}} \frac{\partial\eta}{\partial\bar{r}} \frac{\partial\eta}{\partial\bar{y}} + \frac{1}{2} \left(\frac{\partial\eta}{\partial\bar{y}}\right)^2 \right) \frac{\partial^2 q}{\partial\eta^2} \\ + 2 \frac{\partial q}{\partial\xi} + \left(\frac{\bar{r}^2}{2\bar{y}^2} \frac{\partial^2\eta}{\partial\bar{r}^2} - \left[\frac{\bar{r}}{2\bar{y}^2} + \frac{2\bar{r}^2}{\bar{y}}\right] \frac{\partial\eta}{\partial\bar{r}} \\ + \frac{\bar{r}}{\bar{y}} \frac{\partial^2\eta}{\partial\bar{r}\partial\bar{y}} - \frac{1}{2\bar{y}} \frac{\partial\eta}{\partial\bar{y}} - \frac{1}{2} \frac{\partial^2\eta}{\partial\bar{y}^2} \right) \frac{\partial q}{\partial\eta} - 2q = 0.$$

$$(5.27)$$

For a parabolic equation the choice of $\eta(\bar{r},\bar{y})$ is arbitrary. If we take $\eta = \bar{y}$, Eq. (5.27) reduces to

$$\frac{\partial^2 q}{\partial \bar{y}^2} - \frac{1}{\bar{y}} \frac{\partial q}{\partial \bar{y}} + 4 \frac{\partial q}{\partial \xi} - 4q = 0, \qquad (5.28)$$

and on setting $q(\xi, \overline{y}) = Z(\xi) Y(\overline{y})$ in this equation we obtain

$$\frac{d^2Y}{d\overline{y}^2} - \frac{1}{\overline{y}}\frac{dY}{d\overline{y}} - \gamma Y = 0$$
(5.29)

and

$$4\frac{dZ}{d\xi} + (\gamma - 4)Z = 0, \qquad (5.30)$$

where γ is the separation constant.

Equation (5.29) is solved by setting $Y(\bar{y}) = \bar{y}W(\bar{y})$ and it then becomes

$$\bar{y}^2 \frac{d^2 W}{d\bar{y}^2} + \bar{y} \frac{dW}{d\bar{y}} - (1 + \gamma \bar{y}^2) W = 0, \qquad (5.31)$$

which has the solution

$$W(\overline{y}) = \operatorname{const}_1 I_1(\gamma^{1/2}\overline{y}) + \operatorname{const}_2 K_1(\gamma^{1/2}\overline{y}), \quad (5.32)$$

where I_1 , K_1 are, respectively, the modified Bessel function of the first kind and modified Bessel function of the second kind both of order 1. Equation (5.30) has the solution

$$Z(\xi) = \operatorname{const} \exp[((1 - \gamma/4)\xi]], \qquad (5.33)$$

and thus the solution of Eq. (5.23) is

$$H^{(2)}(\bar{r},\bar{y}) = \bar{y} \exp[((1 - \gamma/4)\bar{y}/\bar{r}](D_1 I_1(\gamma^{1/2}\bar{y}) + D_2 K_1(\gamma^{1/2}\bar{y})), \qquad (5.34)$$

where D_1 and D_2 are constants. However $\overline{y}I_1(\gamma^{1/2}\overline{y}) = O(\overline{y}^2)$ as $\overline{y} \to 0$ and $\overline{y}K_1(\gamma^{1/2}\overline{y}) = O(\gamma^{-1/2})$ as $\overline{y} \to 0$ so in order to satisfy the boundary condition (5.14) we must have $D_2 = 0$ so that

$$H^{(2)}(\overline{r},\overline{y}) = D_1 \,\overline{y} \exp[(1-\gamma/4)\overline{y}/\overline{r}]I_1(\gamma^{1/2}\overline{y}). \quad (5.35)$$

Furthermore this solution satisfies Eq. (5.24) if and only if $\gamma = 4$. Thus the required solution to Eq. (5.22) is

$$H^{(2)}(\bar{r},\bar{y}) = D_1 \,\bar{y} I_1(2\bar{y}), \tag{5.36}$$

or in terms of the r, θ coordinates the required solution to Eq. (3.3) is

$$g^{(2)}(r,\theta) = D_1(2\epsilon)^{1/2} \cos(\theta/2) I_1((8\epsilon)^{1/2} \cos(\theta/2))$$
$$\times \exp[\epsilon r(\cos \theta - 1)]. \qquad (5.37)$$

The final problem in determining $g^{(2)}(r,\theta)$ is to evaluate D_1 . We have, from Eqs. (3.1), (3.2), and (5.37) using the identity⁷ $xI'_1(x) = -I_1(x) + xI_0(x)$,

$$\frac{\partial f^{(2)}}{\partial r} + \left(\frac{1}{r^2} - 2\epsilon \cos\theta\right) f^{(2)}$$

$$= -D_1 (2\epsilon)^{1/2} \left\{ \frac{(2\epsilon)^{1/2}}{2r^2} I_0 \left((8\epsilon)^{1/2} \cos\frac{\theta}{2} \right) + \frac{\epsilon}{r} \cos\frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos\frac{\theta}{2} \right) \right\}$$

$$\times \exp[\epsilon r (\cos\theta - 1)], \qquad (5.38)$$

$$\frac{\partial f^{(2)}}{\partial \theta} + 2\epsilon r \sin \theta f^{(2)}$$

$$= D_1 2^{1/2} \epsilon^{3/2} \sin \frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right)$$

$$\times \exp[\epsilon r (\cos \theta - 1)]. \qquad (5.39)$$

Next consider the Pfaffian differential equation arising from this pair of equations. This is

$$df^{(2)} = \left\{ \left(-\frac{1}{r^2} + 2\epsilon \cos \theta \right) f^{(2)} - D_1 (2\epsilon)^{1/2} \left[\frac{(2\epsilon)^{1/2}}{2r^2} I_0 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right) \right] + \frac{\epsilon}{r} \cos \frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right) \right] \\ \times \exp[\epsilon r (\cos \theta - 1)] dr \\ + \left\{ -2\epsilon r \sin \theta f^{(2)} + D_1 2^{1/2} \epsilon^{3/2} \sin \frac{\theta}{2} \right\} \\ \times I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right) \exp[\epsilon r (\cos \theta - 1)] d\theta.$$
(5.40)

This equation is shown to be integrable in the Appendix. We integrate using the method of Natani.⁸ First we treat the variable r as a constant and solve the resulting differential equation. Taking r as a constant, Eq. (5.40) becomes

$$\frac{df^{(2)}}{d\theta} + 2\epsilon r \sin \theta f^{(2)}$$

$$= D_1 2^{1/2} \epsilon^{3/2} \sin \frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right)$$

$$\times \exp[\epsilon r (\cos \theta - 1)], \qquad (5.41)$$

which implies that

$$f^{(2)}(r,\theta) = \exp[2\epsilon r \cos\theta] \left\{ D_1 2^{1/2} \epsilon^{3/2} \int \sin\frac{\theta}{2} \\ \times I_1 \left((8\epsilon)^{1/2} \cos\frac{\theta}{2} \right) \\ \times \exp[-\epsilon r (\cos\theta + 1)] d\theta + D_3(r) \right\},$$
(5.42)

where $D_3(r)$ is a function of r only.

The right-hand side of Eq. (5.42) may be integrated by the substitution

$$u = (8\epsilon)^{1/2} \cos \left(\frac{\theta}{2}\right) \tag{5.43}$$

and the identity

$$\frac{d}{du}(u^{n+1}I_{n+1}(u)) = u^{n+1}I_n(u)$$
(5.44)

for modified Bessel functions I_n of the first kind of order n. We find that

$$-(2\epsilon)^{1/2} \int \sin \frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right)$$

$$\times \exp\left[-\epsilon r(\cos \theta + 1) \right] d\theta$$

$$= e^{-ru^2/4} \left(I_0(u) + \frac{ru}{2} I_1(u) + \left(\frac{ru}{2}\right)^2 I_2(u) + \cdots + \left(\frac{ru}{2}\right)^n I_n(u) \right) + R_n(u), \quad (5.45)$$

where

$$R_n(u) = \left(\frac{r}{2}\right)^{n+1} \int u^{n+1} I_n(u) e^{-ru^2/4} du.$$
 (5.46)

The remainder can be evaluated as follows. Using the expansion

$$I_n(u) = \sum_{s=0}^{\infty} \frac{1}{s!(n+s)!} \left(\frac{u}{2}\right)^{n+2s}$$
(5.47)

and making the change of variable $v = ru^2/4$ we have

$$R_n = \sum_{s=0}^{\infty} \frac{1}{s!(n+s)!r^s} \int v^{n+s} e^{-v} dv.$$
 (5.48)

But by successive integration by parts

$$\int v^{n+s} e^{-v} dv = -e^{-v} (n+s)! \sum_{p=0}^{n+s} \frac{v^p}{p!}$$
(5.49)

and so

$$R_n = -e^{-\nu} \sum_{s=0}^{\infty} \frac{1}{s!r^s} \sum_{p=0}^{n+s} \frac{v^p}{p!}$$
(5.50)

and

$$\lim_{n \to \infty} R_n = -\sum_{s=0}^{\infty} \frac{1}{s! r^s} = -e^{1/r}.$$
 (5.51)

Thus

$$-(2\epsilon)^{1/2} \int \sin \frac{\theta}{2} I_1 \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right)$$

$$\times \exp[-\epsilon r (\cos \theta + 1)] d\theta$$

$$= \exp[-\epsilon r (\cos \theta + 1)] \sum_{p=0}^{\infty} \left(\frac{r(8\epsilon)^{1/2} \cos (\theta/2)}{2} \right)^p$$

$$\times I_p \left((8\epsilon)^{1/2} \cos \frac{\theta}{2} \right) - e^{1/r}, \qquad (5.52)$$

and Eq. (5.42) therefore gives $f^{(2)}(r,\theta)$

$$= -D_{1}\epsilon \exp[\epsilon r(\cos \theta - 1)]$$

$$\times \sum_{p=0}^{\infty} \left(\frac{r(8\epsilon)^{1/2} \cos (\theta/2)}{2}\right)^{p}$$

$$\times I_{p}\left((8\epsilon)^{1/2} \cos \frac{\theta}{2}\right) + D_{1}\epsilon \exp\left[\frac{1}{r} + 2\epsilon r \cos \theta\right]$$

$$+ D_{3}(r)\exp(2\epsilon r \cos \theta). \qquad (5.53)$$

We next set $\theta = \pi$ in Eq. (5.40), which becomes

$$\frac{df^{(2)}}{dr} + \left(\frac{1}{r^2} + 2\epsilon\right)f^{(2)} = -\frac{D_1\epsilon}{r^2}\exp(-2\epsilon r), \quad (5.54)$$

and which has the solution

$$f^{(2)}(r,\pi) = -D_1 \epsilon \exp(-2\epsilon r) + D_4 \exp(1/r - 2\epsilon r),$$
(5.55)

where D_4 is constant because θ has been given the *definite* value of $\theta = \pi$. Comparing Eq. (5.53) with $\theta = \pi$ and Eq. (5.55) gives

$$D_3(r) = (D_4 - \epsilon D_1)e^{1/r}$$
 (5.56)

and so

$$f^{(2)}(r,\theta) = -D_{1}\epsilon \exp[\epsilon r(\cos \theta - 1)]$$

$$\times \sum_{p=0}^{\infty} \left(\frac{r(8\epsilon)^{1/2} \cos (\theta/2)}{2}\right)^{p} I_{p}\left((8\epsilon)^{1/2} \cos \frac{\theta}{2}\right)$$

$$+ D_{4} \exp\left(\frac{1}{r} + 2\epsilon r \cos \theta\right) \qquad (5.57)$$

and the boundary condition at r = 0 [Eq. (3.6)] therefore gives $D_4 = 1$.

To determine D_1 we have first the identity⁸

$$\exp\left[\frac{x}{2}\left(s+\frac{1}{s}\right)\right] = \sum_{n=-\infty}^{\infty} s^n I_n(x).$$
(5.58)

Taking $x = (8\epsilon)^{1/2} \cos(\theta/2), s = r(8\epsilon)^{1/2} \cos(\theta/2)/2$, Eq. (5.58) gives

$$\sum_{n=0}^{\infty} \left(\frac{r(8\epsilon)^{1/2} \cos(\theta/2)}{2} \right)^n I_n \left((8\epsilon)^{1/2} \cos\frac{\theta}{2} \right)$$
$$= -\sum_{n=1}^{\infty} \left(\frac{2}{r(8\epsilon)^{1/2} \cos(\theta/2)} \right)^n I_n \left((8\epsilon)^{1/2} \cos\frac{\theta}{2} \right)$$
$$+ \exp\left[\frac{1}{r} + \epsilon r(\cos\theta + 1) \right]$$
(5.59)

and using this with $D_4 = 1$ Eq. (5.57) becomes

$$f^{(2)}(r,\theta) = D_1 \epsilon \exp[\epsilon r(\cos \theta - 1)]$$

$$\times \sum_{n=1}^{\infty} \left(\frac{2}{r(8\epsilon)^{1/2}\cos(\theta/2)}\right)^n$$

$$\times I_n\left((8\epsilon)^{1/2}\cos\frac{\theta}{2}\right)$$

$$+ (1 - \epsilon D_1)\exp\left(\frac{1}{r} + 2\epsilon r\cos\theta\right) \quad (5.60)$$

and the boundary condition (3.7) therefore requires that $D_1 = 1/\epsilon$ and so from Eqs. (5.37) and (5.57),

$$f^{(2)}(r,\theta) = -\exp[\epsilon r(\cos\theta - 1)] \\ \times \sum_{p=0}^{\infty} \left(\frac{r(8\epsilon)^{1/2}\cos(\theta/2)}{2}\right)^{p} \\ \times I_{p}\left((8\epsilon)^{1/2}\cos\frac{\theta}{2}\right) \\ + \exp\left(\frac{1}{r} + 2\epsilon r\cos\theta\right)$$
(5.61)

and

$$g^{(2)}(r,\theta) = \left(\frac{2}{\epsilon}\right)^{1/2} \cos\frac{\theta}{2} I_1\left((8\epsilon)^{1/2} \cos\frac{\theta}{2}\right)$$
$$\times \exp[\epsilon r(\cos\theta - 1)]. \tag{5.62}$$

The equivalence of Eq. (5.61) and Onsager's expression¹

$$f^{(2)}(r,\theta) = \frac{1}{r} \exp\left[\frac{1}{r} + \epsilon r(\cos\theta - 1)\right]$$
$$\times \int_0^1 I_0\left((8\epsilon)^{1/2}\cos\frac{\theta}{2}\right) e^{-s/r} ds \qquad (5.63)$$

may be easily demonstrated. Furthermore from Eqs. (2.14) and (5.13), Eq. (5.62) gives

$$\frac{K(X)}{K(0)} = \frac{1}{2} g^{(2)}(r,0) = \frac{I_1[(8\epsilon)^{1/2}]}{(2\epsilon)^{1/2}},$$
 (5.64)

which is Onsager's well-known result.

We further observe that since $\xi = \overline{y}/\overline{r} = 1/r$, the transformation $\overline{r} = r(\epsilon y)^{1/2}$ is unnecessary in the above analysis. The only necessary transformation is $\overline{y} = (\epsilon y)^{1/2}$, which suggests that the characteristic length for associated ions is the transverse length $2qy_{\rm ch}r_{\rm max} = 2q/\epsilon$.

Finally, the following important deduction may be made from Eq. (5.62). Let us define $4\pi \overline{K}(\theta_0;\epsilon)$ as the aver-

age flux of *i* ions away from the axis across the surface of revolution generated by any axial curve joining the reference point Q on the ray $\theta = \pi$ to any point P on the streamline that makes an angle θ_0 with X at r = 0. Suppose that P is constrained to vary only along this streamline. Since $g^{(2)}$ is constant along a streamline $g^{(2)}(P) = g^{(2)}(0,\theta_0)$ and therefore

$$\overline{K}(\theta_0;\epsilon) = \frac{1}{2}g^{(2)}(0,\theta_0)$$

= $\cos(\theta_0/2)I_1[(8\epsilon)^{1/2}\cos(\theta_0/2)]/(2\epsilon)^{1/2}.$
(5.65)

The function $\overline{K}(\theta_0;\epsilon)$ increases monotonically from 0 to K(X)/K(0) as θ_0 decreases from π to 0, which is consistent with the existence of a *source* (for associated ions) at the origin; $\overline{K}(\theta_0;\epsilon)$ is the contribution to K(X)/K(0) made by those trajectories emanating from the origin at angles from $\theta = \pi$ to $\theta = \theta_0$. The fractional contribution ν of this range of "initial angles" compared to K(X)/K(0) is from Eqs. (5.64) and (5.65) given by

$$v = \cos(\theta_0/2) I_1[(8\epsilon)^{1/2} \cos(\theta_0/2)] / I_1[(8\epsilon)^{1/2}].$$
(5.66)

Of special interest is the value of v, where $\theta_0 = \pi/2$. In this case v = 49.75% when $\epsilon = 0.01$ and v = 6% when $\epsilon = 10$. This shows that when the applied field intensity is very small the contribution to dissociation by the forward directions $(\theta_0 < \pi/2)$ is approximately equal to the contribution made by the reverse angles $(\theta_0 > \pi/2)$ but that when the applied field intensity is large, the contribution to dissociation made by the forward directions is much greater than the contribution made by the reverse directions.

APPENDIX: INTEGRABILITY OF THE PFAFFIAN DIFFERENTIAL EQUATION

We have from Eqs. (5.38) and (5.39),

$$df^{(2)} = \frac{\partial f^{(2)}}{\partial r} (f^{(2)}, r, \theta) dr + \frac{\partial f^{(2)}}{\partial \theta} (f^{(2)}, r, \theta) d\theta$$
$$= P(f^{(2)}, r, \theta) dr + Q(f^{(2)}, r, \theta) d\theta,$$
(A1)

say, where

$$P(f^{(2)},r,\theta) = (-1/r^{2} + 2\epsilon \cos \theta)f^{(2)} - D_{1}(2\epsilon)^{1/2} \exp[\epsilon r(\cos \theta - 1)] \times \{ [(2\epsilon)^{1/2}/2r^{2}]I_{0}((8\epsilon)^{1/2}\cos (\theta/2)) + (\epsilon/r)\cos(\theta/2)I_{1}((8\epsilon)^{1/2}\cos (\theta/2)) \}$$
(A2)

and

$$Q(f^{(2)}, r, \theta) = -2\epsilon r \sin \theta f^{(2)} + D_1 2^{1/2} \epsilon^{3/2} \exp[\epsilon r (\cos \theta - 1)] \times \sin(\theta/2) I_1((8\epsilon)^{1/2} \cos(\theta/2)).$$
(A3)

If we write Eq. (A1) as

$$P(f^{(2)}, r, \theta)dr + Q(f^{(2)}, r, \theta)d\theta - df^{(2)} = 0$$
 (A4)

and define

$$\mathbf{Y} \triangleq [\mathbf{P}, \mathbf{Q} - 1], \tag{A5}$$

then Eq. (A4) is integrable if and only if

$$\mathbf{Y} \cdot \mathbf{curl} \ \mathbf{Y} = \mathbf{0}, \tag{A6}$$

(Ref. 8), i.e., if and only if

$$-P\frac{\partial Q}{\partial f^{(2)}} + Q\frac{\partial P}{\partial f^{(2)}} + \frac{\partial P}{\partial \theta} - \frac{\partial Q}{\partial r} = 0.$$
 (A7)

Direct substitution of the derivatives of P and Q in Eq. (A7) confirms Eq. (A6).

- ¹L. Onsager, J. Chem. Phys. 2, 599 (1934).
- ²D. K. McIlroy and D. P. Mason, submitted to SIAM Rev.
- ³D. P. Mason and D. K. McIlroy, J. Chem. Soc., Far. Trans. II 74, 2019 (1978).
- ⁴P. Langevin, Ann. Chem. Phys. 28, 433 (1903).
- ⁵A. G. Hansen, Similarity Analysis of Boundary Value Problems in Engineering (Prentice-Hall, Englewood Cliffs, NJ, 1964), p. 2.
- ⁶D. P. Mason and D. K. McIlroy, Proc. R. Soc. London Ser. A 359, 303 (1978).
- ⁷C. J. Tranter, Bessel Functions with Some Physical Applications (English U. P., London, 1968).
- ⁸I. N. Sneddon, *Elements of Partial Differential Equations* (McGraw-Hill, New York, 1957).